PROVERBS

SBL

# Society of Biblical Literature



The Hebrew Bible: A Critical Edition

Ronald Hendel, General Editor

Number 1

# PROVERBS

# AN ECLECTIC EDITION WITH INTRODUCTION AND TEXTUAL COMMENTARY

Michael V. Fox

SBL Press Atlanta

# Copyright © 2015 by SBL Press

All rights reserved. No part of this work may be reproduced or transmitted in any form or by any means, electronic or mechanical, including photocopying and recording, or by means of any information storage or retrieval system, except as may be expressly permitted by the 1976 Copyright Act or in writing from the publisher. Requests for permission should be addressed in writing to the Rights and Permissions Office, SBL Press, 825 Houston Mill Road, Atlanta, GA 30329 USA.

Library of Congress Cataloging-in-Publication Data

Fox, Michael V., 1940- author.

Proverbs : an eclectic edition with introduction and textual commentary / by Michael V. Fox.

p. cm. — (The Hebrew bible: a critical edition ; no. 1) Includes bibliographical references and index. ISBN 978-1-62837-020-1 (hardcover binding : alk. paper) — ISBN 978-1-62837-021-8 (electronic format) 1. Bible. Proverbs—Commentaries. I. Title. BS1465.53.F69 2014 223'.7077—dc23 2014027666

> Printed on acid-free, recycled paper conforming to ANSI/NISO Z39.48-1992 (R1997) and ISO 9706:1994 standards for paper permanence.



# Contents

Series Foreword	ix
Preface	xi
Abbreviations	xiii
1. Is Text Criticism Possible?	1
1.1. The Challenge	1
1.2. The Goals	2
1.3. Theory and Practice	4
1.4. Authors and Editors	6
1.5. Text and Interpretation	9
1.6. Models of Editorial Practice	10
1.6.1. Shakespeare	11
1.6.2. Jane Austen	13
1.7. In Conclusion	14
2. The Hebrew Texts	17
2.1. Hebrew Resources	17
2.1.1. The Masoretic Text	17
2.1.1.1. The Leningrad Codex (M <sup>L</sup> )	17
2.1.1.2. The Aleppo Codex (M <sup>A</sup> )	17
2.1.1.3. Yemenite Codex $(M^{Y})$	17
2.1.1.4. Kennicott-de Rossi Variants	17
2.1.2. Qumran Fragments	18
2.2. The Treatment of the Hebrew Text	19
2.2.1. Vocalization	19
2.2.2. Accents	21
2.2.3. Ketiv and Qere	23
2.2.3.1. Practices in This Edition	23
2.2.3.2. Ketiv-Qere Pairs in Proverbs	24
2.2.3.3. Theories of the Origins and Functions of the	
Ketiv-Qere Readings	29
2.2.3.3.1. A Variety of Types	29
2.2.3.3.2. Qere as Correction	30

# CONTENTS

2.2.3.3.3. Collation: Ketiv and Qere as Manuscript	
Variants	30
2.2.3.3.4. Two Traditions	31
2.2.3.3.5. The Qerayin as Cues	33
2.2.4. Oriental (מדנחאי) and Occidental (מערבאי) Readings	34
3. The Versions	35
3.0. Introduction	35
3.1. The Septuagint (G)	36
3.1.1. Edition Used	36
3.1.2. The Date of G-Proverbs	36
3.1.3. The Order of G-Proverbs	37
3.1.4. The Character of G-Proverbs	38
3.1.5. A Profile of G-Proverbs	41
3.1.5.0. Introduction	41
3.1.5.1. Mimesis	42
3.1.5.2. Moralism	43
3.1.5.3. Refinement	45
3.1.5.4. Improving the "Logic"	46
3.1.5.5. Resolving or Changing Metaphors	47
3.1.5.6. Disambiguation	48
3.1.5.7. Elaboration	49
3.1.5.8. Overexplanation	51
3.1.5.9. Enhancing Parallelism	52
3.1.6. Translation as Control	54
3.1.7. Indicators of Hebrew Variants in G-Proverbs	55
3.1.7.1. A Component That Does Not Serve the	
Translator's Purposes	55
3.1.7.2. An Awkwardness in G	56
3.1.7.3. Ambiguity in the Hebrew	56
3.1.7.4. External Support	57
3.1.7.5. A Combination of Indicators	57
3.1.8. Factors That Are Neutral with Respect to the	
Underlying Text	58
3.1.8.1. Elegance	58
3.1.8.2. Interpretation	58
3.1.8.3. Alternate Proverbs	59
3.2. The Peshitta (S)	61
3.2.1. The Background of the Syriac Translation	61
3.2.2. S-Proverbs' Dependence on G-Proverbs	62
3.2.2.1. S = M	64
3.2.2.2. S ≈/= M + G	65

CONTENTS	vii
3.2.2.2.1. Borrowing Items from G without	
Exegetical Need	65
3.2.2.2.2. Borrowing Components from G to Solve	
an Ideologically Neutral Interpretive Problem	66
3.2.2.2.3. Borrowing Components from G to Solve	
an Apparent Logical or Ideological Difficulty	66
3.2.2.3. Working from G but Making Changes	67
3.2.2.4. Conflation of M and G: Taking One Line from	
M and One from G	68
3.2.2.5. S = G	68
3.2.2.6. Translating or Paraphrasing G alongside M,	
Thereby Producing Doublets	69
3.2.2.7. Incorporating Greek Additions	69
3.2.2.8. Following G in an Omission	70
3.2.3. A Profile of S-Proverbs	70
3.3. The Vulgate (V)	71
3.4. The Targum (T)	71
3.4.1. Edition Used	71
3.4.2. The Date of T-Proverbs	72
3.4.3. T-Proverbs' Relation to M and S	72
3.4.3.1. T = M	73
3.4.3.2. T ≈ M	74
3.4.3.3. $T = S \neq M$	74
4. Policies and Procedures	77
4.1. Anchor Bible Proverbs	77
4.2. Translations	77
4.3. Ambiguity of Citations in the Apparatus	78
4.3.1. Agreement ≠ Support	78
4.3.2. Lexical Assumptions	78
4.3.3. Atomizing Changes	78
4.3.4. Written and Perceived, Written and Remembered	79
4.3.4.1. Perceptual Variants	79
4.3.4.2. Memory Variants	80
4.3.4.3. "Midrashic" Variants	81
4.4. What Goes into the Critical Text?	81
5. Textual Commentary with Critical Text	83
6. Bibliography	401
7. Index	417
7.1. General	417

# CONTENTS

7.2. Hebrew	418
7.2.1. Letter Interchange and Other Graphic Errors	418
7.2.2. Phenomena and Topics in Hebrew	419
7.2.3. Hebrew Words Given Special Attention	420
7.3. Septuagint	421
7.3.1. Phenomena and Topics in G-Proverbs	421
7.3.2. Greek Words Given Special Attention	425
7.4. Peshitta	426
7.5. Vulgate	427
7.6. Targum	427
7.7. Hebrew Bible and Ben Sira	427
7.8. Other Early Sources	428
7.9. Authors	429
Critical Text of Proverbs (from right end of book)	1*

viii

# Series Foreword

This series, The Hebrew Bible: A Critical Edition (HBCE), offers a new model for a critical edition of the Hebrew Bible. The other current scholarly editions are diplomatic editions that present a single manuscript of the Masoretic Text supplemented by one or more text-critical apparatuses. The HBCE is an eclectic edition that combines the best (or earliest) readings from various sources into a critical text, with the data and analyses provided in the accompanying apparatus and text-critical commentary.

The HBCE editions aim to restore, to the extent possible, the manuscript that was the latest common ancestor of all the extant witnesses. This earliest inferable text is called the archetype. The archetype is not identical to the original text (however one defines this elusive term) but is the earliest recoverable text of a particular book. To be more precise, the HBCE will approximate the corrected archetype, since the archetype, like all manuscripts, will have scribal errors that can be remedied.

Many books of the Hebrew Bible circulated in multiple editions in antiquity, and sometimes these editions can be wholly or partially recovered. In such cases, the HBCE text will be plural, approximating the archetypes of each ancient edition. The critical text will consist of two or more parallel columns, which will be aligned to indicate the differences between the editions. In some cases the later editions are not wholly recoverable, and the available evidence is treated extensively in the text-critical commentary. The presentation of multiple ancient editions distinguishes the HBCE from the other critical editions.

Establishing the earliest attainable text and editions of each book is only part of the task. In the extensive text-critical commentary, we lay out the reasons for the preferred readings (including warranted conjectures), and we analyze the scribal and exegetical motives that gave rise to the secondary readings. Although many variants are simply the result of scribal error, others are deliberate revisions, motivated by the desire to explain, update, harmonize, and even expurgate the text. Our critical

#### SERIES FOREWORD

edition therefore moves both backward and forward in time—backward to the earliest inferable texts and editions, and forward to the plethora of changes and interpretations that occurred during the textual life of the Hebrew Bible. In other words, the critical text establishes a book's form, and the commentary explores the panorama of inner-biblical interpretations that are embedded in the texts, editions, and early translations.

As a new model, the HBCE raises afresh many fundamental issues in textual criticism. What is a biblical book? Which stage of the biblical text is more authentic? Is the biblical text a unitary object, or is it irreducibly plural, dispersed in time and space? What do we mean by the original text? What were the hermeneutical rules of ancient scribal revisions? How did interpretive tendencies differ among the various scribal/textual traditions? How do we read a plural text of the Hebrew Bible? The HBCE reframes these issues in the light of the new data and methodological refinements of the post-Qumran era.

The HBCE does not claim to be a final or perfect text. It offers a sophisticated and comprehensive presentation of textual data, analyses, and conclusions. It consists of rigorous and innovative scholarship and aims to provide an essential resource for future research. In the light of textual resources undreamt of in previous generations, we propose to renew the pioneering researches of our predecessors and raise to a new level the textual criticism of the Hebrew Bible.\*

Ronald Hendel General Editor

\* For further discussion of the theory and method of HBCE, see the project website: hbceonline.org.

# Preface

The present volume aims at reconstructing a significant stage in the development of the biblical book of Proverbs.<sup>1</sup> I will later (in §1.2) define this stage, but first I must emphasize that I am not seeking to produce a different Bible or even a different form of the Bible. As Hugh Williamson rightly says (though intending a criticism), the present edition "is not a Bible, new or old" (2009, 175). "Bible" is a socioreligious concept. Whatever a religious community in the Jewish and Christian traditions considers sacred scripture is the Bible for that community, and external scholarly adjudication is irrelevant. A community's canon alone is relevant to the study of its theology, exegesis, and practices. The HBCE text thus can claim to be a Bible only in an extended sense: it aims to uncover an early stage of books that would enter the Hebrew canon.

The heart of this volume is the commentary, which provides an apparatus, then assembles the data, evaluates them, and decides on the correct form, or sometimes the correct forms, of the verse. The critical text assembles the decisions reached in the commentary into a single text. On a fundamental level, this is what critical exegetical commentaries do as well. They must include text-critical decisions, and these decisions together imply a virtual eclectic text. In fact, once a commentator modifies even a single word, she is accepting the validity of the unmodified words, at least by default. An eclectic edition brings together a scholar's critical decisions into a single text.

An eclectic text is a bolder way of displaying emendations because it places the editor's decisions front and center for the reader's critical engagement. This gives the emendations the prominence they deserve, if one is to take them seriously as the basis for interpretation. Moreover,

<sup>1.</sup> The series that includes the present volume was originally slated for publication as the Oxford Hebrew Bible (Oxford University Press). Earlier publications, including some of my own essays, referred to it that way.

#### PREFACE

they are set in context rather than being put forth singly and marginally. The reader can better judge how well they serve the text that has been constructed.

#### Acknowledgements

I have benefited from the corrections, criticisms, and ideas of my graduate assistants in the course of this project: Jason Michele, Jonathon Wiley, Eric Tully, and Catherine Bonesho. On Ms. Bonesho's work, see §3.3. I also thank the Andrew W. Mellon Foundation for an Emeritus Fellowship for the completion of this project. I have benefited from the corrections and insights of my copy editors: Sarah Shectman, James Spinti, Eve Levavi Feinstein, and Keith Stone. In spite of the diligence of these and other readers, errors may have slipped through. Especially problematic was a tendency for strings of Hebrew and Syriac words to get disordered in the software, and some cases may have escaped notice. If the reader finds any errors, please notify me at mvfox@wisc.edu, and they will be corrected in future printings. This email address can also be used for substantive discussion.

Above all I thank Ronald Hendel, the founder and editor of the HBCE project. We worked together closely throughout the project, and our lengthy discussions of questions great and small helped me understand what had to be done and how to do it.

# Abbreviations and Sigla

Italics in verse numbering designate G additions, which have letter designations in Rahlfs 1979. For example, 9:12*a* is the first G addition after 9:12, whereas 9:12a is the first half of M 9:12 and 9:12*a*a is the first half of G 9:12*a*. It should be noted that some G additions are not given their own numbers in Rahlfs but are attached to the M verses; for example, G 15:1a is an addition and is not the same as M 15:1.

Works cited ad loc. by author's name: Baer (1880), Baumgartner (1890), Clifford (1999), Delitzsch (1873), Ehrlich (1913), Grabe (1707–1720), Jäger (1788), Kennedy (1928), Kuhn (1931), de Lagarde (1863), McKane (1970), Meinhold (1991), Pinkuss (1894), Plöger (1984), Skehan-Di Lella (1987), and Toy (1899). Also cited ad loc. are the medieval commentaries and ABP, BAJP, CSP, H-P, Kenn, and KR (see below).

#### Sigla

г 1 	"ceilings," used in the critical text to mark a reading not in $\mathrm{M}^{\mathrm{L}}$
]	separator in apparatus entry (primary reading to left, sec- ondary reading/readings with comment to the right)
+	plus
+pre	a plus preceding the lemma
>	a minus, an absence in
*	retroversion from Greek, Aramaic, etc., where there is no
	equivalent Hebrew text
< >	conjectural emendation
()	(parentheses) added elements, words judged to be sec- ondary to their literary units though present in proto-M <sup>1</sup>

<sup>1.</sup> These sigla were known as antisigma-sigma in classical Greek texts and occa-

xiv	ABBREVIATIONS AND SIGLA
()	parentheses in English translations: words added to literal translation for greater clarity
[]	brackets in English translations: components that are not part of the translation, such as variants
:	(raised :) plural-marker (syame) in Syriac transliteration
×	(overdot) uncertain letter
#	fragmentary reading in Qumran text
1	first (second, etc.) occurrence in a verse
± conj	a conjunction added or subtracted in copying or transla-
	tion
//	parallel
$\frown$	omitted text in cases of homoi
$\rightarrow$	changes to
$\leftrightarrow$	change is possible in either direction
≠	not equal; substantively not the same
~	approximately equal; close enough so as not to indicate a different text
	ellipsis
"	in Hebrew letters, indicates lexical root (e.g., שמ"ר)
\$	section in the introduction of this volume

# Abbreviations

# Greek Letters, for Hexaplaric Sources

ABP 1

α'	Aquila
αλ'	δ αλλος, οἱ αλλοι (an unidentified marginal reading in the
	Septuagint tradition; taken from Field 1875)
$\gamma'$	The Three
ε′	Quinta
εβρ'	ό έβραιος
ετ'	οί ἕτεροι
θ'	Theodotion
λ'	δι λοιποί
o'	δι έβδομήκοντα
$\sigma'$	Symmachus
	Michael V. Fox, Proverbs 1–9, Anchor Bible 18A, 2000

sionally in Hebrew texts from the Dead Sea. Their use in the Critical Text in the present volume follows an ancient practice. See Tov 2012.

ABP 2 AEL abs	Michael V. Fox, <i>Proverbs 10–31</i> , Anchor Bible 18B, 2009 Miriam Lichtheim, <i>Ancient Egyptian Literature</i> , 1973–1980 absolute (state)
accent(s) add	accents changed or omitted consequent on emendation
	addition (added elements; see Roberts 1950; Zuntz 1956)
adj	adjective
anaph antic	anaphora (transfer of material from texts elsewhere in S) anticipation
assim	assimilation (bringing a word or phrase into conformity
	with a frequent usage elsewhere; this may happen in
	Hebrew transmission or in translation)
assoc	association (the association of two words in a semantic
	field; for example, מאור and מאור in Prov 13:50)
aur	aural variant
Bab	manuscript with Babylonian vocalization
b.	Babylonian Talmud
BAP	David-Marc d'Hamonville, La Bible d'Alexandrie: Les
	Proverbes, 2000
BH	Biblical Hebrew, including the language of Ben Sira
BHQ	Biblia Hebraica Quinta
BHS	Biblia Hebraica Stuttgartensia
c	correction in manuscript, e.g., GAc
CAL	Comprehensive Aramaic Lexicon: http://cal1.cn.huc.edu/
CATSS	Computer Assisted Tools for Septuagint/Scriptural Study:
	http://ccat.sas.upenn.edu/gopher/text/religion/biblical/
	lxxmorph/
CB	Catherine Bonesho, personal communication or online
	annotations (see §3.3)
CD	Covenant of Damascus
CBH	Classical Biblical Hebrew (preexilic)
cf.	<i>confer</i> , compare
CG	Cairo Genizah biblical manuscripts (data from BHQ)
cj	conjecture (emendation)
	the commentary of the present volume
conj	conjunction
context	influence of context on translator or copyist
conv	converse translation (creating synonymy by negation of its opposite)
corr	correction (proposed correction in a version)

# ABBREVIATIONS AND SIGLA

CSP	Johann Cook, The Septuagint of Proverbs, 1997
cst	construct
CTAT	Dominique Barthélemy, Critique, 1982–2005
D-stem	piel (Aram. pael)
dbl	double translation
dbl rd	double reading (conflation of two variants <sup>2</sup> )
DCH	D. J. A. Clines, ed., <i>Dictionary of Classical Hebrew</i> , 1993–2011
dial	dialectal variant
diath	diathesis (change of passive to active or active to passive)
displ	displaced
distant dittog	distant dittography (dittography of a letter or small group of letters at a short distance; for example, Prov 15:15: dis-
	tant dittog '' $\rightarrow$ ''')
distant haplog	distant haplography (haplography of a letter or small group of letters at a short distance; the opposite of the
	preceding)
dittog	dittography (the addition of one of two identical letters or words that are in sequence)
div	words that are in sequence) word misdivision or misdivison of letters
DJD	Discoveries in the Judaean Desert (Oxford: Clarendon
DJD	and Oxford University Press)
Dp-stem	D passive stem, <i>pual</i>
Dr-stem	D reduplicated stem, <i>polel</i>
DSS	Dead Sea Scrolls
Dt-stem	hitpael (Aram. etpaal)
ed(d)	edition(s)
elab	elaboration (addition of words, phrases, or sentences
	to fill out and expand the idea or image of the preced-
	ing stich or verse, as in Prov 13:9: fin ] + 2 stichoi G [9 <i>a</i> ]
	[elab])
equal	equally plausible as primary reading
err	error
etym	etymological (translation based on the etymology, or pseudo-etymology, of a Hebrew word)

<sup>2.</sup> On the phenomenon, see Talmon 1960.

xvi

<sup>3.</sup> The assumption is that the scribe's eye was caught by the ' *ahead* of the point where he was copying, so that he wrote it before as well as after the **1**.

exeg expan explic fem fin G	<ul> <li>exegetical (see §3.1.4)</li> <li>expansion in translation</li> <li>explication (an addition controlling the sense of the text to exclude alternative [and presumed erroneous] readings) feminine</li> <li>final</li> <li>Greek version. G is used in reference to both the translation and the translator.<sup>4</sup></li> <li>G<sup>A</sup>: Codex Alexandrinus</li> <li>G<sup>B</sup>: Codex Vaticanus</li> <li>G<sup>S</sup>: Codex Sinaiticus</li> <li>G<sup>V</sup>: Codex Venetus = H-P MS 23</li> <li>* the original reading of a manuscript</li> <li>c corrector, i.e., the one who has replaced the original and still recognizable text by another text</li> <li>P a part of a group of manuscripts</li> <li>s suppletor, i.e., one who has supplied certain words which either were not originally in the manuscript or which have been lost</li> </ul>
G-stem	qal
GELS	Johan Lust et al., A Greek-English Lexicon of the Septua- gint, 1992–1996
GKC	Gesenius-Kautsch-Cowley, <i>Gesenius' Hebrew Grammar</i> , 1910
Gp-stem	G passive stem
graph	graphic error
gram	change of grammatical feature
H-stem	hiphil
HALOT	Ludwig Köhler et al., <i>Hebrew and Aramaic Lexicon of the Old Testament</i> , 1994–2000
haplog	haplography (loss of one of two identical letters or words that are in sequence)
harm	harmonization (eliminating a tension or contradiction between two passages)

<sup>4.</sup> The edition used is Rahlfs's *Septuaginta*. Unless otherwise indicated, this is treated as the original or differing from the original only inconsequentially. See OG.

xviii	ABBREVIATIONS AND SIGLA
HB	Hebrew Bible (including Aramaic portions)
homoi	homoioarkton, homoioteleuton, homoiomeson, or homoiologon (The abbreviation "homoi" refers to the ocular leap between similar beginnings, ends, middles, or whole words and the omission of text between. It is often hard to distinguish among these phenomena, and "homoi" will include them all.)
H-P	Robert Holmes and James Parson, eds., <i>Vetus Testamen-</i> <i>tum graecum cum variis lectionibus</i> , 1798–1827
Hp-stem	H passive stem, <i>hophal</i>
IBHS	Bruce K. Waltke and Michael O'Connor, <i>Introduction to Biblical Hebrew Syntax</i> , 1990
ideol	ideologically motivated change or rendering
impf	imperfect
impv	imperative
indep	independently (when two versions attest independently to a variant)
inf	infinitive
init	initial
interp	interpretation (an element added to a translation to explain the verse or a component thereof, i.e., a gloss)
JPSV	Jewish Publication Society Version
juss	jussive
Kenn	Benjamin Kennicott, <i>Vetus Testamentum hebraicum cum variis lectionibus</i> , 1776–1780
KR	Kennicott and de Rossi readings taken together.
1., 11.	line, lines
LBH1	Late Biblical Hebrew, first phase (mid-sixth to mid-fifth centuries)
LBH2	Late Biblical Hebrew, second phase (mid-fifth to approx. second centuries)
lig	ligature
lit.	literally
LSF	Friedrich Delitzsch, Lese- und Schreibfehler im Alten Tes- tament, 1920
LSJ	Henry George Liddell et al., A Greek-English Lexicon, 1996
logic	making the text, whether in Hebrew or in translation, conform to a putatively logical standard
М	Masoretic Text

$\begin{array}{llllllllllllllllllllllllllllllllllll$		M <sup>A</sup> : Aleppo Codex
MMSS: manuscripts from Kennicott-de Rossi (KR) $M^{Q:} qere^5$ $M^{RB:}$ Second Rabbinic Bible (1524–1525) $M^Y$ : Yemenite manuscript, Cambridge University, Add.Ms. 1753m.Mishnahmascmasculinemetathmetathesis (reversal or transposition of letters or words, $AB \rightarrow BA$ )mgmarginal readingMGELSTakamitsu Muraoka, A Greek-English Lexicon of the Sep- tuagint, 2009Mid.MidrashMinhat ShaiJedidiah Solomon ben Abraham Norzimisinterpmisinterpretationm.l.mater lectionismodlinguistic modernizationmoralmoralizationMpMasorah Parva (Mp <sup>L</sup> in Leningradensis; Mp <sup>A</sup> in Aleppo)MS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear synonym (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number		M <sup>K</sup> : <i>ketiv</i>
$\begin{array}{llllllllllllllllllllllllllllllllllll$		M <sup>L</sup> : Leningradensis, B19
$\begin{array}{lll} M^{\mathbb{R}\mathbb{B}} & \text{Second Rabbinic Bible (1524-1525)} \\ M^{Y}: Yemenite manuscript, Cambridge University, Add. \\ Ms. 1753 \\ m. & Mishnah \\ masc & masculine \\ metath & metathesis (reversal or transposition of letters or words, \\ AB \rightarrow BA) \\ mg & marginal reading \\ MGELS & Takamitsu Muraoka, A Greek-English Lexicon of the Sep- \\ tuagint, 2009 \\ Mid. & Midrash \\ Minhat Shai & Jedidiah Solomon ben Abraham Norzi \\ misinterp & misinterpretation \\ m.l. & mater lectionis \\ mod & linguistic modernization \\ moral & moralization \\ Mp & Masorah Parva (Mp^L in Leningradensis; Mp^A in Aleppo) \\ MS(S) & manuscript(s) \\ mult & multiple occurrences \\ N-stem & niphal \\ near dittog & dittography of graphically similar letters \\ near syn & near synonym (words sharing the same general sense for practical purposes) \\ NETS & Albert Pietersma and Benjamin G. Wright, eds., A New English Translation (bringing a word or phrase into conformity with normal usage) \\ num & change of grammatical number \\ \end{array}$		M <sup>MSS</sup> : manuscripts from Kennicott-de Rossi (KR)
MY: Yemenite manuscript, Cambridge University, Add. Ms. 1753m.Mishnahmascmasculinemetathmetathesis (reversal or transposition of letters or words, AB $\rightarrow$ BA)mgmarginal readingMGELSTakamitsu Muraoka, A Greek-English Lexicon of the Sep- tuagint, 2009Mid.MidrashMinhat ShaiJedidiah Solomon ben Abraham Norzimisinterpmisinterpretationm.l.mater lectionismodlinguistic modernizationmoralmoralizationMS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number		$M^{Q}$ : $qere^{5}$
Ms. 1753m.Mishnahmascmasculinemetathmetathesis (reversal or transposition of letters or words, AB $\rightarrow$ BA)mgmarginal readingMGELSTakamitsu Muraoka, A Greek-English Lexicon of the Sep- tuagint, 2009Mid.MidrashMinhat ShaiJedidiah Solomon ben Abraham Norzimisinterpmisinterpretationm.l.mater lectionismodlinguistic modernizationmoralmoralizationMpMasorah Parva (Mp <sup>L</sup> in Leningradensis; Mp <sup>A</sup> in Aleppo)MS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number		M <sup>RB</sup> : Second Rabbinic Bible (1524–1525)
m.Mishnahmascmasculinemetathmetathesis (reversal or transposition of letters or words, $AB \rightarrow BA$ )mgmarginal readingMGELSTakamitsu Muraoka, A Greek-English Lexicon of the Sep- tuagint, 2009Mid.MidrashMinhat ShaiJedidiah Solomon ben Abraham Norzimisinterpmisinterpretationm.l.mater lectionismodlinguistic modernizationmoralmoralizationMf(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number		M <sup>Y</sup> : Yemenite manuscript, Cambridge University, Add.
mascmasculinemetathmetathesis (reversal or transposition of letters or words, AB $\rightarrow$ BA)mgmarginal readingMGELSTakamitsu Muraoka, A Greek-English Lexicon of the Sep- tuagint, 2009Mid.MidrashMinhat ShaiJedidiah Solomon ben Abraham Norzimisinterpmisinterpretationm.l.mater lectionismodlinguistic modernizationmoralmoralizationMpMasorah Parva (Mp <sup>L</sup> in Leningradensis; Mp <sup>A</sup> in Aleppo)MS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number		Ms. 1753
Initial metathmetathesis (reversal or transposition of letters or words, $AB \rightarrow BA$ )mgmarginal readingMGELSTakamitsu Muraoka, A Greek-English Lexicon of the Sep- tuagint, 2009Mid.MidrashMinhat ShaiJedidiah Solomon ben Abraham Norzimisinterpmisinterpretationm.l.mater lectionismodlinguistic modernizationmoralmoralizationMpMasorah Parva (MpL in Leningradensis; MpA in Aleppo)MS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	m.	Mishnah
AB $\rightarrow$ BA)mgmarginal readingMGELSTakamitsu Muraoka, A Greek-English Lexicon of the Sep- tuagint, 2009Mid.MidrashMinhat ShaiJedidiah Solomon ben Abraham Norzimisinterpmisinterpretationm.l.mater lectionismodlinguistic modernizationmoralmoralizationMpMasorah Parva (Mp <sup>L</sup> in Leningradensis; Mp <sup>A</sup> in Aleppo)MS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007numchange of grammatical number	masc	masculine
mgmarginal readingMGELSTakamitsu Muraoka, A Greek-English Lexicon of the Sep- tuagint, 2009Mid.MidrashMinhat ShaiJedidiah Solomon ben Abraham Norzimisinterpmisinterpretationm.l.mater lectionismodlinguistic modernizationmoralmoralizationMS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	metath	metathesis (reversal or transposition of letters or words,
MGELSTakamitsu Muraoka, A Greek-English Lexicon of the Sep- tuagint, 2009Mid.MidrashMinhat ShaiJedidiah Solomon ben Abraham Norzimisinterpmisinterpretationm.l.mater lectionismodlinguistic modernizationmoralmoralizationMgMasorah Parva (Mp <sup>L</sup> in Leningradensis; Mp <sup>A</sup> in Aleppo)MS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number		$AB \rightarrow BA$ )
tuagint, 2009Mid.MidrashMinhat ShaiJedidiah Solomon ben Abraham Norzimisinterpmisinterpretationm.l.mater lectionismodlinguistic modernizationmoralmoralizationMpMasorah Parva (Mp <sup>L</sup> in Leningradensis; Mp <sup>A</sup> in Aleppo)MS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	mg	marginal reading
Mid.MidrashMinhat ShaiJedidiah Solomon ben Abraham Norzimisinterpmisinterpretationm.l.mater lectionismodlinguistic modernizationmoralmoralizationMpMasorah Parva (Mp <sup>L</sup> in Leningradensis; Mp <sup>A</sup> in Aleppo)MS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	MGELS	Takamitsu Muraoka, A Greek-English Lexicon of the Sep-
Minhat ShaiJedidiah Solomon ben Abraham Norzimisinterpmisinterpretationm.l.mater lectionismodlinguistic modernizationmoralmoralizationMpMasorah Parva (Mp <sup>L</sup> in Leningradensis; Mp <sup>A</sup> in Aleppo)MS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number		tuagint, 2009
misinterpmisinterpretationm.l.mater lectionismodlinguistic modernizationmoralmoralizationMpMasorah Parva (Mp <sup>L</sup> in Leningradensis; Mp <sup>A</sup> in Aleppo)MS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	Mid.	Midrash
m.l.mater lectionismodlinguistic modernizationmoralmoralizationMpMasorah Parva (MpL in Leningradensis; MpA in Aleppo)MS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear haploghaplography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	Minhat Shai	Jedidiah Solomon ben Abraham Norzi
modlinguistic modernizationmoralmoralizationMpMasorah Parva (Mp <sup>L</sup> in Leningradensis; Mp <sup>A</sup> in Aleppo)MS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear haploghaplography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	misinterp	misinterpretation
moralmoralizationMpMasorah Parva (MpL in Leningradensis; MpA in Aleppo)MS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear haploghaplography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	m.l.	mater lectionis
MpMasorah Parva (MpL in Leningradensis; MpA in Aleppo)MS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear haploghaplography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	mod	linguistic modernization
MS(S)manuscript(s)multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear haploghaplography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	moral	moralization
multmultiple occurrencesN-stemniphalnear dittogdittography of graphically similar lettersnear haploghaplography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	Мр	Masorah Parva (Mp <sup>L</sup> in Leningradensis; Mp <sup>A</sup> in Aleppo)
N-stemniphalnear dittogdittography of graphically similar lettersnear haploghaplography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	MS(S)	manuscript(s)
near dittogdittography of graphically similar lettersnear haploghaplography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	mult	multiple occurrences
near haploghaplography of graphically similar lettersnear synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	N-stem	niphal
near synnear synonymy (words sharing the same general sense for practical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	near dittog	dittography of graphically similar letters
NETSpractical purposes)NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	near haplog	haplography of graphically similar letters
NETSAlbert Pietersma and Benjamin G. Wright, eds., A New English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into confor- mity with normal usage)numchange of grammatical number	near syn	near synonymy (words sharing the same general sense for
English Translation of the Septuagint, 2007normnormalization (bringing a word or phrase into conformity with normal usage)numchange of grammatical number		practical purposes)
normnormalization (bringing a word or phrase into conformity with normal usage)numchange of grammatical number	NETS	
num change of grammatical number		English Translation of the Septuagint, 2007
num change of grammatical number	norm	normalization (bringing a word or phrase into confor-
Ost ostracon	num	change of grammatical number
	Ost	ostracon

<sup>5.</sup> In the apparatus to the critical text, I cite the *qere* as M<sup>Q</sup>, without indicating the source manuscript of each. That information is given in the tables in §2.2.3.2.

XX	ABBREVIATIONS AND SIGLA
Oc	Occidentalis: the reading of the Palestinian Masoretes; cf. Or.
OG	Old (original) Greek, or differing from the original only inconsequentially
OL	Old Latin (Vetus Latina) translation
om	omission
Or	Orientalis: the presumed reading of the Babylonian Masoretes. See §2.2.4.
orth	change (or misconstrual) of orthography
OS	Old (original) Syriac Peshitta
Рар	Papyrus
paleo	paleo-Hebrew script
parab	parablepsis (skipping over letters or words, regardless of the cause)
pass	passive
per	person
pf	perfect
phon	phonetic (a variant induced by a phonetic similarity but not necessarily because the word was actually <i>heard</i> dif- ferently; that would be "aur[al]") <sup>6</sup>
pl	plural
pm	first hand ( <i>prima manus</i> )
poss	possessive
proto-G, S, P	the stemmatic precursor of these versions (see §1.2)
ptcp	participle
xQxx <sup>x</sup>	Qumran text; cited from DJD
rab	rabbinic (a rendering according to a rabbinic usage in an ancient translation)
re	regarding
ref	referenced by; with reference to (used in citing an idea from an unavailable secondary source, often taken from Baumgartner 1890 and de Lagarde 1863)
reform	reformulation (This is a proverb created by modification of an older one. This is a deliberate process and central to

<sup>6.</sup> The similarity may be heard by the "inner ear." A phonetic variant may be dialectal, as perhaps in the case of  $\mathfrak{L}/\mathfrak{c}$  interchanges; see the comment on Prov 16:21.

within the M tradition as well as in later transmission.) rel reliqui (all manuscripts besides the ones cited) rep repetition of noncontiguous word or phrase, e.g., the second אַדְהֶהָ in Prov 22:21) RH Rabbinic (Mishnaic) Hebrew S Syriac: the Peshitta translation (Leiden edition). (S is used in reference to both the translation and the transla- tor. Unless otherwise indicated, S refers to the main text in Leiden.) SD Wolfgang Krauss and Martin Karrer, eds., Septuaginta Deutsch, 2009 SDEK Kraus and Karrer, Septuaginta Deutsch: Erläuterungen und Kommentare sep separation of one letter into two (cf. lig) sfx suffix sg singular sim similarly Sir Ben Sira (Ecclesiasticus) <sup>7</sup> SL Michael Sokoloff, A Syriac Lexicon, 2009 slot slot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated) smooth smoothing (improving the flow of a passage by grammat- ical changes) syn substitution of synonym SyrH Syro-Hexapla TAD Bezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993 T Targum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4) T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ) T <sup>L</sup>		the development of a proverbial transmission. It occurs
reprepetition of noncontiguous word or phrase, e.g., the second אַרְקָה in Prov 22:21)RHRabbinic (Mishnaic) HebrewSSyriac: the Peshitta translation (Leiden edition). (S is used in reference to both the translation and the transla- tor. Unless otherwise indicated, S refers to the main text in Leiden.)SDWolfgang Krauss and Martin Karrer, eds., Septuaginta Deutsch, 2009SDEKKraus and Karrer, Septuaginta Deutsch: Erläuterungen und Kommentaresepseparation of one letter into two (cf. lig) sfxsimsimilarlySirBen Sira (Ecclesiasticus) <sup>7</sup> SLMichael Sokoloff, A Syriac Lexicon, 2009slotslot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated)smoothsmoothing (improving the flow of a passage by grammat- ical changes)synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ)T <sup>L</sup> Targum Proverbs in de Lagarde 1873		
second אדקהRHRabbinic (Mishnaic) HebrewSSyriac: the Peshitta translation (Leiden edition). (S is used in reference to both the translation and the transla- tor. Unless otherwise indicated, S refers to the main text in Leiden.)SDWolfgang Krauss and Martin Karrer, eds., Septuaginta Deutsch, 2009SDEKKraus and Karrer, Septuaginta Deutsch: Erläuterungen und Kommentaresepseparation of one letter into two (cf. lig)sfxsuffixsgsingularsimsimilarlySirBen Sira (Ecclesiasticus)7SLMichael Sokoloff, A Syriac Lexicon, 2009slotslot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated)smoothsmoothing (improving the flow of a passage by grammat- ical changes)synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ)T <sup>L</sup> Targum Proverbs in de Lagarde 1873	rel	
RHRabbinic (Mishnaic) HebrewSSyriac: the Peshitta translation (Leiden edition). (S is used in reference to both the translation and the transla- tor. Unless otherwise indicated, S refers to the main text in Leiden.)SDWolfgang Krauss and Martin Karrer, eds., Septuaginta Deutsch, 2009SDEKKraus and Karrer, Septuaginta Deutsch: Erläuterungen und Kommentaresepseparation of one letter into two (cf. lig)sfxsuffixsgsingularsimsimilarlySirBen Sira (Ecclesiasticus)7SLMichael Sokoloff, A Syriac Lexicon, 2009slotslot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated)smoothsmoothing (improving the flow of a passage by grammat- ical changes)synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ)T <sup>L</sup> Targum Proverbs in de Lagarde 1873	rep	
SSyriac: the Peshitta translation (Leiden edition). (S is used in reference to both the translation and the transla- tor. Unless otherwise indicated, S refers to the main text in Leiden.)SDWolfgang Krauss and Martin Karrer, eds., Septuaginta Deutsch, 2009SDEKKraus and Karrer, Septuaginta Deutsch: Erläuterungen 		
<ul> <li>used in reference to both the translation and the translator. Unless otherwise indicated, S refers to the main text in Leiden.)</li> <li>SD Wolfgang Krauss and Martin Karrer, eds., Septuaginta Deutsch, 2009</li> <li>SDEK Kraus and Karrer, Septuaginta Deutsch: Erläuterungen und Kommentare</li> <li>sep separation of one letter into two (cf. lig)</li> <li>sfx suffix</li> <li>sg singular</li> <li>simi similarly</li> <li>Sir Ben Sira (Ecclesiasticus)<sup>7</sup></li> <li>SL Michael Sokoloff, A Syriac Lexicon, 2009</li> <li>slot slot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated)</li> <li>smooth smoothing (improving the flow of a passage by grammatical changes)</li> <li>syn substitution of synonym</li> <li>SyrH Syro-Hexapla</li> <li>TAD Bezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993</li> <li>T Targum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)</li> <li>T<sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ)</li> <li>T<sup>L</sup> Targum Proverbs in de Lagarde 1873</li> </ul>	RH	Rabbinic (Mishnaic) Hebrew
tor. Unless otherwise indicated, S refers to the main text in Leiden.)SDWolfgang Krauss and Martin Karrer, eds., Septuaginta Deutsch, 2009SDEKKraus and Karrer, Septuaginta Deutsch: Erläuterungen und Kommentaresepseparation of one letter into two (cf. lig)sfxsuffixsgsingularsimsimilarlySirBen Sira (Ecclesiasticus)7SLMichael Sokoloff, A Syriac Lexicon, 2009slotslot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated)smoothsmoothing (improving the flow of a passage by grammat- ical changes)synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3-4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ) TL	S	Syriac: the Peshitta translation (Leiden edition). (S is
in Leiden.)SDWolfgang Krauss and Martin Karrer, eds., Septuaginta Deutsch, 2009SDEKKraus and Karrer, Septuaginta Deutsch: Erläuterungen und Kommentaresepseparation of one letter into two (cf. lig)sfxsuffixsgsingularsimsimilarlySirBen Sira (Ecclesiasticus)7SLMichael Sokoloff, A Syriac Lexicon, 2009slotslot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated)smoothsmoothing (improving the flow of a passage by grammat- ical changes)synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3-4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ) TrL		used in reference to both the translation and the transla-
SDWolfgang Krauss and Martin Karrer, eds., Septuaginta Deutsch, 2009SDEKKraus and Karrer, Septuaginta Deutsch: Erläuterungen und Kommentaresepseparation of one letter into two (cf. lig) sfxsfxsuffixsgsingularsimsimilarlySirBen Sira (Ecclesiasticus)7SLMichael Sokoloff, A Syriac Lexicon, 2009slotslot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated)smoothsmoothing (improving the flow of a passage by grammat- ical changes)synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3-4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ) T <sup>L</sup>		tor. Unless otherwise indicated, S refers to the main text
Deutsch, 2009SDEKKraus and Karrer, Septuaginta Deutsch: Erläuterungen und Kommentaresepseparation of one letter into two (cf. lig)sfxsuffixsgsingularsimsimilarlySirBen Sira (Ecclesiasticus)7SLMichael Sokoloff, A Syriac Lexicon, 2009slotslot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated)smoothsmoothing (improving the flow of a passage by grammat- ical changes)synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3-4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ)T <sup>L</sup> Targum Proverbs in de Lagarde 1873		in Leiden.)
SDEKKraus and Karrer, Septuaginta Deutsch: Erläuterungen und Kommentaresepseparation of one letter into two (cf. lig)sfxsuffixsgsingularsimsimilarlySirBen Sira (Ecclesiasticus)7SLMichael Sokoloff, A Syriac Lexicon, 2009slotslot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated)smoothsmoothing (improving the flow of a passage by grammat- ical changes)synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3-4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ) TrL	SD	Wolfgang Krauss and Martin Karrer, eds., Septuaginta
und Kommentaresepseparation of one letter into two (cf. lig)sfxsuffixsgsingularsimsimilarlySirBen Sira (Ecclesiasticus) <sup>7</sup> SLMichael Sokoloff, A Syriac Lexicon, 2009slotslot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated)smoothsmoothing (improving the flow of a passage by grammat- ical changes)synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ)T <sup>L</sup> Targum Proverbs in de Lagarde 1873		Deutsch, 2009
sepseparation of one letter into two (cf. lig)sfxsuffixsgsingularsimsimilarlySirBen Sira (Ecclesiasticus) <sup>7</sup> SLMichael Sokoloff, A Syriac Lexicon, 2009slotslot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated)smoothsmoothing (improving the flow of a passage by grammat- ical changes)synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ)T <sup>L</sup> Targum Proverbs in de Lagarde 1873	SDEK	Kraus and Karrer, Septuaginta Deutsch: Erläuterungen
sfxsuffixsgsingularsimsimilarlySirBen Sira (Ecclesiasticus) <sup>7</sup> SLMichael Sokoloff, A Syriac Lexicon, 2009slotslot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated)smoothsmoothing (improving the flow of a passage by grammat- ical changes)synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ)T <sup>L</sup> Targum Proverbs in de Lagarde 1873		und Kommentare
sg singular sim similarly Sir Ben Sira (Ecclesiasticus) <sup>7</sup> SL Michael Sokoloff, <i>A Syriac Lexicon</i> , 2009 slot slot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated) smooth smoothing (improving the flow of a passage by grammat- ical changes) syn substitution of synonym SyrH Syro-Hexapla TAD Bezalel Porten and Ada Yardeni, <i>Textbook of Aramaic Documents</i> , 1993 T Targum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4) T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ) T <sup>L</sup> Targum Proverbs in de Lagarde 1873	sep	separation of one letter into two (cf. lig)
sim similarly Sir Ben Sira (Ecclesiasticus) <sup>7</sup> SL Michael Sokoloff, A Syriac Lexicon, 2009 slot slot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated) smooth smoothing (improving the flow of a passage by grammat- ical changes) syn substitution of synonym SyrH Syro-Hexapla TAD Bezalel Porten and Ada Yardeni, <i>Textbook of Aramaic</i> <i>Documents</i> , 1993 T Targum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4) T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ) T <sup>L</sup> Targum Proverbs in de Lagarde 1873	sfx	suffix
SirBen Sira (Ecclesiasticus)7SLMichael Sokoloff, A Syriac Lexicon, 2009slotslot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated)smoothsmoothing (improving the flow of a passage by grammat- ical changes)synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ) TITLTargum Proverbs in de Lagarde 1873	sg	singular
SLMichael Sokoloff, A Syriac Lexicon, 2009slotslot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated)smoothsmoothing (improving the flow of a passage by grammat- ical changes)synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ) TILTLTargum Proverbs in de Lagarde 1873	sim	similarly
slotslot variant (replacement of a redundant or difficult word by a simpler expression in the "slot" vacated)smoothsmoothing (improving the flow of a passage by grammat- ical changes)synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, <i>Textbook of Aramaic</i> <i>Documents</i> , 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ) TLTLTargum Proverbs in de Lagarde 1873	Sir	
by a simpler expression in the "slot" vacated)smoothsmoothing (improving the flow of a passage by grammatical changes)synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ) TLTLTargum Proverbs in de Lagarde 1873	SL	Michael Sokoloff, A Syriac Lexicon, 2009
smoothsmoothing (improving the flow of a passage by grammatical changes)synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ) TILTLTargum Proverbs in de Lagarde 1873	slot	slot variant (replacement of a redundant or difficult word
ical changes) syn substitution of synonym SyrH Syro-Hexapla TAD Bezalel Porten and Ada Yardeni, <i>Textbook of Aramaic</i> <i>Documents</i> , 1993 T Targum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4) T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ) T <sup>L</sup> Targum Proverbs in de Lagarde 1873		
synsubstitution of synonymSyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ) TrLTLTargum Proverbs in de Lagarde 1873	smooth	
SyrHSyro-HexaplaTADBezalel Porten and Ada Yardeni, Textbook of Aramaic Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)T <sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ) TrLTLTargum Proverbs in de Lagarde 1873		ical changes)
<ul> <li>TAD Bezalel Porten and Ada Yardeni, <i>Textbook of Aramaic Documents</i>, 1993</li> <li>T Targum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)</li> <li>T<sup>Be</sup> Targum in MS Berlin Orientalis 1213 (data from BHQ)</li> <li>T<sup>L</sup> Targum Proverbs in de Lagarde 1873</li> </ul>	syn	substitution of synonym
Documents, 1993TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)TBeTargum in MS Berlin Orientalis 1213 (data from BHQ)TLTargum Proverbs in de Lagarde 1873	SyrH	Syro-Hexapla
TTargum, the Aramaic translation (for other manuscripts, see Healey 1991, 3–4)TBeTargum in MS Berlin Orientalis 1213 (data from BHQ)TLTargum Proverbs in de Lagarde 1873	TAD	Bezalel Porten and Ada Yardeni, Textbook of Aramaic
see Healey 1991, 3–4)TBeTargum in MS Berlin Orientalis 1213 (data from BHQ)TLTargum Proverbs in de Lagarde 1873		Documents, 1993
TBeTargum in MS Berlin Orientalis 1213 (data from BHQ)TLTargum Proverbs in de Lagarde 1873	Т	Targum, the Aramaic translation (for other manuscripts,
The second of		•
	-	•
	-	÷ ÷
T <sup>W</sup> Targum Proverbs in Walton's Polyglot	$T^W$	Targum Proverbs in Walton's Polyglot

<sup>7.</sup> The Hebrew text of Ben Sira edited by Pancratius C. Beentjes (1997) is used in this volume. References to Ben Sira specify the manuscript only when there are differences significant to the matter under discussion. For abbreviations of the manuscripts and discussion of textual witnesses, see Skehan and Di Lella 1987, 52–59.

xxii	ABBREVIATIONS AND SIGLA
$T^Z$	Targum Proverbs in Zamora text (= San Bernardo 116-Z- 40): Díez Merino 1984
theol	theologically motivated change
transf	transfer of a phrase or sentence from another verse
transp	transposition (change in sequence of letters, words, or verses, e.g. ABC $\rightarrow$ BAC)
unc	uncertain
VSO	verso
V	Vulgate
$V^A$	Vulgate Codex Amiatinus
var(s)	variant(s)
vocal	difference of vocalization
у.	Jerusalem Talmud (Yerushalmi)

# **1. IS TEXT CRITICISM POSSIBLE?**

There can be no edition of the work of a writer of former times which is satisfactory to all readers, though there might, I suppose, be at least half a dozen editions of the works of Shakespeare executed on quite different lines, each of which, to one group of readers, would be the best edition possible. (McKerrow 1939, 1)

As the demand increases for the plays of Shakespeare, so new editors will arise all with notions and new readings of their own,—till it will end perhaps by every intelligent man turning editor for himself. (anonymous reviewer, *Athenaeum*, 1853)

The editor [of an eclectic text] thus presents to the readers a personal view of the original text of the book of Genesis or Kings. Needless to say, the reconstruction of such an Urtext requires subjective decisions, and if textual scholars indulged their textual acumen, each scholar would create a different Urtext. (Tov 2008b, 246)

## 1.1. THE CHALLENGE

Text criticism of all literature presents severe conundrums to scholars who undertake it. (To those who do not, the problems are still there, just ignored.) The Hebrew Bible, in particular the book of Proverbs, raises a special set of issues. I will discuss them and explain the theoretical basis for my choices, placing them in the context of textual critical theory generally, with examples from editorial practice in other literatures.

The thorniest issue is what the eclectic text will represent, to the degree it succeeds. It cannot be the *Urtext*, the original text of a book, the form it had immediately subsequent to its composition and prior to any corruption. This goal is feasible for some biblical books, at least as a heuristic. But in the case of Proverbs, it is unclear what its *Urtext* could even refer to. Hezekiah's collection? Or the four collections in chapters 10–29 prior to later additions, which in my view include chapters 1–9 and

30–31<sup>?1</sup> Anyway, in a sense the book of Proverbs is *all* additions, since it is, by its own testimony, an anthology of anthologies, themselves agglomerations of proverbs, epigrams, and poems, some deriving from oral literature, others having antecedents in written wisdom, others composed afresh. At what stage do "additions" become "later"? To this complexity we can add the likelihood that proverbs are constantly being reshaped.<sup>2</sup> The border between *Urtext* and additions blurs. The goals have to fit the reality of the text at hand.

### 1.2. The Goals

The primary goal of The Hebrew Bible: A Critical Edition (HBCE) is to reconstruct the corrected archetypes of biblical books. I understand "archetype" as the *latest* form ancestral to the extant text-forms. This is not the *Urtext*, the original form the book had. Nor is the textual archetype the particular manuscript that was ancestral to the extant text-forms. The textual archetype is an ideal: the form of the archetype free of the errors that the authors (one surmises) would have wanted fixed. It is the *work*, as defined below. The goal of reconstructing an archetype is a heuristic, and it is unlikely that it can ever be fully attained.

In the case of Proverbs one cannot reasonably aim to recover the full archetype. The book is too multiplex, the resources too scanty, and the translations too polymorphous and ambiguous to make this goal feasible. Still, the archetype is not entirely beyond reach. When the text makes good sense and is witnessed to by the extant versions, the chances are that we have the archetype, and this is true of much of the book of Proverbs. Moreover, I will sometimes reconstruct words and phrases I consider ancestral

<sup>1.</sup> In ABP 1.44–47, 322–30, and 2.499–500, I trace the development of Proverbs. The earliest level comprises chapters 10–29 as a whole. This stage is, of course, composite. It includes four collections (10:1–22:16; 22:17–24:22; 24:23–34; 25:1–29:27), which are themselves compiled from sayings of an indeterminable array of ages and authors. Later, chapters 1–9 were prefixed as an introduction to the book. Within this introduction I distinguish two stages. The earlier is a cycle of ten poems I call "Lectures": I. 1:8–19; II. 2:1–22; III. 3:1–12; IV. 3:21–35; V. 4:1–9; VI. 4:10–19; VII. 4:20–27; VIII. 5:1–23; IX. 6:20–35; X. 7:1–27. Later, and at different times, five independent poems were interspersed among (but not inside) the Lectures. These I designate "Interludes." They are: A. 1:20–33; B. 3:13–20; C. 6:1–19; D. 8:1–36; E. 9:1–18. It is impossible to determine when the four appendices in chapters 30–31 were added.

<sup>2.</sup> See ABP 1.487–93 and the literature cited there.

to both M and G (more precisely, G's Hebrew source text) and thus belonging to their archetype. But I cannot do this with sufficient consistency to compose a continuous reconstruction of the archetype. We cannot reconstruct the hyparchetypes of G, M, and S, merge them into one form, and call this the archetype. In any case, we should not assume that the text lines of M, G, and S were all that ever existed.

Instead of reconstructing the archetype, I define three practical goals for this volume:

1. To reconstruct the corrected hyparchetype of the Masoretic book of Proverbs, or proto-M. This is what the HBCE text aims to represent. I understand "hyparchetype" as Paul Maas defined them in his classic handbook on editing (1956, 6). Hyparchetypes are "reconstructed variantcarriers," that is to say, deviating text-forms that derive from a single nonextant source text (at some remove) but not from each other. Their relation is collateral. I define my goal as the corrected hyparchetype because I want to correct errors even if (as is very likely) many were already in the text that reached G's lineage.

2. To recover ancient Hebrew variants (regardless of their validity) and to evaluate them. To this end I will examine the ancient translations, primarily G, S, and the regrettably few fragments from Qumran. These variants will be available in the apparatus and commentary for the reader's evaluation.

3. To reconstruct non-M hyparchetypes, which in practice means proto-G and, to a lesser extent, proto-S. These are the text-states ancestral to the source text used by the Greek and Syriac translators. If extensive reconstruction of either were possible, it would receive its own column as a parallel edition, but given the uncertainties of retroverting the translations of this book, reconstruction is feasible only sporadically. The results will be given in the commentary and used in the critical text only when, in my view, they represent the corrected proto-M.

The basic integrity of the hyparchetypes should be maintained. One hyparchetype should not be imposed on the other, certainly not when the differences arise from deliberate innovations rather than just errors. Respecting the integrity of the hyparchetypes is particularly relevant to Proverbs, in which hyparchetypal proverb forms can be equally valid variant proverbs, such as exist within the Masoretic book of Proverbs itself.<sup>3</sup>

<sup>3.</sup> For example, Prov 13:14 and 14:27. Variant proverbs within M are precisely catalogued and interconnected by Snell 1993; see esp. 35–42.

Nevertheless, the text of one line of the stemma can legitimately be corrected with the help of a reading from a parallel edition.

#### **1.3. THEORY AND PRACTICE**

I follow Thomas Tanselle's distinction among "document," "text," and "work" (1989, 14-18). A document is an artifact—such as a manuscript or a printing-that carries a text. The text may be carried by one document or by many, or by oral performance. The text exists apart from the documents. (Thus if all documents of a certain poem were lost but the poem exactly remembered, its text would still exist.) Behind the text is the work, the ideal form of the text.<sup>4</sup> According to Tanselle, "the work is a historical act of intention, an ideal and enduring configuration created by an author but embodied (inevitably) in a flawed form."5 When a text holds errors, we can aim to correct them only by envisioning the work, which has a sort of platonic existence, abstracted from any of its particular textual instantiations. To take the example of Proverbs: The documents I mostly use are my printed copy of BHS and the digitized copy. They (and other copies of BHS) hold the same text. That text has errors, as does every text of Proverbs, including the one preserved in the Aleppo Codex. Text criticism aims to reconstruct the text that best embodies the work, or one stage of it.6

It is very unlikely that there actually was a document that held the exact text of proto-M. That is because changes, deliberate and unintended, were surely introduced at different times, some even before the later parts of the book were added. I wish to be clear that the text I have produced,

<sup>4. &</sup>quot;Ideal" does not mean perfect. Tanselle (1996b, esp. 12–13) argues against this misconstrual of his views. As he explains, "The only sense in which intentionalist editors construct 'ideal' texts is that those texts may not have existed in physical form before the editors produced them; but such editors do not think of their texts as perfect in any sense, nor do these editors believe that they are uncovering the 'idea of a text' underlying any particular executed text" (13).

<sup>5.</sup> Tanselle's principles, as summarized by Cohen and Jackson 1991, 106.

<sup>6.</sup> Hendel (forthcoming) examines these issues in a different conceptual framework, the philosophy of art, but comes to conclusions compatible with the ones described here. Hendel uses a distinction, formulated by Charles Peirce, between "type" and "token." In brief, "A book as a discourse is a type (an abstract semiotic object), and the physical object with its visible symbols is its token" (section I). I understand "type" to be comparable to "work" and "token" to "document." Texts are *instantiated* by tokens but are not identical to them. In Hendel's view, the text also is a type.

however successful, never had physical existence. It is a construct. It can be defined as the proto-M *as it should have* been, the text the authors and editors wanted us to read. This goal is heuristic: approachable but not wholly attainable.

Defining my primary goal as the reconstruction of the corrected Masoretic hyparchetype provides some guidelines for deciding what material to excise as extraneous or to mark as later. I am aiming for a stage that hypothetically existed prior to the text of the proto-MT, without *its* errors and intrusions.

Copyist errors (whether early or late) can certainly be removed. For example, חצבה in Prov 9:1 is (I argue in the commentary) a copyist error, which I emend to הצבה, based on G. Moreover, glosses unrepresented in G can be eliminated as secondary to proto-M and moved to the apparatus (which makes them, in HBCE's format, *marginal* glosses). In Prov 5:22a, for example, the phrase את הרשע is not represented in G or S, and it is not integrated into the Hebrew syntax. In the case of 23:23, an entire verse is eliminated from the HBCE text.

There are a large number of pluses in G-Proverbs, consisting of lines, verses, and even full poems. Some, perhaps most, were composed in Greek; for example, the epigram on the bee in 6:8a-8c certainly was. But even when I think that an addition had a Hebrew basis and that I can retrovert it (for example, 9:12a-12b), I will confine it to the commentary. Such additions arose in the proto-G tradition in or prior to translation. In practice, the distinction between originally Hebrew and originally Greek additions is very hard to determine.

The Septuagint also has minuses of stichoi and couplets, which are unlikely to be the translator's deliberate doing. Many of these minuses are accidental omissions, such as Prov 1:16, "for their feet run to harm; they rush to shed blood." (The verse is essential to the context and probably lost in G through parablepsis.) Other minuses are less easily explained, as, for example, several verses in 15:27a-16:9 (see the introduction to that section in the commentary); 18:23-19:2; and 20:14-19. It is quite likely that M continued to develop after the G transmission went its own way, but unless there is evidence that they are additions to proto-M, I maintain them in the HBCE text. To be sure, scribes are more likely to add than to deliberately subtract, but for an editor to mechanically insist on the shorter text does an injustice to the diversity and complexity of scribal practice.<sup>7</sup>

<sup>7.</sup> For example, the large plus in 4QSam<sup>a</sup> in 1 Sam 11 (about Nahash's gouging

More problematic are presumptive scribal additions that reached G as well as M and were therefore present in proto-M. These may be short glosses such as נשים in 14:1 or sentences such as 8:13a, "The fear of the Lord means hating evil." I am trying to reproduce not just proto-M but a better, more correct form of that text, that is, the corrected hyparchetype. I argue that some components in proto-M are extraneous to the literary integrity of the unit they are in. These are marked in the HBCE text by parentheses, which I use in a way comparable to the ancient scribal sigla antisigma-sigma. These are almost identical in form to modern parentheses and are found in Qumran texts and even in M itself, where they were thought to be inverted-nuns (see Tov 2004, 201-2, and figs. 8.1 and 8.2). These sigla usually marked words as extra, that is to say, words that the scribe knew to be absent from another copy of the work. By identifying some components as additions prior to proto-M, I am able to approximate the earlier work without disrupting proto-M's integrity. The components I identify as secondary are certainly not the only additions embedded in the proverbial collections, but the presence of such components is hard to detect in groups of unorganized or loosely clustered proverbs.

Lateness of one literary unit relative to another does not bring it special marking, since this description would fit most of the book. For example, the first part of the book of Proverbs, chapters 1–9, was prefixed to the proverb collections, chapters 10–29, or perhaps 10–31 (see note 1). But prior to these chapters there was no book of Proverbs.

In all this, it must be emphasized, my reconstructions and literarycritical proposals are based on exegesis, which I first provided in ABP and include, in less detail, in the commentary of the present volume. None of them can claim certainty.

#### 1.4. Authors and Editors

As an editor I try to recover the text the authors intended. My goals are in line with the position formulated by Thomas Tanselle, as developed in

out the right eyes of the men of Reuben and Gad) is original, and its absence in M is accidental (thus Tov 2012, 311–13). In two Esther texts I have worked on, G-Esther and the Alpha Text (Fox 1991), there are demonstrable deliberate omissions. A redactor copied the Septuagint's additions into an older form of the Alpha Text but omitted some material (listed in Fox 1991, 62–68). That at least some omissions are deliberate is shown by the way that the editor will connect the words just before and after into a new and intact sentence. I call this "cutting-and-splicing" (65–68).

his *A Rationale of Textual Criticism* (1989) and in numerous essays. The editor aims to construct the text that best represents the work. This is not a mechanical task but requires engaged exegesis and literary judgments. "An editor, only through his analysis and understanding of the meaning of the work in the light of his knowledge of the author and his times, will be in a position to use authorial active intention as a basis for editorial choice" (Tanselle 1976, 210).

In the 1980s and 1990s, as the author lost his privileged place in literary theory, editorial theory shifted toward a nonintentionalist paradigm. In this conception, the work as a whole exists in diachronicity, in fluctuating and multiple forms, while each text-form is an autonomous "textual moment." This view of text and editing holds considerable interest for biblical text criticism, because the Bible demonstrably exists in temporal extension, and this fact legitimates the production of different types of editions with different goals. But the diachronic perspective does not exclude the intentionalist paradigm, insofar as this aims at authorial intention.<sup>8</sup> But this is an editorial choice, not an absolute verity. An editor could choose to represent the form of the text known to traditional Jewish scholarship, in which case nothing but inner-Masoretic variants would be relevant. The Masoretic enterprise was a preservationist, not creative, effort. The editing of a Masoretic edition, such as the Jerusalem Crown, is oriented not to the moment (or period) of creation but to the time of textual fixation by the Ben Asher Masoretes. Yet even here intention may be relevant, insofar as an editor seeks to recover the intention not of the creators but of the preservationists. The preservationists' intention is relevant insofar as the editor has to determine the intention of numerous Masoretes whose notes often had to be decoded and evaluated. I choose the kind of intentionalist paradigm that aims at *authorship*, which to my mind makes sense only as an intentional act.

"Author" does not mean only the original author. There can be many authors in the course of a book's development, all of whom had something they wanted to communicate. In the case of Proverbs, authorship

<sup>8.</sup> Tanselle (1996a) surveys the increasing interest in the "collaborative or social aspects of text-production" and the "increased concern with textual instability and the significance of versions" (52). He applauds this broadening of perspective but protests at the reductionist tendency to caricature the concern with authorial intention, as if it insisted on one and only one right text by one and only one author. Intentionalist editing accommodates collaborative authorship and textual instability (54–56 and passim). The HBCE is able to do both.

is extremely diffuse and multiplex, distributed among innumerable and indeterminable individuals—authors of sayings, collectors and editors, and scribes who made *intentional* modifications. We may think of the author as a construct comprising that collectivity. It is a collectivity that promoted a certain worldview and ideology, and it is one with a "memory," in the sense that the later contributors knew their predecessors' work and both emulated it and went beyond it with literary and ideological innovations. The book of Proverbs has literary and conceptual coherence thanks to the cohesive force of tradition in the genre. Sages transmit old wisdom and use it as a model in the shaping of their own. Ideas evolve, expand, and retreat, but never in self-conscious opposition to earlier ones.

The book of Proverbs is not a haphazard assemblage of bits and pieces but a deliberate work whose growth was controlled by composers<sup>9</sup> who shaped the work by *selection*, choosing to incorporate sayings that were pertinent to their ideology; by *composition*, whether of stichoi, verses, or longer units; and by *placement*, through frequent clustering of sayings on a single topic. Given the complexity of the book's history, this collective authorship cannot be reduced to a "final editor." The collective author (or composer) is a personification of an abstraction. But given the homogeneity of the book's ideology and style, as well as the fact that the successive contributors to the book were aware of and influenced by the earlier compositional levels, collective authorship is a valid way of conceiving how the book communicates meaning.<sup>10</sup>

Proverbs is ultimately the creation of individuals who intended us to understand certain things. I do not know what we can read for—or write for—other than the communication of intention. Indeed, the very existence of speech and writing is a commitment to the recoverability of intention, and this recoverability is constantly confirmed when other people correctly retrieve the intentions we ourselves encode in speech and writing.

The recovery of intention is the essence of interpretation, and interpretation addresses units larger than the variants in question. That is why a careful holistic reading of a document can override typographical errors. If you read a sentence of mine with a meaningless typo (or, worse, a *mean*-

<sup>9.</sup> This term was suggested to me by Bernard Levinson, who suggests it as a way of embracing authors who create by both original composition and active editing, activities that are in practice often indistinguishable.

<sup>10.</sup> See ABP 1.322-30 for a discussion of part of the process.

*ingful* one), I hope you will mentally emend the text by reference to the "ideal" of the author's intention that you reconstruct from the wider context.<sup>11</sup> This text may never have existed precisely in a single text-form, but it is correct. Nothing other than the intended text is worth the reader's time. As Tanselle says,

Of all the historical activities of textual study, the effort to reconstruct the texts of works as intended by their creators takes us deepest into the thinking of interesting minds that preceded us. We must respect the documents that make our insights possible, but we cannot rest there if we wish to experience the works created by those minds. (1989, 92)

#### **1.5. Text and Interpretation**

Different eclectic texts can be created, depending on stated goals, and no text is definitive and final. This is because, to a large extent, every edition is produced in tandem with interpretation. That is not to say that the text produced is "just" an interpretation, a "reading" that is idiosyncratic or (according to one notion of authorship) spun off by a social or ideological context. It is, rather, a *truth claim* based on interpretation. Each choice is, at some point, right or wrong, not just more or less interesting. Still, critics, true to the word's etymology in *krinein*, are making judgments, and judges cannot remove themselves from their position between facts and verdict and still render a credible decision.

This is not *all* that text criticism is. There are preparatory steps, namely, the gathering of variants (including ones retroverted from the ancient versions) and consideration of bibliography, the history of physical books. When enough exemplars are available, it is also necessary to establish a stemma to determine the historical sequence of variants and eliminate some demonstrably later ones. Mainly one must eliminate erroneous mutations. This leaves an irreducible set of competing variants for consideration by the critic, who must at this point exercise literary judgment,

<sup>11.</sup> I take a simple illustation of "correct archetype" from my own history of errors. In ABP 1:365, line 1, the "archetype" (i.e., the printed form) reads, "The association of two passages on kingship is more striking and may have played a role in the movement of 31:1–9, but it does explain why chapter 30 also was split in two." The correct archetype is "but it does not explain." The physical archetype of this sentence makes sense superficially but not in context. I hope that the critical reader will reconstruct the correct form.

both exegetical and aesthetic. The following commentary is full of literary judgments. For example, in Prov 26:25 I establish two variants, דולקים ("burning") and הלקים ("smooth"). I prefer the latter on the grounds that "smooth [that is, dishonest] lips" better fits the imagery and the topic of the thematic cluster in 26:20–28, namely, hostile speech. In 18:16, of the two Masoretic variants, צדק ("righteous") and ארץ ("earth"), I prefer the latter, on the grounds that "all the judges of the earth" fits the universalism of this chapter and that the paradigmatic parallelism it provides is more coherent here.

A critical edition is constructed by means of an active engagement with the author's meaning, as transmitted by the text and interpreted by the editor. Hence every edition is an ideal construction of meaning, and the editor is a collaborator in its creation. Tanselle writes: "[A]ll editorial work, including that devoted to documentary editions, is-like all other acts of reading—a construction of meaning, which may or may not have a historical orientation; when editors publish their work they are simply offering new documents that can serve in their turn as the grounding for further creations of meaning" (2001, 67). Erne, who titles his study Shakespeare's Modern Collaborators, writes, "Since every Shakespeare play has been perceived to need emendation in a number of passages, every modern editor can be said to be participating in this authorial reconstruction" (2008, 20). The value of the construction will be judged by its conformity with authorial meaning, as recovered by interpretation. It should be stressed that those who prefer a Masoretic reading or an entire Masoretic edition are in effect participating in this construction of meaning, albeit passively, by aligning themselves with one text-state, a medieval one.

#### 1.6. MODELS OF EDITORIAL PRACTICE

Bible scholars are much impressed by the uniqueness of the Hebrew Bible and wonder if editorial practices applied to other texts are valid here as well.<sup>12</sup> The Bible is undoubtedly unique, but other works are unique as well. Biblical text-critical theory and practice must be comparable to what happens in other areas or risk being solipsistic, defensive, and celebratory. The modern editor of a biblical book faces a situation similar to what editors of many other works must deal with, and biblical text critics can ben-

<sup>12.</sup> See, for example, Williamson 2009, 163-64.

efit from attention to the theory and practices of text criticism and editing in other fields, for they often present similar problems and are amenable to similar solutions. The following two examples are intended as models for thinking about editorial practices in biblical text criticism. It goes without saying that no model will completely fit a distant domain of literature.

#### 1.6.1. Shakespeare

Shakespearian editorial practice provides illustrations of well-thought-out approaches to problems familiar to Bible text critics (which is not to say that the Shakespearians have solved *their* problems!). *King Lear* is an interesting example, because it exists in two significantly divergent forms, the Quarto of 1608 (Q1, slightly modified and reprinted in Q2, 1619) and the First Folio of 1623 (F). Scholarly consensus holds that Q was prepared from Shakespeare's holographs or "foul papers" (his working drafts, meant for theater use) in 1608, two years after the completion of the play and while the play was still being performed—and constantly reworked—under Shakespeare's direction. In spite of its proximity to the author, the Q text, especially in its earliest impression, is flawed.<sup>13</sup> F differs from Q in significant ways, including major pluses and minuses that bear strongly on the play's meaning and aesthetics. There are two basic ways to deal with this duality.

Until the 1960s, editors produced eclectic editions. Furness's 1880 Variorum *King Lear* takes F as the copy-text while embedding readings from the Quartos. (Earlier, Q was the copy-text.) Furness regarded Q as by far the inferior. The Variorum's upper apparatus records all variants not in the eclectic text. Subsequent editions as well have typically used F as the copy-text and corrected it by reference to Q, usually with a limited critical apparatus. The important Arden edition (1997, with *Lear* edited by R. A. Foakes) conflates F and Q, embedding in its F copy-text some 300 lines absent from F while retaining some 110 lines absent from Q (and marking all such passages).

Most contemporary Shakespearians, however, oppose conflation. The dominant view now is that Q and F are different recensions, each

<sup>13.</sup> The consistent application of the "lectio difficilior" rule to *Lear* would result in a maximal mess. Albrektson (1981) has shown the failure of this rule in Bible studies. In fact, its original intent was to identify glosses that made a difficult text easier to understand, not to validate whichever variant was the least comprehensible.

with its own integrity.<sup>14</sup> Though the path from Q to F was complex and is in dispute, the consensus is that the latter incorporates some of Shakespeare's own revisions.<sup>15</sup> Hence the Oxford second edition prints the two versions as two distinct plays. Taylor (in Wells and Taylor 1986), recognizing the integrity of Q and F, chose to use Q1 as a copy-text with collations from the "variant states" in other Q printings. Nevertheless, his edition incorporates certain F readings when Q is certainly corrupt. Weis (1993, 41–46) sets Q and F on facing pages but corrects both texts as necessary. Halio, most interestingly, published companion volumes, one (1992) based on F, the other (1994) on Q, each with its own introduction and commentary. Both texts are critical editions and use the other for corrections but are not conflated. Even editors committed to the integrity of each version must sometimes correct Q from F or F from Q (Wells and Taylor 1986, 8).

None of the extant text-states of Lear is considered correct, and unless an editor is producing a diplomatic edition, decisions must be made that aim at a better form than any surviving document. Moreover, it is recognized that there is not necessarily a single correct form. All critical editors of Lear are producing text-forms that did not exist previously. They all aim at restoring some phase of Shakespeare's intention, which is imperfectly recorded in several text-states. None of the major critical editions of Shakespeare merely reproduces a particular printing-Q1b, say-and relegates all variants, even preferred ones, to the apparatus. (That would be the Biblia Hebraica approach.) There are publications that simply reproduce a particular printing for its historical and academic value, but they are not intended to be critical editions. One uncritical edition was produced by G. Holderness (1995), a professor of cultural studies, who is in effect a Q-fundamentalist. He deconstructs the critics (whose preference for F, he explains, entails "a distinct complicity with both authorial intentionality and with a conservative critical agenda that invests its desires in the surviving male protagonists of the tragedy" [41]) and repudiates the practice of emendation as lacking "philosophical justification" and as

<sup>14.</sup> See the comments of Foakes (1997, 110–46); Weis (1993, 34–40); and Taylor (Wells and Taylor, 1986, 81–86). The following account is based primarily on the discussions by Weis (1993, 4–7, 34–40) and Halio (1992, 58–91).

<sup>15.</sup> The relation between Q and F calls to mind the relation between G-Jeremiah and M-Jeremiah, as described by Tov (1985) and others, though the revisions in the latter were not authorial.

"foreground[ing] the editor at the expense of the text" (9). A similar resistance to emending M is not unknown in Bible studies.

The present volume of HBCE will use the first approach, conflation, taking  $M^L$  as its copy-text and correcting from other editions, mainly G-Proverbs, in spite of the priority of the former and the integrity of the other editions. While the HBCE approach is innovative, even radical, in Bible studies, it is the standard practice among serious Shakespearian editors and, I believe, in other fields as well.

The aim of reconstructing a stage prior to any existing documentation legitimates combining elements from different versions or editions—such as Q and F for *Lear* or G and M for the Bible—a procedure that some have objected to.<sup>16</sup> We are using documents and versions not as goals in themselves but as evidence of the work that lies behind them.

#### 1.6.2. Jane Austen

The complexity of the growth and transmission of Shakespearian texts, together with their attendant uncertainties, resembles biblical textual history in significant ways. But we can also find parallels in editorial treatments of works with a very different, simpler kind of history, in which most details of composition, production, and publication are known. R. W. Chapman's 1923 Oxford edition of Jane Austen, recognized as a model of editorial care and brilliance, takes a particular printing as copy-text but make changes as necessary. These are minor, but their presence makes his unquestionably authoritative edition into a "composite." Indeed, the sub-

<sup>16.</sup> G. Brooke objects to eclectic texts of the Bible on the grounds that they "minimize the contribution of individual scribes and the specific creative traditions to which they may severally belong" (2005, 39). But this is precisely what noncritical editions, and even the best diplomatic editions, do by enshrining one text-state as *the* text, giving the impression that any variant from *this* is somehow a divergence from the correct text-form. This, in fact, is precisely the mentality Brooke objects to.

When properly annotated, an eclectic edition, such as the Arden and second Oxford editions of Shakespeare, preserves variants that would otherwise not come to public attention and gives them context in the history of the text. Brooke is certainly right that "each scriptural book has its own complex story to tell" (40). But an eclectic edition, with an apparatus, parallel columns for different editions, and a commentary that inquires into the textual history in all its complexity, is better equipped to tell the "complex story" than a diplomatic text that just lists variants below.

title of Chapman's edition is "the text based on collation of the early editions." It also includes conjectural emendations.<sup>17</sup>

Chapman does not try to reproduce Austen's original orthography, even though we can sometimes be quite sure what it was. (We have the autographs of two "cancelled" chapters from *Persuasion*.) He does not "restore" *friend* to *freind* (which Austen considered correct), *could* to *cd*, *admiral* to *adml*, or *Captain Wentworth* to *Capt*. *W*. (That would be like returning the Hebrew text to its unvocalized state and removing vowel letters.) Nor does Chapman restore Austen's frequent capitalization of nouns and her extensive use of dashes that appeared as paragraph divisions in printing. My point is that an edition can aspire to represent the most authentic text and to correct errors without attempting to replicate or restore the physical details of the original text-form. If this makes Chapman's edition a hybrid (and it does), it is an honor to have that label attached to HBCE.

#### 1.7. IN CONCLUSION

The following principles hold true of all critical texts, whether or not they are presented in eclectic form:

- 1. Every critical edition is a construction of meaning.
- 2. Every critical edition is an attempt to recreate the *work*, which is an ideal verbal construction.
- 3. The primary goal of textual criticism is representation of authorial intent, even when the author is an anonymous collectivity.
- 4. Different valid critical texts can be created, depending on stated goals and the nature of the text under study.
- 5. Every critical edition is an innovation.
- 6. A critical text (whether eclectic or diplomatic) is a hypothesis, which, however uncertain, is still a truth claim, one based on data and its interpretation and falsifiable by arguments in the same mode.

<sup>17.</sup> My favorite emendation: "p. 240, l. 30: *for it* A.C. Bradley: *it for* 1818 This elegant correction is perhaps not absolutely certain, but I have not been able to resist it" (Chapman's note in *Persuasion*, 295).

Textual criticism of the Hebrew Bible, of which an eclectic edition is just a structured manifestation, proceeds in spite of contradictions and insoluble dilemmas. But this is true of text criticism in other literatures too. In "The Monsters and the Textual Critics," T. Davis states three propositions:

- 1. Textual criticism is necessary.
- 2. Textual criticism is impossible.
- 3. Textual criticism is universal.

In other words, we must go on, we can't go on, we go on. (1998, 95)

In this spirit I will go on.

The Athenaeum reviewer's observation about the plethora of editions appearing in 1853 (see the epigraph), foreseeing "every intelligent man turning editor for himself," is, in a modified sense, my goal. It is true that every trained critical reader of the Bible is his or her own editor. When I undertake a serious reading of, say, Isaiah, I use the various tools available, especially the commentaries, to get ideas for variants. Some I accept, others I reject. In the process I form my own virtual critical edition, which is the "book" of Isaiah I read. It does not, and will not, appear in print, but it is still an "edition," and it is my own. I would like every intelligent reader of the book of Proverbs to be able to use the present edition and textual commentary in a similar way: to construct a virtual edition-a text-form that will include some of the variants I propose-and, undoubtedly, others I reject, and others I never thought of. This reader's virtual edition is the one that he or she considers most effective in understanding and appreciating the book of Proverbs in its formative stages. Producing this edition is truly collaborative work, the kind that sustains all scientific efforts.

# 2. The Hebrew Texts

## 2.1. HEBREW RESOURCES

2.1.1. The Masoretic Text (M)

2.1.1.1. The Leningrad Codex (M<sup>L</sup>). Codex Leningradensis, or, more precisely, the digitized form of the *Biblia Hebraica Stuttgartensia*, is the copytext on which the HBCE text is based. This codex is the oldest complete text of the Hebrew Bible. The colophon dates it to 1008 or 1009. Codex Leningradensis underwent corrections based on a Ben Asher text, and its vocalization is close to Aleppo's (see Tov 2012, 45).

2.1.1.2. The Aleppo Codex (M<sup>A</sup>). The Aleppo Codex (or כתר ארם צובה), dated to the tenth century CE, is the best representative of the Ben Asher school of Masoretes and has vocalization and Masoretic notes added by Aaron ben Asher. It is generally agreed to be the most authoritative Masoretic codex. Its text is very close to its Masorah. Most of the Pentateuch is thought to have been burnt in anti-Jewish riots in Syria in 1948 (though, in one view, it was gradually stolen [Friedman 2012]), but two-thirds of the codex, including Proverbs, survived intact. Collation with the Leningradensis of Proverbs shows only orthographic differences and fairly minor differences in vocalization policy, in particular Aleppo's more extensive use of compound (*hatef*) vowels for simple vocalic *shewa*. None of the differences are substantive variants that bear on the meaning of the text.

2.1.1.3. Yemenite Codex ( $M^{Y}$ ); Cambridge University, Add. Ms. 1753. Also consulted was a codex inscribed by Sa'adia ibn Yehoshua ibn Sa'adia in Yemen in 1577 CE. This is available in microfilm and is used by BHQ.

2.1.1.4. Kennicott-de Rossi Variants. Two great collections of consonantal variants were assembled by Benjamin Kennicott in 1776–1780 and Giovanni Bernardo de Rossi in 1784–1788 (KR). The manuscripts and printed editions collated are late medieval and Renaissance. In the absence of more than fragmentary evidence of the Hebrew Proverbs from ancient times, occasional attention will be given to these variants, in spite of the severe problems attendant on their use.

Kennicott's collection is full of errors, de Rossi's less so. Moshe Goshen-Gottstein (1960, 434), one of the few to examine a large number of these manuscripts directly, estimated that 95–99 percent of the hundreds of manuscripts collated are useless as a source of ancient variants. Moreover, the lack of information about the relations among the manuscripts means that counting manuscripts is useless. Multiple manuscripts in the collations may descend from one. Still, a reading in "many" manuscripts<sup>1</sup> is less likely to be a late creation than a reading in only one, since an old reading had more time to penetrate different manuscripts lines.

In the absence of an evaluation of the manuscripts used and their stemma, KR readings are poor evidence for premedieval variants. In a few cases, they agree with an ancient variant, and we may weigh the possibility that the variant survived in a minor Masoretic line. But the KR variant may have arisen for the same reason as the ancient variant and independently of it (Goshen-Gottstein 1954, 433). KR readings are occasionally mentioned as reflecting agreement with a variant and *possible* support for it. It is to be noted that KR manuscripts often agree with the *qere*, suggesting that the latter was sometimes a textual variant—or became one. KR's most important value is to exemplify the *kinds* of changes that arose in the course of transmission. They are cited for their insight into the transmission process and never undergird an emendation on their own.

#### 2.1.2. Qumran Fragments

The manuscript discoveries in the Judean Desert, in particular in the vicinity of Qumran, have energized biblical text criticism since the 1950s. Attention became especially vigorous in the wake of the full publication of the Qumran texts in the 1990s and early 2000s. Unfortunately, only a few fragments of Proverbs remain. These are:

4QProv<sup>a</sup> (= 4Q102), with parts of 1:27–2:1 4QProv<sup>b</sup> (= 4Q103), with parts of 13:6b–9b; 14:6–10; 14:31–15:8; and 15:19b–31.

<sup>1.</sup> BHS designates more than twenty manuscripts as "multi."

In 4QProv<sup>b</sup>, the right column is written stichographically, with one couplet per line, showing that the stichometric division of Hebrew verse, based on parallelism, was recognized in ancient times. The script of 4QProv<sup>a</sup> belongs to the mid-first century BCE, that of 4QProv<sup>b</sup> to the turn of the era. There is also a quotation (or paraphrase) of 15:8 in the Damascus Covenant XI, 20–21.

The critical edition of the Qumran fragments appeared in *Qumran Cave 4* (DJD XVI), 181–86 = Ulrich et al. 2000. Ulrich notes the following variants:

4QProv<sup>a</sup> 1:32a: מושכת for M משובת

4QProv<sup>b</sup>

14:34: וחסד for M וחסר (uncertain) 15:19: סללה for M סוללה 15:28: יהגה יomitted

All the fragments are accounted for in the commentary.

2.2. The Treatment of the Hebrew Text

2.2.1. Vocalization

In accordance with HBCE policy, the eclectic text of Proverbs maintains the vocalization of the copy-text except when emendation is deemed necessary. Emendation of vowels is required either when the consonantal text is modified or when the consonants are maintained but the copy-text's vocalization is deemed mistaken.<sup>2</sup> In the commentary I also vocalize meaningful variants that are not chosen for the HBCE text.

Avoiding vocalic emendation would create a strange situation in which the editor emends consonantal graphemes that are (mostly) ancient while treating as sacrosanct the graphemes that are indisputably medieval. *Matres lectionis* are in principle no different from the Tiberian vowels, except that their *usage* is more ancient. The particular vowel letters in

<sup>2.</sup> Williamson's objections to unvocalized reconstructions (2009, 165–67, referring to the planned Oxford Hebrew Bible) were valid for the earlier stage but are not relevant to HBCE.

M are not necessarily ancient. As a perusal of KR shows, the process of adding vowel letters continued beyond the Masoretic period, including in L and other Masoretic manuscripts.

All words retroverted from the versions are vocalized so as to represent intentions implicit in the translation. When, for example, G renders the second word in Prov 3:24 as  $\varkappa \alpha \delta \eta$ , he not only saw  $\neg \omega \Box$  but also thought—and possibly spoke—a word that means "you will dwell." Whatever the actual sound was, it would be represented in Tiberian vocalization as  $\neg \omega \Box$ .

The intentionalist paradigm of editing, discussed in §1.4, provides a rationale for both maintaining the Tiberian graphemes and emending vocalization in a much earlier consonantal text. In either case, the editor is accepting a certain interpretation of the authorial intention. To be sure, in the period when the archetypes and hyparchetypes were formed, the vocalization existed only in the minds of authors and readers and in oral reading, except insofar as it was represented by vowel letters. But the intentionalist paradigm makes such mental forms relevant and their representation appropriate. The vocalization represents, ideally, the way the authors *would have* vocalized the text had they been using the Tiberian phonemic system and vocalic graphemes. This is a stretch, but we face it every time we read M. And if the HBCE text is a hybrid, so is the Masoretic one.

The Tiberian vocalization maintains important and ancient information, though it is, of course, not infallible (see Joosten, forthcoming). Of course, the Tiberian vocalization does not accurately represent the phonetic realities of the varieties of Hebrew spoken in the biblical period. That kind of representation belongs to historical linguistics. However, the reconstructed portions of the Hebrew text represent the phonetics of the inferred ancient text in the same way that the Tiberian vocalization does for the Masoretic sections. Since there was probably a one-to-one correspondence between Tiberian full vowels and the ancient vocalic phonemes, the distortion is minimal. In any case, this approach to the vocalic elements of the text produces an ancient-medieval hybrid with modern reconstructions that aim to restore an earlier stage of the text's development.

Even when aiming at establishing an ancient text-state, one is justified in both maintaining and modifying the medieval vocalization. Failure to do so would diminish the clarity of the text and conceal (rather than resolve) obscurities. Nevertheless, vocalization is not without cost, for it can exclude intentional and interesting ambiguities. Gary Martin (2010) emphasizes the potential significance of visual ambiguities and insists that they should often be maintained and appreciated.

The book of Proverbs, which claims to hold "enigmas" (1:6), should be a prime source of puns and ambiguities, both aural and visual. Tur-Sinai's 1947 monograph on Proverbs is a treasury of enigmas, though one can never be certain where they actually reside. One striking case is 25:23: The translation I prefer is רוח צפון תחולל גשם ופנים נזעמים לשון סתר "As a north wind brings rain, so a sly tongue brings angry looks" (ABP). Tur-Sinai (1947, 29–30) proposes a secondary meaning in the same words: "Hide [your] spirit [= emotions] [before your superiors] and you will cause a rain [of good will], [but be careful that your expression does not reveal the way you feel,] because an angry face is a language [that can reveal] what is hidden." The Hebrew does permit this amazing expansion of the terse original. Strictly speaking, Tur-Sinai's reading would require the vocalizations for לָשׁוֹן סַתָר, but a reader of the consonantal text who was alert to potential ambiguities could choose to understand it both ways at once. While I must opt for one vocalization in the HBCE text, the scholar should be alert to productive ambiguities in the consonantal text, and the commentator should draw attention to them. Some of them are noted in ABP and in the commentary in the present volume.

#### 2.2.2. Accents

HBCE policy calls for the preservation of the Masoretic accents, except in emended words, where they are deleted. There is little choice but to omit them when emendations other than very minor ones are made in the text of Proverbs, because its accent system (shared by Psalms and Job) is not well enough understood to allow for reshaping the accent structure of a verse, and a change of one accent may undermine the integrity of the accentual unit or the entire verse.

It is admittedly inconsistent to delete accents while supplying vocalic graphemes just because the latter seem more important nowadays. The Tiberian vocalic system and the accentuation grew together and are in many ways interdependent. Thus emending one and not the other can result in incongruence. This incongruence is present, though obscured, in diplomatic editions such as the Biblia Hebraica series, which proposes emendations of words while usually ignoring the accents. It is one of the peculiarities of biblical text criticism that we will emend the ancient level of the text while holding the medieval level sacrosanct. An eclectic edition, however, cannot escape the fact that the emendation of the former has consequences for the latter.

In the HBCE text I delete the accents in emended words, with certain exceptions: (1) I retain the 'atnah when the syntax is undisturbed. A verse without its first dichotomy marked would be anomalous. Since most of Proverbs is composed of couplets with the dichotomy clear from the parallelism and syntax, this is usually unproblematic. (2) I maintain the silluq at the end of the verse because every verse must end with one. (3) The maqqeph is retained or sometimes added when a new vocalization requires it. The maqqeph belongs to the vocalic as well as the accentual system. (4) I also retain the accent when the change I make is trivial and could have no effect on the syntax, as when I use a defectiva ketiv but retain the vowels, which in principle belong to the qere. (5) Finally, I retain the accent even in subordinate dichotomies when an emendation leaves the syntax unaffected, as when replacing הַצָּבָה by הַצָּבָה in 9:1. It seems best to leave the accent system of a verse intact if this can be done with confidence.

But emendations can disturb the accents of phrases outside the marked emendations, as when the change involves omissions, additions, or transpositions, as happens notably in 7:23. I do not see a solution for this besides omitting the accents of the affected unit. This omission is an emendation too, though it is incomplete because I cannot show what the new accents should be, and as such will be marked with ceiling brackets.

It should also be noted that the *petuhot* and *setumot* are represented, as in most printed editions, by  $\mathfrak{D}$  and  $\mathfrak{D}$ , respectively. In the codices they are shown more prominently and accurately as blank lines for *petuhot* and a space within the line for *setumot* (as in Prov 24:23). The *petuhot* mark the beginnings of *pisqaot* (paragraphs), which in Proverbs largely correspond to recognizable literary units. The practice of *pisqaot* is found in some Qumran manuscripts, though their divisions do not usually match the Masoretic ones. See further Yeivin 1980, 42–44.

The analogy of the Shakespearian model, discussed in §1.6.1, can help us think about the treatment of the Masoretic accents. Elizabethan punctuation was ambiguous and fluid, and scribes and printers tended to regard it as *their* prerogative. Shakespeare himself punctuated lightly, since he wrote for his own theater and would be present to prompt the players in their phrasing (see Wells and Taylor 1986, xlii). One of his copyists, Ralph Crane, is known to have imposed his own system of punctuation on the texts he transcribed. Stage directions, act-scene numeration, and speaker designations were introduced largely in the First Folio (1623) and are often wrong. Modern editors typically try to maintain the accidentals of the best texts, but they are often forced to alter some and add others (Oxford second edition [edited by Wells in Wells and Taylor 1987, xlii; cf. McKerrow 1939, 40–43]). The editors do *not* simply wipe out the punctuation marks in an attempt to restore the state of Shakespeare's drafts or even his fair copies. The Masoretic accents, like many of the accidentals in the early Shakespearian editions, were added as an aid to *performance*.

#### 2.2.3. Ketiv and Qere

2.2.3.1. Practices in This Edition. In the case of *ketiv-qere* variants, the default for the HBCE text is the *ketiv*. The *ketiv*, after all, is the *text*. The Masoretes make the *ketiv* (בְּתְבִין ון בְּתִיב), *ketivin*) the default, while they relegate the *qere* (בְּתַבִין ון בְּתִיב), *qerayin*) to the margin, in the Massorah Parva. (In the present volume, the nonpreferred form is relegated to the apparatus, with the commentary). In fact, the entire text in a Masoretic manuscript is the *ketiv*. Another reason to use the *ketiv* as the default is that many of the *qerayin* serve as a "*sic*," a warning *not* to write the *qere* form, even if the word is to be understood this way. We know this to be the case when a stabilizing note in the Mp of one manuscript turns up as a *qere* in another. For example, Mp<sup>A</sup> in 3:27 has just ''תִיר' ("a superfluous *yod*"), which tells the copyist to write <code>TT'</code>, not <code>TT'</code>. Mp<sup>L</sup> has 'part' ("the read"). See further below, §2.2.3.3.5.

In the codices, including Leningradensis and Aleppo, the *ketiv* holds the vowels that properly belong to the *qere*. In the HBCE text, if the vocalization of the *qere* fits the *ketiv*—as it often does, since the *qere* is typically an orthographic disambiguation—it is maintained without further remark. If the *qere* is preferred, it is marked by "ceilings" ( $^{r}$ …  $^{1}$ ), for it is a departure from the copy-text and therefore an emendation. Likewise, if the *ketiv* is preferred but is given a vocalization other than what stands in the copy-text, the departure from the copy-text is indicated by ceilings as well. (In the tables below, the vowels are placed on the *qere*.). In the apparatus to the HBCE text, I cite the *qere* as M<sup>Q</sup>, without indicating the source manuscript. That information is given in the tables below. Mostly the *qerayin* are found in M<sup>L</sup>.

In the apparatus, I align the versions—G, S, V, and T—with the appropriate member of each pair, insofar as this is possible. In some cases, the versions may witness to the presence of the *qere* in their source manuscripts, but the evidence is usually ambiguous. When a translation agrees with a *plene* writing, which is usually the *qere*, this may indicate only that the translator interpreted the *defectiva* writing in the same way as the *qere* does. Moreover, all the versions are flexible in representing grammatical number. When I group the versions with members of a *ketiv-qere* pair, they are to be understood as data, not necessarily support, for one of the members. For the purpose of aligning the versions with members of a pair, I treat the translation as precise and mimetic, though of course it is not necessarily so.

2.2.3.2. *Ketiv-Qere* Pairs in Proverbs. The following are the *ketiv-qere* pairs marked as such in  $M^{I}$ , plus some absent from  $M^{L}$  but present in  $M^{A}$  or  $M^{Y}$ , as noted. Mp notes other than *qerayin* are mentioned. Sometimes a *qere*-type reading is implied by a note in the Mp, and these virtual *qerayin* are given in parentheses.<sup>3</sup> As for the relation between the members of the pairs, it is the *qere* that has to be explained, because it is a comment on the *ketiv*, but that is not to say that the *qere* is somehow wrong. It may be the historically correct text. In the tables, I call the *ketiv* or the *qere* "correct" only when I consider one member alone to fit the context or to have a grammatically correct form, in other words, the one usual in the Tiberian pointing. My considerations are explained in the commentary. Usually both forms are possible. Unless otherwise noted, the *qerayin* in the following table are shared by  $M^{L}$  and  $M^{A}$ . The Mp column gives Mp notes on the word in question other than 'p.

Table 1. Graphic Interchange 1/'					
verse	rerse <i>ketiv qere</i> Mp comments				
3:30	תרוב	תָּרִיב		qere is correct	

<sup>3.</sup> While many Mp notes can serve as virtual *qerayin*, they should be distinguished from actual *qerayin*, which are marked as such in a manuscript. The Mp in BHS, produced by G. E. Weil, does this frequently, but BHQ does not. Dotan also supplies a *qere* in his edition of the Leningradensis "whenever the reader's convenience requires a *qere* and the manuscript has a *yatir* (= superfluous) indication instead of a *qere* … but not everywhere the ms has a *yatir*" (2001, xx–xxi). This seems to me to mix modern hypotheses with a medieval source in an edition that (unlike the present volume) claims to replicate that source. But it is clear that earlier Masoretes created *qerayin* just as some editions, including the Second Rabbinic Bible, BHS, and Dotan, do, and this is one reason not to assume that *qerayin* are textual variants.

4:16	יכשולו	יַבְשִׁילוּ		alternatively, both may be expansions of יכשלו
11:3	ושדם	יְשָׁדֵם		qere is correct
12:14	ישוב	יָשִׁיב		
17:13	תמיש	תְמוּשׁ		qere is correct
17:27	וקר	יְקַר		<i>ketiv</i> is correct
18:17	יבא	וּבָא		
20:30	תמריק	הַמְרוּק		
23:5	התעוף	<u>ה</u> ַתָּעִי <b>ו</b> ּ		
23:5	ועיף	זָעוּף		qere is correct
23:24	גול יגול	גּיל יָגִיל		qere is correct
23:31	בכיס	בַּכּוֹס	ו' ק' Mp <sup>A</sup>	qere is correct
31:4	אי	אַנ		correct form is אָי
31:27	M <sup>A</sup> הילכות [ <i>sic</i> ]; Q in M <sup>A</sup> only: הֲלִיכוֹת		Q in M <sup>A</sup> only: הַלִיכוֹת	

	Table 2. Other Graphic Interchanges, Including Homoioteleuton					
verse	ketiv	qere	Мр	comments		
3:15	מפניים	מִפְּנִינִים		homoio; <i>qere</i> is correct		
19:19	גרל	ڋ۪ڽٙڔ		ד/ר; <i>qere</i> is correct		
20:16	נכרים	נְכְרִיֶּה		ם/ה; <i>ketiv</i> is correct		
20:21	מבחלת	מְבֹהֶלֶת		ה/ח; qere is correct		
21:29	יכין	יָבִיז		ב/ב; <i>qere</i> is correct, with דרכו, see in table 5		

## 2. THE HEBREW TEXTS

	Table 3. Metathesis of Letters					
verse	ketiv	qere	Мр	comments		
1:27	כשאוה	רְשׁוֹאָה		۱/೫; <i>qere</i> is correct		
19:16	יומת	יָמוּת		۲/۱ (or grammar)		
23:26	תרצנה	תִּצֹרְנָה		ר/צ; <i>qere</i> is correct		
31:27	M <sup>A</sup> הילכות [sic]	M <sup>A</sup> הַלִיכוֹת		a correction		

	Table 4. Aural Interchange					
verse	ketiv	qere	Мр	comments		
19:7	לא	לו		uncertain		
26:2	לא	לו		<i>ketiv</i> is correct		
27:20	ואבדה	וַאֲבַדוֹ	<b>ک' </b> Mp <sup>A</sup>			

	Table 5. Number Disambiguation or Number Variants					
			<i>ketivin</i> ending in 1 could ut the <i>qere</i> treats them as			
verse	ketiv	qere	Мр	comments		
2:8	חסידו	חֲסִידְיו	Mp <sup>A</sup> has only 'ג' חס', i.e., this precise word occurs 3x <i>defectiva</i> .	Mp notes that the <i>ketiv</i> should be written <i>defectiva</i> .		
3:27	ידיך	יָּדָ <i>ר</i> ָ	Mp <sup>A</sup> has only 'יתיר י'.			
3:28	לרעיד	לְרַ <u>ע</u> ַד		orthographic normal- ization		
6:13	בעינו	<b>רְּעֵינְיו</b> M <sup>AY</sup>	Mp <sup>L</sup> nul			
6:13	ברגלו	ַבְרַאָלָיו M <sup>AY</sup>	Mp <sup>L</sup> has only 'ח' הס', i.e., 8x <i>defectiva</i> .			

6:16	תּוֹעֲבַות	תּוֹעֲבוֹת M <sup>A Y</sup>	'יתיר ו in Mp <sup>L</sup> .	
16:27	שפתיו	שְׁפָּתוֹ		Unusually, the <i>ketiv</i> is the orthographi- cally ambiguous form; שפתו must have been a known variant, of which the <i>ketiv</i> was a disambiguation.
21:29	דרכיו	דַרְכּוֹ		as in 16:27
22:8	יַקְצוֹר	(יִקְצָר־)	Mp <sup>L</sup> 'ג' מל', Mp <sup>A</sup> ג'	Both Mp and M <sup>A L</sup> remind the copyist to write the word <i>plene</i> ; Mp <sup>L</sup> implies that the correct Tiberian form, though an actual <i>qere</i> , is not given; M <sup>Y</sup> has neither a <i>qere</i> nor a Mp note.
22:11	טהור	יְּשְׁהָר־ (also M <sup>YK</sup> )	Mp <sup>A</sup> has only 'ל' ומל' (i.e., this is the only occurrence written thus; elsewhere this precise form is written <i>plene</i> ).	
22:14	יפול	יַבָּל-	M <sup>Y</sup> has יפל as its <i>ketiv</i> .	
22:20	שלשום	שָׁלִישִׁים		Ketiv and qere are both (mistaken) plene writ- ings of consonantal שלשים, with the ketiv showing a ז→' change; see the commentary.
22:25	ארחתו	אֹרְחֹתָיו	Mp <sup>A</sup> : 'תיו ק	
23:6	תתאו	הִּתְאָיו	M <sup>AY</sup> nul	cf. 23:3: תִּתְאָו
24:1	תתאו	הִּתְאָיו	Mp <sup>⊥</sup> איו ק'; M <sup>A</sup> γ nul	cf. 23:3: הִתְאָו

24:17	איביך	אוֹיִבְידָ M <sup>A Y</sup>	Mp <sup>L</sup> : 'יתיר י	
26:24	בשפתו	בִּשְׂפָתָיו		
30:10	אדנו	אֲדֹנְיו		

\* Singular/plural *ketiv-qere* variants and ostensive singular/plural *ketiv-qere* variants are frequent, and it is usually difficult to give preference to one or the other. Gordis 1971, 86–92, has a list of such pairs. Usually the *qere* is a plene guide to reading the *defectiva ketiv*.

	Table 6. Grammatical Variants					
verse	ketiv	qere	Мр	comments		
2:7	וצפן	יֹצְפֿן		verb tense or mode		
4:16	יכשולו	יַרְשִׁילוּ		different expansions of יבשלו		
13:20	הלוך	הוֹלֵדְ		verb tense or mode		
13:20	וחכם	ָּיָחְכָּם גָּחְכָּם		verb tense or mode		
20:4	ישאל	ٳؗؗؗڣۣ؉ۣۘۯ		verb tense or mode		
22:3	ויסתר	וְנִסְתָּר		verb tense or mode		
23:24	יולד	ויוֹלֵד		± conjunction		
23:24	וישמח	יִשְׂמַח		± conjunction		
27:10	ורעה	וְרַעַ		normalization		
27:24	דור	וָדוֹר		± conjunction		
28:8	ובתרבית	וְתַרְבִּית		+ preposition		
28:16	שנאי	שנא		number		
30:18	וארבע	וְאַרְבָּעָה		normalization		
31:16	נטע	נְטְעָה		normalization		

א בליל 31:18 מיל 31:18 מילי archa
--------------------------------------

Table 7. Erroneous Form in the <i>Ketiv</i> (Mp protects the difficult <i>ketiv</i> while allowing the verse to be read comprehensibly)					
verse <i>ketiv qere</i> Mp comments				comments	
8:17 אֹהַבִי אהביה <i>qere</i> is correct					
8:35 איתיר י' מָצָא MA מצאי Mp <sup>L</sup> <i>qere</i> is correct					

Table 8. Synonyms or Different Word Forms					
verse	ketiv	qere	Мр	comments	
3:34; 14:21; 16:19	ולעניים	וְלַעֲנָוִים		see the commen- tary on 3:34	
6:14	מדנים	מִדְיָנִים			
18:19; 21:19; 23:29; 25:24; 26:21; 27:15	מדונים	מִדְיָנִים			
15:14	ופני	וּפִי		<i>qere</i> is correct	
20:20	באישון	בֶאֱשׁוּן			

2.2.3.3. Theories of the Origins and Functions of the *Ketiv-Qere* Readings. The *ketiv-qere* variants have been explained in various ways. They are surveyed for the entire Bible and evaluated by Robert Gordis (1971, 7–28) and for the Pentateuch and Former Prophets by Maimon Cohen (2007, 2–7). I consider all these explanations as at least partly valid but suggest a way of most effectively explaining the systems in §2.2.3.3.5.

2.2.3.3.1. A Variety of Types. Gordis (1971) argues that the *qere* system includes different types of annotations. He hypothesizes three phases in

the development of the *qere* system: (1) the substitution of readings for the Tetragrammaton and of euphemisms for terms considered indecent; (2) the addition of readings to disambiguate the base text (this must have happened, he notes, before the addition of vowel signs); and (3) the collation of worthwhile readings from other manuscripts.

2.2.3.3.2. *Qere* as Correction. The idea that the *qere* readings are corrections of a faulty text is no longer accepted. It was first proposed by Abrabanel in the introduction to his commentary on Jeremiah and accepted by several Reformation scholars (see Gordis 1971, 12). Various scholars have objected that the Masoretes were conservators of a tradition and correction would have run contrary to "the spirit of the Masorah" (1971, 24). Moreover, many cases of *qere* are faulty and can hardly be considered corrections. In addition, the *qere* is "superior" to the *ketiv* (in grammatical and semantic appropriateness) in only a minority of cases (two hundred in Gordis's calculation; 1971, 23–24), so textual correction does not seem to be the motive.

2.2.3.3.3. Collation: *Ketiv* and *Qere* as Manuscript Variants. Observing that many *qerayin* are inferior to the *ketivin* or even devoid of meaning or contrary to grammar, Harry Orlinsky (1960) argued that they cannot be corrections. Instead he proposed that both the *ketiv* and the *qere* derive from mechanical collation of three manuscripts. The majority reading became the *ketiv*, the minority one the *qere*. Orlinsky conjectures that when there were three variants, the collators (now exercising judgment) dismissed the least likely one (1960, 191–92).

Orlinsky's theory is pure conjecture, and the Qumran Bible manuscripts have not supported it. Moreover, if the purpose of the system were to preserve variants, the Masoretes could have listed two marginal variants as well as one, and it is unlikely that their manuscripts always agreed on at least two forms. Another objection is that the *qere* is very often an expansion of the *defectiva* writing of the third masculine plural suffix, the one in which there is the most frequent ambiguity (see Gordis 1971, 87–92). There is no reason to think that two of three manuscripts would consistently have the *plene*. Also, some Mp notes, such as ' $\aleph$ ' in 24:1 and some others in the tables, are not a means of preserving variants and belong entirely to the realm of scribal practice.

Orlinsky says, puzzlingly, that there was no attempt to mark *plene-defectiva* issues (1960, 189), as would be expected if the *qere* were a correc-

tion. But in fact *plene* writings are common and well-exemplified among the *qerayin* (see Gordis 1971, 86–92 and the lists above in §2.2.3.2). For further arguments against the collation theory, see Albrektson 1978.

2.2.3.3.4. Two Traditions. In a thorough study of all the *ketiv-qere* pairs in the Aleppo Codex of the Pentateuch and Former Prophets, Cohen (2007) argues that the system of *qerayin* originated in the oral reading tradition. (This was earlier suggested, briefly, by James Barr 1981, 36.) Cohen objects to atomistic approaches and seeks to provide a single, inclusive explanation that can account for every detail of the system (2007, 9). Cohen argues that the *ketiv* is the written tradition and the *qere* the oral reading tradition. They are equally valid traditions that (he conjectures) go back to the biblical authors (310). Cohen states his adherence to "the philological approach, the approach that accepts the text as it is written, as it is set before us today" (9). He precludes explanations that posit textual errors, though he does not deny the possibility of their existence.

Certain objections may be raised to the written-oral dichotomy, especially as formulated by Cohen:

1. A monocausal explanation may be reductionist and force diverse phenomena into a single mold.<sup>4</sup>

2. There are certainly textual errors in the *ketiv-qere* systems (examples in Proverbs are 1:27; 19:19; 20:21; 23:5, 26; and 26:2). They do not disappear just because one claims loyalty to a "philological" explanation. Sometimes both forms are wrong, from a literary-philological point of view; see Gordis 1971, 155.

3. Likewise, while one may claim that the two *systems* are, in principle, "equally valid," this is not true of individual variants. Some are simply wrong, and when this is the case, it is almost always the *qere* that is correct or at least closer to the grammatical norm. In Proverbs, I consider the *ketiv* preferable to the *qere* only in 17:27; 20:16; and 26:2, but these are my literary judgments and in the commentary I explain how a scribe could have assumed that the *qere* was better and produced it as a cue to the meaning of the verse.

<sup>4.</sup> By way of analogy—a strong one—consider that modern footnotes serve several different functions (citation of source; addition of information; statement of tangential thoughts or data; interaction with others' ideas). What they share is an attempt to clarify something in the body of the text.

Cohen strains credulity when he rationalizes erroneous forms such as *ketiv* (supposedly meaning "made ten times as much" = "made a lot"; Cohen 2007, 278–79) for עָשָׁר (1 Kgs 22:49; Cohen 2007, 278–72); or 2007, 278–79) for עָשָׁר (1 Kgs 22:49; Cohen 2007, 278–72); or *ketiv* (supposedly meaning that Caleb looked "like his heart") for כלבו meaning "made distant") for בָּלָבִי *qere* (1 Sam 25:3; Cohen 2007, 285–86); or איש ויא ketiv (supposedly meaning "made distant") for <u>ויד</u> *qere* (2 Kgs 17:21; Cohen 2007, 279–81); or *ketiv* (with איש חי for וייד א supposedly meaning an "animal-like man," hence a swift soldier) for עולד *apere* (2 Sam 23:20; Cohen 2007, 296). The midrashlike logic of these explanations is a flaw. However, it does suggest ways the early readers could impose sense on an erroneous *ketiv* without considering it *wrong*. The *qere* was meant to show how to understand the *ketiv*, not how to fix it.

4. We might also question whether the *qerayin* consistently represent an oral tradition. Most of the *qere* readings are clarifications of an ambiguous writing, such as בְּרַגְלָיו/ברגלו , and make no difference to pronunciation. A scribe may have heard the word pronounced, and that pronunciation may have derived from a "tradition" (especially if it is right), but it is also possible that the scribe chose the plural rendering because that made sense to him. A great many times, the *defectiva* plural was vocalized as /āw/with no comment. Why was the "tradition" ignored there? The frequent agreements of *qerayin* with ancient translations may show only that a translator and a Masorete came to the same conclusion. There are hundreds of *plene/defectiva* variants in the KR collections and even in the better controlled group of Masoretic manuscripts listed in Breuer 2003, so we need not assign the *plene* writings in *qerayin* to a tradition. They simply exist.

Even if pronunciations were preserved in oral reading, the *qere* is more than a recording of what was heard. For example, oral readings would have preserved the pronunciation /titāw/ for תתאו, and this was later noted in the Mp as איו ק'. But the oral tradition did *not* preserve the spelling איו ק'. This kind of *qere*, and there are many examples, was meant for copyists.

5. In any case, there is no evidence for a single, continuous oral tradition parallel to the textual transmission. There were undoubtedly reading practices that maintained one or another pronunciation. These are evidenced in the Nahal Hever Minor Prophets manuscript, which adjusts the Greek translation toward M, including its vocalization (Barthélemy 1963, 169–98).<sup>5</sup> But this is not evidence for a single tradition stream extending

<sup>5.</sup> Examples from 8Hev XIIgr ("R"): Hab 1:5, ἐκδιηγῆται is changed to ἐκδιηγ]ήθη = M's passive יָשָׁמֵן; Hab 1:16, ἐλίπανεν is changed to ἐλίπανθη = M's G-stem יָשָׁמֵן. The

from authors to the *qerayin*. Moreover, the books not meant for public reading, such as Proverbs, Job, Ezra-Nehemiah, and Chronicles, could not have supported a precise and unchanging oral tradition.

6. Cohen is right that the *qerayin* readings are not corrections, for, as he says, the nature of the Masoretic enterprise was to preserve texts, not to correct them (2007, 6). Further, if the Masoretes were correctors, why did they leave numerous flawed variants in the written text untouched? But to speak, as many do, of "corrections" is a biased formulation, for whatever the *qerayin* do, they do not "correct" the body text. They are not textual corrections, similar to modern emendations, not even in cases in which a scholar might judge a *qere* preferable from a grammatical, literary, or historical standpoint. The *qerayin* are incorporated in the Mp along with numerous other annotations without tampering with the text proper.

\$2.2.3.3.5. The *Qerayin* as Cues. The *qere* typically provides some form of normalization or disambiguation of the *ketiv*. When one of the members of a pair is wrong by grammatical norms (for example, showing numerical noncoordination) or is semantically inappropriate, it is usually the *ketiv* that is faulty (e.g., Prov 3:15, 30; 4:16; 8:17, 19; 19:16, 19; 20:21; 23:24, 26, 31; 31:4, 16). In Proverbs, the *qere* is in some sense faulty in 17:27; 20:16; and 26:2, and even these instances could have been accounted for reasonably, if mistakenly, by exegesis. The imbalance in favor of the *qere* shows that the assembling of *qerayin* did not proceed by mechanical collation or even the wish for preservation of readings. Their function was rather to help the copyist record the text accurately, which meant reminding the scribe how the text is *not* to be written and, perhaps incidentally, informing the oral reader how it is to be pronounced.

The Masoretes did not create all the *qerayin* on their own. Some *qerayin* may have existed in manuscripts and possibly were known to the Masoretes in that form, but this does not mean that the ones recorded in the Mp, including the correct ones, were intended to preserve an alterna-

following statistics (taken from Tov 1990a, 147–53) show how the sources align with regard to vocalization: (a)  $R = M \neq G$ : 15x (+ 5x in reconstruction); (b)  $R = G \neq M$ : 21x; (c)  $G = M \neq R$ : 2x; (d)  $R \neq M$ , G 6x. The group (a) is where R probably shows a dependence on an oral practice similar to M's. Overall, R moves closer to M's vocalization but not in a radical fashion that might show a fully developed reading tradition at this time. None of the above cases have *qerayin*. There are no cases extant of R correcting toward a *qere*.

tive textual form. It is just that the variant was used as the source of the *gere*-cue.<sup>6</sup>

A clue to the function of the *qere* readings is the way they and their equivalents actually appear in the Mp. The Mp often gives the *qere* in an abbreviated form. For example, in Prov 23:6, where the *ketiv* is אויביד, the note in Mp<sup>L</sup> is 'אוי ק'. Or Prov 24:17, were the *ketiv* is 'אויביד, M<sup>L</sup> specifies 'מורביד' ("an extra *yod*"). That is to say, the word is to be *understood* without the plural marker *yod* but *written* with it. This is not a correction but the equivalent of "*sic*," and it only implies that one is to understand the word as a singlular, אויביד.

The *qerayin* had various origins, but they all serve as prompts to what was supposed to be the correct writing, pronunciation, and understanding of the text. The *qerayin* functioned much like the *sebirin*, except that the latter were to be thought but not pronounced.

#### 2.2.4. Oriental (מדנחאי) and Occidental (מדנחאי) Readings

For the sake of completeness I have noted some places where the Oriental and Occidental readings differ, when the difference is more than trivial and not confined to *qere-ketiv* specification. The identification of the Oriental readings is very problematic, for there are many differences among the manuscripts that purport to list them. The best resource for these readings is Hayut 2013. The sources of the Oriental readings are cited and discussed by Hayut; see esp. 95–98.

<sup>6.</sup> In Gordis's judgment, in cases where the *ketiv* and *qere* are not equally satisfactory or unsatisfactory, the *ketiv* is the better 12.34 percent of the time, while the *qere* is superior in 18.57 percent of the cases (1971, 82).

## 3. The Versions

#### 3.0. INTRODUCTION

In the absence of major documentary evidence in Hebrew other than M itself, the commentary will have to give close attention to the versions, inquiring into the possibility that they transmit ancient Hebrew variants, whether right, wrong, equally valid, or (as often) uncertain.

The most important version is G-Proverbs. It will be examined in considerable detail because it undoubtedly witnesses to a different Hebrew text of Proverbs, indeed to a different edition, and some of its variants can be recovered by careful probing. S-Proverbs will also get close attention. While largely aligned with M, S sometimes witnesses independently to Hebrew variants. Their recovery is complicated by the extensive use S makes of G in this particular book. I do not consistently mention T and V. T-Proverbs is a transposition of S into Western Aramaic characters and morphology, with strong assimilation to M. V basically works from M with consultation of G, and it almost always reflects one or the other. T and V are cited only in limited circumstances: in cases of *ketiv-qere*, when they show some independence of M and G, when they confirm a variant in an unexpected way, and sometimes when I just find their interpretation interesting and want to mention it among the others.

References to textual scholarship prior to the mid-nineteenth century are usually based on Baumgartner, Pinkuss, or (most often) Lagarde. These are commonly cited as "ref," that is, "refers to." Jäger's seminal *Observationes* (1788) proposed some of the best retroversions and emendations, and the present study uses many of them.

It should be noted that "M" very often refers to the consonantal text transmitted in the Masoretic tradition, not the Masoretic Text proper, with its vowels and accents. The versions often use the consonantal M while understanding it differently than the Tiberian tradition does. Such interpretations may be called "implicit vocalization."

#### 3. THE VERSIONS

#### 3.1. The Septuagint $(G)^1$

#### 3.1.1. Edition Used

The Septuagint (G) is by far the most valuable of the versions for the study of the book of Proverbs. It is central to the study of the book's early interpretation, its transmission history, and its textual development. The commentary of this volume will probe it carefully to determine where it provides a different text.

As my working edition I use, *faute de mieux*, Rahlfs's *Septuaginta*. I give attention only to substantive inner-Greek variants (taken from Rahlfs or Holmes-Parsons [HP]; the Brooke-McLean Cambridge Septuagint does not include Proverbs). By "substantive" I mean variants that are, in my judgment, relevant to text-critical decisions. I occasionally call attention to interesting variants in the minuscules by referring to "manuscripts," that is, manuscripts collated in HP. In the absence of a stemma, listing of particular minuscules would be of little value.

The lack of a critical edition of G-Proverbs is unfortunate, but the effects are mitigated by the fact that changes later introduced in G are mostly Hexaplaric in origin and adjust toward M.<sup>2</sup> Hence the inability to identify some of these changes may camouflage true variants but would not create false ones. Peter Gentry's Göttingen edition is some years off, but I am reassured by the assessment he has graciously allowed me to quote: "The problem of the relationship between LXX and MT Proverbs is notorious and vexing. Scholars should not expect miracles from a forthcoming critical edition of the LXX text in resolving this issue, for a critical edition will not yield a radically different text from that of Rahlfs's" (pers. comm., March 17, 2010).

## 3.1.2. The Date of G-Proverbs

The dating of G-Proverbs can be determined only loosely. A mid- to late second-century BCE dating is reasonable. Johann Cook (1993) supports this dating on the grounds of the translator's familiarity with Hellenistic literary style and his supposed suspicion of foreign thought. Michael Dick (1990, 21, 50) advocates an early second-century dating for G-Proverbs

<sup>1.</sup> The latter part of this chapter is based in part on Fox 2005.

<sup>2.</sup> The Hexaplaric variants are listed and discussed by Fritsch 1953.

because it does not accept the belief in an afterlife, which became prominent later, as well as its generally universalistic outlook. However, the "universalistic" outlook is native to wisdom literature. Also, while belief in the afterlife is unlikely to come to expression prior to the second century, it could appear later and is irrelevant to the *terminus ad quem*. External sources are not much help in this matter. The Greek of Ben Sira is not dependent on G-Proverbs, and while the Wisdom of Solomon shows dependency on G-Proverbs in a few verses, most clearly Wis 6:14 on 1:21, its own dating is in dispute.

The unity of G-Proverbs has not been established. There is no reason to assume multiple translators. While different translation styles are present, they are distributed throughout the book, not clumped together in a way that would point to different hands at work in different places. The additions composed in Hebrew, such as 9:12a-12b, were most likely already in the Hebrew source text. (Otherwise we would have to posit that they existed in different manuscripts and were translated and imported later.) Some of the additions composed in Greek may be the translator's work, but others were probably inserted later. For example, the epigram on the bee in 6:8a-c is written in an ornate literary Greek unlike the rest of the book, and its attitudes are distinctively Greek; see the commentary. It is unlikely that the translator would have paused in his work to compose an epigram in a different style. The poem in 9:18a-d also shows evidence of a Greek origin. The Hexaplaric and pre-Hexaplaric adjustments to M are, of course, later than the original.

### 3.1.3 The Order of G-Proverbs

The most striking difference between G-Proverbs and M-Proverbs is in the order of the last third of the book. After 24:22*e*, G proceeds as follows (using M's numbering): 30:1-14; 24:23-34; 30:15-31:9; 25:1-29:27; 31:10-31. There are also small-scale divergences, such as 8:32b appearing after 8:34; several displacements in 15:27a-16:9; and 31:25 and 26 being switched. There are numerous additions and omissions; see these entries in §7.3.1 below. There is no single explanation for all these variations. (The terminology is not meant to prejudice the decision as to which text had the original order.)

I see nothing that would deliberately motivate the changes in either direction. Cook believes that Prov 31:1–9 was displaced from its M order to join 29:27, which mentions the  $\alpha\delta\mu$  so, to 31:10, which speaks of the

#### 3. THE VERSIONS

άνδρεία, thereby creating a contrast (between bad man and good woman) of the sort that the translator favors (CSP 312–15). Also, according to Cook, the translator wanted 31:1–9 immediately before 25:1–8, since both speak of kingship. However, the literary gains are too trivial to justify such a major dislocation, especially since moving a few verses would have accomplished the same thing. In any case ἄδικος and ἀνδρεῖος are not elsewhere in antithesis. M-Proverbs' order is certainly correct at least in chapters 30–31, because 30:1–14 form with 30:15–33 a series of independent poems and epigrams that are radically different from the proverb collections but belong together as an appendix to the book. Prov 30:11–14 in particular belongs with the list-epigrams in 30:15–33.

In G-Proverbs, there are in fact only two large blocks of text differently located: 30:1–14 (15 verses) and 30:15–31:9 (28 verses). It is likely that the Septuagint's order resulted from accidental displacement of these blocks. This is unlikely to have happened by shuffling of leaves in a Greek codex (as happened in Sira 30:25–33:13a/33:13b–36:16), because the two blocks are of different sizes. Possibly a copyist in Greek or Hebrew skipped over major blocks (perhaps by rolling up his scroll then reopening too far along), after which he returned to incorporate the skipped material. In any case, the different large-scale ordering seen in G-Proverbs has no inherent connection to the translator's other changes and shows no signs of being the translator's doing.

### 3.1.4. The Character of G-Proverbs

For the most part, G-Proverbs aims at a faithful representation of the intentions of the Hebrew text and is best understood in terms of that goal. Whether one accepts Cook's dictum that "the Septuagint should principally be seen as an *exegetical writing*" (CSP 12, emphasis original). depends on one's own purposes. It can indeed be approached as exegesis, and G-Proverbs certainly has a place in the study of the history of interpretation of the book of Proverbs. In fact, some of my own comments are relevant to that study. But for text-critical purposes, G-Proverbs must be taken seriously as a translation that aims at representing the message of the Hebrew book and for the most part succeeds.

Of course, all translations, except perhaps the most mindless, are based on exegesis, though this tends to be noticeable only when one disagrees with the interpretation. The category "exegetical" is too broad to help us explain most differences from the Hebrew of M-Proverbs (or, more precisely, from a mimetic translation of that text, with a precise and rigid mapping of components of the source text into the target language).<sup>3</sup> "Exegetical" is more usefully reserved to describe added words, phrases, or lines that are intended to explain another element in the translation.

Cook's main thesis, developed in CSP and numerous articles, is that G-Proverbs is a Jewish Hellenistic document that is both interpretive and religiously conservative. By this he means that it states and promotes specifically Jewish doctrines, above all the centrality of the law of Moses, and does not introduce foreign ideas to any significant degree. I largely agree with this assessment (though I do not think that the law is *central* in this translation; see the comment on 13:15). I am also persuaded by Dick's arguments (1990) that G-Proverbs does not promote Hellenistic doctrines but at most vaguely reflects some common Hellenistic notions. At the same time, I would not deny many of d'Hamonville's observations in BAP on the impress of Greek literature and thought on the translation. We can also accept that the translation was (erratically) influenced by Greek stylistic choices (as argued by Gerleman 1956, 11-35; BAP 92-99). It is only to be expected that an Alexandrian Jew (as the translator presumably was) would imbibe tropes, prosodic patterns, and ideas borne by his environment as well as his ancestral traditions. I only claim that these qualities do not overwhelm the underlying Hebrew text or preclude the recovery of variants.

To call G-Proverbs "free" is imprecise and insufficient, because there are many kinds of freedom, many ways of diverging from the source text. "Free" does not explain what is going on. Moreover—and I think this is the case in G-Proverbs—a translator may diverge from a mimetic rendering toward what seems like freedom because he feels compelled to represent the meaning of M—or of a variant Hebrew source— correctly, as he understands it, not because he is playing loose with the sacred book.

<sup>3. &</sup>quot;Literal" is confusing because it also serves as an antonym of "metaphorical." But a metaphor, if translated "literally," remains metaphorical, whereas if it is translated "freely," it becomes a "literal" statement. The term "mimetic" comes from the translation theorist James S. Holmes (1988, 25–28), who describes translation in terms of mapping theory. A large-scale map (such as, say, a map of a country) represents the basic contours and main landmarks in the territory. A small-scale map (such as a map of a town) can show the details of the territory (such as streets) in their right places and proportions. Mimetic translation is of the latter sort. It attempts to map the maximal number of linguistic features of the source onto the receptor text and aims at consistency in correspondences between the vocabulary of the source and the target. See further Fox 2002.

#### 3. THE VERSIONS

I prefer to describe the translation technique of G-Proverbs as "flexible." By this I mean that the translator uses a variety of techniques.<sup>4</sup> Sometimes he maps his source closely, sometimes paraphrases, sometimes expands the quantitative representation of Hebrew words, sometimes reduces it, and sometimes just guesses at the meaning. This flexibility makes it almost impossible to characterize the translation as a whole in quantifiable terms. The fact that the translator chooses a broad mapping in one verse, grasping the meaning of a Hebrew proverb in a Greek sentence that does not align neatly with the components of the source, proves little about his approach elsewhere. At most it is suggestive of the kinds of things the translator *can* do. It is no more justifiable to extrapolate from the broadly mapped verses to the translator's overall technique and call his translation "free" than to deduce from the (numerous) narrowly mapped ones that he is "literal." We have to proceed verse by verse and decide each case on its own.

When we allow for the translator's flexibility, G-Proverbs proves to be, on the whole, an honest rendering of a Hebrew original *as the translator understood it*. To make this translation usable for text-critical purposes, I attempt to explain its most noticeable departures from the Hebrew. For this purpose, M is, for practical reasons, the starting point but is not presumed to be the translator's source text. We seek to understand the constraints under which the translator operates and to see through the translation to the source text behind it. Only by exegesis of the translation as a literary text is it possible to understand the knowledge, assumptions, motivations, and ideas that resulted in the translation, and thereby to recover some of the variants he worked with.

Four principles should be kept in mind as relevant to all the versions: (1) However interpretive or paraphrastic the translator may be in translating the Hebrew, he is not necessarily translating a text identical to M. (2) When a translation represents the same text as M, it does not actually

<sup>4.</sup> I use "technique" in the limited sense proposed by Aejmelaeus 1993, 65–76. It is what the translator *does*, what appears in the result, without implying a deliberate system or working procedure. Aejmelaeus writes, "Translation technique cannot be anything more than a collective name for all the different renderings used by a translator. Study of translation technique aims at describing the end-product of a translator's work. It cannot be a question of discovering the system used by the translator, because there was none" (69). (However, systems did emerge among the later "revisers.") If one ascribes intentions, attitudes, and inclinations to a translator, these are descriptions of tendencies evident in the end product. But this is also true when we speak about literary texts and their authors.

*support* it, as people commonly say. It only *agrees* with it. (3) A variant in the translator's source may be wrong. One cannot disprove the existence of a variant by showing it to be bad Hebrew or inappropriate to its context. (4) If we cannot determine what text underlay G, we must default to ignorance, not to M.

## 3.1.5. A Profile of G-Proverbs

3.1.5.0. Introduction. G-Proverbs has a set of characteristic features, both large-scale and small. None are unique to G-Proverbs, but taken together they give a sense of its particular character.

These features arise in three phases in the book's development: the Hebrew source text used by the translator, the OG translator's own activity, and the subsequent inner-Greek transmission. It is often difficult to determine which phase a particular instance belongs to. If scholars have tended to ascribe most M-G differences to the translator, this may be due to an inclination to assume that what one sees in a translation originated in translation. Cook says that a determination that the translator handled his text freely "could naturally lend support to a view that would ascribe deviations from the M to the translator" (CSP 31). ("Inevitably" might be a more fitting adverb.) But this reasoning is circular. Cook's assessment of the degree of the translator is working from M. In any case, translational freedom, if that's what it is, may obscure variants without thereby supporting M.

In the commentary I try to determine how each variant arose, but this is often impossible. Still, we should note that all the *types* of variation found in G (and the other translations), other than language-dependent ones, are well attested in single-language redaction and transmission. The appendix to Fox 2005 provides examples from Ben Sira and Ptahhotep.

The production of variants is one way that wisdom literature evolves; indeed, this is true of M-Proverbs itself, as shown by the numerous proverbial doublets, in which one must have emerged as a variant of the other or both emerged as derivatives of a common source. To be sure, variants of all sorts are well attested in other genres as well. Still, though I cannot quantify this, it seems to me that wisdom literature is especially malleable and even invites manipulation. This may take the form of additions, rephrasings, lexical variation, glosses, rearrangement, and more. After all, wisdom literature claims to transmit not the words of God but rather the teachings of sages who learned from earlier sages. The scribes who followed could view themselves as part of the creative wisdom tradition as well.

It is difficult to generalize about the translation technique of G-Proverbs because there are many different practices at work, often in the same verse. They range from strict mimetic translation to paraphrastic and expansive renderings, some of which have only slight contact with the Hebrew. G-Proverbs even includes new passages that are not translations at all. There is no "typical" verse, no characteristic translation which could show us what to expect elsewhere. To quantify the divergences from M and take an average would be like mixing the colors of a painting together and declaring that its typical hue is brown.

Still, whatever the techniques that gave rise to the translation, G does have its own literary profile, a set of features that together give it its distinctiveness. We could describe these features without respect to their origin, because G-Proverbs now exists as a work with its own integrity. (However, I exclude Hexaplaric and other recensional pluses, since these only obscure the translation's distinctiveness. For fuller discussion and additional examples, see the commentary.) The following categories are overlapping. Thus "resolving metaphor" may be a way of enhancing the tone, and so on. They are separated below for purposes of illustration. Significant differences from M are marked in italics.

3.1.5.1. Mimesis. The most frequent translation technique in G-Proverbs is mimetic or formal rendering. Here are three examples out of scores:

10:2  $l^{\circ}$  yw<sup>\</sup>ylw 'wsrwt rš wsdqh tsyl mmwt ούκ ώφελήσουσιν θησαυροί ανόμους, δικαιοσύνη δε ρύσεται έκ θανάτου. 10:9 ylk hwlk btm bth, wm'qš drkyw δς πορεύεται άπλῶς, πορεύεται πεποιθώς, ὁ δὲ διαστρέφων τὰς ὁδοὺς αὐτοῦ vwď γνωσθήσεται. 9:1 hsbh 'mwdyh šb'h hkmwt bnth byth

μκπωτ οπτη ουτη πέρη πωαγή εση ή σοφία ψκοδόμησεν έαυτῆ οἶκον, καὶ ὑπήρεισεν στύλους ἑπτά. The last verse, too, is an exact literal translation, but of a different Hebrew text, with הצבה for הצבה. Translation of a different source text has nothing to do with translational freedom. Of course, it is necessary to tease the variant out of the translation before evaluating the way the translation maps its source. *Circularity is avoided only by bringing a variety of factors to bear on the reconstruction of the source and not simply translating the Greek into Hebrew.* 

In most verses, G is close to M, apart from trivial differences in number, tense, and function of words and a certain looseness in representing parts of speech. When G goes without mention in the commentary, it is because the translation is functionally identical to M. In verses where G maps the Hebrew closely, deviation from M is more likely to be a textual variant.

3.1.5.2. Moralism.<sup>5</sup> M-Proverbs is, to say the least, already a heavily moralizing book, judging behavior as good or bad, rarely with ethical shadings. G-Proverbs only increases this moralism. Where a saying in M might be read as utilitarian, G usually makes sure that it is not. Here are a few examples of many:

- M 2:11 promises that "shrewdness (מזמה) will watch over you, good sense (תבונה) protect you." Both of these faculties are in some cases utilitarian. The former can even be misused (e.g., Prov 12:2; 14:7; 24:8). G defines the effective qualities as βουλή καλή "good counsel" and ἕνοια δσία "pious understanding" to make it clear that only moral and religious virtues provide protection.
- M 3:9 advises making offerings to God. G stipulates that one must give only what was earned *honestly* (though this was never in doubt).
- M 6:30 says, "People don't despise a thief if he steals to fill his belly when starving," thereby recognizing that crimes are not equally reprehensible. G, however, does not want to allow for the possibility that a thief could ever avoid capture and disgrace. He adds the notion of the thief's being caught and changes the public's lack of contempt into lack of surprise, translating, "It is not surprising if

<sup>5.</sup> See further "moral and/or religious emphasis, moralizing" in §7.3.1 below.

one who steals is caught, for he steals in order to fill his soul, when he hungers."

- M 9:17, at the end of Folly's invitation, reads, "Stolen water is sweet, and secret food a delight." It is not certain whether it is Folly or the teacher now speaking. Lest one think it is the teacher and imagine that he is affirming the goodness of illicit pleasures, G places the words unambiguously in Folly's mouth. This makes them a continuation of her deceitful invitation. The Hebrew is much subtler, allowing that the enticements of Folly, especially in female form, though deadly, really are (momentarily) sweet. The Greek does not want you to think this even for a moment.
- M 19:22b, "better a poor man than a deceitful one," might be thought to imply that every poor man is honest. Moreover, the stich does not provide a clear antithesis between the two parties being compared. G excludes the wrong inference and sharpens the antithesis by translating: "better a *righteous* poor man than a *rich* deceiver." There is a certain finesse in the way that the translator infers the notion of righteousness by reversal of its antonym "deceitful" and extracts the concept of wealth from "poor" by reversal of each component. He is probably construing the proverb correctly.
- M 21:22 praises the strategic power of wisdom by declaring that a wise man can take down a fortress. G explains that the fortress was one "on which the *impious* relied." After all, why else would the wise man attack it, and why would the city fall if it did not deserve it?
- M 28:20b warns that "he who hastens to get rich will not go unpunished." But since hastening for wealth is not in itself a crime, G converts the action into evil: "but the evil man will not go unpunished."

Not all moralizing additions in G originated with the translator. The following was present in his Hebrew source text: M 18:22a, "He who finds a wife finds something good," might be thought to mean that *any* woman is a blessing. G begins, "He who has found a *good* wife has found happi-

ness," to make it clear that only marriage with a good woman is a blessing. This verse was cited with טובה in some Hebrew sources as well (b. Ber. 8a; b. Yeb. 63b; Mid. Shoḥer Tov 151ab). This shows that we cannot automatically assign heightened moralism to the translator.

The following addition entered subsequent to OG: G 22:16 reads, "He who lays false charges against the wretched increases his own [sc. wealth]," which is an expression of indignation.  $G^{ScA}$  and some manuscripts enhance moral clarity by adding xaxá, hence "increases his own *troubles*."

Moreover, there are moralizing additions that never reached G. In 5:22a, M has, "His—the evildoer's (את הרשע)—iniquities will trap him." The phrase "the evildoer's" is not integrated in the Hebrew syntax, and it is lacking in G. The phrase is an epexegetical gloss in the Hebrew. We easily recognize moralizing additions in G, such as an added δίκαιος or κακός. This example shows that the same process went on in Hebrew transmission.

A valuable assessment of the place of moralism in G-Proverbs is the essay by Dick (1990, esp. 21–26).

3.1.5.3. Refinement. Sometimes the concern that shapes G's translation is not so much morality as tonality. The translator seeks to make certain actions and persons seem more refined, enhancing their propriety and dignity.

- M 3:34 says that God mocks mockers. G says that he "opposes" them, ascribing to God a less vivid but more dignified action
- M 6:3 urges the man who has given surety to "go and grovel" to his neighbor to get released from the obligation. G converts this into a more decorous "Go without being lax."
- M 9:3 says that Wisdom sends forth her *maidservants*. G uses the masculine δούλους ("manservants"), on the assumption that it would not be proper to send around maidservants to bring in male guests.
- M 22:3a observes that "the shrewd man sees trouble and hides" a behavior which, though prudent, is not particularly noble. By giving a different sense to יסר מולעה and deriving ויסתר from יסר, G pictures a different situation: "When a clever man sees a bad man being punished severely, he takes instruction."

## 3. THE VERSIONS

• M 28:25 says that the man who trusts in the Lord ידשן—literally "will be made fat." G's "will be attentive" has a more edifying quality.

3.1.5.4. Improving the "Logic." The translator sometimes makes a verse more logical, or, more precisely, brings out the logic he considers implicit in it. The logic in question is of course the translator's own, insofar as this can be deduced from the changes he has made.<sup>6</sup> This category overlaps with resolving metaphor (§3.1.5.5) and disambiguation (§3.1.5.6).

- M 4:25 says that you should let your eyelids "go straight." Since eyelids do not actually go straight, G says that they should "assent (νευέτω) to righteousness."
- M 5:9 warns that an adulterer could leave his "glory" (הודו) in a foreigner's house. It is not evident how one can leave one's "glory" somewhere. G helps out by construing הוד as "life." The fool will leave his life there, literally, by getting himself killed.
- M 8:27b has God inscribing a "circuit on the abyss." G-Proverbs has God locating his throne *upon the winds*. This is based on the idea that the circuit of the heavens is on high (Isa 40:22), which is the location of God's throne (Isa 66:1).
- M 9:13a calls Lady Folly the "woman of folly (פתיות)." The translator mistakenly equates של with פתיות ("morsel") and then must add "lacking" to make sense of the sentence.
- M 10:18a reads: "Deceitful lips cover up hatred." This means that dishonest people speak flattery while concealing the hatred they feel. But G is puzzled that covering up hatred should be wrong and goes so far as to reverse the meaning of its source by making "righteous lips" do the covering up. This is close to a "correction" of the source text, making it mean what it somehow *must* mean.
- M 12:23a says that the shrewd man "covers up (כסה) knowledge." A failure to understand why one would "cover up" knowledge is

46

<sup>6.</sup> See "logic (apparent logic), adjustment for" in §7.3.1.

probably what prompted the translator to render כסה (a word he knew) as if it were כָּסָא "throne"; hence: "The intelligent man is a throne of knowledge."

- M 15:12b says that the fool will not go *to* the wise. This is not illogical, but G still adjusts the wording to his notion of what is meant by "going": going *with* someone as companion (influenced by 13:20) rather than going *to* someone for instruction.
- M 16:12a, "The doing of evil is the abomination of kings," might be thought to mean that kings inevitably shun evildoing. G's "An abomination to the king is he who does evil" prevents this construal by ascribing the hated actions to another person. This construal of the Hebrew is grammatically feasible but probably wrong.
- M 26:7 and 9 speak of proverbs in the mouths of fools. By translating as "transgression" and "bondage," G avoids ascribing proverbs of any sort to fools. Proverbs, after all, are the domain of the wise.
- M 30:3 declares, "and I have not learned wisdom"—a strange admission coming from Solomon. By reading אל as if it were ואל and ignoring the grammatical form of למדתי, G has him say the expected: "God taught me knowledge" (cf. 1 Kgs 3:12, etc.).
- In 31:10, G avoids the metaphor of the excellent woman's "price" with its possibly indelicate implications.
- It is in accordance with the translator's "logic" (that is to say, inherited assumptions) to avoid ascription of authorship to anyone besides Solomon; see the comment on 10:1.

3.1.5.5. Resolving or Changing Metaphors.<sup>7</sup> One can resolve figurative language into its literal sense or change the figures themselves in order to improve the logic—as in the preceding category—and to add clarity.

<sup>7.</sup> See "metaphor, resolving and explicating" in the septuagintal index.

- M 12:6a says that "the words of the wicked are an ambush for blood." Lest the reader wonder how words can be an ambush, G substitutes the more expected "deceits."
- M 14:11a promises that the tents of the righteous "will *flourish*." Tents may stand or fall, but they don't really "flourish," so G provides the more obvious "will stand."
- M 17:28 praises the man who "shuts his lips." G explicates this (rather unnecessarily) as "whoever makes himself silent."
- M 23:2a uses a vivid metaphor for self-control at a meal: "put a knife to your throat." G substitutes an expansive, nonmetaphorical sentence affirming the propriety of eating what is given you: "And lay your hand [sc. on what is placed before you], knowing that it is right for you to prepare such things."
- M 27:24 reads, "For wealth [חסז] is not forever, nor does a crown endure generation after generation." The metaphor of a crown has puzzled commentators. G explains it: "For *power and strength* do not belong to a man forever, nor does he transmit (them) from generation to generation." The translator assumed, reasonably, that "crown" was equivalent to the parallel חסן, which can mean "power."

In this category we can include G's tendency to avoid literal translation of body parts and other concrete symbols, either by leaving them untranslated or by replacing them with more abstract terms (SDEK, 1957). For example, in G 8:2–3 "head," "hand," and "mouth" are left untranslated, and in G 1:24 "stretch out hand" becomes "stretch out words."

Occasionally G introduces a metaphor of its own. In 16:28 G adds the vivid image of a "torch of deceits," which the troublemaker kindles. In 23:31, G warns that a drunkard will become "naked as a pestle." But such metaphorical rewritings are rare.

3.1.5.6. Disambiguation.<sup>8</sup> In a sense, all the practices listed here are forms of disambiguation. But a particular form of paraphrase worth noting is G's

<sup>8.</sup> See "disambiguation, paraphrase" in the septuagintal index.

removing the ambiguity from a verse that may be read in more than one way, rather than leaving it to the reader to tease out the intended sense.

- M 5:4a says of the seductress that "her end is as bitter as gall." This might be thought to mean that she herself will suffer a hard death, although it is the foolish youth's death that is relevant to the context. Hence G plays on אחרית to produce the unambiguous "Later you will find her more bitter than gall."
- G 25:13 expansively explicates the terms of the analogy between "the chill of snow in harvesttime" and a reliable messenger by explaining that the snow "is beneficial against heat."<sup>9</sup> This verse might belong in "overexplanation" (§3.1.5.8) as well.
- G 28:14 adds δι' εὐλάβειαν for clearer definition of what kind of fear is being praised, namely, respectfulness.
- M 31:1, beginning with דברי למואל, ascribes the following instruction to an otherwise-unknown wise man. To make it clear that Solomon is the speaker in Prov 31, as G believes is the case for the entirety of Proverbs (see the comment on 10:3), G reads the phrase as דָבְרַי לְמוֹ אֵל "my words are to God."

3.1.5.7. Elaboration.<sup>10</sup> G has (apart from Hexaplaric incursions) numerous additions of a stich or more. Some of these are doublets or internal expansions of a verse. Some have enough independence to be considered a verse on their own, and Rahlfs gives them added letters (which here are italicized). Some additions were introduced in the Hebrew transmission that led to OG's source text; others were produced by the translator; and others are the works of later Greek scribes. I attempt to identify the source of the additions in the commentary (and see "additions" in the septuagintal index), but this is often impossible. Still, they are all similar in character and share the same ideology, except in regard to a few nuances.<sup>11</sup> In

<sup>9.</sup> Examples of paraphrase are 5:4; 6:25; 7:6–17; 8:27; 10:4; 10:5; 10:24; 12:25; 16:7; 17:16*a*; 23:27; 24:15; 25:13; 28:17; 30:1; 30:8; 31:5.

<sup>10.</sup> See "additions" in §7.3.1.

<sup>11.</sup> G 6:8*a*-*c* develops the Greek topos of the bee. G 9:10*a* introduces the theme of (Mosaic) law. G 9:12*a*-*c* and 9:18*a*-*d* warn against foreign ideas and culture, a concern

describing G-Proverbs it is legitimate to bring them all into consideration. All are anchored in their context and almost all develop the idea stated in the preceding M verse.

The first four examples were composed, as far as I can tell, in Greek.

- G 4:27*a* elaborates on the admonition in 4:27. Whereas 4:27 exhorts the reader to make his way straight, inclining to neither the right nor the left, G 4:27*a*, following a Greek, not a Hebrew, notion, distinguishes the right from the left as the good from the bad.
- G 9:10*a* is a doublet of 9:10c, but it also extends the principle stated in 9:10ab—that the beginning of wisdom is the fear of the Lord—by saying that to know the law is good intellect.
- G's proverb on the educated son (10:4*a*) seems to elaborate the verse that comes *after* it, which contrasts two kinds of son.
- G and M of 13:9 say that the light of the wicked will be extinguished. This image can refer to misery, or to death, or to ignorance. G adds 9*a* to show that it means (or *also* means) the darkness of sin.

Some additions were probably inserted later than OG:

- G 6:11*a* is likely a later addition based on a misunderstanding of the preceding verse. It distinguishes two kinds of runners. This recalls G's handling of 4:27*a*.
- The epigram on the bee in 6:8*a*-*c* was added in Greek as a further example of industrious creatures.
- The epigram in 9:18*a*-*d* that warns against close contact with the foreign culture was added to explain that the enticements of Lady Folly in the preceding verses are cultural as well as sexual.

not found in M-Proverbs. G 17:6*a* holds out promise of spiritual wealth, an image not found in M. None of this is in any way *contrary* to M.

The following set of elaborations took place partly in Hebrew, partly in Greek:

M 18:22a reads, "(a) He who finds a wife finds something good, (b) and receives favor from the Lord." This praises marriage as an inherent good. In the Greek, the observation develops in three stages: (1) G 18:22a: "He who finds a good wife has found something good." The first "good" was present in the Hebrew (see the commentary). (2) G 18:22aa: "He who expels a good wife expels happiness [lit. "the good"; pl]." This is based on a non-Masoretic Hebrew text of G 18:22a (with "good") but understood with a different vocalization. (3) G 18:22*a*b: "but he who retains an adulteress is foolish and wicked." This supplements G 18:22aa. Its absence in S shows that it was added after OG. We see three phases of intervention, each one refining the preceding and drawing out its implications. Phase 1 restricts the scope of the original Hebrew, teaching that not every wife, but only a good one, is a blessing. Phase 2 explains why a man should hold such a wife fast. But, lest this be thought to preclude divorce, Phase 3 makes it clear that this action is sometimes necessary.

Like all of G-Proverbs' tendencies, elaboration, too, is not invariable. Occasionally G will do the opposite, omitting words as being redundant. For example, in 25:21 "bread" and "water" are clearly implied by "feed" and "give drink" and are omitted. Similarly 25:28.

3.1.5.8. Overexplanation.<sup>12</sup> All the above features go beyond mimetic translation in clarifying the text, but sometimes the clarifications are not really necessary. One gets the impression that the translator lacks confidence in readers' ability to figure things out for themselves. Some of the above examples illustrate this tendency. Further examples:

• G 8:21*a*: "If I tell you things that happen daily, I shall (also) remember to recount things of old." It is obvious that 8:1–20 describes the present, 8:21–36 the past. Still, the translator inserts G 8:21*a* to distinguish the two phases, taking the reader by the hand through the progress of the narration.

<sup>12.</sup> See "overexplanation" in §7.3.1.

- M 16:11b says that "all the weights of the purse are [God's] work. G explains that "*honest* weight-stones are his works." This precludes the (unlikely) misunderstanding of the verse as meaning that God is responsible for dishonest weights.
- G 22:13 prefixes "The sluggard makes excuses and says" to explain the sluggard's intent, though this is obvious from the rest of the verse.
- In G 22:14*a*, after a warning on the danger of evil speech, a third stich adds a banal conclusion, "But it is necessary to turn away from a crooked and evil way."
- In 25:22b, to "and the Lord will repay you," G-Proverbs adds "with good," though the repayment for the recommended behavior could hardly be otherwise.
- M 26:27 speaks of one "who digs a pit," to which G adds "for his neighbor," lest one think that the warning concerns digging legitimate holes.<sup>13</sup>

3.1.5.9. Enhancing Parallelism.<sup>14</sup> G-Proverbs often enhances the correspondence between the semantic or morphological components of a couplet. According to Gerleman (1956, 23–26), there is a strong tendency to make couplets antithetical, which almost always produces or strengthens parallelism (17–23). In a book about parallelism in G-Proverbs, Tauberschmidt (2004) identifies "secondary parallelism," meaning parallelism created or enhanced by the translator. Though I find Tauberschmidt's study defective in many ways and have criticized it in a review (Fox 2004), it is fair to say that he has described one of this translator's practices, and I will note occurrences in the commentary. However, as the following examples show, the enhancement of parallelism is often linked to other motives and causes as well. Moreover, secondary parallelism often arises in Hebrew too.

52

<sup>13.</sup> Examples of overexplanation are 8:21*a*; 14:20; 16:11; 19:22; 22:13; 22:14*a*; 25:22; 26:27; 27:25.

<sup>14.</sup> See "parallelism" and "antithesis" in the septuagintal index.

- M 1:23bc: "Behold I pour out to you my spirit, // I make my words known to you." G 1:23bc: "Behold, I will bring forth to you the utterance (ῥῆσιν) of my spirit, // I will teach you my word (λόγον)." The translator adds ῥῆσιν as a parallel to λόγον (Tauberschmidt 2004, 33–34). At the same time, the addition provides an explanation of the idiom "pour forth spirit."
- M 1:24: "Because I called and you refused me, I stretched out my hand and no one paid heed." G: "Because I called and you did not listen, I spoke at length and you gave no heed." Where M has a gesture parallel to speaking, G has speaking in both sides of the parallel.
- M 17:4: "An evil man hearkens to the lip of iniquity [i.e., an iniquitous person]. // Deceit [שקר, i.e., a deceitful person] listens to the tongue of destruction." G 17:4b: "but a righteous man does not hearken to deceitful lips" (δίχαιος δὲ οὐ προσέχει χείλεσιν ψευδέσιν). According to Tauberschmidt (2004, 44), G creates antithesis by "changing" שקר into δίχαιος. But the added negative must be noted as well. It is more accurate to say that G preserves parallelism by converse translation. As often, G modifies the parallelism of a verse that already had its own.

Secondary parallelism may also arise in the Hebrew, as in the following case:

M 14:33: "Wisdom rests in the heart of the sensible man, but in the midst of dolts it makes itself known." G: "In the good heart of a man there is wisdom, but in the heart of fools it is *not* perceived (οὐ διαγινώσχεται)." It may be true that "the second colon correspond[s] to the first colon in a contrastive manner" (Tauberschmidt 2004, 204), but, contrary to Tauberschmidt, this is not the translator's doing. The negative is witnessed by Theodotion and Aquila and must have been in G's source text. The negative was inserted in the Hebrew transmission for the sake of logic, because the notion of wisdom being known among fools seemed outlandish.

 M 27:9: שמן וקטרת ישמח לב ומתק רעהו מעצת נפש Literally, "Oil and incense make the heart rejoice, and the sweetness of his friend more than the counsel of the soul." The meaning is uncertain. G 27:9: "By myrrh and wines and incense the heart is pleased, // but by misfortunes the soul is rent." This is based on a different Hebrew text, which ends: וְמַתְאָרֶת נְשָׁשֶׁבֶת נְשָׁשׁ וּמִתְקָרְשָה מֵשַאֶּבֶת נְשָׁשׁ.

Tauberschmidt knows that quite often G-Proverbs *violates* the parallelism found in M, and he even dedicates a chapter (2004, 109–64) to explaining why this happens. Though he lists almost as many verses that disturb parallelism as ones that enhance it, Tauberschmidt feels he must find a motive for each violation, as if the "fondness" for parallelism were a foundational demand requiring any departure from it to have a special reason. (Tauberschmidt does not, in practice, include textual differences among the reasons.)

In fact, the most we can conclude is that sometimes the translator tightens the parallelism, and sometimes he does not. This is not a useless conclusion, for it alerts the text critic to the possibility that in any particular case tighter parallelism may be the translator's doing. This conclusion does not, however, provide a strong heuristic principle that can claim priority over competing explanations.<sup>15</sup>

3.1.6. Translation as Control. This chapter began by questioning the usefulness of the concept "free" to characterize G-Proverbs. I suggested that a better characterization of the translator's approach is *flexible*—meaning that he applies a variety of means to achieve his goals. On the basis of the above profile, I further suggest that his main goal, evinced mostly in ways he departs from a mimetic translation, is *control*: control over the way the reader understands each segment of the book. The translator is trying to guide the reader to the exact intent of the text—*his* text, as *he* understands it. To this end he eliminates ambiguities, stabilizes indeterminacies, and draws stark moral distinctions wherever these tactics are needed to prevent misunderstandings, and often when they are not. He is not merely translating "freely," making the text mean whatever he wishes it to mean. He is letting the text convey what he thinks its author wants it to mean.

<sup>15.</sup> Examples of enhanced parallelism are 1:23bc; 2:15; 7:8; 7:17; 8:4; 8:14; 8:16; 8:20; 14:33; 15:13; 15:20; 15:22; 15:26; 17:4; 21:14; 24:11; 27:6; 27:9; 28:15; 30:23.

The translator is a participant in the creative process that transforms old wisdom into new. These observations also apply, *mutatis mutandis*, to the scribes who shaped and transmitted the Hebrew text that reached the translator and also to the Greek scribes who made contributions of their own to the dynamic book.

3.1.7. Indicators of Hebrew Variants in G-Proverbs. Given the flexible character of the Greek translation, it is admittedly difficult to identify variants in it and apply them to text criticism. There are, however, a number of criteria that help in the task. These cannot catch all variants, and my decisions in the commentary will sometimes suffice with slimmer arguments. But the following indicators are a good starting point.

3.1.7.1. A Component That Does Not Serve the Translator's Purposes. When a divergence from M does not serve a translator's ideological or literary agenda but can be explained in terms of scribal error, the difference is likely due to a Hebrew variant. G-Proverbs' flexibility would have allowed him to produce a rendering that promoted his ideology. For a fuller discussion of the following examples (and many others), see the commentary.

- 13:11. M: "(a) Wealth will grow less than a vapor (מהבל), but he who gathers carefully (על יד) gains increase." G: (a) טהמאלגי להמסטלא שלא שלא מיסטלא שרא מיסטלא מיסטלא שרא מיסטלא שרא מיסטלא מיד מיסטלא מיד מיסטלא מיד מיסטלא שרא מיסטלא מיד מיסטלא מיד מיסטלא מיד מיסטלא מיד מיסטלא שרא מיסטלא מיד מיסטלא מיד מיסטלא מיד מיסטלא מיר מיסטלא מיד מיסטלא מיסטלא מיסטלא מיד מיסטלא מיד מיסטלא מיד מיסטלא מיד מיסטלא מיסטלא מיד מיסטלא מיסטלא מיד מיסטלא מיסטלא מיד מיסטלא מיד מיסטלא מיד מיסטלא מיד מיסטלא מיסטלא מיסטלא מיסטלא מיסטלא מיסטלא מיסטלא מיסטלא מיסטלא מי מול מיסטלא מיסטלא מיסטלא מיסטלא מיסטלא מיטלא מיטלא מיטלא מיטטלא מיטלא מיסטלא מיד מיסטלא מיסטלא מיסטלא מיסטלא מיסטלא מיסטלא מיסטלא מיטא מיסטלא מיטלא מיטלא מ
- 13:19b. M: "What dolts loathe is turning away from evil (αre)." G: "but the works of the impious [pl] are far from knowledge (μακρὰν ἀπὸ γνώσεως)." M's "from evil" would only have enhanced G's moralism, which permeates the entire translation and is underscored

#### 3. THE VERSIONS

in this verse by the addition of "the pious" and "the impious." G's "from knowledge" comes from מדע, understood as מִדַּעַ. Other adjustments were made to accommodate this variant.

3.1.7.2. An Awkwardness in G. The translator uses a good, fluent Greek style, and with rare exceptions the proverbs make good sense in Greek. Awkwardness of style or obscurity of thought may be a sign that the translator is trying to force sense out of a corrupt text. The retroversion of this text must be supported by a scribal explanation for the change.

- 10:21a. M: "The lips of the righteous man shepherd ('rτψ) many."
   G: "The lips of the righteous know (ἐπίσταται) lofty things." G's metaphor of lips "knowing" something is a strange one. It arose because the translator had to make sense of a graphic variant, namely, 'rψ,' which he construed as 'rψ.
- 21:20. M: "A delightful treasure and oil (ושמו) are in the habitation of the wise, but a foolish man swallows it up." In 21:20a, M sets oil apart as something distinct from and parallel to treasure, then in 20b refers back to the two nouns by a singular suffix (on יבעלנו). G: "A desirable treasure will rest (מעמדמטׁסבדמו) on the mouth of the wise man, but foolish men will drink it down." G 20a is smooth, but G 20b asserts that fools will ingest and absorb the wise man's words, which is precisely what fools would not do. The difficulties are clues to corruptions in both versions. M has ושמון (מוכר בנוה הכם ובמיה) אוצר, where G had, correctly, ובנוה הכם ובמיל אדם יבלענו אוצר, אוצר אדם יבלענו

3.1.7.3. Ambiguity in the Hebrew. Sometimes the relation between M and G is best explained by triangulating an ambiguous variant behind both.

10:3. M: "The Lord will not let the soul [= appetite] of a righteous man starve, but he rebuffs the *disaster* (?) [Infin] of evildoers." G: "The Lord will not let (the) righteous soul starve, but he will overthrow the *life* [ζωήν δὲ] of the impious [pl]." It does not make good sense to say that God will rebuff the evildoer's "disaster," as M has it, but neither is G's "life" a suitable object of "rebuff." ζωήν δέ =

מחיה. וחית can mean "life," as G thought, but also "appetite," as in Job 33:20 and 38:39. The original וחית was easily corrupted into M's וחות. Note that G does not have to be *right* in order to witness to a variant.

3.1.7.4. External Support. External support for a variant in the Septuagint is hard to come by. In Proverbs, S depends heavily on G, so agreement may just show dependency. But sometimes S offers independent attestation by agreeing with the posited Hebrew of a G variant while treating it *differently* from the way G does.

- In 11:6, where M has ובהות, G's τῆ δὲ ἀπωλεία αὐτῶν represents
   This is supported by S's באלמס and V's *in insidiis suis*, which have the suffix as G does but render the word quite differently.
- In 11:26a, S's translation, "He who withholds produce in distress shall abandon it to his enemies," is independent of G's "May he who withholds grain leave it to the gentiles." This suggests that אישבקהו represents ישבקהו independently of G's ὑπολίποιτο, where M has יקבהו.

The Greek of Ben Sira occasionally supports G (of which it not derivative). In 6:5a, where M has ממיד, G's פֿא  $\beta \rho \delta \chi \omega \nu = \pi \alpha \gamma i \delta \sigma$ . Sir 27:20, which is based on Prov 6:5a, has פֿא  $\pi \alpha \gamma i \delta \sigma \varsigma$ . This sentence shows that the translator of Ben Sira had a word for "trap" in his Hebrew source.

3.1.7.5. A Combination of Indicators. Most effective in establishing a Hebrew variant is an argument based on mutually reinforcing clues.

14:32b. M: "while the righteous man trusts in his death (במותו)." A close translation would be ideologically unproblematic in G, but G reads, "but he who trusts in his own piety (דָן במטדסט אָסולדאָדו)," reflecting בתומו S supports this variant by rephrasing it independently of G. Also in support of בתומו is the displaced doublet of this verse in S 14:35c, "who trusts that he has no sins."

A combination of indicators can sometimes enable restoration of an entire verse or more, at least approximately. It is of special interest (though often impossible) to know which of the major G additions had a Hebrew souce-text. I propose such a restoration in the commentary on 9:12a-b.

3.1.8. Factors That Are Neutral with Respect to the Underlying Text

3.1.8.1. Elegance. Though awkwardness in translation can be a clue to a variant Hebrew reading (§3.1.7.2), the opposite is not true: literary elegance does not in itself indicate original composition. Elegance in the translation, displayed in literary qualities such as described by Gerleman (1956, 11–35) and d'Hamonville (BAP 92–99), does not disprove the existence of a Hebrew variant or make the source text unrecoverable. Although literary graces would be unlikely to arise from a rigidly mimetic rendering, they do not make the translation opaque to variants in the base text.

- 11:10b-11a. G is missing these lines. As it stands, G 11:10a+11b (11:10-11 in Rahlfs) form a nice couplet, cast in a tidy ABA'B' pattern and the antithetical form that Greek favors even more than M. But the couplet came about by homoioarkton from קריה to קריה, resulting in the loss of 11:10b-11a in G's source text. The translator dealt with a faulty text creatively.
- 24:7–10. There are variants, some quite faulty, in G's source text of these verses. The translator shapes them into a coherent epigram; see the commentary at 24:10.
- 29:22. M: "An angry man *provokes* (ארה) conflict, and a wrathful one is full of transgressions." G: "A hot-tempered man digs up (ὀρύσσει) conflict, and an angry man digs out (ἐξώρυξεν) sins." The translator created a paronomasia by using ἐξώρυξεν in the second stich. But this is still a clue to a variant, because it presupposes רגרה (as in Prov 16:27; 26:27, etc.) instead of M's יגרה.

3.1.8.2. Interpretation. Interpretive motives in the rendering of a verse do not disqualify it as evidence for textual variants. Even when the interpretive elements may seem to violate the intention of the original, the scribe or translator who introduced them very likely thought that he was clarifying the author's intent.

It is important to remember that for purposes of assessing the translator's techniques and ideas, it is not what the Hebrew text means but what the translator *thinks* it means that counts. For that matter, this is true of the text critic too: what the text means cannot be distinguished from what he or she thinks it means. For practical purposes, there is no escape from gauging the translator's interpretation by one's own—often uncertain, sometimes shifting—understanding.

- 1:10-11. Cook calls this "the most convincing example of religious motivation guiding the translator. ... The translator contrasts the religious and the unrighteous even more explicitly than the Hebrew (MT) does" (CSP 408). In fact, however, G offers a straightforward translation, even in Cook's English rendering: "(10) "My son, let not impious men lead you astray, and do not consent, if they invite you saying: (11) 'Come with us; partake in bloodshed, and let us hide a just man unjustly in the earth'" (NETS). The G-M correspondences are found elsewhere, and in any case the Hebrew words are not less religious than the Greek. The translation is representing the religious coloration of M. It is interpreting the verse, but *correctly*.
- 11:7. M: "When the wicked man dies, hope is lost, and the expectation of strength (ותוחלת אונים) perishes." G: "When a just man dies, hope is not lost, but the pride (καύχημα) of the wicked perishes." G's rephrasing, which goes so far as to add a negative, makes the verse allude to the afterlife. Still, καύχημα represents a variant, תהלת.

3.1.8.3. Alternate Proverbs. It is in the nature of proverbs to constantly evolve, both in oral and in written transmission (two media that are, as David Carr has shown [2005, 3–14 and passim], thoroughly intertwined and mutually supportive). The variants that arise include "memory variants" (see §4.3.4.2) but are also, or mostly, *literary* variants, as people adapted old sayings to new uses or simply put the impress of their own style on them.

Often both the M and G forms differ considerably but make equally good sense. We can accept both as variant proverbs, without determination of priority. After all, we find many alternative proverbs *within* M-Proverbs.<sup>16</sup> Readers consider them equally valid and do not assimilate

<sup>16.</sup> These are catalogued and interconnected by Snell 1993.

one to the other. In such cases of alternate proverbs within the versions, I judge both forms "equal." I use M as the default for the HBCE text without thereby implying that it is preferable. In some cases both G and M may have evolved from a common ancestor and have equal priority. But even a determination of the priority of one would not invalidate the other. This reasoning can be used in defense of M's influential במותו in 14:32b. It was not the original form, but it became a significant proverb.

- 14:27a. M: "The fear (יראת) of the Lord is a fount of life." G: "The command (πρόσταγμα) of the Lord is a fount of life." G's πρόσταγμα was probably influenced by 13:14, which says something similar about בחרת חכם. It is not certain just what Hebrew word underlies πρόσταγμα, whether תורת ח תורת הכם, but in either case both M and G have validity as alternate proverbs. G's form arose by synonym substitution and stands in relation to M in the same way as M 14:27 does to M 13:14, or M 2:16 to 7:5. Or—M's form arose by synonym substitution, etc. Neither variant should be adjusted to the form of its doublet.
- 17:24. M: "Releasing water—the start of a quarrel, so before a quarrel breaks out, leave off!" G's reconstructed source text (see the commentary) means, "Releasing words starts a quarrel, so before a quarrel breaks out, leave off."

The phenomenon of alternative proverbs exists not only between M and G, or in different verses in Proverbs, but occasionally even within a single verse, with the *ketiv* and the *qere* carrying different forms of a proverb, as in 13:20a: *ketiv*: הלוך את חכמים ("Go with the wise and become wise"); *qere*: הולך את חכמים ("He who goes with the wise will become wise.")<sup>17</sup>

Scholars (paremiologists) have studied proverb variation extensively, and one clear conclusion is that proverbs are constantly mutating, in both

<sup>17.</sup> For further probing of valid variants, see Martin 2010, who speaks of "multiple originals." In a discussion of Hab 2:4, Martin suggests that variants can arise not only from scribal error but also from "an alternate possibility—that a scribe took no pains to carefully distinguish his 1 from his ' in some cases because the text made sense, in some way, either way. The reading is left intentionally open to allow for both options" (2010, 252).

oral and written transmission, whether because of slippage of memory, or because one form felt more natural to a new speaker, or because someone wished to change the message or its impact.<sup>18</sup>

# 3.2. The Peshitta $(S)^{19}$

### 3.2.1. The Background of the Syriac Translation

A late third- or early fourth-century CE dating for S-Proverbs is most likely. Citations in Aphraat show that Proverbs had been translated by the fourth century. Citations in homilies 337 and 344 show traces of G's influence (influence (influence (influence context)) and  $\beta \circ \eta \theta \circ \eta \omega \omega \omega \omega$  in 25:21). Ephrem, writing about 373 CE, seems to have known S-Proverbs.<sup>20</sup> S's deep familiarity with G suggests a western locale, probably Edessa, a religious center where the influence of Greek Christianity was especially strong.

The debate about the religious origins of S is complex and unresolved. For one thing, the various books of the Bible could have been translated in different social and religious settings. However, S-Proverbs' use of G gives a clue to *its* origins. The translator has a fair knowledge of Hebrew. The only way to attain this was in a Jewish school, where the Bible and other Hebrew texts were studied by endless recitation and repetition. This means that the translator must have been Jewish, at least at first. But he also knew Greek (as some Jews of Edessa did); and not only that, he used G in a way that suggests he regarded it as canonical and authoritative. S not only looks to it for help in hard passages but sometimes translates it alongside of and even instead of M. The Septuagint was not accepted by Jews outside Hellenistic Egypt. Christians in the east, in contrast, held it in high regard and studied it. The translator was very likely a Christianized

<sup>18.</sup> See ABP 2.491–93 for some references and examples. A case that interested me was the evolution of John F. Kennedy's dictum, "Ask not what your country can do for you; ask what you can do for your country." It is possible to trace the saying, or its basic structure, back to 1925. Sometimes it was transmitted orally, sometimes in writing, with changes of all sorts happening in both media.

<sup>19.</sup> This survey is based on Fox 2013, which has further examples and discussion, especially with regard to the religious-social origins of the Peshitta.

<sup>20.</sup> On the dating, see Pinkuss 1894, 107-8.

Jew, either an individual convert or a member of a Jewish group that had accepted Christian doctrines.<sup>21</sup>

# 3.2.2. S-Proverbs' Dependence on G-Proverbs

The most striking feature of S-Proverbs is its deep and frequent dependence on the Septuagint. Pinkuss's 1894 work on S-Proverbs includes a section on S-G agreements. Pinkuss lists many specific agreements and concludes that they are too numerous and too specific to have all arisen independently. While Pinkuss does not categorically exclude the possibility of later adjustments to G, he sees no evidence of this happening. He notes that some S-G agreements are attested as early as the fourth century CE, too early for a major revision toward G. I would add that a great many S-G agreements—such as the ones I use as examples below—are so thoroughly interwoven into the Syriac sentences that they cannot be explained as later interpolations unless we suppose that the interpolator simply exchanged one word with its synonym. J. Joosten (1995, 63-72), examining the doublets in S-Proverbs (see §3.2.2.6), also affirms S-Proverbs' dependence on G. As for the character of S-Proverbs overall, he concludes, "One sometimes gets the impression that the author of Peshitta Proverbs accorded as much importance to the LXX version as to the Hebrew" (65). M. Weitzman too supports S-Proverbs' dependence on G (1999, 70-86).

S's direct dependence on G is not hard to show. Evidence is indisputable in cases where S renders G's major additions, which do not exist in M (see \$3.2.2.7). Direct dependence is demonstrable also when S agrees with G in unexpected and even strange ways, which are required by neither context nor language. In none of these examples (nor in many others) is the S-G affiliation of the sort that can derive from a shared non-M variant. There probably are such variants elsewhere, but they are hard to identify. One must show that G and S are interpreting the same Hebrew consonantal form differently in a way that can be explained by an ambiguity in that form.

7:10b: S's translation of the well-known verb כפוּג by גער is unique and must derive from G's έξίπτασθαι, because the notion of "fly

<sup>21.</sup> The theory of a Judeo-Christian origin has several supporters. Among the recent arguments for a Judeo-Christian origin, see van Peursen 2004, 243–62.

away" or "flee" makes sense only in the context of G's interpretation.

- 10:14: S translates הבסי אויל hardly an obscure phrase—as
   לשוית ("a rash mouth"), which can only come from G's narrow rendering, στόμα δε προπετοῦς.
- 15:15: Where M has "a perpetual feast," S has a more cautious "is always at rest," which is taken from G.
- 16:10: S's ("goes astray") for M's αυξί must be based on G πλανηθη. S uses for πεπλάνηται in Prov 9:12b, where no Hebrew is available. When S is based on the Hebrew, bis not translated Since the notion of straying is not required by the context, S must be taking it from G.
- 17:12: M ("Better to come upon a bear bereft of her young than a dolt in his folly!") is clear and its vocabulary known to S. Still, S follows G's idiosyncratic "Cares will befall an intelligent man, but fools are preoccupied with evils."
- 18:3a: S's ארשא גריא ("to the depth of evils") must derive from εἰς βάθος κακῶν, a phrase with no warrant in M.
- 24:34b: S's rendering of איש מגן as "runner" must be based on G's δρομεύς, because this understanding of מגן is unparalleled in S, even in the doublet in 6:11.

It is theoretically possible that in all the cases of S = G (other than the G additions), both versions were drawing on otherwise-unknown interpretative traditions. But evidence for these traditions is lacking, and to invoke this possibility is to argue from silence. Neither T-Proverbs, it should be emphasized, nor the Jewish commentators who drew on it, can serve as evidence of an independent Jewish interpretative tradition. This targum (§3.4) is essentially a modified transcription of S-Proverbs and as such served as a medieval conduit for G's interpretations into Jewish exegesis.

My working assumption is that when the three versions agree, S is dependent on M. To be sure, it is possible in this case that G is his source,

3. THE VERSIONS

but my working assumption avoids inflating G's impact. Similarly, when S agrees with G in miscellaneous components of a verse but the basic idea is closer to M, I treat M as the source of the other components so as not to exaggerate the dependency on G. When, however, G is the controlling force in a verse and determines its gist, its imagery, or its poetic structure, I treat G as the source of the minor shared components as well, since it would have been difficult and unnecessary for S to have looked back to M for the details.

In the following I give one or two examples of each type of use of G and cite some others (without providing a complete catalogue). The examples are discussed in the commentary. Italics indicate where S follows G, not M.

3.2.2.1. S = M. S most often provides a straightforward, reasonably accurate translation of M. The following is one example among many. In the commentary, when S is an uncomplicated rendering of M (as S understands it), it usually receives no further discussion.

1:31

ויאכלו מפרי דרכם וממעצתיהם ישבעו

And they will eat of the fruit of their way and will be sated by their plans.

ممطه حر همته المحمد المحمد المتعلم محمد المتعلم محمد المعند المعند المحمد محمد المحمد المحم محمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحمد المحم المحمد المحم محمد المحمد الم

This verse is typical of the way that S often translates M word-for-word, with slight variations in number. Variations of this sort will not usually receive further mention.

6:16

שש הנה שנא יהוה ושבע תועבות נפשו

(a) There are six things the Lord hates, (b) and seven that are the abominations of his soul.

ὅτι χαίρει πᾶσιν, οἶς μισεῖ ὁ κύριος, συντρίβεται δὲ δι' ἀκαθαρσίαν ψυχῆς (a) For he rejoices in all things that the Lord hates, (b) and he is shattered through impurity of soul.

a) There are six things that the Lord hates, (b) and the seventh his soul despises,

Here S reflects the consonantal text of M but not its vocalization. S also reflects the *defectiva* writing תועבת. S may be implicitly vocalizing the word as a participle (though elsewhere תע"ב does not appear in the G-stem). In any case, S makes it clear that תועבת נפשו is a subjective genitive.

3.2.2.2. S  $\approx$ /= M + G. Sometimes the translator is basically oriented toward M but takes some items from G. This kind of "cherry-picking" shows S's active engagement in both his sources at once. In some cases, S and G may have the same non-M variant, but often no alternative Hebrew is likely. There are three subcategories, depending on the translator's motives.

3.2.2.2.1. Borrowing Items from G without Exegetical Need

6:28a M: "on coals": על הגחלים G: "on coals *of fire*": פֿת׳ מֿאטּףמֹאנשע תּטףסָ S: "on coals *of fire*": הנהוא הנהוא

always means burning coal, but ἀνθραξ can refer to nonignited charcoal. Therefore G adds precision by adding "of fire." S follows G's lead in adding "of fire" even though אמליים itself means "burning coals." For similar examples see 6:28; 7:20; 10:13; 18:3; and 22:15.

7:20b

M: "at mid-month": ליום הכסא G: "after many days": δι' ήμερῶν πολλῶν S: "and after many days": הריגהאה הם מביאה

G is guessing at the rare noun COR and gives an approximate rendering. S follows G exactly, though the exact cognate belongs to the Syriac lexicon and is used to translate COR (COR) in Ps 81:4.

3.2.2.2.2. Borrowing Components from G to Solve an Ideologically Neutral Interpretive Problem

20:5

מים עמקים עצה בלב איש ואיש תבונה ידלנה

(a) A plan is deep water in the heart of a man, (b) but a man of understanding can draw it up.

ύδωρ βαθὺ βουλὴ [var λόγος] ἐν καρδία ἀνδρός, ἀνὴρ δὲ φρόνιμος ἐξαντλήσει αὐτήν.

(a) A counsel [var: a word] is deep water in the heart of a man, (b) but a prudent man will draw it out.

محلمہ حلحہ مدی مرحلہ محلمہ محلمہ مدین (a) Deep is a word in the heart of the king, (b) but a wise man can draw it up.

Here S is basically working from M, though he mistakenly attaches מים to the preceding verse, in spite of the grammatical noncoordination in the resulting עמקים עצה Then אלאים הבלאים אוסיקים איז איז לא היא manuscripts; see comment. Next, on the basis of 25:3, S replaces "man" by "king." S may consider it necessary to specify what kind of man has counsel deep in his heart, seeing that many men lack this trait. Thus by a combination of mistakes and choices, S creates a proverb of his own. Other examples are 5:6; 13:2; and 24:34.

3.2.2.2.3. Borrowing Components from G to Solve an Apparent Logical or Ideological Difficulty

12:23

(a) The shrewd man covers up knowledge, (b) but the heart of dolts cries out folly.

ἀνὴρ συνετὸς θρόνος αἰσθήσεως, καρδία δὲ ἀφρόνων συναντήσεται ἀραῖς.
(a) The intelligent (man) is a *throne* of knowledge, (b) but the heart of the foolish will meet with curses.

منه جنحک منه ماه ماه ماه منه منه منه منه منه (a) A clever man is a *throne* of knowledge, (b) but the heart of fools calls evil [or "for evil"].

On the surface, at least, M seems to say that a wise man should conceal knowledge, whereas Proverbs often recommends speaking it. To solve this ostensive logical problem, G treats כסה as if it were בסה. S does the same. This is a deliberate exegetical choice, since S knows what כסה means. In the second stich, S looks to the Hebrew text, deriving 'קרא', not 'קרא', and using a broader term for אולת, namely, הביצא', as in 15:14. S uses both his sources while giving his own interpretation to the verse as a whole. Other examples are 12:17; 28:21; and 29:21.

3.2.2.3. Working from G but Making Changes. Sometimes S takes the thought or basic structure of the verse from G but looks back to M for individual components or introduces innovations of his own. In a type already discussed (\$3.2.2.1), M is foundational; in this type, however, G is the starting point. In examples of the latter type, since S follows G in some notable specifics and in its overall sense, it is reasonable to assign the entire verse to G's influence.

12:25a

דאגה בלב איש ישחנה

Worry in a man's heart brings him low [translation uncertain].

φοβερὸς λόγος καρδίαν ταράσσει ἀνδρὸς δικαίου. A frightening word disturbs the heart of a just man.

A *frightening word disturbs* the heart of a man.

Attempting to make sense of an obscure sentence, S incorporates most of G 12:25a, adopting the meaning given to T and the syntax of the clause. But he immediately returns to M and omits  $\delta i \alpha a \omega$ , thereby, surprisingly, removing the moralizing element.

3.2.2.4. Conflation of M and G: Taking One Line from M and One from G

12:17

יפיח אמונה יגיד צדק ועד שקרים מרמה

(a) The faithful witness speaks what is right, (b) but the witness of lies—deceit.

ἐπιδεικνυμένην πίστιν ἀπαγγέλλει δίκαιος, ὁ δὲ μάρτυς τῶν ἀδίκων δόλιος. (a) A righteous man declares what is *clearly* honest, (b) but the testimony of the wicked is deceitful.

مص محمال العمامي محمل العمامي معمود العمامي معمود (a) The righteous man speaks faithfulness *that is visible*, (b) but the witness of iniquity is deceitful.

3.2.2.5. S = G. Sometimes S translates a verse of G in its entirety, usually in order to deal with an exceptical or theological difficulty. In these cases, it is clear that S is relying on G alone and is doing so to preserve some of the components he found only in that source.

7:10

והנה אשה לקראתו שית זונה ונצרת לב

(a) And now: a woman (comes) toward him, (b) in harlot's garb, her intent hidden.

ή δὲ γυνὴ συναντῷ αὐτῷ, εἶδος ἔχουσα πορνικόν, ἡ ποιεῖ νέων ἐξίπτασθαι καρδίας.

(a) The woman meets him, having a whorish *appearance*, (b) *who/which makes the hearts of youths fly away*.

```
במהש אינאנא לאחובת באמבודא גועלא. גדפוגא לבתום, גראוניגא
```

(a) And a woman comes out to meet him, *in the appearance* of a harlot,(b) *who/which makes the heart of youths flee*.

68

G is responding to an apparent problem in M: guarding one's heart should be a virtue, as in Prov 4:23, not a sign of the woman's wickedness. S follows G's guess, with no reference to M. S's حصنات ("scatters" or "disturbs") is an interpretation of the *hapax* فلاألم ("scatters" or "disturbs") is an BHQ notes, translates the verse similarly, but he comes on this notion via T, which received it from S. Other examples are 11:31 and 19:7c.

3.2.2.6. Translating or Paraphrasing G alongside M, Thereby Producing Doublets. Joosten (1995, 63–72) examines seven double translations in S and argues that they were available to the S-translator rather than being produced by an inner-S interpolator. Here is one example:

14:23

בכל עצב יהיה מותר ודבר שפתים אך למחסור

(a) In all striving there is profit, (b) but mere talk just leads to neediness.

έν παντὶ μεριμνῶντι ἔνεστιν περισσόν, ὁ δὲ ἡδὺς καὶ ἀνάλγητος ἐν ἐνδεία ἔσται.

For [lit. "in"] everyone who takes care there is a profit, but the comfortable and indifferent will be in need.

```
כבל דישים אולי עדא הי דיבאיא אישא דבישי בבמדיה עעא
מכסעדא נהמאי כל באב דינא דאמא. מדדללא דיספמאמה
דבהלא בעסו להמי
```

(a) In all about which you are anxious, there is one profitable thing. (b) And he who is poor in his life [or "dwelling"] will be comfortable and happy. (c) The Lord heals all pain, (d) but the utterance of evildoers' lips will cause them loss.

S 14:23ab ≈ G. The second couplet, 14:23cd, is dependent on a text close to M. Neither translation is a correction toward M. See Joosten 1995, 67–68. Other examples are 11:29; 14:9; 14:22; 14:23; 14:32; 18:22–22*a*a; and 23:31.

3.2.2.7. Incorporating Greek Additions. G has numerous additions: epigrams, verses, and lines that are not found in M. There are about twentyfive additions of at least one couplet in length. S transmits six of them: 9:12*a*-*c*; 9:18*a*-*d*; 11:16*cd*; 13:13*a*; 25:20*a*; and 27:21*a*. S clearly considers these additions integral to the book and wishes to preserve them in his

#### 3. THE VERSIONS

own version. The puzzle, which remains open, is why he does not translate them all.<sup>22</sup>

In no case is there reason to think that S has a Hebrew source of the additions he translates. To the contrary, there is some evidence that S is working from the Greek. For example, in 9:18a, S ice "jump" reflects, but misunderstands,  $\dot{\alpha}\pi\sigma\pi\gamma\delta\eta\sigma\sigma\nu$ , which can mean "jump" or (as intended here) "get away."

3.2.2.8. Following G in an Omission. S follows G in omitting the ascription משלי שלמה ("Proverbs of Solomon") in 10:1a, though he does not consistenly avoid implications of non-Solomonic authorship elsewhere, as G does. See the commentary on 10:1.

3.2.3. A Profile of S-Proverbs

The translator of S-Proverbs was not a passive servant of the Hebrew or the Greek. Rather, he actively exercised his judgment about which components of each source, if any, he would import into his own product. This is translational flexibility, a quality to be measured not by the degree of divergence from the wording of the original but by the scope of the translator's autonomy vis-à-vis his source text, as manifest in the range of options he employed himself in representing his sources. Weitzman aptly compares the OT-Peshitta's translation technique to Jerome's (1999, 79). Jerome worked basically from the Hebrew but often—and to varying degrees—turned to the Septuagint. Verwijs's remarks on S-Amos can be applied to S-Proverbs as well: "The translator of the Peshitta shows respect for the Septuagint in using it to make choices about words, to solve problems with the Hebrew text, and to inform about the theological meaning of the text" (2005, 40).

Given S's wide-ranging dependence on G, S is usually shaky evidence for Hebrew variants when the two versions agree. (Again, aligning S with G in the apparatus indicates agreement, not support.) Shared variants may lie behind the versions, but S cannot simply serve as an independent wit-

<sup>22.</sup> The additions not translated in S are G 3:22*a*; 4:27*ab*; 6:8*a*-*c*; 6:11*a*; 7:1*a*; 8:21*a*; 10:4*a*; 12:13*a*; 13:9*a*; 16:17*a*-f; 17:6*a*; 17:16*a*; 18:22*a*; 19:7cd; 22:8*a*; 22:9*a*; 22:14*a*; 24:22*a*-*e*; 25:10*a*; 26:11*a*; 27:20*a*. This is apart from corrupt doublets (11:12*a*) and displaced verses (15:27*a*; 15:28*a*; 15:29*a*b; 20:9*a*-*c*; 28:17*a*). In any case S does not follow G's arrangement.

#### 3.4. THE TARGUM (T)

ness to them. To use S as a textual witness, we need to see independence in S's handling of a Hebrew variant that can be retroverted from both versions. For further features and practices of S, see "phenomena" and "topics" in the Syriac index.

#### 3.3. VULGATE

Jerome translated Proverbs into Latin in 398 CE, toward the end of his career (d. 405 CE). He worked from the Hebrew with the help of the Septuagint and with heavy reliance on the literalistic translations, especially Aquila.

Jerome's Hebrew source was almost identical with the consonantal M. He drew on Christian and rabbinic exegesis (C. Gordon 1930). The Vulgate offers little evidence for textual variants in Proverbs. I mention it on the few occasions when it arguably supports a variant to M. I also occasionally mention an interesting interpretation in V, but its main value, not exploited here, is for the history of interpretation. V is cited in *ketiv-gere* notations; see §4.3. The electronic text I used is based on Fischer and Weber, *Biblia Sacra*, 1983, and variants are cited from its apparatus.

I have consulted the Old Latin version only sporadically. This muchneglected version requires a study in its own right. Jerome's use of it in Proverbs is unclear.

I have benefited from a set of annotations prepared by Catherine Bonesho. It is to be hoped that these annotations will grow into a full study of V-Proverbs. In the meantime, Ms. Bonesho has made a draft of her annotations available at https://www.academia.edu/8534503/Annotations\_to\_ Vulgate-Proverbs. The document includes an index of translational phenomena in V-Proverbs.

#### 3.4. THE TARGUM (T)

### 3.4.1. Edition Used

No critical edition of T-Proverbs is available. My default text is de Lagarde's *Hagiographa Chaldaice* (1873) (T<sup>L</sup>), which is based on *Rabbinic Bible* II (Venice 1525). (Unmodified "T" in practice refers to  $T^L$  and is used when there are no significant variants.) I sometimes cite  $T^Z$  = Zamora (= San Bernardo 116-Z-40; Díez Merino 1984),  $T^W$  (Walton's Polyglot, 1628–1655), and  $T^{P-110}$  (Paris Hebreu 110, taken from Goldstein's collation in

#### 3. THE VERSIONS

chapters 1–9 and elsewhere from Healey's textual notes). As a rule of thumb, whichever variant is closest to S is the earliest. Binyamin Goldstein of Yeshiva University has graciously let me use a draft of his diplomatic edition of T-Proverbs. This work in progress had reached chapter 9 when I used it. When completed it will be an important contribution to the field.

# 3.4.2. The Date of T-Proverbs

M. Weitzman (1994, 81–83) brings evidence for Jewish use of T-Proverbs by medieval rabbis, allowing him to date this translation to the last third of the first millennium CE. The Targum of Proverbs, unlike many other targumim to the Hagiographa, shows very little midrashic paraphrase or expansion. T-Proverbs' features are described most thoroughly by Melammed (1972, 18–20, 54–72).

# 3.4.3. T-Proverbs' Relation to M and S

The main peculiarity of T-Proverbs is that it is directly and extensively dependent on the Syriac. T-Proverbs transcribes the Syriac into square script, with dialectal and orthographical adjustments toward Western Aramaic. T departs from S only to conform to M (as T understands it) or to make some minor changes in orthography and lexical choices.

The theory of T-Proverbs' dependence on S, first proposed by Johann A. Dathe (1764), is now the consensus. Dathe's hypothesis has been supported by Hermann Pinkuss (1894, 110–13), Ezra Z. Melammed (1972), Luis Díez Merino (1984, ch. 3 and p. 307), and Robert J. Owens (1998). The latter surveys and evaluates the various theories.

Other theories have fallen by the wayside. Siegmund Maybaum (1871) argued for Syriac dependence on the Targum but had few followers. S's strong dependence on G (see §3.2.2) shows this to be impossible. Armand Kaminka (1931–1932) maintained that the numerous T-G agreements are evidence of a common source in a pre-Tannaitic Hebrew text. This tortuous theory is vitiated, however, by the fact that almost all the T-G agreements appear also in the Syriac (e.g., 1:21; 6:30; 7:22; 8:13), whereas there are a great many instances where S and G agree against T. Pinkuss (1894, 110), counts more than one hundred instances of T = S = G; see also Kaminka (1931–1932, 178–79). The few cases where T seems closer to G than to S are mostly incidental to harmonization of T with M (e.g., 6:7a) or due to coincidental agreement in interpretation.

T-Proverbs is nearly identical to S-Proverbs in about 300 of its 915 verses (Kaminka 1931–1932, 171). Daniel C. Snell (1993) counted the words that T and S share (setting aside G pluses) and found a 72 percent correlation. The thorough study by Melammed (1972) found 410 verses dependent on S. Disagreements of T with S are almost always due to T's attempt to adjust to M. In some cases, T's Syriac source text apparently differed from the extant S manuscripts.

Because T-Proverbs is derived from S and adjusts to M, it has little text-critical value. Very rarely, T may support a minor variant independently. This may happen when T uses a synonym of a word in S without thereby adjusting to M. One case is הצבה for הצבה in 9:1.

In the commentary, "S (T)" or "T = S" means that T treats the verse in the same way as S, apart from trivial adjustments. Since this is usually the case, T is mentioned only when an apparent divergence might lead one to think otherwise. When there is a *ketiv-qere* pair, T is aligned in the apparatus with the appropriate item as possibly representing the medieval source text T used when adjusting S to M. But this datum is really relevant only when T and S agree with different members of the pair.

Following are examples of T-Proverbs' transliteration and modification of S-Proverbs into Aramaic script and usages. In these examples, the Syriac script is transposed into the Aramaic for clarity. A raised colon (<sup>:</sup>) is used for *syame* (the Syriac plural marker).

3.4.3.1. T = M

3:25

M: ואל־תירא מפחד פתאם ומשאת רשעים כי תבא Do not fear a sudden terror or the disaster of wicked men when it comes.

S: לא תדחל מן דלוחיא דמן שליא ומן חאפא דחטיא<sup>:</sup> מא דאתא Do not fear sudden turmoil or the violence of wicked men when it comes.

T: לא תדחל מן דחליא מן שלייא ומן חיפא דרשיעי כד ייתי Do not fear sudden terror or the violence of wicked men when it comes.

T might be considered a close translation of M, though it in fact copies S, with slight variations (דחליא) is semantically closer to M). T uses a western form of "when" and removes the initial conjunction, in agreement with S. 3.4.3.2. T ≈ M

6:10

מעט שנות מעט תנומות מעט חבק ידים לשכב M: A bit of sleep, a bit of snoozing, a bit of clasping hands to lie down....

קליל תנום וקליל תדמד וקליל תסים אידד על חדיד S: You briefly sleep and briefly slumber and briefly place your hand on your breast....

קליל שינתא קליל נומתא קליל תחבק ידד למדמכא T: A little sleep and a little slumber, and you briefly clasp your hands to sleep....

T uses S as a starting point but then adjusts to M. Whereas S (influenced by G) treats the three actions as verbs, T makes the first two into nouns, as in M. The third action, however, it translates as a verb, following S. T adjusts to M also by using the same Aramaic verb as in the Hebrew (חבק) and adding an equivalent of לשכב, which is lacking in S.

3.4.3.3.  $T = S \neq M$ 

26:7

M٠

דליו שקים מפסח ומשל בפי כסילים Legs dangle from a cripple, and a proverb in the mouth of dolts.

אן תתל הלכתא לחגירא תקבל מלתא מן פומה דסכלא S: If you could give walking to the cripple, you could receive a word from the mouth of the fool.

אין תיתל הליכתא לחגירא תקביל מילתא מן פומיה דסיכלא T: If you could give walking to the cripple, you could receive a word from the mouth of the fool.

T maintains S's radical paraphrase exactly, in spite of its difficulty, though the Hebrew is not especially difficult.

74

11:26

M:

מנע בר יקבהו לאום וברכה לראש משביר

He who withholds grain—the nation will curse him, while there will be blessing for the head of a distributor.

S: דכלא עבורא באולצנא נשבקיוהי לבעלדבבוהי<sup>:</sup> ובורכתא תהוא על רשה דאינא דמבזן

He who withholds grain in famine will leave it to his enemies, but there will be blessing on the head of whoever sells (it).

T: מן דכלא עיבורא באולצנא נשבקוניה לבעלי דבבוי וברכתא תהוי
 לאינא דמבזין

He who withholds grain in famine will leave it to his enemies, but there will be blessing on the head of whoever sells (it).

T copies S's interpretation of 11:26a (which comes from G) exactly. Melammed (1972, 71–72) lists twelve cases where  $T = S \neq M$  although M is not obscure: 3:27; 11:26; 16:4; 16:6; 16:21; 19:6; 20:4; 20:14; 21:30; 23:4; 26:10; 31:4.

25:20*a* 

M: absent

S: איך ססא למאנא ואיך בלטיתא לקיסא הכנא כריותא מהרא ללב דגברא

Like a moth to a garment and a worm to wood, so does illness trouble a man's heart.

T: היך ססא למנא ואיך בלטיתא בקיסא היכנא כריותא מהרא
 לביה דגברא

Like a moth to a garment and a worm [read בלטיתא; cf. T<sup>z</sup> בלטיתא] to wood, so does illness trouble a man's heart.

S took Prov 25:20*a* from G. T transcribed S 25:20 and kept on going, not noticing that this additional stich is missing in M. (He even takes over the Syriac word for "like," namely, איך.) Postulating a Hebrew variant in T's Hebrew text for this and only this G/S addition (Healey 1991, 53) is gratuitous. This verse is decisive proof of T's dependence on S.

# 4. Policies and Procedures

### 4.1. Anchor Bible Proverbs

I often refer to ABP, my Anchor Bible Proverbs commentary. The ABP includes textual notes at the end of both volumes, which I sometimes cite here, usually ad loc. I have sometimes changed my mind since writing those notes, and I have even more often developed and sharpened my argumentation. I do not usually remark on differences from ABP, for that commentary now stands on its own, and the reader might sometimes find its decisions preferable. Note that in the present volume, "commentary" refers to the textual comments in chapter 5 of the present volume, not the ABP.

### 4.2. TRANSLATIONS

I frequently quote verses or stichoi<sup>1</sup> of the versions, particularly G, as needed for the clarity of the commentary. I translate foreign languages (Hebrew, Greek, Syriac, Aramaic, and Latin) into English. Though I assume knowledge of Hebrew on the reader's part, translation should help readers who do not know all the other languages well enough to follow the argument, and it should provide them with some tools to make critical decisions of their own. Moreover, the meaning of the Hebrew and the versions is not self-evident. By translating citations, I clarify my own understanding for the reader's consideration. Verses and stichoi that are not translated can be presumed to be close representations of M.

<sup>1.</sup> In this volume, "stich" (or "line," as in ABP) refers almost always to a component of the primary dichotomy, which is usually marked by an *`atnaḥ*. A verse is typically a couplet (two stichoi). Rarer are monostichs and tristichs. One may also speak of a quatrain, a unit of two closely associated couplets (e.g., 1:15 + 16).

# 4.3. Ambiguity of Citations in the Apparatus

# 4.3.1. Agreement ≠ Support

In the apparatus, the association of a translation with a variant indicates that the translation agrees with the variant but does not necessarily add evidence in support of its validity or even its existence. The notation "G S," for example, may mean only that S used G. Since S sometimes does witness independently to variants, it must be cited even if I think that S derives from G or that the agreement is coincidental or translation-dependent. These matters are considered in the commentary. I have not usually cited V and T in the apparatus, because V is dependent on M and G, and T is dependent on M and S, so the agreement of V and T with one of their sources may be presumed, except in rare cases. Given the willingness of all the versions to vary grammatical number and tense, these features will usually go unmentioned.

# 4.3.2. Lexical Assumptions

To construct the apparatus I must sometimes make the working assumption that the versions understood the variants as I do. Of course, if a word occurs frequently enough, the concordance can show how the translators understood it, but this is often not the case. Words may be difficult and sentences obscure. For example:

20:20 באישון  $M^{K}$  G (ai δè κόραι) ] באישון  $M^{Q}$ ; ראישון S (איד ארונא) T (איך אתונא) ( $M^{Q}$  M<sup>K</sup> G: orth, equal; S T: graph כבא)

In the commentary I draw a distinction between אישון ("time") and אישון ("pupil" of the eye) that the translators might not have recognized, but in order to align the versions with the variants, I must assume that the distinction was known to the sources cited. This is only a starting assumption and may be modified after further examination.

# 4.3.3. Atomizing Changes

The change from one word-form to another can take place by a series of steps or all at once. For the sake of the analytical description, I try to factor

the change into its minimal components, but I do not assume that they actually occurred in sequence. For example:

In other words, a series of graphic errors led from ערב בן איעל עד בליעל. In fact, it seems likely that the confusion happened at once, in the moment that a scribe glanced at the letters and misunderstood them. In 25:27 I emend בבדם כבוד אוחקר כבדם כבוד I describe the theoretical steps leading to M as רבדם לוחקר לבדם לוחקר. I describe the theoretical steps leading to M as רבדם לוחקר לוחקר לבדם, but I do not think that the changes were discrete. Other examples of textual changes atomized for descriptive purposes are found in 2:18; 6:5; and 18:6.

4.3.4. Written and Perceived, Written and Remembered

It is very difficult, and sometimes impossible, to distinguish a written, textual variant from a perceptual or "pseudo" variant, one that did not exist in writing, and it must be granted that some of the variants I propose may be of the latter sort.<sup>2</sup> Nontextual variants are of three sorts:

4.3.4.1. Perceptual Variants. If a translator looks at a Hebrew letter or word and mistakes it for a similar-looking one, he creates what I would term a perceptual variant. This too has a claim to reality. Whatever the ontology of the perceptual variant, it existed for at least a moment in a translator's mind and (like the lost textual variant) is preserved only in a translation (or possibly in a citation elsewhere). The trace at least shows us what *could* have happened in textual transmission and makes the retroversion more credible.

Sometimes the relation of the translation to the source enables us to make a case for written variation. An example is 13:19b. M: "What dolts

<sup>2.</sup> Tov (1997, 88–89, 162–71) speaks of pseudo-variants that may have only existed in the translator's mind. He observes that there are hardly any criteria for distinguishing these from written variants. However, in §3.1.7 I do offer some ways to recognize written variants (proposed earlier in Fox 2005). Whenever we sense that the translator is struggling with his text, it is likely that he has slowed down to focus on the physical form of his text.

loathe is turning away from evil (מרע)." G: "but the works of the impious [pl] are far from knowledge (μακρὰν ἀπὸ γνώσεως)." M's "from evil" would only have enhanced G's moralism, which permeates the entire translation and is underscored in this verse by the addition of "the pious" and "the impious." G's "from knowledge" comes from מדע, understood as מִדַע Other adjustments were made to accommodate this variant. In this case, a perceptual (mental) error is less likely than a graphic one. People are likely to see what they expect to see. The translator would be less likely to see the מרע and transform it mentally into the difficult מרע confusion was in his source text and not just a slip of perception.

4.3.4.2. Memory Variants. Memory variants arise in oral transmission but can accompany a written text. Like scribal errors, these too result in written variants. David Carr argues that literary transmission often involves "a mix of oral and written dynamics" (2011, 17); the former belong to memory variants. These are "the sorts of variation typical of memory-reconstructive processes: substitution of synonymous terms, radical adaptation of the tradition, etc." (2011, 17).

Carr takes the book of Proverbs as "a potential illustration" of these dynamics (2011, 25-34). He states his conclusions cautiously and emphasizes that we cannot know how any particular variant was formed, but he believes that there are some indicators of orality. To take one example, הוה איש ישר בעיניו ותכן לבות יהוה. Assuming that 21:2 was written later than 16:2, it may be that a scribe had memorized (or simply remembered) 21:2 but accidentally substituted synonyms. Since the changes in meaning are slight, "[t]hey are as likely or more likely to be the results of shifts in memory as the product of conscious alteration" (28). This assumes that conscious alteration produces only significant changes. It may be that a scribe chose to vary the earlier proverb slightly, to produce a new one. An example that runs counter to Carr's explanation is 19:9 (עד שקרים) עד שקרים לא ינקה ויפיח (לא ינקה ויפיח כזבים יאבד) in relation to 19:5 (לא ינקה ויפיח כ כזבים לא ימלט). It is hard to believe that the scribe forget the second stich of 19:5 while inscribing the intermediate three verses. This must be deliberate variation, even though-or perhaps because-the message does not change, allowing the second verse to drive home the teaching of the first.

Carr identifies some types of variants as characteristic of memory transmission: "exchange of synonymous words, word order variation,

presence and absence of conjunctions and minor modifiers, etc." (2011, 33). But these are not exclusive to oral transmission or even predominant in it. As far as I can tell, there are no diagnostic criteria distinguishing memory variants from visual ones—though some variants, such as graphic confusions and metatheses of letters, are diagnostic of visual errors. (Aural variants too can occur in written transmission, when a scribe dictates the text to a copyist.) Some variant types could have arisen either way.

The oral-written dichotomy (which Carr himself has done much to elide) may not after all be of fundamental importance to interpreting and editing a text. A text may be transmitted in different media, including speech that is recalled and speech that is recorded in writing. In the end, the process is not crucial to the resulting text, which in either case comes to us in written form, in documents.

4.3.4.3. "Midrashic" Variants. A translation might reflect not what the translator saw but what he thought it must say. There are mental manipulations that could make sense of a text that is wrong, obscure, or unacceptable. I call them (loosely) "midrashic" because they recall the sorts of exegetical rereadings familiar from classical midrash. For example, Pesher Habakkuk transforms למול (1QpHab 1:5-6), probably without having the latter in his source text, and CD 7:14–15 revises כמות of Amos 5:3 into סכות tabernacle," which may actually be original. These examples and sources are taken from Bernard M. Levinson (2011, 183), who discusses the way the Septuagint manipulates Deut 13:10. For other examples, see Scriba 1995, 148–50 (though these examples could be graphic errors); Junker 1936, 162–74; and de Waard 1993. There is no clear way to identify this sort of pseudovariant, especially since they may well seem to be the better, more expected reading.

4.4. What Goes into the Critical Text?

For reasons explained in §1.3, I use parentheses in the critical text to mark components I consider to have been in proto-M (because they are shared by G). I do this only for additions that are, in my view, intrusive in the units in which they are embedded.

An emendation must finally stand on its own. Support from the ancient translations and even other Hebrew texts can at best show that a variant *existed*, not that it is valid. Even a Hebrew variant may be wrong. Retroverting a variant from the versions adds layers of difficulty, and even

### 4. POLICIES AND PROCEDURES

if the recovery of the Hebrew source text is successful, the variant may be wrong. When no extant variant provides a meaningful reading, I may resort to conjecture, which is sometimes necessary but never certain.<sup>3</sup> Fully aware of these obstacles to recovering the text from the sources, I will embed variants only when they have a strong enough claim to validity that they could stand on their own without versional support.

<sup>3.</sup> For a vigorous defense of the necessity of conjecture in New Testament text criticism, see Wettlaufer 2012.

5. Textual Commentary with Critical Text

1:1-8

84

גוו גָשְׁלֵי שְׁלֹמֵה בֶּן־דָּוֹד מֶׁלֶדְ יִשְׂרָאֵל:
גַּקַתַּת הַפְּהֵי שְׁלֹמֵה בֶּן־דָּוֹד מֶׁלֶדְ יִשְׂרָאֵל:
גַּקַתַת הַפְהָה וּמוּסֵר לְהָבִין אִמְרֵי בִינֵה: 3 לְקַתַת מוּסַר הַשְׂבֵּל גַּעָר דַעַת וּמְזִמֶה:
גָשָׁרַים: 4 לְתַת לִפְּתָאים עָרְמֵה לְנַעַר דַעַת וּמְזִמֶה:
זִמִשְׁבָּט וּמֵישְׁרִים: 4 לְתַת לִפְתָאים עָרְמֵה לְנַעַר דַעַת וּמְזִמֶה:
זִמִשְׁבָט וְהַיֹשֶׁרִים: 4 לְתַת לִפְתָאים עָרְמֵה לְנַעַר דַעַת וּמְזִמֶה:
זִשְׁמַע חְכָם וְיוֹסֶף לֶכֵח וְנָבוֹז תַּחְבָּלוֹת יִקְנֶה: 6 לְהָבֵיז מֲשָׁל
וּמְלָיצֵה דִּבְרֵי חְכָמִים וְחִידֹתֵם: 7 יִרְאַת יְהוֶה רֵאשִׁית דֶעַת חְכָמָה וֹמוֹסָר אָבִיז וְמָשִׁל וֹמוֹלָר אָנִים בַּזוּ:

1:1

G translates מלך ישראל as δς ἐβασίλευσεν ἐν Ισραηλ ("who ruled in Israel"). Compare the rendering of מלכי יהודה by a relative clause in Isa 1:1 and 1 Kgs 4:1. G does not represent אשר מלך (= מָלַדָ) בישראל.

1:3

G: (a) δέξασθαί τε στροφὰς λόγων, (b) νοῆσαί τε δικαιοσύνην ἀληθῆ (c) καὶ κρίμα κατευθύνειν. "(a) To receive twistings of words, (b) to understand true righteousness, (c) and to guide justice straight." G creates three stichoi from M's two.

G 1:3a: στροφὰς λόγων ("twistings of words," i.e., convoluted sayings) is a puzzling translation of מוסר, which is otherwise rendered by παιδεία, σοφία, and synonyms. In both Sir 39:2–3 (no Hebrew preserved) and Wis 8:8, στροφή is associated with αἰνίγματα ("enigmas"). Sir 6:22 (Hebrew) reveals an "etymological" interpretation of מוסר in the sense of twisted and difficult, probably associating it with מוסר ("turn aside") (CSP 50). G may also be playing on שׁכל and the homonymous שׁכל "cross"; cf. Gen 48:14 (Kuhn).

G 1:3b: G adds ἀληθῆ, an adjectival elaboration of "righteousness" for moral emphasis. CSP (64) notes that G often adds adjectives for explication, e.g., (παιδὶ +) νέω in 1:4 and (κλοιὸν +) χρύσεον in 1:9.

G 1:3c: G translates ומישרים as an infinitive, possibly thinking of an Aramaic infinitive. De Lagarde retroverts אמדפטטעוע, an Aramaizing infinitive, but this is distant graphically. G shows considerable flexibility in the treatment of grammatical forms.

S has אוסָר הַשְׁבָּל ("instruction and fear") for מוסָר הַשְׁבָּל. Since S is flexible in its treatment of the *waw*-conjunctive, this is not evidence for השכל (contra BHS), especially since S's idea that השכל means "fear" is peculiar and unparalleled in its other occurrences in Proverbs. 1:4

G translates פתי as ἀxάκοις "innocent." In contexts where the פתי has not yet sinned, G uses ἀxακος (1:22; 8:5; 14:15; 21:11). If one has sinned, he is an ἀφρων (7:7; 9:4, 16; 14:18; 19:25; 22:3; 27:12). (ἀκακος is also used for forms of ממ"ם in 2:21 [some G MSS] and 13:6.) In this way the translator reinforces moral polarities by dividing an ambiguous class into two morally antithetical groups (see Giese 1990, §2.11–13). For M's compasize that the verse refers to the earliest stage of instruction and more clearly distinguishing the elementary stage of education in 1:2–4 from the advanced stage in 1:5–6.

1:7 init + 2 stichoi G (7ab) (anaph)

G: (a) ἀρχὴ [G<sup>BS</sup>; var: ἀρετή G<sup>A</sup>] σοφίας φόβος θεοῦ, (b) σύνεσις δὲ ἀγαθὴ πᾶσι τοῖς ποιοῦσιν αὐτήν. (c) εὐσέβεια δὲ εἰς θεὸν ἀρχὴ αἰσθήσεως, (d) σοφίαν δὲ καὶ παιδείαν ἀσεβεῖς ἐξουθενήσουσιν. "(a) The beginning [var: 'best'] of wisdom is the fear of God, (b) and (it is) a good understanding for those who perform it. (c) And piety toward God is the beginning of knowledge, (d) but the wicked hold wisdom and instruction in contempt." Fritsch (1953, 170) regards 1:7ab as OG and 1:7cd as Hexaplaric (but unmarked). Within 1:7a, the variant ἀρετή is an alternative interpretation of **Γκυντ** and not a permutation of the majority reading ἀρχή (see ABP 1.67). Whether **Γκυντ** means "first" or "best" is still a matter of dispute.

The prefixed couplet, 1:7ab, is taken from Ps 111[110]:10: רָאשִׁיהָם רָאשִׁיהָם (For the phenomenon of anaphora, the transfer of material from another part of G, see Heater 1982, 6 and passim.) CSP considers all four stichoi to be OG, with the translator responsible for the quotation from Psalms. However, since the addition in G 1:7a differs somewhat from the Greek of Ps 111[110]:10a (which reads ἀρχὴ σοφίας φόβος χυρίου), as well as from G-Prov 9:10, it was likely transferred in the Hebrew transmission and present in G's source text.

1:8

G: (a) ἄχουε, υἰέ, παιδείαν [G<sup>B</sup>; var: νόμους G<sup>ScA</sup>] πατρός σου (b) καὶ μὴ ἀπώσῃ θεσμοὺς μητρός σου. "(a) Hear, son, the instruction [var: 'laws'] of your father, (b) and do not reject the rules of your mother."

טוֹצ: In accordance with Greek style, G-Proverbs never represents the semantically superfluous possessive suffix of the vocative בָּנ

תּוֹרַת אָמֶדְ: 9 כָּי | לִוְיָת חֵׁן הֵם לְרֹאשֶׁדְ וַׁעֲנָקִים לְגַרְגְּרֹתֵידְ: 10 בְּנִי אִם־יְפַתּוּדְ חַטָּאִים אַל־תּּבֵא: 11 אִם־יאׁמְרוּ לְכֶה אָׁתְּנוּ נָאֶרְבֵה לְדֶם נִצְּפְּנֶה לְנָקֵי חִנֵּם: 12 נֵבְלָעֵם כִּשְׁאַוֹל חַיֵּיֵם וּתְמִימִים כְּיִוֹרְדֵי בור: 13 כָּל־הַוֹן יֵקָר נִמְצֵא נְמַלֵּא בָתַּינוּ שָׁלֵל: 14 גּוֹרָלְדָ תַּפֵּיל בַּתוֹבֵנוּ כִּיס אֶׁחָד יִהְיֵה לְכֵלֵנוּ: 15 בְּנִי אַל־תֵלֵדְ בְּדֶרֵד אָתֵ

similar differences in the treatment of the vocative belong to translation style and will not be noted further, but see the index.

παιδείαν in G<sup>B</sup> looks like an adjustment to the expected rendering of . CSP regards it as original and BHQ explains it as an assimilation to G 1:8b. By the variant reading νόμους, G is contrasting the "laws" of the father with the "rules" of the mother; cf. G 6:20. The weightier word is νόμος, being the standard designation of the divine as well as the paternal *torah*. θεσμός means "custom," "rule." (In Jewish Hellenistic literature, θεσμός means "custom," "rule," "mode of practice," both pagan [5 Sib. 19] and Jewish [T. Naph. 8:10]. θεσμός refers to the Torah proper only in the mouth of a hostile gentile (4 Macc 8:7). (In Sir 28:19 read δεσμοῖς with G<sup>BS</sup>.) The choice of θεσμός ascribes lesser authority to the mother's words.

S גראסש for מוסר is based on the variant אנסטע, Elsewhere S translates מוסר ("instruction").

1:9

At the end of 1:9a, G adds کלא ("will receive") to the end of the sentence. It apparently made more sense to the literal-minded translator to say that you will *receive* a reward, represented by beads, than that Torah *is* beads. For M's וענקים G has אלסוסע עלטסנסע ("golden necklace"), for rhetorical heightening. S (sim T) explicates לוית חן ("a wreath of grace") as אסגא ("beauty"; also in 4:9), perhaps because it does not know the meaning of the *dis legomena* לויה (BHQ).

1:10

G's βουληθής ("[nor] should you consent") agrees with M's vocalization אב"ה (from אב"ה "be willing"), as does S's אב"ה ("be persuaded"). Many MSS KR write this *plene:* תאבה. Two MSS KR normalize this as תאבה, i.e., אָבוֹא ("come"), a reading reflected also in the Greek tradition; see BHQ. ונכרית מארץ זכרו 🖁 (καὶ ἄρωμεν αὐτοῦ τὴν μνήμην ἐκ γῆς) (anaph)

#### 1:13

G τὴν κτῆσιν αὐτοῦ ("his property"): G and S often add possessive pronouns, generally for clarity but also sometimes when the Hebrew is unambiguous. Such additions will henceforth go unmentioned, except in the rare cases where they are relevant to determining the source text. More often, G removes possessive pronouns it considers unnecessary. For examples, see "possessive adjective/pronoun" in the index.

#### 1:14

G: (a) (b) = M; (c) καὶ μαρσίππιον ἕν γενηθήτω ἡμῖν ("and let us have one purse"). The last stich is a Hexaplaric doublet of 1:14b. It is marked in part by an asterisk in SyrH and is lacking in OL.

# 1:15 בני M S (בו, ] > G (unc)

G does not represent "my son." Both readings are possible in Hebrew. The additional "my son" in M does seem to overload the line, but the same imbalance is caused by "my son" in M 1:10, and compare the asymmetries in 3:12, 28. G has the vocative vié where M lacks أج in 2:17; 9:12; 23:22; 24:1; and 27:27. M's קרי is not represented in G 1:15; 5:20. M's plural is translated with a singular in 5:1; 7:24; 8:32. In the other nineteen cases (see index),  $\Box = vié$ . When  $G \neq M$  in this regard, there is no criterion for determining originality, since either version may have added the word to adjust to the common usage.

For M נתיבתם (sg), G, S, and 2 MSS KR have the plural "paths," perhaps because the verse is speaking about a plurality of people. This need not indicate an original consonantal variant. Variation in grammatical number is frequent in the versions, including G, V, and S (but less so in T, which tends to correct toward M). There is also considerable number variation in Hebrew manuscripts. The variations usually have no clear motivation. Only variations that are significant to interpretation or present in ancient manuscripts will be mentioned.

ַרַגְלְדְּ מִנְּתִיבָתֵם: 16 פִּי רַגְלֵיהֶם לְרַע יְרָוּצוּ וִימַהֲרוּ לִשְׁפְּדְ־דֶּם: 17 פִּי־חֻנָּם מְזֹרֵה הָרֶשֶׁת בְּעֵינֵי כָל־בִּעַל כְּנֵף: 18 וְהֵם לְדָמֵם יֶאָּרָבוּ יִצְפְנוּ לְנַפְשׁתֵם: 19 בֵּן אָרְחוֹת כָּל־בִּצַע בֶּצַע אֶת־נֵפָש בְּעָלָיו יִצְחָנוּ הַנַפְשׁתֵם: 20 הַכְמוֹת בַּחַוּץ תְּרָנָה בְּרָחֹבוֹת תִּמֵּן קוֹלֵה: 12 בְּרָאשׁ הֹמִיוֹת הִֿקְרָא בְּפִתְחֵי שְׁעָרִים בָּעִיר אָמֶרֶיהָ תֹאמֵר:

1:16 M S  $G^{AMSS}$  ] >  $G^{BS^*} = OG$  (homoi כי $\cap$ с)

OG lacked this verse. (In the MSS that have this verse [notably G<sup>ScA</sup>], it is sometimes placed before, sometimes after 1:17, supporting the likelihood that in those MSS it is a later restoration.) De Lagarde considers M 1:16 a later addition based on Isa 59:7. Nevertheless, the verse is integral to context and original. Its literary function is to present an ambiguity (Whose blood are they really hastening to shed?) that is resolved in the continuation. This technique is a feature of the entire passage; see ABP. Isa 59:7 may draw on Proverbs, or both may draw on a common wisdom source. The latter is more likely because, as Ehrlich notes, it is unlikely that a glossator would remove "innocent" while importing it into Isaiah.

1:17

G: οὐ γὰρ ἀδίκως ἐκτείνεται δίκτυα πτερωτοῖς. "For not unjustly are nets spread out for birds." The added negative probably indicates that the translator understood the verse as a rhetorical question, which he then converted into a negative indicative to the same effect. (For examples of this technique in G-Job, see Orlinsky 1958, 244–46.) The translator understood "birds" as a metaphor for the wicked or gullible youths, whose fate is "not unjust." ἀδίκως corresponds to nullible youths, whose fate is "not unjust." ἀδίκως corresponds to the adage. The "birds" who are ensnared "not unjustly" must be the fools and criminals spoken of in the next verse.

BHS is mistaken in saying that כל is omitted in the versions. In fact, כל is translated as a unit by a word meaning "birds."

1:18

G: (a) αὐτοὶ γὰρ οἱ φόνου μετέχοντες θησαυρίζουσιν ἑαυτοῖς κακά, (b) ἡ δὲ καταστροφὴ ἀνδρῶν παρανόμων κακή. "(a) For the very ones who take part in murder store up evils for themselves, (b) and the downfall of lawbreakers is evil." μετέχοντες associates M's ארבו (b) with ער"ב "mix," etc. This is an aural confusion (though perhaps in the mind rather than the ear of the copyist) that would be more likely to arise in Hebrew copying than in translation. G understands this verse to mean that the bandits are harming themselves and states this plainly. (M's "for their blood" is deliberately ambiguous.) רמים is construed as רמים. S follows G's interpretation.

G 1:18b develops the idea of G 1:18a, emphasizing the disaster awaiting the wicked. It is too unlike M 1:19 to be a paraphrase of it and thus to serve as evidence for reading אחרית there (*pace* Mezzacasa; BHS on 1:19). (In any case, that word is never translated καταστροφή.) G frequently adds κακός for moral emphasis; see the index.

1:19

G τῆ γὰρ ἀσεβεία ("for by impiety") is a moralizing expansion and does not represent בעולה for M's בעליו (contrary to Jäger; de Lagarde; Mezzacasa; Toy). Rather, בעליו is implicit in G's "their own soul." Characteristic of G-Proverbs is the frequent use of words emphasizing the religious dimension of wickedness, in particular by using ἀσέβεια/ἀσεβής as the standard translation of  $\gamma$  (SDEK 1958). This is not an innovation but a matter of foregrounding an aspect of the Hebrew word's semantic content.

#### 1:20

G: (a) Σοφία ἐν ἐξόδοις ὑμνεῖται; (b) = M. "(a) Wisdom praises herself in the exits." The middle ὑμνεῖται (for תרנה "cries aloud") is to be understood reflexively (sim 8:3), since Lady Wisdom is singing her own praises in this chapter. Similarly S

## 1:21 המיות M S (הומיות [ (בדּגאm ) (קר ה הומיות G (דנאנישע) (graph ה $\rightarrow$ ה

G: (a) ἐπ' ἄκρων δὲ τειχέων κηρύσσεται, (b) ἐπὶ δὲ πύλαις δυναστῶν παρεδρεύει, (c) ἐπὶ δὲ πύλαις πόλεως θαρροῦσα λέγει. "(a) On the heights of the walls she cries, (b) at the gates of the princes she sits constantly, (c) at the gates of the city she says boldly" (continues in 1:22).

In G 1:21a, rather than M's hapax המיות ("bustling places," "cross-

22 עַד־מְתַי | פְּתָיִם מֵאַהַבֿוּ פֶּתִי וְלַצִּים לֻצוֹן חָמְדַוּ לָהֶם וּכְסִילִים יִשְׂנְאוּ־דֲעַת: 23 תָּשׁוּבוּ לֱתוֹכַחְתִּי הִגָּה אַבִּיעָה לְכֵם רוּחֵי אוֹדֻיעָה יִשְׁנְאוּ־דֲעַת: 23 תָּשׁוּבוּ לֱתוֹכַחְתִּי הִגָּה אַבִּיעָה לְכֵם רוּחֵי אוֹדֻיעָה יְשְׁנְאוּ־דֲעַת: 23 תָּשׁוּבוּ לֱתוֹכַחְתִּי הִגָּה אַבִּיעָה לְכֵם רוּחֵי אוֹדֻיעָה יְבָרֵי אֶתְכֶם: 24 יַעַן קָרָאתִי וַתִּמְאֵנוּ נָטִיתִי יְׁדִי וְאֵין מַקְשִׁיב: 25 תַּקַרְעוּ כָל־עֲצְתֵי וְתוֹכַחְתִּי לָא אַבִיתֶם: 26 גַם־אֵנִי בְּאַידְכֶם 125 אַתְכָם: 24 יַעַן קָרָאתִי וַתּמְאֵנוּ נָטִיתִי יְׁדִי וְאֵין מַקְשִׁיב: 25 אַתְכָם: 24 יַעָן קָרָאתִי וַתּכַחְתִּי לָא אַבִיתֶם: 26 גַם־אֲנִי בְּאַידְכֶם אָשִיחֶק אֶלְעַג בְּבָא פַחְדְכֶם: 27 בְּבָא רְכָשוֹאָה' | פּחְדְכָם וָאַידְכָם אָשִׁחָק אֶשְׁחָק אֶלְעַג בְּבָא פַחְדְכֶם: 27 בְּבָא רְכָשוֹאָה' | פּחִדְכָם וָאַידְכָם הָאַיחָק בְּסוּתָרָי זַיַרָאָת יוְהוֹבָקהי לָא אַבָּתָרָי זַיַרְאָת יוּזַבָּקוּרָיָי בָּאַיַרָכָם בָּבָא רְכָשוֹאָה' ו פַחְדְכָם וָאַידְכָם וְאַידְכָם בְּרָנוּ זְלָזי יִמְצָאָרָי וּבוּאַרָבין בָּתוּרָבָר וּצוּקָרָעוּ בָרָרָין בָּהָאוּרָכָם בָּתוּרְבָם בָּבָית בַיַם בָּתוּרָבָם וָאַידְכָם בָּבָית בְּבָיתוּ הַבָּרוּתוּים הַיָּים בּבוּתוּבָם בּבָא רְכָם בָּבָית בְרָבוּים בָּרָבָם בָּתוּבָרָם בָּבָם בָּרָתוּידָים: 30 אַרָּבָם בָּבָיתוּבָם בַּרָבָים בָּרָים בָּרָבָם בָּרָבוּים בַיקרָם בּבָרָם בָּרָבָם בָּרָים בָּרָבָם בָּתוּבָרָם בָּתוּבָם בָּאַנוּין כָים בַיַיָּים בָּים בַּקוּים בּבּבָא בַיַים בָּרַיָם בָּרָם בָּתוּתוּים בַים בָיקרָם בַים בַּבּיים בָּייִינָם בַּרָים בָּרָים בַּרָים בָּרָים בַים בָּרָים בָּים בַיַים בַיּתוּים בּבָים בָּרָים בָּרָם בָּים בַייִים בַים בּים בָּתוּים בּביים בָּיירָים בּיירָים בַים בָּרָים בָּיתוּין בָּים בַיּים בַים בָּים בַים בַיתוּים בּיים בַיין בָים בַין בָּים בַייוּים בַים בּיתוּים בָים בַיהיים בַיה בָּיים בָיהוּין בָיוּים בָייוּיבָים בָים בַירָים בָיהיים בַיוּים בַיין בָים בָּיוּים בָיהיים בַיהיים בַיהוּים בַייין בָיים בָיים בַיים בַייין בָּין בָייוּיין בָיים בַיין בָּיים בַיין בָיין בַיין בָים בַיין בָים בייין בָיים בייוּיין בָיים בייין בָיים ביים בייז בַיין ב

roads"), G's דפועלש = חומיות סר חומיות ("walls"). The latter would be a variant plural of חומות, a form attested at Amarna (Tur-Sinai 1947, 105) and in Ugaritic and Phoenician (Dahood 1963, 5). (BHQ ascribes the rendering to ignorance of the meaning of M's word, but the translator caught the gist of המ"ה in 7:11; 9:13; and 20:1 and could have rendered it appropriately here.) Nevertheless, M is right, for Wisdom is going about the busy streets of the city, to people, not on the walls. G 1:21b is taken from G 8:3, q.v.

S's המיות ("streets") is a correct interpretation of בבאה "bustling places." T saw בירתא but understood it as Aramaic בירתא ("fortress," "palace"). This reading may be earlier than T; see b. Ber. 53b: בראש הבירה ("at the top of the fortress").

1:22

G (continuing 1:21): (a) Όσον ἂν χρόνον ἄκακοι ἔχωνται τῆς δικαιοσύνης, οὐκ αἰσχυνθήσονται. (b) οἱ δὲ ἄφρονες, τῆς ὕβρεως ὄντες ἐπιθυμηταί, (c) ἀσεβεῖς γενόμενοι ἐμίσησαν αἴσθησιν (23a) καὶ ὑπεύθυνοι ἐγένοντο ἐλέγχοις. "(a) As long as the innocent possess righteousness, they will not be put to shame, (b) but the fools, being desirous of pride, (c) having become impious, hated knowledge." What is notable about G 1:22–23a is not the introduction of antithesis (observed by Dick 1990, 22) but the way the verse is cast as a generalization in intricate Greek syntax. This style makes Sophia into something of a philosopher. 1:23

G (continuing 1:22): (a) καὶ ὑπεύθυνοι ἐγένοντο ἐλέγχοις. (b) ἰδοὺ προήσομαι ὑμῖν ἐμῆς πνοῆς ῥῆσιν, (c) διδάξω δὲ ὑμᾶς τὸν ἐμὸν λόγον. "(a) and became subject to reproofs. (b) Behold I will extend to you the utterance of my breath, (c) and I will teach you my word." G 1:23a continues 1:22c, with its syntax ignored for the sake of cohesion. In 1:23a, G adds ῥῆσιν ("utterance") to parallel λόγον ("word"). This addition clarifies what is meant by "pouring out" one's spirit.

#### 1:24

G: (a) ἐπειδὴ ἐχάλουν καὶ οὐχ ὑπηκούσατε (b) καὶ ἐξέτεινον λόγους καὶ οὐ προσείχετε. "(a) Then I would call but you did not listen, (b) and I would stretch out words but you did not pay attention." Since Wisdom is chastising fools for their obtuseness, G assumes that she is speaking of past events in 1:24–25. The verbs of Wisdom's actions are imperfects, to show that her efforts were ongoing.

καὶ οὐχ ὑπηκούσατε turns ותמאנו into its antonym and then negates it. Klein (1976, 515–37) and R. Gordon (1999, 3–21), writing about this device in the Targumim, call it "converse translation." Orlinksy (1958, 231–38) discusses its use in G-Job. For examples, see the comments on 9:6b; 11:7a; 11:29a; 15:22a, 32b; 17:4b; 26:20a; 30:3a; 31:30b. In 1:24b, G's substitution of λόγους for "ד" is based on the logical assumption that Wisdom works through words, not gestures.

S's ملر ("my voice") represents the same interpretation as G but expresses it independently. Since علر is not an adjustment to G, it is probably the original, with the variant ܐܝܝܕ, ("my hand") being a later adjustment to M.

# 1:27 (metath $M^Q$ ] כשאוה $M^K$ (metath או $\rightarrow$ או

The *ketiv* form is not found elsewhere and is clearly a mechanical error.

S omits 1:27aα by homoi פחדכם ∩פחדכם.

דַרְבֶּם וְּמִמֹעֲאָׁתֵיהֵם יִשְׂבֵּעוּ: 32 כֵּי מְשׁוּבַת פְּתָיֵם תַּהַרְגֵם וְשׁלְוָת כְּסִילִים תְּאַבְּדֵם: 33 וְשׁׁמֵעֵ לֵי יִשְׁכָּן־בֶּטַח וְשַׁאַנֵּן מִפִּחַד רְעֵה: פּ 1:1 בֵּנִי אִם־תַּקַח אַמְרֵי וֹמִצְוֹתֵׁי תִּצְפָּן אִתֵּד: 2 לְהַקְשִׁיב לֵחָכְמָה אָזְגֵד תַּטֶה לְבְּדְ לַתְּבוּנֵה: 3 כֵּי אֵם לַבִּינֵה תִקְרָא לַתְּבוּנָה תִּתֵּן קוֹלֶדְ: 4 אִם־תְּבַקְשֶׁנָה כַכְּסָף וְכַמַּטְמוֹנֵים תַּקרֶא לַתְּבוּנָה תַּתֵן קוֹלֶדְ: 4 אִם־תְּבַקשֶׁנָה כַבְּסָף וְכַמַּטְמוֹנֵים תַּקרְכָּא לַתְּבוּנָה תַּתֵן קוֹלֶדְ: 4 אִם־תְּבַקשֶׁנָה כַבְּסָף וְכַמַּטְמוֹנֵים תַּחְפְּשֶׁנָה: 5 אָּז תְּבִין יִרְאַת יְהוֶה וְדֻעַת אֶלֹהֵים תִּמְצָא: 6 בְּידֵיהוָה יִתֵּן חְכְמֵה מִפִיו דַעַת וּתְבוּנֵה: 7 <sup>-</sup>וְאָפַנַן לָיִשְׁרִים תּוּשִׁיֶה מְמֵצָ

In 4QProv<sup>b</sup> מושכת is clearly a mechanical error deriving in two steps from (= M).

G: (a) ἀνθ' ὦν γὰρ ἠδίχουν νηπίους, φονευθήσονται, (b) καὶ ἐξετασμὸς ἀσεβεῖς ὀλεῖ. "(a) For inasmuch as they wronged the innocent, they shall be murdered, (b) and interrogation will destroy the wicked." As in G 1:17, the translator regards naive youths—such as the son being addressed—as potentially the direct victims of the wicked. ἐξετασμός associates muth with muth cuestion," etc.) and pictures judicial interrogation. ἐξετασμός in Wis 4:6 is part of the final judgment of the sinner.

2:1 אמרי ומצותי (הבע, הבהשונה) אמרי ומצותי G (ῥῆσιν ἐμῆς ἐντολῆς) (near haplog יי יי ו)

G reads אמרי מצותי, implicitly vocalized as אָמְרֵי מִצְוְתִי הַאָּמְרֵי אָמְרֵי אָמְרֵי הַאָּמְרֵי הַאָּמְרֵי הַאָּמְרֵי הַאָּמְרֵי הַאָּמָר הַ הוּסוֹט singular = plural אמרי ווו אַמרי ווו אַמרי ווו אַמרי ווו אַמרי הי אַמרי הַ אַמרי אַמרי הַ אַמרי אַמרי הַ אַמרי אַנאַרי אַמרי אַנאַזאַ אַמרי אַנאַראַ אַרי אַמרי אַמרי אַמרי אַמרי אַנאַן אַרי אַג

2:2 (בנ  $\leftarrow$  ב M S ( $\perp \omega$  ) ( $\perp \omega$  לבנך ( $\perp \omega$   $\vee \omega$  ) (near dittog  $\perp \omega$ 

G: (a)(b) = M; (c) παραβαλεῖς δὲ αὐτὴν ἐπὶ νουθέτησιν τῷ υἰῷ σου. "(c) and you will direct it to admonition for your son." τῷ υἰῷ σου = לִבְנָף This arose by near dittography of the similar-looking ⊐ and J (for the phenomenon, see Kennedy 1928, 44). This stich has the obelus in SyrH and is OG. However, both the verb παραβαλεῖς and the direct object αὐτήν in 2:2c presuppose καρδίαν in 2:2b, and that stich too must be original. This implies that the translator is accommodating two variants he is aware of: לבך hough arising from scribal error, G's "to your son" introduces the interesting idea of the listener's gaining wisdom to teach his own son; see ABP 1.81.

G 2:2 is best read as consequent upon verse 1, since verse 2 does not start with  $\kappa\alpha i$ . In this way G turns the understanding of wisdom into a *reward* for accepting the father's precepts. (For a similar notion in Egyptian wisdom, see ABP 1.76–77.)

2:3

G: (a) ἐἀν γὰρ τὴν σοφίαν ἐπικαλέσῃ (b) καὶ τῇ συνέσει δῷς φωνήν σου, (c) τὴν δὲ αἰσθησιν ζητήσῃς μεγάλῃ τῇ φωνῇ. "(a) For if you call to wisdom, (b) and to understanding give your voice, (c) and seek knowledge in a great voice." בינה is here translated σοφία (elsewhere in the HB only in Prov 3:5). This rendering is perhaps under the influence of 1:20, where the figure that calls out is called σοφία. (a) is usually translated φρόνησις or σύνεσις.) Though 2:3c is lacking in G<sup>BS</sup> and given an asterisk (incorrectly) in SyrH, it is OG. The more literal 2:3b is Hexaplaric and missing from important MSS (Fritsch 1953, 178; CSP). G 2:3c is a second translation of 2:2b.

V has *cor tuum* ("your heart") for M G "your voice," to supplement the demand to *call* to wisdom by the injunction to *pay attention* to it as well.

T translates אימא as אימא ("mother") = אָם, though it was certainly using a vocalized Hebrew text. This is a midrashic parsing found in b. Ber. 57a and elsewhere. It is unusual for T to diverge from both M and S.

2:6 (מדט M S (מדב בהכיח ( מפניו ( מרבי המכניי) א מפניו ( מרבי מיל מיל א מפיו) ( syn)

G's "presence/face" for M's "mouth" suggests to Toy that G is avoiding the notion of verbal inspiration of wisdom. But theological inhibitions about such an idea would not have been necessary at a time when wisdom could easily be identified with Torah. מפניו both mean, approximately, "from him," and the latter could appear as an accidental synonym variant of the former. On this synonym pair, and this verse in particular, see Talmon 1975, 350 and n. 109.

**2:7 יִצְפָ**ון [M<sup>K</sup>] יִצְפָן (gram, equal)

להלבי תם M S (גרמאבי גא הסב האלט א הספומע מטידשי) להליכתם (גרמאני הסב האליכתם (נגרמי הספומע א ליני שG (דא  $\pi$ סר מטידש)

The *qere* and the *ketiv* are both future tense.

G's לְהַלִיכָתָם ("for their going") arose by graphic error and does not make good sense.

ּתְּם: 8 לְּגַצֹּר אָרְחַוֹת מִשְׁפֶּט וְדֶרֶדְ חֲסִידָו יִשְׁמְר: 9 אָׁז הֲבִיז צֶדֶק וּמִשְׁפֵּט וּמֵישָׁרִים כָּל-מַעְגַּל-טְוֹב: 10 כִּי־תָבַוֹא חָכְמָה בְלִבֶּדְ וְׁדַעַת לְנַפְשְׁדְ יִנְעֵם: 11 מֵזמָה תִּשְׁמָר עָלֶידְ הְּבוּגָה תִנְצְרֶכָּה: 12 לֱהַצִּילְד מִדֶּרֶדְ רֶע מֵׁאִישׁ מְדַבֵּר תַּהְפֵּכְוֹת: 13 הַעַּזְבִים אָרְחוֹת יֶשֶׁר לְלֶכֶת בְּדַרְכֵי־חְשֶׁדְ: 14 הֲשְׁמַחִים לַעֲשְׂוֹת רֶע יְגִילוּ בְּתַהְפָּכִוֹת רֵע: 15 אֲשֶׁר אָרְחֹתִיהֶם עִקְשֵׁים וֹּנְלוֹזִים בְּמַעְגְלוֹתֶם: 16 לֱהַצִּילְד מַאִשֶׁה זֶרֶה מִנְּכְרִיָּה אֲמָרֶיהָ הֶחֶלֵיקָה: 17 הַעָּזֶבָת אַלְוּף נְעוּרֶיהָ

2:8 חסידו M<sup>K</sup> ] חַסִידֵיו M<sup>Q</sup> G (εὐλαβουμένων ἀὐτόν) S (געמעהסת,) ≈ V (sanctorum) ≈ T (דצריקין) (orth)

The plural, indicated by the *qere*, is correct. Since the singular אסיד does not take first- or third-person singular suffixes, the *ketiv* is a *defectiva* writing of the plural. V and T<sup>L</sup> do not represent the possessive suffix.

Singular/plural *ketiv-qere* variants, and ostensive singular/plural *ketiv-qere* variants, are frequent, and it is usually difficult to give preference to one or the other. Gordis has a list of such *ketiv-qere* pairs (1971, 86–92). Usually the *qere* is a *plene* guide to reading the *defectiva ketiv*, as here. See §2.2.3.2, n. 21.

2:11

G: (a) βουλὴ καλὴ φυλάξει σε, (b) ἔννοια δὲ ὁσία τηρήσει σε. "(a) Good counsel will guard you, (b) and pious understanding will keep you." G makes it clear that only good counsel and pious understanding will protect you (and not just מומה, which often refers to a morally dubious shrewdness). The faculties mentioned in the Hebrew, both מומה מומה, can be misused and are not inherently virtuous. (Examples of the former are Ps 10:22; 21:12; of the latter, Ezek 28:4.) See ABP 1.34, 37–38.

#### 2:15

G's ai דרסעומו מידשי ("their courses") omits the preposition of במעגלותם to tighten the parallelism. V, S, and (consequently) T follow suit.

#### 2:16-20

G replaces M's warning against the "Strange Woman"—that is to say, the adulteress (see ABP 1.139-41)—with one against "evil counsel,"

mentioned in G 2:17. The "evil counsel" (κακή βουλή) in 2:17 is neither Hellenistic wisdom (pace Hengel 1974, 1:155; CSP 138) nor the rabbinic concept of the evil inclination (as argued by CSP 136-37). When the passage is read in the context of G-Prov 2 as a whole, it becomes clear that the evil counsel is the urgings and insinuations of the evil man, who "speaks nothing reliable" (2:12) and whose influence would dislodge the naive from the right path (2:16). "Evil counsel" comprises everything that such people, male and female, encourage susceptible young men to do. An example is quoted in 1:11-14. This is the evil counsel that is incidentally personified in G 2:18. Such blandishments can be withstood only with the support of the βουλή καλή "good counsel" (2:11), which is the wisdom that comes from teachers and God. Given the conceptual reshaping of this passage and the syntactic reformulations (notably, the translator overriding the second-person plural setting of the passage), Hebrew variants are not retrievable, except for those in 2:18 and, perhaps, 2:19.

Since in 5:3; 6:24–25; and 7:6–17 G recognizes the Strange Woman as an actual woman, he could have read the present passage similarly. The decision to view the woman of 2:16–20 as an allegory for all illicit counsel was a deliberate reinterpretation, in the midrashic vein.

2:16

G: (a) τοῦ μακράν σε ποιῆσαι ἀπὸ ὁδοῦ εὐθείας (b) καὶ ἀλλότριον τῆς δικαίας γνώμης. "(a) So as to make you far from the straight way (b) and alienated from the righteous judgment." Thinking that the agent of was the evildoers, the translator found it necessary to rewrite this and the next verse (BHQ).

#### 2:17

G: (a) υἰέ, μή σε καταλάβῃ κακὴ βουλὴ (b) ἡ ἀπολείπουσα διδασκαλίαν νεότητος, (c) καὶ διαθήκην θείαν ἐπιλελησμένῃ. "(a) My son, let not evil counsel capture you, (b) (the counsel) which abandons the teaching of youth (c) and forgets the godly covenant." G starts the new section by adding a sentence (G 2:17a), cf. 8:21a. G begins the new line with υίέ, assimilating it to the customary M usage. This happens also in 9:12; 19:20; 23:22; 24:1; and 27:27. G associates <code>ψ</code> with <code>ψ</code> with <code>ψ</code> "tearn," "teach."

S's אלוף ("foster-father") also associates אלוף with teaching, perhaps influenced by G. T<sup>Z</sup> מרביתא ("training"). T<sup>L</sup> מרביינא ("trainer"), from S. 2:17-3:4

וְאֶת־בְּרָית אֶלֹהֵיהָ שָׁבְחָה: 18 כִּי שָׁחָה אֶל־מָוֶת <sup>ר</sup>וְתִיבְתָה<sup>¬</sup> וְאֶל־ רְפָאִים מַעְגְלֹתֵיהָ: 19 כִּל־בָּאֶיהָ לָא יְשׁוּבְוּן וְלָא־יַשִׁיגוּ אָרְחוֹת חַיִּים: 20 לְמַעַן הֵלֵך בְּדֶרֶד טוֹבִים וְאָרְחוֹת צַדִּיקִים תִּשְׁמְר: 21 כְּי־יְשָׁרִים יִשְׁפְנוּ־אֶרֶץ וּתְמִימִים יוֶתְרוּ בֵה: 22 וְרְשָׁעִים מֵאֶרֶץ יִבָּרֵתוּ וּבוֹגְדִים יֹשְׁפְנוּ־אֶרֶץ וֹתְמִימִים יוֶתְרוּ בֵה: 22 וְרְשָׁעִים מֵאֶרֶץ יִבָּרֵתוּ וּבוֹגְדִים יֹשְׁפְנוּ־אֶרֶץ וֹתְמִימִים יוֶתְרוּ בֵה: 22 וְרְשָׁעִים מֵאֶרֶץ יִבָּרֵתוּ וֹבוֹגְדִים יֹשְׁפְנוּ־מֶמֶנָה: פּ זוֹמִינִים וְאָרָחוֹת חַיֵּים מְאָרָץ יִבָּרָמוּ וֹמִצוּ לְבֶּדְּ: 2 כִּי אַׁרָך יָמִים וּשְׁנִוֹת חַיִים וְשָׁלוֹם יוֹסִיפּוּ לֶדְ: 3 (תֶסֶד וָאֲמֶת אַל־יַעַׁזְבֵדָ) קַשְׁרֵם עַל־גַּרְגִרוֹתֵיךּ כָּתְבָם עַל־לְנוּח לְבֵּדָ: 4 וּמִצָּארָחֵן

2:18 שחה M ] שתה G (ਫੱθετο) σ' (ωρισεν) (graph  $\rightarrow$   $\rightarrow$  )

<br/>- ביתה [ <נתיבתה MG (דיא מטא מטל ק<br/>ל) א (near haplog ביתה נת metath ביתה) (near haplog בי $\to$ יב

M makes better sense if we emend נתיבתה to נתיבתה ("her path") (BHS), which provides a feminine singular subject for שחה. It is the path to the woman's house (rather than the house itself) that is heading down to Sheol.

τῷ θανάτῷ τὸν οἶκον αὐτῆς, (b) καὶ παρὰ τῷ ặδῃ μετὰ τῶν γηγενῶν τοὺς ἄξονας αὐτῆς. "(a) For she has placed her house beside death, (b) her courses beside Hades with the earthborn." G reads אל courses beside Hades with the earthborn." G reads אל courses beside Hades with the earthborn." G reads אל, a graphic error facilitated by the difficulty of M's image of the house sinking down to death. However, שית would not govern אל in this sense. (In Job 7:17 has its expected meaning, "toward.") In 2:18b, καὶ παρὰ τῷ ặδῃ is OG; μετὰ τῶν γηγενῶν is from θ' and εβρ'.

S אבחה, which BHQ says comes "via" אכחה, is actually an inner-S scribal error. The word was taken accidently from S 2:17 (Toy). אימורא ("thresholds") represents אלמות אלמות, associated with "porch" (BHQ).

ארחות חיים  $M \approx S$  (הועד העיבא M  $\approx S$  (אונע היים  $M \approx G$  (ליומטדשא לשקר) (assim)

Neither G 2:19b (with τρίβους εὐθείας "straight paths") nor the doublet in 2:19c (with ἐνιαυτῶν ζωῆς "years of life") translates M's "ways of life." De Lagarde considers 2:19c original. Perhaps G is conflating two variants, שנות חיים In G 4:10c (= OG), "ways of life"  $\approx$  M "years of life." Both 2:19c and 4:10c are rather awkward and unlikely to have been the translator's innovation. In both cases, the changes prob-

ably occurred in Hebrew copying and were influenced by the same phrases elsewhere (שנות חיים in 3:2; 4:10; and 9:11; ארח חיים and the like in 2:19; 5:6; 10:17; 15:24).

#### 2:21

G 2:21ab, though absent in G<sup>BMSS</sup>, are OG (CSP 144–45), while 2:21cd are Hexaplaric.

# 2:22 יָסָחוֹ CG, Bab MS Ba ] יָסָחוֹ M (vocal, gram)

Instead of ווסי, a CG MS and one with Babylonian vocalization (Ba = Berl. Or. 680; cf. BHK, BHQ) vocalize יסחו (or יסחו), a G-stem passive. BHQ weighs the possibility (likely, I think) that this is the earlier vocalization. With the disappearance of the G-passive, יסחו was vocalized as an active G-stem, understood as an impersonal plural. The unvocalized form was necessarily treated as a passive in the versions.

#### 3:1

For M's ומצות' (= S האסמוע, G has tà dè אמרי ("my words"). Elsewhere in Proverbs (4:5; 7:24; 8:8; 17:27), אמרי(ם) אמרי(ם). Nowhere else does אמרי(ם). Still, there is much variation in the treatment of words for "words," "commandments," and "teachings," since they are functional synonyms, and variations can occur in Hebrew or in translation.

#### 3:2

S אלאסשים ("will be added to you") may reflect consonantal אוסשים ("will be understood as an H-passive (for M's יָסִיפוֹ "they shall add"), but S may simply be compensating for the lack of an explicit subject.

# 3:3 חסד ... יעזבך M G S (add)

בתבם על־לוח לבך MS (הבאהב אנה בא להשיים) ]>G (homoi [ר~ך]

In M, the first line, 3:3a, "Let not kindness and faithfulness abandon you," is probably a gloss on the preceding verse (thus Meinhold; Plöger; ABP). The line intrudes between the pronoun "them" in 3:3b and its antecedents ("teaching" and "precepts") in 3:1–2. Also, the word ¬DOT ("kindness") appears only here in Prov 1–9, suggesting that it is not native to the usages of this unit. While thematically pertinent in a general way, 3:3a adds a message intrusive at this point, namely, that you should regard the

ןשֵׂכָל־עִוֹב בְּעֵיגֵי אֶלֹהֵים וְאָדֵם: פּ 5 בְּטַח אֶל־יֱהוָה בְּכָל־לִבֶּדְ וְאֶל־בְּינְתְדָ אַל־תִּשְׁעֵן: 6 בְּכָל־דְרָכֵידְ דָעֵהוּ וְהוּא יְיַשִׁר אִּרְחֹתֵידִ: 7 אַל־תִּהֵי חָבֶם בְּעֵינֵידְ יְרָא אֶת־יְהוָה וְסוּר מֵרֶע: 8 רָפְאוּת תִּהֵי <sup>ר</sup>לְשֵׁרֶדּ<sup>י</sup> וְ</sup>שִׁקוּי לְעַצְמוֹתֵידָ: 9 בַּבַּד אֶת־יֵהוָה מֵהוֹנֶדְ זֹמֵראשִׁית כְּל־תִבוּאָתֶדָ: 10 וִימָּלְאַוּ אֲסָמֵידָ שָׁבֵע וְתִירוֹש יְקָבֶידָ זִפְרְצוּ: פּ 11 מוּסַר יֵהוָה בְּנֵי אַל־תִּמְאֶס וְאַל־תְׁקֹץ בְּתוֹכַחְתִּוֹ: 12 כִּי אֶת אֲשֶׁר יֶאֶהַב יְהוֶה יוֹכֵיח וֹּכְאָב אֶת־בֵּן יִרְצֶה:

blessings that come with piety as the bounty of God's grace rather than as simply your own deserts. (This is the only viable interpretation of this line. The הסד in this line can only be God's kindness toward the pupil, not the pupil's toward others, since הסד is always conferred by a superior on an inferior [in status or power]; see ABP 1.145. In any case, a person's own "kindness" and "faithfulness" would not be said to "abandon" him. The failure would be for *him* to abandon *them*.)

G: (a) ἐλεημοσύναι καὶ πίστεις μὴ ἐκλιπέτωσάν σε, (b) ἄφαψαι δὲ αὐτὰς ἐπὶ σῷ τραχήλῳ, καὶ εὑρήσεις χάριν. "(a) Let mercies and faithfulness not abandon you. Bind them on your neck, and you will find favor." OG (G<sup>BS\*</sup>) lacks 3:3c. The line was supplied (from θ', according to SyrH) in various forms and positions in G<sup>AMSS</sup>. Since the third stich is entirely germane and needed for the parallelism, the omission was almost certainly accidental, arising from a homoioteleuton from ¬ at the end of 3:3b to ¬ at the end of 3:3c. OG (G<sup>B</sup>) joins 3:4a to 3:3c.

S: (a) جامعت حل حامعتمام محامع "And peace and blessing and truth will not abandon you." S makes the first clause into an indicative statement of promise and adds another term to the two blessings in M: "kindness" and "faithfulness." 3:3bc = M.

3:6 (הד א S (הד הדעהו [ הביסה, M S (הד הדעהו [ (הד לעמט מט מיל היש ה) + 1 stichos G (later addition)

G: (a) ἐν πάσαις ὁδοῖς σου γνώριζε αὐτήν, (b) ἵνα ὀρθοτομῆ τὰς ὁδούς σου, [(c) ὁ δὲ πούς σου οὐ μὴ προσκόπτη]. "(a) In all your ways know her, (b) that she may make your ways straight, [(c) and your foot not stumble]." G probably had הדעהו (= הדעהו), because it uses γνώριζειν only for the H- and A-stems of ידע (or a synonym), never for the G-stem, and there would be no reason for דעהו

understood the verb in 3:6a as "make known" rather than "know," he took the direct object to be *wisdom* (hence the feminine αὐτήν). The result, "In all your ways, declare [or 'teach'] it," accords with G's assumption that the wisdom mentioned in 3:5 is of the virtuous sort. (G has "do not exalt yourself for your wisdom," rather than M's "Do not rely on your wisdom." The translator assumes that though one *should* rely on his wisdom he should not *boast* about it.).

Some G manuscripts add the thematically related line from 3:23b,  $\delta \delta \hbar \pi o \upsilon c \sigma \upsilon v \mu \eta \pi \rho \sigma \kappa \delta \pi \tau \eta$  (asterisked in SyrH; omitted in G<sup>BSA</sup>). This example is significant because it shows that the transference of clauses and lines from elsewhere in the book is a reality of scribal practice in the Greek tradition and need not be ascribed to the translator.

M's "healing ... for your navel" does not make good sense. We should follow G and S in reading a word for "your flesh." The minimal change would require vocalizing אירך = לְשָׁרָך שִׁרָך שׁׁרָר לבשרך אולש with the א nomenon see GKC §23f ). Also possible are לבשרך or לבשרך G 3:22a, which is based on this verse, refers to flesh and indicates that the author of that verse saw one of the synomyms for flesh here.

#### 3:9

G: (a) τίμα τὸν κύριον ἀπὸ σῶν δικαίων πόνων (b) καὶ ἀπάρχου αὐτῷ ἀπὸ σῶν καρπῶν δικαιοσύνης. "(a) Honor the Lord from your honest labors (b) and offer to him from your fruits of righteousness." By adding "honest" and "of righteousness," G makes it explicit that one should make sacrificial donations only from *honest* earnings. καρπὸς δικαιοσύνης appears in 11:30 and 13:2 as well. "All" is omitted in G 3:9b.

#### 3:10

#### 3:12

Where M has וּכְאָב ("as a father"), G's μαστιγοῖ δέ ("and afflicts") construes the word as וְיָבָאָב or perhaps וְכֵאָב, though the D-stem is not 13 אַשְׁרֵי אֲדָם מְצָא חָכְמֵה וְאָדָם יִפֵּיק תְּבוּגֵה: 14 כֵּי טִוֹב סֵחְרָה מִסְחַר־כֵּסֶף וֹּמַחְרוּץ תְּבוּאָתֵה: 15 יְקָרָה הֵיא <sup>ה</sup>מִפְּנִיגֵים<sup>י</sup> וְכָל־ <sup>ה</sup>ְחַפָּצִים<sup>י</sup> לָא יְשְׁוּ-בְה: 16 אַרֶך יְמִים בְּימִיגֵה בִּשְׂמאוֹלָה עַשֶׁר וְכָבְוֹד: 17 דְּרָכֵיהָ דַרְכֵי־גְעַם וְכָל־נְתָיבוֹתֵיהָ שָׁלְוֹם: 18 עֵץ־חַיֵּים הֵיא לַמַּחַזִיקִים בֶּה וְתֹמְכֵיהָ מְאָשֵׁר: פּ 19 יְהוָה בְּחָכְמָה הֵיא לַמַּחַזִיקִים בָּה וְתֹמְכֵיהָ מְאָשֵׁר: פּ

actually used in Hebrew (it is in Aramaic). An *'al tiqre'* interpretation in Mid. Ps. 94:2 says, אל תקרא וכאב אלא וכאב ("Do not read וְכָאָב hut rather אל תקרא וכאב). This shows that G's construal does not require consonantal alteration, so G is not evidence for ויכאָב אוי וויכאָב (contra BHS). Perhaps the *defectiva* spelling was used to allow for a visual pun. G adds πάντα, as often, for emphasis.

S: (a) = M; (b) איבא גוגא גבוס יישר "and like a father who punishes his son." S understands ירצה as "punishes," perhaps influenced by the phonemic similarity of ארצה and possibly aware of the midrashic interpretation mentioned above. T follows S.

## 3:15 מפנינים [ M<sup>Q</sup> ] מפניים ( homoi י∩י)

מפנינים ] + 2 stichoi G (15bc)

הפצים הוועט א G (דועוטי) א א (quae desiderantur) T (כותעם) א [ הַפָּצָיד ] הַפָּצָיד M (graph  $\Sigma \to \Sigma$ 

G: (a) τιμιωτέρα δέ ἐστιν λίθων πολυτελῶν, (b) οὐκ ἀντιτάξεται αὐτῆ οὐδὲν πονηρόν. (c) εὔγνωστός ἐστιν πᾶσιν τοῖς ἐγγίζουσιν αὐτῆ, (d) πᾶν δὲ τίμιον οὐκ ἄξιον αὐτῆς ἐστιν. "(a) She is more precious than costly stones. (b) No evil can oppose her. (c) She is familiar to all who approach her. (d) No valuable equals her worth." Stichoi (a) and (d) adjust to M, except that stich d witnesses to תפצים. Stichoi (b) and (c), which diverge sharply from M, have the obelus in SyrH and are OG. They cannot be reconciled to M, and there is no evident motive for the translator to have composed them. M's sentence was translated in the expected fashion in 8:11b and would have presented no problem here. De Lagarde (following Grabe) emends πονηρόν to ποθητόν ("longed for"), but the latter never corresponds to  $\gamma$ D. Moreover, ἀντιτάσσεσθαι ("resist," "oppose") usually connotes hostility (cf. 3:34), making πονηρόν a suitable subject.

It is unclear if G 15bc arose in Hebrew or Greek. For the sentiment, see Sir 6:18–22, a passage influenced by Prov 3:13–18. Ben Sira advises the reader to draw close (קרב) to Wisdom, for she yields her bounty to those

who accept her discipline but is troublesome to the fool. Prov 3:15c is partly echoed in Sir 4:24a, כי באומר נודעת חכמה, translated לי אָאָ אַלאָ אָלאָשָ אָאַאָר אָלאָשָ אָאַאָר גודעת די אָאַר אָלאָשָ אָאָאָאָ אָאָאָ

Where M has הפצים, all the versions reflect הפצים, the form used also in 8:11. S's כבוכ ("anything") witnesses to הפצים, independently of G. V's *quae desiderantur* ("which are desired") = הפצים, understood as "desired things." (הפצים could have been rendered "which you desire"; cf. Isa 58:13.) The suffix of M's הפציך is the only second-person singular in this literary unit (3:13–18 = Interlude B in my terminology; see §1.1; footnote 1). In chapter 8 (Interlude D), which is based on the present unit, the addressees are "people" (8:4) or "sons" (8:32), addressed in the plural, not an individual student.

# **3:16** fin ] + 2 stichoi G (16*a*) (elab)

G: (a) μῆκος γὰρ βίου καὶ ἔτη ζωῆς ἐν τῆ δεξιặ αὐτῆς, (b) ἐν δὲ τῆ ἀριστερặ αὐτῆς πλοῦτος καὶ δόξα. "(a) For length of living and years of life are in her right hand, (b) and in her left are wealth and honor." To "length of life," G adds καὶ ἔτη ζωῆς, a phrase imported from 3:2 in Hebrew or Greek. According to Gerleman (1956, 25), this was intended to make the line "materially conformable to" (i.e., quantitatively balanced with) its parallel (1956, 25, comparing 3:16; 17:15; 23:21). But since many additions produce imbalance (e.g., 3:23a; 5:22b; 10:22b; 12:4b), the occasional creation of balance can be fortuitous.

#### 3:16a

G: (a) ἐκ τοῦ στόματος αὐτῆς ἐκπορεύεται δικαιοσύνη, (b) νόμον δὲ καὶ ἔλεον ἐπὶ γλώσσης φορεῖ. "(a) From her mouth proceeds righteousness, (b) and she bears mercy on her tongue." G 3:16*a*a resembles Isa 45:23a, and 16*a*b is based on Prov 31:26b[25b]. Since in both cases the affinity is to the Hebrew rather than the Greek, the couplet was probably present in the source text (Tov 1990b, 49). Tov retroverts the couplet to מפיה געא צדקה ותורת חסד על לשונה the practical benefits noted in 3:16.

3:18

G: (a) ξύλον ζωῆς ἐστι πᾶσι τοῖς ἀντεχομένοις αὐτῆς, (b) καὶ τοῖς ἐπερειδομένοις ἐπ' αὐτὴν ὡς ἐπὶ κύριον ἀσφαλής. "(a) It is a tree of life for them who cleave to it, (b) and for those who lean on her, (she is) as firm as [leaning] upon the Lord." G enhances the religious message by asserting that to embrace wisdom is to embrace God.

יֶסַד־אָּרֶץ כּוֹגֵן שְׁמִׁיִם בִּתְבוּגַה: 20 בְּדַעְתּוֹ תְּהוֹמִוֹת נִבְקֶעוּ וּשְׁחָלִים יִרְעַפּוּ־טֵל: 21 בְּנִי אַל־יָלָזוּ מֵעֵינֵיֶדְ נְצְׂר תִּאִיָּה וּמְזמֵה: 22 וְיְהְיַוּ חַיִּיִם לְנַפְשֶׁדְ וְׁחֵן לְגַרְגְּרֹתֶידְּ: 23 אָז תַּלֵדְ לָבֶטַח דַּרְכֶּדְ וְרַגְלְדְ לָא תִגְּוֹף: 24 אָם־<sup>ר</sup>תֵּשֵׁב<sup>ּי</sup> לָא־תִפְחֵד וְשָׁרַבְהָ ווְעָרְבֶה שְׁנָתֶדָ: : 25 אַל־ תִּגְוֹף: 24 אָם־<sup>ר</sup>תֵּשֵׁב<sup>י</sup> לָא־תִפְחֵד וְשָׁרַבְהָ ווְעָרְבֶה שְׁנָתֶדָ: 25 אַלי תִּגִּוֹף: מִנִי מִפַּחַד פִּתְאֵם וּמִשׁׁאַת וְלִשָּׁעִים כֵּי תָבְאַ: 26 כֵּי־יֶהוָה יִהְיֶה

As for ἐπερειδομένοις, G-Proverbs assumes that πας "support," hence "make/be firm" (as in RH). G is not familiar with the sense "grasp," which is required in Prov 3:18 (ἐπερειδομένοι); 4:4 (ἐρειδέτω); 5:5 (ἐρείδεται), 22 (N-stem; translated σφίγγεται, as required by context); 11:16a (emend to ἐρειδει; see the comment), 16b (ἐρείδονται; in G 16d); 29:23 (ἐρείδει); and 31:19 (ἐρείδει). πας "support" in 28:17 and is translated, relevantly, ἐν ἀσφαλεία. G's limited understanding of the meaning of Tας

De Lagarde believes that שָׁכָ לאוֹ עטוּקטֹא reflects a dittography, ותמכיה, but the unique rendering of מאשר as "firm" (as in MH מאושר "strong") shows that the translator is deliberately importing a religious theme.

In S 3:18b, גרשמרבי, ("who hope") should be emended to (Dp-stem, "support"), as in 5:5 (Pinkuss). The plural (= T) belongs to the Syriac idiom, and it is not evidence for מאשרים, as BHS implies.

3:20 אירעפו M G (<br/>ל $po' \eta \sigma \alpha \nu )$  S ( $\infty i) ] אירעפו M^{\rm MSS} (metath <math display="inline">\to \to \to )$ 

The metathesis found in some MSS KR seems to have been ancient; cf. Sir 43:22, מערף ענן טל. But only M's "drip" is meaningful here.

3:21 איזלו [ M ילזו G (παραρρυῆς) (metath לז → לז)

G translates the difficult ילזו as "(do not) slip aside" (= יזלו) (thus K<sup>95pm</sup>), from נזל ("flow") (cf. Baumgartner). G is unlikely to be seeing and construing it as "flow" because in the nearly identical 4:21 יליזו is translated ἐκλίπωσιν, which would fit in the present verse, too.

S has  $\Delta u$  ("be trivial"); sim T ניזלן. Since S does the same in 4:21, this is probably an exceptical association with  $\Delta u$  "be trivial," rather than a graphic metathesis. It may be a case of "exceptical metathesis"; on the phenomenon, see de Waard 1993, but also the comment on 11:27.

**3:22** fin M ] + 2 stichoi G (22*a*) (assim to 3:8)

# 3:22*a* רפאות תהי לשרך ושקוי לעצמותיך G

G: (a) ἔσται δὲ ἴασις ταῖς σαρξί σου, (b) καὶ ἐπιμέλεια τοῖς σοῖς ὀστέοις. "(a) And it will be healing for your flesh, (b) and care for your bones" (continues in 3:23). This translates 3:8 but in a wording different from the Greek of that verse, which is τότε ἴασις ἔσται τῷ σώματί σου καὶ ἐπιμέλεια τοῖς ὀστέοις σου. If the translator wished to copy G 3:8, he would have reused the wording he had just composed, whereas in translating he might render the same Hebrew with slight differences. Hence this expansion (which embellishes the statement of wisdom's benefits in 3:22) was probably in G's Hebrew source text (Tov 1990b, 50), probably in the wording of M 3:8. Like 3:8, 3:22*a* also reflects "your flesh."

#### 3:23

G's πεποιθώς ("being confident") and ἐν εἰρήνῃ ("in peace") are a doublet corresponding to M's לבטח. The former is closer to M and presumably later, but the latter is lacking in SyrH (בה אבע אונא) and G<sup>V</sup>.

3:24 אָזשב 3:24 (אנαθη) SyrH (אות) אותשב G (אמפן) אוגבא (near dittog ב → (כב

G: (a) ἐἀν γὰρ κάθῃ, ἄφοβος ἔσῃ, (b) ἐἀν δὲ καθεύδῃς, ἡδέως ὑπνώσεις. "(a) For if you sit, you will be without fear, (b) and if you lie down, you will sleep pleasantly."

M's "(when) you lie down" in both stichoi is pointlessly repetitious. G's "(when) you sit" fits into a sequence of actions that represent the totality of a day's activities: walking (3:23), sitting down (3:24a), going to sleep (3:24b).

S turns the conditional into an indicative, "You will sleep (גאגבא), etc." S reflects the repeated תשכב, as in M, but creates variation by rendering it differently in each stich.

3:26 בבסלך M ~ S (ح<br/>  $M \approx$  S (ح<br/>  $M \approx$  G (<br/>לתן המסגיי לא מעס השט שי האר) (graph  $\rightarrow$  ה)

G: (a) ό γὰρ κύριος ἔσται ἐπὶ πασῶν ὁδῶν σου, (b) καὶ ἐρείσει σὸν πόδα, ἕνα μὴ σαλευθῆς. "(a) For the Lord will be on all your paths, (b) and will keep your foot firm, so that you not be shaken." For ἐρείσει, de Lagarde prefers τηρήσει (G<sup>297</sup> ≈ V), which, however, looks like a harmonization.

G has בְּמְסִלֹתָד ("in your paths"); thus Toy. אָמָסלֹת ("in Judg 20:31, 32, 45. במסלה are very similar in several scripts, early and late. More

בְכִסְלֶהְ וְשָׁמֵר רַגְלְהְ מִלְּכָד: 27 אַל־תִּמְנַע־טָוֹב מִבְּעָלֶיו בִּהְיוֹת לְאָל יּיְדְהָ' לַעֲשְׂוֹת: 28 אַל־תּׁאמָר <sup>ו</sup>לְרֵעֲהָ' לֵהְ וָשׁוּב וּמָחָר אָתֵׁן וְיֵשׁ אִתְּדָ: 29 אַל־תַּחֲרָשׁ עַל־רַעֲהָ רָעֶה וְהוּא־יוֹשֵׁב לָבֶטַח אִתְּדָ: 30 אַל־<sup>ו</sup>תָרִיב' עִם־אָדָם חִנֶּם אִם־לְא גְמָלְדְ רָעֶה: 31 אַל־תֵּקַנֵּא בְּאֵישׁ חָמֶס וְאַל־תִׁבְתַר בְּכָל־דְרָבֶיו: 32 כֵּי תוֹעֲבַת יְהוֶה נָלֵוֹז וֵאֶת־

problematic is the + ת. Perhaps במסלתך had behind it the lexical variant ובסלתך in Job 4:6).

G uses σαλεύειν for לכד only here, but proposed corrections of the Greek (see de Lagarde; Baumgartner) are graphically distant. G 3:26b is explicating the metaphor of keeping the foot from being trapped.

S בכסלך indirectly attests to M by paraphrasing בכסלך as "with you," associating כסל with the homograph that means "side" ("at your side" = "with you"). Thus V *in latere tuo*.

3:27 אדך M<sup>Q</sup> G (ή χείρ σου) Τ (בידך [בידך] א<sup>K</sup> S (היהרא איד) (num)

The *qere* is preferable because the expression לאל ידך consistently uses the singular of "hand" in the main Masoretic tradition (Gen 31:29; Deut 28:32; Mic 2:1; Neh 5:5).

G: (a) μὴ ἀπόσχῃ εὖ ποιεῖν ἐνδεῆ, (b) ἡνίκα ἂν ἔχῃ ἡ χείρ σου βοηθεῖν. "(a) Do not refrain from doing good to the needy, (b) whenever your hand may be able to help." G's phrasing, especially in translating לבעליו ("its possessor") as ἐνδεῆ ("needy"), turns the proverb into an exhortation to almsgiving.

S (T) omits מבעליו perhaps puzzled as to how one could do good to its "owner." (The Hebrew actually means "one to whom it [a benefit] is due"; see ABP.)

3:28 לרעך M<sup>Q</sup> ] אָרֵעֶידְ M<sup>K</sup> (norm)

fin ] + 1 stich G (28c) (elab = 27:1b)

The *ketiv* is a *plene* writing of a singular noun, namely, רְשֶה (Gordis 1971, 180 n. 219; GKC §93*ll*). (The singular is required by the following verse.) The *qere* derives the word from the more common רָשָ, cf. the *ketiv-qere* pair in Prov 27:10, where also רֵשָ is the *qere*.

G: (a) μὴ εἴπης Ἐπανελθών ἐπάνηκε καὶ αὔριον δώσω, (b) δυνατοῦ σου ὄντος εὖ ποιεῖν, (c) οὐ γὰρ οἶδας τί τέξεται ἡ ἐπιοῦσα. "(a) Do not say, 'Go

away and return and tomorrow I will give, (b) when you are able to do (it), (c) for you do not know what the morrow will bring forth."

G lacks a word for לרעך, for unclear reasons (BHQ suggests homoioarkton). G adds a third stich, 3:28c. This is the equivalent of M 27:1. Since the Greek of 27:1 is different (γινώσχεις for οἶδας), 3:28c was not taken from there but was present in G's source text and translated separately. Judging by M 27:1, G's source text would have had, approximately, גי לא תדע מה ילד יום. G 3:28c is a deliberate scribal expansion that derives from an ancient wisdom tradition. The logical association between doing good to other people (3:28ab) and ignorance of the morrow (3:28c) is not self-evident. (Toy calls 3:28c "a not very appropriate gloss, taken from 27:1.") Yet the topos is used often in Egyptian wisdom, especially to bolster advice to treat others well, on the grounds that one does not know what will happen in his own future. See the instructions of Anii (21.4-10; cf. AEL 2.142); Ptahhotep (lines 339-50, especially 343, 345; cf. AEL 1.69); Amenemope (§18 = 19.11–15; cf. AEL 2.157); O. Petrie 11 (vso. 5); and Pap. Ramesseum I (A 18). (For Egyptological bibliography and discussion of sources, see ABP 1.19-23; 1.427-28.)

3:30 תרוב  $M^{\mathbb{Q}}$  תרוב  $M^{\mathbb{K}}$  G ( $\phi$ וא געלף איז א  $M^{\mathbb{Q}}$  (sraph '  $\rightarrow$  1) (graph '  $\rightarrow$  1)

G makes 3:30b a warning: "lest  $(\mu \eta)$  he do some harm to you," rather than M's "if he has not done you harm," so as not to seem to tolerate revenge.

S omits 3:30b to the same effect (Pinkuss).

#### 3:31

G: (a) μὴ κτήσῃ κακῶν ἀνδρῶν ὀνείδῃ, (b) μŋδὲ ζηλώσῃς τὰς ὁδοὺς αὐτῶν. "(a) Do not aquire the disgrace of evil men, (b) and do not envy their ways." G derives 𝑘µμ ("acquire"). (Compare how G equates may and translates it θρόνος in Prov 12:23.) G adds κακῶν to "men" for moral emphasis (or, to the same effect, double-translates to "men" for moral emphasis (or, to the same effect, double-translates may. G has ζηλώσῃς. G uses ζηλοῦν for אשר (D-stem) in Prov 4:14b and הנחר 6:6b. Hence the choice of ζηλώσῃς is probably a deliberate translation of μαπι , α way of warning against the sin of envy. It does not represent a variant, παπις ).

#### 3:32

G: (a) ἀκάθαρτος γὰρ ἕναντι κυρίου πᾶς παράνομος, (b) ἐν δὲ δικαίοις οὐ συνεδριάζει. "(a) Impure before the Lord is every transgressor, (b) and he

יְשָׁרִים סוֹדְוֹ: 33 מְאֵרַת יְהוָה בְּבֵית רָשֶׁע וּנְוָה צַדִּיקֵים יְבָרֵדְ: 34 אִם־לַלֵּצִים הְוּא־יָלֵיץ 'וְלַעֲנִיִּים' יִשֶּן־חֵן: 35 בָּבוֹד חַכָמִים יִנְחֵלוּ זֹכְסִילִים 'מְרִימִים' קָלְוֹן: וֹהַקִשִׁיבוּ לָדַעַת בִּינֶה: 2 כִּי לֶכֵח טוֹב נְתַתִּי לָכֶם תְּוֹרָתִי אֵל־תַּעֲזְבוּ: 3 כִּי־בֵן הָיֵיתִי לְאָבֵי רֵדְ וְיָהִיד לִפְנֵי אִמֵי: 4 וַיֹּבֹנִי וַיָּאֹמֶר לִי יְתְמָדְ־

will not sit among the righteous." Having misidentified the antecedent of the suffix of "his counsel" as the transgressor, the translator added the negative to make sense of the sentence (cf. 1:17). Less likely is de Lagarde's surmise (following Grabe) that où is a dittography of the following  $\sigma v$ , an event unlikely in uncials. ἀxάθαρτος for  $\pi u$  is a peculiarity of G-Proverbs. In translating תועבת יהוה, G-Proverbs often but inconsistently prefers the Aramaic-sounding circumlocution "before the Lord," using ἐναντι (3:32), ἐνώπιον (11:1; 20:10), or παρά (16:5; 17:15); cf. 8:7. This usage reduces anthropopathism.

3:34 ולענוני) M<sup>K</sup> [ אולענוים (M<sup>Q</sup> V (*et mansuetis*) T<sup>L</sup> (ולענוני) ] (syn, equal) Hebrew ענוים ("humble") and עניים (usually "poor") are very similar graphically and not always distinguishable semantically. The pair occurs in the Bible eight times, with each member serving sometimes as the gere, sometimes as the ketiv (see Gordis 1971, 121). Note especially that in Ps 9, ענוים is the gere in verse 13, while ענוים is the gere in verse 19. In Proverbs, the pair occurs in Prov 3:34; 14:21; and 16:19, always with as the gere, regardless of the sense required by context. Neither ענוים form can be explained as dialectal preference, modernization, or normalization. Possibly a combination of graphic and semantic similarities gave rise to scribal variants in both directions. ענוים occurs in Proverbs only in this ketiv-qere pair. עני, which always means "poor," occurs in 15:15 (ὀφθαλμός err); 22:22 (πτωχός); 30:14 (τοὺς ταπεινούς); 31:5 (τούς ἀσθενεῖς), 9 (πένητα), 20 (πένητι). The plural עניים occurs in 30:14 (ταπεινούς) with no gere. In Proverbs, με can mean "poor" (as is clear in 14:21) and "humble" (as is clear in 3:34 and 16:19). Apparently, yet could also mean "humble," as in Isa 66:2; see HALOT 856ab. V's mites and mansuetis as well as T's ענוים witness to ענוים, at least in Proverbs. G does not distinguish the two words.

G, S, T, and V ignore the difficult particle אם, on which see ABP. Instead of the undignified image of God "mocking" (יליץ) humans, G

says that God מידודמֹסדּדמו "opposes" (used also in 3:15b) mockers. For similar reasons, S uses סופענספ ("and he will demolish"), though Lady Wisdom is said to "mock" (ארא און אין האווא הא

For הכל שבזרא וולענוים [ולענוים ("and he protects [lit. 'hovers over'] the wise"). This should be emended to הכל ניא מר ("and will have mercy toward the meek"). Also in 13:15 S understands this phrase as "show mercy," though it translates differently there.

3:35 (ימ - ימימ M (haplog מֵרֶים [ מרימ<ים) א מַרָים (ימ - ימימ א מון א מַרָים) א מ

Since the versions had little choice but to use plural verbs to match the plural "fools," they do not necessarily represent a plural מרימים in their source text. Nevertheless, מרימים is a reasonable conjecture (ABP 1.163), with M explicable as haplography. הרים can mean "separate," "remove," hence "take away." Whereas wise sons inherit honor (14:29), fools receive society's contempt as their portion (3:35). See the discussion in ABP 1.169. V treats מרים as a noun, *exaltatio*. S leaves the sense vague: מרים ("receive trouble").

In 3:21–35, G places greater emphasis on moral and charitable obligations than does M. In G, 3:27 requires charity to the poor. This is a possible but not necessary interpretation of M; cf. ABP. In 3:29–30a, G precludes the inference that devising evil and quarreling without cause sometimes *are* permitted. In 3:30b, G similarly excludes the inference that revenge is permissible.

4:3 (graph G (לבני האססש א G (בר ש האססש א G (בר (מרכן) א לבני (מרכן) א לבני (graph  $J \to J$ ) For M's רך ויחיד G has שא אסע אינא אינאטע ("obedient and beloved"). "Obedient" is a reasonable paraphrase of רן (lit. "soft"), and רוי is translated מאמא אינט in Gen 22:2, 12, 16; Amos 8:10; Zech 12:10; and Jer 6:26.

For M's לפני, several MSS KR have the awkward לפני ("to the sons of"), a graphic error. (This happened in the reverse direction in Ps 80:3a.) A Masoretic notation (see Baer 1880, 33) identifies this verse as one of two verses (along with Ps 80:3) in which סבירין לבני וקריין לפני, that is, readers *think* לבני but *pronounce* and write לפני. This note attests to the antiquity of the mistake but validates the other reading.

4:4-5a מצותי M ] מצותי G (ἐντολάς) (near haplog ו→ ۱) M ] > G (unc)

```
קנה חכמה קנה בינה M S (מע, מסבלא) ] > G (theol)
G: (4a) οἱ ἔλεγον καὶ ἐδίδασκόν με, (b) Ἐρειδέτω ὁ ἡμέτερος λόγος εἰς σὴν
```

דְּבְרֵי לְבֶּדְ שְׁמְׂר מִצְּוֹתֵי וֶחְיֵה: 5 קְגֵה חֲכְמָה קְגֵה בִיגָה אַל־תִּשְׁפָּח וְאַל־מֵּט מֵאִמְרֵי־פִּי: 6 אַל־תַּעַזְבֶה וְתִשְׁמְרֵדְ אֶהְבֶה וְתִצְרֶדָ: 7 רֵא שַׁית חֲכְמָה קְגֵה חָכְמֵה וּבְכָל־מִנְיָנְדְ קְגֵה בִינֵה: 8 סַלְסְלֶה וְּתְרוֹמְמֵדְ תְּכַבֵּדְדְ בִּי תְחַבְּמֶגָה: 9 תִּתֵּן זְרֹאשְׁדְ לִוְיַת־חֵן עֲשֶׁרֶת תִּפְאֶרֶת תְּמַגְנֶדָ: 10 שְׁמַע בֵּנִי וְקַח אַמְרֵי וְיִרְבְּוּ לְדָ שְׁנִוֹת חַיִים:

אמר (5a) שָטאמסד בידסאלג, שָׁן באואלטאַן. "[The subject is 'father' and 'mother,' from the preceding verse] (a) who would say [pl] and teach [pl] me: (b) 'Let our word rest firmly on your heart. (5a) Keep the commands, do not neglect (them).' "The only meaning of ממ"ך known to the translator of G-Proverbs (and G elsewhere, with one or two exceptions) is "lean on," "support"; see the comment on 3:18. By translating 'mathing the verbs in 4:4a as plural, G—and this must be deliberate—makes both parents the teachers (see ABP 1.82–83). G reverses the order of the verbs, putting them in their "logical" sequence: saying before teaching. G begins verse 5 with the end of M 4:4b (מצות שמר), without representing היה, which was lost for unclear reasons. (It is supplied in  $G^{MSS}$  and SyrH.)

BHQ (following CTAT) suggests that the translator hesitates to concede that wisdom is the possession of humanity rather than the sole possession of God. This explanation is reasonable, but it must be noted that the translator is not consistent in this concern. G is willing to translate איקנה דעת literally in 18:15, though יקנה דעת is a synonym of the other wisdom words. In any case, I consider 23:23 in its entirety a later intrusion in M.

S translates אמעיבע as אמעיבע ("Let [my word] become strong"). Like G, S understands תמך to mean "support," "rest [something] upon," hence "be strengthened." We should not retrovert to יסמך, contrary to Pinkuss.

At the end of 4:4b, S adds متحمص محمد معرف ("and [keep] my law as the pupil of the eye"), taken from 7:2b. The motive of such transferences, though inconsistently applied, seems to be to enhance the resemblance of passages that are near doublets, on the assumption that what is true in one passage is worth expressing in others.

S lacks אל תשכח. S probably omitted it as being adequately covered by אל השכח ("and do not turn aside").

4:7 M S ] > G (ideol)

This verse is absent in OG. The possibility that it is a later addition in M must be weighed, since it interrupts the exhortations to keep wisdom (4:6) and cherish it (4:8). It is unlikely that the translator omitted this verse since it is largely a repetition of 4:5 (contrary to SDEK), because OG lacks much of that verse anyway. Omission of repetitions could explain, perhaps, 11:4 (cf. 10:2) but not most of the G minuses. Note how 9:16 repeats 9:4 and 19:9 repeats 19:5, in spite of their proximity. Snell (1993, 23–33) says that G reflects most repetitions while varying the wording.

Whatever the reason for the absence, if M 4:7 is retained, the four verses are organized in a parallel structure:

A. Get wisdom (4:5).

B. Keep Wisdom and she'll reward you (4:6).

A'. Get wisdom (4:7).

B'. Keep Wisdom and she'll reward you (4:8).

4:10 ארחות [ (בייה M S (גייה G (όδοί) (assim)

G: (a) Ἄκουε, υἱέ, καὶ δέξαι ἐμοὺς λόγους, (b) καὶ πληθυνθήσεται ἔτη ζωῆς σου, (c) ἵνα σοι γένωνται πολλαὶ ὁδοὶ βίου. "(a) Listen, son, and receive my words, (b) and years of your life will be multiplied, (c) so that the ways of life may be many for you." G 4:10c is OG (and alone present in Sahidic, per BHQ). The doublet in G 4:10b is closer to M 4:10b and is recensional. The assimilation of "years" to "ways" (ặᇊ rồi nh 4:10b and is discussed in the comment on 2:19. In 4:10bc, as in 9:11, G and S add "your," to bring out the possessive implicit in 7.

11 בְּדֶרֶך חֲכִמְה הֹרַתֵּיף הֹדְרַכְהִיּדְ בְּמַעְּגְּלִי־יְשֶׁר: 12 בְּלֶכְתָּד לֹא־ יֵצֵר צַעֲדֶדְ וְאִם־תִּרִיץ לָא תִכְּשֵׁל: 13 הַחַזֵק בַּמּוּסָר אַל־תֶּרֶף גִצְעָה כִּי־הִיא חַ״ֶּיך: 14 בְּאַרַח רֲשָׁעִים אַל־תָּבָא וְאַל־תְּאַשֵׁר בְּדֶרֶך רָעִים: בּזַרְהָיא חַ״ֶּיך: 14 בְּאַרַח רֲשָׁעִים אַל־תָּבָא וְאַל־תְאַשׁׁר בְּדֶרֶך רָעִים: 15 פְּרָעֵהוּ אַל־תַּעֲבָר־בָּוֹ שְׁטֵה מֵעָלָיו וַעֲבְוֹר: 16 כִּי לָא יִשְׁנוּ אִם־ לָא יְרֶעוּ וְנִוְזְלָה שְׁנָתָם אִם־לָא יֹיַכִשְׁינוּי]: 17 כִּי לָחַמוּ לֶחֶם רֶשַׁע לָא יְרֶעוּ וְנִוְזְלָה שְׁנָתָם אִם־לָא יוַכִשְׁיהוּי]: 17 כִּי לָחַמוּ לָחֶם רֶשַׁע וְיֵיז חַמָסִים יִשְׁתּוּ: 18 וְאַרַח אֲדִיקִים כְּאוֹר גְגַהּ הוֹלֵדְ וְאוֹר עַד־נְכָוֹז הַיּזֹם: 19 דֶרֶך רְשָׁעִים בְּאַכֵּלָה לָא יְדִעוּ בַּמָּר יִכָּשִׁלוּי: 20 בְּנִי לִדְבָרֵי הַקְשָׁיבָה לַאֲמָרִי הַט־אָזְגֵדְ: 21 אַל־יּיָלָוּזוּן 20 בָּנִי לַדְבָרֵי הַקִשְׁיבָה לַאַמְלִי הַט־אָזְגָדָ: 21

#### 4:12

G: (a) = M; (b) ἐἀν δὲ τρέχῃς, οὐ κοπιάσεις. "(b) and if you run, you will not grow weary." G replaces M 4:12b with a reminiscence of M-Isa 40:31: ירוצו ולא ייגעו ילכו ולא ייעפו, but in the second-person singular. (G of that verse is quite different.)

#### 4:14

G translates אשל־תאשר ("do not go") as μηδὲ ζηλώσης ("and do not envy"). This construes אשר as the homonymous "declare/consider fortunate" and associates it with the admonition not to envy the wicked, as in Prov 3:31; 23:17; Ps 37:1. Also α' and θ' (μηδὲ μαχαρίσης) parse the verb in this way.

S La rela = G.

#### 4:15

S: (a) הבא גבו האבר געי, (b) = M. "(a) And do not cross through the place they encamp." S follows G's puzzling construal of ברעהו but stays closer to the Hebrew for the rest of the verse.

4:16 יכשילו M<sup>Q</sup> V (*subplantaverint*) T (דעבדין תיקלא) [ (דעבדין M<sup>K</sup>; ישנו M<sup>K</sup>; ישנו (M<sup>K</sup>: gram; G: distant dittog ישנו 4:16a)

The ketiv (יְבְשׁוֹלוֹ) and the qere (יְבְשׁוֹלוֹ) are different expansions of consonantal יכשלו (ketiv = "stumble"; qere = "cause to stumble"). The defectiva writing allows for the metathesis that produced G's ישכבו. The qere is preferable, because the sentence calls for an evil act, not a bad consequence.

G: (a) où yàp µỳ ὑπνώσωσιν, ἐἀν µỳ κακοποιήσωσιν. (b) ἀφήρηται ὁ ὕπνος αὐτῶν, καὶ où κοιµῶνται. "(a) For they will not sleep if they do not do evil. (b) Their sleep is removed, and they do not fall asleep." Probably a copyist duplicated ישנו from 4:16a (cf. BHQ), which G translated differently for variety's sake. It is also possible that κοιµῶνται = ישכבו (BHS; Heidenheim) for יכשולו or, earlier, יכשלי. In either case, the translator had to ignore אם־לא

S has בגבא ("until") for אם־לא (twice). In 4:16bβ, בגבא גברא ("until they do their will"), S was apparently trying to make sense of the awkward intransitive יכשולו or a *defectiva* יכשולו. S understood "if they do not stumble" to mean that the wicked stumble into *sin*, and only then can they sleep. Pinkuss retroverts S to i" משולו ("rule"); that, however, would not produce S.

#### 4:19

G's דאפלה fo כאפלה ("dark") lack the כאפלה. Several MSS KR have the more obvious (and prosaic) באפלה ("in darkness").

# 4:21 ניליזו [ יליאו (graph 1→ י)

G (continuing 4:20): (a) อัสพร นท์ ร่หงโสพชร์ง σε ล่ สฤγล์ σου, (b) φύλασσε מטֹדמׁג בֹּט סון אמסלוֹם. "(a) so that your springs not abandon you. (b) Guard them in you heart." The translator misread מעיניך ("from your eyes") as מעיניך ("your springs") and made the apparently missing direct object "them" (מטֹדמֹג) refer to the springs rather than to the father's words. G correctly translated מעיניך in the nearly identical 3:21a, and it may be that an exegetical tradition led the translator to treat the word as "your שְׁמְוֹם בְּתוֹדְ לְבָבֶדְ: 22 כִּי־חַיֵּים הֵם לְמֹצְאֵיהֶם וְּלְכָל־בְּשָׂרָוֹ מַרְפֵּא: 23 מְכָּל־מֵשְׁמָר נְצִׂר לְבֶּדְ כִּי־מִמָּנוּ תּוֹצְאָוֹת חַיִּים: 24 הָסֵר מֵמְדָ 23 מִכְּל־מֵשְׁמָר נְצִׂר לְבֶּד כִּי־מִמָּנוּ תּוֹצְאָוֹת חַיִּים: 44 הָסֵר מֵמְדָ 34 מִקְשִׁוּת פֶּה וּלְזוּת שְׁפָתַים הַרְחֵק מִמֶּדָ: 25 אֵינֶידְ לְנַכַח יַבָּיטוּ 34 זְעַפְעַפֶּיד יַיִשְׁרוּ נֶגְדֶדָ: 26 פַּלֵס מַעְגַל רַגְלֶדֶ וְכָל־דְרָכֶידְ יִבְּנוּ: 37 אַל־תֵּט־יִמֵין וּשְׂמָאול הָסֵר רַגְלְדָ מֵרֶע: 5:1 בְּנִי לְחָכְמָתֵי

springs" in 4:21 in spite of the resulting complication, namely, the implication that the springs should be preserved in the heart. G must understand "springs" as Torah. The teachings of Torah or the father are the "founts" of life in 10:11; 13:14; 14:27; 16:22. These verses use מקור not מקור but G uses  $\pi\eta\gamma\dot{\eta}$  there as in 4:21.

The different preposition of S's حمتيم resulted from understanding the verb as "be trivial."

#### 4:22

In G 4:22a, the direct object αὐτάς must again refer to the springs. (The variants αὐτῶν in SyrH and some manuscripts and αὐτήν in G<sup>B</sup> [sc. ῥήσει in 4:20] are inner-G adjustments.) καὶ πάσῃ σαρκί ("and for all flesh") for M's "and for all his flesh" universalizes the benefit of wisdom.

#### 4:25

G: (a) = M; (b) τὰ δὲ βλέφαρά σου νευέτω δίκαια. "(b) and let your eyelids assent to righteousness." G has the gist of M but a different image. In Prov 21:1, νεύειν ("to nod," hence "affirm") = Tar ("wish"). G is being logical: the eyelids do not actually go straight, but they can gesture affirmation. Sir 27:22 (G) uses διανεύειν of winking the eye.

# 4:26 (אמדבי) א S ( (נאמבר) [ \* (נאמב) G (אמדεύθυνε) מ' σ' θ' (ἐτοίμαζε) (unc)

G: (a) ὀρθὰς τροχιὰς ποίει σοῖς ποσὶν (b) καὶ τὰς ὁδούς σου κατεύθυνε. "(a) Make tracks straight for your feet, (b) and make your ways go aright." G lacks "all" in 4:26b, but it seems necessary in the Hebrew both for line length and for emphasis. For M's יכנו (N-stem), G has κατεύθυνε, a transitive singular imperative; α', σ', and θ' have ἐτοίμαζε, likewise singular imperative transitive; and Heb 12:13a has ὀρθὰς ποιεῖτε, a plural imperative transitive. (Hebrews recasts the sentence as plural to fit the context.) Thus there is independent evidence for the existence of a variant [, c, c] why

the change would take place in translation). Perhaps the imperative in the second stich was influenced by the imperative in the first.

S: ארבי ז גלא גע שושא. (b) = M. "(a) Remove your foot from bad paths (b) and all your ways will be right." S 4:26a is based on M 4:27b (הסר רגלך מרע) (Pinkuss), not on M 4:26a.

## **4:27** fin ] + 4 stichoi G (27*a*-27*b*) (elab)

G: (a) μὴ ἐκκλίνης εἰς τὰ δεξιὰ μηδὲ εἰς τὰ ἀριστερά, (b) ἀπόστρεψον δὲ σὸν πόδα ἀπὸ ὁδοῦ κακῆς. "(a) Do not turn aside to the right or to the left, (b) but turn your foot away from the evil way." See 4:27*a*, *b*.

#### 4:27*a*

G: (a) δδοὺς γὰρ τὰς ἐx δεξιῶν οἶδεν ὁ θεός, (b) διεστραμμέναι δέ εἰσιν αἰ ἐξ ἀριστερῶν. "(a) For God knows the ways on the right, (b) but those on the left are perverted." In contrast to 4:27, the added 4:27*a* equates the right and left with good and evil respectively, which is a Greek, not a biblical, cliché. This identification disturbs the original metaphor of a single true path. The injunction in 4:27 *not* to incline to the right puzzled a Greek scribe, who "corrected" it in 4:27*a*.

#### 4:27b

G: (a) αὐτὸς δὲ ὀρθὰς ποιήσει τὰς τροχιάς σου, (b) τὰς δὲ πορείας σου ἐν εἰρήνῃ προάξει. "(a) And he will make your paths straight, (b) and your journeys he will guide in peace." This sentence rephrases M 4:27 in a way that introduces God's activity (Dick 1990, 27) and construes the straightness as well-being rather than rectitude.

G 4:27*a* and 27*b* are two distinct restatements of 4:27. Since the OG translator found M 4:27 clear, we may regard both as later scribal interpretations of that verse. Both additions are absent in S but are present in some manuscripts of V. Since 4:27*b* is directly related to 4:27, it was probably the earlier addition. Then a Greek scribe inserted 4:27*a* directly after the verse to which *it* responds. (My assumption is that a scribe who is "responding" to a verse in OG would have inserted his addition right next to it. Consequently, whichever of two additions is adjacent to it is the later.)

# 5:1 הלתבונתי M [ לאמרי [ ( έμοῖς δὲ λόγοις) S ( ארבאמרי [ syn)

For לתבונתי G has לאסוג אין אַמָרַי = אָאָמָרַי) (S supports this rather than לתבונתי cf. 4:10, 20; etc.). תבונה is not translated אלאסג elsewhere. This

ַהַקְשֵׁיבָה לְּתְבוּנְתִׁי הַט־אָזְגֶדְ: 2 לִשְׁמִׁר מְזָמֵוֹת וְדַעַת שְׂפָתֶידְ יִנְאָרוּ: 3 <sup>ו</sup>אַל תַּקְשִׁיב לְאֵשֶׁת אָוֶלֶת כִּי נְפֶת תָּטֹפְנָה שִׂפְתֵי זֶרֶה וְחָלֶק

synonym interchange is due to assimilation to 4:20b, probably in the Hebrew, since the wording in Greek there is somewhat different.

5:2

G: (continuing 5:1) (a) ίνα φυλάξης ἔννοιαν ἀγαθήν. (b) αἴσθησιν δὲ ἐμῶν χειλέων ἐντέλλομαί σοι. "(a) so that you may guard good understanding. (b) And the knowledge of my lips I command you." There are three differences between M and G. (1) G adds מעמאי ("good") to מזמות, to provide a moralizing enhancement, though έννοιαν itself introduces the moral connotation that מזמות lacks. (2) G has "my lips" for M's "your lips." The translator was apparently convinced that דעת שפתיך is a bound phrase, and since the wisdom to be kept was the father's not the son's, he simply adjusted the possessive to "my." (3) G translates ינצרו as ἐντέλλομαι, though the latter verb always renders צוה. Not finding a subject for ינצרו (it is actually "your lips"), the translator takes the "I" that he has provided in the possessive of "my lips" and makes the verb into something that the father is doing. Confusion with or assimilation to the frequent צוה (BHQ) is unlikely, and no variants are indicated. The translator is not being "free." Rather, his own misconceptions about דעת שפתיך lead him astray, step by step.

5:3 אַל תַקְשִׁיב לְאֵשֶׁת אָוֶלֶת G (μή πρόσεχε φαύλη γυναικί; cf. V [ne intenderis fallaciae mulieris], rab) ] > M S (unc)

דרה M S (הבואה) ] אונה G (אטאואטעד לא א האר) (phon א ר ה והמישאה), near syn)

G: (a) μὴ πρόσεχε φαύλῃ γυναικί, (b) μέλι γὰρ ἀποστάζει ἀπὸ χειλέων γυναικὸς πόρνης, (c) ἡ πρὸς καιρὸν λιπαίνει σὸν φάρυγγα. "(a) Do not hearken to the base woman, (b) for honey drips from the lips of the harlotrous woman, (c) who for a season anoints your throat."

M lacks a sentence after 5:2. The motivation in 5:3a ("For the strange woman's lips drip honey") does not motivate the exhortation in 5:1–2 to listen to wisdom and hold fast to it. The other Strange Woman passages follow up the exhortation to gain wisdom with a sentence promising protection from the woman's temptations: "to save you from a strange woman, an alien who speaks smooth words" (2:16); "to guard you from another man's wife, from the smooth talk of the alien" (6:24); "to guard

you from a strange woman, an alien who speaks smooth words" (7:5). Literary considerations justify the conjectural addition of a sentence to fill the logical gap between verses 2 and 3.

Midrash Proverbs begins a homily on 5:3 with בני הזהר מאשה זונה ("My son, beware of a whoring woman"). This sentence is not cited as part of the verse proper, but it closely resembles G and suggests that a sentence like this was known even if it was not in the text the midrash uses. One might supply a couplet along the lines of 7:5: לְשָׁמָרֶדְ מֵאָשָׁה זָרָה מְנָכִרְיָה אָמֶרִיהָ הֶחֵלִיקָה ("to keep you from the strange woman, from the alien who speaks smooth words"). Thus Toy, Scott, Whybray (1994, 23), ABP, and others. An alternative is to retrovert G 5:3a to (approximately) אל הַקשיב לאשה זונה. Other linking sentences that use the vetitive (+ impf) are 1:10; 3:27; and 4:14. It is, to be sure, possible that the translator himself provided the linking sentence for the same reason I do, and it is possible that he reformulated an existing Hebrew sentence. However, I am using G as evidence for the linking sentence because it is not purely conjectural, as it would be to use 7:5 here (thus ABP). My basic claim is that there was such a sentence; the precise wording is harder to recover. Perhaps φαύλη אשת אולת = אשת אולת, on the pattern of in 9:13; cf. מיאָם אויל = איש אויל in 29:9. Some V MSS have a similar sentence at the end of 5:2: ne intenderis fallaciae mulieris (with variants) ("Pay no heed to the wiles of a woman"). The mechanism of the loss is unclear.

In G 5:3b, γυκικός πόρνης ("harlotrous woman") represents אָוָה, which is an error for דרה (likewise in 23:27; see Kaminka 1931–32, 178; and compare Prov 5:3 with 6:7; 7:4; and probably 21:6). Though the graphic resemblance between 1 and ר is not great, it seems certain that the letters were occasionally interchanged. ר/ז interchange can also be induced by the phonetic resemblance between the liquids 1 and ר. Examples are בוכדראצר – נבוכדראצר (2 Kgs 25:22; sim 2 Kgs 24:1, etc.). This is a phonetic change, not a scribal error. See also Kgs 24:1, etc.). This is a phonetic change, not a scribal error. See also Phonetic (Ps 91:10)/ (Neh 7:7; Ezra 2:2); ויתר (2 Sam 22:33) ויתר (Ps 18:33); תרא[ה] (11QPsAp<sup>a</sup>); see also Kennedy 1928, 100–101. In the present case (and in Prov 23:27), another catalyst for דנה/זרה interchange could be the pragmatic synonymity of the terms in the context of the Strange Woman.

In 5:3c, G has a puzzling η πρός καιρόν ("who, for a season") for M's indicating h μρός καιρόν ("who, for a season") for M's presents itself. Nor is a graphically similar variant evident for either the Greek or the Hebrew. De Lagarde emends πρός καιρόν to πρό έλαίου =

מִשֶּׁמֶז חִבְּהּ: 4 וְאַחַרִיתָהּ מָרָה כַלַּעֲנָה חַדָּה כְּחָרֶב פִּיְוֹת: 5 רַגְלָיהָ יְרְדַוֹת מֵוֶת שְׁאוֹל צְעָדֶיהָ יִתְמְכוּ: 6 אַׂרַח חַיִּים פֶּז־הְפַלֵּס נְעִוּ מַעְגְלֹהֶיהָ לָא תֵדֶע: פּ 7 וְעַתָּה בֶנִים שִׁמְעוּ־לֵי וְאַל־הָסוּרוּ

משמן. The latter, however, is already represented by G  $\lambda \imath \pi \alpha i \nu \epsilon \imath$  ("lubricates").

For M's  $\exists \Box \subseteq ($  "her palate"), G has σὸν φάρυγγα ("[lubricates] your palate," i.e., kisses you), and may have read  $\exists \Box \subseteq \Box \subseteq \Box \subseteq \Box \subseteq \Box$  ("your palate")—a dittography of  $\Box$ . However, this is uncertain, since G will add (and subtract) possessive pronouns fairly freely.

S explicates the metaphor of palate as تحلينه ("her words"), cf. S 2:16; 7:5. In S, the adjective  $\pi d = \pi d = \pi d$  is understood to mean "smooth" only here; elsewhere S usually construes the word etymologically, from the notion of "divide," and translates it with a form of  $d = \pi d = \pi d$ .

5:4

G: (a) ὕστερον μέντοι πικρότερον χολῆς εὑρήσεις (b) καὶ ἠκονημένον μᾶλλον μαχαίρας διστόμου. "Later, however, you will find her more bitter than gall, (b) and sharper than a two-edged sword." G rephrases 5:4a to avoid a misunderstanding. M's "but her end is as bitter as gall" might be thought to mean that she herself dies a hard death, whereas it is the fool's death that is relevant to context. This is an example of how G-Proverbs guides and limits interpretation without necessarily distorting the meaning or imposing an ideological shift. The comparative degree of the adjectives "more bitter" and "sharper than" serves for literary intensification (cf. 6:6). However,  $\beth$  and 𝔅 are very similar in many square scripts, and the variants αμ" (from the sword").

5:5

G: (a) τῆς γὰρ ἀφροσύνης οἱ πόδες κατάγουσιν (b) τοὺς χρωμένους αὐτῆ μετὰ θανάτου εἰς τὸν ἄδην, (c) τὰ δὲ ἴχνη αὐτῆς οὐκ ἐρείδεται. "(a) For the feet of folly lead (b) those who consort with her down to Hades with death. (c) Her steps are not firm." G substitutes "folly" for M's suffixed "her." De Lagarde and Jäger explain ἀφροσύνη as a synecdoche for the foolish woman. This is possible, but the use of the abstraction favors an allegorical identification of the woman described here with folly itself, and this is reinforced by 5:19.

Though both G and S translate ירדות ("go down") as causal, "bring down" (cf. G 7:27), they do not reflect a variant (מורידות\*?). Both versions seek to make it clear that not only does the wicked woman die; she *kills*. In Jer 9:17, both G and S treat the G-stem of ירדו as causative. Even Naḥmias construes 'רדות' in the present verse as causative and compares 1 Sam 23:6 and Jer 9:17.

G's assumption that תמך means "lean on," hence "be supported," "be firm," rather than "grasp" (see the comment on 4:4) causes a number of tangles. The translator is obliged to supply a negative (here, in où געמנ (here, in où געמנ) so as to avoid an intolerable statement, namely that folly's steps *are* firm. He joins "Sheol" to the first clause (5:5b) rather than taking it as the direct object of תמך. To make sense of מות שאול, he treats an adverb of means rather than location.

Such contortions show that it is inadequate to characterize G as "free" just because it differs from M (or, more precisely, our interpretation of it). The translator is trying to render his Hebrew text as precisely as possible. When his linguistic assumptions differ from the ones current today, his translation may still be bound to the Hebrew text *as he understood it*; see §3.1.4 in the Introduction.

S understands תמך as "support" (حמרבאה) but takes "Sheol" as its subject: המד האבאה איבאה במרבאה ("Sheol supports her steps").

5:6

G: (a) όδοὺς γὰρ ζωῆς οὐκ ἐπέρχεται, (b) σφαλεραὶ δὲ αἱ τροχιαὶ αὐτῆς καὶ οὐκ εὖγνωστοι. "(a) For she goes not upon the ways of life. (b) Her courses are slippery and not well-known." σφαλεραί "slippery" means more precisely "likely to make one stumble." She goes on paths that are rough or slippery. G produces a more obvious metaphor than M's "her courses wander." M's  $\forall \mathsf{M}$  was parsed as second-person singular ("you do not know") and converted into an adjectival construction: "not well-known." In other words, the paths are foreign or alien; hence one who treads them stumbles (5:6a) and gets lost (5:6b). Compare the image in 9:12b.

S: Line of the area on the way of life, (b) for her paths are a trackless wasteland and not known." The singular "way" = M (and adjusts to the usual metaphor of the Two Ways, one of which is the way to death). S learned "not well-known" from G but dealt with the difficult metaphor of paths "wandering" independently.

5:7-17

מַאִמְרֵי־פִּי: 8 הַרְחֵק מַעָּלֵיהָ דַרְכֵּדְ וְאַל־ּחִקְרַב אָל־פֶּתַח בֵּיתֵה: 9 פּן־תִּתֵּן לַאָחַרֵים הוֹדֶדְ וּשִׁנֹתֶידְ לְאַכְזָרִי: 10 פּן־יִשְׁבְּעַוּ זָרַים כּּחֵדְ זְעַצְּבֶּידְ בְּבֵית נְכְרֵי: 11 וְנָהַמְתָּ בְאַחֲרִיתֶדְ בִּכְלִזֹת בְּשָׂרְדָ וּשְׁאֵרֶדְ: 12 וְאָמַרְתָּ אֵידְ שָׁנֵאִתִי מוּסֵר וְתוֹכַּחַת נָאַץ לִבְּי: 13 וֵלֹא־שָׁמַעְתִי בְּקוֹל מוֹרֵי וְלְמְלַמְדֵׁי לֹא־הִפֶּיתִי אָזְנֵי: 14 בְּמְעַט הְיֵיתִי בְכָל־רֶע בְּתוֹדְ קָהֵל וְעֵדֵה: 15 שְׁתֵה־מַיִם מִבּזֹרֶדְ וְנֹזְלִים מִתְוֹד בְּאֵרֶדָ: 16 <sup>-</sup>פֶּן' יְפָוּצוּ מַעִינֹתֶידְ תוּצָה בְּרָחֹבּזֹת פַּלְגֵי־מֵים: 17 יְהוּ־לָדָ

5:7

G changes "sons" to the singular vié for the sake of consistency with 5:1; similarly in 7:24 and 8:32.

#### 5:8

G: (a) = M; (b) μὴ ἐγγίσῃς πρόθυραις [G<sup>A</sup>] οἶκων αὐτῆς. "Do not approach the front doors of her house." (The variant used in Rahlfs, πρὸς θύραις [G<sup>BS</sup> etc.], is ungrammatical and corrupt [Zuntz 1956, 129].) πρόθυραις (often for פתח) is emphatic: Don't approach *even* her outer doors. The plural οἴκων is not due to "scribal inadvertence" (Toy) but used in reference to a residential complex; cf. 7:8.

#### 5:9

For M's הודך ("your glory"), G has  $\zeta \omega \eta \nu$  ססט ("your life"). Since it is unclear how one could leave his "glory" in a foreigner's house, G construed הוד (uniquely) as *life*, influenced by the parallel "your years" (contrary to BHS and ABP, which emend to הוד).

S has אולא ("your wealth" or "your strength"). Pinkuss believes that S read העולד ("your wealth"), which אולא sometimes means. It is more likely that S intended "your strength" as a clarification of M's "your glory." Strength, referring here to sexual power, is a metonymy for offspring; see ABP 1.195.

For M's אאבזרי, G has the plural ἀνελεήμοστν ("to cruel men"), under the influence of the parallel plural. G often smoothes out sudden shifts in number. S also uses the plural: לבר עממין גרא ביעבי, T<sup>Z</sup> has לבר עממין ("to a foreigner") and T<sup>L</sup> has לנוכרין ("to aliens"). These may reflect an association of אכזר ("foreign") (as in Rashi to Job 19:13), under the influence of Prov 5:10.

נוהמת [M] אונהמת הא<br/>מאסיא האמישי הער האמישי האמין א אונהמת האמישי ה<br/>  $(graph \to \neg \to \neg)$ 

G has אמו שבדמשבאחָטָאָק ("and you repent") = וְנָחַמְהָ your soul repents"). כאחריתך געלדעי translates באחריתך. It should be noted that G-Proverbs frequently omits the possessive suffix of אחרית (14:13; 20:21 [G 20:9b]; 25:8; 29:21) and once (24:14) adds a suffix of its own.

For M's בשרך ושארך G has σάρχες σώματός σου and S has , both meaning "the flesh of your body." This is probably a correct interpretation of the hendiadys.

#### 5:13

Whereas M divides the line between מורי מורי הלמלמדי, G ( $\pi$ מוט ( $\pi$ מוט ( $\pi$ מוט ( $\pi$ מוט ( $\mu$ ε אמו) א ווא שיט ( $\pi$ מוט ( $\mu$ ε) joins the words as direct objects of "I did not listen" and consequently creates an imbalance between the two parts of the verse.

#### 5:15

G: (a) πῖνε ὕδατα ἀπὸ σῶν ἀγγείων, (b) καὶ ἀπὸ σῶν φρεάτων πηγῆς. "(a) Drink water from your vessels, (b) and from the wells of your spring." It is difficult to align G with M, except in the basic image of drinking from one's own water sources. בולים is not translated by ἀγγείον elsewhere; נולים is ignored (perhaps considered as included in "water"); σῶν φρεάτων πηγῆς is a double translation of מתוך בארך; and קרבית are treated as plurals. The motives for these divergences from M are hard to discern. Baumgartner understands "vessels" as an allusion to a wife, noting that a wife is called a cup (כוס) in b. Ned. 20b and perhaps a vessel (σκεῦος) in 1 Thess 4:4. Certainly this is the intention of the verse's metaphor in the Hebrew, but the plural of "vessels" and "wells" runs contrary to it.

# 5:16 (יפוצו - פניפוצו G ? (<br/> $(\mu \eta)$ ] > M S (near haplog יפוצו - פניפוצו )

פלגי M ] שלגי האפלגו (graph  $^* \sim G$  ( $\delta$ ומ $\pi$ ορευέσθω) S ( $\mathbf{u}$  וווס) (graph  $\mathbf{v} \rightarrow \mathbf{1}$ )

It is uncertain if  $\mu \eta$  (G<sup>BS\*</sup>) represents בן ("lest"), which is missing but seems called for in M, or if the translator was forced to supply a negative particle to make sense of the verse (thus BHQ), an expediency taken in 1:17a; 3:32b; 5:5c; et al. (It is removed in G<sup>ScA</sup> 5:16.) בן is usually translated <sup>i</sup>va  $\mu \eta$  in G-Proverbs, but it is represented by a simple  $\mu \eta$  in 25:10. I would supply it in any case as a conjectural emendation. The translator construes "מַעִינתִיך ("your springs") as מַעִינתִיך ("from your springs"). He must then supply a subject ("the waters") from 5:15. Gerleman (1956,

#### 5:17-19

# ּלְבַדֶּדְ וְאֵיז לְזָרִים אַתֵּדְ: 18 יְהִי־מְקוֹרְדָּ בְרָוּדְ וּשְׂמַׁח מֵאֵשֶׁת נְעוּרֶדְ: 19 אַיֶּלֶת אֲהָבִים וְיַעֲלַת־תֵז דַדֶּיהָ יְרַוָּדָ בְכָל־גֵעָת בְּאַהַבָתָה תִּשְׁגֶה

18) offers this verse as an example of a Greek predilection for antithetical parallelism. The antithesis arose, however, either because G had i in his source text or because he wished to keep the reader from thinking that a man's "waters" (sexual powers) *should* be spilt outside.

S: معذف متعميه معنهميه. "(a) And your waters will overflow in your streets (b) and in your streets they will flow." S combines M's "your springs" and "your waters" into حتبه ("your waters"), which are probably to be understood as the male generative powers or as the semen itself (Isa 48:1; RH). S (lacking a negative in 5:16b) understands the spilling of waters as a blessing, namely fecundity, not as something to be avoided. Since S is independent of G in this verse, its ("will flow," "let [them] flow") is some support for the variant up, a verb treated loosely as a future/jussive.

#### 5:17

G: (a) ἔστω σοι μόνω ὑπάρχοντα, (b) καὶ μηδεὶς ἀλλότριος μετασχέτω σοι. "(a) Let her belong to you alone, (b) and let no stranger partake with you." G adds a verb in 5:17b. (In Hebrew, the force of the verb יהיי carries into the second clause.) M's זרים is treated as singular, probably reflecting the assumption of a singular paramour.

S: And let them be to you alone, and strangers not partake with you." S too supplies "partake" in the second clause but maintains the plural of M. Since S would not need this verb, it is probably getting it from G. This is another example of S combining elements of the two versions.

5:18 באשת [ M ] באשת rab  $M^{\rm MSS}$  G (עבד אינא איטע איט א געע איט א ט א V (cum muliere) (graph  $\square \to \square$ 

G: (a) ή πηγή σου τοῦ ὕδατος ἔστω σοι ἰδία, (b) καὶ συνευφραίνου μετὰ γυναικὸς τῆς ἐκ νεότητός σου. "(a) Your fount of water—let it be for you alone, (b) and rejoice with the wife of your youth." Where M has τη, G has σοι ἰδία ("to you alone"). This was accidentally introduced into this verse by the translator or copyist, from the preceding verse. לבדך ("for you alone") (Baumgartner ref Vogel) is graphically distant from LCC.

For M's ושמח מאשת ("get pleasure from the wife [of your youth]"), G has גמו סטיבט ("get pleasure from the wife [of your youth]"), G has גמו סטיבט שבדמ ("and rejoice with") = -a reading supported by the above-mentioned sources and found in Mid. Shoher Tov, ad loc. See ABP 1.202.

#### 5:19

G: (a) ἕλαφος φιλίας καὶ πῶλος σῶν χαρίτων ὁμιλείτω σοι, (b) ἡ δὲ ἰδία ἡγείσθω σου καὶ συνέστω σοι ἐν παντὶ καιρῷ. (c) ἐν γὰρ τῆ ταύτης φιλία συμπεριφερόμενος πολλοστὸς ἔση. "(a) A doe of affection and a filly of your graces, let her be together with you. (b) And may your own one guide you and associate with you at all times, (c) because, consorting in the affection of this one, you will become very great."

There are some significant variants in implicit vocalization: (1) דּדֶיהָ (G ή δὲ ἰδία) for M דָדֶיהָ Ner translator correctly understood דריה love" to mean her lovemaking, which he euphemized as "your own one"; cf. S הוישנה (C) יְרוּך (G ήγείσθω σου;  $\approx$  S (.(. Le S)), for M הוישנה (G) אסוינאלים (G πολλοστός ἔση), for M הַשָּׁנֶה

G a: This stich is expansive. It adds a possessive pronoun to η m—"your graces" (σῶν χαρίτων)—as well as a verb, ὁμιλείτω σοι ("let her *be together* with you"). The idea of being "together with" is taken from stich c.

G b:  $\dot{\eta}$  δὲ ἰδία corresponds to דדיה, which G understood as דָּיָהָ ("her love") and construed as an abstract-for-concrete metonymy meaning *her*—your wife herself, who is "your own one." It is the opposite of τῆς μἡ ἰδίας ("the one who is not your own") = נכריה in 5:20. Baumgartner suggests that ἡ δὲ ἰδία ("your own one") = לבדה, for M's דיה, but that is graphically distant.

The clause ἡγείσθω σου shows that the translator is not only thinking of a mortal woman, because he would not have considered it desirable to let a man's wife "lead" him. The verb ἡγεῖσθαι, especially with the genitive, implies rule and control as well as guidance. In G it commonly translates words for various types of rulers and chiefs, particularly and Without displacing the literal sense of the verse, the translator implies an תָמִיד: 20 וְלָמָה תִשְׁגָּה בְנֵי בְזָרֶה וּתְחַבֵּק חֵק נְכְרִיֶּה: 21 כֵּי נֵׁכַח | עֵינֵי יֶהוָה דַּרְכֵי־אֵישׁ וְכָל־מַעְגְלֹתֵיו מְפַלֵּס: 22 עֲווֹנוֹתִיו יִלְבְדָנִוֹ יּ י וּבְחַבְלֵי חַשָּׁאתוֹ יִתָּמֵך: 23 הוּא יֵמוּת בְּאֵין מוּסֵר וּבְרָב אוּלְתוֹ

allegorization of the woman as personified Wisdom, an idea inspired by the implicit vocalization of ירוּך as "guide you"; see below. M<sup>K188</sup> introduces this vocalization into the orthography as יורוך.

συμπεριφερόμενος is based on אָשָׁגָה "go astray," for which G provides a more decorous "go about with" (Baumgartner). The second rendering, πολλοστός ἔση, parses the word as תִשְׁגָא = אַשְׁגָא ("you will become great") (Cappelli; BHQ recognizes this as a doublet). G prefers spiritual to sexual allusions and neutralizes them in 5:19–20.

G's reading is supported by Zalcman, who argues that שׁג"ה elsewhere has negative connotations, whereas here אָשָׁגָה would refer to the blessing of a large family. And indeed there may be a visual pun residing in the consonantal תשגה, suggesting that a man loyal to his wife will both "lose himself" in his wife and increase. As for תִּשְׁגָה, we should be open to the possibility of playful misuse of words, especially in sexual contexts. Further, the parallel line urges the youth to "slake [his] thirst on her lovemaking [בֹּדִרִם]," a phrase recalling Cant 5:1b, where the lovers are told to "get drunk" on love; see further ABP.

G 5:19b and 19c are probably an OG doublet preserving two interpretations of M 5:19b. As G understands it, the Hebrew would be vocalized as אַיֶּלֶת אֲהָבִים וְיַעֵלַת־חֵן, דֹדֶיהָ יֹרוּדְּ בְכָל־עֵת בְּאַהֲבָתָה תִּשְׁגֶה/תִּשְׂגֶה/תִּשְׂגָה תָמִיד.

Following S's line of interpretation, T reads אילתא דרחמותא ודיצתא איליא דרחמותא דרחמות דרחמות איליף בכל זמן וברחמות הגרם תדירא (A gazelle of love and a doe of kindness, learn good behavior (הוונא איליף) at all times and in her love grow strong (תגרם T<sup>LZ</sup>; var: הגרם T<sup>P110</sup>, 'study')." The latter variant is in line with rabbinic interpretation of Torah study as love and is probably original. On T's interpretation and its rabbinic sources, see ABP 1.393.

# 5:20 בני M S (בו, ] > G (equal)

G: (a)  $\mu \dot{\eta} \pi \sigma \lambda \dot{\upsilon} \varsigma$  (σθι πρὸς ἀλλοτρίαν (b) = M. "(a) Be not great with another woman." G again implicitly points הִשְׁגֶה as הִשְׁגֶה. "Be great" here means to have many progeny. G's τῆς μὴ ἰδίας correctly construes there means up woman other than a man's own wife. On +/- "my son," see the comment on 1:15.

# 5:22 אָת־הָרָשֶׁע א S (גרבנו (>G) (gloss, explic)

G: (a) παρανομίαι ἀνδρα ἀγρεύουσιν, (b) σειραῖς δὲ τῶν ἑαυτοῦ ἁμαρτιῶν ἕκαστος σφίγγεται. "(a) Transgressions hunt a man, (b) and each one is bound in the cords of his own sins." G supplies "a man" and "each one" for explicitness. It lacks את־הרשע ("the evildoer"). The absence is original. "The evildoer" is an epexegetical gloss clarifying the object of retribution, a person's iniquities trap *him*, and not others, and in any case the direct object "a man" is already present in the preceding verse.

S: בעאר האפרי, אלאעד באלא הפערלא געלמסת, גאפרי (a) The evildoer will be caught by his transgressions, (b) and by the rope of his sins he will be bound." S puts the sentence in the passive to solve the syntactical difficulty of M and also to match the passive of 5:22a. יִשְׁגֶה: וּ:6 בְּנִי אִם־עָרַבְתָּ לְרֵעֶדְ תָּקַעְתָ לַזֶר וּפַפֶּדְיּ: 2 נוֹקַשְׁתָּ בְאִמְרֵי־פֵּידְ גִׁלְפַׁדְתָּ בְּאִמְרֵי־פִידְ: 3 עֲשֵׁה זָאת אֵפָוֹא | בְּנִי

6:1 כפך M<sup>MSS</sup> G (σην χεῖρα) αλ' (χεῖρά σου) S (אנר) T (דדך) V (manum tuam) ] בְּפֵיך M (num)

We should read the singular כפך. The gesture accompanying an agreement uses one hand, not two. The consistent witness of the versions supports the singular, though, to be sure, a translator could have chosen the more natural singular "your hand" even if he had כפיך in his source text. Possibly M is a *plene* writing of the singular; see Sperber 1966, 261 and the comment on 3:28.

G-Proverbs chapter 6: See the detailed study of van der Louw 2007, 249–356, whose observations are used extensively in the comments on this chapter. Van der Louw describes the particulars of the translational transformations in Prov 6 and examines its literary-aesthetic features, notably rhyme and assonance.

G: (a) Υίέ, ἐἀν ἐγγυήσῃ σὸν φίλον, (b) παραδώσεις σὴν χεῖρα ἐχθρῷ. "(a) Son, if you go surety for your friend, (b) you will deliver your hand to an enemy." ἐχθρός translates T only here. The translator assumes that 6:1b is the consequence of 6:1a and that the T here cannot be the same person who was called "your friend" in the preceding verse. The translator understood the verse to mean that going surety for a friend would turn him into your enemy.<sup>1</sup> No further explanation is needed for ἐχθρός, such as the similarity of T to T (noted by BHQ). CSP observes that Sir 6:5–17 uses the same words in speaking of a "friend" (φίλος) who turns into an "enemy" (ἐχθρός). As for παραδώσεις σὴν χεῖρα, striking hands in making a bargain was not a Greek usage and required explanation (van der Louw 2007, 259). Every time the idiom appears, G tries a different formulation; see the comments on 17:18 and 22:26.

6:2

G: (a) παγὶς γὰρ ἰσχυρὰ ἀνδρὶ τὰ ἴδια χείλη, (b) καὶ ἁλίσκεται χείλεσιν

<sup>1.</sup> In fact, a זר is anyone outside the sphere of affiliation relevant to context, while a  $ilde{\mathcal{L}}$  is another person within the pertinent sphere of affiliation. In the present passage, the  $ilde{\mathcal{L}}$  is an acquaintance, probably someone living in one's proximity, while the is an outsider, not necessarily a foreigner, who needs someone to vouch for him; see ABP 1.137–38.

### 6:3

G: (a) ποίει, υἱέ, ἁ ἐγώ σοι ἐντέλλομαι, καὶ σῷζου. (b) ἥκεις γὰρ εἰς χεῖρας κακῶν διὰ σὸν ϕίλον. (c) ἴθι [G<sup>BSAMSS</sup>; var: ἴσθι G<sup>MSS</sup>] μὴ ἐκλυόμενος, (d) παρόξυνε δὲ καὶ τὸν ϕίλον σου, ὃν ἐνεγυήσω. "(a) Do, son, what I command you, and be safe, (b) for you have come into the hands of bad men through your friend. (c) Go without being lax [var: "be not lax"] (d) and also importune your friend, for whom you have gone surety." (Van der Louw [2007, 263] argues for the minority variant ἴσθι.) The issue facing modern commentators and ancient translators is, who is doing what to whom? In the following, note that G 6:3cd corresponds to M S 6:3c.

In 6:3a, G omits אפוא but consequentiality is still implied by å έγώ σοι έντέλλομαι.

In G 6:3b, G has a double rendering of M's ארעך, with אמאמע associating אין ("bad") and סטי קואסי reading רעך = M. G assumes that the source of the trouble is evil men (namely, the moneylenders) rather than the friend, who is the borrower. The double rendering refers back to the parties in 6:1 by G's understanding (and is therefore the translator's doing). The gloss in G 6:3d, "for whom you have gone surety," shows that the translator is carefully sorting out the various parties in the transaction.

In 6:3c, M's undignified advice, "go and grovel," is replaced by a warning to avoid being lax (or despondent, באלטט(געסע). Rather, you should go so far as to anger your neighbor by demanding to be released from your pledge. A retroversion of שי באלטט(געסעסע to הרפה אל (CSP; van der Louw 2007, 265) should be avoided, because it must add a negative particle and assume an uncommon graphic change,  $D \rightarrow n$ . The translator derives the notion of persistence from the next verse while guessing at the meaning of the rare verbs.

# וְהִנְּצֵׁל בִּּי בֲאתָ בְכַף־רֵעֶדְ לֵדְ הִתְרַפֵּס וּרְהַב רֵעֶידְ: 4 אַל־תִּתַּן שֵׁנָה לְעֵינֶיֶדְ וּתְנוּמְה לְעַפְעַפֶּידְ: 5 הֵנָּצַל בִּצְבֵי <sup>ר</sup>ִמִמְּצְוֹד<sup>י</sup> וּכְצִפֿוֹר מִיַּד

In G 6:3d, the singular τὸν φίλον σου represents M's -νψ, which looks like a plural but may be a *plene* writing of the singular, cf. 2 Sam 12:11; for the phenomenon, see Sperber 1966, 261.

S: אבר הבא כז, כאל גבל אים, עבו אים, עבו כאיגר, S: אבר הבא כז, כא גער אים, אים איז בא כא כא כז, בא אים אים איז א באר אסת, (a) Do thus, my son, and save yourself, (b) because for the sake of your friend you have fallen into the hands of your enemy. (c) Therefore coax your friend, for whom you have gone surety."

S's treatment of this verse gives insight into its flexible use of G. First, on its own, S ignores אפוא. Then it follows G in some regards, too specific to be chance resemblances: (1) Like G, S attempts to identify the parties in the scenario. (2) S understands the misfortune in 6:3b as falling into hostile hands and adds a phrase, "of your enemy" ( $\approx$  G's "of bad men") to clarify this. (3) It combines the rare verbs in M 6:3c, התרפס and רהב, into one; and (4) it adds "for whom you have gone surety" in 6:3c to define the friend's role. At the same time, S diverges from G. (1) In 6:3a,  $S = M \neq G$ . (2) In 6:3b, S identifies the lenders as "your enemy" rather than "bad men." Perhaps this reduces the moralizing element, for the lenders are hostile to the borrower (when he defaults) but are not necessarily evil. (3) S combines the two verbs in 6:3c differently, summarizing them as "coax." G did not understand the Hebrew verbs; S did. (4) S omits לך ("go") in 6:3b. Thus we see a translator who calls upon G for exegetical guidance but has independent control of the Hebrew and enough confidence to use it flexibly.

6:4

G μηδὲ ἐπινυστάξης ("nor drowse off") construes ותנומה as a verb, probably because the verb is gapped in the second stich in the Hebrew. For suppletion of gapped verbs see CSP 159 and van der Louw 2007, 267. Examples are 1:12, 33; 2:9; 5:10. Compensating for gapping may involve adding a new verb or treating a noun like a verb.

6:5 אמצוד G (ἐκ βρόχων) S ( (ג) אין ממצוד G (ἐκ βρόχων) S ( (ג) אין ממצוד M (near haplog and graph  $m \to m$ ) paleo  $m \to m$  ( $m \to m$ )

מיד יקוש Μ<sup>MSS</sup> G (ἐκ παγίδος) (syn var)

G (continuing 6:4): (a) ἵνα σώζη ὥσπερ δορκὰς ἐκ βρόχων, (b) καὶ ὥσπερ ὄρνεον ἐκ παγίδος. "(a) so that you may save yourself like a gazelle from traps and (b) like a bird from a snare."

M's מיד ("from a hand"), followed by V's in manu, is problematic. It is not a good parallel to מיד יקוש ("from a fowler's hand") in 6:5b. Moreover, one does not catch a gazelle by hand. G (ἐκ βρόχων) and S (ح. ממצוד = T ממצוד) witness to ממצוד ("from a trap"), which G translates with change of number, as often. Ben Sira provides an early witness to the presence of a word for trap in Prov 6:5a. Sir 27:20b reads, και ἐξέφυγεν ώς δορκάς έκ παγίδος (no Hebrew preserved). This verse draws on the phraseology of Prov 6:5 independently of G-Proverbs. In Prov 22:25, βρόχος = מוקש. In 7:21, βρόχος corresponds to חלק ("smoothness"), which is the snare for the gullible youth. (Note the plural for singular correspondence, as is also presupposed in the retroversion suggested here.) βρόχος is not used elsewhere in a translated text. מצוד sometimes means, or is thought to mean, "fortress." G understands it to mean "trap" only in Qoh 7:26, where it is translated θηρεύματα ("snares"). The morphological variant מצודה is likewise always construed as "fortress" or the like, except in Ps 66:11, where it is translated  $\pi\alpha\gamma$ ic. Thus there is little evidence from elsewhere as to how G-Proverbs would have translated מצוד, but βρόχος would have been a reasonable rendering, as  $\pi \alpha \gamma i \varsigma$ would have been in Sir 27:20b.

In G 6:5b, בא המצולסכ possibly reflects מפח, a synonym variant for M's מיד יקוש. This would be a secondary simplification in the Hebrew transmission behind OG. It is supported by Syriac מיד יקוש). כא יקוש (אסטלא יקוש). The fact that several MSS KR have in the second stich shows that the hypothesized change was feasible.

#### 

6:6

(a) "Iôt πρòς τὸν μύρμηκα, ὦ ὀκνηρέ, (b) καὶ ζήλωσον ἰδὼν τὰς ὁδοὺς αὐτοῦ (c) καὶ γενοῦ ἐκείνου σοφώτερος. "(a) Go to the ant, O sluggard, (b) and seeing its ways, emulate (him) (c) and become wiser than that one." G turns the couplet into a tristich, adding "emulate" to make the practical lesson clear. G uses the comparative degree, σοφώτερος ("wiser"), because to the literal-minded translator, it is hardly adequate to be merely *as wise* as the ant. The comparison assumes that the ant *is* wise, a notion that could have been learned from Prov 30:24–25, where too the comparative is used. The use of animals as moral examples that humans should surpass is a Hellenistic topos (van der Louw 2007, 271).

6:7 קצין M ] א קציר (ער בG ( $\gamma \varepsilon \omega \rho \gamma i o \upsilon$ ) ((6:6] ( $m c \tau s \tau$ ) (graph  $t \to \tau$ )

G's γεωργίου ("field") must represent קִצִיך ("harvest"). The error would have been occasioned by  $L/\Gamma$  phonetic similarity (see the comment on 5:3), reinforced by the appearance of קצִיך in the next verse. The translator must have understood "harvest" as a metonymy for field. γεώργιος corresponds to שָׁדֶה in Prov 24:30 and 31:16. Cook (CSP) thinks that γεωργίου was derived from "immediate context," but immediate context actually calls for קצין, which G could have translated easily enough if it were present. (קצין was translated βασιλεῦσιν ["kings"] in 25:15.) Van der Louw, more subtly, argues that "field" was the shortest way to summarize the difference between crops of ants and crops of men, namely, that ants do not own land (2007, 274). But this is not the distinction relevant to this passage. The virtue at issue in this verse is the kind of wisdom that enables one to work diligently even when lacking supervi-

6:5-8

sion, and "field" is not relevant to it. Moreover, "field" is not well coordinated with the other two items that the ant lacks, while the repeated  $\mu\eta\delta\epsilon$  leads us to expect three items in the same category. More likely, G is making do with a graphic error, which produces a certain awkwardness. In any case, "harvest" is confirmed by S.

S treats 6:6-7 quite differently from G, and its אנא ("harvest") would not have derived from G's "field." Hence S independently witnesses to העברא in Proverbs [6:8; 10:5; 20:4; 25:13; 26:1] and usually elsewhere.) T אנד = S.

**6:8** fin ] + 6 stichoi G (8*a*–*c*) (elab)

### 6:8*a*-*c*

G inserts an epigram here: (a) η πορεύθητι πρός τὴν μέλισσαν, καὶ μάθε ώς ἐργάτις ἐστὶν τήν τε ἐργασίαν ὡς σεμνὴν ποιεῖται [var: ἐμπορεύεται G<sup>V</sup>], (b) ἦς τοὺς πόνους βασιλεῖς καὶ ἰδιῶται πρὸς ὑγίειαν προσφέρονται, ποθεινὴ δἑ ἐστιν πᾶσιν καὶ ἐπίδοξος. (c) καίπερ οὖσα τῆ ῥώμῃ ἀσθενής, τὴν σοφίαν τιμήσασα προήχθη. "(a) Or go to the bee, and learn how diligent she is and how seriously she performs work—(b) (she) whose products kings and commoners use for health, and who is desired and well-known to all. (c) Though weak in strength, having honored wisdom she excels." In 6:8*a*, ἐμπορεύεται ("traffics in") in G<sup>V</sup> is arguably original (thus Zuntz 1956, 136).

The epigram in G 6:8a-c is an original Greek composition, written in an ornate Greek style not typical of G-Proverbs. It was inserted here by association with the ant. It is unlikely that it was created for this context. Whereas Prov 6:6-8 describes the ant as an example of independent enterprise whose efforts benefit itself, the G addition lauds the bee for her value to *others*. The epigram's opening, "Or go to the bee," is an editorial attempt to link it to the preceding parable. The epigram's author is probably not the OG translator. It is unlikely that a translator would pause in the middle of his work to compose a long, not entirely relevant epigram in a different style.

Admiration for the bee is characteristically Greek. Aristotle praises the industriousness ( $\epsilon\rho\gamma\dot{\alpha}\tau\iota\varsigma$ ) of the bee, which he describes immediately after the ant (*Historia animalium* 622b; noted by Gerlemann 1956, 31). Cook (CSP 166) holds that the *hapax*  $\epsilon\rho\gamma\dot{\alpha}\tau\iota\varsigma$  shows that the translator had access to Aristotle. Van der Louw (2007, 278) disputes the parallel and observes that Aristotle mentions the bee alongside the hornet and the wasp as the most industrious animals (622b) and later characterמַאָּכָלֶה: 9 עַד־מָתַי עָצֵל | תִּשְׁבָּב מְׁתַׁי תָּקוּם מִשְׁנָתֶדּ: 10 מְעַט שִׁנוֹת מְעַט תְּנוּמֵוֹת מְעַט | חִבָּק יִדַיִם לִשְׁבְּב: 11 וּבְא־כִמְהַלֵּד באשֶׁד וּמַחְסְרָדָ בְּאַישׁ מְגַן: פּ 12 אָדָם בְּלִיַעַל אַישׁ אָגָן

izes different *kinds* of bees (627a). Nevertheless, the motif of the bee's industry was known and assumed. The G translator too is aware of this motif, even if direct influence from Aristotle is hard to prove. Van der Louw's idea that this is the pharaonic bee (2007, 278) is certainly wrong, because the bee in Proverbs lacks royal features. It works for others, not to aggrandize its own power, and is not a warrior.

The point of praising the bee in Prov 6:8a-c is that its wisdom compensates for its physical weakness (Giese 1992b, 411). This epigram teaches that the combination of work and wisdom can overcome a lack of power. Accordingly, even an impoverished wise man can supplant an ungodly rich one (411). Cook believes that the translator is interpreting a "dualism" between the rich and the poor in a "religious way" (CSP 168). But this contrast is well in the background and cannot be the *purpose* of the addition, nor is there a particularly religious message here.

Ben Sira (11:3) says of the bee: "The bee is the weakest [var: 'smallest'] among flying animals, but her fruit is the best of products of sweet things." Pseudo-Phocylides too uses this topos in emphasizing the bee's industry (ll. 171–174). This immediately follows an epigram on the ant, in ll. 164–170, indicating dependency on G-Proverbs. Sa'adia uses a variant of the bee-topos: "The ants as well as the bees, by nature lacking the faculty of intelligence, gather their food at the appropriate season."

G: (a) ὀλίγον μὲν ὑπνοῖς, ὀλίγον δὲ κάθησαι, μικρὸν δὲ νυστάζεις, (b) ὀλίγον δὲ ἐναγκαλίζη χερσὶν στήθη. "(a) You sleep a while, you sit a while, you drowse a bit, (b) and you embrace your breast with (your) hands a while." Following the cue of the next verse, G uses the second person here, too.

 literary objections to the retroversion (2007, 282) are irrelevant, because the variant is a copyist's error. G treats the verbal nouns in this sentence as second masculine singular verbs. The identical sentence is rendered differently in G 24:33. S also translates the nouns as verbs (2nd masc sg impf) but has nothing corresponding to  $\partial\lambda$ (yov  $\delta$ è xá $\partial\eta\sigma\alpha\iota$ . In 6:10, T = M; in 24:33, T = S.

6:11 fin ] + 2 stichoi G (11*a*) (elab)

G: (a) εἶτ' ἐμπαραγίνεταί σοι ὥσπερ κακὸς ὁδοιπόρος ἡ πενία (b) καὶ ή ένδεια ώσπερ άγαθός δρομεύς. "(a) Then poverty will come upon you like a bad wayfarer (b) and your need like a good runner." By adding "bad" and "good," G creates a contrast, absent in M, between κακός όδοιπόρος ("bad wayfarer"—i.e., a highwayman) and ἀγαθός δρομεύς ("good runner"). This must have happened in the process of translation, since כמהלך רע would not have an appropriate sense. Cook sees this antithesis as further evidence of the translator's penchant for a "dualism of good versus bad" (CSP 171). This penchant is undeniable-for M as well as G. Here, however, the antithesis is superficial and devoid of moral content. The antithesis is also asymmetrical, for the bad wayfarer is not necessarily speedy—and the good runner is presumably good only in his speed—otherwise his visit would not be a threat. The actual relationship between the stichoi is one of heightening: Not only will poverty come unexpectedly (6:11a); it will come quickly (6:11b) (van der Louw 2007, 288). The Hexaplaric sources render etymologically: α' ε' ὡς ἀνὴρ  $\dot{a}\sigma\pi$ ίδος; σ' (ώς)  $\dot{a}$ νήρ χαθωπλισμένος; θ' ώς  $\dot{a}$ νήρ δορυφόρος. See the doublet in 24:34.

S = M but translates איש מגן as איש באיא ("diligent [or 'vigorous'] man") (= T).

### 6:11*a*

G: (a) ἐἀν δὲ ἄοκνος ἦς, ἥξει ὥσπερ πηγὴ ὁ ἀμητός σου, (b) ἡ δὲ ἔνδεια ὥσπερ κακὸς δρομεὺς ἀπαυτομολήσει. "(a) But if you are tireless, your harvest will be there like a fountain, (b) but your neediness will depart like a bad runner." ἥκειν more precisely means "be present," hence, "will be there" (van der Louw 2007, 285). The second part of 6:11*a* has the obelus in SyrH. Fritsch (1953, 179) believes that all of G 6:11*a* is OG, while 6:11 is Hexaplaric. But "bad runner" in 6:11*a*b shows an awareness of the Greek of 6:11b, which itself diverges from the Hebrew. G 6:11*a* plays on the structure and terms of G 6:11 (cf. Tov 1990b, 46). Perhaps misunderstanding 6:11, the addition provides a positive contrast to that

#### 6:12-15

# הוֹלֵדְ עִקְשִׁוּת פֶּה: 13 קֹרֵץ בְּעֵינָו מֹלֵל בְּרַגְלָו מׁרֶה בְּאֶצְבְּעֹתֵיו: 14 תַּהְפָּכוֹת | בְּלִבּוֹ חֹרֵשׁ רָע בְּכָל־עֵת <sup>ו</sup>מְדֹנִים<sup>י</sup> יְשַׁלֵּח: 15 עַל־בֵּן

verse (van der Louw 2007, 291). Here, "bad runner" must be a runner with a bad message; it is not really the antithesis of the "good runner" in 6:11 (who is simply fast). G 6:11*a* is likely a later addition based on a misunderstanding of the preceding verse. V has 6:11*a* in some MSS. S omits.

### 6:12

G: (a) Ἀνὴρ ἄφρων καὶ παράνομος (b) πορεύεται ὁδοὺς οὐκ ἀγαθάς. "(a) A foolish and lawless man (b) goes in ways that are not good."

מיא אדם בליעל אדם גמו האדם גמעסעסק combines the two phrases אדם בליעל and אדם בליעל. (In G-Proverbs, בליעל is consistently rendered מאָסשע, one of the translator's favorite words.) For M's "goes (in) crookedness of mouth," an idiom not in Greek, the translator substitutes a more predictable image: הסףביטבוג אסאר מאמאל ("goes on paths that are not good").

S summarizes עקשות פה ("in oppressiveness"), a word that nicely combines the ideas of oppression and slander. This seems like a genuine case of exegetical metathesis or simply of homoiophony.

6:13 בעינו M<sup>K</sup> G (ὀφθαλμῷ) ] בעינו M<sup>Q</sup> S (הבינסת,) V (oculis) T (בעינוי) (orth)

ברגלו M<sup>K</sup> G (ποδί) S (פוֹ אָליי) V (pede) ] ברגלו (num or orth, equal)

The two *qerayin* in this verse are marked in M<sup>A</sup> but not (contrary to BHS) in M<sup>L</sup>. Mp<sup>L</sup> marks שרגלי as  $\neg \Box \Box$  (*defectiva*), meaning that it regards the word as a plural. G treats it as singular, S as plural. G and S saw ברגלי and understood it as singular. In both cases the *ketiv* is a defective writing of the plural; see the comments on 2:8 and 6:3. The gesture involves the singular eye in Prov 10:10a and Ps 35:19 but the plural eyes in Prov 16:30. G associates the *hapax* מֹלָל ("shuffle") with the common Aramaic מֹלֵל ("speak"), hence σημαίνει "signs."

In 6:13c, διδάσχει δὲ ἐννεύμασιν δακτύλων ("He teaches with hints of his fingers") assumes that מרה means "teaches," though it actually means "points." Since it would not make sense to say "teaches with his fingers," the translator must add a seemingly appropriate modifier: "with hints." This is another example of how G's apparent freedoms are often attempts to deal with a difficult or misunderstood Hebrew text. The translation implies that the scoundrel teaches others his wicked ways.

6:14 מִדְיָנָים [ M<sup>K</sup> מִדְיָנָים M<sup>Q</sup> (dial? norm)

The *ketiv* מדנים and the *qere* מדנים are variant forms of the frequent pair *ketiv* מדנים *qere* מדנים, which occurs in Prov 18:19; 21:9 (see comment); 21:19; 23:29; 25:24; 26:21; and 27:15. מדנים appears alone in 6:19 and 10:12. (The Oriental reading is מדינים for both the *qere* and the *ketiv*.) The plural of this word is a peculiarity of Proverbs. (The singular occurs in 2 Sam 21:20.) The only form of the singular is (האור), which does not have a plural outside *ketiv-qere* pairs. The plural is (occurring in the above-listed *ketiv-qere* pair as well as in Prov 19:13). The reading מדינים appears in the body of the text of many MSS KR in all of the above-listed verses. This indicates that the *qere* is a variant active in manuscript transmission. The pair is too frequent to be explained as scribal error in either direction, but its cause and significance have not been accounted for.

Given the consistency with which the *ketiv* uses מדונים and the *qere* מדינים, the likely explanation is dialectal. I would hypothesize that the form מדונים, produced simply by pluralizing the singular מדונים, belongs to a colloquial register, while the literary form was מדינים (similar to the distinction between "formulas" and "formulae," "cherubs" and "cherubim," and the like). I maintain the *ketiv* forms (מדונים and the eclectic text. Given the rarity of these plurals, I am inclined to consider them the earlier forms, with מדינים a normalization. As for vocalization, as a plural of מִדונים, should be vocalized מִדונים is either than assume a third form of the plural.

It is not usually possible to determine which reading the versions represent, because either the *ketiv* or the *qere* could be rendered by words for strife and offensiveness. Sometimes, as here, it is possible to determine the underlying Hebrew.

G renders 6:14c as ὁ τοιοῦτος ταραχὰς συνίστησιν πόλει ("such a one causes troubles to a city"). "Troubles" and "to a city" are a doublet, both translating the *qere* מדינים. The first renders it "troubles"; the second associates it with מדינים in its Aramaic sense, "city." Certainly the translator knew what מדינים means. The second translation is not a mistake but a midrash. It locates the events mentioned in this verse in the public sphere, in distinction to 6:19, which speaks of dissension in the private sphere. The translator seems to have in mind the ταραχάι that often wracked Alexandria (van der Louw 2007, 300).

S הכאמפרו ("and he turns about [in his heart]"), treats ההפכות as a verb, thereby connoting scheming or unreliability rather than per-

אָרָאָם יָבָוֹא אֵידָוֹ אָּתַע יִשְׁבָר וְאֵין מַרְאָּא: מַ 16 שָׁשׁ־הָנָּה שְׁנֵא יְהוֶה וְשָׁבַע <sup>ר</sup>ּתּוֹעֲבַת<sup>י</sup> נַפְּשְׁוֹ: 17 עֵינַיִם רָמוֹת לְשׁוֹן שָׁקָר וְיָדַיִם שׁפְּכְוֹת <sup>ר</sup>ַדַם<sup>י</sup>־נָקִי: 18 לֵב חֹבשׁ מַחְשְׁבַוֹת אֶוֶן רַגְלָיִם מְמַהַרוֹת לְרָוּץ לֶרְשָׁה: 19 יָפִיַח בֶּזָבִים עֵד שֶׁקֶר וּמְשַׁלֵח מְדָנִים בֵּין אַחִים: 20 נִצְׂר בֵּנִי מִצְוַת אָבֵידְ וְאַל־תִׁטֹּשׁ תּוֹרַת אָמֶדָ: 21 קַשְׁרֵם עַל־לִבְּדָ

versity. S-Prov 2:16 says of the Strange Woman that she מספבא השנה ("turns her words about"). To מדנים ישלח S adds בעל האדט ("between two people") for clarity.

6:15 אישבר M S (נאגאבו (נאגאבר G (אמו סטעדףוβή) (graph  $' \rightarrow 1$ )

Having שבר in 6:15b, the translator saw a sentence that on the face of it meant "Suddenly and a breaking and there is none to heal." The difficulty of this text, and the apparent pairing with "breaking," led him to assign a unique meaning to בתע, "breach," "wound," perhaps by association with בתע ושבר (BHQ). Elsewhere διακοπή renders p or p or c. There is no reason to assume that בתע ושבר was the translator's misreading (van der Louw 2007, 301) rather than a scribal error.

6:16 ושבע M S (ה.ברב) א ושבר (מנשבע הסנשבע העבע א G האברט) (graph  $\mathcal{Y} \to \mathcal{T},$  paleo  $\mathcal{P} \to \mathcal{A})$ 

תועבות M^Q G (מאמאמאסילמע)  $\approx$  S (המלאה MK (num, orthog) תועבות MK (num, orthog)

G: (a) ὅτι χαίρει πᾶσιν, οἶς μισεῖ ὁ κύριος, (b) συντρίβεται δὲ δι' ἀκαθαρσίαν ψυχῆς. "(a) For he rejoices in all things that the Lord hates, (b) and he is shattered through impurity of soul." G 6:16–19 continue the description of the villain begun in 6:12.

Instead of שבע, G read ושבר, which he parsed as וְשָׁבַר ("shattered"). The letters were similar in some varieties of the archaic script:  $A/\sigma$ . Other examples are 3:10 (שבר/שבע); 19:27 (שבר/שמע/לשמר)—but uncertain); 15:4 (שבר/שבע); and possibly 8:3. BHQ explains the interchange here as phonetic (applying this explanation only to 15:32 and 19:27). The problem with the phonetic explanation is that some of the examples involve  $\psi/\psi$  confusion (exhibited also in  $\psi\psi/\psi\psi$  in 6:16), which requires that the words were read visually, not heard. For possible examples of  $\gamma/\gamma$  interchange, see LSF §131.

Since his source text had ושבר, the translator assumed that 6:16b refers to the wicked man, for it is he who can be said to be "shattered." There-

fore הועבת נפשו ("the loathing of his [sc. God's] soul") must describe the wicked man, the object of God's loathing.

Lacking "seven," G did not recognize  $\mathcal{W}\mathcal{W}$  as a numeral but construed it as  $\mathcal{W}\mathcal{Y}$  ("he rejoices"). Van der Louw believes that the translator overrode the number sequence to avoid the suggestion that the Lord hates (only) six things, "whereas the Torah makes it clear that there are many more things He hates" (2007, 304). But to say that there are six things God hates would not preclude others, any more than Prov 30:18 implies that *only* four things are beyond human understanding. All these differences in G were triggered by a slight scribal error, not by interpretive choices (such as van der Louw tries to trace). The translator will often ignore syntactical constraints in favor of representing the individual word meanings as he understands them; see the note on 5:5.

S had הועבת, which he construed as הועבת, hence: אסערא ("loathes"); thus T and V. Several MSS KR have the *defectiva* writings, including and העבת and הועבת. The writing הועבות probably arose as an adjustment to the plural subject.

נגבא וביא) (vocal) דָּם־נָקָי א S (גרא וביא) (גרא וביא) (vocal)

Whereas M's vocalization דָם־נָקי construes the phrase to mean "innocent blood," G's מוֹש אסגמיט understands it as a bound form, דַם־נָקי ("blood of the innocent"), which is probably correct; see ABP 1.85.

# 6:18

G combines ממהרות and לרוץ into the intensified  $\underline{\acute{e}\pi\iota}\sigma\pi\epsilon\dot{\upsilon}\delta\sigma\nu\tau\epsilon\varsigma$  ("rushing") (van der Louw 2007, 310).

### 6:19

To distinguish this verse from similar ones about מדינים, G situates the present warning in the context of private strife—אָרְעָנים, meaning "lawsuits" (van der Louw 2007, 311–13). See the comment on G 6:14.

### 6:20

G: (a) Υίέ, φύλασσε νόμους πατρός σου (b) καὶ μὴ ἀπώσῃ θεσμοὺς μητρός σου. "Son, keep the laws of your father, (b) and do not reject the rules of your mother." G treats πιεπαι a plural to provide an antecedent for  $\Box(\neg \varphi \nabla)/\alpha$ ύτούς in the next verse in spite of the gender incongruity in the Hebrew (van der Louw 2007, 314). Since the singular νόμος is used for parental instruction unequivocally in 4:1 and 13:14 and sometimes elsewhere, the human origin of the teaching in this verse is not the reason G

תָמֵיד עָׁנְדֵם עַל־גַּרְגְּרֹתֶדּ: 22 בְּהִתְהַלֶּכְדָּ | תַּנְשֶׁה אֹתָדְ בְּשָׁכְבְּדָ תִּשְׁמִר עָלֶיֶד וַהַקִיצוֹת הֵיא תְשִׂיחֶדּ: 23 כִּי גַר מֵצְוֶה וְתוֹרָה אֵוֹר וְדֶרֶךְ חֵׁיִּים תּוֹכְחֵוֹת מוּסֵר: 24 לֵשְׁמָרְדָ מֵאֵשֶׁת <sup>ו</sup>רֵעֶדְ<sup>ז</sup> מֵחֶלְלֵת

renders it as plural (contrary to van der Louw 2007, 314). In fact, there is little to choose between M's vocalization of the word as a singular and G's as a plural. In any case, the teaching(s) here cannot be distinguished from the  $\pi$  called νόμος in the singular in 6:23. On G's parallelism between the νόμος of the father and the less authoritative θεσμός of the mother, see ABP 1.368.

### 6:22

G: (a) ἡνίκα ἂν περιπατῆς, ἐπάγου αὐτήν, καὶ μετὰ σοῦ ἐστω. (b) ὡς δ' ἂν καθεύδῃς, φυλασσέτω σε, (c) ἵνα ἐγειρομένω συλλαλῆ σοι. "(a) Whenever you go about, lead it and let it be with you. (b) Whenever you lie down, let it guard you, (c) so that when you are awake, it may converse with you."

ἐπάγου αὐτήν = תְּנָחָה (≈ S). αὐτήν refers to wisdom, implicit in the context, and not necessarily to any particular word for it, such as ἐντολή in the next verse. G understands אתן as "with you" (אָתָד), which it expands into καὶ μετὰ σοῦ ἔστω to explain in what sense "you"—the boy being addressed—can "lead" wisdom. Having parsed the third feminine singular verb wrongly as second masculine, the translator adjusted the rest of the verse to his preconception.

In G, the imperatives make the verse into a continuation of the exhortation rather than a promise of reward. The addition of  $i\nu\alpha$  in 6:22c implies that the *purpose* of Wisdom's protection is her "conversation" with her disciples, this being the study of wisdom. Wisdom is a reward in and of itself; see the comment on 2:2. V: *gradiantur tecum* ("let them go with you") = G.

nent to context: lying down and rising up. In 6:22c, S's העיה ("meditation," "object of meditation") associates Hebrew תשיחך with one sense of תיתדבר עמך ("conversation," "meditation"). T<sup>L</sup>: תיתדבר עמך ("it will accompany" [lit. 'be led'] with you"). T<sup>L</sup> realizes that there is only one verb in this clause but still assumes that אתך means "with you." T<sup>Z</sup> תדבר עמך ("lead [her] with you") adjusts to M's תַּנְתָה but still implicitly vocalizes אָתָד (though the editor undoubtedly was working from a vocalized text). Since T<sup>L</sup> is adjusting further to M, it is secondary. In any case, context and parallelism favor M's vocalization.

### 6:23

G: (a) ὅτι λύχνος ἐντολὴ νόμου καὶ φῶς, (b) καὶ ὁδὸς ζωῆς ἔλεγχος καὶ παιδεία. "(a) For a commandment of law is a lamp and a light, (b) and reproof and instruction are the way of life." The expected and natural translation of M is "For the precept is a lamp and the teaching a light, and the reproof of discipline is the way of life." G construes the verse in a forced fashion. If retroverted precisely, G 6:23a would be כי נר but to retrovert to this would, מצות תורה ואור ודרך חיים תוכחת ומוסר require assuming a number of unmotivated variants. CSP suggests that the restructuring, especially in ἐντολή νόμου, is the translator's attempt to make a clearer reference to the Mosaic law. However, such an allusion could be seen in M, too, and subsequent Jewish interpreters did just that. Van der Louw (2007, 320) says that G reformulated the sentence in order to avoid an awkward literal translation: ὅτι λύχνος ἐντολὴ νόμου καὶ φῶς. However, the expected literal translation would be ὅτι λύχνος ἐντολή καὶ νόμος φῶς, which is a compound verbless sentence no more awkward than G's 6:23a as it stands. The reasons for G's variations from M in both parts of the sentence remain unclear.

# 6:24 העיד א S (הידאאה) (near haplog מ $\rightarrow$ העיד) ( $\alpha \leftarrow \alpha$

Though M's "woman of evil" is grammatically possible, this is not the way אשת רְעָדְ is used elsewhere in Proverbs. We should read מאשת רְעָדָ ; note the possessive pronoun in 6:29 and compare Exod 20:17. The emendation provides a better parallel to נכריה. M's vocalization would have been occasioned by the loss of  $\Box$  by haplography with the similar-looking  $\omega$  of the following word.

לְשִׁוֹז נְרָרִיֶּה: 25 אַל־תַּחְמִּד יְיִפְיָה בִּלְבָבֶדְ וְאַל־תִּקְחָדְ בְּעַפְעַפֶּיהָ: 26 כִּי בְעַד־אִשְׁה זוֹנָה עַד־כִּפַּר לָחֶם וְאָשֶׁת אֵישׁ גָפָשׁ יְקָרָה תְצְוּד: • 72 הַיַחְתֶּה אִישׁ אֵשׁ בְּחֵיקוֹ וּבְגָדִיו לָא תִשְּׂרַפְנָה: 28 אִם־יְהַלֵּךְ אֵישׁ עַל־הַגֶּחָלֵים וְרַגְלָיו לָא תִכָּוֶינָה: 29 כֵּז הַבָּא אֶל־אֵשֶׁת רַעֵּהוּ לָא יִנָּקָה בְּל־הַנֹּגַע בְּה: 30 לֹא־יְבַוּזוּ לַצַּנְּב בְּי יִגְגָוֹב לְמַלֵּא נַפְשׁוֹ בִי יִרְעֵב: 31 וְנִמְצָא יְשַׁלֵם שִׁבְעָתָיִם

"from a woman/wife of evil" = "from an evil woman/wife"). G's vocalization fits the context better, but a possessive is still needed.

G understands חלקת ("smoothness") as διαβολῆς—slander rather than flattery or seduction. G is perhaps influenced by the homonym "division" (van der Louw 2007, 322). Various idioms using חלק in regard to speech are associated with dishonesty (G-Ps 5:10; 12[G 11]:3, 4). In Prov 28:23, however, G uses λωσσοχαριτοῦντος ("flattering").

S אואאה ("evil woman") = M ≠ G. S also understands חלקת as slander (האבל מו<sub>צ</sub>ה).

### 6:25

G: (a) μή σε νικήση κάλλους ἐπιθυμία, (b) μηδὲ ἀγρευθῆς σοῖς ὀφθαλμοῖς (c) μηδὲ συναρπασθῆς ἀπὸ τῶν αὐτῆς βλεφάρων [G<sup>BSAMSS</sup>; μηδὲ συναρπασάτω σε τοῖς βλεφάροῖς αὐτῆς G<sup>V 252</sup>]. "(a) Let not the desire of beauty overcome you, (b) nor let yourself be caught by your eyes, (c) nor let yourself be seized by her eyelids [var: 'nor let her catch you by her eyelids']." In 6:25a, G is deliberately paraphrastic. The Hebrew would have presented the translator with no difficulties. Whereas M demands that the listener "not covet her beauty," G assumes that the desire is a reality, and though you cannot prevent the *feeling* of desire, you *can* refuse to let it control your actions. Similarly, the Mekilta to Exod 20:17 interprets the tenth commandment as prohibiting the *carrying out* of forbidden desires. G has "your eyes" in 6:25b because it is concerned with the male's desire, as G 6:25a shows, rather than the woman's wiles.

Although SyrH gives 6:25b the obelos, it is more likely that 6:25b is OG and G 6:25c a later revision toward M (de Lagarde), with  $\sigma\nu\nu\alpha\rho\pi\alpha\sigma\theta\tilde{\eta}\varsigma$  in diathesis. CSP believes that either the original translator or a later hand added 6:25c as an explication of the second line, but that is not what it does.

S: אבריאס א אלם היא אלם אדריאס א אלים איביא האריביג בעוניס א אלים א אריביא (a) Do not desire her beauty in your heart, (b) and let her not trap you with her eyes, (c) and let her not capture you with her eyelids." S again maneuvers between M and G. The second translation in 6:25c is based on the variant in  $G^{V 252} = M$  (BHQ).

# 6:26

S المحين ("the likeness [of a harlot]") should be emended to الاحدين ("the price [of a harlot]") (Pinkuss).

### 6:30

G: (a) οὐ θαυμαστὸν ἐἀν ἀλῷ τις κλέπτων, (b) κλέπτει γὰρ ἵνα ἐμπλήσῃ τὴν ψυχὴν πεινῶν [G<sup>A MSS</sup>; var: πείνωσαν G<sup>BSV</sup>]. "(a) It is not surprising if one who steals is caught, (b) for he steals in order to fill his soul, when he hungers [var: 'when it (the soul) is hungry']." "Is caught" is imported from the next verse to improve the logic, since a thief must be caught before people take note of him. Baumgartner postulates that G had 'C C' κίcτ', a corruption of L' C' κίετ however, the translator who is introducing a moralism here. He adds the notion of the thief's being caught and changes the public's lack of contempt into lack of surprise. G did not want to allow for the possibility that a thief could ever avoid capture and disgrace.

S follows G, with κείνωσαν.

### 6:31

G: (a) ἐἀν δὲ ἀλῷ, ἀποτείσει ἑπταπλάσια, (b) καὶ πάντα τὰ ὑπάρχοντα αὐτοῦ δοὺς ῥύσεται ἑαυτόν. "(a) But if he is caught, he will repay sevenfold, (b) and he will save himself by giving all his possessions." M's "all his wealth" was considered too grand for this context, so the translator rendered mas "all his possessions" and dropped "his house" as (consequently) redundant. G adds "will save himself" to make the contrast with the adulterer explicit: the thief, but not the adulterer, can save his life (van der Louw 2007, 334).

אָת־כְּלֹ־הָוֹז בִּיתוֹ יִהֵז: 32 נֹאָף אִשָּׁה חַסַר־לֵב מַשְׁחִית נַפְשׁוֹ הַוּא יַעֲשֶׂנָּה: 33 נֶגַע־וְקָלָוֹז יִמְצֵא וְׁחֶרְפָּתׁוֹ לָא תִמְחָה: 34 כִּי־קִנְאָה חַמַת־גָּבֶר וְלָא־יַחְמוֹל בְּיוֹם נָקֵם: 35 לֹא־יָשָׂא פְּנֵי כָל־כְּפֶר וְלָא־ יֹאבֶה כֵּי תַרְבֶּה־שְׁחִד: פּ זוּך בֵּנִי שְׁמִר אֲמָרֵי וֹּמִצְוֹתַי יֹאבֶה כֵּי תַרְבֶּה־שְׁחִד: פּ זוּך בֵּנִי שְׁמִר אֲמָרֵי וֹּמִצְוֹתַי תַּצְפָן אִתְּדָ: 2 שְׁמִר מִצְוֹתַי וֶחְיֵה וְתוֹרָתִי בְּאִישִׁוֹז עֵינֵידָ: 3 קָשְׁרֵם עַל־אֶצְבְּעָתֶידְ כָּתְבֵם עַל־לְוּחַ לְבֶּדָ: 4 אֶמְר לַחְכְמָה אֲחָתִי אֶתְ וֹמֹדֶע לַבִּינֵה תִקְרֵא: 5 לָשְׁמְרָדְ מַאִשְׁה זְרֵה מְנָּכִריָּה אֲמָרֵיהָ

# 6:32

G condenses this verse into a single stich and identifies the direct object of משחית as עשנה/משחית יעשנה/משחית as עשנה/משחית data v rather than the act of adultery (van der Louw 2007, 334). V = G.

## 6:34

According to van der Louw (2007, 337–40), G 36:32–35 envisions a trial before a formal court—a  $x\rho$ i $\sigma$ i $\varsigma$  rather than  $\Box \zeta \Box$ —that punishes the adulterer but does not kill him. Even if the trial itself could be settled by compensation (unlike in M), the husband's wrath would never be assuaged.

7:1 fin ] + 2 stichoi G (1*a*) (elab)

### 7:1a

G: (a) טוֹצָ, דוֹעָם דָטָע אַטָּרָטָט, אָמוֹ וֹסְעָטָסָבּוּגָ, (b) אָאָש אָרָאָש אָרָ אָסָרָסָטָ מָאָ אָסָרָסָטָ מֹאָאָריי (a) Son, honor the Lord and you will be strong, (b) and besides him fear no other." This addition interrupts the exhortation with an irrelevant thought and is secondary. Notably, M-Proverbs never mentions the possibility of worshiping other gods. The elaboration may have been introduced in Greek or Hebrew, perhaps as בבד את יהוה וחזק כבד את יהוה וחזק. The idea is frequently expressed, but see especially Isa 44:8 for the sentiment.

### 7:2

For M ותורתי in 7:2b, G has דסטי אל געסטי ("my words"), as in 4:10. This is a synonym variant that could occur in either the Greek or the Hebrew (as אמרי דברי); cf. 3:1b.

### 7:3

Instead of "on your fingers," S has رون ("on your neck"), assimilating to 3:3b, because "neck" is used in similar imagery in 3:22 and 6:21.

# 7:5

G (continuing 7:4): (a) ἵνα σε τηρήσῃ ἀπὸ γυναικὸς ἀλλοτρίας καὶ πονηρᾶς, (b) ἐάν σε λόγοις τοῖς πρὸς χάριν ἐμβάληται. "(a) so that it may keep you from a woman strange and wicked, (b) if she should assail you by words (meant) to charm." Since G does not have a distinct equivalent for μCr<sup>i</sup>π j, it avoids repetition of ἀλλοτρίας by replacing it with a modifier that underscores the woman's corruption: καὶ πονηρᾶς. (Γις μαραβαραματίας) is handled in various ways, with ἀλλότρια // ἀλλότρια only in 23:27.) G paraphrases 7:5b, emphasizing that the woman's smooth words are really an assault. The same sentence is handled very differently in 2:16b.

# 7:6-17

G: The scene of the Strange Woman in Prov 7 is conceived differently in G. Many differences from M, especially in 7:6–17, stem from the translator's notion that it is the Strange Woman, not the male teacher (the father), who is looking out the window. The woman in G's scene goes forth from her house (though this is not stated) and meets the passerby in 7:10.

G. Boström (1935, 120–23) proposes that the woman looking out the window is Aphrodite Parakyptousa, the Aphrodite who leans out the window. In fact, the woman is not that goddess—isn't even modeled on her—for the speaker in G 7:6–17 is hidden behind a lattice and is not on view to passersby. Boström believes that G preserves elements of the original text, while M was changed for the sake of discretion, to avoid revealing the Strange Woman as the Aphrodite figure. (Boström makes several emendations to achieve this effect.) But if the author was warning against the Aphrodite cult, why would later scribes *obscure* the danger it presents?

Few of G's divergences here can be ascribed to textual processes; in any case, variants are very hard to recover because of the paraphrastic nature of the translation in this passage. G's interpretation of the chapter is discussed in detail in ABP 1.403–6.

S: Throughout this section S shows strong G influence but differs in many particulars. In spite of having recourse to M, S has the *woman* looking through the window in 7:6–8. T, as usual, uses S as the basis and adjusts toward M, in this case restoring the first-person narrator. V does

הֶחֶלִיקָה: 6 בֵּי בְּחַלֵּוֹן בֵּיתֵי בְּעָד אֶשְׁנַבְּי נִשְׁחֵפְתִי: 7 וָאֵֶרָא בַפְּתָאיִם אָּבִינָה בַבְּנִים נַעַר חֲסַר־לֵב: 8 עֹבֵר בֲשׁוּק אֵצֶל פִּגָה וְדֶרָדְ בֵּיתָה יִצְעֲד: 9 בְּנֵשֶׁר<sup>-ר</sup>בַּעֲרָב<sup>-</sup> יָוֹם בְּאִישִׁוֹן לַיְלָה וַאֲפֵלֶה: 10 וְהִנֵּה אֲשָׁה לִקְרָאתוֹ שִׁית זוֹנָה וּנְצֵרַת לֵב: 11 הֹמִיָה הֵיא וְסֹרֶרֶת בְּבֵיתָה לֹא־ יִשְׁפְּנָוּ רַגְלֵיהָ: 12 פַּעַם | בַּחוּץ פַּעַם בְּרְחֹבְוֹת וְאֵצֶל כְּל־פִּנְה תֶאֱרָב:

the same. For the details of this adaptation see ABP ad loc. Since this section is tightly knit, I will translate all of G to provide context. Note that sentences often continue from one verse to the next.

### 7:6

G: (a) ἀπὸ γὰρ θυρίδος ἐκ τοῦ οἴκου αὐτῆς (b) εἰς τὰς πλατείας παρακύπτουσα. "(a) For from the window, out of her house, (b) she peers onto the plazas." G defines the perspective as the Strange Woman's.

S: متل مح حصوب . "For from the window of her house and from the balcony she looked out." (منه حصوب can mean "upper room" as well as "balcony.")

### 7:8

G: (a) παραπορευόμενον παρὰ γωνίαν ἐν διόδοις οἶκων αὐτῆς. "Passing by the corner on the thoroughfares of her houses." The plural "her houses" is used of the woman's dwelling in 5:8 and of the domain of the Woman of Strength in 31:27, in reference to a stately residential complex. G condenses the verse and collapses the parallelism. G does not recognize the suffix on the unusual פָּנָה (on which see GKC §91*e*, though GKC is uncertain on the point). עולד (M 7:8b) is used in the next sentence (and subsequently corrupted).

S (continuing 7:7): האבדי, בשלה גער ומילא גער מילא גער (מילא גער): "who pass in the street at the corners of the paths." S also joins איצעד to the next sentence, though it handles this verse very differently from G and assumes that the text continues to describe the plurality of youths mentioned in 7:7.

# רכבו (vocal) (ברבי א G (έσπερινῷ) S (ברבי ( < בַּעֲרֹב ( אַבָּעָרָב )

G: (a) καὶ λαλοῦντα [corr: ἀλύοντα] ἐν σκότει ἑσπερινῷ, (b) ἡνίκα ἂν ἡσυχία νυκτερινὴ ἦ καὶ γνοφώδης. "(a) And speaking [corr: "wandering"]

in the evening darkness, (b) when (there is) nocturnal silence and darkness."

According to BHQ, λαλοῦντα is used for variety. However, that verb is a masculine accusative participle and would refer to the object of the woman's "seeing" in 7:7, namely, the boy. This makes it appear that he is speaking, though his words are not quoted and have no further relevance. He is not "conversing" (NETS) or "calling," "uttering a cry" (BHQ). Instead, we should emend to ἀλύοντα ("be fretful, distraught," Jäger). This renders "צעד" in a way that suggests the woman's agitation.

M's vocalization אָבְשֶרֶב יוֹם (an unparalleled phrase) means "at the evening of day." A more natural vocalization is בַּעֲרֹב יוֹם (lit. "when the day was becoming evening"). For the locution, see Judg 19:9.

7:10

G: (a) ή δὲ γυνὴ συναντῷ αὐτῷ, εἶδος ἔχουσα πορνικόν, (b) ἡ ποιεῖ νέων ἐξίπτασθαι καρδίας. "(a) The woman meets him, having a whorish appearance, (b) which makes the hearts of youths fly off." G 7:10b is a guess at μεντα de la cutually, "guarded of heart"). The translator probably found the phrase illogical because "guarding" (נצרו לב) one's heart should be a virtue (4:23). ἐξίπτασθαι = ἐκπέτασθαι ("fly away," "flee"). For the youths' hearts to fly off means that they lose all sense.

S: הבועל המפוז האשר המסוב האשר לאמיה אבססי הבוער המפוז היישר האשר המסום הבוער (a) And a woman comes out to meet him, in the appearance of a harlot, (b) who [or 'which'] makes the heart of youths flee." S follows G's guess, with no reference to M. האשר ("scatters" or "disturbs") is an interpretation of the hapax ἐξίπτασθαι, not a rendering of גערת phrase אשר הבי היים occurs also in S 22:15; see the comment. T  $\approx$  S. Sa'adia translates the verse similarly (as noted by BHQ), but this comes via T.

7:12

13 וְהֶחֶזִיקָה בְּוֹ וְגָשְׁקָה־לֵּוֹ הֵעָזָה פְׁנָּיָה וַתַּאֹמַר לְוֹ: 14 זִבְחֵי שְׁלָמִים עָלֵי הַיּוֹם שִׁלַמְתִי נְדְרֵי: 15 עַל־בֵּן יָצָאתִי לִקְרָאתֶדְּ לְשַׁחֵר פְּנָיד וָאֶמְצָאֶדְ: 16 מֵרְבַדִּים רָבַדְתִי עַרְשֵׁי חֲטָבׁוֹת אֵטְוּז מִצְרֵים: 17 נַפְתִי מִשְׁכְּבֵי מִר אֲׁהָלִים וְקַנְּמוֹז: 18 לְכֵה נִרְזֶה דְדִים עַד־הַבְּקֶר גְּתְעַלְּסָה בָּאֲהָבְים: 19 בִּי אֵיז הָאֵישׁ בְּבִיתֵוֹ הָלַדְ בְּדֵרֶד מֵרְחוֹק: 20 צְרוֹר־ הַכֶּסֶף לָקַח בְּיָדִוֹ לְיוֹם הַבִּסָא יָבָא בֵיתְוֹ: 11 הַטַּתוּ בְּרָב לִקְחֵה

S follows G but omits "every" because, logically, the woman cannot lie in wait at *every* corner. T<sup>L</sup> seems to be dealing with the same problem differently, translating not simply "at every corner" but כל פנתא דיאיא ("at every *suitable* corner")  $\approx T^{Z}$ .

7:13

G: (a) εἶτα ἐπιλαβομένη ἐφίλησεν αὐτόν, (b) ἀναιδεῖ δὲ προσώπῷ προσεῖπεν αὐτῷ. "(a) Then she seized and kissed him, (b) and with a shameless face she spoke out to him."

### 7:14

G: (a) Θυσία εἰρηνική μοί ἐστιν, (b) σήμερον ἀποδίδωμι τὰς εὐχάς μου. "(a) I have a peace offering; (b) today I pay my vows." The present tense indicates that she is about to bring the offerings.

S omits עלי.

### 7:15

G: (a) ἕνεκα τούτου ἐξῆλθον εἰς συνάντησίν σοι, (b) ποθοῦσα τὸ σὸν πρόσωπον εὕρηκά σε. "(a) For this reason I went forth to an encounter with you. (b) Longing for your face, I found you." G has a bipartite structure lacking in M.

S: אבעולא שבאר מישא לבעור אומא אבעולא מאבעילים אומא אראש. (a) For this (reason) I went out to meet you, (b) for I was expecting to see you, and I found you." S also creates a bipartite verse.

T 7:15b: "for I was expecting to perceive [לרגשא] (you) and I said, I will guide you [אידברינך]." T presents the woman as an alternative teacher who will lead the youth away from Wisdom. 7:16

# 7:17

G: (a) διέρραγκα τὴν κοίτην μου κρόκω, (b) τὸν δὲ οἶκόν μου κινναμώμω. "(a) I have sprinkled my bed with saffron, (b) my house with cinnamon." The translator, thinking that wcr (actually: "aloes") means "tents," and knowing that the woman does not live in one, translates it as "house" (as happens also in G-Gen 9:21, 27; 31:33; etc.) and ignores the plural for the sake of tighter parallelism with τὴν κοίτην μου.

### 7:18

G: (a) ἐλθὲ καὶ ἀπολαύσωμεν φιλίας ἕως ὄρθρου, (b) δεῦρο καὶ ἐγκυλισθῶμεν ἔρωτι. "(a) Come and let us delight ourselves with lovemaking till morning; (b) come and let us roll up in love." εγκυλίειν has the abstract sense of "be involved in," but the etymological connotation of physically *rolling* is vividly germane here. עלט is not understood thus elsewhere (see Job 20:18; 39:13), nor is the cognate y.

V translates דדים as *uberibus* ("breasts").  $\alpha', \sigma',$  and  $\theta'$  have  $\tau i \tau \theta \tilde{\omega} \nu$ ; cf. the comment on 5:19. S understands it, correctly, as הערא" (love-making).

### 7:20

G: (a) ἐνδεσμον ἀργυρίου λαβών ἐν χειρὶ αὐτοῦ, (b) δι' ἡμερῶν πολλῶν ἐπανήξει εἰς τὸν οἶκον αὐτοῦ. "(a) He has taken a bag of silver in his hand; (b) after many days he will come back to his home." G's "after many days" is a contextual guess for the rare ROS. α', correctly: εἰς ἡμέραν πανσελήνου ("at the day of the full moon"). V in die plenae lunae.

בְּחֵלֶק שְׁפָּגֶיהָ תַּדִּיחֶנּוּ: 22 הָוֹלֵדְ אַחֲרֶׁיהָ פְּׁתְאָם בְּשׁוֹר אֶל־(טֶבַח' יְבְוֹא 'וּכְעָכֹס' אֶל־'מוֹסֵר אַיֶּל': 23 ' ' כְּמַהֵר צִפּוֹר אֶל־פֶּח וְלְא־ יְּדֵע כְּי־בְנַפְשָׁוֹ הוּא 'עַד יְפַלַּח חֵץ כְּבֵדְוֹ': פּ 24 וְעַתָּה

### 7:21

G: (a) ἀπεπλάνησεν δὲ αὐτὸν πολλῃ ὁμιλία, (b) βρόχοις τε τοῖς ἀπὸ χειλέων ἐξώχειλεν αὐτόν. "(a) She led him astray by much conversation, (b) and by snares from (her) lips she drove him aground." (ὁμιλία can also mean "instruction.") G replaces the benign-sounding חלק ("smoothness") with the scarier image of traps, perhaps influenced by the imagery (but not the wording) of 6:2. The translator had already eliminated smoothness of speech from the woman's features in 7:5. The motif of the seductress's smooth speech is lacking also in G 2:16. It seems that the translator thought that verbal smoothness was not an unambiguous indicator of danger and deceit, because mellifluous rhetoric can be a worthy talent.

7:22 אַבָח  $M^A$ ] אַבָח  $M^L$  (graph  $\mathcal{Y} \to \mathcal{Y}$ )

<דֹכְעָבֹס [ <וּבְעָבֹס M (vocal)

מוֹסָר ( לאים א נוֹסָר ( לאים א G (פֿתו א נוּסָר S (גאים א מוֹסָר M (vocal)

איל G (<br/> לגמקסג in G 23) S (הער in S 23) ] M אויל (near dittog ' $\rightarrow$ י<br/>, context)

M is incomprehensible. The various attempts to translate without emendation are along the lines of the KJV: "or as a fool to the correction of the stocks." David Qimhi and other traditional commentators paraphrase: as they bring a fool running to prison and put him in chains [עכס]. But all such proposals are unwarranted by the Hebrew. Following a widely accepted emendation, supported by G, we can translate "like a stag [איל] bounding [אָרָטָכס]! to bonds [מֹנַסַר] " This is in harmony with the other similes. For  $v_{CO}$  meaning "prancing" or the like, see Isa 3:16.

G: (a) δ δὲ ἐπηκολούθησεν αὐτῆ κεπφωθείς, (b) ὥσπερ δὲ βοῦς ἐπὶ σφαγὴν ἄγεται (c) καὶ ὥσπερ κύων ἐπὶ δεσμοὺς. "(a) And he followed her, begulled, (b) as an ox is led to slaughter, (c) and as a dog to bonds." G resulted from M by the following steps:

(1) κεπφωθείς for פתאם ("suddenly"): The κέπθφος is a pigeon or seagull, so English "be gulled" works precisely. To arrive at this sense, the translator probably associated פתאם (whose meaning he knew; see 6:12) with פתאים ("callowness," "naiveté") and translated it by an elegant

word that suggests the youth is as naive as a bird. For this motif, see the next verse and Prov 1:17; Hos 7:11; and Qoh 9:12.

(2) ש័σπερ ציטע להו לצסעטיג: For the obscure עכס, the translator substituted a Greek cliché, "a dog to bonds" (Gerleman 1956, 33). להו לצסעטיג implicitly vocalizes מוֹסָר ("bond"; cf. Isa 28:22; 52:2; Ps 116:16).

(3) ἐλαφος (G 7:23): G read איל ("stag") for M's אויל ("fool") and spliced it, in violation of the syntax, to 7:23. M's vocalization and the change from אויל to איל, though probably accidental, were induced also by the context: The entire chapter describes a fool and speaks of the "discipline" he undergoes.

For the verse as a whole, see the comment after 7:23.

7:23 אָד יְפַלָּח הֵא בבדוי transp to end of 7:23 (עָד יְפַלָּח הֵא בבדוּ M init (displ; accents)

G (continuing 7:22): (a) η ώς ἔλαφος τοξεύματι πεπληγώς εἰς τὸ ἦπαρ, (b) σπεύδει δὲ ὥσπερ ὄρνεον εἰς παγίδα. (c) οὐκ εἰδὼς ὅτι περὶ ψυχῆς τρέχει. "(a) or as a stag struck by an arrow in the liver. (b) And he hurries as a bird to a snare, (c) not knowing that he runs for [= at the cost of] life."

As noted, G reads איל and attaches it to the following 7:23. S's אולא supports this reading. To make sense of the impossible syntax that resulted when איל was removed from 7:22, G turned 7:23b into the main clause. In 7:23c, דף לא איל ("runs") was added to provide a verb for the indefinite "he" (M הוא) at the end of the verse.

The three stichoi in 7:23 are out of order. Verse 23a interrupts the sequence of three similes. Also, if the stag (the proposed original reading) is killed by an arrow, the trap and bonds are irrelevant. Moreover, a *trapped* stag would be butchered by a knife, not an arrow. Or, if we stick with M, the comparison between a fool going to his punishment (7:22c) until (*sic*!) he is killed by an arrow (7:23a), on the one hand, and a bird hurrying to a trap (7:23b), on the other, is awkward and hardly meaningful. Meinhold suggests that 7:23a is a later interpolation, though it is hard to see why the interpolator would have considered it helpful. It is likely that 7:23a was displaced from the end of verse 23, which is

ְּבָנִים שִׁמְעוּ־לֵי וְהַקְשִׁיבוּ לְאִמְרֵי־פִּי: 25 אַל־יֵשָׂטְ אָל־דְּרָכֵיהָ לְבֶּדְ אַל־תַּע בִּנְתִיבוֹתֶיהָ: 26 כִּי־רַבִּים חֲלָלִים הִפֵּילָה וַעֲצָמִים כִּל־ הַרָגֶיהָ: 27 דַּרְבֵי שִׁאַוֹל בֵּיתֶה יֹרְדוֹת אָל־חַדְרֵי־מֶוֶת: 1:8 הַלְאִ־חָכְמָה תִקְרֵא וֹתְבוּנָה תִּתֵּן קוֹלֵה: 2 בְּראשׁ־מְרוֹמֵים עֲלֵי־ דֶרֶדְ בֵּית נְתִיבִוֹת נִצְּבָה: 3 לְיַד־שְׁעָרֵים לְפִי־קֶרֶת מְבָוֹא פְתָחֵים

its natural place (Hitzig 1858; Delitzsch 1874–75; and others). When 7:23a is moved to the end of 7:23, the verse effectively culminates in the youth's death, pictured in a dramatic image. Other tropes for the death of fools are a storm (1:27), a sword (5:4), and ropes (5:22). By the above-suggested changes, 7:22–23 should be read:

הולך אחריה פתאם כשור אל טבח יבוא וכעכס אל מוסר איל כמהר צפור אל פח ולא ידע כי בנפשו הוא עד יפלח חץ כבדו

22	Impulsively he follows her,
	like an ox going to slaughter,
	like a stag bounding to bonds,
23b	like a bird rushing to a trap.
23c	He wasn't aware that he'd pay with his life-
23a	till an arrow split his liver.

The Masoretic accent scheme does not apply to the rearranged verses.

# 7:24

G adjusts M's "sons" to "son" for the sake of consistency, as in 5:7 and 8:32.

# 7:25

OG lacks 7:25b. Apparently the translator (or copyist) was misled by the repeated  $3 \times 10^{10}$  into thinking that he had already translated (or copied) that sentence. It is restored in Hexaplaric manuscripts (according to SyrH, from  $\theta'$ ) and present in the other versions.

7:27

As in 5:5, G parses ירדות as a causative: אמדמאסטסמו ("lead down").

# 8:1

G: (a) Σὺ τὴν σοφίαν κηρύξεις, (b) ἵνα φρόνησίς σοι ὑπακούσῃ. "(a) You [sg] shall call wisdom, (b) so that prudence may respond to you." The translator misconstrued πησία as second person. Consequently, the audience of chapter 8 becomes the youth who is addressed in chapter 7. In 8:1b, G's "respond to you" for M's more indefinite "gives forth her voice" is influenced by the mutuality formula of 8:17: "I love those who love me, and those who seek me find me."

# 8:2

G: (a)  $\dot{\epsilon}\pi\dot{\imath}$  γὰρ τῶν ὑψηλῶν ἀκρων ἐστίν, (b) ἀνὰ μέσον δὲ τῶν τρίβων ἔστηκεν. "(a) For she is upon the high heights, (b) and stands between the roads." G omits עלי דרך, perhaps by condensation with τῶν τρίβων representing עלי דרך as well as μέσον δ, V, and T correctly parse בית as Aramaic "between."

S adds אבכא'ד ("wisdom") as the subject in 8:2a for clarity and transposes איז (with עלי דרך with בית נתיבות for no apparent reason.

תּּרְנָּה: 4 אֲלֵיכֶם אִישִׁים אֶקְרֶא וְׁקוֹלִי אֶל־בְּגֵי אָדֶם: 5 הָבִינוּ פְתָאיֵם עָרְמֶה וּּכְסִילִים הָבִינוּ לֵב: 6 שֵׁמְעוּ בִּי־נְגִידַים אֲדַבֵּר וּמִפְתַּח שְׁפָתַי מֵישָׁרִים: 7 בִּי־אֶמֶת יֶהְגֶה חִבֵּי וְתוֹעֲבָת שְׂפָתַי רֶשַׁע: 8 בְּצֶדֶק כָּל־ אִמְרֵי־בֶּי אֵין בְׁהֶם נִפְתָּל וְעַקֵשׁ: 9 כַּלֶם גֲכֹחִים לַמֵּבֵין וִישָׁרִים

8:3

G: (a) παρὰ γὰρ πύλαις δυναστῶν παρεδρεύει, (b) ἐν δὲ εἰσόδοις ὑμνεῖται. "(a) For by the gates of the princes she sits; (b) in the entrances she praises herself." G 8:3a = 1:21b; 8:3b  $\approx$  1:20a.

The puzzling δυναστῶν for שערים (subsequently taken into 1:21b) apparently represents שרים, either in writing or in the translator's interpretation. The first could occur by near haplography  $\rightarrow \rightarrow \neg$  in the archaic script. On the similarity of these letters, see the comment on 6:16. Alternatively, "princes" may have seemed appropriate in this chapter, because they are said to rule by wisdom in 8:15 and should be part of Wisdom's explicit audience.

The unfamiliar קרת is ignored. This word is handled differently at each occurrence (8:3; 9:3, 14; 11:11); see de Waard 2006, 262–64.

S אלפי ("calls by her mouth") is a misunderstanding of לפי (actually "at the opening") as if it meant "by [her] mouth." S derives קרת from קרת, even though Syriac has the cognate קרא ("city"), a meaning recognized in 11:11a, where it is translated as הבגעא". It is translated הביה ("calls") again in 9:14 (S 9:15) and Job 29:7 and by סניתים ("they say") in Prov 9:3.

### 8:4

S prefixes מאפיא ("and she says") to the verse to introduce Wisdom's words, because the translator does not understand (8:3) as a verbum dicendi.

### 8:5

For הבינו לב (lit. "understand heart"), G has ἔνθεσθε καρδίαν ("take in heart," i.e., "absorb wisdom," "become intelligent"). This is commonly assumed to represent הכינו לב (Jäger, de Lagarde, BHS, etc.), but ἔνθεσθε

(which means "place in," "incorporate," "instill") is closer in sense to הבינו than to הבינו; cf. T. Naph. 2:2 and 3 Macc 5:28, where the verb means to implant something in the mind. G varies the verbs rather than repeating הבינו (McKane).

S reformulates the difficult אמאבאס באבתס ("will understand [or 'may they understand'] in their hearts"), again using a future for an imperative and making this a purpose clause dependent on the preceding. The result of this reformulation is that, unlike in M, Wisdom does not speak to the foolish (who in any case would ignore her call) but addresses people in general.

### 8:6

G's מֿעסוֹסש ("I will bring forth") is a smoothing of the seemingly awkward predication in M, ומפתח שפתי, "and the opening of my lips is straight things"; compare the adjustment in G 6:34.

### 8:7

G: (a) ὅτι ἀλήθειαν μελετήσει ὁ φάρυγξ μου, (b) ἐβδελυγμένα δὲ ἐναντίον ἐμοῦ χείλη ψευδῆ. "(a) For my throat speaks truth, (b) and abominable before me are deceitful lips." Since G assumed that "mæm" is a plural construct ("lips of deceit" = "deceitful lips"), it was obliged to add "before me," because "abomination is deceitful lips" would not make sense. G's circumlocution "abominable *before me*" reduces anthropopathism when said of God and recalls Aramaic קדם (see on 3:32). Here it is applied to Wisdom, as if she were a god.

S has במכל ("my mouth"), the usual term for the organ of speech, for "my palate").  $\Pi C$ 

### 8:9

G's ἐνώπια ("in front of") parses נכחים (actually, "honest") as an adjective from the adverb נֹכַח ("before").

S renders אברים as אוג ("are revealed"), working directly from the Hebrew text but taking exegetical guidance from G. S 8:9b reads האוד היב אבא גבר איני ("and they are straight to the one who wishes [var: "those who wish"] to know them"). This means that wisdom's words are accessible to anyone desirous of wisdom and not only to the educated. ּלְמִּצְאֵי דֲעַת: 10 קְחְוּ־<sup>ר</sup>מּוּסָר<sup>ז</sup> וְאַל־בָּסֶף וְׁדַעַת מֵחָרָוּץ נִבְחֶר: 11 כִּי־טוּבָה חֲכְמָה מִפְּנִינֵים וְכָל־חֲפָצִׁים לָא יִשְׁווּ־בָהּ: 12 אֲנִי־ חֲכְמָה שָׁכַנְתִי עָרְמֵה וְדֻעַת מְזִמְּוֹת אֶמְצָא: 13 (יְרְאַת יְהוָה שֵׁנֹאת רֶע גַּאֶה וְגָאוֹז | וְדֶרֶך רֶע) וּפִי תַהְפָּכוֹת שָׂנֵאתִי: 14 לְי־עֵצָה וְתוּשִׁיֶה אֲנֵי בִׁינָה לֵי גְבוּרֵה: 15 בֵּי מְלָכֵים יִמְלֶכוּ וְרוֹזְנִים יְתִׂקְקוּ

8:10 מוּסָרִי [ (מרדותא) T<sup>L</sup> (מוֹגָהאָ M V (disciplinam meam) T<sup>Z</sup> (מרדותי) (near dittog ויו → 1)

נבחר M ] + גבחר G (מיטאנגיס לוונס) (dittog בחר  $\rightarrow$ 

G: (a) λάβετε παιδείαν καὶ μὴ ἀργύριον (b) καὶ γνῶσιν ὑπὲρ χρυσίον δεδοκιμασμένον. (c) ἀνθαιρεῖσθε δὲ αἴσθησιν χρυσίου καθαροῦ. "(a) Take education and not silver, (b) and knowledge over tested gold. (c) And prefer knowledge to pure gold."

The possessive of מוסרי is lacking in G, S,  $T^L$ , and one MS KR, but represented in V and  $T^Z$  (the latter adjusting to M). Since the very frequent מוסר is not given the first-person suffix elsewhere in the Bible, the ' probably arose accidentally here, by near dittography.

In 8:10b, δεδοχιμασμένον = נְבְחָר parsed—perhaps correctly—as an Aramaism, from בח"ר "assay." G 8:10c = OG (de Lagarde; Fritsche). Though absent in G<sup>BS</sup>, G 8:10c is farther from M than is G 8:10b and shows a variant, בחר (parsed as impv). G 8:10b is likely a corrective addition. Actually, G 8:10c witnesses to both בחר (ἀνθαιρεῖσθε) and (καθαροῦ). The Greek words are not a double translation, since they fill different syntactic slots and presuppose different Hebrew words. Apparently the source text read, awkwardly and erroneously, בחר It is also possible that the translator's source text had one of the two variants and he chose to combine it with another variant he was aware of.

8:11

it stands, S 8:10c ("and nothing equals it") gives a reason, albeit tautologous, for S 8:10b. S diverges from both M and G in this verse.

### 8:12

G: (a) ἐγὼ ἡ σοφία κατεσκήνωσα βουλήν, (b) καὶ γνῶσιν καὶ ἐννοιαν ἐγὼ ἐπεκαλεσάμην. "(a) I, Wisdom, inhabited counsel, (b) and I called upon knowledge and insight." The aorists suggest that Wisdom is speaking of the time she first inhabited these principles, rather than her ongoing proximity to them (though that too is undoubtedly true). Similarly Sir 24:8, in dependence on the present passage, speaks of the moment when Wisdom first inhabited Zion. ערמה, which is a morally neutral term for cleverness and cunning, is translated cautiously by βουλή, which usually refers to counsel put to proper uses. G's ἐπεκαλεσάμην ("I have called upon") is an explanatory rendering of wat set.

V: intersum ("I am within") = אֶמְעַא

For שכנתי S has ברית ("I created"), which is the source of T<sup>L</sup> ברית. A variant is מענתי ("I acquired," "created"). Both are puzzling.

# 8:13 יראת MGS (add)

Prov 8:13a is probably a pietistic insertion intended to counterbalance the possibly amoral overtones of cunning and shrewdness. The line interrupts the connection between verses 12 and 14, which speak of the excellence of wisdom, and has nothing to do with the rest of the chapter, which does not aim at inculcating the proper attitude toward God. See further ABP.

G: (a) φόβος κυρίου μισεῖ ἀδικίαν, (b) ὕβριν τε καὶ ὑπερηφανίαν καὶ ὁδοὺς πονηρῶν. (c) μεμίσηκα δὲ ἐγὼ διεστραμμένας ὁδοὺς κακῶν. "(a) The fear of the Lord hates injustice, (b) insolence and arrogance and the ways of wicked people. (c) I hated the twisted ways of evildoers." G construes micked people. (c) I hated the twisted ways of evildoers." G construes  $V \ odit$ . In G 8:13b, "way" is pluralized. In 8:13c, G has ὁδοὺς for 'D. It seems that the translator is introducing the motif of the twisted path here, influenced by the parallel ITCT. Something similar happens in 11:20, where διεστραμμέναι ὁδοἱ renders ψ.

### 8:14

G translates תושיה as ἀσφάλεια ("security"). G in general associates תושיה not with intellectual faculties but with strength and firmness. For תושיה ("I am understanding"), G's פֿעָל φρόνησις, S's הבל, דמה מהבל

אֶדֶק: 16 בֵּי שָׂרִים יָשֻׁרוּ וּנְדִיבִּים כָּל־שִׁפְטֵי וּאֶרָץ': 17 אֲנִי וּאִהְבִי' אֵהֶב וֹמְשַׁחֲרֵי יִמְצָאֻנְנִי: 18 עְׂשֶׁר־וְכָבְוֹד אִתִּי הְוֹן עְׁתֵׁק וּצְדְקֶה: 19 (טְוֹב בְּרְיִי מֵחָרָוּץ ומִבֶּז וּתְבוּאָתִי מִבֶּסֶף נִבְחֶר:) 20 בְּאִׁרַח־ צְדְקֵה אֲהַלֵּך בְּתוֹך נְתִיבִוֹת מִשְׁפֵּט: 21 לְהַנְחֵיל אֹהֲבֵי ו יֵשׁ

and V's *mea prudentia* use the possessive ("mine is understanding") in imitation of the parallel clauses.

ארץ  $M^{\text{HilleliMSS}}$  G ( $\gamma\tilde{\eta}\varsigma$ ) S<sup>7h6</sup> = OS (הביא )  $M^{\text{ALedd}}$  S<sup>7a1etc.</sup> (הבסגאיי) V (*iustitiam*) (ideol)

G's κρατοῦσι might seem to represent ነយשנט (Toy; BHS) or אַפטי for M's שׁפטי. V also has a verb—*decernunt* ("they discern"). It is more likely that G (followed by V) made שׁפטי into a verb to parallel ישרי. A verb in G's source text would have required the absence of ג, but there is no reason to delete the word, and its presence is appropriate to the universalizing thrust of the verse.

Where M<sup>L</sup>, M<sup>A</sup>, and most other MSS have צֶּדֶק, there is evidence of another Masoretic variant: ארץ. In MS Rossi 314, an annotation assigns this reading to the important Hilleli Codex, of which now only the Pentateuch remains. It is also the reading in the Second Rabbinic Bible (Venice, 1525). This reading is confirmed by the fourteenth-century Spanish commentator Yosef Naḥmias. The medieval Masoretic tractate Minḥat Shai, ad loc., testifies to the two ancient readings but prefers (and see Baer 1880, 36–37). This reading is also witnessed by many MSS KR (and even conflated in K 76). Thus ארץ is *not* an emendation to M.

ארץ is witnessed by OG and OS. (The variant in MS 7h6 is original, not an inexplicable adjustment to G. Both אראהא appear in seventh-century Syriac MSS.)

T צֵדֵק = תריצותא.

later formulation allows itself to be understood as confined to Israelite rulers, or at least those of them who judge righteously. Read with "judges of the earth," the verse is in line with the universalistic perspective of the chapter, which tells of the creation of the world and describes how Wisdom calls to all humanity and serves them all.

8:17 אהביה [ (זער אהביה [ אהביה Μ<sup>Q</sup> G (τοὺς ἐμὲ φιλοῦντας) S (זער M<sup>K</sup> (unc)

The *ketiv* is a mechanical error and is impossible in context. (It would mean, "I [Wisdom] love those who love her." The cause of the error (+ $\pi$ ) is unclear.

8:19 M G S (add)

ייי פריי M^{Or}  $\mathrm{M}^{\mathrm{Or}\mathrm{Q}}$  ] פריי M^{\mathrm{Or}\mathrm{K}} (\mathrm{M}^{\mathrm{Or}\mathrm{K}} \operatorname{graph}^{2} \to \mathfrak{l})

The *ketiv* is a graphic error.

This verse was probably added as a pietistic precaution based on 8:10 and, especially, 3:14, which has the same chiasm reversed. In M, Wisdom takes a precautionary step away from her enthusiastic promise of material blessings to remind us (in an echo of 8:10) that wealth is still inferior to the fruits of wisdom. The latter are intellectual and ethical, not only material. But the antithesis between Wisdom's wealth and fruit is not really appropriate here, since Wisdom has just declared that her fruits include wealth (8:18). Moreover, the relative depreciation of wealth—though certainly consonant with the author's beliefs—weakens the promise of affluence in the next verse.

G: (a) βέλτιον ἐμὲ καρπίζεσθαι ὑπὲρ χρυσίον καὶ λίθον τίμιον, (b) τὰ δὲ ἐμὰ γενήματα κρείσσω ἀργυρίου ἐκλεκτοῦ. "(a) Better to enjoy (my) fruit than gold and precious stone, (b) and my produce is better than choice silver." καρπίζεσθαι ("to enjoy [my] fruit") and κρείσσω fill an ellipsis in each stich.

S's מענדא ("refined gold") combines התרא מנידא and ד.

### 8:20

A shewa missing in  $\begin{subarray}{c} M^L \begin{subarray}{c} M^L \begin{s$ 

G adds a verb, ἀναστρέφομαι ("I walk about") at the end of 8:20b to eliminate an ellipsis and tighten the parallelism.

8:21 fin ] + 2 stichoi G (21*a*) (transition)

For M's שי ("possessions" = G  $\delta \pi \alpha \rho \xi \omega$ ), S has שבי ("hope") and T<sup>L</sup> has אבי ("many years"). Both translations mute the material-

וְאֹצְרְתֵיהֵם אֲמַלֵּא: <br/>
<br/

ism of Wisdom's promises. G adds  $\dot{\alpha}\gamma\alpha\theta\hat{\omega}\nu$  ("with good things") to the end of the verse for specificity.

### 8:21*a*

G: (a) ἐἀν ἀναγγείλω ὑμῖν τὰ καθ' ἡμέραν γινόμενα, (b) μνημονεύσω τὰ ἐξ αἰῶνος ἀριθμῆσαι. "(a) If I tell you things that happen daily, (b) I shall (also) remember to recount things of old." The prosaic insertion provides a transition to the new topic. The addition appears in OL.

# 8:22

For a discussion of the versions' interpretation and treatment of 8:22–36, see ABP 1.411–16.

G: κύριος ἔκτισέν [var: ἐκτήσατο G<sup>V 252</sup>] με ἀρχὴν ὁδῶν αὐτοῦ εἰς ἔργα αὐτοῦ. "The Lord created [var: 'acquired'] me at [or 'as'] the beginning of his ways, for his work."

יקנני : The versions (like the subsequent commentators) are divided on whether קנה; ד בראני : (OG בראני, S) סייה, דייה, דייה, דייה, דייה, דייה, דייה, המנויד, דייה, acquired" ( $G^{V\,252} \alpha' \sigma' \theta' \dot{\epsilon} \kappa \tau \eta \sigma \alpha \tau \sigma$ ; V *possedit*). On the interpretations of this freighted word, see ABP 1.279–80, 411–12.

In *Epistle* 140, *ad Cyprian*, Jerome transliterated 8:22a as *adonai canani bresith dercho*. *Bresith* agrees with the likely adverbial function of ראשית (cf. V *initium*); the ב is probably influenced by Gen 1:1. G  $\delta\delta\tilde{\omega}\nu = \pm c c$ .

אנסדתי (לנסבתי א נסדתי ה<br/>סן M; אנסדתי הערא האמער, א נסדתי (M: vocal; G S: graph  $\to \to$  , diath)

וְסַבֹּתִי, "was woven" = was formed, best accords with the context, which tells of Wisdom's figurative birth.  $\Box$ " is used in Ps 139:13b and

Job 10:11 to describe how God crafts an embryo by weaving together bones and sinews.

G: (a) πρὸ τοῦ αἰῶνος ἐθεμελίωσέν με ἐν ἀρχῆ. "(a) Before the ages he founded me, in the beginning." G changes the verbs from passive to active in 8:23–24, thereby focusing on God as sole creator (CSP). ἐθεμελίωσέν με = 'ὑִ̣סָרָתָ' (by diathesis) for M's 'Locαri (deμελιοῦν almost always represents τος', never τος).) θ' ἡτοίμασέ με supports 'Locaronantally. Although ד/⊂ interchanges are not well attested, they are possible in the square script. α' κατεστάθην = μος) (cf. Mic 5:4; Ezek 32:30) as does σ' and another reading of θ': προκεχείρισμαι.

S has גֹסָדְתָּ' ("he established me"), either reading גֹסָדְתָּ' or influenced by G's έθεμελίωσέν με. S also provides a verb, גאם, ("establishes"), in 8:23b.

8:24 נְכָבַּדֵי הער הער גַרָבָר (דמ'ג האָזאָז מיז) א גַרָבַרי של א (near dittog הכד  $\to$  בכ, metath בכ)

G: (a) πρὸ τοῦ τὴν γῆν ποιῆσαι (b) καὶ πρὸ τοῦ τὰς ἀβύσσους ποιῆσαι, (c) πρὸ τοῦ προελθεῖν τὰς πηγὰς τῶν ὑδάτων. "(a) Before making the earth (b) and before making the abysses, (c) before the springs of the waters came forth," (continues in 8:25). The first part of G 8:24a (in Rahlfs) = M 8:23b. G supplies verbs in 8:24a and 24c to fill gaps. In 8:24b, the verb ποιῆσαι is made to apply to the abyss. That this change was not intended to skirt the suggestion that wisdom was "begotten" or "born" (Jrited to shown by the use of γεννῷ in the next verse.

For M's נבכי, G read נבכי (Landes 1956, 31–33; BHS). נבכי is not specifically represented in G (דמֹג דמָאָמֹג דῶν ὑδάτων), but it is unlikely that G would have simply ignored נכבדי (thus BHQ), whose basic meaning ("honorable") was well known. נכבדי מים מעינות however, could easily have been conflated with its synonym גבכי מים מעינות are the channels through which the underground waters feed the seas and rivers. They are called מבכי נהרות Job 28:11. For the form with J, see נבוכי מים ועולים in 1QHod III,15; cf. Ugaritic nbk/npk ("well") in Kirta col. v, 216. Even without G's support, נכבדי שטעום be a good conjectural emendation and makes better sense than נכבדי. The change from ינכבדי to ינכבדי ould have occurred in various ways, given the graphic similarity of the letters involved:  $\Box/\Box$ . (On  $\Box/\Box$  confusion, see the comment on 8:23; on  $\Xi$ , see LSF 131.)

8:25

G: (a) πρό τοῦ ὄρη ἑδρασθῆναι, (b) πρό δὲ πάντων βουνῶν γεννῷ με. G

# לִפְגֵי גְבָעַוֹת חוֹלֲלְתִּי: 26 עַד־לָּאׁ עֲשָׂה אֶֶרֶץ וְחוּצֵוֹת וְׁרֹאשׁ <sup>ר</sup>ּעַפְרָוֹת<sup>י</sup> הֵּבֵל: 27 בַּהַכִינִוֹ שֶׁמַיִם שֵׁם אֶנִי בְּחוּקוֹ חוּג עַל־פְּגֵי תְהוֹם: 28 בְּאַמְצִוֹ שְׁחָקִים מִמֶּעַל <sup>ר</sup>ְבְעַזְּוֹוֹי עִינְוֹת תְּהוֹם: 29 בְּשׁׁוּמָוֹ לַיֶּם | חַקּוֹ וְמַיִם לָּא יַעַבְרוּ־מֵיו <sup>ר</sup>ַבְּחַזְקוֹיי מָוֹסְבֵי אֱרֶץ: 30 וֵאֶהְיֶה אֶצְלוֹ

(continuing 8:24): "(a) before the establishment of the mountains, (b) and before all hills, he begat me."

In G, the use of γεννά for חוללתי is hardly intended to reduce mythic overtones (*pace* Küchler 1992, 137). On the contrary, treating הוללתי as an active verb only sharpens them. γεννά ("begat") serves as the main verb of 8:24–25. The present tense of γεννά in a series of aorists—the aorist is used elsewhere in this narrative—is puzzling. S has ארעעעל ("I was conceived/born").

# **8:26 עַפְרִוֹת** [ M<sup>A</sup> ] עַפְרָוֹת (vocal)

M<sup>L</sup> vocalizes עָפְרוֹת, from an unattested עָפְרוֹת; M<sup>A</sup> and many manuscripts correctly read עַפְרוֹת (from עָפָרוֹת), as in Job 28:6.

S translates הוצות as הוצות ("wadis"); similarly V *flumina*. These renderings are unique in their respective versions and are hard to explain.

8:27

G: (a) ἡνίκα ἡτοίμαζεν τὸν οὐρανόν, συμπαρήμην αὐτῷ, (b) καὶ ὅτε ἀφώριζεν τὸν ἑαυτοῦ θρόνον ἐπ' ἀνέμων. "(a) When he prepared the heavens, I was present with him, (b) and when he demarcated his throne upon the winds." συμπαρήμην αὐτῷ ("was-with-and-near him") emphasizes Wisdom's proximity to God. G thus reformulates M's wac ("there"), since no place has been mentioned for "there" to refer to. "His throne" is

based on the idea that the heavens are God's throne (Isa 66:1) and that he is located on the "circuit of the heavens" (Job 22:14). G's  $\dot{\epsilon}\pi'$   $\dot{\alpha}\nu\dot{\epsilon}\mu\omega\nu$  ("on the winds") for ההום is a reflex of the same picture and arises from the assumption that the deeps are not the "logical" place for God's throne (CSP). Note that different words for "when" are used in this and the next verse. Since no interpretive motive is relevant in cases such as this, they demonstrate the translator's inclination to enhance variety for its own sake (e.g., 3:13; 4:15).

S ברכת "with him" ≈ G.

### 8:28 (וו בעז ווי) ביע M (metath $H \rightarrow H$ )

### 8:29 (ב∩ב M ] > G (homoi ב∩ב)

רחזקו (ז א בחזקו G (ז ליטעטא א דאוילי  $M \approx S$  (מכג ברג) (graph  $t \rightarrow 1$ )

G: (M a-b) >; (c) καὶ ἰσχυρὰ ἐποίει τὰ θεμέλια τῆς γῆς. "(c) and was strengthening the foundations of the earth." M 8:29ab are lacking in OG. They are supplied in G<sup>Ss A</sup> etc. from θ' (asterisked in SyrH). The missing stichoi are essential to the picture and original. The reason for their disappearance is not clear. The mythological background of the line is already submerged in the Hebrew and would not have required suppression. The omission was perhaps due to homoioarkton from though to charge to homoioarkton from the picture of the line is shared. D. N. Freedman (personal communication) observes that if the lines were written stichometrically, the words could be vertically aligned, which would facilitate parablepsis.

אמע לאסטבע האסנגו בחוקו ("when he made strong"), for M בחוקו ("when he engraved"). The former is correct, since foundations are built by laying them down and making them sturdy, not by engraving them. For 1/1 interchange, see LSF §121ab; Kennedy 1928, 65.

V: (a) quando circumdabat mari terminum suum (b) et legem ponebat aquis suos. (c) Quando adpendebat fundamenta terrae. "(a) When he אָמְוֹז וְאָהְיֶה שַׁעֲשָׁעִים יָוֹם | יִוֹם מְשַׂחֶקָת לְפָנְיו בְּכָל־עֵת: 31 מֲשַׁחֶקֶת בְּתֵבֵל אַרְצְוֹ וְשַׁעֲשָׁעִׁי אֶת־בְּנֵי אָדֲם: 32 וְעַתְּה בְנִים שִׁמְעוּ־לֵי וְאַשְׁבִי דְּרָכֵי יִשְׁמְרוּ: 33 שִׁמְעוּ מוּסֶר וַחֲכָמוּ וְאַל־תִּפְרֶעוּ: 34 אֵשְׁבֵי אָדָם שֹׁמֵעַ לֵי לִשְׁלֵד עַל־דַּלְתֹתִי יָוֹם | יָוֹם לִשְׁמֹר מְזוּזָת פְּתָחֵי: 35 בֵּי מָצְאִי 'מְצָא' חַיָּיֶם וַיָּפֶק רְצוֹז

compassed the sea with its limit, (b) and set a law to the waters, (c) and weighed out the foundations of the earth." In 8:29b, V explains what it means for the waters not to transgress God's mouth. For V 8:29a, cf. T; for V 8:29b, cf. S.

S's  $a \to a \approx M$ , since the notion of "strengthening" would have been conveyed lexically, whereas the unclear notion of "engraving" foundations can be implied by "making" them.

### 8:30

G: (a) ήμην παρ' αὐτῷ ἁρμόζουσα. (b) ἐγὼ ήμην ἧ προσέχαιρεν. (c) καθ' ήμέραν δὲ εὐφραινόμην ἐν προσώπω αὐτοῦ ἐν παντὶ καιρῷ, "(a) I was with him, arranging (things). (b) I was the one in whom he rejoiced. (c) Daily I rejoiced in his presence at all times" (continues in 8:31). The crux אמון has been understood variously, according to the different senses of אמ"ן. G: ἁρμόζουσα, which may mean "suiting myself (to him)," "arranging [sc. all things]," "fitting together" (CSP), or (literally) "harmonizing," "being in tune" (Gerleman 1950, 26). The latter rendering recalls the Pythagorean notion of the "music of the spheres," which is called ἁρμονία. Gerleman (1950, 26-27) finds in this verse the Stoic conception that wisdom brings all things into harmony. This is possible, but the idea need not have come directly from Stoic philosophy.  $\sigma'$ ,  $\theta'$ : ἐστριγμένη ("set firm," "made steady"). α': τιθηνουμένη ("nursed"). Both treat אמון as a passive participle. V: cum eo eram cuncta conponens ("I was with him arranging everything"). This is based on G but does not necessarily capture the intended meaning of ἁρμόζουσα. I understand the word as an infinitive absolute and translate, "And I was with him, growing up" (ABP 1.285-87, q.v. for the history of interpretation; also ABP 1.414–15).

S: מהימנתא שאמשה ("I was creating with him"). T: מהימנתא ("trusted one [fem]").

G: (a) ὅτε ἐυφραίνετο τὴν οἰκουμένην συντελέσας (b) καὶ ἐνευφραίνετο ἐν υἱοῖς ἀνθρώπων. "(a) While he rejoiced, having completed the inhabited world (b) and took pleasure in the sons of men." G makes 8:31 a temporal clause dependent on verse 30, thereby associating 8:31 with God's rejoicing at creation, in accordance with Gen 1:31. The assumption that God was the one who rejoiced led G to override the feminine gender of משחקת in Prov 8:31a.

Jäger retroverts סטעדבאלסמג to תכל. This would not fit grammatically, but a scribal error is possible. More likely, the creation story is the source of סטעדבאלסמג, which is used in Gen 2:1–2 for כלה. דלה סוגטעוליאן conflates ארצו ושעשעו ושעשעו ושעשעו ושעשעיו (parsed as plural ושעשעיו or (ושעשעיו) or ושעשעיו and rephrased it as a verbal clause.

S: (a) = M; (b) הכבא המעה המער המני, אובא "(b) and I was praised by the sons of man." This reflects ושעשעי ("my delight," understood as "humans delighted in me"). This harks back to 8:3.

### 8:32-34

At this point  $G^B$  (= OG) proceeds in the order M 8:32a, 34a, 32b, 34bc, 35ab, 36ab, omitting 8:33. G's order makes sense but is inferior in literary balance. The reason for the reordering is unclear. One possibility is that M 8:32b and 33 were lost from G's source text by homoioteleuton from from 'אשר' (8:32b) to אשר' (8:34a). Then 8:32b was restored, but after 8:34a. OG thus read M 8:32a, 34abc, 35ab, 36ab. Verses 32b and 33 are supplied in SyrH with an asterisk and are found in G<sup>A</sup>. The following comments proceed in M's sequence.

### 8:32

G uses vié singular, for M's plural, as in 5:7 and 7:24.

### 8:33

G omits 8:33 for unclear reasons. The verse is appropriate to context.

# 8:35 (משכח $M^Q V (inveniet) T (משכח) מצאי M^K \approx G (<br/>נאססט) <math>\approx S ($ מבסטא (reminiscence of מצאי')

The ' of the first מצא' arose by a sort of distant dittog of the ' of the first מצא' G: (a) ai yàp žźodoi μου žźodoi ζωῆς, (b) καὶ ἑτοιμάζεται θέλησις παρὰ κυρίου. "(a) For my exits are the exits of life (b) and favor is readied with the Lord."

מִיְהוֶה: 36 וְחֹטְאִי חֹמֵס נַפְּשֵׁוֹ כָּל־מְשַׂנְאֵׁי אָָהֲבוּ מֶוֶת: • 1:9 חֻכְמוֹת בִּנְתָה בֵיתֶה <sup>ו</sup>הִאָּבָה<sup>י</sup> עַמּוּדֵיהָ שִׁרְמֵה: 2 טִבְחָה טִבְחָה מִסְבָה יֵינֶה אַׁף עֵרְבָה שִׁלְחָנֵה: 3 שֵׁלְחָה נַעֲרֹתֵיהָ תִקְרָא עַל־גַּבִּי מְרָמֵי בֶרָת: 4 מִי־בֶּתִי יָסֵר הֵנָּה חֲסַר־לֵב יּאַמְרָה' לְּוֹ: 5 לֲכוּ לַחֲמֵוּ בְלַחֲמֵי וֹּשְׁתוּ בְּיַיָן מְסֵכְתִּי: 6 עִזְבָוּ פְתָאיָם

G 8:35a corresponds consonantally to the *ketiv* but implicitly vocalizes מֹצָאֵי ("my exits are the exits [of life]"). Insofar as G had something specific in mind, "exits" may refer to ways of escape from difficulty or may be a metonymy for "roads." GELS proposes the former, but 1:20 supports the latter, for the second interpretation conveys the idea, appropriate to these chapters, that wisdom is on the roadways (though they are not called "exits" there). Still, although the *ketiv*, G, and S are meaningful, the context requires the notion of "finding" (the antithesis of "missing"; 8:36a). Moreover, the plural is awkward before the singular verb in 8:35b. Also, 18:22 supports **XX** by rephrasing it and not the *ketiv*. For the association of **XX** with **XX** see also 10:13 and 18:22a.

In G 8:35b, ויפק is parsed as a passive, and its sense (<br/>לגים, only here) is surmised from context. הפיק is inconsistently translated in G.

S 8:35a ראפמע, ראפמנא = G.

### 8:36

G has the plural "sinners" in 8:36a and omits "all" in 8:36b. S = G.

9:1 הצבה האבה האבה האמ<br/>מ' האמי האנה האמי האנה האבה האנה האמי של ה<br/>  $M \ ({\rm graph} \ \Pi \to \Pi)$ 

הצבה ("has set up") provides a stronger parallel to בנתה ("has built") than does M's "has hewn." Also, M's order, "building"—"hewing," is nonsequential. Reading with the variant, the first line mentions building generally and is paralleled by a specific action, the first step in building, namely, "setting up columns." הצבה (possibly הצבה (in G, S, and T. T's ועתידת is neither an adjustment to M nor simply a conversion of S to Western Aramaic. Hence T attests independently to a non-M reading, one of the few places it does so. (The synonym האסר של was available to T.)

G στύλους elides the possessive pronoun of עמודיה, as often when the antecedent of a possessive is clear.

9:2

G adds εἰς κρατῆρα ("into a krater") in 9:2b to elaborate the picture of a symposium for the Hellenistic reader. (CSP rightly observes that κρατῆρα is too broadly used of vessels to be a specific allusion to Hellenistic mystery cults.) G does not represent a lost character (pace de Lagarde), because claways plene in the abs) is almost always translated ποτήριον (as in Prov 23:31), never κρατήρ, which was the bowl from which cups were filled.

9:3

G: (a) ἀπέστειλεν τοὺς ἑαυτῆς δούλους (b) συγκαλούσα μετὰ ὑψηλοῦ κηρύγματος ἐπὶ κρατῆρα λέγουσα. "(a) She has sent forth her manservants, (b) inviting [fem sg] [people] to a feast [lit. 'cup'] with a lofty declaration, saying [fem sg]:" (continues in 9:4).

In this paraphrastic verse, G's use of the masculine plural τοὺς ἑαυτῆς δούλους rather than the feminine plural is for the sake of propriety, since it would not be seemly to send out maids to bring in male guests; cf. Jdt 8:10; Matt 22:3. G derives קרת קרת ("call"); cf. 8:3. κρατῆρα is taken from 9:2 but nicely resonates the sound of σ.

S: אַזאָ מאָדָאָ מאָדָאָס. "(a) And she sent forth her manservants (b) that they may call on the heights and say [masc pl]:" (continues in 9:4). S is based on G but smoothes out an apparent unevenness by having the *manservants* rather than Wisdom doing the calling. T: עשינתא ונצירן ("mighty and guarded").

9:4 אָמָרָה [ (מארי: S (מארי: C (מארי: M G (נוֹתני) אַמָרָה ) אַמָרָה אַמָרָה אַמָרָה אַמָרָה אַמָרָה אַמָרָה אַמָרָה

S: אמרה האכי וארא לאלא, האנצמי וארא מארי שלא לאלא. "Whoever is simple let him come to me, and he who lacks sense, and I will say to him:" (continues in 9:5). S אמרה parses אמרה as first-person singular and supplies a conjunction. The first person is preferable to the third, as the word is vocalized in M and G. The switch to third person is awkward within the first-person discourse. (Wisdom is speaking in verses 4a and 5–6.) The first-person cohortative yields a smoother sentence and conveys volition or determination: "I will surely say to him." Compare 9:16.

9:6

G: (a) ἀπολείπετε ἀφροσύνην, καὶ ζήσεσθε, (b) καὶ ζητήσατε φρόνησιν, ἵνα βιώσητε, (c) καὶ κατορθώσατε ἐν γνώσει σύνεσιν. "(a) Abandon folly and live, (b) and seek prudence that you may be alive, (c) and keep understanding straight through knowledge." In place of καὶ ζήσεσθε, G<sup>BSAMSS</sup> וְחִזִּוּ וְאִשְׁרוּ בְּדֶרֶךְ בִּינְה: (7 יָּמֵׁר ו לֵּץ לֹמֵחַ לָוֹ קָלְוֹז וּמוֹכִיחַ לְרָשָׁע מוּמְוֹ: 8 אַל־תּוֹכַח לֵץ פּן־יִשְׁנָאֶדְ הוֹכַח לְחָכָם וְיָאֶהָבֶדָ: 9 תַּן לֵחָכָם וְיֶחְכַּם־עֲוֹד הוֹדַע לְצַדִּיק וְיַוֹסֶף לֶקַח:) פּ (10 תְּחַלַּת חֶרָמָה יִרְאַת יְהוֶה וְדַעַת קְדֹשִׁים בִּינֶה:) 11 כִּי־בֵי יִרְבַּוּ יָמֵידְ וְיוֹסִיפּוּ לְּדָ

have <sup>i</sup>va εἰς τὸν αἰῶνα βασιλεύσητε ("so that you may rule forever"), an expansion based on Wis 6:21 and reflecting the notion that Wisdom's words are (as in Wisdom of Solomon) addressed to the world's rulers. ἀφροσύνην is not "exegetical" (contra BHQ) but a precise translation of the abstract plural פָּתָאיָם. V and γ' translate similarly. CSP calls G 9:6b a "free rendering" of M, but in fact it is not a rendering but a converse restatement of 9:6a.

### 9:7-10

This passage, which may be titled "Advice to the Advisor," is undoubtedly an addition. It interrupts Wisdom's speech and is inappropriately interposed between the injunction in 9:6 and its natural sequel in verse 11, which gives the reason (introduced by causal 'D) for the injunction. The later scribe who inserted these verses apparently doubted that the "callow" and "senseless" are likely to be amenable to a call to seek wisdom (9:4–6) and so introduced a caveat. The passage does not echo the vocabulary of the preceding section and does not confront Wisdom's invitation directly but speaks instead to the issue of effective chastisement. It appears, then, that the present passage originated as an independent epigram and was subsequently inserted as a response to Wisdom's summons of the ignorant. (ABP, q.v. for further discussion.)

Verse 10 seems to have originated separately from verses 7–9. It is abrupt and isolated, for the practical advice in 9:7–9 does not prepare the way for the definition of wisdom in 9:10. Still, in the context of Wisdom's summons, 9:10 converts the entirety of 9:7–10 into a comment on the religious preconditions of education. Verse 10 has the appearance of a statement of principle appended to 9:7–9.

Interlude E, which tells of the banquets of Wisdom and Folly, originally consisted of 9:1-6 + 11 + 13-18. The purpose of 9:7-10 is to warn against unrealistic hopes in educating fools, in spite of Lady Wisdom's invitation to them. Behind this warning is an ancient debate in wisdom literature about who is capable of learning (see further ABP 1.309–17).

### **9:**7–**9** M G S (add)

For הון לחכם G has δίδου σοφῷ ἀφορμήν ("Give the wise man an opportunity"), supplying a direct object. The direct object in M is "reproof," implicit in the preceding verse, but it may have seemed strange to the translator that a wise man would need a reproof. S אפאר and V occasionem are based on G.

### 9:10 M G S (add)

fin ] + 1 stich G (10*a*) (elab)

G: (a) ἀρχὴ σοφίας φόβος κυρίου, (b) καὶ βουλὴ ἀγίων σύνεσις. "(a) The beginning of wisdom is the fear of the Lord, (b) and the counsel of the holy ones is understanding."

There is considerable inner-G variation in this verse. The use of βουλή for Fυπ shows that the translator has in mind the counsel that saints provide; cf. S. After all, he may reason, humans cannot really *know* the angels. ἄγιος may refer to either (human) saints (e.g., Pss 16[15]:3; 34[33]:10) or angels (e.g., Ps 89[88]:6, 8; Job 5:1). The former is probably intended here, since it is holy people who give counsel; and see Wis 6:10.

S has הוו-שאה ("knowledge of saints") for ודעת קדשים. Thus fear of God is defined as the knowledge that the saints possess.

### 9:10*a*

G (10*a*): τὸ γὰρ γνῶναι νόμον διανοίας ἐστὶν ἀγαθῆς. "For to know the law is [the part of] a good intellect." G 9:10*a* (Rahlfs), which is actually the third stich of G 9:10, is a doublet of 9:10b. The distinctively Greek syntax of διανοίας ἐστὶν ἀγαθῆς supports an origin in Greek rather than Hebrew. G 9:10*a* has the obelus in SyrH, and de Lagarde considers it the older translation of M 9:10b. Toy calls it a gloss by a "legalistic scribe." Seeligmann (1953, 179) considers it a midrashic expansion that is based on Prov 13:15 and that identifies wisdom with study of Torah. Prov 9:10*a* certainly has the latter function, but wisdom was identified with Torah before G-Proverbs, and the expansion could be the work of the original translator (thus CSP) no less than a later scribe. In G 9:10*a*, M's <code>¬rw¬</code> is understood as "holy matters," hence "the law." Wis 6:10 says that "those who observe holy things in a holy way shall become holy." Prov 9:10*a* is reused in G-Prov 13:15b or possibly comes from there.

### שְׁנִוֹת חַיֶּים: (12 אִם־חֶכַמְתָּ חָכַמְתָּ לֶדְ וְלַצְתָ לְבַדְדָ תִשְׂא:) שִׁנִוֹת חַיֶּים:

9:11 (gram, smooth) (حرة Μ ] حم\* « G (τούτῷ γὰρ τῷ τρόπῷ) (gram, smooth)

G: (a) τούτω γὰρ τῷ τρόπῳ πολὺν ζήσεις χρόνον, (b) καὶ προστεθήσεταί σοι ἔτη ζωῆς σου. "(a) For in this way you will live much time, (b) and years will be added to your life." BHQ assumes that the change from ' $\Box$ ("in me") to "in this way" is the doing of the translator, who is adapting to "immediate context."

More likely, a change from בה סו בי סכנעודים ליד Hebrew transmission, since S reflects the latter without dependency on G's expansive "in this way." The change was induced by the fact that Wisdom has not been speaking since 9:6. 9:7–10, I argue in ABP 1.306–7, is a later insertion. That passage interrupts Wisdom's speech and is interposed between her injunction in 9:6 and its natural sequel in 9:11, which gives the reason (introduced by causal 'C') for the injunction. On the function and message of the insertion, see ABP 1.306–7. The change to שם smoothed out an incongruity produced by the insertion. In G, the future passive προστεθήσεται ("will be added") remedies the lack of a subject for ויוסים.

S הבה = בה independently of G's circumlocution. S הנאלהשב = G (הראלהשב).

fin ] + 7 stichoi G S (9:12*a*-12*c*) (elab)

Prov 9:12 is a later addition apart from the larger insertion in 9:7–10 (as I argue in ABP 1.317–18). The present proverb was attached to this location by association with the words  $\Box and \forall \gamma$  in 9:7–10.

G: (a) υἰἑ, ἐἀν σοφὸς γένῃ σεαυτῷ, σοφὸς ἔσῃ καὶ τοῖς πλησίον. (b) ἐἀν δὲ κακὸς ἀποβῆς, μόνος ἀναντλήσεις κακά. "(a) Son, if you become wise, you will be wise for yourself and for your neighbors, (b) but if you prove evil, you alone will draw up evils." On the prefixed "son," see the comment on 2:17. The phrase "and for the neighbors" is a cautionary addition, lest one think that the wise man benefits *only* himself. M's **wwn** ("you will bear") is understood (inaccurately) as *drawing up*—metaphorically drawing up evils as one draws up water. κακός and κακά are added in order to place the contrast between 9:12a and 12b on an ethical axis.

S 9:12 ≈ G.

<sup>9:12</sup> M G S (add)

9:12 <i>a</i> -c		
12a	a	ὃς ἐρείδεται ἐπὶ ψεύδεσιν, οὗτος ποιμανεῖ ἀνέμους,
	b	ό δ' αὐτὸς διώξεται ὄρνεα πετόμενα.
12 <i>b</i>	а	άπέλιπεν γὰρ όδοὺς τοῦ ἑαυτοῦ ἀμπελῶνος,
	b	τοὺς δὲ ἄξονας τοῦ ἰδίου γεωργίου πεπλάνηται.
12 <i>c</i>	а	διαπορεύεται δὲ δι' ἀνύδρου ἐρήμου
	b	καὶ γῆν διατεταγμένην ἐν διψώδεσιν,
	С	συνάγει δὲ χερσὶν ἀκαρπίαν.
12a	а	He who supports himself on deceits—he will shepherd
		the winds,
		THE WINDS.
	b	and he will pursue a flying bird.
12b	b a	
12b		and he will pursue a flying bird.
12 <i>b</i> 12 <i>c</i>	a	and he will pursue a flying bird. For he has abandoned the roads of his own vineyard,
	a b	and he will pursue a flying bird. For he has abandoned the roads of his own vineyard, and has forgotten the paths of his own field.

This addition has no counterpart in M. It is an addition in G, partly based on a nonextant Hebrew text. The passage remonstrates against relying on deceits, by which foreign doctrines and beliefs are intended. What one *should* rely on is the true wisdom (3:18), which has been granted to Israel. This addition is an interpretation of the Strange Woman; see the discussion in ABP 1.418–20.

The evidence for a Hebrew source text in 9:12*a*–12*b* is as follows:

(1) When we retrovert ἐρείδεται to תמך, we see an ambiguity that can explain a peculiarity in G and forms a better proverb. In the Hebrew Proverbs, תמק always means "grasp," "hold" (3:18; 4:4; 5:5, 22; 11:16 [2x]; 28:17; 29:23; 31:19). The only meaning that G knows for this verb, however, is "lean on" or "support oneself on," as in RH; see the comment on 3:18. Although G's δς ἐρείδεται ἐπὶ ψεύδεσιν does make sense, the proposed Hebrew (תומך שקר) would be better translated, "he who grasps deceit." This is more in harmony with the image cluster. Grasping is an active attempt to get something. Here it is appropriately parallel to shepherding (= העה) and pursuing (= הדר). G's "support himself on," in contrast, is static.

(2) דסט אלי מעגלי (or מעגלית), understood as "axles," hence, *pars pro toto*, "wagon"; cf. 2:9, 18. An original Greek composition would have used a more common word for paths.

## 13 אֵשֶׁת בֵּסִילוּת הְמִיֶּה פְּתַיּוּת וּבַל־יָדְעָה מֵּה: 14 וְיָשְׁבָה לְפָּתַח בִּיתֶה עַל־בִּפֹא מְרַמֵי קֶרֶת: 15 לִקְרָא לְעָׂבְרֵי־דֶרֶדְ הַמְיַשְׁרִים

(3) The middle  $\pi \epsilon \pi \lambda \dot{a} \nu \eta \tau a \iota + accusative is unnatural Greek. To be sure, in the middle/passive this verb can govern the accusative of place and mean "wander about in" (LSJ 1411a), but that meaning would not be appropriate here, since the man in question has$ *left*and wandered*from*his proper territory. G 9:12*b*b looks like a mechanical rendering of is an Aramaism meaning "go astray" and "forget."

The Hebrew underlying G 9:12*ab* can be retroverted approximately to the following:

12a	а	תּוֹמֵך שֶׁקֶר יִרְעֶה רוּחַ
	b	וִיִרְדֹף עוֹף מְעוֹפֵף
12b	а	כִּי עָזַב דַּרְכֵי כַּרְמוֹ
	b	וּמַעְגְלֵי שָׂדֵהוּ טָעָה

Alternatives: ומעגלי for אפור זמעגלי, ומעגלי for ומעגלי.

In ABP 1.419 I retroverted 9:12*b*b to ו<br/>אמעגלי שדהו תעה, restoring a preposition since none is reflected in G. If we retrovert to טעה, however, none is required. The sense is much the same.

It is difficult to reconstruct a Hebrew source text for 9:12*c*, especially for stich 9:12*c*c. The description of the desert there is influenced by G-Jer 2:6b, which includes the phrases ἐν τῆ ἐρήμω, ἐν γῆ ἀπείρω καὶ ἀβάτω, ἐν γῆ ἀνύδρω καὶ ἀκάρπω. Prov 9:12*c* is an inner-Greek expansion of 9:12*b*. It teaches that the deceitful man has chosen a barren, fruitless land in place of his rightful field and vineyard, that is to say, his Jewish religious culture.

S has this addition. S 9:12*a*-*c* (applying Rahlfs numbering) reads:

- 12a אינא געוב גאאינא. איבאי געוב געויין גע
- 12b . הבושה גבושה 12b . הבות פהלעותה.
- אדיד בעהובא גלא דיש. 12c הדי איגא גגוישא וגא ב-משא. אב מה גבע לא דגר

- 12*a* a He who rebels deceitfully shepherds winds,
  - b and pursues a bird of the heavens.
- 12b a For he has abandoned the way of his vineyard,
  - b and forgotten the paths of his tillage,
- 12*c* a to travel on a desert without water,
  - b and from that (path) which is trodden, he traveled in thirst.
  - c He, too, will gather nothing.

### 9:13

G: (a) Γυνὴ ἄφρων καὶ θρασεῖα ἐνδεὴς ψωμοῦ γίνεται, (b) ἡ οὐκ ἐπίσταται αἰσχύνην. "(a) A foolish and brazen woman comes to lack (even) a morsel. (b) She knows no shame."

G construes ששת כסילות as "foolish woman," which is what the Hebrew idiom requires ("woman of folly" = foolish woman). Still, "foolish woman" is probably a figure for folly in the abstract, which is then personified (ABP). G derives the *hapax* פתיות from ליס ("piece of bread," "morsel"), which it then renders ψωμοῦ. It must then add ἐνδεής to make sense of the sentence.

G apparently found "what" in ארעה מה ובל ידעה שעב ip uzzling and so substituted a clearer direct object: מוֹסעִיעָיע. According to Jäger (followed in ABP), מוֹסעִיעָיע reflects ארמה, for M's אה. To be sure, the image of a brazen woman refusing to be ashamed (הכלם) is found in Jer 3:3, but it would be hard to account for the loss of the letters כל Most likely, the translator had difficulty with the rare use of הם in the sense of "anything" and so provided "shame" as a logical direct object of "does not know."

V renders ובל ידעה מה, correctly, as *et nihil omnino sciens* ("and knows nothing at all").

### 9:14

G gives the gist of the Hebrew, translating על כסא מרמי הרמי אין as  $\dot{\epsilon}\pi \lambda$  δίφρου  $\dot{\epsilon}\mu\phi\alpha\nu\omega$ ς  $\dot{\epsilon}\nu$  πλατείαις ("upon a chair, publicly, in the plazas"). S

אִּרְחוֹתֶם: 16 מִי־ְפֶתִי יָסֵר הֵנְּה וַחֲסַר־לֵב <sup>ו</sup>וְאַׂמְרָה<sup>ז</sup> לְּוֹ: 17 מַיִם־ גְּנוּבִים יִמְתֶּקוּ וְלֶחֶם סְתָרֵים יִנְעֶם: 18 וֲלֹאֹ־יָדַע בְּי־רְפָאַים שֶׁם בְּעִמְהֵי שְׁאוֹל קְרֵאֶיהָ: פ

condenses the phrase to  $\neg$  in  $\neg$  in  $\neg$  ("on a high chair"), taking the image from G but diverging from its wording. M's  $\neg$  ("city"), understood as "call" in S 8:3, is either ignored or bundled into a picture appropriate to context. S ignores it.

### 9:15

Hebrew TרT is not explicitly represented in OG (it is supplied in G<sup>A</sup> as δδόν), but it is probably implicit in τοὺς παριόντας ("those passing by" [CSP]); compare the condensed treatment of TTTT in 9:6.

9:16 אַמְרָה ה האמי<br/>א $* \approx G \;(\pi \alpha \rho \alpha \varkappa \epsilon \lambda \epsilon \acute{y}$ סטסמ) א (יא מיד  $M \;(vocal, gram)$ 

For יהעריה ("innocent," "callow"), G has a much harsher ἀφρονέστατος ("most foolish"). In 9:16b, the versions agree on the consonantal ואמרה but parse it differently. G's παρακελεύομαι λέγουσα ("I exhort, saying") is an expansive rendering of אַמְרָה that maintains the first-person singular; see the comment on 9:4. S also renders the verb as first person but without G's expansiveness. The first person is correct. It is more natural in the context of Folly's speech. T<sup>L</sup> puts the entire sentence in the third person and restructures the syntax: "Whoever is foolish comes (נאתה) to her, and (likewise) the senseless one. And she says (ואמרה) to him."

### 9:17

G: (a) Ἄρτων κρυφίων ἡδέως ἄψασθε (b) καὶ ὕδατος κλοπῆς γλυκεροῦ. "(a) Take hidden breads with pleasure, (b) and the very sweet water of theft." G transposes the clauses and treats ממתקו as an adjective and שנעם as an adverb. (V renders both as comparative adjectives.) In the Hebrew these words are apparently spoken by the teacher, though this is not certain. The reworking puts them into Folly's mouth, lest the reader think they are true and that illicit pleasures really are sweet.

**9:18** fin ] + 8 stichoi G (18*a*–*d*) (elab)

G's קראידק associates קראיה with קראיה "meet," with diathesis.

### 9:18*a*-*d*

G adds four couplets:

18a	a b	άλλὰ ἀποπήδησον, μὴ ἐγχρονίσης ἐν τῷ τόπῳ μηδὲ ἐπιστήσης τὸ σὸν ὄμμα πρὸς αὐτήν.
18b	а	ούτως γὰρ διαβήσῃ ὕδωρ ἀλλότριον
	b	καὶ ὑπερβήσῃ ποταμὸν ἀλλότριον.
18c	а	άπὸ δὲ ὕδατος ἀλλοτρίου ἀπόσχου
	b	καὶ ἀπὸ πηγῆς ἀλλοτρίας μὴ πίῃς,
18d	а	ίνα πολὺν ζήσης χρόνον,
	b	προστεθῆ δέ σοι ἔτη ζωῆς.
18a	a	But get away, tarry not in the place,
18a	a b	But get away, tarry not in the place, nor direct your eye toward her.
18a 18b		÷ · · · -
	b	nor direct your eye toward her.
	b a	nor direct your eye toward her. For thus you will pass through strange water
18b	b a b	nor direct your eye toward her. For thus you will pass through strange water and cross over a strange river.
18b	b a b a	nor direct your eye toward her. For thus you will pass through strange water and cross over a strange river. From strange water keep away,

This addition was composed in Greek and is aimed at a diaspora audience. It admonishes Jews to avoid close contact with the surrounding foreign culture. (For an interpretation, see ABP 1.422-23.) Evidence for a Greek origin is the fact that 9:18d is based on G 9:11, using its wording, with none of the distinctive features of M 9:11, namely, M's "by me," "your days," and "shall add" in the active. Since the addition interprets the symbolism of the foolish woman in a way not found elsewhere in G-Proverbs, we may conclude that it was inserted later in the Greek transmission.

G 9:18bb is lacking in G<sup>BA</sup>. Since the other verses of the epigram are couplets, the loss of this line was accidental. The stich is confirmed by S, which agrees with  $\delta_{i\alpha}\beta_{i\sigma}$  as opposed to  $G^{s}\delta_{i\alpha}\beta_{i\sigma}$ 

S is a close translation of G here and does not witness to a Hebrew text. Evidence for dependency on G is in 9:18*a*, where initates one of the senses of ἀποπήδησον, "jump."

In 9:18d, متنب الملاة معدل المعالمة معدل ("So that many days and years of life may be added to you"), S reformulates the line under the influence of S 9:11, where it has סנאלם (= G, ≠ M ויוסיפו).

פ

משלי שלמה

בֶּן חֲכָם יְשַׂמַּח־אֶב וּבֵן לְּסִׁיל תּוּגַת אִמְוֹ: 2 לֹא ְיִוֹעִילוּ אוֹצְרָוֹת רֶשַׁע וֹצְדָלֶה תַּצִיל מִמְוֶת: 3 לְא־יַרְעֵיב יֵהוָה גָפָשׁ צַדֵּיק וּוְחַיַּת' רְשָׁעֵים יֶהְדְּוּ: 4 וֹבאשׁ עֹשָׂה' כַּוּ־רְמִיֶּה וְיֵד חָרוּצִים תַּעֲשִׁיר: 5 אַגֵר בָּקַּיִץ

10:1

In 10:1a, the start of the second major collection in Proverbs, several versions omit משלי שלמה. G, S, and V\* omit 10:1aa, "Proverbs of Solomon." S and V must be following G in the omission, even though V and S did not consider it necessary to obscure the ascription to Agur in 30:1. The heading in M 10:1 is original. For one thing, the collection that begins in 10:1 was the original beginning of the book (see ABP 2.499) and must have had a title. There would have been no need to add a title when 10:1 was no longer the start of an independent document.

The versions handle the ascriptions in various ways:

M: 1:1 Solomon; 10:1 Solomon; 24:23 the wise; 25:1 Solomon ("which the men of Hezekiah transcribed"); 30:1 Agur; 31:1 Lemuel ("which his mother taught him").

G: 1:1 Solomon; 10:1 omitted; 24:23 none ("*to* the wise"); 25:1 Solomon ("which the friends of Ezekias king of Judah copied out"); 30:1 none (translated "fear my words"); 31:1 none (translated "my words have been spoken *by* God"). G's source text had an ascription to "the wise" in 22:17, now obscured in translation, probably deliberately. See the comments on these verses.

V: 1:1 Solomon; 10:1 Solomon (but the absence of the verse in some MSS is likely original and learned from G); 24:23 none ("*to* the wise"); 25:1 Solomon ("which the men of Ezechias king of Juda copied out"); 30:1 "Gatherer the son of Vomiter"; 31:1 King Lamuhel ("which his mother taught him").

S: 1:1 Solomon; 10:1 omitted; 24:23 none ("*to* the wise"); 25:1 Solomon ("which the friends of Hezekiah king of Judah wrote"); 30:1 Agur son of Yaqiy; 31:1 Muel ("which his mother taught him").

T: 1:1 Solomon; 10:1 Solomon; 24:23 implicitly Solomon ("These too I say *to* the wise"); 25:1 Solomon ("which the friends of Hezekiah king of Judah wrote"); 30:1 Agur; 31:1 Lemuel ("which his mother taught him").

G is determined to avoid even oblique concession of non-Solomonic authorship. In 10:1, a new ascription to Solomon might imply that 1:1 does not apply to the entire book. G obscures the other headings except

10:1

for the one in 25:1. This heading was less troublesome because it refers to the editorial role of Hezekiah's men rather than primarily assigning authorship to Solomon and thereby raising questions about the other material. In G 31:1, Solomon speaks and introduces his teaching as words spoken by God *which his mother taught him*. In 24:23 G, S, V, and T, the teachings are "*to* the wise," and the speaker is implicitly Solomon, who continues to speak in the following verses.

10:3 וחית G ( $\zeta \omega \dot{\eta} \upsilon \delta \dot{\epsilon}$ ) ו הוו (graph  $\tau \to \pi$ ; graph  $\tau \to \iota$ )

G: (a) οὐ λιμοκτονήσει κύριος ψυχὴν δικαίαν, (b) ζωὴν δὲ ἀσεβῶν ἀνατρέψει. "(a) The Lord will not let the righteous soul starve, (b) but he will overthrow the life of the impious." G construes רַשָּׁע (not implausibly) as רַשָּׁע then pluralizes it, as it often does for character types.

Neither M nor G is entirely apropos here. It does not make good sense to say that God will rebuff (יהדף) the evildoer's "disaster" (which sounds like protection) as M has it, but neither is G's "life" a fitting object of  $\dot{\alpha}\nu\alpha\tau\rho\dot{\varepsilon}\psi\varepsilon\iota.\,\dot{\zeta}\omega\dot{\gamma}\upsilon\dot{\delta}\dot{\varepsilon} = \pi$ ית (Jäger). Hebrew היה can mean "life," as G renders it, but also "appetite," as in Job 33:20 (// נפש) and Job 38:39. The latter sense is what was intended in the translator's source text. G read life." This example shows that G does not have to be *right* in order to witness to a variant; in fact an awkwardness in translation can be helpful in retroversion; see further \$3.1.5.2 and Fox 2005.

The proposed original וחית was corrupted to והות in M. Alternatively, instead of emending we might maintain והות but construe it as "desire" (a by-form of הוה . אוה is supposed to have this meaning in Mic 7:3 (thus, e.g., HALOT). There, however, it could as well mean "deceit."

S החנול ("property") looks like a rendering of הוון, though there is no graphic reason for  $\pi$  to become  $\gamma$  or 1, nor is there an exegetical motive for associating with that word.

Several medieval Hebrew MSS KR have רוגדים in place of רשעים. This is due to the influence of 11:6 (BHS). The motive for the change is unclear, but it does show that synonym substitutions can occur within the Hebrew tradition even at a late stage in the transmission.

```
10:4 אָלאש [ (מסכנוא) T (מסכנוא) [ מסכנוא) M (vocal)
העשָה איש הענוא אין איש G (דמאנואט) אין איש אין איש אין איש M (vocal)
fin ] + 2 stichoi G (4a) (elab)
```

G: (a) πενία ἄνδρα ταπεινοῖ, (b) χεῖρες δὲ ἀνδρείων πλουτίζουσιν. "(a) Poverty humbles a man, (b) but the hands of the vigorous make [them]

בֵּן מַשְׂבֵּיל נִרְדָּם בַּקָּצִּיר בֵּן מֵבְישׁ: 6 בְּרָכוֹת לְרָאשׁ צַדֵּיק וּפִּי רְשָׁעִים יְכַסֶּה חָמֶס: 7 זַכָּר צַדִּיק לִבְרָכֶה וְשֵׁם רְשָׁעִים יִרְקָב: 8 חַכַם־לֵב יַקַּח מִצְוֹת וֶאָוֵיל שְׁפָתַים יִלְבֵט: 9 הוֹלֵך בַּתּם יֵלֶך בָּטַח

rich." G makes Hebrew ראש (understood as ראש) ) the subject and associates איז (subdues," "forces"). This allows for G's paraphrase, "poverty humbles a man." This is better than M's pointing, which would have to be translated, "A poor man makes a deceitful hand." רמיה is not represented in G. The choice of מאספונשי for G's ("diligent") introduces a moral factor, but subtly so, since the Greek word embraces a broad range of pragmatic and moral virtues.

V: (a) egestatem operata est manus remissa ("The slothful hand has produced poverty")  $\approx$  G. V rightly understands רמיה as "slothful."

S: איברא א הארא א הארא ("Poverty humbles a man") = G.

 $T^{L}$ : מסכנוא ממככא לגברא רמייה ("Poverty humbles a deceitful man") = S, but representing ממכתא ( $T^{Z}$  ממכתא is a graphic error.)

### 10:4*a*

G: (a) υίος πεπαιδευμένος σοφος ἔσται, (b) τῷ δὲ ἄφρονι διαχόνῳ χρήσεται. "(a) An educated son will be wise, (b) and will use the fool as a servant." G 10:4*a*b ≈ M 11:29b ≠ G 11:29b. The use of διάχονος shows 10:4*a* to be Greek in origin. A Hebrew text would have used  $\forall z = \forall z$  for "servant" in this context, a word never rendered by διάχονος. This addition is most closely connected to the next verse.

### 10:5

G: (a) διεσώθη ἀπὸ καύματος υἰὸς νοήμων, (b) ἀνεμόφθορος δὲ γίνεται ἐν ἀμήτῷ υἰὸς παράνομος. "(a) A thoughtful son is saved from heat, (b) but a lawless son becomes wind-blasted in the harvest." G radically recasts the Hebrew using some unique lexical correspondences. καύματος ("heat") paraphrases 𝔅 𝔅 𝔅 (lit. "summer"). διασώζειν ("preserve [through a danger]") is loosely related semantically to the rare ¬𝔅 ("gather"), which is translated differently elsewhere. In view of the metaphor in 10:5b, the idea in 10:5a may be that a thoughtful son, like good grain, is "gathered" or harvested for his protection in timely fashion, before he is desiccated by the heat, whereas a lawless one is wind-blasted and blown away. Possibly G's ἀνεμόφθορος reflects ζτΓ ("blown away"), as in Isa 19:7,

or a graphically closer נרדף ("pursued") for M's נרדם ("drowses off") (Mezzacasa). But given that the translator is deliberately reshaping the picture, no Hebrew variants can be securely recovered. The purpose of the paraphrase is to fit the Egyptian climate better, for summer is not the harvest time in Egypt. Note also that the blasted ears of corn in Pharaoh's dream are called ἀνεμόφθοροι (Gen 41:6, 7, etc.).

G-Proverbs overall gives less attention than M to the flaw of laziness. G 10:26 substitutes lawlessness for laziness and reshapes the proverb strongly, as here. G 19:15 substitutes timidity for laziness. G 19:24, which in Hebrew concerns the sluggard, becomes a warning against bribes. G 20:13 condemns "babbling" rather than (excessive) sleep. G 21:26 replaces the sluggard with the impious. G 24:30–34 changes the emphasis from laziness to folly. G deemphasizes the dangers of laziness, a practical failing, in favor of moral virtues.

### 10:6

G: (a) εὐλογία κυρίου ἐπὶ κεφαλὴν δικαίου, (b) στόμα δὲ ἀσεβῶν καλύψει πένθος ἄωρον. "(a) The blessing of the Lord is on the head of the just, (b) but untimely grief will cover the mouth of the wicked." G expands M's "blessings" into εὐλογία κυρίου ("blessing of the Lord"), taken from 10:22, to emphasize God's role in retribution (Dick 1990, 27–28). (Similarly V<sup>MSS</sup> benedictio domini.) G interprets πασ αενθος ἄωρον ("untimely grief") because it assumes that 10:6b alludes to the evildoers' future punishment rather than to their deeds. (M 10:6b is actually to be translated, "but the mouth of the wicked covers up lawlessness.")

### 10:7

(a) μνήμη δικαίων μετ' έγκωμίων, (b) ὄνομα δὲ ἀσεβοῦς σβέννυται. "(a) The memory of the just [pl] is praised [lit. "with praises"], (b) but the name of the impious is extinguished." ברכה is uniquely translated as ἐγκωμίων to direct one's attention to the funeral eulogy (BAP). σβέννυται assimilates to 13:9b, as well as 20:20 and 24:20b. Turn is more appropriate in the other verses, which use metaphors of light and fire.

### 10:8

G: (a) = M; (b) ό δὲ ἄστεγος χείλεσιν σκολιάζων ὑποσκελισθήσεται. "(b) But he who is unguarded with (his) lips, being crooked, will stumble."

G's ἄστεγος is lit. "unroofed," "exposed" (GELS: "unguarded"; MGELS: "not being able to keep silence"). In 26:28, στόμα δὲ ἄστεγον corresponds

### 10:9-12

#### PROVERBS

וּמְעַמֵּשׁ דְּרָכְּיו יִוָּדֵעַ: 10 לְּרֵץ עֵיוִ יִתֵּן עַצְּבֶת וּוְתוֹכַחַת מְגֵלָה שָׁלוֹם': 11 מְקוֹר חֵיִים פֵּי צַדֶּיק וּפִּי רְשָׁעִים יְכַסֶּה חָמֶס: 12 שֵׂנְאָה הְּעוֹרֵר

to שה חלק ("slippery mouth"), meaning seductive and deceitful speech. In the present verse, G is trying to say precisely what kind of fool is in question, as in 10:14b. G 10:8 seems to be influenced by mention of מעקש שפתיו ("one who makes his lips stubborn") in 19:1, even though that verse is lacking in G. Or perhaps מעקש intruded from the next verse.

V: (a) = M; (b) *stultus caeditur labiis* ("but the fool is beaten with the lips"); similarly in 10:10.

S (T) האאנג ("seized") for ילבט is a guess from context. It is unlikely to reflect the graphically distant ילכד, contrary to Pinkuss and BHQ. It is followed by T.

### 10:9

Toy, BHS, and others would sharpen the antithesis by emending M's "יָרוֹעַ ("be found out") to יָרוֹעַ ("suffer"), as in 11:15 and 13:20. This is a possible reading, but since the disgrace of discovery would be an adequate threat, the change is not necessary. עודע means "be recognized for what one is," as in Sir 12:8 and Jer 28:9. Prov 10:9b means, "while he who makes his ways crooked will be found out."

# 10:10 ותוכחת מגלה שלום אותוכחת מגלה שלום יותוכחת מגלה שלום (א לא פֿגליַאָני אָצָבָט ( אותוכחת מגלה שלום M (vertical dittog from 10:8)

M 10:10b is meaningful in itself, but the verse is nevertheless awkward and asymmetrical. Stich 10:10a describes how one undesirable type of person affects others, while stich 10:10b speaks of the punishment awaiting a different unworthy type. It is hard to relate the stichoi to each other or even to explain them as a disjointed proverb (on which see ABP 2.494–97). There is a good chance that a different Hebrew underlies G 10:10b; see below. There would be no reason for the translator to avoid M's 10:10b entirely if it were in G's source text. Possible retroversions of G 10:10b are שלים ישלים ("and he who reproves directly makes peace"; Gemser); ותוכחת מגלה שלום ("and open reproof [makes] peace"; Ehrlich); ומוכיח יעשה שלום ("and the reprover makes peace"; Clifford; cf. Isa 27:5). I choose Ehrlich's proposal for its resemblance to Prov 27:5 and for the antithesis it forms to 10:10a. In the retroverted text, שלום is the second object of the verb יתן. However, G handles Prov 27:5 differently.

As emended, the verse reads: "He who winks his eye causes grief, while he who reproves openly makes peace." The second stich balances the first nicely. Winking the eye is a hostile, sneaky gesture; see 6:13 and 16:30. Far better to rebuke someone frankly and honestly: "Better an open rebuke than hidden love" (27:5). M 10:10b was very likely a vertical dittography from 10:8b.

G: (a) ὁ ἐννεύων ὀφθαλμοῖς μετὰ δόλου συνάγει ἀνδράσι λύπας, (b) ὁ δὲ ἐλέγχων μετὰ παρρησίας εἰρηνοποιεῖ. "(a) He who signals with (his) eyes with deceit causes sorrows to men, (b) but he who reproves openly makes peace." G makes it clear that  $\gamma$   $\gamma$  ("squints," "winks") is a pernicious act. The antithesis in G's second line shows that the deceitful eye-signals are thought to be false signs of friendship. (Ben Sira [27:22–23] also interprets the gesture in this way.) G's "to men" makes it clear that harm to others rather than oneself is the issue here. (Possibly ἀνδράσι was added also for the rhythm; thus BAP.) BHQ says that G produced 10:10b for the sake of antithesis, but G does not elsewhere make such radical adjustments for that slight reason, and in any case G eliminates the parallel eye // lips. V follows M.

S:  $rest and reproves openly makes peace." S <math>\approx$  G, but not adding "to men." Since S does not follow G slavishly here, it provides some evidence for the different form of 10:10b.

T = M, but follows S 10:8 in translating מתאחד as ילבט.

10:11

G ἐν χειρὶ ("in [the] hand"), where M has 'בו' ("and the mouth"), is borrowed from 18:21. De Lagarde (with reference to Grabe) suggested that this is an inner-G error for ἐν χειλί ("in the lip"), but the singular is not used in G-Proverbs and would not be appropriate here. G again interprets Data a punishment, this time translating it as ἀπώλεια ("destruction") ≠ G 10:6.

מְדָגְיִם וְעֵל כָּל־פְּשָׁעִׁים תְּכַסֶּה אַהֲבֶה: 13 בְּשִׂפְתֵי גָבוֹן תִּמְצֵא חְכְמֶה וְשֵׁבֶט לְגֵו חַסַר־לֵב: 14 חַכָמִים יִצְפְּנוּ־דֶעַת וּפִי־אָׁוִיל מְחִתָּה קַרְבֶה: 15 הַוֹן עֲשִׁיר קִרְיֵת עֻזְּוֹ מְחִתָּת דַּלֵּים רֵישֶׁם: 16 פְּעֻלַּת צַדְּיק לְחַיֶּים תְּבוּאַת רְשָׁע לְחַטֶּאת: 17 אַרַח לֲחַיִּים שׁוֹמֵר מוּסֵר וְעוֹזֵב תּוֹכַחַת מַתְעֶה: 18 מְכַסֶּה שִׁנְאָה שִׂפְתֵי־שֶׁקֶר וּמוֹצָא דְׁבָּה

10:12

G: (a) μῖσος ἐγείρει νεῖκος, (b) πάντας δὲ τοὺς μὴ φιλονεικοῦντας καλύπτει φιλία. "(a) Hatred stirs up conflict, (b) but love covers [i.e., protects] all who are not contention-loving." G assumes that פשעים means "offenders" (i.e., [ἐψψ], rephrased for the sake of tighter parallelism as "contention-loving." But, puzzled by the notion that any virtue could "cover"—i.e., hide or protect—offenders of any sort, G adds a negative to produce a more acceptable sentiment.

S: אלאאס הסא אבשא כבלקסס בא ארשא מעוג (a) Hatred will provoke strife, (b) and shame will cover all evildoers." Like G, S misunderstands פשעים as "offenders" and "covers" as protects, but he solves the logical problem this causes by converting "love" into "shame."

10:13

G: (a) δς ἐκ χειλέων προφέρει σοφίαν, (b) ῥάβδῷ τύπτει ἄνδρα ἀκάρδιον. "(a) The one who brings forth wisdom from (his) lips (b) smites the mindless man with a rod." In G, the syntax is rearranged but no variants are indicated. מוֹצָא ("will be found") is associated with מעא ("brings forth"). מעא ("find") is associated with (מוֹצָיא = ) מֹצָא (find") also in 8:35 and 18:22*a*. G conflates the two stichoi of 10:13 into one sentence and equates the speaker of wisdom with the one who smites the fool.

10:14

G: (a) σοφοί κρύψουσιν αἴσθησιν, (b) στόμα δὲ προπετοῦς ἐγγίζει συντριβῆ.

"(a) The wise man will hide knowledge (b) but the mouth of the reckless approaches [or 'brings (him) near to'] destruction." M's אויל is rendered, more narrowly, as "reckless." אויל was understood similarly in 10:8 but there translated as  $\delta$   $\delta$ è ਕਰτεγος ("unguarded").

S, who knows very well what אויל means, chooses to follow G's "reckless" and translates حصن محمد. This makes the proverb more focused by saying what kind of fool is in view. This is something that G often does in spite of his tendency to tighten parallelism. S chooses to follow G's example for what can only be aesthetic reasons.

### 10:15

G: (a) κτῆσις πλουσίων πόλις ὀχυρά, (b) συντριβὴ δὲ ἀσθενῶν [G<sup>Ant V 336</sup> <sup>613</sup>; var: ἀσεβῶν G<sup>BSA</sup>] πενία."(a) The property of the wealthy is a strong city, (b) but poverty is the disaster of the weak [var: 'impious']." By the majority reading ἀσεβῶν, G moralizes the severe observation in 10:15b. However, the variant ἀσθενῶν is supported by the third-century Antinoe Papyrus and is correct; cf. 21:13 (Zuntz 1956, 161). Poverty in this case is not a punishment but a misfortune. This example shows how the processes and motives usually assigned to the translator often persist in subsequent transmission.

10:17

G: (a) όδοὺς δικαίας ζωῆς φυλάσσει παιδεία, (b) παιδεία δὲ ἀνεξέλεγκτος πλανᾶται. "(a) Education guards the right ways of life, (b) but the unchastised education goes astray" (>  $G^{Asc Ant}$ ).  $G^{Ant}$  10:17a reads όδὸς ζωῆς φυλάσσει ἀκάκους ("The way of life guards the innocent"). Zuntz (1956, 161–62) supports this reading.

G ignores or lacks the conjunction of אועוזב. G construes מוסר as the subject in 10:17a and repeats παιδεία in 10:17b, making παιδεία the topic of the couplet. G then reads the words מוסר עוזב תוכחת מתעה as an independent clause (in violation of the correct stichometry) and takes it to mean, "education which leaves off [i.e., 'neglects to provide'] chastisement goes astray." ἀνεξέλεγχτος here (but not in 25:3) means "unchastised" or "without chastisement" (ἐξελέγχειν means "rebuke" in Mic 4:3; Wis 12:17; and 4 Macc 2:13). Translations of ἀνεξέλεγχτος vary: "incapable of disproof or criticism" (GELS); "inscrutable" (MGELS); "sans examen" (BAP), which does not quite fit here; "unchastened" (NETS); "ohne Überprüfung" (SD). G adds διχαίας for moral clarity.

ַהָּוּא כְסִיל: 19 בְּרָב דֲבָרִים לָא יֶחְדַּל־פָּשַׁע וְחֹשֵׂדְ שְׂפָתְיו מַשְׂבִּיל: 20 בֶּסֶף וֲבְחָר לְשִׁוֹן צַדֵּיק לֵב רְשָׁעֵים בִּמְעֵט: 21 שִׂפְתֵי צַדִּיק יִרְעַוּ רַבֵּים וֶאֶוילִים בַּחֲסַר־לֵב יָמְוּתוּ: 22 בִּרְבַּת יֱהוָה הֵיא תַעֲשָׁיר וְלָאֹ־

### 10:18

G: (a) καλύπτουσιν ἔχθραν χείλη δίκαια, (b) οἱ δὲ ἐκφέροντες λοιδορίας ἀφρονέστατοί εἰσιν. "(a) Righteous lips conceal hatred, (b) and he who brings forth abuse is most foolish."

G reverses M, which reads: "Deceitful lips cover up hatred, while the slanderer is a fool." The first stich means that dishonest people (for which "deceitful lips" is a synecdoche) speak honeyed flattery but are really concealing hatred. (S, correctly: حدر المحمد), "conceal [as in ambush] hostility"). Here and in 10:12, the translator seems to reverse the meaning of his source. The present verse cannot be a converse translation or a case of turning a rhetorical question into an implied negative indicative (see the comment on 1:7 and "negatives" in the index). If 10:18a were a rhetorical question, the implied indicative would be "Deceitful lips do not cover up hatred." It seems that G did not understand why it is wrong to conceal hatred and so converted "deceitful lips" into "righteous lips" with no justification in the Hebrew. But Proverbs does affirm concealment in some cases; see also 10:12b; 10:19; 11:13b; 17:9a.

### 10:19

G: (a) ἐκ πολυλογίας οὐκ ἐκφεύξῃ ἁμαρτίαν [G<sup>BSMSS</sup>; var: οὐκ εκφεύξετια ἁμαρτία G<sup>A 157</sup>], (b) φειδόμενος δὲ χειλέων νοήμων ἔσῃ. "(a) By many words you will not escape sin [var: 'sin will not escape'], (b) but, (by) being sparing of lips, you will be intelligent."

S: متعلم هم معمل معدال معدال

and G in tandem. He relies on G—actually,  $G^A$ —for oúx εxφεύξεται, but instead of replicating G's מָׁשָמָדוֹמ he goes back to the Hebrew for שָשָ which he understands as בֹּשָׁשָ.

### 10:20

G הבתוף ("burnt," "purged") understands נבחר as "assayed" (used of metallurgy), as in Aramaic and Isa 48:10 and Sir 4:17.

S has a puzzling אלא ("bile," "bitterness") for M's כמעט, a word that is correctly translated elsewhere as סמעל or איסי, Pinkuss emends to איסי, which he translates "Herabfallendes, Schlackeslag," meant as an antonym of מחתא; whence T מחתא. The Syriac dictionaries and Jastrow, however, do not assign מחתא the required meaning. Healey 1991 translates T "a deficiency" (presumably an A-passive participle of 1991 translates T "a deficiency" (presumably an A-passive participle of געטל). T would then be a reasonable translation of געטל), and this might be a transcription of געמלא.

10:21 אידעו א S (דעו (פדעמבעב) (פדענמבעי) א ידעו (פרע מדמנס) (G (לבונמדע מו) א ידעו ) א לב M S (גער י) G (transf to 22)

G: (a) χείλη δικαίων ἐπίσταται ὑψηλά, (b) oἱ δὲ ἄφρονες ἐν ἐνδεία τελευτῶσιν. "(a) The lips of the righteous [pl] understand lofty things, (b) but the fools die in neediness." G ἐπίσταται = 1μτ' construed as 1μτ' ("will understand"). A better vocalization would be 1ττ" ("inform," "teach"). The latter is supported by V *erudiunt* ("instruct"). The correct translation of the reconstructed Hebrew is "The lips of the righteous teach many people." This is a valid text but not, I think, the archetype. C. Gordon (1930, 392) thinks V is based on the rabbinic understanding of Torah as the food par excellence, so that "feed" is a metaphor for teaching Torah (an interpretation maintained by David Qimḥi). A simpler explanation is that both G and V represent 1τ", though they vocalize it differently.

G ביאסצוק = ("in lack"). G does not represent לב ("heart"), but there is an extra "heart" in 10:22, which ends in בי אמסלוק. It seems that a copyist incorrectly transferred לב into the next line and the translator adjusted it to the Greek syntax.

S's הדער ("conciliatory") associates אבן ("desire," etc.). T's בל ("desire," etc.). T's conciliatory") derives רעיין

### 10:22

G: (a) εὐλογία κυρίου ἐπὶ κεφαλὴν δικαίου. (b) αὕτη πλουτίζει, (c) καὶ οὐ μὴ προστεθῃ αὐτῃ λύπῃ ἐν καρδία. "(a) The blessing of the Lord is on the

יוֹסָף עָצָב עִמֲה 23: כִּשְׁחַוֹק לֵכְסִיל עֲשָׂוֹת זִמְּה וְׁחָכְמָה לְאָישׁ הְּבוּגַה: 24 מְגוֹרַת רָשָׁע הֵיא תְבוֹאֶנוּ וְתַאֲוָת צַדִּיקֵים יִתֵּן: 25 כַּעֲבוֹר סוּפָה וְאֵין רָשֶׁע וְצַדִּיק יְסוֹד עוֹלֶם: 26 כַּחָׂמֶץ | לַשִּׁנִּיִם וְכָעָשֶׁן לְעֵינְיִם כֵּן הֶׁעָצֵׁל לְשֹׁלְחֵיו: 27 יִרְאַת יֵהוָה תּוֹסִיף יָמֵים

head of the righteous [sg]. (b) It makes rich, (c) and misery is not added to it in the heart." G adds ἐπὶ κεφαλὴν δικαίου (from 10:6) for moral clarity, lest one think that just anyone's wealth is evidence of divine favor; see Giese 1992a, 417. The superfluous "in the heart" was likely displaced in the Hebrew of the source text from 10:21b.

While G does not denigrate wealth as such or deny that wisdom brings material rewards, it does teach, often by adding words, that wealth is to be valued only when accompanied by righteousness and wisdom. This is a matter of emphasis, not a fundamental departure from the Hebrew. There is nothing specifically Stoic in G's ideas of wisdom and wealth (Giese 1992a, 417).

G also has a tendency to remove or temper warnings against overwork and "haste." Whereas M says that further *striving* (עצב) does not add to what God grants, G points out, rather unnecessarily, that misery is not added to God's blessing. Similar modifications to this effect occur in G 23:4 and 28:20. Two proverbs that warn against haste and overwork are omitted altogether, namely 19:2 and 21:5, though the topic is not necessarily the motive for the omissions.

### 10:23 בשחוק [ M<sup>MSS</sup> G (לא $\gamma$ ל $\lambda\omega\tau$ ו) (graph $\rightarrow \rightarrow$ )

G: (a) ἐν γέλωτι ἄφρων πράσσει κακά, (b) ή δὲ σοφία ἀνδρὶ τίκτει φρόνησιν. "(a) In laughter a fool does evils, (b) but wisdom engenders discernment for a man." The source text in 10:23a had (a clearly erroneous) בשחוק.

S's جسر ("when he laughs") could represent either variant.

10:24

G: (a) ἐν ἀπωλεία ἀσεβὴς περιφέρεται, (b) ἐπιθυμία δὲ δικαίου δεκτή. "(a) A wicked (man) is tossed about by destruction, (b) but the desire of the righteous man is acceptable." It is difficult to associate G 10:24a with M or to reconstruct a different source text related to it. In 10:24b, δεκτή ("acceptable") paraphrases 'μ' understood as 'μ', on the assumption that

what is "granted" by God is "acceptable" to him. V *dabitur* ("shall be given") also treats the verb as a passive. See ABP ad loc.

After 10:24a, G<sup>A</sup> adds δουλεύσει δὲ ἄφρων φρονίμω, borrowed from 11:29b, and after 10:24b it adds καρδία δὲ ἀσεβοῦς ἐκλείψει, borrowed from 10:20b. Two new couplets result:

24a	The wicked man wanders about in destruction,
	and the fool will serve the prudent.
24b	The desire of the righteous is acceptable,
	but the intelligence [lit. "heart"] of the impious will cease.

This illustrates how even in inner-Greek transmission, a proverbial text can remain dynamic, modifying sayings and even producing new ones; see the comments on 10:15 and 11:16.

10:25 יסוד M S (הא<br/>למער באלאישה (elab]) סור (לאגאלועמג) (graph ר $-\tau$ 

G: (a) παραπορευομένης καταιγίδος ἀφανίζεται ἀσεβής, (b) δίκαιος δὲ ἐκκλίνας σώζεται εἰς τὸν αἰῶνα. "(a) When the whirlwind passes, the impious one disappears, (b) but the righteous, turning aside, is saved forever." ἐκκλίνας = , 'oις', a graphic error producing an ungrammatical sentence. σώζεται is added to specify the benefit that the righteous man receives from turning aside.

### 10:26

G: (a) ὥσπερ ὄμφαξ ὀδοῦσι βλαβερὸν καὶ καπνὸς ὄμμασιν, (b) οὕτως παρανομία τοῖς χρωμένοις αὐτήν. "(a) As unripe grapes are harmful to the teeth and smoke to the eyes, (b) so is lawlessness to those who practice it." G moralizes the verse by substituting lawlessness for laziness (see the comment on 10:5) and having the unworthy person rather than the sender suffer the consequences of the bad behavior.

וּשְׁנִוֹת רְשָׁעֵים תִּקְאְרְנָה: 28 תּוֹחֶלֶת צַּדִּיקִים שִׁמְחֶה וְתִקוּת רְשָׁעֵים תּאבִד: 29 מְעוֹז לַתּם דֶּרֶדְ יְהוֶה וֹמְחִתָּה לְפַׁעֲלֵי אֶוָן: 30 צַדִּיק לְעוֹלֶם בַּל־יִמְוֹט וֹרְשָׁעִים לָא יִשְׁפְנוּ־אֶרָץ: 31 פִּי־צַדִּיק יְנוּב חָרְמֵה וּלְשָׁוֹן תַּהְפָּכוֹת תִּכְּרֵת: 32 שִׂפְתֵי צַדִּיק <sup>ו</sup>יַבִּעַוּן רָצֵוֹן וּפֵי רְשָׁעִים תַּהְפָּכוֹת: 11:1 מֹאזְנֵי מֶרְמָה תּוֹעֲבַת יְהוֶה וְאֶבֶן שְׁלֵמֶה רְצוֹנוֹ: 2 בָּא־זָדוֹן וַיָּבָא קָלֶוֹן וֵאֶת־צְנוּעֵים חָרְמֵה: 3 תִּמַת יְשָׁרֵים תַּנְחָם זְסֶלֶף בּוֹנְדִים יִישָׁדֵם': 4 לֹא־יוֹעֵיל הוֹן בְּיוֹם עֶבְרֶה וֹּצְדָלָה תַּצִיל

### 10:29

A variant in 10:29a (G<sup>BA</sup>) is φόβος χυρίου ("fear of the Lord") for M's דרך יהוה. Dick (1990, 37) says that "fear of the Lord" is the translator's interpretation of the unusual "way of the Lord" and that G was perhaps influenced by 10:27. BAP observes that by repeating this phrase, G gives 10:27–29 a certain cohesiveness.

### 10:31

G renders נוב ("yields fruit") as ἀποστάζει ("drips") and translates תהפכות ("perversity") as ἀδίκου ("unjust man"), construing the Hebrew words as abstract-for-concrete tropes.

10:32 יובעון האסטדי<br/>לגנו) [ איבעון M S (גד $\rightarrow$ ר ב שון (graph  $\rightarrow$ ר ב)

Since lips cannot be said to "know" favor, as M has it, it is better to emend to "pour out," "express" (Toy, BHS, and others), as in 15:2 and 28, which also define the speech characteristics of different kinds of people. The emendation assumes only a  $\Gamma/\Gamma$  confusion, which is possible in the early square script.

G's ἀποστάζει is probably an etymologizing translation of μαμί. To be sure, ἀποστάζειν is not used for μαμί elsewhere, but G is inconsistent in its rendering of μαμί see Prov 15:2 (ἀναγγελεῖ) and 15:28 (ἀποκρίνεται). However, it could be repeated from the preceding verse to eliminate the difficulty posed by "know" in 10:32b.

### 11:1

 $G \approx M$  but explains שלמה ("complete") as dixauou ("just"); similarly S  $\sim ($ "just" [pl]). The translator highlights the moral message already inherent in the text.

### 11:3 ושדם [ $M^{Q}$ ] ושדם (graph ' $\rightarrow$ )

The *ketiv* is an error and impossible in context. We cannot determine which variant the versions had, because they would be forced to produce the sense of the *qere* even if they had the *ketiv*.

 $G^A$  (= OG): (a) ἀποθανὼν δίκαιος ἔλιπεν μετάμελον, (b) πρόχειρος δὲ γίνεται καὶ ἐπίχαρτος ἀσεβῶν ἀπώλεια. "(a) When the righteous man dies he leaves behind regret, (b) but the destruction of the wicked is speedy and joyful." That is to say, their destruction will evoke vindictive joy.

G has a double translation of תנחם (actually "guides them," from (נחה, (נחה): (1) באוגד, associating the word with הניח ("to place," "to leave") and (2) שבדמשבלסי, associating it with נחם ("regret"). Understanding as "regret" prompted the translator to connect מגרח with death (מו"ת), hence מתספמיליע. This verse is an example of how a translator can translate lexical elements rather literaly while putting them together in a "syntactic ensemble which is entirely free" (Barr 1979, 323). It is, moreover, an example of how the text may bear unexpected interpretations. The translator knew that הנחה means "lead" (6:22; 18:16) and was certainly familiar with the frequent הנחה. But, perhaps aware that the plain message of 11:3a will be restated two verses ahead, he chose to elicit other potential messages hidden in the verse in a technique used often in midrash.

### **11:4** M S ] > OG (unc, equal)

This verse is missing in OG. It is supplied in some MSS from  $\theta'$ . M 11:4 is a near-doublet of 10:2, which *is* translated in G. M 11:4 fits nicely in the proverb cluster 11:4–7, which contrasts the deliverance of the righteous with the destruction of the wicked. This fit could be evidence of originality, especially as there is no good way to account for the verse's loss in G. But literary assessments cut both ways, and the thematic appropriateness could have inspired a later scribe to insert the verse in proto-M. It is maintained in the HBCE text as at least native to proto-M.

Heidenheim, whom BHQ cites and provisionally accepts, says that the verse was omitted for dogmatic reasons. The verse was important for rabbinic interpreters as a prooftext for the doctrine of justification, ַמִמֶּוֶת: 5 צִּדְקַת הֲמִים הְיַשֵּׁר דַּרְבָּוֹ וֹּבְרִשְׁעָתוֹ יִפְּל רָשֶׁע: 6 צִדְקַת יֵשִׁרִים תַּצִּילֵם <sup>ר</sup>ּוּבְהַוָּתָם<sup>י</sup> בּּגְדִים יִלְכֵדוּ: 7 בְּמָוֹת אָדָם רָשָׁע תּאבַד הִקְוֵה וְתוֹחֶלֶת אוֹנִים אָבָדָה: 8 צַדִּיק מִצְּרָה נֶחֶלֶץ וַיָּבָא רָשֶׁע תַּחְתֵּיו: 9 בְּפָה חֲנֵף יַשְׁתַת רֵעֵהוּ וֹּבְדַעַת צַדִּיקִים יֵחָלֵצוּ: 10 בְּטוּב אַדִּיקִים תַּעַלִץ קַרְיָה וּבַאֲבָד רְשָׁעֵים רִנֶּה: 11 בְּבִרְפַת יֵשָׁרִים תָּרָים

which is to say, immortality, and a scribe opposed this belief. But there is no hint of antirabbinic (Sadducean?) tendencies elsewhere in the Hebrew transmission or the G translation. In any case, the doublet in 10:2 was not affected, nor was 11:7, which can easily be read as affirming immortality.

11:6 (דבהותם G (τῆ δὲ ἀπωλεία αὐτῶν) S (בבהַלתם V (in insidiis suis) ] ובהותם M (near haplog ב – ב

M, reading ובהות (followed by α' σ' θ' ἐν δὲ τῆ ἀδικία), must mean, "and in the deceit [or 'disaster'] of traitors they shall be captured." This is awkward, because the only antecedent available for ילכדו ("they shall be captured") is ישרים ("the upright") in 11:6a, which would make no sense. We should read ובהותם גרים. The change could have occurred by near haplography with the next word: ובהותם בגדים שו

רַוָּה refers both to wicked speech, whether deceits or threats, and to disaster. The differences among the versions show that they are independently rendering ובהותם. G: דָאָ לוֹ מֹשׁעוֹנוֹ מִשׁלוֹע ("and in their destruction"); S: בּאָלַתּם ("in their iniquity"); V: *in insidiis suis* ("in their snares"); T: בשלומיהון ("by their own retributions"; Healey 1991).

11:7 ותוחלת M S (סמבויסים א היא ה<br/>לת הלת הוחלת א ותוחלת א ותוחלת ה<br/>שבו א ותוחלת ה, orth)

G: (a) τελευτήσαντος ἀνδρὸς δικαίου οὐκ ὅλλυται ἐλπίς, (b) τὸ δὲ καύχημα τῶν ἀσεβῶν ὅλλυται. "(a) When a just man dies, hope is not lost, (b) but the pride of the wicked [pl] perishes." G 11:7a creates antithetical parallelism by converse translation (see the comment on 1:24). The device allows the translator to mirror M while introducing an allusion to the blessed afterlife awaiting the righteous. G associates אונים (actually "strength") with אָוֶן ("wickedness")—though this word does not have a plural; similarly Ps 94[93]:23. τὸ δὲ καύχημα = μίτρι ("and the glory/ praise of").

G suggests (though not unambiguously) that the righteous have a hope beyond death, while the pride of the wicked comes to an end. G 10:28 uses similar language to declare that the *hope* of the wicked perishes when they die. Allusions to the afterlife in G-Proverbs are few and subtle, so much so that Dick (1990, 28–32) concludes that G-Proverbs resists the idea. In my view, although G-Proverbs does not vigorously promote the afterlife belief as a solution to earthly injustice, it still hints at it in 10:28; 11:7; and 14:32. The Greek translation comes from a time the second century BCE—when the belief had just emerged but was not yet widely accepted. Qoh 3:21 knows of the claim that the soul rises at death but is skeptical about its validity.

S's אונים construes אונים as "wicked" = G, but הראישה = M.

11:8–10a MGS ] > 6Q30 (homoi אבד∩אבד)

מצרה M S (כך הסלך איז הארה א מצדה (כך אין האר מצרה א מצרה א הר ר= G (לא א מצרה) מצרה א מצרה א מ

G's פֿא טאָסָאָק ("from [the] hunt") = מַצָּדָה). (טְּאַדָה). (טְּאַדָה). (טָאַדָה). (טָאַדָה) Gen 27:3 ketiv.) G explains איבא מאסאטלוסידמו ("is delivered"), alluding to the surrender of the wicked to divine punishment.

6Q30 omits 11:8–10a. This is due to parablepsis from אבדה in 11:7 to ובאבד in the middle of 11:10 (Eshel 2003, 545).

11:9 איצלחו [ (נא בינה א S (נא בינה) א א יחלצו פור א יולא א א א יחלצו א יחלצו א יחלצו (צלח א דו א יחלצו א יו

G: (a) ἐν στόματι ἀσεβῶν παγἰς πολίταις, (b) αἴσθησις δὲ δικαίων εὔοδος. "(a) In the mouth of the wicked [pl] there is a trap for [their] countrymen, (b) but the knowledge of the just [pl] is successful." G's παγίς either reflects שחת ("pit," "pitfall"), though this equivalence is unique, or, more likely, ignores the ' of שחת . εὔοδος = ישחת (N- or H-stem) (de Lagarde); cf. Num 14:41. (M's שמול ישמא translated correctly in Prov 11:8.) G's variant arose by a chiastic transposition of the letters m, and the error necessitated adjustments in syntax. G πολίταις = ψ] in Prov 11:12; 24:28; Jer 29[36]:23; 31[38]:34. Prov 11:10 and 12 make the πόλις the setting of the event described.

S's ארעיה ("be strengthened") represents ארעיה = M (as in Pss 50:15; 91:15). T's ארעיה ("his land") (≠ S געבוית "his friend") is a careless reading of M's רעהו.

11:10b–11a MS] > OG (homoi קרת ∩קריה)

G: (10a) ἐν ἀγαθοῖς δικαίων κατώρθωσεν πόλις, (11b) στόμασιν δὲ ἀσεβῶν κατεσκάφη. "(10a) By the good deeds of the just [pl], the city

11:11-16

קֶרֶת וּבְפִי רְשָׁעִׁים תֵּהֶרֵס: 12 בָּז־לְרֵעֵּהוּ חֲסַר־לֵב וְאָישׁ תְּבוּנִוֹת יַחֲרִישׁ: 13 הוֹלֵך רֶכִיל מְגַפֶּׁה־סֵּוֹד וְנָאֶמַזְ־רוּחַ מְכַסֶּה דָבֶר: 14 בְּאֵין הַּחְבֵּלוֹת יִפָּל־עֵם וּתְשׁוּעָה בְּרֵב יוֹעֵץ: 15 רַע־יֵרוֹע כִּי־עָרַב זֶר וְשוֹגָא תֹקְעֵים בּוֹטֵחַ: 16 אֵשֶׁת־חֵז תִּתְמָך כְּבֵוֹד וְעָרִיצִים יִתְמְכוּ־עִשֶׁר:

is put aright, (11b) but by the mouths of the ungodly it is torn down." OG lacks M's 11:10b–11a. (They are supplied in G<sup>B\*S\*</sup>, from  $\theta'$ .) BHQ (and cf. de Waard 2006, 264–65) explains the treatment of 11:10b–11a as "telescoping" rather than accidental omission. More likely, the loss was due to homoioarkton from קרת סל קרת סל קרת the loss of does not have to be explained separately as "facilitation" (BHQ), since G does try to deal with קרת between the facilitation" (BHQ), since G does try to deal with אַגעדליף שנאריה, in order to sharpen the antithesis with G 11:11 (i.e., M 11:11b), which was originally contiguous to G 11:10 (i.e., M 11:10a). The translator thereby shaped the remaining stichoi into a nice couplet:

By the good deeds of the just, the city is set aright, but by the mouths of the ungodly it is torn down.

The  $\dot{\alpha}\gamma\alpha\theta oildot$  of the just, standing in antithesis with "mouths" (a destructive power) of the wicked, must refer to the good things the just do for others rather than to their own prosperity. An expanded form of M 11:10b appears in G 11:3b; see the comment there.

S translates אבי ("be strong"), explaining the apparent metaphor of a city "rejoicing." It is less likely that S is reading החלץ (BHQ, comparing S 10:9).

### 11:12

G, translating רעהו ("his fellow") as πολίτας ("citizens"), continues the theme of the polis from 11:9 and 10. Toy observes that "a political interpretation is natural in a city like Alexandria." This theme is introduced in G 24:28 as well.

### 11:13

G's expanded translation of σινεδρίω ("counsels in the assembly")—brings out two senses of the Hebrew word: "counsel"

(as in Jer 23:18, 22; Job 15:8) and "assembly" (as in Ps 89:8; Ezek 13:9; Job 19:19). The latter is elicited by the context of 11:9–13, namely, civic life. On the assembly in G-Proverbs, see the comments below on 15:22 and 31:23.

11:14 עם M S (ححت [ (ححت M S ( ححت) א כעלה [ (בראה מטווו:128)

G: (a) οἶς μὴ ὑπάρχει κυβέρνησις, πίπτουσιν ὥσπερ φύλλα, (b) σωτηρία δὲ ὑπάρχει ἐν πολλῆ βουλῆ. "(a) Those to whom there is no strategy [lit. 'steering'] fall like leaves, (b) but salvation exists in much counsel."

Where M has של עם ("a people will fall"), G has πίπτουσιν ὥσπερ φύλλα ("will fall like leaves"), which can hardly render anything but יִפּל גַעָלָה Jäger proposes that this is a contamination by association with יפל in 11:28. This is not in the source text of G 11:28, but the error in 11:14 could have happened earlier.

By changing "in many counselors" to "in much counsel," G shows a lack of enthusiasm for royal administrators. A similar motive explains "in the heart of" in 15:22 and 24:6.

S concretizes תחבלות ("stratagems") as תחבלות ("leader"); cf. S 1:5; similarly V gubernator ("helmsman," "leader") ≠ V 1:5; 20:18.

### 11:15

G: (a) πονηρός κακοποιεῖ, ὅταν συμμείξη δικαίω, (b) μισεῖ δὲ ἦχον ἀσφαλείας. "(a) A bad man does harm whenever he consorts with a just man, (b) and he hates the sound of safety." Several disturbances in G were occasioned by the assumption that  $V \neg$  means bad man. δικαίω is used to supply an antonym for  $V \neg$  thus understood.  $\Box \neg \nabla \nabla$  is associated with  $\nabla \nabla$  "to sound" (sc. the horn),  $\Box \nabla \nabla$  with  $\Box \nabla$  "safety."

### 11:16

G: (a) γυνὴ εὐχάριστος ἐγείρει [corr: ἐρείδει] ἀνδρὶ δόξαν, (b) θρόνος δὲ ἀτιμίας γυνὴ μισοῦσα δίκαια. (c) πλούτου ὀκνηροὶ ἐνδεεῖς γίνονται, (d) οἱ δὲ ἀνδρεῖοι ἐρείδονται πλούτῳ. "(a) A gracious woman arouses [corr: 'supports'] honor for a man, (b) but a woman who hates righteous things is a seat of dishonor. (c) Slothful men come to lack wealth, (d) but the vigorous are supported by wealth." ἐγείρει should be corrected to ἐρείδει

17 גַּמֵל גַפְשׁוֹ אֵישׁ חֶסֶד וְעֹבֵר שְׁאֵרוֹ אַרְזָרִי: 18 רָשְׁע עֹשֶׂה פְּעֻלַּת־ שְׁקֶר וְזֹרֵעַ צְׁדָקָה שֶׂכֶר אֲמֶת: 19 בֵּז־צְדָקָה לְחַיֵּים וּמְרַדֵּף רָעֲה לְמוֹתוֹ: 20 תּוֹעֲבַת יֶהוָה עִקְשִׁי־לֵב וֹּרְצוֹנוֹ תְּמֵימִי דֶרֶדְ: 21 יָד לֵיָד לֹא־יִגַּקָה הֶע וְזֶרַע צַדִּיקִים נִמְלֵט: 22 גָזֶם זֶהָב בְּאַף חֲזֵיר אָשֶׁה

("supports");  $\Gamma$  and P are similar in uncials. This reading is confirmed by S הממרבא.

In G 11:16a, ἀνδρί makes it clear that this woman is a support for her husband's honor, not only her own. Stich 11:16b is lacking in M. G's wording has affinities with its style elsewhere (cf. θρόνος αἰσθήσεως [ $\neq$  M], used of an intelligent man in 12:23a), and the stich is OG; see Tov 1990b, 46. Tov says that the translator added 11:16bc to provide antitheses to M's two lines. Stich 11:16c is connected to M only by the word "wealth." Its wording recalls G 9:13a, which may have been called to mind by the mention of the disgraceful woman in G 11:16b. Stich 11:16d follows M 11:16b, with  $\forall \mathsf{Crvvc} \mathsf{T}$  translated ἀνδρεῖοι. ἀνδρεῖος =  $\mathsf{T}$  in 10:4 and 13:4. The aural association with  $\mathsf{T}$  could have prompted this choice. It is very likely a deliberate move to eliminate the moral awkwardness of M, which according to G's understanding of  $\mathsf{T}$  wealth."

In its expanded form, G 11:16 forms two distinct couplets, 11:16ab and 11:16cd. The first concerns two contrary kinds of women. The statement about the virtuous woman was inspired by Prov 31, especially 31:23. The second (found also in  $V^{MSS}$ ) contrasts the lazy man with the vigorous or virtuous one. Compare the way that  $G^A$  creates two couplets in 10:24 by inserting two lines.

### 11:18

G: (a) ἀσεβὴς ποιεῖ ἔργα ἄδικα, (b) σπέρμα δὲ δικαίων μισθὸς ἀληθείας. "(a) The impious man does unjust work, (b) but the reward of truth [or 'the true reward'] is righteous seed [= 'righteous descendants']."

The implicit vocalization אָזָרָע rather than M's אָזָרָע is reflected in G (סתוֹב והבסמ'ש), V (seminanti ... iustitiam), S (סתוֹב והבסמ'ש), and T (ודוריע לצדקתא). The phraseology of M is used differently in G 11:21.

### 11:19 בן $M \subseteq G$ (טוֹטָ) אבן (graph $\Box \to \Box$ ) (בי ה)

G and S reflect בן צדקה, but בן צדקה is a peculiar and erroneous formation. G סומיע סאָל ("and the pursuit of") implicitly vocalizes אַמְרָדָן, where M has אַמְרָדָן ("and he who pursues"). Since S does not follow G in the construal of this word, S is more likely to be independent in witnessing to בן M is correct. Other verses that begin with כן, a linking device, are 1:19; 6:29; 24:14; 26:19; and 30:20; see ABP on 11:19. T = M.

V praeparat is based on M כן, understood to be from כן"ן.

Another possible variant, reflected in εἰς θάνατον and גבהא, is למות, is ("to death") for M's למותו (homoi ת $\cap$ ת). This is, however, uncertain, because both versions are flexible in their treatment of suffixes.

### 11:20

### 11:21

G: (a) χειρὶ χεῖρας ἐμβαλών οὐκ ἀτιμώρητος ἔσται, (b) ὁ δὲ σπείρων δικαιοσύνην λήμψεται μισθὸν πιστόν. "(a) He who strikes (his) hands to (another's) hand unjustly shall not be unpunished, (b) but he who sows righteousness shall receive a faithful reward." ἀδίκως corresponds to consonantal 𝒴, though the word order is violated. (For other examples of overriding word order, see 11:21; 15:26, 33; 16:15, 24; 17:12; 19:7*a*; 31:1). The translator has in mind clasping hands to seal an agreement; thus too in 16:5. However, ἐμβαλλεῖν is not used for this idiom in verses where the meaning of this gesture is clear (6:1; 11:15; 17:18; 22:26).

δ δε σπείρων δικαιοσύνην is based on וְזֹרֵעַ צְּדְקָה in M 11:18, though G translated differently there. In the present verse, it probably seemed more logical to say that the person who sows righteousness will receive ּיָּפָּה וְסָרַת טֵעַם: 23 תַּאָוַת צַדִּיקִים אַדְ־טָוֹב תִקוּת רְשָׁעִים עָבְרָה: 24 יֵשׁ מֱפַזַר וְנוֹסָף עֵוֹד וְחוֹשֵׂדְ מִיּשֶׁר אַדְּ־לְמַחְסוֹר: 25 גֶפָּשׁ־בְּרָכָה תְדֻשָׁן וֹמַרְוָה גַּם־הָוּא יּיִרְוֶא<sup>ַ</sup>: 26 מְגַעַ בָּר יִקְבָהוּ לְאֵוֹם וֹבְרָכָה

a reward than to say that his descendants will do so. G drew on 11:18 to provide a translation in 11:20 that he found less appropriate in 11:18 itself. (A CG MS has יְוֹרֵע in 11:21, too.) The translator may have also found it puzzling that the righteous man is said to "escape" without the danger being mentioned, so he substituted the broad promise of "a faithful reward."

S interprets "hand to hand" differently: אדבים אוד האבת ("He who stretches out his hand against his fellow"); similarly S 16:5. S 11:21b = M.

### 11:22

G: (a) ὥσπερ ἐνώτιον ἐν ῥινὶ ὑός, (b) οὕτως γυναικὶ κακόφρονι κάλλος. "(a) As a ring in the snout of a pig, (b) thus is beauty to a malevolent woman." OG omits "gold" (restored in G<sup>BS\*A</sup>), perhaps thinking that any ring would be ridiculous in those circumstances. G makes the analogy between ring and beauty explicit and emphasizes the woman's nastiness rather than her stupidity. (κακόφρων = ματί πι 19:19.) Cognates of κακαφρῶν mean "malevolent," "evil-minded" in T. Abr. A 14:11 and Sib. Or. 4.171.

S translates סרת as סרת ("stinking"), associating Hebrew סרת with the similar-sounding ("stink"). C. Hyun (2000, 71–77) maintains that phonetic resemblance was a guiding principle in word choice in S-Proverbs. In any case, for the ancient translator, homoiophony (which has a major role in midrash) was closer to a lexical principle—the association of apparently related words—than a merely phonetic one.

### 11:23

G translates אָבְרָה ("wrath") as מֹתסאצּוֹדמו ("perishes"). The translator may have found it unclear how the hope of the wicked could *be* wrath, as M has it, so he substituted a more predictable "perishes" (though the correspondence אָבר = מֹתסאלאטיגעי is unique). Compare G's treatment of the difficult predication in 11:18. Hexaplaric מֹתלאטדמו and מֹתצּבְעִדמו (ascribed to מא') shows a similar treatment of the word. M is correct. Both stichoi are examples of "blunt juxtaposition," on which see ABP on

13:1. The literal translation is, "The desire of the righteous is but good [noun]; the hope of the wicked is wrath"; see further ABP.

### 11:24

G: (a) εἰσὶν οῦ τὰ ἴδια σπείροντες πλείονα ποιοῦσιν, (b) εἰσὶν καὶ οῦ συνάγοντες ἐλαττονοῦνται. "(a) There are those who, sowing their own (property), make abundance, (b) and there are also those who gather (yet) lack." G attempts to pin down the meaning of a paradoxical and ambiguous saying. G construes the imagery as agricultural. מפור (withhold") is understood to mean the (purposeful) scattering of seed.

V: (a) *alii dividunt propria et ditiores fiunt* (b) *alii rapiunt non sua et semper in egestate sunt.* "(a) Some divide up their own property and become richer. (b) Others seize what is not theirs and are always in want." V gives the verse a moral message.

11:25 יוֹרָא (et qui inebriat) ] יוֹרָא (metath וור  $\rightarrow$  ווֹר א איַרָוא M

M's אוֹרָא should be emended to יוֹרָא ("be slaked"), to fit the subject מרוה ("he who gives drink"). M's vocalization calls for the translation "He who slakes (others') thirst [for wisdom]—he too teaches" (cf. Joseph Qimhi). The spelling with the א is anomalous whether the root is ה'ר"ה זי רר"ה The spelling is confirmed by the Masorah but normalized to יורה in many MSS and cited thus by the twelfth-century Spanish-Jewish commentators Joseph Qimhi and Moshe Qimhi.

 ַלְרָאשׁ מַשְׁבְּיר: 27 שִׂחֵר טוֹב יְבַקֵּשׁ רָצְוֹז וְדֹרֵשׁ רָעֲה תְבוֹאֶנּוּ 28 בּוֹטַח בְּעָשְׁרוֹ הַוּא יִפֶּל וְׁכָעָלֶה צַדִּיקִים יִפְרֶחוּ: 29 עוֹבֵר בֵּיתוֹ

favor," which is an equivalent of "blessing," its parallel. The other correspondences are common in Proverbs. This Hebrew text is an alternative proverb, equal to M in principle but too conjectural to belong in the apparatus. There is no basis for deciding whether it or M is the earlier.  $\sigma' \pi \rho \omega i \nu \sigma \varsigma$  ("early," said of rain) = M.

V's et qui inebriat ("and he who inebriates") represents יָרָוָא (ירוה = : יָרָוָא).

S: (a) = M; (b) אסב נאלגע נאלגע. "(b) and the curser will also be cursed." S associates מרוה מרוה with ארר "("curse") but does not show a variant.

T: ומאן דמאליף אף הוא יליף ("and he who teaches, he too will learn"). T understands 11:25b as ומוֹרֶה גַם הוּא יוֹרָה. T's understanding of both verbs is probably based on the rabbinic understanding of the metaphorical significance of "slaking" thirst, found already in Midrash Proverbs.

G: (a) ό συνέχων σῖτον ὑπολίποιτο αὐτὸν τοῖς ἔθνεσιν, (b) εὐλογία δὲ εἰς κεφαλὴν τοῦ μεταδιδόντος. "(a) May he who withholds grain leave it to the gentiles, (b) but a blessing is upon the head of him who shares." For M's monotone (b) the set of the set of the will leave it? or "may he leave it." or "may is an Aramaism, clearly erroneous. In 24:24, G (and S) understood שבק as "curse," and there would be no reason to mistake it here. There is no mechanical explanation for the additional  $\boldsymbol{\psi}$ . It must have been due to a copyist's expectation of a word meaning "abandon." Because of the error, was parsed as  $\boldsymbol{\zeta}$  with the set of the set of the error.

S: (a) אבלא כבהאס, לבלא ברהאס, (b) = M. "(a) He who withholds produce in distress shall abandon it to his enemies." S diverges from G by adding "in distress" to specify the circumstances of withholding grain and having the recipients of the grain be one's enemies rather than the nations. Being distant from G, S provides some independent support for ישבקהו.

11:27 אחר MS ( (דבאה) אחר G (דבאדמויט(transp הרש  $\rightarrow \square$ ) G's דבאדמויט("he who devises") = הרש, a chiastic transposition. Jan de Waard (1993, 250) believes that this sort of metathesis is a trans-

lational device rather than a scribal error. The change in this verse, he argues, was motivated by the uniqueness of שחר in the G-stem alongside the familiarity of the locution שחר + words for good or evil. But this would be a weak motivation. שרש is unproblematic, because D-stem verbs frequently have a corresponding participle in the G-stem (e.g., קוֹנֶה, קוֹנֶה, The translator would not have been obliged to parse consonantal שחר as a D-stem.

Other examples of exegetical metathesis, according to de Waard, are in Prov 24:21; 29:9; and 30:10. Exegetical metathesis is indeed a possibility, since ancient exegesis allows for the association of any two words that sound alike. However, it is almost impossible to distinguish exegetical from graphic metathesis.<sup>2</sup> It is not clear that there are any exegetical metatheses in M.

11:28 ובעלה M S (האך איזא העלה [ העלה א מעלה ה א הובעלה א היגאמע<br/>  $\beta$ מעלה (graph  $\Box \to \alpha$ 

G: (a) = M; (b)  $\delta \delta \dot{\epsilon} \dot{a}\nu\tau i\lambda a\mu\beta a\nu \delta\mu\epsilon \nu o\varsigma \delta ixa (\omega\nu, o vot co \dot{a}\nu a \tau \epsilon \lambda \epsilon \tilde{i}$ . "(b) but he who assists the righteous [pl]—he shall spring up." Though the correspondence would be unique, it is hard to see how  $\delta \delta \dot{\epsilon} \dot{a}\nu\tau i\lambda a\mu\beta a\nu \delta\mu\epsilon \nu o\varsigma$ could arise except as a rendering of ומעלה ("raise up," implying assistance) for M's ובעלה ("like foliage") (Jäger). See the comment on 11:14.

### 11:29

G: (a) ὁ μὴ συμπεριφερόμενος τῷ ἑαυτοῦ οἴκῷ κληρονομήσει ἄνεμον, (b) δουλεύσει δὲ ἄφρων φρονίμῷ. "(a) He who is not in concord with his own house will inherit wind. (b) And the fool will serve the intelligent." ὁ μὴ συμπεριφερόμενος is a converse translation of עוכר ("he who troubles"). συμπεριφερέσθαι refers to conjugal amity in Sir 25:1.

S: הבא באש באלא אנגאא, אנגאא איבראם איברא האש בראא אנגאא גלא באש בראא איברא איברא געס איברא געס איברא גערא (a) He who builds his house deceitfully will leave groans to his children. (b) He who does not remain tranquil in his house will allocate winds to his children. (c) And the fool will be slave to the wise." S 11:29ab is a double translation of M 11:29a. S 11:29a is too unlike M to be a later correction based

<sup>2.</sup> Even graphic metathesis may sometimes be intentional. 2 Sam 6:5 reads בכל עז ובשירים, which appears in 1 Chr 13:8, correctly, as בכל עז ובשירים. When the sibilants t and צ were interchanged, ובשירים became unintelligible. The t was changed to ', and ברושים (perhaps written *defectiva*) was turned into a type of tree, ברושים ("cypresses"), to accord with *עצי* See Junker 1936, 163–64.

יִנְחַל־רְוּחַ וְעָבֶד אֶׁוֹיל לַחֲכַם־לֵב: 30 פְּרִי־אֲדִיק עֵץ חַיֵּים וְלֹקֵח <sup>ר</sup>ְנְפָשׁוֹת<sup>י</sup> חָבֶם: 31 הֵן אֲדִיק בָּאָרֶץ יְשָׁלֶם אַׁף כְּי־רָשֶׁע וְחוֹטֵא: 12:1 אֹהֵב מוּסָר אַהֵב דֶעַת וְשֹׁגֵא תוֹכַחַת בְּעַר: 2 טוֹב יְפִיק רָצוֹן מיְהוֶה וְאָישׁ מְזִמְוֹת יַרְשִׁיַעַ: 3 לֹא־יִכְּוֹן אָדֶם בְּרֵשֹע וְשִׁרָשׁ צַׁדִּיקִים בַּלֹימִוֹט: 4 אֵשֶׁת-חֵיל עֲטֶרֶת בַּעְלֵה וּכְרָקָב בְּעַצְמוֹתֵיו מְבִישֵׁה:

on G (contrary to de Lagarde and Pinkuss; see Joosten 1995, 67). Rather, the translator is influenced by G 11:29a, from which he gets "his children"; see Joosten 1995, 66–67. At the same time, S must be looking at the Hebrew and implicitly vocalizing ינָקל.

11:30 (מפרי האווים, א מפרי הפרי השרי מפרי מפרי מפרי מפרי אמ<br/>מMS (מפ $\to$ ש<br/>אוים, אמפרי מפרי  $M^{\rm A\,Y}$  נפשות M^L (vocal)

חכם M ] המס ( $\pi$ מףמעטעשט) S (הבהא) (graph  $\rightarrow$  המס ( $\sigma \rightarrow \sigma$ ,  $\sigma \rightarrow \sigma$ )

 $S \approx M$ , but  $\pi = \pi \Delta$ . Since S does not follow G in other divergences here, it provides independent evidence for this reading.

11:31

G: (a) eỉ ὁ μèν δίκαιος μόλις σῷζεται, (b) ὁ ἀσεβὴς καὶ ἁμαρτωλὸς ποῦ φανεĩται; "(a) But if the righteous man is barely saved, (b) where shall the wicked and sinner appear?" (quoted in 1 Pet 4:18) M means, "Since the righteous man receives what he deserves on the earth, how much the

more so do the wicked man and the sinner!" M's יָשָׁלָם means, "receives what he deserves," literally "is repaid," for good or ill. The translator understood ישלם to mean "be saved" or "kept safe," by association with understood ישלם. (Similarly he translates ישלם [M ישלם] as טָאַמוּטוּט (Similarly he translates ישלם it would make no sense to say that the wicked will "be saved," the translator added "barely" in 11:31a, so as to maintain the *a fortiori* structure and deny that the wicked can possibly escape disaster. On G μόλις, see Barr 1975. G's version of 11:31 assumes that all human beings are flawed; consequently even the righteous are *barely* saved.

S follows G slavishly, even in the treatment of "wicked and sinner" as a hendiadys and giving it a singular verb. Where M has ישלם, S has, uniquely, ישלם ("lives"). S is actually glossing G  $\sigma \phi \zeta \varepsilon \tau \alpha \iota$ . See further Weitzman 1999, 255–56.

12:2

G: (a) κρείσσων ὁ εὑρὼν χάριν παρὰ κυρίω, (b) ἀνὴρ δὲ παράνομος παρασιωπηθήσεται. "(a) Better is the one who finds favor with the Lord, (b) but the lawless man will be passed over in silence." The awkward Greek of 12:2a may be understood as comparing the favored man in 12:2a to the "wicked and sinner" of 12:1b, but the comparative structure is not carried through in the second stich. Jäger retroverts παρασιωπηθήσεται to  $\psi$  τητις, but this is graphically distant from  $\psi$ . More likely, the translator understood  $\psi$  (by diathesis) as "be convicted," hence "be silenced in court."

### 12:3

G's ἐξ ἀνόμου = ΣΓΡΟΥΤΑΙ (M). The preposition I is represented by ἐx six times out of 195 occurrences in G-Proverbs. It can be translated this way especially when it means "by." In 12:3b, ἐξαρθήσονται ("be removed") for civil not totter") is intended to improve the logic, since a tree's root does not actually totter.

5 מַחְשְׁבִוֹת צַדִּיקֵים מִשְׁפֵּט תַּחְבָּלְוֹת רְשָׁעֵים מִרְמֶה: 6 דִּבְרֵי רְשָׁעֵים אֶרְב־דֶם וּפִי יְשָׁרִים יַצִּילֵם: 7 הָפָּוֹדְ רְשָׁעֵים וְאֵינֶם וּבֵית צַדִּיקֵים יַעֲמְד: 8 לְפִי־שָׁרְלוֹ יְהֻלַּל־אֵישׁ וְנַעֲוֵה־לֵב יִהְיֶה לְבְוּז: 9 טוֹב וַמְלָה וּוַעֲבֶר<sup>ּז</sup> לְוֹ וּמִמִּתְבַּבֵּׁד<sup>י</sup> וַחַסַר־לֶחֶם: 10 יוֹדֵעַ צַדִּיק נֶפָשׁ בְּהֶמְתֵּוֹ וְרַחַמֵי רְשָׁעִים אַכְזָרִי: 11 עַבֵּד אַדְמָתוֹ יְשְׂבֵּע־לֶחֶם וּמְרַדֵּף

### 12:5

 $G \approx M$ , but restructuring the syntax.  $S \approx M$ , but translating as מחבלות as מחבלות ("their perversities") to make it clear that the stratagems in question are evil.

### 12:6

G ≈ M, but where M 12:6a has "The words of the wicked are an ambush for blood," G resolves the metaphor as δόλιοι ("deceits"). This rendering is less ambiguous (the translator may have wondered how words can *be* an ambush) and also connects the present verse better to the themes of the preceding (stratagems, deceit) (BAP). Some G MSS partially adjust to M by adding εἰς αἶμα ("for blood"), which is, however, inappropriate after "deceits."

A credible conjectural emendation, suggested by Tur-Sinai (1947, 102) and accepted by ABP, is אפר אועבד ("[agricultural] produce"). The latter is a synonym of תבואה ("produce") in Josh 5:11, 12. עבר is written עבר in Arad 31.10, a spelling that could easily have been corrupted to עבד Read as emended, the saying would be translated "Better a lowly man who has produce than one who glorifies himself and lacks bread." This saying advocates working one's land, similarly to Prov 12:11; 27:23–27; and 28:19.

G δουλεύων ἑαυτῷ ("serving himself") implicitly vocalizes וְעָבֵד לוֹ. The translator may have thought it illogical that owning a slave would some-

how outweigh disgrace—a notion seemingly implied by M—whereas he could accept self-sufficiency as a virtue.

S المعدم ("who serves himself") implies the same vocalization as G, though S translates the verse with direct reference to the Hebrew; similarly V *et sufficiens sibi* ("and supports himself").

# **12:11** fin ] + 2 stichoi G (11*a*) V (partial dbl)

(G 12:11*a* should be read in conjunction with v 12.) G: (11*a*a) ὅς ἐστιν ήδὺς ἐν οἶνων διατριβαῖς, (b) ἐν τοῖς ἑαυτοῦ ὀχυρώμασιν καταλείψει ἀτιμίαν. (12a) ἐπιθυμίαι ἀσεβῶν κακαί, (b) ai δὲ ῥίζαι τῶν εὐσεβῶν ἐν ὀχυρώμασιν. "(11*a*a) He who gets pleasure from diversions [or 'banquets'] of wine (b) will leave behind disgrace in his own fortresses. (12a) The desires of the wicked [pl] are bad, (b) but the roots of the pious [pl] are in fortresses."

G 12:11*a* is an example of a proverb created from a corrupt Hebrew text. It must be seen against the background of M 12:11–12. G 12:11 corresponds to M 12:11. The next couplet, G 12:11*a*, present also in V, picks up the theme of pursuing vanities from M 12:11, but it is primarily a distorted variant of M 12:12. G 12:12 is corrupt as well. It is not close enough to M 12:12 to be reckoned a later adjustment "correcting" 12:11*a*.

In G 12:11*a*a, סווע המר a corrupt doublet of המר. The latter is first translated broadly as os נסע אוז א או (where M has המר), then wine is being condemned. In 12:11*a*b, אמדמאנועבו = יתר); H-stem), a common correspondence. יתן is a synonym-variant of יתר, taken from the end of 12:12. (Compare ויתר in Ps 18:33 for ויתן in 2 Sam 22:33.) G's έν τοῖς ἑαυτοῦ ὀγυρώμασιν in 12:11ab corresponds to מצוד in M 12:12a. מצוד means "net," but the translator equates it with מצודה ("fortress"). In G 12:12b, ἐν ὀγυρώμασιν appears at the end of the verse. The rather unnatural notion of leaving something behind in *fortresses* (12:11*a*b) shows that the translator was constrained by his source text or, more precisely, by the way he understood it. Likewise, the strained metaphor of "roots" in "fortresses" (12:12b)-where they would in reality be no more secure than elsewhere—reveals the translator struggling with a variant Hebrew text. The source text of G 12:11a (the first doublet of M 12:12) read approximately (with implicit vocalization): המד המד בְמִצוּדִיו רַעִים יֹתָר. διατριβαίς was added to give a context to "wine"; רעים was understood as "bad" then translated more narrowly as "disgrace."

ַרִיקִים חֲסַר־לֵב: 12 חָמַד רָשָׁע מְצַוֹד רָעֵים וְשָׁרֶשׁ צַדִּיקִים יִתָּז: 13 בְּפָשַׁע שְׁפָתַיִם מוֹקַשׁ רֶע וַיֵּצֵא מִצְּרָה צַדְּיק: 14 מִפְּרֵי פִי־אָישׁ יִשְׂבַּע־טֶוֹב וּגְמְוּל יְדִי־אָדָם <sup>ר</sup>יָשָׁוּב לוֹ: 15 דֶרֶך אֶויל יָשָׁר בְּעֵינֶיו וְשֹׁמֵע לְעֵצֶה חָבֶם: 16 אֶוֹיל בַּיּוֹם <sup>ו</sup>יוֹדַע<sup>-</sup> בַּעְסֵוֹ וְכֹסֶה קָלָוֹן עָרוּם:

The example of G 12:11*a* shows that distance between M and G need not be due entirely to "free translation" or "interpretation," even when these are factors in the rendering, and moreover it shows that these processes do not necessarily preclude retroversion. A freely composed Greek proverb would not produce reminiscences of errors that could arise only in Hebrew. In the present passage, a fairly meaningful Greek text emerges from corrupt variants, which can only be partially retrieved.

G 12:12a, which is a doublet of G 12:12a, can be retroverted to, approximately. חמד רשעם רעים ושרש צדיקים במצודותיו. The translator picked up a few words from the corrupt doublet and shaped them into a couplet that condemns a king (since he alone would possess fortresses) who pursues frivolities, much in the spirit of Prov 31:4 and Qoh 10:16–19.

T's איתן ("endures") associates איתן with איתן ("enduring") but does not necessarily read the latter. Nevertheless, איתן is a reasonable conjectural emendation, since it makes good sense and fits the parallelism (Hitzig, Toy, ABP). By this reading, 12:12b is a rephrasing of 12:3b, with "endures" equivalent to "not totter."

**12:13** fin ] + 2 stichoi G (13*a*) (elab)

G: (a) δι' ἁμαρτίαν χειλέων ἐμπίπτει εἰς παγίδας ἁμαρτωλός, (b) ἐκφεύγει δὲ ἐξ αὐτῶν δίκαιος. "(a) Through sin of the lips, the sinner falls into traps, (b) but the righteous man escapes from them." G parses שוקש as an Hp-stem participle (מוקש, lit. "is trapped") and construes ארע as the evildoer. ἐξ αὐτῶν paraphrases מצרה מצרה and refers to the traps. α' σκῶλον ("obstacle") = M.

S באאטו: ("is seized") = מוּקַשׁ = G.

### 12:13a

G: (a) δ βλέπων λεĩα ἐλεηθήσεται, (b) δ δὲ συναντῶν ἐν πύλαις ἐκθλίψει ψυχάς. "(a) He who gives gentle looks will receive mercy, (b) but he who meets (men) in the gates will afflict souls." The additional couplet draws a contrast between mild men and litigious ones. It also gives an example of "sin of the lips" mentioned in verse 13 and its opposite. (συναντῶν implies seeking a quarrel; BAP.) The idioms of the addition do not reflect a Hebrew source text.

# 12:14 ישוב $M^{K}$ ] ישיב $M^{Q}$ (graph $\rightarrow$ )

The intransitive *ketiv* fits the context better than the causative-transitive *qere*, since the latter lacks a subject. It is, however, possible that ישיב is the original, with God as implicit subject, while ישוב is an attempted improvement, facilitated by graphic similarity. The versions are indeterminate.

G: (a) ἀπὸ καρπῶν στόματος ψυχὴ ἀνδρὸς πλησθήσεται ἀγαθῶν, (b) ἀνταπόδομα δὲ χειλέων αὐτοῦ δοθήσεται αὐτῷ. "(a) From the fruits of the mouth, the soul of a man will be filled with good things, (b) and the recompense of his lips will be given to him." The added ψυχή means "appetite," like Hebrew **UD1**, and specifies just *what* will be "filled." χειλέων is a corruption of χειρῶν (de Lagarde, ref Grabe), though the appearance of the latter in G<sup>V 15</sup> SyrH is probably Hexaplaric and secondary. Compare the doublet of 12:14a in 13:2a.

12:16 איוֹדָע G (ἐξαγγέλλει) θ' (γνρωίσει) σ' (δηλώσει) S (כהגב) ] איוֹדָע M (vocal, gram)

M's אָדָע treats the verb in 12:16a as passive (hence: "The fool—his anger is known on the same day"), whereas G and S, as well as  $\theta'$  and  $\sigma'$ , treat it as transitive and causative, implicitly vocalizing it as an H-stem (hence: "The fool makes his anger known on the same day"). The latter gives a tighter parallel to the second stich, with both lines describing how one chooses to deal with his anger, whether by flaring up at an insult or by overlooking it.

17 יִפְּיַח אֲמוּנָה יַגַּיד אֲדָק וְעֵד שְׁקָרֵים מִרְמֲה: 18 יֵשׁ בְּוֹטֶה כְּמַדְקְרָוֹת חֶרֶב וּלְשׁוֹן חַכָמֵים מַרְפָּא: 19 שְׂפַת־אֲמֶת תִּכְּוֹן לָעֵד וְעַד־אַרְגִּיעָה לְשׁוֹן שְׁקֶר: 20 מִרְמָה בְּלֶב־חִרְשֵׁי רֶע וּלְיׁעֲצֵי שָׁלָוֹם שִׁמְחֶה: 21 לֹא־יְאֻנֶּה לַצַּדִּיק כָּל־אָוֶן וּרְשָׁעִים מֶלְאוּ רֵע: 22 תּוֹעֲבַת יֵהוָה שִׂפְתֵי־שֶׁקֶר וְעֹשֵׁי אֱמוּנְה רְצוֹנְוֹ: 23 אָדָם אֶרוּם כְּסֶה דְּעַת וְלֵב כְּסִילִים יִקְרָא אָוֶלֶת: 24 יִד־חָרוּצֵים תִּמְשָׁוֹל וּרְמִיֶה תִּהְיָה לָמַס:

### 12:17

G: (a) ἐπιδεικνυμένην πίστιν ἀπαγγέλλει δίκαιος, (b) ὁ δὲ μάρτυς τῶν ἀδίκων δόλιο. "(a) A righteous man declares what is clearly honest, (b) but the testimony of the wicked is deceitful." πίστις (like אמונה) means "honesty" as well as "faith" and "loyalty." G is using πίστις in the classical sense of "argument," "proof," hence "preuve convaincante" (BAP; see LSJ 1408a.II.2; cf. the usage in 3 Macc 3:10). 2 is understood as a metonymy (abstract for concrete) and taken as the subject of the sentence.

12:18 בוֹטָה M<sup>OrQ</sup>; בוֹטָח M<sup>MSS</sup>  $\theta'$  ( $\pi \epsilon \pi \circ i \theta \omega_{\varsigma}$ ) V (pungitur) (M<sup>Or</sup>: norm; M<sup>MSS</sup>,  $\theta'$ , V: graph  $\rightarrow \rightarrow$ )

There is evidence for a variant, בוטח, in  $\theta'$ , V, and some medieval Hebrew MSS—though it doesn't make much sense. This could be a case of MSS KR witnessing to an ancient variant, though such a slight graphic misreading could arise later and independently. The Oriental reading (see Baer 1880, 58) looks like an adjustment to the usual spelling of "utter," but both spellings continue in RH.

G: (a) εἰσὶν οῦ λέγοντες τιτρώσκουσιν μαχαίρα; (b) = M. "(a) There are those who wound with a sword when they speak." This is a reasonable expansion of M's metaphor.

12:19

G: (a) χείλη ἀληθινὰ κατορθοῖ μαρτυρίαν, (b) μάρτυς δὲ ταχὺς γλῶσσαν ἔχει ἀδικον. "(a) True lips give honest testimony [lit. 'make testimony straight'], (b) but a hasty witness has an unjust tongue." G uses the active of κατορθοῦν to render the N-stem of cit in 12:3 and 25:5 as well. G twice understands  $\forall$  as  $\forall$  as in 29:14, and associates  $\forall$  with  $\forall$  with  $\forall$  with 12:17 and encourages us to read 12:18 as a comment on testimony, true and false.

S's השמה ("hasty witness")  $\approx$  G.

12:21 אינאה [ אנאה G (מא לספיט S (שבאי ) (metath אנאה ), graphic assoc, theol)

G: (a) טֿא מֹףָלָסָדָ דָשְׁ אָזאָגעלש טֿטֿצֿא מֹטָגעָי; (b) = M. "(a) No injustice will please the righteous person." מֹרְלָב and יִדָּשָּׁה This probably arose by association with גָּשָׁה Perhaps M's flat promise that "no misfortune will happen to the righteous man" did not seem credible. The antithesis in M is sharper: the righteous avoiding suffering // the punishment of the wicked. Still, both G and M offer meaningful proverbs. G describes the good man's attitude, M the consequences of his righteousness.

12:23 אולת MS (אולת [ (גרייאא MS (אולת G (מֹףמוֹג)) אולת אולת (לו → ול

G: (a) מׁאָאָ סטעבדטֹג אָפָאָאָסטע גוֹסאָאָסבעג, (b) אַמאָאָלוּע אַפּאָטע טעמעדאָלַדעו מֹאָמוֹג. "(a) The intelligent man is a throne of knowledge, (b) but the heart of the foolish [pl] will meet with curses." G differs from M by (1) construing אָרָרא בסה ("throne"), (2) equating אָלוּת ("calls") with יקרא ("meets," "happens"), and (3) reading אָלוּת ("curses") in its source text. (דעל is translated דענ מָאָל in Deut 29:20.) Difference 1 gets rid of an apparent logical difficulty: that the wise man should conceal wisdom, whereas he presumably should be revealing it to all. Differences 2 and 3 are trivial errors. The result was a meaningful new proverb.

S follows G's בסוּסטד, then returns to M, which is not problematic in stich b. אולת is a broad translation of אולת.

12:24

G: (a) χεὶρ ἐκλεκτῶν κρατήσει εὐχερῶς, (b) δόλιοι δὲ ἔσονται εἰς προνομήν. "(a) The hand of the chosen will readily rule, (b) but the deceitful will be for plunder." The use of ἐκλεκτός for μητις ("choice gold") and προνομή for σι ("forced labor") are unique. The former is based on an etymological association with the homonym "choice gold." It also has a

25 דְּאָגֶה בְּלֶב־אֵישׁ יַשְׁחֶנְּה וְדָבֶר טִוֹב יְשַׂמְחֶנָּה: 26 יְתַר מֵרַעֵּהוּ צַדְּיק וְדֶרֶדְ רְשָׁעֵים תַּתְעֵם: 27 לֹא־יַחֲרָדְ רְמִיֶּה צֵידְוֹ וְהוֹן־אָדֶם יָקֶר חְרוּץ: 28 בְּאִׂרַח־צְדָקָה חַיֵּיֵם וְדֶרֶדְ נְתִיבֶה <sup>ו</sup>אֶל<sup>ז</sup>-מֵוֶת: 13:1 בֵּן

theological motivation, for it is the *elect* who one day will rule, according to a common motif of eschatology.  $e\dot{\nu}\chi\epsilon\rho\tilde{\omega}\varsigma$  is added for emphasis and wordplay with  $\chi\epsilon\iota\rho$ .

### 12:25

G: (a) φοβερὸς λόγος καρδίαν ταράσσει ἀνδρὸς δικαίου, (b) ἀγγελία δὲ ἀγαθὴ εὐφραίνει αὐτόν. "(a) A frightening word disturbs the heart of a just man, (b) but a good report makes him happy." G 12:25a is a loose paraphrase of M's obscure Hebrew. It reasonably clarifies the first stich by reference to its antithesis in the second.

S: העראי העובא האין האין אראי גראי געראי געראי אין אראי געראי געראי (a) A frightening word frightens the heart of a man, (b) but a good word makes him happy." S is clearly dependent on G for the meaning of the sentence but tries to accommodate M by omitting διχαίου.

12:26 ייותר [ M יתר V (qui neglegit) (near dittog  $\rightarrow$  י)

G: (a) ἐπιγνώμων δίκαιος ἑαυτοῦ φίλος ἔσται, (b) ai δὲ γνῶμαι τῶν ἀσεβῶν ἀνεπιεικεῖς. (c) ἑμαρτάνοντας καταδιώξεται κακά, (d) ἡ δὲ ὁδὸς τῶν ἀσεβῶν πλανήσει αὐτούς. "(a) A just judge [or 'advisor' or 'arbitrator'] will be his own friend, (b) but the decisions of the wicked are unreasonable. (c) Evils will pursue sinners, (d) and the way of the wicked will lead them astray." Alternatively, 12:26a can be translated "A just man will be the judge of his own friend" (reading φίλου, with Baumgartner). Stichoi 12:26ad are OG, and 12:26bc have been inserted between them. M 12:26a is difficult, and G is virtually a guess. It is hard to see how מתר ימר ימר ימר έπιγνώμων, but the idea of 12:26b is based on G (not M) 13:10b: oi δὲ ἑαυτῶν ἐπιγνώμωνες σοφοί ("but those who are their own judges are wise"). Stich b (supplied with an asterisk instead of an obelus in SyrH) is absent in  $G^{BSV}$  and is later. Nevertheless, it is unlike M 12:26b, and it is not clear that it belongs to one of the Hexaplaric translators. It seems to be a later elaboration in the Greek transmission. G 12:26c = G 13:21a. G 12:26d = M 12:26b.

S renders יתר as אבא הלא הלבא ("gives good advice"), apparently taking a cue from פֿתועשעע, which can mean "advisor."

V: (a) qui neglegit damnum propter amicum iustus est; (b) = M. "(a) He

who overlooks a loss for the sake of a friend is just." V has יותר, which it understands as a D-stem (יְוָהֵר) meaning "concede," "overlook a loss," in accordance with rabbinic usage. It then treats the verb as a participle.

#### 12:27

The versions agree that the obscure החרך means "come upon," "meet": G פֿתודנטֹצָבדמו; V *invenient*; S נסתקבל, this commentary. It is likely that these all derive, directly or indirectly, from G's guess. הרוץ is understood as "pure gold" and translated literally in V (*auri pretium*) and T (הרא יקריא) and figuratively as "pure" in G (ממשמפל) and S (הרא יקריא) (BHQ).

12:28 אָל־ [ (ל-) V (*ad*) T (ל-) אַל־ [ אַל־ M (vocal, theol)

In 12:28b, V reads *iter autem devium ducit ad mortem* ("but a detour leads to death"), an interesting way of dealing with the obscure Hebrew "way of path."

S: (a) = M; (b)  $\rightarrow$  גבאלא לביאליה וואים. "(b) and the way of angry people is to death." S 12:28b = G; similarly T.

 $M^{Y}$ , V, S, and T vocalize אל מות *to* death," though אל מות, the usual M vocalization, would be ideologically acceptable to all. M has traditionally been understood to mean "the way of its [sc. righteousness's] path is [i.e., leads to] nondeath," that is to say, immortality (Delitzsch, etc.). The syntax, however, is impossible, because אל marks a negative wish, not an indicative. See further ABP.

נערה M S (באלא) ] אנועה (לא מ $\pi\omega\lambda\epsilon lq$ ) (graph אנערה א M S (האלא) (graph א וועה ) שו

G: (a) υἰὸς πανοῦργος ὑπήκοος πατρί, (b) υἰὸς δὲ ἀνήκοος ἐν ἀπωλεία. "(a) A clever son is obedient to [his] father, (b) but a disobedient son is in destruction." (G<sup>A\*</sup>, interestingly, has μητρί ["to (his) mother"] in 13:1a.) G seems to be parsing מוסר as a passive of יסר, "is (effectively) chastised," hence "is obedient." ἀνήκοος construes ממוסר חָכָם מַוּסַר אָב וְלֵץ לֹא־שָׁמַע גְּעָרֶה: 2 מִפְּרֵי פִי־אָישׁ יִאכַל טֵוֹב וְגָפֶשׁ בּגְדֵים חָמֶס: 3 נֹצֵר פִּיו שׁמֵר נַפְשָׁו פּשֵׁק שְׁפָתָיו מְחִתָּה־לְו: 4 מִתְאַוֶּה וָאַיִן נַפְשָׁו עָצֵל וְגֶפָשׁ חָרֵצִים מְדַשֵׁן: 5 דְבַר־שֶׁקָר יִשְׁנָא צַדְקָה תִּצַּר תִּם־דֶרֶד וְרִשְׁעָה תְּסַלֵּף

clause. בי מאטאנו ("dying"). A noun גועה is not attested, but it is structurally possible. (In Num 20:3, לי גערה מאטאנו is clause) קאר גערה קאטאנו וו האטאנו וו G, who knows what גערה וערה difficult (and erroneous) variant.

S: ארארב באראר מישא ג'א מסבל באראר מישא ג'א מסבל באראר מיא מישא ג'א מסבל. "(a) A wise son obeys his father, (b) but a bad son, who does not accept reproof, will perish." S combines components of M and G in 13:1b, using M's "reproof" together with G's "is in destruction" and its treatment of א שמע ל.

13:2 יאַכל M G (φάγεται) ] ישבע M<sup>MSS</sup> S (גמרבי) V (saturabitur) (syn, assim)

Several MSS KR have ("be sated") for אכל, assimilating the verb to 12:14. The former is found in V and S (as well as T, which has taken it from S). The agreement of these diverse sources justifies listing v = v as a Hebrew variant.

G: (a) ἀπὸ καρπῶν δικαιοσύνης φάγεται ἀγαθός, (b) ψυχαὶ δὲ παρανόμων ὀλοῦνται ἄωροι. "(a) The good [man] shall eat from the fruits of righteousness, (b) but the souls of transgressors will untimely perish." The syntax of the Hebrew is difficult. G seeks to make sense of it by making "good" the subject and substituting "righteousness" for "mouth" ( $\approx$  G 11:30). G thereby prevents the inference—allowed but not intended by M's word-ing—that even the unworthy man will enjoy the fruit of his speech. (G does not handle 12:14a this way.) In 13:2b, G avoids an apparent truism by treating DΩT as the violence done *to* transgressors and rendering it ᠔λοῦνται ἄωροι.

S follows G's cue but minimizes its divergence from M by skipping over  $\ddot{\alpha}\omega\rho\sigma$ .

For M's "will eat," S (مصحد), T (لتصحر), and V (*saturabitur*) have "will be satisfied," using 12:14a to emphasize the value of the reward.

### 13:4

G: (a) ἐν ἐπιθυμίαις ἐστὶν πᾶς ἀεργός, (b) χεῖρες δὲ ἀνδρείων ἐν ἐπιμελεία. "(a) Every sluggard is in longing, (b) but the hands of the virtuous [pl] are cared for." G creates a strict antithesis. ἐπιμελεία could mean "à l'enterprise" (BAP) or "are diligent" (NETS), but the antithetical "in longing" suggests that the second stich refers to the benefits of God's care, that is, being cared for. Whereas the lazy man just yearns for satisfaction, the virtuous (i.e., industrious) have their desires fulfilled. ἐπιμέλεια = [Ψ] (lit. "fattened") in 28:25; cf. 3:8, 22, where the Greek term is used, but for different Hebrew words. G χεῖρες ("hands") (for M's 10:4 (where too G uses the pl) and 12:24.

G,  $\sigma'$ , V, and S do not represent נפשו. Since there is no good explanation for the loss of this word, it was probably present in their source texts (contrary to BHQ) but considered as adequately implied by the notion of desiring. The difficulty of the syntax may have motivated this approach.

### 13:5

G: (a) = M; (b) מֹסדּβאָς δὲ αἰσχύνεται καὶ οὐχ ἕξει παρρησίαν. "(b) but the wicked will be disgraced and not have confidence [or 'fluency of speech']." οὐχ ἕξει παρρησίαν is a converse translation of ויחפיר to avoid repetition of the synonyms meaning shame. However, G does not take this course when the pair of synonyms recurs in 19:26. בו"ש is associated, correctly, with שו"ש (ABP).

# **13:6** M 4QProv<sup>b</sup> S ] > $G^{BSMSS}$ (unc)

This verse is lacking in G<sup>BSMSS</sup>; it is supplied in G<sup>A</sup> and has an asterisk in SyrH. Still, G<sup>A</sup> is not a literal translation of M and does not look Hexaplaric. The reason for the verse's omission in G<sup>BSMSS</sup> is unclear. G treats רשעה as an abstract-for-concrete metonymy: דסטָ אֹד מֹסבּאָבוֹך ("the wicked" [pl.]). In fact, the metonymic words are דָרֶרָך (lit. "innocence חַשָּאת: 7 וַשׁ מֻתְעַשֵּׁר וְאֵיז כָּל מִתְרוֹשֵׁשׁ וְהַוֹן רֶב: 8 לַפֶּר גֶפָשׁ־אֵישׁ עָשְׁרֵוֹ וְׁרָשׁ לֹא־שָׁמַע גְּעָרֶה: 9 אוֹר־צַדִּיקִים יִשְׂמָח וְגַר רְשָׁעֵים יִדְעֶדָ: 10 רַק־בְּזָדוֹן יִתֵּן מַצֶּה וְאֶת־נוֹעָצִים חָכְמֵה: 11 הוֹז <sup>ר</sup>מְבֹהָל<sup>ז</sup> יִמְעֵט

of way" = the innocent) and either רשעה or רשאת (wickedness or sin = the wicked or sinner; see ABP).

13:8

G: (a) λύτρον ἀνδρὸς ψυχῆς ὁ ἴδιος πλοῦτος, (b) πτωχὸς δὲ οὐχ ὑφίσταται ἀπειλήν. "(a) The ransom of a man's life is his own wealth, (b) but a poor man cannot endure a threat." G shades yau in a way that makes sense of M's puzzling "but a poor man does not hear a rebuke." S = M.

**13:9** fin ] + 2 stichoi G (9*a*) (elab)

G: (a) φῶς δικαίοις διὰ παντός, (b) φῶς δὲ ἀσεβῶν σβέννυται. "(a) Light (lasts) forever for the righteous, (b) but the light of the wicked is extinguished." For the difficult image of the light of the righteous "rejoicing" ("שמת), G substitutes a serviceable διὰ παντός ("forever"). (Actually, ש απ = "shine"; see ABP.) The translator generally avoids repeating a word in parallelism with itself, but sometimes this does happen, as in 13:9; 14:20; and 29:22.

### 13:9a

G: (a) ψυχαὶ δόλιαι πλανῶνται ἐν ἑμαρτίαις, (b) δίκαιοι δὲ οἰκτίρουσιν καὶ ἐλεῶσιν. "(a) Deceitful souls wander in sins, (b) but the righteous have compassion and show mercy." This is a new proverb whose two lines seem unrelated. It was located here because of its connection to 13:9. That the light of the wicked will be extinguished (13:9b) means (or also means) that they wander in the darkness of sin (9*a*a). 13:9*a*b ≈ 21:26b; cf. G-Ps 112[111]:5a. G-Prov 13:11c reuses this stich.

13:10

G: (a) κακὸς μεθ' ὕβρεως πράσσει κακά, (b) οἱ δὲ ἑαυτῶν ἐπιγνώμονες σοφοί. "(a) An evil man does evils with insolence, (b) but those who are their own judges are wise." G probably understands rgamma (lit. "empty") and moralizes it as "evil" to clarify the nature of the blemish. It then underscores the evil by adding κακά; see the comment on 1:18.

On פֿאַנאָעשע אין, see the comment on 12:26. פֿאַנאָעשע כסיד corresponds to נועצים ("those who take counsel"), understood as reflexive. G 13:10b recalls the Delphic dictum "Know thyself" (de Lagarde; Gerleman 1956, 29–30), though the wording differs.  $\alpha', \sigma', \theta'$  do not represent אין, which means that they understood it as אין, which did not seem to add anything to the meaning of the sentence. S כויש  $\alpha$  G. T לחוד M.

13:11 אבהל G (ἐπισπουδαζομένη) σ' ε' (ὕπερσπουδαζομενη) V (festinata) ] געבהל א G (געבן א S (געבן א א מהבל א: געבה); + בה G (μετὰ ἀνομίας) (M S: metath מַהָבָל; G: dbl rd)

G: (a) ὕπαρξις ἐπισπουδαζομένη μετὰ ἀνομίας ἐλάσσων γίνεται, (b) ὁ δὲ συνάγων ἑαυτῷ μετ' εὐσεβείας πληθυνθήσεται. (c) δίκαιος οἰκτίρει καὶ κιχρῷ. "(a) Property gathered hastily with illegality dwindles, (b) but he who gathers for himself with piety will be increased. (c) The righteous man has mercy and lends."

As often, G enhances the religious and moral message. By rendering  $\pi as \dot{\alpha}\nu o\mu i\alpha$  (only here and Sir 49:2), the translator would have us understand the fault in question to be illegality, not hard work as such; see Giese 1992a, 418; 1993a, 112. Also, by paraphrasing the difficult  $\psi$ ' as  $\mu \epsilon \tau$ ' εὐσεβείας, G further implies that the secure increase of wealth requires piety. Perhaps "hand" inspired the notion of generosity to the poor (thus T), hence piety.

G 13:11c, like G 13:9*a*b, of which it is a variant, is based on Pss 37[36]:21b and 112[111]:5a. The addition explains the meaning of gathering wealth "with piety" and further tempers the praise of "gathering" in 13:11b, lest it be misread as affirming greed.

S reads מהבל as מהבל ("which is from iniquity" ≈ μετά מיסעומק) and follows G, but loosely, in interpreting על יד as על יד ("gathered in righteousness"). S is thus using G's construal of the verse

וְּקֹבֵץ עַל־יָד יַרְבֶּה: 12 תּוֹחֶלֶת מֲמֻשְׁכָה מַחֲלָה־לֵב וְעֵץ חֵיִּים תַּאֲוָה בְאֶה: 13 בְּז לֲדָבָר יֵחֶבֶל לְוֹ וִירֵא מִצְוָה הָוּא יְשֵׁלֶם: 14 תּוֹרַת חֲכָם

while moving it closer to M's wording. T's ויהיב למסכינא ("and gives to the poor") follows S but with an eye to M's על יד.

The idea of the verse as emended is found also in 20:21: "An inheritance gained in a rush [נחלה מבהלת] at first—its outcome will not be blessed."

13:12 אוחלת M G (12b) (ἐλπίδα) ] אתחלת G (12a) (ἐναρχόμενος) (orth) אטובה [ M ] אטובה G σ' θ' ε' (ἀγαθή) (moral)

G: (a) κρείσσων ἐναρχόμενος βοηθῶν καρδία (b) τοῦ ἐπαγγελλομένου καὶ εἰς ἐλπίδα ἄγοντος. (c) δένδρον γὰρ ζωῆς ἐπιθυμία ἀγαθή. "(a) Better is he who begins to help with heart (b) than he who promises and leads (others) toward hope, (c) for a good desire is a tree of life."

G is distant from M, and it is hard to reconstruct G's source text. Stichoi b and c render components of M 13:12b expansively without regard to M's syntax. What sent G off track was the erroneous אָתחלת (= אָקלַת), which was treated loosely as a participle, ἐναρχόμενος. The awkwardness in G 13:12a, with its superfluous "begins" and "with heart," indicates that G is struggling with a difficult Hebrew text rather than composing freely. Apparently M's אווהליג is represented in 13:12b by εἰς ἐλπίδα, a doublet. βοηθῶν for ממשכה is hard to explain, but note that in Ps 109[108]:12 משך חסד (lit. "draw mercy") is translated ἀντιλήμπτωρ ("helper"). In Prov 13:12c, G has "good desire" for M's mathematicates that has come to pass").

G or its source text hesitated to give a blanket endorsement to the fulfillment of desire. The change to "good" is moralizing and introduces an interpretation not evident in M. It probably occurred in Hebrew transmission, with שובה for גבאה SyrH,  $\sigma'$ ,  $\theta'$ , and  $\varepsilon'$  agree with G in representing, and they would be unlikely to make the change without a textual basis. G means that actually helping someone wholeheartedly is better than raising false hopes. As 13:12c explains, only wholehearted help qualifies as ἐπιθυμία ἀγαθή.

 suggests that it has a variant text, which, however, cannot be fully reconstructed.

### **13:13** fin ] + 3 stichoi G (13*a*) (elab)

### 13:13a

G: (a) υίῷ δολίψ οὐδἐν ἔσται ἀγαθόν, (ba) οἰκέτῃ δὲ σοφῷ (bβ) εὔοδοι έσονται πράξεις, (c) και κατευθυνθήσεται ή όδὸς αὐτοῦ. "(a) For a deceitful son, there shall be nothing good,  $(b\alpha)$  but for a wise servant,  $(b\beta)$  affairs will be successful, (c) and his way will prosper." G provides an additional proverb. The first two stichoi, 13:13aab, are inspired by 17:2-in the Hebrew, not the Greek. In its present place "the deceitful son" adumbrates "he who scorns a word" in 13:13a, and "a wise servant" echoes "he who fears a command" in 13:13b. Stich 13:13ac is redundant. In fact, 13:13*a*b $\beta$  and 13*a*c have the look of doublets.

V lacks 13:13*a* but is provided with it in some MSS.

S: with King a and my Komp Ki Kiklan Kight a) To the deceitful man, there معمد لائع will be nothing good, (b) but the wise man, whose deeds are honest-(c) his way too will rise to the top." S has the equivalent of G 13:13a but differs in the persons it describes-the deceitful man and the wise man rather than the deceitful son and the wise servant. S is generalizing the addition's teaching to give it wider validity. Instead of G's doublet  $(13:13ab\beta+c)$ , S has a single statement (c), which may be a condensation of the G doublet but alternatively can reflect S's Hebrew source text.

θανεΐται) (div)

G: (a) νόμος σοφοῦ πηγή ζωῆς, (b) ὁ δὲ ἄνους ὑπὸ παγίδος θανεῖται. "(a) The law of the wise is a spring of life, (b) but the fool will die by a trap." M is easy to understand, and there is no ideological motivation for the divergence in the second stich. G 13:14b is based on a variant text, namely, וכסיל ממקש ימות (Baumgartner). The change from לסור to ממקש ימות was provoked by a different word division, וכסיל This left the resulting verb, "he will die," without a subject. The change of to כסיל to כסיל provided the subject, but it is unclear whether this was a scribe's or the translator's doing. Since G translated לסור as הטובי מצ מסובי של as הסובי έκκλίνειν in the identical 14:27b, he could have done the same here. Though G 13:14b creates an antithesis to 13:14a, it is a very loose one

מְקוֹר חַיֵּיִם לְׁסוּר מִמִּקְשֵׁי מֲוֶת: 15 שֵׂכָל־ְטוֹב יִשֶּׂן־חֵזְ וְדֶרֶדְ בּּגְדֵים <sup>ר</sup>אֵידֵם<sup>י</sup>: 16 כָּל־אֲרוּם יַעֲשֶׂה בְדֶעַת וּּכְסִיל יִפְרָשׁ אַוֶּלֶת: 17 מַלְאָד רְשָׁע יִפְּל בְּרֶע וְצִיר אֶמוּנִים מַרְפֵּא: 18 רֵישׁ וְקָלוֹז פּוֹרֵעַ מוּסֵר וְשׁוֹמֵר תּוֹכַחַת יְכַבֶּד: 19 תַּאֲוָה גָהְיָה תֶעֶרַב לְגָפָשׁ וְתוֹעֲבָת בְּסִילִים

and certainly not the motive for the differences from M (contrary to Gerleman 1956, 20).

S: (a) = M; (b) אדאראה גרבאלא געלא געלא (b) to those who turn aside from the traps of death." S treats לסור, ungrammatically, as a participle, as in 14:27.

Both 9:10*a* and 13:15b are insertions teaching that knowledge of the Law is the best part of wisdom. Dick (1990, 26, 41) regards  $\nu \dot{\rho} \mu \sigma \nu$  in 13:15b as sapiential teaching, while Cook (2002, 280–88) identifies it as the law of Moses. Dick says that references to  $\nu \dot{\rho} \mu \sigma \varsigma$  tend to allude to the law of Moses, but in G-Proverbs this remains rather vague, so that "law" could be understood as communal law rather than specifically the Pentateuch (1990, 40, 42). Still, the mention of "law" with no indication of source (such as "of the wise") often does seem to allude to the law of Moses; see the comment on 6:23. This is clearer when  $\nu \dot{\rho} \mu \sigma \varsigma$  has the definite article. Occurrences without the article are 6:23; 9:10*a*; 13:15; 28:7, 9; with the article: 28:4 (2x); 29:18.

It is noteworthy that the law of Moses is mentioned in G-Proverbs only rarely and allusively, though its centrality was undoubtedly a given for the translator. Identifying wisdom with the revealed Torah is not a major goal of the G translator, who was largely content to adhere to the older concepts of wisdom.

V guesses that the difficult איתן means vorago ("pit" or abyss").

S's איתן for איתן is probably taken from G.

### 13:17

G: (a) βασιλεὺς θρασὺς ἐμπεσεῖται εἰς κακά, (b) ἀγγελος δὲ σοφὸς [=  $G^{BScA}$  (= OG); var: πιστὸς  $G^{S^*103253 \text{ Rahlfs}}$ ] ῥύσεται αὐτόν. "(a) A rash king will fall into evils, (b) but a wise [var: 'faithful'] messenger will save him." The poorly attested πιστός (= with M) looks like a correction of an original σοφός.

For unclear reasons, G has "king" for מלארים. Perhaps it read מלאכים (cf. מלבים for מלאכים in 2 Sam 11:1), though there is no mechanical explanation for the loss of the  $\aleph$ , and G-Proverbs does not show clear signs of oral variants. In any case, this error gave rise to interpretive differences: the replacement of evil with rashness and the unique rendering of מרפא ("healing") as "saving." The translator shifted the proverb to a different situation, one in which a reckless king is saved by one of his emissaries, who is wiser and more prudent than he.

## 13:18

G: (a) πενίαν καὶ ἀτιμίαν ἀφαιρεῖται παιδεία, (b) = M. "(a) Instruction removes poverty and disgrace."

S: (a) א מבשהא היב שוא אין איז א מבעוע (b) = M. "(a) Poverty and disgrace nullify instruction."

G and S identify the subject of the sentence differently. On the syntax of this difficult sentence, see ABP.

13:19 א מרע [ $\dot{\alpha}\pi\dot{\sigma}$  א מדע ( $\dot{\alpha}\pi\dot{\sigma}$  א א הבאלא) (graph  $\tau \to \tau$ ) (graph  $\tau \to \tau$ )

G: (a) ἐπιθυμίαι εὐσεβῶν ἡδύνουσιν ψυχήν, (b) ἔργα δὲ ἀσεβῶν μακρὰν ἀπὸ γνώσεως. "(a) The desires of the pious please the soul, (b) but the works of the impious are far from knowledge." G 13:19a makes it clear that only pious desires are sweet.

Though the versions modify the verse in ways that can be called moralizing or "interpretive," we can still recover a variant. מתט איש שעש שיש שיש (understood as מדע) (de Lagarde). This variant forced the substitution of the morally neutral έργα ("works") for the very negative תועבה ("abomination") to avoid banality. סור is reflected indirectly in the emphatic "far."

S's אשר איב איב איב י ("proper desire") shows the same moralizing concern as G but expresses it differently. אבאא הי ("from knowledge") confirms מדע. סַוּר מֵרֶע: 20 <sup>ר</sup>הָלִוֹדְּ אֶת־חֲכָמֵים <sup>ר</sup>וַחֲכֶם וְרֹשֶׁה כְסִילִים יֵרְוֹעַ: 21 חֲטָּאִים תְּרַדֵּף רָשֶׁה וְאֶת־צַׁדִּיקִים יְשַׁעָּּם־טְוֹב: 22 טוֹב יַנְחֶיל בְּגֵי־ בְּגִים וְצָפְוּן לַצַּדִּיק חֵיל חוֹמֵא: 23 <sup>ר</sup>רַב אֹבַל נִיר רָאשָׁים וְיֵשׁ נְסְפָּה בְּגִים וְצָפְוּן לַצַּדִּיק חַיל חוֹמֵא: 23 <sup>ר</sup>רַב אֹבַל נִיר רָאשָׁים וְיֵשׁ נְסְפָּה בְּגִים וְצָפְוּן לַצַּדִיק חַיַל חוֹמֵא: 23 <sup>ר</sup>ַב אַבָל וְאַהֲבֹו שָׁחַרִוֹ מוּסֵר: בְּלָא מִשְׁפֵּט: 24 חוֹמֵיִד שָׁבְטוֹ שוֹנֵא בְגָו וְאֹהֲבֹו שְׁחַרִוֹ מוּסֵר: בּזִיק אָבֵל לְשַּׁבַע נַפְּשָׁו וּבֶטֶן רְשָׁעֵים תָּחְסֵר: בּוּזַרַד וּחָכְמַוֹת וְנָשִׁים) בְּנְתָה בֵיתֶה וְאוֹיָלֶת בְּיָדֵיה תָהָרְסֵנּוּ: 2 הוֹלֵד

13:20 הלוך M<sup>K</sup> ] הולך M<sup>Q</sup> G (א הלוד M<sup>Q</sup> G (המאטת איז א M<sup>Q</sup> G (המסגאי) V (*qui* ... graditur) T (דמהליך) (gram, orth, equal)

אוחכם  $M^K$  [ הרא של  $M^Q$  G (soft éstai) מל (softseftsetai) S (האסא שבית) V (sapiens erit) T (נתחכם) ] (gram, equal)

ירוע M S (הא האד שG (עראי<br/>שG (אישטסאאספדמו) (graph  $\to \to$  , metath <br/>ד  $\to$  דו)

Each of the two *ketiv-qere* pairs is indivisible; either "He who goes with the wise will become wise" (*qere*) or "Go (הָלוֹד) with the wise and become wise (וַהְלוֹד)" (*ketiv*). Hence they are not miscellaneous copying errors but two variants of the proverb. The versions all follow the *qerayin*.

V's *efficietur similis* ("will become like them") derives  $\tau$  לרע"ה (or  $\tau$ " ( $\tau v$ "  $\tau$ ) "associate with" and understands this as qualitative similarity.

13:23 (ירַב אֹכֵל אַ C (vocal, accents) אַ רָב־אָכֶל M (vocal, accents)

M is difficult both linguistically and theologically; see the discussion and references in ABP. The two proposed conjectural changes in vocalization (and the attendant accents) make the verse clearer. It can then be translated: "The great man devours the tillage of the poor, and some people are swept away without justice."

G: (a) δίκαιοι ποιήσουσιν ἐν πλούτῷ ἔτη πολλά, (b) ἄδικοι δὲ ἀπολοῦνται συντόμως. "(a) The righteous [pl] will spend many years in wealth, (b) but the unjust [pl] will perish suddenly." G has no contact with M in 13:23a, and in 23b it diverges from M's meaning. The differences aim to override M's apparent denial of the principle of recompense. G treats αμeq as a noun phrase: "one who is in injustice," hence unjust people, ἄδικοι. G expresses confidence in recompense, while M recognizes the

existence of inequities. Since G's ideological motivation is clear, it seems far-fetched to surmise that in 13:23b G read רשעים for ראשים, translating it מאואט, for which he created an antithesis in 13:23a (אוסגטנו) based on a treatment of ראשים as a permutation of ישרים (BHQ).

V: (a) multi cibi in novalibus patrum, (b) et alii congregantur absque iudicio. "(a) Much food is in the cultivated fields of fathers, (b) but others are gathered without justice." V translates ראשים ("poor") with "fathers," probably by associating איז איז ("poor") with "fathers," fathers." This interpretation was prompted by the notion of inheritance in 14:22, and "father" is implicit in 13:22 and 24. V understands נספה as הספה ("gathered"), as in Gen. Rab. 49:15 (§57b) (C. Gordon 1930, 390). V, unlike G, does not deny the possibility of injustice. It grants that some people gain their wealth as a legitimate inheritance while others get rich wrongly.

# 14:1 <חָכְמָוֹת [ <חָכְמָוֹת M (vocal)

נשים MG (אונאיב, S (האגאא)  $T^{L}$  (דנשיא) V mulier (add)

The first clause was originally רמות בנתה היתה, as in 9:1a. Though the versions represent נשים, it is very likely an early gloss to the unusual singular חכמות M's הכמות נשים is meaningful: "the wisest of women"; compare the superlative construction in Judg 5:29. However, the verb "builds" is singular, whereas "women" is plural. נשים overloads the line and is probably a gloss added by someone who assumed that הכמות was plural. It is to be understood as in apposition, as if to say, "The wise [sc. women]." But in spite of its plural appearance, הכמות (as in 1:20 and 9:1) is a singular (see ABP on 1:20). Its literal meaning is "wisdom," but it here serves as an abstract-for-concrete metonymy meaning "wise woman." When הַכְמוֹת (a noun).

The versions all represent שכמות נשים but construe the phrase variously. G: דיסקמו איטע ("wise women"); V: sapiens mulier ("wise woman"). S: איאאה אייארא אייארא ("wise woman"). S: איאאה אייארא אייארא ("wise woman"), S: איאאה אייארא אייארא אייארא ("wise woman"), S: איאאה אייארא אייארא אייארא אייארא ("wise woman").

ή δὲ ἄφρων ("the [fem sg] folly [masc sg]") = M's אולת (lit. "folly"), parsed as an abstract-for-concrete trope or as the feminine equivalent of אויל.

ַּבְּיָשְׁרוֹ יְרֵא יְהוֶה וּנְלְוֹז דְּרָכֵיו בּוֹזֵהוּ: 3 בְּפִי־ְאָוִיל חַׂטֶר גַּאָוֶה וְשִׂפְתִי חְׁכָמִים <sup>ר</sup>ּתִּשְׁמְרְוּם<sup>י</sup>: 4 בְּאֵין אֲלָפִים אָבָוּס בֶּר וְרָב־תְׁבוּאוֹת בְּכַח שִׁוֹר: 5 עֵד אֶמוּנִים לָא יְכַזֵּב וְיָפִיחַ בְּזָבִים עֵד שֵׁקֶר: 6 בִּקֶשׁ־לֵץ חָכְמָה וָאֵין וְדַעַת לְנָבִוֹן נָקֵל: 7 לֵך מֻנֶּגֶד לְאִיש בְּסֵיל וּבַל־יְׁדַעְתָ שִּׁפְתֵי־דֶעַת: 8 חָכְמַת עֶרוּם הָבֵין דַּרְכֵּו וְאוֶגֶלת בְּסִילִים מִרְמֶה: 9 אֵוּלִים יְלֵיץ אָשֶׁם <sup>ו</sup>וּבֵית<sup>י</sup> יְשָׁרֵים בָצְוֹן: 10 לֵב יוֹדַע מְרַמָה

### 14:2

G translates d'("despises him") by diathesis as ἀτιμασθήσεται ("will be despised"). This states that the man of twisted ways will be held in contempt and not just that he despises God, which is a truism.

# 14:3 הַשָּׁמוּרֵם [ הַשָּׁמוּרֵם M (metath ור $\rightarrow$ רו)

M is a metathesis for תשמרום, a rare but regular third feminine plural imperfect + third masculine plural suffix, as required by the syntax.

G: (a) ἐκ στόματος ἀφρόνων βακτηρία ὕβρεως, (b) χείλη δὲ σοφῶν φυλάσσει αὐτούς. "(a) From the mouth of fools is a rod of arrogance, (b) but the lips of (the) wise guard them." It is impossible to determine if φυλάσσει αὐτούς represents תשמרום or M's anomalous תשמורם, since G would have had to translate the latter as "will keep them" in either case.

## 14:5

4QProv<sup>b</sup> has Prov 14:5–10 (fragmentary) = M.

### 14:6

G: (a) ζητήσεις σοφίαν παρὰ κακοῖς καὶ οὐχ εὑρήσεις, (b) αἴσθησις δὲ παρὰ φρονίμοις εὐχερής. "(a) (If) you [sg] seek wisdom with [i.e., 'near'] the wicked, you will not find (it), (b) but knowledge is readily (available) with sensible people." Since it might be unclear why a scoffer would seek wisdom at all, G ignores M's syntax for the sake of what seems like a more logical proposition.

14:7 (כל M ] (של המעדמ) S (של השמכ) (metath כל  $\rightarrow \Delta$ ) (metath כל M ] (של העת [M] ובל דעת [M] ובל דעת (M] ובל דעת (M] הכ  $\rightarrow \rightarrow$ , div) G: (a) המעדמ לימעדוֹמ מעסףו מקסטו, (b) אוכלי במסל מוסטיק צנוֹאן ססקמ. "(a) Everything is against a foolish man, (b) but wise lips are weapons of

knowledge." G reads כָּל וּכְלֵי דַעַת שָׁפְחֵי דַעָת (Jäger). G's variants are mechanical errors, as described in the apparatus, but they create a meaningful (though not very graceful) proverb. De Waard (1993, 250) believes that these changes were deliberate attempts to deal with interpretive problems and to create antithesis and alliteration. But M is not difficult, and there is no reason to ascribe the differences to a complex interpretive process which somehow mimicked common scribal errors in order to produce a rather fuzzy antithesis with no theological gain.

### 14:8

G: (a) σοφία πανούργων ἐπιγνώσεται τὰς ὁδοὺς αὐτῶν, (b) ἄνοια δὲ ἀφρόνων ἐν πλάνη. "(a) The wisdom of the clever [pl] will understand their ways, (b) but the ignorance of fools goes astray [lit. 'is in going astray']." According to Baumgartner, ἐπιγνώσεται = 「ユニヾ (3rd fem sg) for M's infinitive תָבִין, but the translator would have had to treat the latter as a finite verb anyway. On the face of it, M 14:8b—"the folly of dolts is deceit"—is redundant and banal. (Who else's folly, after all? And how could folly be other than deceit?) G sharpens the proverb. ἐν πλάνη reinforces the "ways" motif of 14:8a (BAP).

S is close to G but diverges in rendering the apparently redundant אולת כסילים ("and the folly of fools") as השבלה ("and the way of fools"), thereby extending the theme of "ways" from the first stich to the second.

14:9 הכיע (M: norm; S: metath [of M] הכיע (ג'א אובית β (ג'א אובין M; אובית) (M: norm; S: metath [of M] וני → ינ

G: (a) οἰχίαι παρανόμων [var: ἀφρόνων G<sup>A</sup>] ὀφειλήσουσιν καθαρισμόν, (b) οἰχίαι δὲ διχαίων δεχταί. "(a) The houses of transgressors [var: 'fools'] will require purification, (b) but the houses of the righteous are acceptable." M is obscure. Still, G does offer clues to variant words that make the verse more comprehensible, as well as providing its own interpretation. In the following, it is assumed that G treats grammatical number flexibly, as is often the case. (1) οἰχίαι = אהל (אולים M אהל, ואילים (חוש אהל, אהלי (מולים עולים (מולים מולים) (סווע הואים) (סווע הואיים) (סווע הואים) (סווע הואיים) (סווע הואיים) (סווע הואיים) (סווע הואיים) (סווע הואיים) (סווע הואיים) (סווע ה

וּבְשִׂמְחָתוֹ לֹא־יִתְעָרַב זֶר: 11 בֵּית רֲשָׁעִים יִשְׁמֵד וְאָהֶל יְשָׁרִים יַפְרִיחַ: 12 יֵשׁ דֶּרֶד יֲשֶׁר לִפְנֵי־אֵישׁ וְאַחֲרִיתָה דַּרְכֵי־מֶוֶת: 13 גַּם־

standard רצון (5) ובין כמח mean "what is desired," "object of favor" (as in Prov 11:20; 12:22; 15:8; etc.). The text that reached G was, approximately, prov 11:20; 12:22; 15:8; etc.). The text that reached G was, approximately, ("The tents of mockers are guilt [= 'guilty'], but the house of the righteous is favor [i.e., 'favored']"). (BAP notes the cultic resonances of  $\kappa \alpha \theta \alpha \rho \iota \sigma \mu \delta \kappa \tau \alpha \ell$ .) Since there is no good scribal explanation for אהלי אולים or ליין אים אוכי, and since most of the correspondences are unparalleled elsewhere, the proposed variants are too conjectural to introduce into the HBCE text.

S: בוגנשא המבלא גיבא גמביא מכלא באי גרגנשא המבלא. "(a) The houses of evildoers truly need cleansing, (b) but the houses of the righteous are acceptable. (c) Fools commit sins, (d) but the sons of the upright (do) what is favored." Stichoi 14:9ab = G. S 14:9cd attempts to adjust to M, but 14:9d reflects יבני (ובני (ובני בו

14:10 אד M S (הבוא G ( $\subset{U}\beta \rho \epsilon \iota$ ) (graph  $\rightarrow \rightarrow$  )

G: (a) καρδία ἀνδρὸς αἰσθητική, λυπηρὰ ψυχὴ αὐτοῦ. (b) ὅταν δὲ εὐφραίνηται, οὐκ ἐπιμείγνυται ὕβρει. "(a) (As for) the knowledgeable heart of a man—his soul is sad, (b) but when it [he?] rejoices, it [he?] is not mingled with arrogance." The idea of 14:10a is similar to Qoh 7:4: "The heart of wise men is in the house of mourning, while the heart of fools is in the house of merrymaking." ὕβρει = ¬<u>1</u> "arrogant," which is treated as an abstract: "arrogance." The same error occurred in 27:13.

S: לבא הסאנא כבורך בוגסאלא מס אמער ארא הבער שישים. אינאען איז אינען איז אינען איז אינען איז אינען אינען אינען אינען. ((a) The knowledgeable heart is anxious for itself, (b) and in its [his?] joy, no stranger shall take part." S is based on M but construes אידע as an adjective, געבאנא ("knowledgeable") = G מוֹסטאַדעאָל. S says that the wise man keeps his emotions to himself.

### 14:11

Where M has the unusual metaphor "(the tents of the upright) will flourish (יפריה)," G has the more obvious "will stand" (סדאָסטאדמו).

#### 14:12

4QProv<sup>b</sup> 14:12–13 (fragmentary) = M.

G: (a) ἔστιν όδὸς ἡ δοκεῖ ὀρθὴ εἶναι παρὰ ἀνθρώποις, (b) τὰ δὲ τελευταῖα αὐτῆς ἔρχεται εἰς πυθμένα ἅδου. "(a) There is a way that seems to be straight to men, (b) but its ends go to the depth of Hades." G interprets M correctly. A way that "is straight before a man" is behavior that *seems* straight and smooth (cf. 21:2a). M's "its end is ways of death," though somewhat awkward in grammatical number, means that this deceptive way ends up as paths *to* death (ABP 1.129, 2.577). Since Translated the same way in 16:25, G is not evidence for Translated the Section (contrary to BHS).

S translates אחריתה ואחריתה האחריתה. This may seem to represent a graphic error in Hebrew: ארחותיה  $\rightarrow$  אחריתה. However, אחריתה is translated by ארחותיה also in Prov 19:20 and by הסונגלים in Prov 16:25. (The variant הסונגלים in the latter is a later adjustment in S 14:12. In 16:25, (The variant הסונגלים is a deliberate play on הביבורים.) In the present verse and in 16:25, "its ways" (סביבורים and הביבורים) were prompted by the adjacent הררים. In 19:20, the translator may have resisted saying "so that you be wise in your end," because, logically speaking, the youth should become wise *now*. It seems that the translator could treat אחרים s "ways" because of the similarity of אחרית this conflation includes an "exegetical metathesis" but other changes as well.

14:13 אַחַרית (האשמחה [ אחרית ה שמחה ] M אַחַרית האַ שמחה הG (דבאבטדמוֹם אל א $\approx$  G (דבאבטדמוֹם) (div)

G: (a) ἐν εὐφροσύναις οὐ προσμείγνυται λύπη, (b) τελευταία δὲ χαρὰ [var: χαρᾶς G<sup>A</sup>] εἰς πένθος ἔρχεται. "(a) Pain does not mingle in pleasures, (b) but in the end, happiness comes to sorrow [var: '(but the end of joy comes to) sorrow']." G imports some of the phraseology of G 14:10 in slightly different form (in 14:10: λυπηρά, εὐφραίνηται, ἐπιμείγνυται). G reverses the meaning of M 14:13a. If the motive were to introduce a positive attitude, it is puzzling that the same was not done to 14:13b, for the verse remains pessimistic. Rather, G means that while one is happy (or: enjoying pleasures) he does not feel grief, but ultimately his joy turns into sorrow. The majority G reading, τελευταία δὲ χαρά, represents the same problematic word division as M—ικηΓης m (On the face of it, this division is supported by  $G^{A V 103 MSS H-P}$ . However, ἔρχεται requires a subject, so OG must have had the nominative χαρά, not the genitive χαρᾶς.)

S's העומא העומא ("the end of *its* happiness") is not evidence for העומחה, since the possessive suffix of שנהמה must represent the שנהמה of האחרית השמחה, even though it is attached to the *nomen rectum*, הענהמה ה.

14:13-21

בִּשְּׁחִוֹק יִכְאַב־לֵב <sup>וּ</sup>וְאַחֲרִית הַשִּׂמְחָה<sup>ז</sup> תוּגֶה 14 מִדְּרָבֵיו יֵשְּׁבַּע סַוּג לֵב <sup>וּ</sup>וֹּמִמַעֲלָלָיו<sup>ז</sup> אַישׁ טִוֹב: 15 בָּתִי יַאָמַין לְכָל־דָּבֵר וְעָרוּם יָבִין לַאֲשִׁרוֹ: 16 חָכָם יֵרָא וְמַר מֵרֶע וּכְסִיל מִתְעַבֵּר וּבוֹטֵחַ: 17 קַצַר־ אַפַּיִם יַעֲשָׁה אָוֶלֶת וְאָישׁ מְזִמּוֹת יִשְׂגֵא: 18 נְחַלָּוּ פְתָאיֵם אָוֶלֶת וַעֲרוּמִים יַכְתָּרוּ דֵעַת: 19 שַׁחַוּ רָעִים לִפְגַי טוֹבֵים וֹרְשָׁעִים עַל־שַׁעַרֵי צַקִּיק: 20 גַּם־לֶרֵעָהוּ יִשְׁנֵא רֶשׁ וְאֹהֲבֵי עָשִׁיר רַבִּים: 11 בְּז־לְרַעֵהוּ

14:14 וממעלליו המעלליו ה<br/>מעלליו המעלליו אוממעלליו היש אוממעלליו אוממעלליו או (haplog לל הי haplog או המע $\to$ ה המרוס א

As emended, the proverb means that a person's own deeds come back to reward or punish him. Compare מעלליו // דרכיו in Jer 17:10.

S: כד אסושלה ערב של גרביע ברבה. אשרב של ארא אשרב של האיני שרב של האיני שרב של האיני שרב של האיני שרב של היושר "He who is rash in his heart will be sated from his ways, and a good man will be sated from the fear of his soul." S 14:14b is a guess at M's מעלליו (?).

### 14:15

G: (a) = M; (b) πανοῦργος δὲ ἔρχεται εἰς μετάνοιαν. "(b) but a clever man comes to a change of heart." G 14:15b seems to be a broad rephrasing of M's "while the shrewd man watches his step." G implies a chain of reasoning: while the simple man believes everything, the shrewd one observes his own behavior and (when appropriate) has a change of heart.

S  $rac \sim rac$  ("distinguishes between good and evil") offers a different interpretation of what "watching his step" means.

T מתביין לטבתה ("examines his well-being") associates לאשרו with אשרו עיד אישרו אַשָּרי אישרי.

### 14:16

G: (a) σοφός φοβηθεὶς ἐξέκλινεν ἀπὸ κακοῦ, (b) ὁ δὲ ἄφρων ἑαυτῷ πεποιθὼς μείγνυται ἀνόμω. "(a) The wise man, fearing, turned aside from

a bad man, (b) but the fool, trusting in himself, associates with the lawless man." G's μείγνυται equates מתעבר with מתערב, meaning "mix in," "interfere"; thus S האנוער בהי ("intermingles with it"). G understood מתעבר to mean "interfere" in 26:17, translating it as δ προεστώς ("he who puts himself forward," i.e., interferes). Hence we need not suppose that G is reading המתערב here (contrary to BHS). The two similar-looking verbs could easily become confused because their meanings, "interfere" and "anger," are related as cause and effect. In 14:16b, G rearranges the syntax and expands the sentence to provide a logical sequence: trusting the bad man leads to associating with him.

14:17 יענא M ] אינשא G (טהסאר אינשא היא א ינשא (metath  $M \to M$  (נש $\to \Psi \to M$ 

G: (a) ὀξύθυμος πράσσει μετὰ ἀβουλίας, (b) ἀνὴρ δὲ φρόνιμος πολλὰ ὑποφέρει. "(a) He who is quickly angered acts with carelessness, (b) but the prudent man endures much." πολλὰ ὑποφέρει = **κυν**, understood (ungrammatically) as transitive. G often changes the voice of a verb (diathesis). Consequently, **παται** ("a man of schemes" = "a shrewd man") was interpreted positively. This creates a loose antithetical parallelism. The point of the proverb in G is that the intelligent man can or will put up with much irritation.

S: מעלא באל גבוי א האראי יש אורא באל איי מער (a) The hasty man—(in) all that he does he does not take counsel, (b) but a wise man is thoughtful." S paraphrases G. ("moderate," "thoughtful") is based on G, not M.

### 14:19

G 14:19b clarifies that "the wicked will bow before the gates of the righteous" (M) means θεραπεύσουσιν θύρας δικαίων ("will serve at the doors of the righteous"). On the Greek idiom, see LSJ 793a.II.2.

### 14:20

G: (a) φίλοι μισήσουσιν φίλους πτωχούς, (b) φίλοι δὲ πλουσίων πολλοί. "(a) Friends will hate poor friends, (b) but friends of the wealthy are many." G identifies the  $\mathfrak{W}$  ("poor") more closely as "poor friends," though that seems obvious from context. G also puts the clause in the active voice.

14:21 אַניים M<sup>K</sup> S (כל משביא) V (*pauperi*) ] עניים M<sup>Q</sup> (syn, equal) On this *ketiv-gere* pair, see the comment on 3:34.

G's πένητας ("the wretched [pl]") interprets M's רעהו ("neighbor") narrowly, in light of 14:20a and the parallel עניים.

חוֹטֵא וּמְחוֹגֵן <sup>ר</sup>עַנִיִיִם אַשְׁרֵיו: 22 הַלוֹא־יִתְעוּ תִּרְשֵׁי רֶע וְחֵסֶד וֹטֵא וּמְחוֹגֵן יעַנִיִים אַשְׁרֵיו: 23 הַכָּל־עָצָב יִהְיֶה מוֹתֶר וּדְבַר־שְׁפָתַיִם אַדְ־ וְמַחְסוֹר: 24 עֲטֵֶרֶת חַכָמִים עָשְׁרֶם אוֶלֶת בְּסִילִים אוֶלֶת: 25 מַצְיל

14:22 (אידעו [ מעל M ] אידעו G 22c (לדוסדמידמו) אידעו (graph ידעו (graph הרשי M G 22c (גברגן) של הרשי G 22a (גברגן) א הרשי (graph ' - ו)

ישרשי G 22b (דנאדמועסט<br/>סט) א הרשו G 22b (דנאדמועסטסט) א הרשו (graph ' $\rightarrow$  <br/>ו)

G: (a) πλανώμενοι τεκταίνουσι κακά, (b) ἔλεον δὲ καὶ ἀλήθειαν τεκταίνουσιν ἀγαθοί. (c) οὐκ ἐπίστανται ἔλεον καὶ πίστιν τέκτονες κακῶν, (d) ἐλεημοσύναι δὲ καὶ πίστεις παρὰ τέκτοσιν ἀγαθοῖς. "(a) Those who go astray devise evils, (b) while the good devise mercy and truth. (c) The devisers of evils do not understand mercy and loyalty, (d) but compassions and loyalties are with those who devise good [lit. 'good devisers']." G 14:22a, c ≈ M 14:22a; G 14:22b, d ≈ M 14:22b. G 14:22cd, has the obelus in SyrH and is presumably OG. G 14:22ab, however, is not simply a revision toward M. G 14:22ab and 22cd are renderings of slightly different Hebrew texts, both of them awkward and erroneous. BAP observes that the expression ἐλεημοσύναι δὲ καὶ πίστεις in 14:22b is systematically used for μοστικαι αλήθεια and αλ'.

G twice read הְרְשׁוּ for M's הרשי. It also had ידעו for ידעו, a graphic error probably occasioned by the loss of the left vertical of the ת. Other departures from M were occasioned by these graphic errors.

S saw חרשו רע in 14:22a and thought it was a verbal clause, "do evil." This mistake was possible only when reading the Hebrew. The subject "evildoers" was supplied from context.

T<sup>L</sup>: עוולי תעיין דחשלין בישתא וחסדא וקשוט חשלין טאבי. "(a) Evil-

doers go astray because they plan evil, (b) but the good plan kindness and truth." T has finite verbs rather than the (nominal) participles הרשי, הרשי. T bases itself on S—note the omission of הלא. (Its presence in some T MSS is certainly a later adjustment to M.) T took the verbal forms from S but then adjusted toward M by substituting a more specific verb, להשל, for S's vague ----.

G: (a) ἐν παντὶ μεριμνῶντι ἕνεστιν περισσόν, (b) ὁ δὲ ἡδὺς καὶ ἀνάλγητος ἐν ἐνδεία ἔσται. "(a) For [lit. 'in'] everyone who takes care there is a profit, (b) but the comfortable and indifferent will be in need." עצב ("toil," "misery") is rather construed as a metonymy for people who toil assiduously. G 14:23b is a reasonable paraphrase of M, understanding "word of the lips" as a metonymy for those who merely talk and take no action. See the contemptuous use of this phrase in 2 Kgs 18:20 and Ps 59:13.

V paraphrases דבר שפתים as *verba sunt plurima* ("when there are many words").

In 14:23ab, S's Greek text had בָא בָסדעי ("there is one") =  $G^{S \, 122 \, 149 \, 161}$ for באסדעי (de Lagarde). The subject and predicate, "comfortable" (or "sweet") and "in need" (i.e., "poor"), were reversed to provide a theological explanation for the absence of pain in the afterlife. This verse exemplifies S-Proverbs' tendency—Christian in character—to introduce affirmative statements about poverty (Joosten 1995, 68). In fact, 14:23b seems to allude to the afterlife. In 14:23c ( $\approx$  M 14:23a), S's source text had הוה for M's ההה 'or M's יהיה' and possibly כל for M's .

#### 14:24

G: (a) στέφανος σοφῶν πανοῦργος, (b) ή δὲ διατριβὴ ἀφρόνων κακή. "(a) The crown of the wise [pl] is clever, (b) but the pastime of fools is evil."

There are two ways to explain  $\pi \alpha \nu \delta \tilde{\rho} \gamma \rho \varsigma$ , where M has עשרם: (1) דעשרים mavoũργος may reflect ערמה מין (fem adj), though the גְּפָשׁוֹת עֵד אֶמֶת וְיָפְחַ בְּזָבִים מִרְמֶה: 26 בְּיִרְאַת יֲהוָה מִבְטַח־עֵׂז וּלְבָנְיו יִהְיֶה מַחְסֶה: 27 יִרְאַת יֲהוָה מְקוֹר חַיֵּיֵם לְסוּר מִמְּקְשׁי מֶוֶת: 28 בְּרָב־עֵם הַדְרַת־מֵלֶדְ וּבְאָפֶס לְאֵם מְחִתַּת <sup>ר</sup>רוֹזֵן<sup>ז</sup>: 29 אֶרֶדְ אַפַּיִם רַב־הְבוּגָה וּקְצַר־רוּחַ מֵרָים אָוֶלֶת: 30 חַיֵּיַ בְשָׂרִים לֵב מַרְפֵּא

In 14:24b, διατριβή and κακή eliminate M's tautology ("the folly of dolts is folly") by interpreting the first אולת as a particular type of behavior and the second as a moral quality; cf. the similar avoidance of an apparent tautology in 16:22.

S: (a) = M; (b) אמטארא אלאמס (b) and the perversion of fools is their folly." S too eliminates the tautology. T departs from S but instead of adjusting entirely to M (as it usually does), it avoids tautology by translating שטיותהון דסבלי שטיותהון ("and the naïveté of fools is their folly"). The versions' efforts to avoid the tautology in 14:24b show that they are working from M.

# 14:27 אמצות [ (העלאס M S (העלאס ) אמצות (reform, equal)

לסור [ M ] לסיר G ( $\pi$ οιεῖ δὲ ἐκκλίνειν) (graph  $1 \rightarrow 1$ 

G: (a) πρόσταγμα κυρίου πηγή ζωῆς, (b) ποιεῖ δὲ ἐκκλίνειν ἐκ παγίδος θανάτου. "(a) The command of the Lord is the fount of life. (b) It makes (one) turn away from the trap of death." πρόσταγμα = גַּצְוָת as usually in G-Proverbs (though תורת is also possible). There is no graphic similarity or ideological motive to explain the change. G is simply a variant proverb: מצות יהוה מקור חיים (27b = M). This is a variant of 13:14. There are many variant proverbs within M-Proverbs (catalogued by Snell 1993, 34–59), and it would make no sense for the critic to assimilate them to each other. Likewise, variant proverbs preserved in G-Proverbs are of equal validity to the ones preserved in M-Proverbs. In 14:27b, it is impossible to determine whether G read ' לָּהָסִיר = לֹסִיר ("to cause to turn away") or chose to treat לסור S translates לסור as לסור ("to those who turn aside"; thus T); cf. 13:14. S was prompted to take this step because an infinitive dependent on a noun could seem awkward. In other words, it felt more natural to say "The fear of the Lord is a fount of life *for those who turn away* from snares of death" than "... a fount of life *to turn aside from* snares of death."

14:28 רוזן G? (δυνάστου) ] רוזן Μ (metath ד → וז)

The form רווז, used in the plural in Prov 8:15; 31:4; and elsewhere, is correct. S's حلک ("king") is a serviceable guess and could have been used for either form.

G δυνάστου may reflect ((as in M), interpreting the latter as "potentate" out of contextual necessity.

#### 14:29

G: (a) μακρόθυμος ἀνὴρ πολὺς ἐν φρονήσει, (b) ὁ δὲ ὀλιγόψυχος ἰσχυρῶς ἀφρων. "(a) A man of patient temper is abundant in good sense, (b) but an impatient one is exceedingly foolish." In 14:29b, "exceedingly foolish" is a contextual guess at a difficult מרים אולת ("exalts folly"). In 3:35, מרים אולת was translated ὕψωσαν ("lift up, exalt"). Greek ὀλιγόψυχος and cognates include a range of undesirable emotions, including faintheartedness (e.g., Isa 35:4; Sir 4:9), despondency and distress (e.g., Ps 54[55]:9; Jdt 7:19; Isa 25:5), and impatience (as here and probably Exod 6:9, where it renders but rather embrace a bundle of related emotions.

#### 14:30

G: (a) πραύθυμος ἀνὴρ καρδίας ἰατρός, (b) σὴς δὲ ὀστέων καρδία αἰσθητική. "(a) The man of gentle temper is a healer of the heart, (b) but a knowledgeable heart is a moth [i.e., devourer] of bones." πραύθυμος is added to echo μακρόθυμος in the preceding verse and to reinforce the Stoic tone (BAP). G 14:30a continues the praise of patience from 14:29a, ignoring מרפא מון ("life of the body") and construing מרפא as the direct object of אברים, which is implicitly parsed as a participle in the H- or D-stem. G 14:30b is out of line with M-Proverbs' consistent affirmation of wisdom as a source of happiness, but G 14:10 expresses a similar attitude. (αἴσθησις/αἰσθητική translates words for knowledge/knowledgeable and is always a virtue.) וּרְקַב עַצָּמָוֹת קַנְאֶה: 31 עַׂשֵׁק־דָּל חֵרֵף עֹשֵׂהוּ וּמְכַבְּדוֹ חֹנֵן אֶבְיוֹן: 32 בְּרָעָתוֹ יִדֶּחֶה רְשֵׁע וְחֹסֶה <sup>ו</sup>בְתַמָּוֹ<sup>ז</sup> צַדֵּיק: 33 בְּלֵב גָבוֹן תִּנִוּחַ

14:31

4QProv<sup>b</sup> has 14:31–15:31.

G translates "Π ("insults") euphemistically as παροξύνει ("angers"), as in 17:5. Apparently the translator recoiled from speaking of an insult to God even when condemning the person guilty of this. (G 27:11 shows that the translator knew that "Π actually means "insult.") μυσι is translated narrowly as δ συχοφαντῶν (sim Ps 119[118]:122; Prov 28:3, 16; etc.). The classical meaning of συχοφαντεῖν is "to slander, accuse."

וברעתו 14:32 א ברעתו M G (לא גמגלמ מטינסט) א ברעתו S (at 14:35b) (סבעבדא נפאס) (graph  $\to \to \to$  )

M 14:32b means, at least in the traditional reading, "but the righteous hath hope in his death" (KJV). The hope in immortality is otherwise foreign to Proverbs. As emended, 14:32b means, "but the righteous man trusts in his innocence" (ABP). The proverb teaches that when disaster befalls the wicked, the upright can חוסה (take refuge and find shelter) in their the innocence. This is the message of Prov 10:2b and 11:4b as well.

The versions treat this verse as follows:

G: (a) ἐν κακία αὐτοῦ ἀπωσθήσεται ἀσεβής, (b) ὁ δὲ πεποιθὼς τῆ ἑαυτοῦ ἱσιότητι δίκαιος. "(a) The wicked man will be shoved down in his wickedness, (b) while the righteous man trusts in his piety." α', σ', and θ' read πέποιθεν ἐν τῷ θανάτῳ ἑαυτοῦ δίκαιος ("The righteous man trusts in his death" i.e., when he dies).

V: (a) *in malitia sua expelletur impius*, (b) *sperat autem iustus in morte sua*. "(a) The impious man will be cast out in his wickedness, but the just has hope in his death.

S: מה הכשאנים הלאביל לח ידער אין אווישי האמציים. (a) The evildoer is swept away by his evil, (b) but he who is confident that he has no sins is righteous."

T: בבישותיה מסתחף רשיעא ודתכיל דמאית צדיקא הוא (a) The evildoer is swept away by his evil, (b) but he who has trust, who dies, is righteous."

M's במותו ("in his death"), supported by 4QProv<sup>b</sup>, T, V, and the Hexa-

Although G 14:32b means "while the righteous man trusts in his piety," the Hebrew behind it, דתמו צדיק, would be better translated, "but the righteous man trusts in his innocence." This makes good sense in the context of Proverbs. It teaches that when faced with רעה ("trouble"), the righteous man relies for protection on his innocence rather than his own cunning (cf. 28:26) or his wealth (11:28). Prov 11:6 puts this message in very similar terms: "The righteousness of the upright will save them, while the treacherous will be trapped by (their) deceit."

We might weigh whether the change from במותו to בתמו was deliberate. Geiger explains M's reading as a theologically motivated alteration hinting at "die Belohnung, die ihm in der erneuten Welt werden wird," in accordance with Pharisaic doctrine (1857, 175). Geiger reasonably compares M's variant here to the Masoretic vocalization of Qoh 3:21, which attempts to obscure Qohelet's doubt about the afterlife by pointing the interrogative הָעָלָה ni ה הָעָלָה an article. However, the cases are not really comparable. The Masorah of Qoh 3:21 chooses a vocalization to make a doctrinally difficult text acceptable. This is a less radical move than a consonantal change in a theologically unexceptionable verse.

M's reading in 14:32b probably was due to an accidental  $\pi/\pi$  metathesis followed by the addition of the vowel letter ו במותו is unlikely to have been the original reading, because it strains the syntax. הסה always governs the preposition -= and is not used absolutely, in the sense of "have faith." הסה ב- is not followed by mention of an event, whether

<sup>3.</sup> Contrary to Jan de Waard (1998), the fact that 4QProv<sup>b</sup> agrees with M does not "reinforce the thesis that the reading in G is due to a kind of 'al tiqrâ' treatment" (93). It shows only that the reading arose sometime before the mid-first century CE, which we knew in any case from the Hexaplaric sources. Michael B. Dick (1990, 30) says that OG is introducing the Hellenistic Jewish concept of self-justification. However, that concept is not well established and would not force the deliberate elimination of an allusion to the afterlife.

# ַחְכְמֵה וּבְקֶרֶב לְּסִילִים תִּוְדֵעַ: 34 צְדָקָה תְרוֹמֵם־גֵּוֹי וְחֶסֶד לְאָמִים חַפֶּאת: 35 רֲצוֹן־מֵלֶדְ לְעֶבֶד מַשְׂבֵּיל וְעֶבְרָתוֹ תִּהְיֶה מֵבִישׁ:

good or bad. In other words, -ם חסה means "rely on" or "seek refuge in," not "expect" or "have hope for." In fact, the connotation of refuge and shelter is probably always present when this verb is used.

Once in existence, however, the variant with במותו would reasonably have been read as an allusion to the afterlife, as has indeed been done consistently by traditional interpreters. Another approach to M 14:32b is suggested by B. Vawter, who translates, "and the just man rejoices in his [sc. the wicked man's] death" (1972, 167). But -חסה is always followed by mention of the source of hope and protection.

# 14:33 (בבון M ] נכון G (מאמ<br/> $\beta \tilde{\eta})$ S (הבהיבה) (graph $\to \to )$

(logic) אלא G מ' θ' (où) S (גא) (logic)

G: (a) ἐν καρδία ἀγαθῆ ἀνδρὸς σοφία, (b) ἐν δὲ καρδία ἀφρόνων οὐ διαγινώσκεται. "(a) In the good heart of a man is wisdom, (b) but in the heart of fools it is not perceived." ἀγαθῆ = J, as in Job 42:8 G<sup>A (θ')</sup>. In 14:33b, a negative is added. The negatives in θ' and α' attest to its existence in some Hebrew MSS. (It could not have come to G-Proverbs as a Hexaplaric incursion, because the sentence makes no sense without it.) The negative was inserted within the Hebrew transmission for the sake of logic, because the notion of wisdom being known among fools seemed outlandish. (M actually means that the *wise* man's wisdom manifests itself among fools, in the form of chastisement.)

V (erudite) vocalizes תּוֹדְע. G S (אארב) and T (תתידע) follow M's passive.

S: גאאא הל הלבסה הכושה הידא היהבא היהבא היהבא (a)

In the heart of honest men wisdom dwells, (b) but in the heart of fools it is not known." S = G.

T ends the verse with שטיותא תתידע ("folly will become known"), to remove the apparent logical discrepancy in M.

14:34 אוחסר [ M וחסד G (לאמססטטטט אין א א אוחסר (מראין) (graph  $\to \to \to$ 

Ulrich et al. (2000, 185) transcribe וחסר in 4QProv<sup>b</sup>, but the ר is damaged and וחסד is quite possible. In any case, G έλασσονοῦσι δὲ and S witness to הסרו (קפר) (G recognized the rare Aramaism הסרויי). (Grecognized the rare Aramaism הסרויי) "disgrace" in 25:10 and could have done so here.)

V's *miseros facit* ("makes miserable") is indeterminate. But neither the verb הָסֵר nor the noun הֶסֶר fits the sentence as well as M's הֶסֶר ("dis-grace").

14:35 בשביל ] + 2 stichoi S (S 14:35bc) (displ from 14:32)

ועברתו [ M ] ווערמתו ה<br/>לG (ד<br/>ק $\delta \grave{\epsilon}$  המטדסט בטסד<br/>ססקלם) (graph ה $\to$ ה metath רמ $\to$ ה)

G: (a) δεκτὸς βασιλεῖ ὑπηρέτης δὲ νοήμων, (b) τῆ δὲ ἑαυτοῦ εὐστροφία ἀφαιρεῖται ἀτιμίαν. "(a) The intelligent servant is favored by the king, (b) and by his own adaptability he removes disgrace." G 14:35 and 15:1 form a unit advocating professional adaptability or even "opportunisme politique" (BAP). The *hapax* εὐστροφία means "the ability to adapt," "to change one's view" (BAP). Since it stands in place of M's ועברתו, it should be retroverted to the graphically similar וְּשֶׁרְמָתוֹ ("his cleverness") (Jäger).

For אנברתו, 4QProv<sup>b</sup> has either אועב תו, with a space where the א should be, or (according to Ulrich et al. 2000, 185) עברתו, with the hamaged or faded away, which is in any case correct. It is far-fetched to explain the writing in 4QProv<sup>b</sup> as an unattested עברתו, supposedly a form of עב"ת ("twist") and the basis for G's εὐστροφία (BHQ; de Waard 1998, 93). In any case, this supposition does not explain the space. It is no argument 15:1 מִשְׁנֶה־הֵדְּ יָשֵׁיב חֵמֶה וּדְבַר־עָּׁצֶב יַשְׁלֶה־אֶף: 2 לְשִׁוֹן חֲכָמִים הֵיטֵיב דֶעַת וּפִי לְסִילִים יַבִּיעַ אָוֶלֶת: 3 בְּכָל־מֲקוֹם עֵינֵי יְהוֶה צֹׁפֿוֹת רָעִים וטוֹבִים: 4 מַרְפֵּא לָשׁוֹן עֵץ חַיֵּים וְסֶלֶף בָּה שֶׁבֶר בְּרְוּחַ: 5 אָויל יִנְאַץ מוּסַר אָבֵיו וְשׁמֵר תּוֹבַחַת יַעְרֶם: 6 יּבְבֵית<sup>-1</sup> אֲדִיק חָסֶן רֶב

against ארמתו to observe that ערמה לערמה does not have a suffix elsewhere (de Waard 1999, 3). The nonattestation of ערמה with the suffix may be happenstance, since the noun occurs only five times. Moreover, we do not know if the noun v could take a suffix.

15:1 init + 1 stichos G (1a) (transition)

G: (a) ὀργὴ ἀπόλλυσιν καὶ φρονίμους, (b = M 15:1a) ἀπόκρισις δὲ ὑποπίπτουσα ἀποστρέφει θυμόν, (c = M 15:1b) λόγος δὲ λυπηρὸς ἐγείρει ὀργάς. "(a) Anger also destroys the intelligent. (b) But a submissive answer turns aside wrath, (c) and a painful word stirs up anger [pl]." G 15:1a was added to introduce the warning against anger and connects with 14:35b. By changing the notion of softness or gentleness (¬¬) in M 15:1a into submissiveness, G continues depicting the scenario of G 14:35, in which a royal official placates the king. B. Berachot 17a quotes this verse using ¬𝒴 C, V (T) = M.

15:2 אולת M ≈ G (אמאמ) [אלות M ≈ G (אמאמ) (metath לו → לו → לו )

 $G \approx M$ , but using אמאל as a broad equivalent for אולת, as in 14:24. For S has אולת ("curse"), apparently reading אולת ("curses") (thus S-Deut 29:20) but ignoring the plural; see the comment on 12:23. The same metathesis occurs in 12:23. BHQ explains G's אמאל פֿתוֹסדמדמו as based on טובות תדע for M's אויעיב דעת via י-i interchange and "a kind of *'al tiqra*' treatment, namely, a twofold metathesis." This explanation seems rather strained and unnecessary.

נפלס (<br/>transp מיל M ] ופלס אופלס הופלס (ל $\kappa$ סטל<br/>פMופלס הופלס (ל $\kappa$ סטל<br/>פ $\to$ סלפ (לס

ό δὲ συντηρῶν = פלס (Jäger). The translator thought that one meaning of פלס is "watch" or the like, drawing that notion from context in

Prov 5:21, where it is rendered σκοπεύει ("watch"). From "watching" to "guarding" is a reasonable semantic move.  $\sigma$  (noun and verb) is elsewhere translated by words meaning to cause harm or degradation, not by words for guarding or watching.

Instead of M's שבר ("breaking"), which is certainly correct, G  $\pi\lambda\eta\sigma\theta\eta\sigma\epsilon\tau\alpha = \psi$ שבע or שבע ("be sated," Jäger). (G could have read שבע and treated it as a future passive verb.) To be sure, it is conceivable that the translator understood שבר as its homonym "provisions," which he then paraphrased as "be sated." But this does not happen elsewhere, whereas  $\pi\iota\mu\pi\lambda$ '  $\mu\pi\lambda'$  is the usual correspondence in G-Proverbs and elsewhere. On  $\psi$  confusion, see the comment on 6:16. The phrase "filled with spirit" is unparalleled in M and G, though it is prominent in the NT (Luke 1:15, 41; Acts 2:4; 4:8; etc.). The unusual locution in G-Prov 15:4 supports the reading  $\psi$  See the comment on 6:16.

 $V \approx M$  but translates וסלף בה as *quae inmoderata est* ("that which is immoderate")—a unique understanding of סלף. This too seems to associate stich 15:4b with the theme of eating.

#### 15:5

G has לידטאלג ("commandments") for תוכחת ("reproof"). "Keeping the commandment(s)" is the usual expression, but since שמר תוכחת occurs also in 13:18, the phrase was not unfamiliar. The motive for the assimilation from commandment to reproof (which could have occurred in Hebrew or in Greek) is unclear.

G: (a) ἐν πλεοναζούσῃ δικαιοσύνῃ ἰσχὺς πολλή, (b) οἱ δὲ ἀσεβεῖς ὁλόρριζοι ἐκ γῆς ὀλοῦνται. (c) οἴκοις δικαίων ἰσχὺς πολλή, (d) καρποὶ δὲ ἀσεβῶν ἀπολοῦνται. "(a) In abundant righteousness there is much strength,

<sup>15:6</sup> ברבות\* ] הַיָא M; \* ברבות G (פֿא האנסטמלסט'סן) (M: haplog ברבות; G: near dittog בו  $\rightarrow$  בר, graph י  $\rightarrow$  ו)

<sup>ר</sup>וּתְבוּאַת ּ רָשָׁע נֶעְבֶּרֶת: 7 שִׂפְתֵי חֲכָמִים יְזְרוּ דֲעַת וְלֵב כְּסִילֵים לא־בֵן: 8 זֶבַח רֲשָׁעִים תּוֹעֲבַת יְהוֶה וּתְפַלֵּת יְשָׁרֵים רְצוֹנְוֹ: 9 תּוֹעֲבַת יֵהוֶה דֶּרֶדְ רָשֶׁע וּמְרַדֵּף צְדְקָה יֶאֶהֶב: 10 מוּסָר רָע לְעוֵב אֶׂרַח

(b) but the wicked [pl] will perish, completely uprooted from the earth. (c) There is much strength for the houses of the righteous, (d) but the fruits of sinners will perish." Stichoi ab are OG; cd are an adjustment to a text close to M.

because the verse mentions nothing that could be said to be "in" the produce of the wicked. The additional  $\beth$  could arise through a careless reading of  $\neg$  or a dittography with the preceding  $\neg$ , in spite of the intervention of the 1. G καρποί δέ in G 15:6d (which is an adjustment toward a Hebrew text of 15:6b) seems to reflect ותבואת, as does S's متللامهم, but both would have ignored the preposition in any case. However, the Hexaplaric  $\lambda \circ \gamma \circ \sigma \mu \circ \delta \epsilon (\alpha \lambda')$  supports the omission of the preposition. As emended above, the verse is to be translated, "In the house of the righteous there is much wealth, but the produce of the wicked is troubled." Baumgartner suggests that in G 15:6b, δλόρριζοι associates עקר with איז ("root"); however,  $\zeta/\sigma$  interchanges or wordplays do not occur elsewhere in G-Proverbs. Rather, G translates with όλλύναι, as in 11:17 and 15:27 (with έξ-), and adds δλόρριζοι and ἐκ γῆς for emphasis. S and V do not represent G 15:6ab, but only G 15:6cd.

15:7

G: (a) χείλη σοφῶν δέδεται αἰσθήσει, (b) καρδίαι δὲ ἀφρόνων οὐκ ἀσφαλεῖς. "(a) The lips of the wise [pl] are bound to [or 'by'] knowledge, (b) but the hearts of fools are not secure." In G 15:7a, δέδεται ("is bound") parses 'ί("disperse") as a passive of אז"ר ("gird on"). In 15:7b, ἀσφαλεῖς associates ciates ciates

15:8 תועבת יהוה M G (βδέλυγμα κυρίφ) S (כמארא מיאה CD A 11.21 (reform)

ישרים M G (אמדנטטעטעטעטעט S (אדיבא ) א G (אמדנטטעטעטעט א דקים (אדיבא ) CD A 11.21 (reform); + כמנחת CD A 11.21 (elab)

רצונו M G (δεκταὶ παρ' αὐτῷ) S (רצון [ (כ בענת) CD A 11.21 (reform)

4QProv<sup>b</sup> has 15:1-8 (fragmentary) = M. The Covenant of Damascus (CD A 11.20-21 = 4QD<sup>f</sup> [4Q271] 1.14) quotes this verse as דבח רשעים Since this quotation is treated as Scripture and introduced by תועבה ותפלת צדקים כמנחת רצון ("as it is written"), it may be considered a textual variant. M's רעובת יהוה ("abomination to Yahweh") is required for M, because ראונים ("his delight") assumes הוה cedent. As cited in CD, however, the divine name is not necessary. The CD sentence is a reformulation, not the original. כמנחת רצון ("like an acceptable cereal offering") is an expression foreign to Proverbs (and not found in the Bible). Moreover, the CD form of the proverb uses the cereal offering as a point of comparison for excellence, showing greater esteem for the temple cult than Proverbs usually does. There are no clear reasons for the other reformulations.

15:10 לעזב M ] אלעבר (<br/>טהא דŵי דמריט (graph ד $\to$ ר ה<br/>, metath ר ר ר ה הר ה ה אלעבר (בר ד

G: (a) παιδεία ἀxάxου [corr: xαxοῦ] γνωρίζεται ὑπὸ τῶν παριόντων, (b) οἱ δὲ μισοῦντες ἐλέγχους τελευτῶσιν αἰσχρῶς. "(a) "The education of the innocent [corr: 'wicked'] man is known by those who pass by, (b) and those who hate reproofs will perish shamefully." ἀxάxου is an error (dittog) for xαxοῦ = 𝒴 (Jäger). ὑπὸ τῶν παριόντων = ¬Ţ ('he who passes on the way," with number variance). The adjustments that G makes in other regards were in response to the somewhat puzzling text that G used here. The point of G 15:10a is that the discipline of the evildoer will be public and visible; similarly in 10:9. αἰσχρῶς is added to emphasize this point.

S: גארא גארא גראה גראה אין ביצא אין אין געוין געויא געוין געויען געוין געוין געוין

שׁוֹגָא תוֹכָחַת יָמְוּת: 11 שְׁאַוֹל וַאָּבַדּוֹז גָגֶד יְהוֶה אַׁף כִּי־לִבְּוֹת בְּנִי־ אָדְם: 12 לָא יֶאֶהַב־לֵץ הוֹבֵחְ לְוֹ אֶל־חְׁכָמִים לָא יֵלֵדְ: 13 לֵב שְׁמִח יֵיטֵב פָּגִים וּבְעַצְבַת־לֵב רַוּחַ נְבֵאֶה: 14 לֵב גָבוֹז יְבַקָּשׁ־דֶעַת <sup>ו</sup>וּפִי<sup>ז</sup> לְסִילִים יִרְעֶה אָוֶּלֶת: 15 כְּל־יְמֵי עָנִי רָעֵים וְטוֹב־לֵב מִשְׁתֶּה תָמִיד: 16 טוֹב־מֵעַט בִּיִרָאַת יִהוָה מֵאוֹצֵר לֵב וּמִהוּמַה בִוֹ: 17 טִוֹב אֵרָחַת

shame is exposed, (b) and those who hate reproof will die." S works from M but picks up from G the motifs of public exposure and shame.

#### 15:12

G: (a) = M; (b) μετὰ δὲ σοφῶν οὐχ ὁμιλήσει. "(b) and he will not consort with the wise." μετὰ δέ does not show a variant ואת for M's ואל. G (and subsequently S محر) was likely motivated by the image of going with someone as companion, influenced by 13:20, rather than going to someone for instruction. The same motive would explain ואת in Codex Vaticanus Urbinati Ebr. 1 (cited by BHQ).

# 15:13

G: (a) καρδίας εὐφραινομένης πρόσωπον θάλλει, (b) ἐν δὲ λύπαις οὔσης σκυθρωπάζει. "(a) When the heart is happy, the face flourishes, (b) but when it [the heart] is in sadness, it [the face] is sorrowful." G tightens the parallelism of 15:13b with 15:13a by removing "spirit" and "heart" in 15:13b. The subject of σκυθρωπάζει is πρόσωπον; compare how σκυθρωπός modifies πρόσωπον in Gen 40:7; θ'-Dan 1:10; and Sir 25:23. Both stichoi focus on face as an indicator of mood.

S uses בגים ("body") for פנים, perhaps reasoning that good cheer affects the whole body. S also uses גהה for גהה for גהה ("body"?) in 17:22a.

ני 15:14 אופני א $M^Q\,G \;(\sigma \tau \acute{o} \mu \alpha \; \delta \acute{e}) \; S \;($ הפהדא MK (syn) ופני MK (syn)

ירעה M ] ידעה  $* \approx G (\gamma \nu \acute{\omega} \sigma \epsilon \tau \alpha \iota) (\text{graph} \rightarrow \tau)$ 

G: (a) καρδία ὀρθὴ ζητεῖ αἴσθησιν, (b) στόμα δὲ ἀπαιδεύτων [var: ἀσεβῶν  $G^{S}$ ] γνώσεται κακά. "(a) The straight heart seeks knowledge, (b) but the mouth of the uneducated [var: 'impious'] will know evils."

Jäger retroverts مَوْمُ to لَحَان ("established") (for M's (لحاز), but that correspondence is not found elsewhere. S uses حجمت ("of the honest"), which might seem like homoiophonic rendering of الحاز (associating it

with בון), but that correspondence is unique, whereas נבון is translated by איז in 14:33 as well.

ים is correct, for the mouth, but not the face, can be said (metaphorically) to "shepherd" or "graze on" knowledge. The erroneous ופני because "mouth" and "face" are pragmatic synonyms in many contexts (see 2:6), though "face" does not work here. The appearance of ג would have been facilitated, if not occasioned, by near dittography, since ב and are very similar in some square scripts, for example, the Nash Papyrus (see Yardeni 1997, 173, where sometimes ב is distinguished from ג only by a slight bulge to the right at the top of the vertical).

For M's ידעה, אילעה ידעה ידעה or an ungrammatical ידעה, which G would have had to render "knows" anyway. A similar error occurred in 10:21. In the present verse, S apparently read ירעה, and, not understanding the metaphor, made do with the serviceable  $\rightarrow$  ("speaks").

The G<sup>S</sup> variant  $\dot{\alpha}\sigma\epsilon\beta\tilde{\omega}\nu$  continues the tendency, begun in OG (e.g., 10:26), of rendering words for ignorance and folly by words for immorality. G = qere.

15:16

By translating מהומה as ἀφοβίας ("lack of fear," sc. of the Lord), G creates a sharper antithesis to "fear of the Lord" and places מהומה in the realm of religious concerns.

G: (a) κρείσσων ξενισμός λαχάνων πρός φιλίαν καὶ χάριν (b) η παράθεσις μόσχων μετὰ ἔχθρας. "(a) Better a feast of greens for (the sake of) love and favor (b) than a meal of beef with hostility." (ξενισμός is entertainment of

ְיָרָק וְאַהֲבָה־שֶׁם מִשִּׁוֹר אָׁבוּס וְשִׂנְאָה־בְוֹ: 18 אֵישׁ חֲמָה יְגָרֶה מְדְוֹן וְאֶרֶך אַפַּיִם יַשְׁמִיט רִיב: 19 דֶּרֶך עֲצֵל בִּמְשֵׂכַת חֲדָק וְאָׁרַח יְשָׁרִים וּסֶלְלֶה": 20 בֵּן חֻכָם יְשַׂמַּח־אֶב וּכְסִיל אָׁדָם בּוֹזֶה אִמְוֹ: 21 אֵוֶעָת שִׁמְחָה לַחַסַר־לֵב וְאִישׁ הְּבוּנְה יְיַשֶׁר־לֶכֶת: 22 הָפֵּר מֲחַשָּׁבוֹת בְּאֵין

S's מעסבא גברא ("and love of name/reputation") implicitly vocalizes ואָהָבָת שָׁם (Pinkuss).

# **15:18** fin ] + 2 stichoi G (18*a*) (elab)

G: (a) ἀνὴρ θυμώδης παρασκευάζει μάχας, (b) μακρόθυμος δὲ καὶ τὴν μέλλουσαν καταπραύνει. "(a) A hot-tempered man prepares conflicts, (b) but a patient man calms even one in the offing." Patience prevents quarrels before they start.

#### 15:18a

G: (a) μαχρόθυμος ἀνὴρ κατασβέσει κρίσεις, (b) ὁ δὲ ἀσεβὴς ἐγείρει μᾶλλον. "(a) A patient man extinguishes contentions, (b) but an impious one stirs (them) up (even) more." Paronomasia of μαχρόθυμος (15:18b–18*a*a) with θυμώδης (15:18a) helps link the two couplets (BAP).

G 15:18*a* has the obelus in SyrH, which would seem to mark G 15:18 as a revisionary addition (Fritsch 1953, 174). However, in spite of the obelus and G 15:18's differences from M, G 15:18 is probably not Hexaplaric. Verse 18 is probably OG and 15:18*a* a later addition, composed independently and placed here by association with 15:18. Verse 18*a* shows signs of being based on G 28:2. Note also the unhebraic word order in 15:18*a*a and the adverbial use of  $\mu \tilde{\alpha} \lambda \rho \nu$ , which in verses translated from the Hebrew serves only as a comparative particle.

Interestingly, G 15:18a is not a "pertinent religious interpretation"

(Cook 2000, 174)—the religious enhancement is too slight to be the motive—but rather an example of the ongoing emergence of new proverbs within the Greek scribal tradition.

# 15:19 סָלָלָה = סְלָלָה (mod) א סָלָלָה M (mod)

4QProv<sup>b</sup> has much of 15:19–31, with different readings in verses 19 and 28. For M's סָלָלָה, 4QProv<sup>b</sup> has סוללה, presumably to be vocalized סוללה (smoothed," "leveled"), a true G-passive participle. Since this form is the more archaic, it is probably what was originally intended by M's consonants סללה, rather than M's סַלָלָה (also a G-stem passive participle), especially since the long  $\hat{u}$  of the latter form is regularly written with a 1, unless (as in Gen 31:39) it has a possessive suffix. On the G-passive, see GKC §58 and IBHS §22.6. The perfect and participle of true G-passives merged with D-passives, e.g., אָבָל (Exod 3:2).

In G-Proverbs, מֹעסׁרָבוֹעִים renders ארוצים ("diligent, industrious") when the context refers to diligence (10:4; 13:4; 11:16?). It can also render terms for strength: אבר (12:4; 31:10) and גבר (28:3). מֹעסֹרָבוֹנה (12:4; 31:10) and גבר 11:30, where the context is military skill. מֹעסֹרְבוֹנה in 21:30, where the context is military skill. מֹעסֹרְבוֹנה is not used of distinctly moral virtues. In the present verse, the choice of דְשׁׁע מֹעסׁרְבוֹנה was probably prompted by its antithesis עִצַל ("sluggard").

# 15:20

G renders וכסיל אדם (lit. "a fool of a man," or "a fool-man") as viòç dè ส്фрыv ("a foolish son"), to tighten the parallelism with "father." The phrase is translated สัфроисс dè สังปัวะุ in 21:20.

# 15:21

G: (a) ἀνοήτου τρίβοι ἐνδεεῖς φρενῶν, (b) ἀνὴρ δὲ φρόνιμος κατευθύνων πορεύεται. "(a) The paths of the mindless person lack sense, (b) but the prudent man walks straight." Perhaps thinking it absurd that folly could be a joy to anyone, G substitutes a more predictable statement about the mindless.

S resolves the same difficulty differently: המבל עמעי לבא הס גראי היא גמעי ("A man who is foolish lacks sense").

#### 15:22

G: (a) ບໍ່περτίθενται λογισμούς οἱ μὴ τιμῶντες συνέδρια, (b) ἐν δὲ καρδίαις βουλευομένων μένει βουλή. "(a) Those who do not respect assemblies delay plans, (b) but counsel abides in the hearts of those who take counsel." G ບໍπερτιθέναι (only here for הָפָר ["confound," "violate"]) means סִוֹד וּבְרָב יוֹעַצִים תָּקוּם: 23 שִׁמְחָה לֲאִישׁ בְּמַעַנֵה־פֵּיו וְדָבֶר בְּעָתִוֹ מַה־טְוֹב: 24 אַרַח חֵיִּים לְמַעְלָה לְמַשְׂבֶּיל לְמַעַן סוּר מִשְׁאוֹל מֵטָה: 25 בֵּית גֵּאִים יִפָּח | יְהוֶה וְיַצִּב גְּבַוּל אַלְמָנֶה: 26 תּוֹעַבַת יֲהוָה מַחְשְׁבָוֹת רֶע וּטְהֹרִים אָמְרֵי־נְעַם: 27 עֹבֵר בֵּיתוֹ בּוֹצֵע בֶּצַע וְשוֹנֵא

"put off," "defer" (GELS), and also "to omit" (LSJ). Baumgartner retroverts έν δὲ καρδίαις to ובלב . However, since μετὰ καρδίας βουλευτικῆς corresponds to ברב יועץ in G 24:6, a translational explanation is better. Rather than praising a large number of advisors, G prefers to stress the intellectual quality ("heart" = wisdom) of the advice. In a similar vein, in 11:14, while M praises the efficacy of "many advisors," G praises the safety that resides in πολλῆ βουλῆ, "much counsel"; similarly in G 11:14; 24:6. In 15:22b, G supplies βουλή as the subject of "stand," "succeed"), perhaps not finding it easily in the first stich. G thereby creates a wordplay βουλευομένων-βουλή.

G is saying that if people fail to respect deliberations in the assembly, their own plans will fail, whereas those who confer with others receive wise counsel and thus will succeed. G has changed the verse from a recommendation to have many advisors (probably with reference to the royal court) into an affirmation of public councils or assemblies. Jewish assemblies existed in the Hellenistic period in various forms—for example, the Gerousia in Palestine and Alexandria and the Lesser Synhedria in Palestine; see the comment on Prov 31:23.

S: הכבוי, העשברא אילא אילא גרעסיי, בנסשא איז גילבע. (a) Those who honor the assembly make their plans pass, (b) and by a multiplicity of counselors [lit. 'counselors of counsel'] it [sc. the assembly] is established." S 15:22a is a converse translation of G 15:22a; S 15:22b = M.

15:23

G: (a) οὐ μὴ ὑπακούσῃ ὁ κακὸς αὐτῇ (b) οὐδὲ μὴ ἐἴπῃ καίριόν τι καὶ καλὸν τῷ κοινῷ. "(a) The bad man will surely not obey it [sc. counsel; see 15:22], (b) nor will he say anything that is timely or good for the commonweal." Whereas M praises the satisfaction of answering back, G formulates a new proverb loosely based on M 15:23. G 15:23 continues the thought of 15:22. The idea is that since a *bad* man does not obey counsel, he cannot say anything of value to others. In 15:23b, TELC

as וְדָבֵר, with the negative carried forward from 15:23a. The notion of the commonweal is taken from συνέδρια in the preceding verse.

# 15:24

G: (a) όδοὶ ζωῆς διανοήματα συνετοῦ, (b) ἵνα ἐκκλίνας ἐκ τοῦ ἄδου σωθῆ. "(a) The thoughts of the intelligent man are ways of life, (b) so that, turning aside from Hades, he may be saved."

G has διανοήματα for M למעלה. This rendering may have a vague relation to the idiom העולה על רוחכם ("that which comes up in your spirit," i.e., "what you plan") in Ezek 20:32, though על רוחכם would seem necessary to convey the thought. See the comment on 14:14.

 $\sigma\omega\theta\tilde{\eta}$  for מטה ("below") is hard to explain. מלט (BHQ) is too dissimilar to be a secure retroversion, in the absence of further evidence. G is probably clarifying what it means to "turn aside" from Hades.

# 15:26

G: (a)  $\beta \delta \epsilon \lambda \nu \gamma \mu \alpha \kappa \nu \rho i \omega \lambda \delta \gamma \iota \sigma \mu \delta \varsigma$  ädikos, (b) a vue de phoeis σεμναί. "An abomination to the Lord is the thought of the unrighteous, (b) but the utterances of the pure are august [lit. 'of the pure, the utterances are august']." In 15:26b, G is overriding the word order, and consequently the syntax, to better parallel the first stich; cf. the comment on 11:21.

#### 15:27

G: (a) ἐξόλλυσιν ἑαυτὸν ὁ δωρολήμπτης, (b) ὁ δὲ μισῶν δώρων λήμψεις σψζεται. "(a) The gift-taker destroys himself, (b) but he who hates the taking of gifts is saved." G (correctly) understands ביתו ("his house") as equivalent to one's possessions, hence "himself." G construes LEV ("he who takes unjust gains") narrowly (and probably correctly) as bribe taking, in light of the second stich. S ≈ G.

# 15:27*a*-16:9

G's verse sequence differs from M's. (S follows M fairly closely.) Using Rahlfs's numbering of G and comparing it to M, G's sequence is:

G	Μ
[15:27]	15:27
[15:27 <i>a</i> ]	16:6
[15:28]	15:28
[15:28 <i>a</i> ]	16:7
[15:29]	15:29

[15:29 <i>a</i> ]	16:8
[15:29 <i>b</i> ]	16:9
[15:30]	15:30
[lacking]	15:31
[15:32]	15:32
[15:33]	15:33
[lacking]	16:1
[16:2a]	≈ 16:2a
[16:2b]	16:2b
[lacking]	16:3
[lacking, but cf. 16:9]	16:4
[16:5]	16:5
[lacking]	16:6; cf. G 15:27a
[16:7]	<i>≠</i> 16:7
[16:8]	<i>≠</i> 16:8
[16:9]	≈ 16:4
[16:10ff]	= M

The shared core is M 15:27–30, 32, 33; and 16:4–9; G maintains the sequence of M 16:6–9 while interlacing these verses in 15:27–30. On this section, see Tov 1990b, 50.

# EXCURSUS ON G-PROV 15:27a-16:9

It is likely that a single person, whether a scribe in the Hebrew transmission, or the Greek translator, or a scribe in the early Greek transmission, was responsible for the relocations. Apart from the relocations, the translator enhanced the cohesiveness of the unit in several ways. G 15:28, 28*a*, 29, 29*a*, 29*b* are linked by the root  $\delta_{1\times\alpha_1-}$ . This clustering depends on the presence of  $\delta_{2\times\tau\alpha_1}$  in 15:28*a* and  $\delta_{1\times\alpha_1\alpha}$  in 15:29*b*, verses which probably originated with the translator. Moreover, the paronomasia of  $\pi_{1\circ\tau\tau\circ\tau\nu}$  in 15:27*a* and  $\pi_{1\circ\tau\tau\circ\tau}$  in 15:28 was clearly the translator's doing (D. A. Teeter, pers. comm.). Still, this enhanced cohesiveness cannot explain most of the relocations.

There is no satisfactory explanation, mechanical, ideological, or literary, for the absence in G of verses present in M. Some of these verses are tightly integrated into their context in M, which speaks of Yahweh's control of human destiny (16:9). Moreover, some of the G minuses are really displacements. We must admit the possibility that scribes in the proto-M transmission were inspired by the context to add additional relevant proverbs. Still, I am inclined to accept the originality of content and location of this section in M. It shows the kind of thematic clustering typical of the proverb collections. The question is whether we can go beyond these general observations and discern a tighter, more deliberate plan of organization. I am not persuaded by the attempts to do so.

(1) BAP (45–47) proposes that M arose from an editorial reworking that created a block of "Yahwistic" sayings (15:33–16:9), while G displays a "theistic" series marked by the use of  $\theta \epsilon \delta \varsigma$ . This section has an unusually large number of cases in which G has  $\theta \epsilon \delta \varsigma$  for M's הוה (12 out of 35 times). Elsewhere,  $\kappa \delta r = 2 \delta \delta \varsigma$ , יהוה  $\delta \epsilon \delta \varsigma$ , with few exceptions. The problem with BAP's theory is that throughout Proverbs. There than הוה '(rather than הוה BAP's theory is that throughout Proverbs, and this distribution is reflected for the most part in G. BAP's theory requires presuming that at an early stage, in proto-Proverbs, Yahwistic redactors did their work everywhere *except* in 15:33–16:9. BAP's theory also does not account for other disruptions in the section.

(2) R. Scoralick argues that G-Prov 15:27a-16:9 is a "kreative Umgestaltung des MT durch die LXX" that forms a carefully designed proverb composition (2002, 57). If such a reorganization had been undertaken in the Hebrew transmission, this would reinforce my belief (following Tov 1990b) that in these verses (and elsewhere) G represents a different recension. Scoralick, however, maintains that the changes are the *translator's* doing (2002, 49), not a Hebrew scribe's. She believes that this translator reveals himself as "ein poetisch begabter Gelehrter ... der erstaunlich frei mit dem Text umgeht" (72).

The only argument Scoralick offers in support of this hypothesis is that in 15:29, 33; 16:2, 5, and 7 הוה 'is translated θεός, not κύριος, as it usually is. (In 15:27*a*, 28*a*; 16:8, 9, κύριος is used.) The use of θεός, she says, does not match the M of these verses. "Die griechische Übersetzung hat durch den Wechsel der Gottesbezeichnungen, der keinen Anhalt im MT hat, einen Rahmen und einen Innenteil im Bereich unserer Verse gestaltet" (2002, 68). But this is a rather strange argument, since of the eightyseven times ההוה is used in Proverbs, it is translated by θεός in eighteen or twenty cases (two are uncertain). G (like other, even more literal, translators) felt free to render the Tetragrammaton by θεός and probably saw no distinction between these two synonyms. There is no reason for the intended reader of G-Proverbs to assign special importance to the clustering of θεός here and to recognize the use of χύριος in 15:28*a*, 29; 16:8, and 9 as instrumental in framing the composition.

Especially confusing is the way the segments of Scoralick's design

מַתְּנִת יִחְיֶה: 28 לֵב אֲדִיק יֶהְגֵּה לַעֲגָוֹת וּפִּי וְׁשָׁעִים יַבִּיעַ רָעְוֹת: 29 רְחַוֹק יֲהוָה מֵרְשָׁעֵים וּתְפִלֵּת צַדִּיקֵים יִשְׁמֲע: 30 <sup>ר</sup>מַרְאֵה־-אֵינַיִם יְשַׂמֵּח־לֵב שְׁמוּעֵה טוֹבָה תְּדַשֶׁן־עֲצֶם: 31 אֹזֶן שׁׁמַעַת תּוֹכַחַת חַיֵּיָם

overlap and clash. For example, she sees 15:27a-28a + 16:8-9 as forming a framework distinguished from the "Innenteil" by the use of χύριος (68). Independent of this tripartite segmentation, Scoralick maintains, the unit is also to be divided into two unequal parts: 15:27a-33 and 16:2-9. But the reader would have no reason to identify 15:27a as the start of a unit or 16:9 as its end (nor has any reader besides Scoralick done so, as far as I can tell). In fact, the proposed design is incoherent and irrelevant to comprehension of the passage.

However, the inner resonances that Scoralick points out are relevant to interpretation. In G, the theme of dependency on God is indeed less accentuated than in M, while the theme of righteousness is made more prominent. This change in emphasis could have come about simply by the clustering of proverbs on righteousness and by the addition of "righteous" in 15:28*a* ( $\approx$  M 16:7); 15:29*b* ( $\approx$  M 16:9); and 16:7, 8 ( $\neq$  M 16:8), 9 ( $\approx$  M 16:4) and of "unrighteously" in 16:5 (where an idiom is misunderstood). Some of this enhancement may indeed be the translator's doing. The diminution of the theme of dependence on God results from the omission of M 16:1 ("The dispositions of the heart belong to a man, but the answer of the tongue is from the Lord") and M 16:3 ("Entrust your deeds to the Lord, and your plans will succeed"). Still, M 16:9, which insists on divine control, is maintained in G 15:29*b*, at the center of the unit in question. There is no great shift of theme.

In the comments on this section, M's order and numbering will be followed. Hexaplaric restorations will not be discussed.

# 15:28 יהגה M G (μελετῶσιν) S (ינה) ] > 4QProv<sup>b</sup> (unc)

and T [בהימנותא] is indeterminate because there was no plural available.) According to Baumgartner, G associates ענוה לענות ("humility"), but  $\pi$ נסדו; nowhere else renders words for humility.

4QProv<sup>b</sup> lacks יהגה, for unclear reasons. The word is essential to the sentence.

#### 15:29

G 15:29a = M 16:8; G 15:29b = M 16:9 (see below).

15:30 אראה G ( $\theta$ בש<br/>קשור [ (שמיה) M S (המסויה) (metath אר  $\to \to +$  , orth, assoc)

G: (a) θεωρῶν ὀφθαλμὸς καλὰ εὐφραίνει καρδίαν, (b) φήμη δὲ ἀγαθὴ πιαίνει ὀστᾶ. "(a) The eye, beholding good things, makes the heart rejoice, (b) and a good saying fattens the bones." θεωρῶν = ϫηκη, treated as a verb, though a G-stem participle would be required. (אור גראה is never translated by a word for seeing in the Septuagint of the Hebrew Bible.) The change was partly graphic, due to the similarity of and and and, but it was facilitated by the association of both words with "eyes" (see below). Two directions of change are possible:

(1) מאור הראה האור. If מאור is original, the Hebrew would be translated, "The light of the eyes makes the heart glad, etc." This means something like "joy makes the heart glad" or "vitality makes the heart glad." (Compare this phrase with idioms using אור [verb and noun] + עינים + 1 Sam 14:27 [qere], 29; Ps 13:4; Prov 29:13, and elsewhere.) This reading is possible but rather awkward and tautologous.

בְּקֶרָב חֲכָמֵים תָּלִיז: 32 פּוֹרַע מוּסָר מוֹאֵס נַפְּשֵׁוֹ וְשׁוֹמֵע תוֹבֹחַת קוֹנָה לֵב: 33 יִרְאַת יְהוָה מוּסַר חָכְמֵה וְלִפְנֵי כְבוֹד עַנָוֶה: 16:1 לְאָדָם מֵעַרְבִי־לֵב וֹמִיְהוָה מַעֲנֵה לָשׁוֹז: 2 בֶּל־דַּרְבִי־אֲישׁ זַדְ בְּעֵינֵיו וְתֹבֵן רוּחוֹת יְהוֶה: 3 גַּל אֶל־יְהוֶה מַעֲשֶׂיד וְיִבֹּנוּ מַחְשְׁבֹתֶידְ: 4 כָּל בְּעַל יֶהוָה <sup>ר</sup>ּלְמַעֲנֵהוּ וְגַם־לִשָּׁע לְיָוֹם רָעֵה: 5 תּוֹעַבַת יֶהוָה בָּל־אָבַהּ־

15:31 M S ] > OG (unc)

Absent in OG, see above on 15:27*a*–16:9.

15:32 אושומע M S (הברב) א יושומע האומע א א א ושומע (syn, equal)

15:33

G: (a) φόβος θεοῦ παιδεία καὶ σοφία, (b) καὶ ἀρχὴ δόξης ἀποκριθήσεται αὐτῆ. "(a) Fear of God is education and wisdom, (b) and the beginning of honor is to respond to it." The καί in 15:33a keeps "education" and "wisdom" distinct, as they are in M 1:2, 7; and 23:23. G 15:33b has the asterisk in SyrH because of its difference from M. The differences from M were provoked by misunderstanding ענוה "answer." Possibly the word was implicitly vocalized as "utile, an imperative, which was treated as an infinitive. (In other words, "and the beginning of honor—answer it!" is paraphrased as "the beginning of honor is to answer it.") "It" refers to Wisdom's summons; cf. 1:20; 8:1. G<sup>Ss A</sup> and some minuscules add (with minor variations) (c) προσπορεύεται δὲ

דמדבועסוֹג אָלאָמ, "(c) and honor goes before the humble." De Lagarde retroverts this stich to ולפני ענוים כבוד. But G varies number and ignores word order on several occasions (see the comment on 11:21), and the additional stich is best explained as a partial adjustment to M.

S: אבאס, השנאס הביבאס, ארבישה, הסגערט ארביבאס, ארבישה, האנשה, הארבישה, ארבישה, העולש, "(a) The fear of the Lord is the teaching [var: "fount"] of life, (b) and the glory of the humble goes before him." "Teaching of life" or "fount of life" (the latter very likely the original) is influenced by 14:27a. Also, אנשה, מולשה, and אנשה, are conjoined in S 16:21–22. S 15:33b looks like a rephrasing of G 15:33c.

**16:1** M S ] > OG (unc, equal)

Absent in  $G^B$  (= OG); see above on 15:27*a*–16:9. It is supplied in SyrH, to which is added a variant of Sir 3:18 ("The greater you are, the more you should humble yourself, and you will find favor before the Lord").

16:2

G: (a) πάντα τὰ ἔργα τοῦ ταπεινοῦ φανερὰ παρὰ τῷ θεῷ, (b) οἱ δὲ ἀσεβεῖς ἐν ἡμέρα κακῆ ὀλοῦνται. "(a) All the works of the humble are visible to God, (b) but the wicked [pl] will perish in the evil day." G 16:2a is related to M only loosely and cannot be Hexaplaric. G 16:2b is lacking in G<sup>B</sup> = OG, and in its place is a stich based on 16:4b (which, in G, is at 16:9b). It cannot be determined if the change is the work of the translator or an earlier scribe. A retroversion based on M usages elsewhere in Proverbs (and ignoring number differences as resulting from G's flexibility in number) would be הרשע יאבד ביום רעה.

S uses רוחות ("his way") for רוחות in 16:2b. This makes it clear that the verse speaks about human spirits rather than actual winds.

# **16:3** M S ] > OG (unc, equal)

Absent in OG; see the introduction to 15:27*a*–16:9.

# 16:4 לַמַעַנֵהוּ [ לַמַעַנֵהוּ M (vocal err)

M's anomalous לְמַעֵנְהוּ must be vocalized לְמַעֵנְהוּ; see ABP.

An approximation of M 16:4 is found at G 16:9: (a) πάντα τὰ ἔργα τοῦ κυρίου μετὰ δικαιοσύνης, (b) φυλάσσεται δὲ ὁ ἀσεβὴς εἰς ἡμέραν κακήν. "(a) All the works of the Lord are (done) with righteousness, (b) and the wicked man is kept for the evil day." "With righteousness" attempts to clarify what it means for God to do something למענהו, which could mean either "for his own sake" (V propter semet ipsum) or "for its purלֵּב יָד לְיָד לָא יִנָּקֶה: 6 בְּחֶסָד וָאָמָת יְכַפַּר עָוֹן וּבְיִרְאָת יְהוָה סִוּר מֵרְע: 7 בִּרְצַוֹת יְהוָה דַּרְכֵי־אֵישׁ גַּם־אוֹיְבָיו יַשְׁלָם אִתּוֹ: 8 טוֹב־מֲעַט בִּצְדָקָה מֵרְב וּתְבוּאוֹת בְּלָא מִשְׁפֶּט: 9 לֵב אָדָם יְחַשֵּׁב דַּרְכָּו וִיהוָה יְכִין צַעֲדְוֹ: 10 קֶסֶם | עַל־שִּׁפְתֵי־מֶלֶך בְּמִשְׁפָט לָא יִמְעַל־פִּיו: 11 פֶּלֶס | וּמֹאוְנֵי מִשְׁפָּט לִיהוֶה מַעֲשֵׁהוּ כָּל־אַבְנֵי־כִיס: 12 תּוֹעַבַת מֵלְכִים עֲשִׂוֹת רֶשַׁע כִּי בִאָדָקָה יִכּוֹן כִּמֵא: 11

pose." Ben Sira (39:30, 34) interprets this concept similarly; see ABP on the present verse.

S (T) understands למענהו as "to him who responds (ענה) to him" and rephrases this as גרשאבים לה ("to them who obey him").

# 16:6

G of this verse is at 15:27*a*. See the introduction to 15:27*a*–16:9.

נרצות הברצות M S ( $_{\Box_{2}}$  ברצות ( $_{\Box_{2}}$  ( $_{\Box_{2}}$  ברצות G (at 15:28) ( $\delta \epsilon \kappa \tau \alpha l$ ) (graph  $\rightarrow J$ )

G (G's 15:28*a*): (a) δεκταὶ παρὰ κυρίω όδοὶ ἀνθρώπων δικαίων, (b) διὰ δὲ αὐτῶν καὶ οἱ ἐχθροὶ φίλοι γίνονται. "(a) The ways of the just [pl] are acceptable to the Lord; (b) through them even enemies become friends." This is a paraphrase, probably occasioned by reading LCVLC ("acceptable") for M's LCVLC (Baumgartner). G moralizes by substituting "of the just" for "of a man." G overrides M's word order; for further examples, see the comment on 11:21.

At 16:7, G has a different proverb: (a) ἀρχὴ ὁδοῦ ἀγαθῆς τὸ ποιεῖν τὰ δίκαια, (b) δεκτὰ δὲ παρὰ θεῷ μᾶλλον ἢ θύειν θυσίας. "(a) The beginning of the good way (is) to do righteous deeds. (b) (These are) more acceptable to God than offering sacrifices." G 16:7a is based on M 1:7; G 16:7b  $\approx$  M 21:3b.

S איס vocalizes יִשָּׁלִם, for M's יִשָּׁלִם, producing a sentiment ("[God] will also punish his enemies") that is nearly the opposite of the Hebrew.

16:8

M 16:8  $\approx$  G 15:29*a*. At 16:8, G has a different proverb: (a) ό ζητῶν τὸν κύριον εὑρήσει γνῶσιν μετὰ δικαιοσύνης, (b) οἱ δὲ ὀρθῶς ζητοῦντες αὐτὸν

εύρήσουσιν εἰρήνην. "(a) He who seeks the Lord will find knowledge with righteousness, (b) and those who seek him uprightly will find peace."

# 16:9

G 15:29b: (a) καρδία ἀνδρὸς λογιζέσθω δίκαια, (b) ἕνα ὑπὸ τοῦ θεοῦ διορθωθῆ τὰ διαβήματα αὐτοῦ. "(a) Let the heart of a man plan righteous things, (b) so that his steps may be made straight by God." G makes M's apparently deterministic statement into a moral instruction by construing the first verb as jussive and substituting "righteous things" for "his way." G 16:9 = M 16:4; see the comment on the latter.

# 16:11

# 16:12

G: (a) βδέλυγμα βασιλεῖ ὁ ποιῶν κακά, (b) μετὰ γὰρ δικαιοσύνης ἑτοιμάζεται θρόνος ἀρχῆς. "(a) An abomination to the king is he who does evil, (b) for the throne of rulership is established by righteousness." G treats the infinitive  $\eta$  as a participle ( $\eta$ ψψ) and adds "rulership." G thereby excludes the inference that kings naturally and inevitably hate evil. This inference is allowed for by M's "Doing of evil is the abomination of kings." G shows less enthusiasm for kings and counselors and less confidence in their virtue than does M-Proverbs, which was composed in part by royal courtiers; see ABP 2.500–503. See the comments on 11:4; 15:22; 20:5, 8; 24:6.

S seems bothered by the same problem. With little attention to the Hebrew grammar, he translates 16:12a as منحم متلجم ("impure are the kings who do iniquity").

אָדָק וִדֹבֶר יִשְׁרֵים יָאָהֶב: 14 חַמַּת־מֶלֶדְ מַלְאַבֵי־מֶוֶת וְאָישׁ חָבְם יְכַפְּרֶנָּה: 15 בְּאוֹר־פְּגִי־מֶלֶד חַיֵּים וֹּרְצוֹנוֹ כְּעָב מַלְקוֹשׁ: 16 קֵנֹה־ חְכִמָּה מַה־טִּוֹב מַחָרָוּץ וּקְנִוֹת בִּינָה נִבְחָר מִבְּסָף: 17 מְסִלַּת יֲשָׁרִים סְוּר מֵרֶע שׁמֵר נַפִּשׁוֹ נֹצֵר דַּרְכָּוֹ: 18 לִפְּגִי־שֶׁבֶר גָּאָוֹן וְלִפְגַי כִשָּׁלוֹן אַנִּבַהּ רְוּחַ: 19 טִוֹב שִׁפַּל־רְוּחַ אֶת־יּעֲנִיֵים מַחַלֵּק שֶׁלָל אֶת־גַּאִים: גַּבַהּ רְוּחַ: 19 טִוֹב שְׁפַל־רְוּחַ אֶת־יּנְצִיִים מַחַלֵּק שְׁלָל אֶת־גַּאִים: 20 מַשְׂבִּיל עַל־דֶּבָר יִמְצָאַרטֵוֹב וּבוֹטֵחַ בַּיהוֶה אַשְׁרֶיו: 21 לַחַכַם־לֵב יִקְרֵא נָבְוֹן וּמֶתֶק שְׁפָּתִים יֹסִיף לֶקַח: 22 מְקוֹר חֵיִים שֵׁכֶל בְּעָלֶיו

ני 16:15 אבני [ (גפי מפת) א מעז G (טוֹטָן) (graph  $\to \to \to$ 

G: (a) ἐν φωτὶ ζωῆς υἰὸς βασιλέως, (b) οἱ δὲ προσδεκτοὶ αὐτῷ ὥσπερ νέφος ὄψιμον. "(a) The son of the king is in the light of life, (b) and those who are acceptable to him are like the late rain." Neither ideology nor the target language required the translator to violate the word order or insert the royal son into the picture. For M's יב, G's source text must have had באור (treated as sg because of the sg suffix of ורצונו). G treats and באור as a bound phrase, though the words are not conjoined. D. N. Freedman argues for the existence of the broken construct chain in Hebrew (1972). But it is difficult to tell if the G translators recognized (or imposed) this (doubtful) construction, since M's word order is overridden in other circumstances as well; see the comment on 11:21. S = M.

# 16:16 קנה M S (סגמא ) א קנות ( מראבי ) א קנה 16:16 קנות ( מראבי ) א קנה

טוב M S (ארא) ] גבחר G (αίρετώτεραι) (syn)

G: (a) אסססומן ססקומן מוֹפָדעֹדבּרָמו אַרְטָטוֹט, (b) אסססומן אוֹד פֿרָ אָרָאַדעָרָבָעָר אָרָאָדעָרָ גווּ אָרָאָדעָרָ גווּ אָרָאָדעָרָ גווּ אָרָאָדעָרָ אוֹדע אַרָאָרָאָרָ אַרָעוּרָ גערי. (a) Nests of wisdom are preferable to gold, (b) and nests of understanding are preferable to silver." In 16:16a, G's source text had הקנות, a normalization of M's rare infinitive construct form קנות G rendered קנות in both 16:16a and 16b as if it were קנות ("nests"), though קנות ("to acquire") was rendered correctly elsewhere. (In Gen 6:14 and in RH, the pl of קנין גוון פון אַרָאָשָרָע.) Probably "acquire" was avoided for theological reasons; see the comment on 4:4–5a.

The exclamatory מה is not represented in G 16:16a. Possibly it was missing in the source text and its presence in M was the result of dittography. However, since a similar מה is not reflected in G 30:13 either, it may have been omitted for the sake of the target-language syntax, as happened also in V, S, and T.

In 16:16a, where M has אוב , G has αἰρετώτεραι. Since αἰρετός and cognates do not translate אוב elsewhere, but often render words from גבח" G probably had נבחר in its source text, as in M 16:16b and 22:1a. Since the translator often chooses to *vary* words in parallelism, he is unlikely to have eliminated the variation here. The reason for the synonym change—is hard to determine, but M's form seems preferable rhetorically.

# 16:17

G: (a) τρίβοι ζωῆς ἐκκλίνουσιν ἀπὸ κακῶν, (b) μῆκος δὲ βίου ὁδοὶ δικαιοσύνης. (c) ὁ δεχόμενος παιδείαν ἐν ἀγαθοῖς ἔσται, (d) ὁ δὲ φυλάσσων ἐλέγχους σοφισθήσεται. (e) ὃς φυλάσσει τὰς ἑαυτοῦ ὁδούς, τηρεῖ τὴν ἑαυτοῦ ψυχήν. (f) ἀγαπῶν δὲ ζωὴν αὐτοῦ φείσεται στόματος αὐτοῦ. "(a) The paths of life turn away from evils, (b) and the ways of righteousness are length of living. (c) He who receives instruction will be in good (fortune), (d) and he who keeps reproofs will become wise. (e) He who keeps his own ways guards his own soul, (f) and he who loves his life will restrain his mouth." ("Receives instruction" means to absorb it, to take it to heart; "keeps reproofs" means to accept and follow them; "keep one's ways" means to stick to the right path, to avoid evildoing.)

G 16:17a  $\approx$  M 16:17a; G 16:17e  $\approx$  M 16:17b. Possibly G 16:17b is OG and 16:17a a revision. In any case, new stichoi, combining phraseology found elsewhere in Proverbs, are added to form three couplets: 16:17ab; 16:17cd; 16:17ef. It is unclear whether this happened in Hebrew or Greek. Both creative expansions and recombination of phrases can be found throughout wisdom literature and can be the work of a scribe, a translator, or a copyist in subsequent transmission.

16:19 עניים  $M^K$  ] עניים  $M^Q$  V (*mitibus*) T<sup>L</sup> (ענוונא); איניס S ( $M^Q$ : syn, equal; S: metath [in  $M^K$  form] ינ  $\leftarrow$  ני')

On the ketiv-qere, see the comment on 3:34.

S: (a) העביע בונא המעביע בווא, (b) = M. "(a) Better is one humble of spirit and humble of eyes." S read an obviously erroneous עינים for the *ketiv* עונים. (Possibly he simply mistook the latter for the former.) Then the second הביעי was added by a translator.

נבון 16:21 M S (שהבלא) א נבל (מהבלא) (phon  $t \to t$ , orth)

G: (a) τοὺς σοφοὺς καὶ συνετοὺς φαύλους καλοῦσιν, (b) οἱ δὲ γλυκεῖς ἐν λόγω πλείονα ἀκούσονται. "(a) They call the wise and intelligent base, (b) but those who are sweet in word shall hear more." The rendering

וּמוּסַר אָוִלִים אוֶּלֶת: 23 לֵב חֲכָם יַשְׂבֵּיל פֵּיהוּ וְעַל־שְׁפָתָׁיו יֹסִיף לֶקַח: 24 צוּף־דְּבַשׁ אִמְרֵי־גְׁעַם מָתִוֹק לַנָּפָשׁ וּמַרְפֵּא לְעֵצֶם: 25 יֵשׁ דֶרֶדְ יֶשָׁר לִפְנִי־אֵישׁ וְאַחֲרִיתָה דַּרְבֵי־מֶוֶת: 26 גָפָשׁ אָמֵל עֵמְלָה לֶוֹ

"wise and intelligent" takes each component of the phrase הכם לב as a word for wisdom. G's φαύλους must reflect גבל ("knave") for גבון ("intelligent"), because the resulting sentence is awkward and could have arisen only if constrained by a textual peculiarity. There is no graphic similarity between ל and J, and the interchange is not frequent, but see J. Kennedy 1928, 89. The interchange may be phonetic in origin, since the liquids have similar sounds. This is indicated by the lexical variants לשכה (2 Kgs 23:11; Ezek 40:17, 45, etc.) and נשכה (Neh 13:7; 3:30; 12:44). In 16:21b, G understands לקח as "taking in," that is, "hearing" (מצטיסטיבו).

# 16:22

G: (a) πηγὴ ζωῆς ἔννοια τοῖς κεκτημένοις, (b) παιδεία δὲ ἀφρόνων κακή. "(a) For those who have acquired it, understanding is a font of life, (b) but the education of fools is bad."

In ABP I say that דסוֹג אבעליו (dittography  $\rightarrow \rightarrow \rightarrow$ ), and this is possible. However, G's (and S's) treatment of prepositions is often flexible. In 16:22a, the antecedent of "its possessor" may seem to be "font" but is actually "understanding." G's rephrasing resolves the ambiguity. In fact, G chooses to paraphrase בעליו every time it occurs (1:19; 3:27; 16:22; 17:8), and indeed, in all these cases there is an ambiguity or unclarity to be resolved. In the present verse, the parallel construct pair, ומוסר אולים, supports M.

G eliminates the apparent tautology of M 16:16b ("the education of fools is folly") by substituting  $\varkappa \alpha \varkappa \eta$  for "folly." This means that it is worthless to try to educate fools; similarly G 14:24b. This idea is taught in G and M of 9:7–8 and some commentators find it in M 16:22 as well; see ABP.

S eliminates the tautology differently, by rendering אולת as אולת as אולת ("contempt"); compare S 14:24.

#### 16:24

G: (a) κηρία μέλιτος λόγοι καλοί, (b) γλύκασμα δὲ αὐτῶν ἴασις ψυχῆς. "(a) Beautiful words are honeycombs (b) and their sweetness is a heal-

ing of the soul." G 16:24b overrides M's word order, treats certain morphological features loosely, and renders מתוק לנפש ומרפא מתוק לנפש מרפא. Given the differences from M, it is hard to detect variants. On G's treatment of word order in Proverbs, see the comment on 11:21. לעצם was left without function and was not translated.

S restates للات as محمد ("wise man"), so that the verse praises an intellectual-moral quality (wisdom) rather than a neutral skill (pleasant speech). Pleasant speech, after all, can be misused; see the comment on 16:21, and compare the cautiousness exhibited about speech in 6:24 and 13:2.

#### 16:25

See the comment on the identical 14:12.

# 16:26 ה א פידו [ M פיהו ה למעד<br/>ס $\star$ G (למ $\tau$ על ד<br/>ע $d\pi \omega\lambda εומי) S (ל<math display="inline">\to$ ) (graph $\to$ )

G: (a) ἀνὴρ ἐν πόνοις πονεῖ ἑαυτῷ (b) καὶ ἐκβιάζεται ἑαυτοῦ τὴν ἀπώλειαν, (c) ὁ μέντοι σκολιὸς ἐπὶ τῷ ἑαυτοῦ στόματι φορεῖ τὴν ἀπώλειαν. (Divided differently from Rahlfs.) "(a) A man who toils [lit. "who is in toils"] toils for himself (b) and pushes away his destruction. (c) The perverse one, however, bears destruction on his own mouth." ἑαυτοῦ τὴν ἀπώλειαν = ידֹי (de Lagarde). Being unable to make good sense of τὴν ἀπώλειαν C κας τ κς τ κς τ τ΄, G chose to produce a proverb asserting the value of toil. This is reinforced by the moralizing addition in G 16:26c, which is an active rewriting of M 16:26b that shows awareness of both יפידו and יפידו Hebrew or Greek translation). The result is antithetical parallelism (Gerlemann 1956, 20), but there are too many other changes to indentify the creation of parallelism as the main motive.

בְּי־אָכַף עָלְיו בְּיהוּ: 27 אֵישׁ בֵּלִיַעַל בֹּרָה רָעֲה וְעַל־<sup>ר</sup>שִׂפְתָׁיו ּ בְּאֵשׁ אַרְבָת: 28 אֵישׁ תַּהְפָכוֹת יְשַׁלַּח מְדְוֹן וְנִרְגָּן מַפְרִיד אַלְוּף: 29 אֵישׁ חֶמָס יְפַתָּה רַעֲהוּ וְהוֹלִיכּוֹ בְּדֶרֶד לֹא־טְוֹב: 30 עָׁצֶה עֵינִיו לַחְשֵׁב תַּהְפָכֵוֹת לְרֵץ שְׁפָתָיו בִּלֶּה רָעֲה: 31 עֲטֵרֶת תִּפְאֶרֶת שִׁיבֶה בְּדֶרֶד אָדְלָה תִּמְצֵא: 32 טָוֹב אֶרֶד אַפַּיִם מִגַּבְּוֹר וּמֹשֵׁל בְּרוּחוֹ מִלֹבֵד עֵיר: 31 בְּחַיָּק יוּטַל אֶת־הַגּוֹרֶל וּמֵיְהוָה בָּל־מִשְׁפָּטְוֹ: 17:1 טָוֹב פַּת חֲרַבָה

16:27 שפתיו MK G (ד<br/>גשע למטדסט אַנואנאט V (labiis eius) T (ובספוותיה) ] ובספוותיה) א<br/>שָׁפָתו M<br/>Q (num)

ערבת ארבת ארבת ארבת א S (הארא א S (הבר $\star$  ) בר  $\star \approx \mathrm{G}$  (שאי $\star \approx \mathrm{G}$ 

The *ketiv*'s plural is certainly right. According to the common idiom, words are on a person's *lips*, not lip. In this case, the *ketiv* is an orthographic clarification of the שפתו, which was retained as the *qere* and should have been vocalized שָׁפָתוּ. In the preponderance of cases, it is the *qere* that inserts a *yod* to indicate a plural (see the list in Gordis 1971, 86–92), but there are exceptions, as in 21:29.

θησαυρίζει ("stores up") retroverts to צבר (Jäger) or a grammatically impossible צברת (cf. the correspondence in Zech 9:3) misconstrued as a masculine. The translator would have overlooked the final  $\pi$  in any case.

S has همجر ("his mouth") for "his lip." This is a synonym variant that could belong to the translator or a Hebrew scribe.

S: (a) = M; (b) הסגים היא הוא הוא הואס. "(b) and a worthless man persecutes friends." S has מרדך, a transposition of מפרד (to which the vowel letter ' was subsequently added).

16:30 כלה רעה M S (הארעה [ (הארעה 16:30 M כלה רעה G (πάντα τὰ κακά) (div) fin ] + כר הרעה הוא G (οὖτος κάμινός ἐστιν κακίας) (transfer from 16:27a, elab)

G: (a) στηρίζων ὀφθαλμοὺς αὐτοῦ λογίζεται διεστραμμένα, (b) ὀρίζει [G<sup>BS</sup>; var: ὀργίζει G<sup>A</sup>; ἐπιδάχνων G<sup>MSS</sup>] δὲ τοῖς χείλεσιν αὐτοῦ πάντα τὰ κακά, (c) οῦτος κάμινός ἐστιν κακίας. "(a) Fixing his eyes, he plans crooked things, (b) and he marks out [vars: "enrages"; "bites"] with his lips all the evils. (c) This one is a furnace of evil." The textual situation of G in this verse is complex, but the text in G<sup>BS</sup> is best, with ὀρίζει representing basically the same understanding of Υ¬γ as in 6:13, where it is translated ἐννεύει, which means to mark out, hint at, or the like. ἐπιδάχνων (preferred by de Lagarde and Baumgartner), attested in minuscules and some secondary translations, also goes back to the Hebrew text but understands γ¬γ as "pinch," hence—since the action is done with the lips/mouth—"bites." p means "pinch" in Job 33:6, but this sense is not recognized elsewhere in G.

 $\pi \acute{a} \varkappa \varkappa \varkappa \varkappa \acute{a} = ect$ קר הָרָשָה, for M's פָּל הָרָשָה ("completes evil"). G handles similar phraseology quite differently in 6:13–14. 16:30c is an addition based on 16:27a which says that the knave ברה רעה ("mines evil"). Transferred to the present verse, the phrase was misdivided as מר הרעה מו understood as כָּר הָרעָה לויצָר הָרעָה הוא פָר הַרעָה הוא . The division error shows that the transfer took place in Hebrew, not Greek. The source text read, approximately, הוא איז הוא ברעה הוא . The added stich enhances the description of the lawless man by transferring to him one of the characteristics of another bad type, the man of Beliyya'al.

#### 16:32

G = M, but using "anger" ( $\delta \gamma \gamma \gamma \varsigma$ ) to render רוח for greater specificity.

#### 16:33

G: (a) εἰς κόλπους ἐπέρχεται πάντα τοῖς ἀδίκοις, (b) παρὰ δὲ κυρίου πάντα τὰ δίκαια. "(a) Everything comes into the bosom for the unjust [pl], (b) and from the Lord are all just things." So as not to give credibility to lot casting (which was a common way of inquiring of gods among foreign peoples but not in Hellenistic Judaism), the translator transforms the verse into a new proverb. See the comment on 18:18.

וְשַׁלְוָה־בֶהּ מִבַּׁיִת מְלֵא זִבְחֵי־רִיב: 2 עֶבֶד־מַשְׂבִּיל יִמְשׁל בְּבֵן מֵבֵּישׁ וּבְתִוֹדְ אַׁחִים יַחַלְק נַחַלֶּה: 3 מַצְרֵף לַכָּסֶף וְכַוּר לַזָּהֶב וּבֹחֵן לִבּוֹת יְהוֶה: 4 מֵרַע מַקְשֵׁיב עַל־שְׁפַת־אֶוֶן שֶׁקֶר מֵׁזִין עַל־לְשִׁוֹן הַוֹת: 5 לֹעֵג לְרָשׁ חֵרֵף עֹשֵׁהוּ שָׁמֵח לְאֵיד לָא יִנְקָה: 6 עֲטֶרֶת זֲקַנִים בְּנֵי בְנֵים וְתִפְאֶרֶת בְּנֵים אֲבוֹתֶם: 7 לֹא־נָאוֶה לְנָבֵל שְׁפַת־יֶתֶר אַׁף כִּי־לְנָדִיב

17:1

G: (a) κρείσσων ψωμὸς μεθ' ἡδονῆς ἐν εἰρήνῃ (b) ἢ οἶκος πλήρῃς [var: πλήρῃς > G<sup>B</sup>] πολλῶν ἀγαθῶν καὶ ἀδίκων θυμάτων μετὰ μάχῃς. "(a) Better a morsel with pleasure in peace (b) than a house full of many good things and unrighteous sacrifices with contention." This is an expansive translation, showing the influence of μετὰ ἀδικίας in G 15:29*a* (M 16:8). μεθ' ἡδονῆς does not render חרבה (or אערבה , as Jäger suggests). Rather, μεθ' ἡδονῆς does not render שלוה בה is ignored while שלוה בה is ignored while שלוה בה is translated twice. The translator may intend to emphasize the goodness of the experience of eating in peace, though he is actually missing the point of the ratio ("Better A [something not so good] with B [something very good] than A' [something better than A] with B' [something much worse than B].") In 17:1b, G details the ratio by separating A' and B'. See the comment on 15:17 in ABP. V sicca ("dry") =  $\mu \epsilon \theta$ ' ἡδονῆς considered as representing a.

17:2

For M's  $\Box$  ("a disappointing son"), G uses δεσποτῶν ἀφρόνων ("foolish masters") to intensify the assertion of wisdom's power. Whereas M says that the wise servant will rule over the foolish son (but not over the other sons), G has him lording it over his own master; see Giese 1992b, 408. (The plural "masters" is for generality.)

17:3 אובחר [ M ] ובחר ( לאגאנאלא M ] ובחר ( לאנאנאנא) (syn, Aram, equal)

G: (a) ὥσπερ δοχιμάζεται ἐν χαμίνω ἄργυρος καὶ χρυσός, (b) οὕτως ἐχλεχταὶ καρδίαι παρὰ χυρίω. "(a) As silver and gold are tested in a furnace, (b) so are hearts chosen by the Lord." G supplies coordinating adverbs as it often does; e.g., 21:1; 25:14, 19, 20; 26:14. For M's [LCT] ("tests"), G has ἐχλεχταί, which reflects LCTC used in Prov 10:20; Sir 4:27; Isa 48:10). G gave LCTC is standard Hebrew sense of "choose."

(To be sure, אֶבֶן בֹחַן is translated λίθον πολυτελη ἐκλεκτόν in Isa 28:16, but there the word is difficult and the notion of "choice" was probably prompted by context.) ובחר could be original or have arisen in proto-G transmission.

S's ובחר or ובחן = בעני.

# 17:4

G: (a) κακὸς ὑπακούει γλώσσης παρανόμων, (b) δίκαιος δὲ οὐ προσέχει χείλεσιν ψευδέσιν. "(a) An evil man obeys the tongue of evildoers, (b) but a righteous man does not hearken to deceitful lips." G 17:4b is a converse translation of M's "deceit (= a deceitful man) hearkens to a deceitful tongue," creating an antithesis between the stichoi. It transposes "tongue" and "lips," apparently indifferent to the placement of these pragmatic synonyms.

S = G.

# 17:5

G: (a) (b) = M; (c) ό δὲ ἐπισπλαγχνιζόμενος ἐλεηθήσεται. "(c) But he who shows compassion will receive mercy." G adds 17:5c to provide an antithesis to 17:5a and emphasizes the importance of compassion (see BAP 122–23).

17:6 fin ] + 2 stichoi G (17:6*a*) (elab)

# 17:6*a*

G: (a) τοῦ πιστοῦ ὅλος ὁ κόσμος τῶν χρημάτων, (b) τοῦ δὲ ἀπίστου οὐδὲ ὀβολός. "(a) To the trustworthy man (belongs) the entire world of possessions, (b) but to the unfaithful one, not a penny." This is an independent saying that connects to the following by the theme word πιστός. Most MSS add this to 17:4, but G<sup>B</sup> appends it to verse 6. Its syntax suggests Greek origin. The notion of the spiritual riches of the sage is a commonplace of Greek rhetoric; see BAP.

17:7

G: (a) οὐχ ἀρμόσει ἀφρονι χείλη πιστὰ (b) οὐδὲ δικαίω χείλη ψευδῆ. "(a) Trustworthy lips do not befit a fool, (b) nor deceitful lips the righteous man." M has "Excessive speech is not fitting for a scoundrel; how much less so false speech for a noble!" This is puzzling, because excessive speech does not befit *anyone*. BHS seeks to improve M 17:7a by reading 'ψ' for 'π', but a  $\mathcal{U} \rightarrow \pi$  change would be hard to explain.

שְׁפַת־שֵׁקֶר: 8 אֶבֶז־תֵז הֲשׁׁחֵד בְּעֵינֵי בְּעָלֵיו אֶל־כְּל־אֲשֶׁר יִפְגָה יַשְׂבְּיל: 9 מְכַסֶּה־פֶּשַׁע מְבַקֵּשׁ אַהֲבֶה וְשׁׁגָה בְׁדָבָר מַפְרִיד אַלְוּף: 10 תַּחַת גְּעָרָה בְמֵבֵין מֵהַכְּוֹת כְּסֵיל מֵאֶה: 11 אַדְ־מְרִי יְבַקֶּשׁ־רֶע וּמַלְאָדְ אַכְזָרִי יְשָׁלַּח־בְּוֹ: 12 פָּגווש דַּב שַׁכּוּל בְּאֵישׁ וְאַל־בְׁסִיל

G's "trustworthy lips" was probably a deliberate change from "excessive speech." G makes the proverb more logical by changing "excessive" to "trustworthy" (on the basis of the antithesis). Trustworthy lips may be said not to befit a fool in the sense that he does not *have* them, just as a righteous man does not have deceitful ones.

S ארחיבנא ("faithful") = G.

17:8

G: (a) μισθὸς χαρίτων ἡ παιδεία τοῖς χρωμένοις, (b) οὖ δ' ἀν ἐπιστρέψῃ, εὐοδωθήσεται. "(a) Instruction is a reward of grace for those who use (it), (b) and wherever it turns, it succeeds [or, 'and wherever he turns he succeeds']." (By the latter translation, the antecedent of "he" is "those who use it," in spite of the difference in number.) G backs away from M's praise of bribes. The G translators generally understood ¬mu as a gift and rendered it δῶρον. G-Proverbs does not condemn gift giving altogether (see 18:16; 21:14), but it does reject bribes in 15:27 and 19:24. On see the comment on 16:22.

V: (a) gemma gratissima expectatio praestolantis (b) quocumque se verterit prudenter intellegit. "(a) The expectation of him who waits is a lovely jewel, (b) wherever he turns, he observes wisely." V likewise avoids praise of bribes, though taking a different tack from G.

Although S does not avoid affirming gift giving consistently, he may be shying away from it in this case. In 21:24, he uses מהמכולא, a less weighted term, saving געניד for illegitimate bribes (6:35; 15:27; 17:23). In the present verse S avoids praise of bribes by translating with , ממכולא מהלא, מגלא, מגלא, אין ("is lovely") and אישריל with , ממכולא ("is intelligence").

17:9

G's אושנה ("hates") misconstrues ושנה. G possibly read ושנא. The latter is found in a Hebrew manuscript (see BHQ). However, both G and a later Hebrew scribe might have used "hates" because it is hatred

that could be expected to divide friends. (The Hebrew proverb actually condemns gossip.) φίλους και οἰκείους ("friends and kin") is a double translation of אלוף ("friend").

S: (a) = M, (b) = G. Read גרבסי ארשי א אראיז (Pinkuss). (Pinkuss).

#### 17:10

G: (a) συντρίβει ἀπειλὴ καρδίαν φρονίμου, (b) ἄφρων δὲ μαστιγωθεὶς οὐκ αἰσθάνεται. "(a) A threat breaks the heart of a prudent man, (b) but a fool, (even when) beaten, perceives nothing." M's חֵתַת is a G-stem imperfect from נח"ת ליכחים down," hence "impact." This word gave the translators difficulty. G parses תחת as a causative H-stem (תְּחֵת), as if from גע"ר. תח"ת is rendered ἀπειλή also in 13:8; Isa 50:2; 54:9. G's "perceives nothing" (or "does not take notice") is a contextual guess for 𝔅, which became obscure once 𝔅 𝔅 𝔅

V proficit ("benefits") and  $\alpha' \pi \lambda \tilde{\eta} \xi_{1\zeta}$  ("a blow") are contextual guesses.  $\theta' \eta \xi_{\ell}$  ("will reach") and T  $\psi \eta \xi_{\ell}$  ("enters") are basically correct.

17:12

G: (a) ἐμπεσεῖται μέριμνα ἀνδρὶ νοήμονι, (b) οἱ δὲ ἀφρονες διαλογιοῦνται κακά. "(a) Cares will befall an intelligent man, (b) but fools are preoccupied with evils." G's point is that although the intelligent man is not immune to worries, the fool is constantly in anxiety about pending misfortune. To express this idea, G uses wordplays: דאבה ("worry") ראבה ("worry") and שכול (properly: "bereft") with שכול ("be intelligent"). It is clear that the wordplays are deliberate and homiletically motivated, because TT is unproblematic and, outside of Proverbs, consistently translated ἄρχος ("bear"). In Prov 28:15, G identifies דב with Aramaic אר and translates λύχος ("wolf"), though T means "bear" in Aramaic also. To enable his homily, G also ignores the Hebrew word order in ἀνδρὶ νοήμονι (see comment on 11:21). ואל בסיל was understood as elliptical for "and cares fall upon (ואָל) a fool." גאאל is loosely based on M באולתו; cf. 13:6. The result is an idea out of line with M-Proverbs' belief that wisdom brings security and confidence (1:33; 3:23; 10:9; 28:1). In 17:10, too, G concedes the occasional vulnerability of the wise. The sentiment of G 17:12a is closer to that of Qoh 1:18. The idea of the evildoer living in anxiety may בְּאַוּלְתְּוֹ: 13 מֵשִׁיב רֻעָה תַּחַת טוֹבֶה לא־'תָמָוּשׁ' דְׁעָה מִבֵּיתְוֹ: 14 פּוֹטֵר מֵיִם רֵאשִׁית מְדֵוֹן וְלִפְנֵי הִתְגַּלַּע הָרָיב נְטְוֹשׁ: 15 מַצְהַיק רֶשָּׁע וּמַרְשֵׁיעַ צַדֵּיק תּוֹעַבַת יְהוָה גַּם־שְׁנֵיהֶם: 16 לָמָה־זֶה מְחֵיר בְּיַד־בְּסֵיל לִקְגוֹת חָכְמֵה וְלֶב־אֶיִן: 17 בְּכָל-עֵת אֹהֵב הָרֵעַ וְאָח

have been imported from Prov 17:13. Given these uncertainties, further retroversion is not feasible.

17:13 תמיש [ (תפסוק) א M<sup>Q</sup> G (אבנו) S (אבנו) V (recedet) T<sup>L</sup> (תפסוק) M<sup>K</sup> (graph  $1 \rightarrow )$ 

The intransitive *qere* is required by context.

17:14 מים M ] אלים (געליסנ;); דמים אדמים (G: near dittog/haplog ' $\leftrightarrow$ ', equal; S: near dittog  $\rightarrow \rightarrow$  רד

M reads, literally, "Releasing water—the start of a quarrel, so before a quarrel breaks out, leave off!" In other words, provoking a quarrel is like opening a sluice gate. The flow begins as a trickle but quickly surges out of control; see ABP.

G: (a) έξουσίαν δίδωσιν λόγοις άρχη δικαιοσύνης, (b) προηγεῖται δὲ τῆς ένδείας στάσις και μάχη. "(a) Righteous rule gives authority to words, (b) but sedition and strife go before poverty." G's translation is a guess at a difficult verse, connecting to M only in miscellaneous words. (G does not know the meaning of התגלע but renders it differently in each occurrence [17:14; 18:1; 20:3; see de Waard 2006, 266-68]. In fact, modern lexicographers also guess its meaning.) Nevertheless, it is possible to retrovert גאלים, where M has מים. The validity of G's reading is supported by the idiom in Ps 22:8b, יָפָטִירוּ בְּשָׂפָה ("they let loose with the lip"), i.e., "shoot off their mouths" in insult. The graphic mechanism of the dittography or haplography  $\leftrightarrow \dot{}$  is unclear, but it is also unclear why the translator would make the change. G's source text should be translated, "Releasing words starts a quarrel, so before a quarrel breaks out, leave off." That is to say, uncontrolled speech provokes conflict-a teaching found often in wisdom literature, such as in Prov 10:19 and 17:27. Both M and G's reconstructed source text are valid proverbs. In

fact, G's rendering, too, though not in my view correct, creates a valid proverb, because proverbs can emerge in the act of translating.

V construes the proverb as a warning against imprudence, translating 17:14b as *et antequam patiatur contumeliam iudicium deserit* ("and before he suffers insult he abandons judgment").

#### 17:15

G: (a) ὅς δίκαιον κρίνει τὸν ἄδικον, ἄδικον δὲ τὸν δίκαιον, (b) ἀκάθαρτος καὶ βδελυκτὸς παρὰ θεῷ. "(a) He who judges the unjust as just, and the just as unjust, (b) is impure and disgusting before God." G has a double translation of תועבת, probably for emphasis but perhaps also for the sake of quantitative balance (Gerleman 1956, 25). Since G speaks of only one person in 17:15a, it ignores שניהם in 17:15b.

17:16 fin ] + 2 stichoi G (17:16*a*) (elab)

## 17:16a

G: (a) ὅς ὑψηλὸν ποιεῖ τὸν ἑαυτοῦ οἶκον, ζητεῖ συντριβήν. (b) ὁ δὲ σκολιάζων τοῦ μαθεῖν ἐμπεσεῖται εἰς κακά. "(a) He who raises high his own house seeks disaster, (b) and he who turns aside from learning will fall into evils." G 17:16*a*a ≈ M 17:19b. G's raising one's house is a clearer image of haughtiness than M's idiom in 17:19b of raising the door (פתח). G 17:16*a*b represents and paraphrases a few words taken from M 17:20, namely, יפול ברעה ("will fall into evil"). G 17:16*a* probably originated as a Hebrew verse composed (like many sayings in M-Proverbs itself) from components of other verses. G 17:16*a* can be retroverted approximately to שבר ועקש ללמוד יפל ברעה מגביה ביתו מבקש in the HB) foreign to BH usage and was composed by a later scribe. Some MSS of V include 17:16*a* in 17:16.

#### 17:17

G: (a) εἰς πάντα καιρὸν φίλος ὑπαρχέτω σοι, (b) ἀδελφοὶ δὲ ἐν ἀνάγκαις χρήσιμοι ἔστωσαν. (c) τούτου γὰρ χάριν γεννῶνται. "(a) Let there be to you a friend for every occasion, (b) and let (your) brothers be useful in

ּלְצָרָה יִוָּלֵד: 18 אָדָם חָסַר־לֵב תּוֹחֵַע בֶּף עֹרֵב עְׁרָבָּה לִפְגֵי רֵעֵהוּ: 19 אַהֵב פֶּשַׁע אֹהֵב מַצֶּה מַגְבִּיהַ פִּתְחוֹ מְבַפֶּשׁ־שֵׁבֶר: 20 עִפֶּשׁ־לֵב לָּא יִמְצָא־מֶוֹב וְנָהְפָּדְ בִּלְשׁוֹנוֹ יִפְּוֹל בְּרָעֵה: 21 יֹלֵד בֵּסִיל לְתַוּגָה לְוֹ וְלָא־יִשְׁמַח אֲבִי נְבָל: 22 לֵב שָׂמַח יֵיטֵב גֵּהֶה וְרָוּח נְבאָה תְּיַבָּשׁ־ גֶּרֶם: 23 שִׁׁחִד <sup>ו</sup>ְבְּחֵיק<sup>י</sup> רְשָׁע יְקֶח לְהַטּוֹת אָרְחָוֹת מִשְׁפֵּט: גַּרֶם: בַּזַ שֶׁחַד <sup>ו</sup>ְבָחֵיק<sup>י</sup> רְשָׁע יִקֶּח לְהַטּוֹת אָרְחָוֹת מִשְׁפֵּט: 42 אֶת־פְּגַי מֵבִין חָכְמֶה וְעֵינֵי כְׁסִיל בִּקְצַה־אֶרָץ: 25 בַּעַס לֵאָבִיו בֵּן

tribulations, (c) because for this were they born." G turns M's statement into advice. Stichoi 17:17bc recapitulate M 17:17b expansively, lest one get the grim impression that a brother being "born" for trouble means that he is destined to suffer it.

# 17:18

G: (a) ἀνὴρ ἄφρων ἐπικροτεῖ καὶ ἐπιχαίρει ἑαυτῷ (b) ὡς καὶ ὁ ἐγγυώμενος ἐγγύῃ τὸν ἑαυτοῦ φίλον. "(a) A foolish man applauds and takes pleasure in himself, (b) just like one who pledges himself by (giving) surety for his friend." תוקע כף ("strikes the hand") is translated twice in 17:18a and is understood as an expression of joy, as in Ps 47[46]:2, where the phrase is rendered κροτήσατε χεῖρας ("clap hands"). (G misunderstands the idiom in Prov 11:15 as well.)

G is unbalanced, comparing the fool's smug pleasure in himself with the very different happiness one gets when providing a friend with financial help. This awkwardness results from the translator's assumption that the guarantor is offering surety *for* his friend, as in G 6:1. (What is actually happening is that a loan guarantor shakes [lit. "strikes" or "clasps"] the lender's hand to seal an agreement to provide surety *on behalf of* a third party, the borrower [ABP].) The lender in this situation is the guarantor's  $\mathcal{V}\mathcal{T}$  ("neighbor," "acquaintance.") G assumes that providing surety for a friend is a worthy deed; thus too in Sir 29:14–20. (Both G and Ben Sira may be reflecting an awareness of the necessity of loan guarantees in a more commercialized economy.) At the same time, G 6:1 (like Sir 8:13; 29:20) acknowledges the risks involved in doing so.

#### 17:19-21

G shapes these verses into three couplets, whose pairing is marked by the particle δέ (BAP): 17:19–20a, 20b–21a, and 21b–21c.

17:19

G 17:19 = M 17:19a. M 17:19b is lacking in G here but appears in different form in G 17:16aa; see the comment there. The cause of the dislocation is uncertain.

17:20 עקש M G (סגאאססאלאסט א G (ד. אישק (metath  $\psi \rightarrow \varphi)$  (metath  $\psi \rightarrow \varphi$ )

G = M, but the latter also appears partially in G 17:16*a*b. Though σκληροκάρδιος is not the standard translation of עקש לב, it is a reasonable approximation, and it is not clear that the Hebrew should be retroverted to קשה לב (as BHQ does). In any case, what BHQ calls Sa'adia's "identical translation" does not witness to קשה, since his comment on the verse presupposes M.

S: אבים גרס ("he whose heart is oppressed"). S had שַשָּק לב for M's עקש לב S says that "the man whose heart is oppressed will not find good." This pessimistic statement arose by a metathesis that was certainly accidental, not exegetical.

17:21 fin ] + 1 stichos G (21c) (antithesis)

G: (a) καρδία δὲ ἄφρονος ὀδύνη τῷ κεκτημένῷ αὐτήν. (b) οὐκ εὐφραίνεται πατὴρ ἐπὶ υἰῷ ἀπαιδεύτῷ, (c) υἰὸς δὲ φρόνιμος εὐφραίνει μητέρα αὐτοῦ. "(a) and the heart of the fool is a misery to its possessor. (b) A father does not rejoice over an uneducated son, (c) but a prudent son makes his mother rejoice."

G's additional 17:21c is a converse translation of 10:1b. It provides an antithesis to 17:21b and with it forms a couplet, which disappeared when G paired 17:21a with 20b.

# 17:23 מַחֵיק G (כא גיאָא ה) מַחֵיק M (graph מַחֵיק מ ה ב $\to b$

G: (a) λαμβάνοντος δῶρα ἐν κόλπῷ ἀδίκως οὐ κατευοδοῦνται ὁδοί, (b) ἀσεβὴς δὲ ἐκκλίνει ὁδοὺς δικαιοσύνης. "(a) The ways of him who unjustly takes gifts in the bosom do not prosper, (b) and a wicked man perverts the ways of righteousness." ἐν κόλπῷ = ¬𝒴. M's 𝒴 ("evildoer") has a double translation: first used adverbially, ἀδίκως, then as the subject of 17:23b, ἀσεβὴς. (S does not translate the word.)

# 17:24

S's בארשויה ("in the depths [of the earth]," for M's בארשויה) alludes to the underworld and hints that this is the fool's (early) destination.

ַכְּסֵיל וּּמֶׁמֶר לְיוֹלַדְתְּוֹ: 26 גָּם עֲנִוֹשׁ לַצַּדְּיק לאֹ־טֵוֹב לְהַכְּוֹת נְדִיבִים עַל־יְשֶׁר: 27 חוֹשֵׂדְ אֲמָרָיו יוֹדֵעַ דֶּעַת וְקַר־רוּחַ אֵישׁ תְּבוּנְה: 28 גָס אֶוִיל מֲחַרִישׁ חָבֶם יֵחָשֵׁב אֹטֵם שְׂפָתָיו נְבְוֹן: 18:1 לְתַאֲוָה יְבַקֵּשׁ נְפְרֵד בְּכָל־תוּשִׁיָה יִתְגַלֶּע: 2 לְאֹ־יַחְפָּץ בֵּסִיל בִּתְבוּנֶה כִּי אֶם־

#### 17:26

G: (a) ζημιοῦν ἄνδρα δίκαιον οὐ καλόν, (b) οὐδὲ ὅσιον ἐπιβουλεύειν δυνάσταις δικαίοις. "(a) To punish a righteous man is not good, (b) nor is it pious to scheme against righteous princes." G's ὅσιον—a restatement of שוב ("good")—makes disloyalty to princes a religious offense. G explains סוב ("smiting") as "scheming," which is the more usual way of harming princes. G loosely renders," which is the more usual way of harming princes. G loosely renders שול ישר without attention to syntax. In a similar vein V has principem qui recta iudicat ("the prince who judges right"). S's הבי אה ביא ("righteous men who speak the truth") is more accurate; see ABP.

וקר 17:27 און M<sup>K</sup>  $\approx$  G (μακρόθυμος δέ)  $\approx$  S (סגעבויא רוחיה)  $\approx$  T (ודמכיכיא רוחיה) ]  $M^{Q}$  V (pretiosi) (graph  $1 \rightarrow i$ )

The *ketiv* וקר רוח ("cool of spirit"), rather than the *qere* וקר רוח יקר ("precious of spirit"), is earlier and preferable. קר רוח קר recalls the Egyptian idiom "the cool man," meaning the man who is characterized by repose (see ABP). קר רוח is supported by the doublet in 14:29, with קר רוח ("patient" or "long-suffering"). The *qere*'s "precious of spirit" is a graphic change prompted by an unfamiliarity with the idiom "cool of spirit." "Precious spirit" does not occur elsewhere, but קרה נפש יקרה רוחיה 6:26. T's המכיריא רוחיה ("and he whose spirit is humble") is an interpretation of the unparalleled idiom וקר רוח than a translation of the easier יקר רוח נפש יקרה.

G: (a) δς φείδεται ἡῆμα προέσθαι σχληρόν, ἐπιγνώμων, (b) μαχρόθυμος δὲ ἀνὴρ φρόνιμος. "(a) He who refrains from uttering a harsh word, is intelligent, (b) and the patient man is prudent." G adds "harsh" to explain what kind of words should be curbed.

#### 17:28

G: (a) ἀνοήτῷ ἐπερωτήσαντι σοφίαν σοφία λογισθήσεται, (b) ἐνεὸν δέ τις ἑαυτὸν ποιήσας δόξει φρόνιμος εἶναι. "(a) Wisdom will be ascribed to the

unintelligent man who consults wisdom, (b) and whoever makes himself mute will be thought to be prudent." The translator understood מחריש ("be silent") in the extended sense of "be quiet and pay attention" (as in M-Isa 41:1 [ $\neq$  G]), hence "consults." הכם is used twice:  $\sigma o \phi i \alpha \nu$  and  $\sigma o \phi i \alpha$ . This is not exactly a double translation, since the words do not fill the same syntactic slot. Rather, once the translator understood מחריש as implying attention, hence consultation, he had to explain just what the object of consultation was. The image "shutting the lips" is (rather unnecessarily) demetaphorized.

18:1 לתאוה M S ( $t \to t$  ( $t \to t$ ) אלתאנה ( $t \to t$  א א G ( $\pi po\phi \acute{a}\sigma eis$ ) (graph  $t \to t$ 

G: (a) προφάσεις ζητεϊ ἀνὴρ βουλόμενος χωρίζεσθαι ἀπὸ φίλων, (b) ἐν παντὶ δὲ καιρῷ ἐπονείδιστος ἔσται. "(a) A man who wishes to separate himself from friends seeks pretexts, (b) and at all times he will be disgraced." προφάσεις = ܪ̈́תאֵנָה (Cappelli, ref de Lagarde). This correspondence is unique but cannot be excluded, because πρόφασις occurs only four times in translated texts. The error could have arisen by 1 → 1 confusion, such as mentioned by Kennedy (1928, 68–69). The rest of the verse is a reasonable attempt to render a difficult text. βουλόμενος is distant from times in placement and meaning and is not a doublet (contra BHQ). V = G.

# 18:2

G: (a) οὐ χρείαν ἔχει σοφίας ἐνδεὴς φρενῶν. (b) μᾶλλον γὰρ ἄγεται ἀφροσύνῃ. "(a) He who lacks good sense has no use for wisdom, (b) because he is instead led by folly." G 18:2b departs considerably from M. G's ἄγεται associates התגלות with התגלו" "go into exile," as in, e.g., Amos 7:11, 17; Isa 23:1 (erroneously, as in the present verse); and Jer 40[47]:1.

S: جمامی معلی الدی محللہ اللہ الدی (a) The fool does not desire wisdom, (b) because his heart dwells on folly." S uses some of G's phrasing but continues the theme of the silent fool from 17:27–18:1, in spite of a Hebrew text that says almost the contrary.

בְּהִתְגַּלְּוֹת לִבְּוֹ: 3 בְּבוֹא־רֲשָׁע בְּא גַם־בְּוּז וְעִם־קָלָוֹן חֶרְפָּה: 4 מַיִם אַנַמָקִים דִּבְרֵי פִּי־אֵישׁ נַחַל נְבֹעַ מְקוֹר חָכְמָה: 5 שְׂאֵת פְּגִי־רָשָׁע לֹא־טֶוֹב לְהַטּוֹת צַׁדִּיק בַּמִּשְׁפֶּט: 6 שִׁפְתַי רֵסִיל יָבַאוּ בְרֵיב וּפִיו לְמַהַלֵּמְוֹת יִקְרֶא: 7 פִּי־רֵסִיל מְחִתָּה־לֵוֹ וּשְׁפָתָיו מוֹקֵשׁ נַפְשְׁוֹ: 8 דִּבְרֵי גָרְגָן כְּמְתְלַהַמֵים וְהֵם יְרְדָוּ חַדְרֵי־בֶטָן: 9 גַם מִתְרַפָּה

18:3

G: (a) ὅταν ἔλθῃ ἀσεβὴς εἰς βάθος κακῶν, καταφρονεῖ, (b) ἐπέρχεται δὲ αὐτῷ ἀτιμία καὶ ὅνειδος. "(a) When a wicked man comes to the depth of evils he shows scorn, (b) and dishonor and disgrace come upon him." Jäger suggests that G's "comes to the depth of evils" paraphrases  $\square$  (dittog of  $\aleph \square$ ), lit. "in a pool," hence "to the depths"; then "of evils" was added to explain the apparent metaphor. But the semantic distance between "pool" and "depths" is great. It is more likely that G did not understand that  $\square$  is the subject of "comes" and found it necessary to add the phrase to give the movement of "coming" an appropriate terminus.

S: גאמעעד בארס, אראי אראי ארס, אראי אראי אראס, אראי אראס, אראי אראס, אראי אראס, אראי אראס, אראעט, אראעע, אראע, א

# 18:4 דברי M S (حتل (λόγος) (reform)

שלב [ (במכת K S (ג (במכת K S (ג (מרכת C ( באר שי)

חכמה M S (געבדא א) ] הרמה\* G (געבא א) מאיים [ (געבדא א)

G: (a) ὕδωρ βαθὺ λόγος ἐν καρδία ἀνδρός, (b) ποταμὸς δὲ ἀναπηδύει καὶ πηγὴ ζωῆς. "(a) A word is deep water in the heart of a man, (b) a welling river and a fount of life."

G's differences from M in this verse may have arisen in Greek or in the Hebrew source text. For ש, G has לא אמאלג, a choice that BHQ explains as assimilation to 20:5. For M's "fount of wisdom" G has  $\pi\eta\eta\eta$  ζωής. This may be an assimilation to the more common מקור חיים (10:11;

13:14; 14:27). "Assimilation," it should be noted, is not in itself a reason to ascribe a difference to the translation process and reject its status as a variant. Assimilation is also a means of creating new proverbs from existing sayings. If there was an underlying Hebrew in the present verse, it was מים עמקים דבר בלב איש, נחל נבע מקור חיים.

Several medieval MSS KR have רחיים, which almost certainly is an adjustment, probably accidental, to the expected idiom. Whatever the motivations for the changes, both variants form valid proverbs.

18:5

G has tò δίκαιον where M has צדיק. G several times translates a person type as an abstraction, e.g., 5:5; 14:10; 15:6; 18:5; 23:28.

# 18:6 (מייתן) א יבאו M S (ביאן [ (בא: G (מאטטטע) T (מייתן) (gram)

למהלמות [ M למהלמות (ליס לא למות (ליס לא למות המה למות א ההמה למות (G: div, scrambling ההמה  $\rightarrow$  לההמה למה (למ $\sim$ למ למ

G: (a) χείλη ἄφρονος ἄγουσιν αὐτὸν εἰς κακά, (b) τὸ δὲ στόμα αὐτοῦ τὸ θρασῦ θάνατον ἐπικαλεῖται. "(a) The lips of a fool lead (him) to evils, (b) and his rash mouth summons death." G had יביאו (or understood the verb as יביאו). The direct object "him" was supplied from context. In 18:6b, τὸ θρασῦ θάνατον = הָהמָה לְמָוֶת (lit. "which murmurs for death"). הומיה ז translated θρασεῖα in 9:13. G's consonants, ההמהלמות , a redivision and scrambling of M's consonants, למהלמות the cause was the repeated consonant sequences, but exact steps of the process cannot be determined.

S: אשבאא גע להאאי איז פאראי איז פאראי איז פאראי גע גע גע גע גע גע (a) The lips of the fool enter into litigation, (b) and his mouth brings him to death." S's "to death" reflects למות, which partially coincides with G's reading. S's reading must have arisen by homoio in הלמ, skipping over הלמ

T's (ליה) מייתן ("bring [him]") takes באו as causative. This is one of the rare cases in which T agrees with G against S and M. T's rendering is surprising, because he must be using a pointed text. But the agreement with G is coincidental, as shown by the fact that T and G supply different words as complements, eic xxxx and xxxx." ("judgment," "conflict").

18:8 עצלה תפיל תרדמה (במתלהמים נרגן כמתלהמים M  $\approx$  S ] אונפש רמיה עצלה תפיל תרעב למעלה תפיל הרעב המיה הרעב המיה הנע<br/>מ $\pi$ נועלסטסטסט (transf from M 19:15, unc)

G: (a) ὀκνηρούς καταβάλλει φόβος, (b) ψυχαὶ δὲ ἀνδρογύνων πεινάσουσιν.

בְמְלַאְכְתֵּוֹ אֶח הוּא לְבַעַל מַשְׁחִית: 10 מִגְדַל־עָׂז שֵׁם יְהוֶה בְּוֹ־יָרָוּץ צַדְּיַק וְנִשְׁגֵב: 11 הַוֹן עֲשִׁיר קִרְיַת עֻזְוֹ וּכְחוֹמָה נְשְׁגָבָה בְּמַשְׂבִּיתְוֹ: 12 לִפְנֵי־שֶׁבֶר יִגְבַּה לֵב־אֵישׁ וְלִפְנֵי כְבַוֹד עַנְוֶה: 13 מַשֵּׁיב דְּבָר בְּטֶרֶם יִשְׁמֶע אָוֶלֶת הִיא־לוֹ וּכְלִמֶה: 14 רְוּחַ־אֵישׁ יְכַלְבֵּל מַחֲלֵהוּ וְרְוּחַ נְכַאָה מֵי יִשְׁאֶנָה: 15 לֵב גָבוֹן יִקְנָה־דֶעַת וְאָזֶן חְׁכָמִים הְּבַקָּשׁ־ דְעַתֵּ: 16 מַתְּן אֶדָם יַרְחֵיב לֵוֹ וְלִפְנֵי גְדֹלֵים יַנְחֶנָוּ: 17 צַדְיק הָרָאשׁוֹן

"(a) Fear casts lazy people down, (b) and the souls of effeminate (men) will starve." G here is fairly close to M 19:15, but there the Greek is quite different. The translation in 19:15 is tied to context. Apparently G's source text had M 19:15 in 18:8 as well, but it is hard to explain the disparity between the translations. M 18:8 = M 26:22, but G  $26:22 \neq G$  18:8. In other words, assimilation was at times one of the translator's motives, but it was not applied consistently.

18:9 מתרפה M $\approx$ S (גאינאסת, אינאסת, מתרפא M<sup>MSS</sup> G (לש<br/>  $M^{MSS}$  G (לשה גומינש) למערט) (aur)

G: (a) ὁ μỳ ἰώμενος ἑαυτὸν ἐν τοῖς ἔργοις αὐτοῦ (b) ἀδελφός ἐστιν τοῦ λυμαινομένου ἑαυτόν. "(a) He who does not heal himself by his labors (b) is the brother of the one who harms himself." Since G could easily have understood מתרפה as "slack off," its natural sense in this context, his source text probably had מתרפא. The latter writing, found also in a Geniza fragment (Rüger 1959, 275) and a number of medieval MSS KR, arose in Hebrew copying by careless association of the two homophones.

The error forced the addition of the negative and created a rather awk-ward statement.

## 18:10

G ἐκ μεγαλωσύνης ("from the greatness"), which is supported by αλ' (ἐκ μεγέθους ἔργων), misconstrues מְגָדַל as מְגָדַל.

## 18:11

G: (a) ὕπαρξις πλουσίου ἀνδρὸς πόλις ὀχυρά, (b) ἡ δὲ δόξα αὐτῆς μέγα ἐπισκιάζει. "(a) The property of a rich man is a strong city, (b) and its honor gives much shade." ἡ δὲ δόξα explains the metaphor of "wall." Hitzig retroverts the Greek to הוכבודה, but that is graphically distant from M's במשכיתו In 18:11b, G's "shade" associates M's וכתומה with שכ"כ /שֹכ"כ ("to cover over"). (כר"כ) is rendered by ἐπισκιάζειν in Ps 91[90]:4 and by σκιάζειν in 1 Chr 28:18 and Job 40:22.)

S: איסיא strong city is glory and wealth, (b) and in a strong wall is his dwelling." S continues the description of the righteous man's rewards (hence איסא "wealth" for "rich man" in 18:11a). S thereby avoids praising wealth, as M seems to do, and enhances the moral message. איסיא derives the difficult איסי from איס", but differently than G. That S had the very different משכיתו (BHQ) is doubtful.

## 18:14

G: (a) θυμὸν ἀνδρὸς πραΰνει θεράπων φρόνιμος, (b) ὀλιγόψυχον δὲ ἀνδρα τίς ὑποίσει; "(a) A sensible servant eases a man's anger, (b) but who will bear a discouraged man?" The first stich seems like a guess at the difficult Hebrew. G's guess creates a proverb with a rather subtle lesson: although it is prudent to appease the anger of one's master, faintheartedness, with the refusal to say anything that may irritate him, is intolerable. (M would be best translated, "A man's spirit can sustain him in sickness, but an ill spirit—who can bear it?")

נבון 18:15 א נבון M G (לסטינעסט) ] אנרון S (אמרא) (graph ב $\to \to$ 

גם corresponds to a form of כו"ן everywhere in S-Proverbs except for here and 20:24. Here too it probably represents גכון But M's נבון best fits the context and the parallel הרמים.

## 18:17 ובא M<sup>K</sup> ] וורא M<sup>Q</sup> (gram, equal)

Either the ketiv (יָבא) or the gere is possible, but the gere's use of the

בְּרִיבְוֹ יּיָבֹא<sup>ּז</sup>־בִׁיֹמָהוּ וַחֲקָרְוֹ: 18 מֲדְיָנִים יַשְׁבִּית הַגּוֹרֶל וּבֵין עַצוּמִים יַפְרִיד: 19 אָח נִפְשָׁע מִקּרְיַת־עֵׂז <sup>וּ</sup>וּמְדוֹנִים<sup>ז</sup> בּּבְרָיִח אַרְמוֹן: 20 מִפְּרֵי פִּי־אֵישׁ תִּשְׂבַע בִּטְגֵו תְּבוּאַת שְׂפָתֵיו יִשְׂבֵע: 21 מֵוֶת וֲחַיִּים בְּיַד־ לְשָׁוֹן וְׁאֹהַבֶּיה יֹאכַל פִּרְיֶה: 22 מְצֵא אֲשָׁה מֵצָא עוֹב וַיָּפֶּק רְצוֹז

conjunction (reflected apparently in G's and S's "and when") seems to be an attempt to smooth out the abrupt introduction of the second stich as the *ketiv* has it. Abruptness, however, is often characteristic of Proverbs's style. One type of this asyndesis is discussed in ABP 1.256.

G: (a) δίχαιος ἑαυτοῦ κατήγορος ἐν πρωτολογία. (b) ὡς δ' ἂν ἐπιβάλῃ ὁ ἀντίδιχος, ἐλέγχεται. "(a) A righteous man is his own opponent at the start of the plea, (b) but when the litigant responds, he [i.e., the latter] is refuted." πρωτολογία is the prosecutor's right to speak first (GELS). G ἐπιβάλλειν here means "respond to an accusation," and ἐλέγχειν has its classical sense of "refute," "disprove" (BAP). Using technical judicial terminology, the verse says that the righteous man can effectively plead his own case and rebut accusations. Righteousness endows a man with rhetorical powers.

#### 18:18

G: (a) ἀντιλογίας σιγηρὸς [G<sup>B S\* A MSS</sup>; var: κλῆρος G<sup>Sc V MSS Rahlfs</sup> SyrH] παύει κλῆρος, (b) ἐν δὲ δυνάσταις ὁρίζει. "(a) The silent man [var: "the lot"] stops conflicts (b) and separates princes." σιγηρός is OG, while κλῆρος ("lot") (= M) is the revision. The translator used "silent man" out of a hesitation to praise lot casting; similarly 16:33.

# 18:19 אנפשע [ M [ נפשע G (βοηθούμενος) S (גדע ברו (graph?) אינושע [ M<sup>K</sup>] אמדינים [ M<sup>K</sup>] ומדונים (M<sup>CQ</sup> M<sup>Or</sup> (dial? norm)

G: (a) ἀδελφὸς ὑπὸ ἀδελφοῦ βοηθούμενος ὡς πόλις ὀχυρὰ καὶ ὑψηλή, (b) ἰσχύει δὲ ὥσπερ τεθεμελιωμένον βασίλειον. "(a) A brother helped by a brother is like a fortified and lofty city (b) and is as strong as a wellfounded palace." G is largely a guess at an obscure verse, rendered with little attention to word order. Nevertheless, βοηθούμενος can be plausibly retroverted to yyii (lit. "is saved," i.e., "helped"). βοηθεῖν renders y"w'(H-stem) in, e.g., Deut 22:27; 28:29, 31; Josh 10:6; and most significantly (in the N-stem) in Prov 28:18.

S דמתעדר ("is helped") (whence T<sup>L</sup> דמתעדר; cf. BHQ) is probably

dependent on G. The mechanism of change from נפשע to נפשע is unclear, but נפשע would not have been rendered "is helped." Given the difficulties of the verse, G שֹׁכ דילג is uncertain evidence for "בקרית", though that is a reasonable emendation if the original said something positive about the brother.

On the *ketiv-qere*, see the comment on 6:14.

## 18:20 תבואת $M^{Oc} M^{OrQ}$ ] תביאת M<sup>OrK</sup> (graph $i \rightarrow i$ )

G's ἀπὸ καρπῶν ("and from the fruits") in 18:20b for M's תבואת ("the produce of [sc. his lips]") was influenced by the phrase "the fruit of the mouth" in 18:20a as well as in 12:14a and 13:2a.

## 18:21

Jäger and others retrovert G's oi δε κρατοῦντες αὐτῆς ("and those who seize it") to אָהָרָיָה, for M's ואָהרֵיה, but there is no graphic similarity between t and ב to explain the error. The notion of "loving" the tongue (an organ that can produce either good or evil) may have puzzled the translator (as it does some modern commentators, e.g., Delitzsch and Clifford). Moreover, M may have seemed susceptible to a misreading advocating garrulousness. (It actually means to cherish fine speech.) For caution's sake, the translator substituted κρατεῖν. By this translation, the verse inculcates the familiar teaching of controlling one's tongue; compare the advice in 21:23 and 16:32. The notion of "control" accords with the metaphor of ἐν χειρί in the first stich.

18:22 אשה rab G (מֹעְמּטָאָ) מא' (ערָאָדאָד) SyrH (מֹעְמּטָאָ) S (גראַג) T (אראַא) (> M) (explic)

fin ] + 2 stichoi G (22a: cf. S) (dbl, vocal)

G: (a) δς εὖρεν γυναῖκα ἀγαθήν, εὖρεν χάριτας, (b) ἔλαβεν δὲ παρὰ θεοῦ ἱλαρότητα. "(a) He who has found a good wife, has found happiness, (b) and has received joy from God."

"Woman" is modified by "good" in several sources (see above), including Hexaplaric translations that must have had it in their Hebrew. Note especially the rabbinic sources, b. Ber. 8a; b. Yeb. 63b; Mid. Shoher Tov 151ab, with this addition. Some manuscripts of V ( $^{AMSS}$ ) add *bonam*.

M 18:22a speaks of the good fortune of finding a wife, thereby praising marriage itself. (The sages were, however, well aware that unfortunate matches are possible; cf. 19:13; 21:9, 19; 25:24; 27:15). G and other witnesses add "good," to make it clear that only a virtuous woman is a blessing. "Good" must have been present in the Hebrew source text of G

מֵיְהוֲה: 23 תַּחֲנוּנִים יְדַבֶּר־רֶלֶשׁ וְׁעָשִׁיר יַעֲנֶה עַזְוֹת: 24 אֵישׁ רֵעִים לְהִתְרֹעֵע וְיֵשׁ אוֹהֵב דְּבֵק מֵאֶח: 19:1 טְוֹב־רָשׁ הוֹלֵך בְּתֻמֵּו מֵעִמֶשׁ שְׁפָתִיו וְהַוּא 'עָשִׁיר'': 2 גַּם בְּלֹא־דַעַת נֶפָשׁ לֹא־עָוֹב וְאֶץ בְּרַגְלַיִם

18:22a, since אשה טובה is reflected in G 18:22a, which, in Hebrew, is a doublet of G 18:22a. The Hexaplaric translations (see above) must have had it in their Hebrew as well. To be sure, "good" could have been added without a textual basis, as in V MSS. In fact, the medieval commentators, though using M without this word, emphasized that אשה refers only to a good woman. Nevertheless, the diversity of the sources cited, and the presence of שובה in talmudic and midrashic citations, show that the word was found in some ancient Hebrew MSS and that the expanded form had become a proverb in its own right.

#### 18:22*a*

The second stich, 18:22*a*b, was not available to S (see below). Hence it is probably a later addition in Greek. It supplies an antithesis to the first stich, 18:22*a*a, and also reminds the reader of the importance of divorcing an adulteress, which was obligatory in rabbinic law.

After M 18:22, S adds only 18:22*a*: مرحم جمع جمع ("And he who expels a good wife expels good from his house"). S had the equivalent of G 18:22*a* in its source text but not 18:22*a*b. Joosten (1995, 71–72) says that the second rendering of 18:22 (i.e., G 18:22*a*) is a translation of G but with attention to M. However, if S were dependent on G here, it would probably have copied the entire G addition. Also, S 18:22*a*b is not identical to G 18:22*a* but adds "from

his house" to explain (unnecessarily) where the woman must be expelled from. The likelihood is that S had a Hebrew form of 18:22*a* in its source text, which was implicitly vocalized as מַצָיא אָשָׁה טוֹבָה מַצָא אוֹם.

**18:23** M S ] > G (unc)

G lacks 18:23–19:2. I see no mechanical or ideological reason that would motivate the omission of these four verses from G.

**18:24** M S ] > G (unc)

S: (a) איש דיעבא די גועבא (b) = M. "(a) There are friends who are (truly) friends." S construes איש (correctly) as שי, in accordance with the *sebir*; see ABP.

#### **19:1** M ] > G (unc)

עשיר S (حאיד) V<sup>MSS</sup> (dives) ] כְסֵיל M (reform)

Since a similar verse is omitted in 21:5, one might weigh whether G chose to omit 19:1 because the second stich seems to disparage ambition. Still, the first stich is entirely acceptable to G, and that too is absent. Also, 19:1 is part of a block of four omitted verses, so its absence cannot be explained individually.

V: A variant in V has *dives* ("rich man") rather than *insipiens* ("fool"). Since the phrase is absent from both G and OL, and V here differs from the way it translates 28:6b (*quam dives pravis* ["than a crooked rich man"]), the unexpected *dives* seems to be original and thus to witness independently to a Hebrew variant. T  $\sigma \subset \Lambda$  M.

M 19:1 is ideologically unexceptionable, and it is unlike V to make radical changes for the sake of aesthetics or poetic logic (S, admittedly, might do so.) Earlier scribes in the Hebrew transmission, however, felt more at liberty to transfer verses and verse components, so the S-V variant has a claim to antiquity. As a rule we would not expect a significant Hebrew variant in Jerome. Nevertheless, the independent testimony of S and V support the hypothesis that there was a variant proverb, with 28:6b in place of 19:1b. The fact that G lacks 18:23–19:2 suggests that this part of the text was in flux fairly late.

M reads, "Better a poor man who walks in his innocence than a man of crooked lips who is a dolt." This saying is unbalanced and rather banal, for an innocent poor man is better than either a liar *or* a fool, and

חוֹטֵא: 3 אַוָּלֶת אָדָם תְּסַלֵּף דַּרְבָּוֹ וְעַל־יְהוָה יִזְעַף לִבְּוֹ: 4 הוֹז יִסִיף ַרַעִים רַבָּים וְדָל <sup>ר</sup>מֵרַעֵהוּ יִפְּרֵד: 5 עַד שֲׁקָרִים לְא יִנָּקֶה וְיָפְיַח בְּזָבִים לָא יִפְּלֵט: 6 רַבִּים יְחַלִּוּ פְנֵי־נָדֵיב וְכָל־הְרַעַ לְאֵישׁ מַתֶּז: 7 בְּל אֲחֵי־רָשׁ ו שְׁנֵאָהוּ אָף בְּי מֲרַעֵהוּ רְחַקוּ מִמֶּנוּ מְרַדֵּף אֲמָרִים לֹא הַמָּה: 8 לְנֶה־לֵב אֹהֵב נַפְּשׁׁוֹ שׂמֵר תְׁבוּנָה לִמְצֹא־טְוֹב: 9 עֵד שְׁקָרִים

combining the two only weakens the force of the contrast. In a "better than" proverb like this, the second line should mention a quality that, other things being equal, is more desirable than indigence; see ABP. In any case, both the M and the S-V variant are valid proverbs.

**19:2** M S ] > G (unc)

## 19:4 מֵרֵעָהוּ [ M<sup>A</sup>Y ] מֵרֵעהוּ M<sup>L</sup> (vocal err)

19:6

G: (a) πολλοὶ θεραπεύουσιν πρόσωπα βασιλέων, (b) πᾶς δὲ ὁ κακὸς γίνεται ὄνειδος ἀνδρί. "(a) Many serve the faces of kings, (b) but every bad man becomes a reproach to (another) man." Similar phraseology occurs in G 29:26a, where θεραπεύουσιν translates מבקשים (lit. "seek"). In 19:6a, G has "kings" for M's דִרִיב ("noble"). As in 25:15; 28:16; and 29:12, G elevates a high-ranking person (משל, גגיד, קצין) to royal status so as to underscore the seriousness of the advice. κακός construes אהרע as אהרע for M's מרון (for M's מתן), but that correspondence does not occur elsewhere. It seems that having begun the translation on a wrong track in 19:6b by assuming that begun the translation on a wrong track in 19:6b by assuming that to plug in a word ("disgrace") that made sense of the clause. This antithesis was not created for literary elegance but to make sense of a misunderstood text.

S is based on M but largely ignores its syntax.

19:7 ממני ] + 4 stichoi G (7c-f) (elab)

 $M^{K}$ ] לא  $M^{V}$  (aur, equal)

The *ketiv* and the *qere* are "equal" insofar as 19:7c is incomprehensible by either variant.

G: (a) πᾶς, ὅς ἀδελφὸν πτωχὸν μισεῖ, (b) καὶ φιλίας μακρὰν ἔσται. (c) ἔννοια ἀγαθὴ τοῖς εἰδόσιν αὐτὴν ἐγγιεῖ, (d) ἀνὴρ δὲ φρόνιμος εὑρήσει αὐτήν. (e) ὁ πολλὰ κακοποιῶν τελεσιουργεῖ κακίαν. (f) ὅς δὲ ἐρεθίζει [var: περικεντεῖ G<sup>MSS</sup>] λόγους [var: λόγοις G<sup>Complut MSS</sup>], οὐ σωθήσεται. "(a) Everyone who hates a poor brother (b) will also be far from friendship. (c) Good understanding will come to those who know it, (d) and a sensible man will find it. (e) He who does much evil fully accomplishes evil, (f) and he who provokes [var: "pierces"] by words will not be saved." (The dative λόγοις in 19:7f, though poorly attested, seems necessary.)

G 19:7ab: G converts the first couplet from a cynical assertion to a moral lesson, partly by ignoring the Hebrew word order and by treating the morphology loosely. After 19:7b, G adds two couplets, verse 7cd and verse 7ef (not marked in Rahlfs as additions).

G 19:7cd: The second couplet is an independent proverb absent from M. It is an addition present in G's source text and can be retroverted to M. It is an addition present in G's source text and can be retroverted to dition, to clarify the elliptical Hebrew construction "X to Y" (found also in Ps 111:10). Behind ຂັບບາລ ຂ່າຍ (found also in Ps 111:10). Behind ຂັບບາລ ຂ່າຍ (construction are not modified by "good," as שׁכָל טוב ליודעיו ואיש תבונה שׁכָל טוב ליודעיו ואיש תבונה שׁכָל טוב (construction in 3:4; 13:15; Ps 111:10; and 2 Chr 30:22.) The retroverted Hebrew could also be translated: "(c) There is good intelligence for those who know it, (d) and the man of good sense will attain it." ("It" in 19:7cd refers to "good intelligence.")

G19:7ef: The third couplet, like the second, had a Hebrew source. דבאבסוסטאָצוֹ אמאלמע represents the Hebrew idiom כלה רעה (see 1 Sam 20:7, 9; 25:17). The couplet is related to the enigmatic M 19:7c (אמרים לא המה מרדף) by λόγοις = אמרים לא המה (לא המה) by λόγοις מרדף) and "provoking" (לאביו). The Hebrew behind this couplet cannot be retroverted except by guesswork. This is unfortunate, because it might have provided a clue to the correct form of M 19:7c, an isolated and meaningless clause that looks like the mangled fragment of a lost couplet. G's negative (où) seems to support the *qere*, but the sentence is too distant from M to determine just what was in its source. V is influenced by G.

S 19:7c (ناب جنہ, محلم میں مخلہ مرب جنہ; "He who is malicious in his words is not strong") ≈ G 19:7f.

19:8

S has המבונה ("faith") for תבונה. BHQ says, opaquely, that S arises

לַא יִנְּקֶה וְיָפֶיחַ כְּזָבִים יאבִד: פּ 10 לְא־נָאוֶה לִכְסֵיל תַּעֲגָוּג אַּף כְּי־לְעֶבָד | מְשׁׁל בְּשָׂרִים: 11 שֵׂכָל אֲדָם הֶאֶרִידְ אַכּּוֹ וּןְתִפְאַרְתּוֹ עַבְר עַל־פָּשׁׁע: 12 נַהַם כֵּכְּפִיר זַעַף מֶלֶד וּרְטַל עַל־ עֵשֶׁב רְצוֹנְוֹ: 13 הַוֹת לֱאָבִיו בֵּן כְּסֵיל וְדֶלֶף טֹבִד מִדְיְגֵי אִשֶׁה: 14 בַּיִת וַהוֹן נַחֵלֵת אֶבִוֹת וֹמֵיִהוָה אִשָּׁה מַשִׂכֵּלֵת: 15 עֵצְלָה תַּפִּיל

"via אמונה." In fact, אמונה." interchange is possible in the paleo script (◄, ◄). For the phenomenon, see Talmon 1981, 504–21. The error was facilitated by the fact that "keep faith" felt more natural.

#### 19:10

G: (a) = M; (b) καὶ ἐἀν οἰκέτης ἄρξηται μεθ' ὕβρεως δυναστεύειν. "(b) or if a home slave begins to rule with arrogance." G's addition of μεθ' ὕβρεως allows for the possibility that a home slave can properly gain control, as 17:2 asserts (even more emphatically in G than in M).

## 19:11 וְתִפָּאַרְתוֹ [ MAY ] וְתִפָּאַרְתוֹ (vocal err)

M<sup>L</sup>'s reading is an obvious scribal error. M's האריך is understood as an infinitive (for H-stem infinitives absolute with this writing, see GKC \$53k) and translated as a noun in S (גנירות "lengthening"); T<sup>L</sup> (גנירות); V (*per patientiam* "by patience" [+ *noscitur* "is known"]). (In fact, M's הַאָּרִיךָ

For unclear reasons, T has אורחיה ("his way") for אפו.

#### 19:12

For <sup>τ</sup> τος ("roar like a lion"), G substitutes δμοία βρυγμῷ λέοντος ("like the bite of a lion"). G-Proverbs may not know the meaning of <sup>1</sup>. It is translated differently in each of its occurrences in Proverbs (θυμοῦ ["anger"] in 20:2 and πεινῶν ["hungry"] in 28:15; on 5:11 see the comment). Elsewhere in the Bible <sup>1</sup> tis usually translated, correctly, with a word meaning "shout" or "roar."<sup>4</sup> In a sense, G is more logical than M, since it is the lion's bite, not his roar, that is dangerous. G translates 19:12a quite differently in its doublet, 20:2a.

<sup>4.</sup> Both GELS and LSJ give "roaring" as one sense of  $\beta \rho \nu \gamma \mu \delta \varsigma$ , but both refer only to this verse. This is a case of imposing the meaning of the Hebrew word on the Greek.

#### 19:13

G: (a) αἰσχύνη πατρὶ υἰὸς ἄφρων, (b) καὶ οὐχ ἁγναὶ εὐχαὶ ἀπὸ μισθώματος ἑταίρας. "(a) A foolish son is a disgrace to his father, (b) and vows (paid) from a harlot's wages are not sacred." G's αἰσχύνη ("disgrace") for הות ("disaster") is probably influenced by knowledge of what a foolish son is said to cause in 19:26 and 29:15: to disgrace his parents. S ܐܘܬܬܐ = G  $\neq$  M 19:13b. (A variant of 19:13b is rendered fairly literally in G 27:15.)

It is possible that G's source text read מדיני ("from the gifts"); for אתנן (BHQ; see Ezek 16:33, though the term used there is אתנן). More likely, G has chosen to provide a more specific interpretation of מדיני אשה). More likely, G has chosen to provide a more specific interpretation of מדיני אשה) by introducing a teaching based on Deut 23:19. The present verse looks like a cautionary response—albeit belated—to Prov 7:14, in which the Strange Woman says, "I have peace offerings. Today I pay my vows [εὐχάς]." In 21:9 as well, G redirects the topic from the contentious woman to one of greater moral import. (In 21:19; 25:24; and 27:15, however, G maintains M's complaint about the obnoxious wife.) G<sup>A</sup>, continuing the ancient practice of combining monostichs to make new proverbs, adds 19:13c: καὶ ὀδύνη τῇ τεκούσῃ αὐτοῦ ("and misery to his mother") = 17:25b.

## 19:14

G: (a) οἶχον καὶ ὕπαρξιν μερίζουσιν πατέρες παισίν, (b) παρὰ δὲ θεοῦ ἁρμόζεται γυνὴ ἀνδρί. "(a) Fathers allocate a house and possessions to children, (b) but a woman is matched [or 'betrothed'] to a man by God." In a similar vein, Gen. Rab. §68 says that God has been arranging marriages ever since creation. The translation of משכלת as "matched"/"fit together" may be playing on שכל ("cross," hence "interleave"; de Lagarde vocalizes גְמָשֶׁבְּלֶת, with reference to Gen 48:14). Lady Wisdom uses ἑρμόζουσα to describe what she was doing "with God" at creation (8:30).

#### 19:15

G: (a) δειλία κατέχει ἀνδρογύναιον, (b) ψυχὴ δὲ ἀεργοῦ πεινάσει. "(a) Timidity seizes the effeminate man, (b) and the soul of the lazy will starve." (α Catually "deep sleep") is understood to mean fear also in G-Gen 15:12 and G-Job 33:15. The lazy man is called "effeminate" just as the diligent one is called ἀνδρεῖος, that is to say, possessing the manly virtues (10:4; 13:4; etc.). BAP sees a connection between 19:14–15 and Plato's *Symposium* 189de, which speaks of man, woman, and androgyne as the three primordial human types. But the similarity is slight. תַּרְדֵּמֶה וְגָפֶשׁ רְמִיָּה תִרְעֲב: 16 שׁמֵר מֲצְוָה שׁמֵר נַפְּשֵׁוֹ בּוֹזֵה דְרָבְיו <sup>ה</sup>יָמְוּת<sup>י</sup>: 17 מַלְוֵה יֵהוָה חִוֹנֵן דֶּל וֹּגְמָלוֹ יִשַׁלֶּם־לְוֹ: 18 יַפֵּר בְּנְדָ כִּי־יֵשׁ תִּקְוָה וְאֶל־הֲמִיתוֹ אַל־תִּשָּׂא נַפְשָׁדָּ: 19 <sup>-</sup>וּגֶדָל<sup>י</sup>־חֵמָה נַשׁא עָׁנָשׁ כִּי אִם־תַּגִּיל וְעוֹד תּוֹסְף: 20 שְׁמַע עֵצָה וְקַבֵּל מוּסֵר לְמַעַן תֶּחְכַּם בּאַם־תַּצִיל וְעוֹד תּוֹסְף: 10 שְׁמַע עֵצָה וְקַבֵּל מוּסֵר לְמַעַן תָּחְכַּם בּאַם־תִּגִיה וְאָרַם חַסְדָי וְטוֹב־יָרשׁ מַאָישׁ נַעָּצָה וְקַבֵּל בּאַחַרִיתֶדָּ: 21 רַבְּוֹת מַחַשְׁרָוֹת בְּלֶביאֵישׁ וַעַצְת יְהוָה הִיא תָקוּם: 22 הְבוּאַת מָקוּם: זְשָׁבָע יְהוָה בָּל־יִמָּקָד וְטוֹב־יָרשׁ מַאָישׁ בָּזֵב: 23 יִרְאָת יְהוָה לְחַיֵּים וְשָׁבֵע יְלִין בַּל־יִפָּקָד רֵע: 24 טְׁמָן עָצֵל יְדוֹ בַּצַלֶחֵת גַּם־אֶל

**19:16** ימות M<sup>Q</sup>G (ἀπολεῖται) V (mortificabitur) T (מות M<sup>K</sup> (gram) The qere "will die" rather than the ketiv "be put to death" is certainly right, because the penalty for being contemptuous is not capital punishment.

In S, Щала represent either the *ketiv* (Hp-stem) or the *qere* (G-stem).

#### 19:18

Where M reads וְאֶל־הֲמִיתוֹ, G's εἰς δὲ ὕβριν ("to arrogance") implicitly vocalizes המה = איאזדסוקט ("and to his insult/arrogance") = G.

19:19 גדל  $M^Q G (\pi \circ \lambda \dot{\alpha}) \approx S (use M^K (graph <math>\tau \to \tau)$ 

The *ketiv* is a meaningless mechanical error. G πολλά and S ("wrathful") witness to the *qere*, so it must have existed as a textual variant.

G: (a) κακόφρων ἀνὴρ πολλὰ ζημιωθήσεται, (b) ἐἀν δὲ λοιμεύηται, καὶ τὴν ψυχὴν αὐτοῦ προσθήσει. "(a) A malicious man will be severely punished, (b) and if he is obnoxious [lit. 'pestilent'], he will also add (punishment) to himself." λοιμεύηται = ν'τ' ("is arrogant" or the like) for M's תציל (de Lagarde). Or possibly G read ארליץ, which is ungrammatical but closer to M's consonants. (Compare how G treats קום) art masc sg)

S: In 19:19a, S's intense "wrathful" implies the notion of "great" = qere. ("sin") does not translate יליץ. S may be influenced by G for the general sense, since M is difficult. (M says, "if you save him," and one may wonder how and why this would happen.)

19:20 שמע בני [ (ברב הל מוסר MS (או שמע עצה וקבל מוסר 19:20 שמע בני [ (ברב הלבא המוסר הבני ( מאטע בני ( מאטע בני ( מאטע בני ( מאטע בני שמע בני ( מאטע בני מוע בני ( מאטע ב

G: (a) מאסטב, טוֹב, המוסבוא המדףלה סט, (b) = M. "(a) Hear, son, the instruction of your father." G 19:20a = שָׁמַע בְּנִי מוּסָר אָביך, taken from 1:8a, either in the Hebrew source text or in the Greek. The motive for this assimilation is not clear.

For M's באחריתך S has האסיעה ("in your ways"), which probably does not represent a variant; see the comment on 14:12.

#### 19:21

G adds εἰς τὸν αἰῶνα ("for ever") in 19:21b; compare Ps 33[32]:11.

## 19:22 תבואת G ] האות M S (unc)

M's "A man's desire (תאות) is his kindness" is not very meaningful. With G's reading, the verse teaches that even a poor man has a "fruit," a benefit he can provide to others, namely, his kindness (ABP). M 19:22b is what I call a disjointed proverb, on which see ABP 2.494–98. The reader is to complete the message by supplying what each clause lacks; thus: "and better (an honest) poor man than (a rich) liar."

G: (a) καρπὸς ἀνδρὶ ἐλεημοσύνη, (b) κρείσσων δὲ πτωχὸς δίκαιος ἢ πλούσιος ψεύστης. "(a) Kindness is fruit for a man. (b) And better a righteous poor man than a rich liar." G adds "righteous" to make it clear that not every poor man has preference to a rich liar, or any liar. καρπός = (ABP; de Lagarde; cf. G 10:16b and G 15:6d). G fills in the blanks of the disjointed proverb in 19:21b to guide the reader.

S has איש י ("desire") (= M) but adds אוי ("rich") (= G). ד ריגתא = S.

## 19:23 דע M S (הידע היא א M S (הידע היא ה) ] דע G (ארע היא ה) (graph $\neg \rightarrow \neg$

G: (a) φόβος κυρίου εἰς ζωὴν ἀνδρί, (b) ὁ δὲ ἄφοβος αὐλισθήσεται ἐν τόποις, οὖ οὐκ ἐπισκοπεῖται γνῶσις. "(a) The fear of the Lord is [i.e., leads] to life for a man, (b) but the unfearing will dwell in places where knowledge does not visit." The erroneous שֵׁת made the verse appear to mean that he who fears the Lord "dwells sated, without being visited by knowledge," which could hardly be said of the God-fearer. The translator avoided this implication by clever parsing of the rest of the clause. He construed this implication by clever parsing of the rest of the clause. He construed the first stich) not fearing God. Such a person will dwell (ילין) in places where knowledge (שָׁת) does not visit (בל יִפְקָד). These are spiritual wastelands such as mentioned in G 9:12c.

פִּיהוּ לָא יְשִׁיבֶנָּה: 25 לֵץ תַּכֶּה וּפָתִי יַעְרֶם וְהוֹכִיחַ לְנָבוֹז יָבִיז דֶעַת: 26 מְשַׁדֶּד־אָב יַבְרַיחַ אֵם בֵּן מֵבִישׁ וּמַחְפִּיר: 27 'חֲדַל'-בְּנִי לִשְׁמַעַ מוּסֵר לִשְׁגוֹת מֵאִמְרֵי־דֶעַת: 28 עֵד בְּלִיַעַל יְלָיץ מִשְׁפֶּט וּפִי רְשָׁעִים

ό αἰώνιος ("the eternal one") in  $G^{V \log (*161 \text{ mg}) 252} = \forall y$ , an error for  $\forall \forall \forall$ . Both variants must have arisen in Hebrew.

#### 19:24

G: (a) ὁ ἐγκρύπτων εἰς τὸν κόλπον αὐτοῦ χεῖρας ἀδίκως, (b) οὐδὲ τῷ στόματι οὐ μὴ προσαγάγῃ αὐτάς. "(a) He who hides (his) hands in his bosom unjustly (b) certainly cannot bring them forth to (his) mouth." The translator misunderstood גלחת (actually "dish") as "bosom" (as in 26:15a, q.v.), making the present verse into a warning against taking bribes. Consequently, M's עצל was replaced by ἀδίκως in order to make the deed a moral issue; see the comment on 10:5. It is possible that "hides his hands in his bosom" means to be stingy, like the idiom κόλπω χεῖρας ἔχειν ("to keep one's hand in one's pocket") (LSJ 974, κόλπος II; suggested by D. A. Teeter, pers. comm.). However, the added ἀδίκως suggests that a distinctly immoral act is in question.

S: ערענאז געלאז אינגסס, באסרס, אב גא לא לאסרס עוד אוש. "(a) A sluggard who hides his hands in his bosom (b) does not even bring them near to his mouth." S  $\approx$  M, but "in his bosom" = G. T בשחאתיה, a'  $\sigma' \mu a \sigma \chi a \lambda \eta \nu$ , and V *ascella* understand the word to mean "armpit(s)," probably independently. This rendering is suggestive of a man hugging himself in inactivity, as in Qoh 4:5.

#### 19:25

S: שברא כא גראיבר ערביבא כמגמא כל האיבר עוביבא כמאבר. "(a) When a fool is beaten, the wise man takes warning, (b) but if you rebuke a wise man, he understands knowledge." The translator follows the syntax of G (turning the imperatives into conditionals) but reverses the meaning of the first clause by translating "wise man." It may seem more "logical" that a wise man rather than a fool would learn a lesson.

## 19:27 חַדַל [ M<sup>AY</sup> ] חַדַל M<sup>L</sup> (vocal err)

M is impossible--it commands the son to cease hearing words of

instruction and, concomitantly, (to cease) straying from it. The versions offer no help.

G: (a) υίὸς ἀπολειπόμενος φυλάξαι παιδείαν πατρὸς (b) μελετήσει ῥήσεις κακάς. "(a) A son who ceases to keep the instruction of (his) father (b) will meditate on bad words." While G is meaningful in itself, it is hard to reconstruct a meaningful Hebrew source text. G forces sense out of his Hebrew—probably the same as M—by using some of M's words and imposing a radical revision on the rest. If this is freedom, it is the freedom of desperation. Given G's license in this verse it would be risky to retrovert ψυλάξαι to ¬𝔅ψας , but see the comment on 6:16.

V: (a) *non cesses fili audire doctrinam* (b) *nec ignores sermones scientiae*. "(a) Do not cease, my son, to hear instruction, (b) nor ignore discourses of knowledge." V deals with the difficulty by supplying negatives.

Some commentators construe the verse as ironic, but this reading does not really work. On this and other attempts to deal with the problems, see ABP. There is very likely at least one textual error in this verse. Instead of לשמע, the source text must have had a word implying repudiation, such as לשמע, ("to hate" [Toy; cf. 5:12]) or the graphically closer diation, such as לשמע ("to hate" [Toy; cf. 5:12]) or the graphically closer ("to spurn," "to throw off"; compare Prov 8:33; 13:18; and 15:32). A scribe might have looked at לשמע מוסר in the present verse and carelessly transformed it into the familiar לשמע מוסר. But in view of the distance of the conjectures from M and the absence of external evidence for them, I have not included them in the apparatus.

און M ] אדין G (געאיז G (הערא S (הערא sc. גער) (graph paleo?  $\bigstar \rightarrow \bigstar$ 

G: (a) ὁ ἐγγυώμενος παῖδα ἀφρονα καθυβρίζει δικαίωμα, (b) στόμα δὲ ἀσεβῶν καταπίεται κρίσεις. "(a) He who gives surety for a foolish child dishonors what is just, (b) and the mouth of the wicked [pl] swallows down judgments." M's ערב בן בליעל became עד בליעל (ערב בן בליעל). This could have happened by a series of graphic errors (de Lagarde) or by one confused misreading. In either case, it was a mechanical error. There is no ethical or religious motive to explain the change, which involves changing condemnation of a dishonest witness into a strange warning against giving surety—specifically on behalf of a foolish childיְבַלַּע־אֶוֶן: 29 נְכִוֹנוּ לַלֵּצִים <sup>ר</sup>ּשְׁבָטֵים וּמַהַלֻמוֹת לְגֵו בְּסִילִים: 20:1 לֵץ <sup>ר</sup>ְהַיַּיִן המֶה שֵׁכֶר וְכָל־שִׁגֶה בּוֹ לְא יֶחְבֶם: 2 נַהַם בַּבְּפִיר אֵימַת מֶלֶד מִתְעַבְּרוֹ חוֹטֵא נַפְשְׁוֹ: 3 בְּבִוֹד לָאיש שֶׁבֶת מֵרֵיב וְכָל־ אֶׁוִיל יִתְגַּלֵּע: 4 מֲחֹרֶף עָצֵל לֹא־יַחֲרָשׁ יוִשְׁאָל בָּקָצִיר וָאֵין: 5 מַיִם

and then condemning this act not as foolish leniency but as a dishonor to justice. The very awkwardness of G's advice indicates that the translator was struggling with a difficult Hebrew source. G's  $\varkappa \rho (\sigma \varepsilon \varsigma = \gamma \tau) r$  for M's The  $\tau \to \tau$  is, of course, a common change, but  $\varkappa \to \tau$  is not. Still, the confusion may be due to the similarity between  $\tau$  and  $\varkappa$  in the paleo script ( $\sigma, \ll$ ).

19:29 שבטים G (עמסדו אברטים S ? (גיאראי) א שבטים M (graph ב - ב א ) משבטים M (graph ב - ב א לגוי (near dittog ו - א לגוי ()

שׁבָּטִים is the preferable reading. שׁבּטים, as Delitzsch points out, is used only of divine judgments, never of penalties inflicted by a human court of justice. Elsewhere stupid people face mundane chastisements, usually beatings. The near-doublets of 19:29b in 10:13b and 26:3b support the appropriateness of reading שׁבטים. M's לגו is the better reading, since the context speaks of individual, not national, punishment.

G: (a) ἑτοιμάζονται ἀχολάστοις μάστιγες (b) καὶ τιμωρίαι ὤμοις ἀφρόνων. "(a) Blows are readied for the licentious, (b) and punishments for the backs of fools." G associates this verse with the following context by using ἀχολάστοις for ±λτέτα Δτιγες = ± as in Job 21:9.

S: حجب حمّت العرب المعنى المعنى المعنى المعنى المعنى المعنى (a) Pains are readied for those who strengthen litigation [i.e., the litigious], (b) and blows for the foolish nation." حمّت حمال المعنى معنى المعنى المعنى

## 20:1 הַייָן M<sup>AY</sup> ] הַיַין M<sup>L</sup> (vocal err)

#### 20:2

G: (a) οὐ διαφέρει ἀπειλὴ βασιλέως θυμοῦ λέοντος, (b) = M. "(a) The king's threat does not differ from the rage of a lion." G gives the converse of M 20:2a: not differing from the lion's anger (G) is equivalent to being the same as its roar (M). G<sup>MSS</sup> add καὶ ἐπιμιγνύμενος ("he who mixes"), offering an alternate translation, one associated with מתערבו ; see the comment on 14:10. G renders this sentence differently in 19:12a; S uses the same wording for both. S understands and a participle.

#### 20:3

Compare 17:14 and 18:1. See BHQ for the various attempts to deal with the unknown יתגלע.

20:4 ישאל M<sup>K</sup> ] אין M<sup>Q</sup> G (אמו ה לאטיג) S (גדים) (gram, equal)

S: האנעמו ערנעא סבא באם משאל בעניה מלא איז (a) The sluggard reviles and does not shut up, (b) and he asks [sc. for water] in the harvest and there is no water." S construes מחרף (actually "after the rainy season") as מחרף and מחרף as "plow." (מנעמון "after the rainy season") as מחריש as "plow." (מנעמון "after the rainy season") as מחריש is also possible.) S also includes מים, from M 20:5, in 20:4. T = M.

20:5-14

אַמָקִים עַצָּה בְּלֶב־אֵישׁ וְאָישׁ תְּבוּגָה יִדְלֶנָּה: 6 רָב־אָדָם 'יִקָּרַא' אַישׁ 'חֶסֶד' וְאִישׁ אָמוּנִּים מִי יִמְצֵא: 7 מִתְהַלֵּדְ בְּתַמִּוֹ צַדָּיק אַשְׁרֵי בְנֵיו אַחֶכֶיו: 8 מֶׁלֶדְ יוֹשֵׁב עַל־כִּפַּא־דִין מְזָרֶה בְעֵינֵיו כָּל־רֵע: 9 מִי־יָאׁמַר זַכִּיתִי לִבֶּי טְׁהַרְתִי מֵחַטָּאתִי: 10 אָבֶן וָאָבֶן אֵיפָה וְאֵיפֶה תּוֹעֲבַת יְהוֹה גַּם־שְׁנֵיהֶם: 11 גַּם בְּמַעַלָּלִיו יִתְנַכָּר־גָעַר אִם־זַדְ וְאִם־יִשְׁר פְּעֵלוֹ: 12 אֵׂזֶן שִׁמַעַת וְעַיִן רֹאֶה יְהוָה עָשָׂה גַם־שְׁנֵיהֶם: 13 אַל־ תֵּאֶהַב שְׁנָה כָּן־תִּנְרָשׁ פְּקַח עֵינֵידְ שְׂבַע־לֵחָם: 14 רַע רָע יאמַר

#### 20:5

## 20:6 יֵקָרָא S (حאםד ) V (vocantur) ] יֵקָרָא M (misinterp)

חסד הסד G (לאנאנאט) א הסד G (לאניעשט) א הסד M (dittog ו  $\to$  11)

M would have to be translated, "Many men call a man his kindness," which is obscure. יקרא makes more sense and offers better parallelism when vocalized as a passive. The verse should be read רְב-אָדָם יִקְרָא אמונים מי ימצא ("Many a one is called a kind man, but a trustworthy man, who can find?"). איש חסד is used in a similar sense in Ezek 23:42 and Zech 2:8.

G: (a) μέγα ἄνθρωπος καὶ τίμιον ἀνὴρ ἐλεήμων, (b) ἄνδρα δὲ πιστὸν ἔργον εὑρεῖν. "(a) A man is great, and a merciful man is precious, (b) and it is (hard) work to find a faithful man." G (a) = יָקָר אָישׁ הֶטֶד, דְרַב אָדָם, יָקָרָא Apparently G parsed יקרא as an Aramaism, though it may have had יקר by haplography. G's חסד is preferable to M's חסדו, with its otiose possessive pronoun.

V *misericordes* ("merciful") represents הסד, as does S's מדיערביא ("merciful"). V and S are independent of G elsewhere in the verse. *vocantur* = אָקָרָא

is also passive. حدامة ب

20:8

G: (a) ὅταν βασιλεὺς δίκαιος καθίσῃ ἐπὶ θρόνου, (b) οὐκ ἐναντιοῦται ἐν ὀφθαλμοῖς αὐτοῦ πᾶν πονηρόν. "(a) When a righteous king sits on the throne, (b) no evil can stand before his eyes." G adds a moralizing δίκαιος, to make it clear that the proverb applies only to righteous kings. G omits to represents it obliquely by δίκαιος. In 20:8b, G states the *effect* of scattering, understanding the verb as passive (implicitly vocalized ἰς) and paraphrasing the action.

 $S \,$  מוֹרָה ("are scattered") = מוֹרָה S . מוֹרָה ("from before him") is not necessarily based on מן עיניו but was produced by the assumption that the verb is passive. Also, it may have been unclear how a king can scatter people *with* his eyes.

#### 20:9

G 20:9 = M 20:9. G 20:9*a*, 9*b*, 9*c* = M 20:20, 21, 22, respectively.

#### 20:10

G: (a) στάθμιον μέγα καὶ μικρὸν καὶ μέτρα δισσά, (b) ἀκάθαρτα ἐνώπιον κυρίου καὶ ἀμφότερα. "(a) A big and a small stone, and a double measure: (b) both are unclean before the Lord" (continues in 20:11). G 20:10a explains an idiom ("a stone and a stone, an ephah and an ephah") that might not be clear if translated mechanically.

## 20:11

G: (a) καὶ ὁ ποιῶν αὐτὰ ἐν τοῖς ἐπιτηδεύμασιν αὐτοῦ συμποδισθήσεται. (b) νεανίσχος μετὰ ἱσίου, (c) καὶ εὐθεῖα ἡ ἱδὸς αὐτοῦ (continues from 20:10). "(a) and he who does these things will become entangled in his practices. (b) A youth (who goes) with a pious man—(c) his way is straight." G 20:11a continues 20:10; G 20:11bc form a new proverb. Jäger and de Lagarde consider ὁ ποιῶν αὐτά to be a graphic distortion of μεται de the end of 20:10b, and Jäger retroverts συμποδισθήσεται to τιπό, but the graphic similarities are not great, and the latter equivalence is not otherwise attested. Overall the contacts of the verse with M are loose, making retroversion very uncertain in the absence of recognized mechanical explanations.

#### 20:13

G: (a) שָׁן מֹאמֹת אמדמאמאניי, ווא שָׁן באַרָאָפּאָדָ; (b) = M. "(a) Do not love to babble, so that you not be removed." G associates שנה (M שָׁנָה) with שנה (שַׁנָה M שנ"ה "to repeat" (Jäger), hence "babble." שנ"ה אני"ה

ַהַקּוֹגֶה וְאֹזֵל לֹוֹ אָז יִתְהַלֵּל: 15 יֵשׁ זְהָב וְרָב־פְּנִיגֵים וּכְלֵי יְׁקָׂר שִׂפְתֵי־ דְעַת: 16 לְקַח־בָּגְדוֹ כִּי־עָרַב זֶר וּבְעָד <sup>וּ</sup>נָכְרִים<sup>י</sup> חַבְלֵהוּ: 17 עָרֵב לָאִישׁ לֶחֶם שֶׁקֶר וְאַחֵׁר יִמְלֵא־כִּיהוּ חָצֵץ: 18 מַחֲשָׁבוֹת בְּעַצְה תִכְּוֹן וּבְתַחְבָּלוֹת עֲשֵׂה מִלְחָמֶה: 19 גְוֹלֶה־סּוֹד הוֹלֵך רָכֵיל וּלְפׁתֶה שְׁפָתִיו לָא תִתְעָרֵב: 20 מֵקַלֵּל אָבֵיו וְאָמָוֹ יִדְעָך וֹרוֹ וּבְאִישׁוֹן חְשָׁך:

17:9, but G does not recognize it there. ἐξαίρειν renders שר"ש several times elsewhere (e.g., Judg 1:20, 21; 1 Kgs 14:24; 2 Kgs 21:2) and so does not necessarily reflect תגרש here, contrary to Baumgartner.

#### 20:14-19, 20-22

OG lacks 20:14–19, an absence that has not been explained. The verses are supplied in many MSS from  $\theta'$ . G has an equivalent of M 20:20–22 at G 20:9*a*–9*c*,

#### **20:14** M S ] > G (unc)

S twice misconstrues רע as רַע ("friend").

**20:15** M S ] > G (unc)

## **20:16** M S ] > G (unc)

נכרים M<sup>K</sup> V (extraneis) S (גרים  $M^{Q MSS}$  (graph  $t \to t$ 

The qere is בְּרָיָה (= many MSS KR) as in the doublet in Prov 27:13. In both verses, the better reading is נְכָרִים. When זו is parallel to נְכָרִים elsewhere, the genders always match. (Singular // plural parallelism, however, is frequent and not a problem.) נכריה is problematic also by implying that it is a woman—an alien or stranger, no less—who is taking a loan and needs a guarantor. Even if a woman could in special circumstances receive a loan, this could not have been common in ancient Israel, and the proverb here and in 27:13 requires a typical case to make its point. (The verb  $\chi$  takes as its object the borrower. Other possible direct objects are the item borrowed and the person for whom the guarantor is responsible [see HALOT], but these are not applicable here.) The variant  $\iota$  can be explained as a graphic error facilitated by the references to the  $\pi$   $\tau$  216; 5:20; 6:24; and 7:5; see further ABP 1.214–16.

V: (a) tolle vestimentum eius qui fideiussor extitit alieni (b) et pro extraneis aufer pignus ab eo. "(a) Take away the garment of him that is

guarantor for a stranger, (b) and on account of the strangers, take away the pledge from him." In V, the *guarantor* himself becomes an alien or stranger. According to 20:16b, he loses the pledge (CB). V reads the *ketiv*.

S understands חבלהו as a noun meaning "his pledged item" (= V).

20:17 M S ] > G (unc)

#### **20:18** M S ] > G (unc)

 $S \approx M$ , except that S has האבר ("is done") instead of the imperative. S possibly had תעשה, which it parsed as a passive. But since S also transforms the doublet in 24:6 into the passive, it seems that the reformulation in both places was deliberate. Advice on how to make war would not be relevant to an audience that is not in a position to make war, unlike the courtiers to whom much of M-Prov 10–29 speaks.

## **20:19** M S ] > G (unc)

S: הכבסי אראה אראה, אראה באיז איז איד אראה. אראה אראה אראה אראה איד אראה איד אראה אראה אראה אראה אראה אראה אראה derer, (b) and he who is faithful in his spirit hides a matter. (c) And with him whose lips are rash, do not mix." In 20:19b, S inserts a line (copied from 11:13b) to complement 20:19a. However, some ninth-twelfth century S manuscripts lack this line, and it looks like a later addition.

20:20 באישון M<sup>K</sup> G (ai dè אלאמו) ] באישון M<sup>Q</sup>; רבאישון S (איז הבאיד) T (איז הבאיד) ( $M^Q M^K G$ : orth, equal; S T: graph ב $\rightarrow \supset$ )

אישון are pragmatic synonyms, the first meaning "pupil" (of the eye, hence the dark of the night), the second meaning "time"; see ABP 1.239–40.

G (at 20:9*a*): (a) κακολογοῦντος πατέρα η μητέρα σβεσθήσεται λαμπτήρ, (b) αἱ δὲ κόραι τῶν ὀφθαλμῶν αὐτοῦ ὄψονται σκότος. "(a) The light of him who curses (his) father or mother will be extinguished, (b) and the pupils of his eyes will see darkness." G divides the verse after 17, then overrides the syntax of 20:20b by making אישון (the *ketiv*) the subject and supplying the verb "will see." The resulting syntax has a Greek, not Hebrew, quality.

S האישון ("like the pupil") = אישון (Pinkuss). At the end of the verse T has איך אתונא דחשוכא ("like the essence of darkness," or, we might say, "like the heart of darkness"). איך אתונא נוא בחשוכא apparently represents איך אתונא (or נאשון) and is not simply dependent on S.

11 נָחַלָּה 'מְבֹהֶלֶת' בָּרִאשׁנֶה וְאַחַרִיתָּה לָא תְבֹרֵדְ: 22 אַל־תּאׁמִר אַשִׁלְמָה־רֶע קַוָּה לֵיהוָה וְיַשָׁע לֵדְ: 23 תּוֹעַבַת יֱהוָה אָבֶן וָאָבֶן וּמֹאוְנֵי מִרְמֵה לֹא־טִוֹב: 24 מֵיְהוֶה מִצְעַדִי־גָבֶר וְאָדָם מַה־יָּבִין דַּרְכָּוֹ 25 מוֹמַשׁ אָדָם יָלַע לֵדָשׁ וְאַחַר נְדָרֵים לְבַקּר: 26 מְזֶרֵה רֲשָׁעִים 25 מוֹמַשׁ אָדָם יָלַע לֵדָשׁ וְאַחַר נְדָרֵים לְבַקּר: 26 מְזֶרֵה רֲשָׁעִים מֶלֶדְ חָבֶם וַיֶּשָׁב עַלֵיהֶם אוֹפּן: 27 נֵר יֱהוָה נִשְׁמַת אָדֶם חֹפִשׁ כָּל מַלֶדְ חָבֶם וַיֶּשָׁב עַלֵיהֶם אוֹפּן: 27 נֵר יֶהוָה נִשְׁמַת אָדֶם חֹפָשׁ כָּל חַדְרֵי־בֶטֶן: 28 תָסֶד וְאָמֶת יִצְרוּ־מֵעָלֶדְ וְסָעַד בַּתֶסֶד כִּסְאוֹ: 29 תִּקְרֵיק' בְּרָע וֹמַבוּוּ חַדְרֵי־בְטָן: 1:12 פַּלְנֵי-מַים גָבָרוֹת אָצַע יהַמְרֵיק' בְּרָע וֹמַבּוֹת חַדְרֵי־בְטָן: 1:12 פַּלְנִי-מָים גָברוּת הָצַע יְהַמְרֵיק' בְּרָע וֹמַבּוֹת חַדְרֵי־בְטָן: 1:12 פַּלְנֵי-מָים גָברוּת הָצָרוֹ יְהַמְרֵיק' בְּרָע וֹמַבּוּת יַחְבָרי בְטָלָ וַמְנֵים מוֹבָרָר זְקַנִים מַיּבָהי גַמַנְרֵיק' בְּרָל אָשֶׁר יַחָפָּץ יַשֶּנוּי ב 2 בָּל־בָּרֶד-אָישׁ יִשָּר בְּתָים גָרוּמָר הַמָּרִיק הַרָהוּ גַעַרים בּחוּרִים בָּתָם וּמִידָרי בְטָשָר זַמָנִים מַיּבָרוּ הַמָּנִים יוּהוָה גַין בָּרִיקּים בּרְמָם וַהָרָים בּמָם וַיָּרָי מִינִים 'וְהַתַבּין אָבָרוּרָים בּתוּרִים הַמִינִים וּשָׁאָד יִמְדָע אָדָר רָמָרָים וּדָרָים גַּבָּרָים מְזּמָרי זַשְׁים מֶעינִים וּוּהַיַם-לְמָר וְזָשָׁים אַדָר בְרָריםים מַשָּאַתוּ זימוָר בּלִשְׁיָן

20:21 מבהלת M<sup>Q</sup> G (at 20:9*b*) (<br/>לתוסתטעם איז שההלת שבהלת א M<sup>Q</sup> G (at 20:9*b*) (<br/> festinatur) [ ספור א שרחלת M<sup>K</sup> (graph  $\rightarrow \rightarrow$  ח)

G (at G 20:9*b*): (a) μερὶς ἐπισπουδαζομένη ἐν πρώτοις (b) ἐν τοῖς τελευταίοις οὐκ εὐλογηθήσεται. "(a) A portion achieved hurriedly at first (b) in the end will not be blessed." The *ketiv* is a graphic error. "Nauseated" or "an object of disgust," as some explain בחל to mean (e.g., Moshe Qimḥi, comparing Zech 11:8), does not fit the context; see ABP. G, σ', θ', V, S, and T = qere.

#### 20:22

G at 20:9*c*.

#### 20:23

G adds לעמוי מטינס מטינס to the end of the verse, and T adds קדמוי, apparently to balance "the Lord's" with a parallel term. S does not do this. This is a rare case of  $T = G \neq S$ .

20:24 יבין M G (עט (גנאסב) אירין א  $M^{MSS}$  (graph  $\rightarrow \rightarrow$  )

S יְבִין = הָאָה for M's יְבִין; see the comment on 18:15. S agrees with several MSS KR.

## 20:28

G translates בחסד as έν δικαιοσύνη ("in righteousness") to avoid repeating έλεημοσύνη ("mercy") (BHQ).

## 20:29

G uses σοφία to translate  $\Box \square \Box$  ("their strength") to explain that a young man's true glory is his wisdom.

## 20:30 תַמְרָוּק [ M<sup>Q</sup> (graph י ↔ ו, equal)

G: (a) ὑπώπια καὶ συντρίμματα συναντῷ κακοῖς, (b) πληγαὶ δὲ εἰς ταμίεια κοιλίας. "(a) Black eyes and fractures happen to evil (people), (b) and blows to the chambers of the belly." G construes μ as two items and substitutes a verb ("happen") for the obscure μ as two items (T μαριη as two for the action of the same fractures for the off evil"), deriving follows G. σ' has ἀποσμήξει κακίαν ("will wipe off evil"), deriving μαριής. S μαριής ("rub" or the like). Similarly, V absterget ("wipes away").

## 21:2

G: (a) = M; (b) אמדפטטיטיט של אמסטיט איז (b) but the Lord guides hearts (aright)." (א ותכן (M וחכן) is associated with כו"ן, which is often translated by אמדניטטיטיע, e.g., Prov 4:26.

## 21:3

G's א טינע מוֹע מוֹע ("than blood of sacrifices") is a deliberate elaboration of ( ("than sacrifice"). S  $\psi = \pi + \pi + \pi$ 

## 21:4 איְרְחַב [ איָרחַב M (vocal)

נר [ (λαμπτήρ) S ( (۲) א נר M (vocal)

G: (a) μεγαλόφρων ἐφ' ὕβρει θρασυχάρδιος, (b) λαμπτήρ δὲ ἀσεβῶν ἁμαρτία. "(a) An arrogant man is insolent-hearted with respect to haughtiness, (b) and the light of the wicked is sin." G explains the metaphor "high of eyes" to mean "insolent in respect to haughtiness." I suggest two changes of vocalization: גור ("lamp") (= G), because it is a clearer metaphor than M's גר ("tillage"); and גר לב , because the abstract "breadth of heart" (see 1 Kgs 5:9) is better coordinated with the parallel רום־עֵינַיָם (lit. "height of eyes").

## **21:5** M S ] > G (ideol?)

Possibly G omitted the verse to avoid the appearance of disparaging ambition; cf. 19:2. This motive is clearly at work in the maneuvers of

שְׁמֶר הֶבָל יֹרְדֵׁף בְּמֹקְשֵׁיּ־מֶוֶת: 7 שׁׁד־רְשָׁעִים יְגוֹרֵם כִּי מֵאֲנוּ לַעֲשְׂוֹת מִשְׁפֶּט: 8 הַפַּרְפַּׁד דֶרֶד אֵישׁ יוֹזֶר יְוֹזֶד יָשֶׁר פְּעֵלוֹ: 9 טוֹב לְשָׁבֶת עַל־פִּנַּת־גֶּג מֵאֵשֶׁת ימְדוֹנִים' וּבֵית חֶבֶר: 10 גֶפָשׁ רָשָׁע

various translations. Among the Greek MSS that add this verse (from  $\theta'$ ), some negate  $\gamma \varkappa$  by  $\mu \eta$ , reflecting the tradition of reversing the verse's meaning.

V: (a) cogitationes robusti semper in abundantia (b) omnis autem piger semper in egestate. "(a) The thoughts of the industrious are always in abundance, (b) but every sluggard is always in want." By rendering איץ its antonym, "sluggish," V sharpens the contrast with דרוץ ("diligent") and does not condemn industry. However, V does not consistently refrain from warning against "hurrying."

S translates المدحيك as مدحيك ("and of the wicked"); cf. 28:20. T follows M.

As emended, the verse reads, פּעַל אוצרות בלשון שקר הבל רֹדֵף , "He who earns treasures by a deceitful tongue pursues a vapor [or 'vanity'] on deadly snares." He is not only pursuing something ephemeral and trivial but is risking his life to do so.

M reads, "(a) The making of treasures by a lying tongue (b) is a driven vapor [or 'vanity'] which seek [pl, *sic*] death [הבל נדף מבקשי־מות]." This is awkward for three reasons: (1) An action ("making," לפעל (a driven vapor") and to *seek* something (death). These things would be more meaningfully predicated of a person. (2) The plural מבקשי is not coordinated with the singular (3) The image of a passive, helpless driven breeze does not well describe the apparent referent, namely people who actively seek death. The proverb as emended is better syntactically and has a more fitting metaphor.

The reading במקשי or במקשי has some support in Rashi, who at least uses these words exegetically when he says, "These things [mentioned in 21:6a] are snares of death for him [מוקשי מות לו]." The variants appear in several MSS KR. Perhaps Rashi was familiar with these readings and incorporated them exegetically.

G: (a) δ ἐνεργῶν θησαυρίσματα γλώσσῃ ψευδεῖ (b) μάταια διώκει ἐπὶ παγίδας θανάτου. "(a) He who earns treasures by a deceitful tongue (b) pursues vanities on snares of death." G reflects several variants, as noted above. Most significantly, G reads "במקש" ("snares of"), a reading suggested by Hitzig, Toy, etc. and found in KR<sup>245pm</sup>; cf. 13:14b. V ad laqueos mortis ("on the snares of death") = במקשי מות –

## 21:8 אזר S (הבישה) V (aliena est) ] א וור M (near dittog $t \rightarrow t$ )

M is obscure. The best that can be done without emendation is to translate: "Changeable [or 'perverse' or 'tortuous'] is the way of a man and strange; but the pure man—his behavior is honest." But this has the first stich condemning the "way" of a man without saying what kind of man he is. Various expediencies to deal with this verse, none of them satisfactory, are discussed in ABP. The versions naturally struggle to make sense of this verse and can provide only uncertain evidence for variants.

Emending דר סו וזר (= G, V, S), the proverb reads, "A strange man is changeable [or 'tortuous'], but the pure man—his behavior is straight." (דו and אישר are synonymous virtues in 20:11.) אישר would have arisen by near dittog  $t \rightarrow t1$  or by anticipation of the 1 in the following word. (As for the supposed root t1", see Ehrlich and BHQ.)

G: (a) πρòς τοὺς σκολιοὺς σκολιὰς ὁδοὺς ἀποστέλλει ὁ θεός (b) ἁγνὰ γὰρ καὶ ὀρθὰ τὰ ἔργα αὐτοῦ. "(a) God sends crooked ways to the crooked [pl], (b) for pure and straight are his works." G reworks the obscure sentence, adding "God sends" and redoing the syntax. It is difficult to see what variants, if any, underlie it.

V: (a) perversa via viri aliena est (b) qui autem mundus est rectum opus eius. "(a) The perverted way of a man is strange, but as for him who is pure, his work is right." V may be reading או ווה the first stich, but the evidence is uncertain because it is already loose in its treatment of the syntax in treating הפכפך as an attributive adjective.

S: אויבה אויעה אבואה אויעה אויעה אויעה אויעה אויעה אויער (a) He who changes his ways is an alien man, (b) but he who is pure, his works are righteous." S, while staying close to the Hebrew, does not represent the 1 of אוור ווור Likewise V aliena est ("is strange").

## 21:9 מדונים M<sup>RB-KMSS</sup> (dial? norm) מדינִים M<sup>ALRB-QMSS</sup> (dial? norm)

G: (a) אָרָבוֹססט סוֹאבוֹע בָאוֹ אָשעוֹמג טָׁדָמוֹלאָסט (b) אן בֿע אַבאַסטוּמעָבעסוג עָבדע מֿטֿאנוֹמג אַמוֹ בֿע סוֹאבע אָסויאָשָ. "(a) It is better to dwell in a corner in open air (b) than in plastered (rooms) with injustice and in a common house." In G as it stands, the translator has simply replaced מאשת מדינים —a

אַוְּתָה־רֶע לֹא־יָחֵן 'בְּעֵינֵי' רֵעֵהוּ: 11 'בַּעֲנָשׁ'-לֵץ יֶחְכַּם־פֶּתִי וּבְהַשְׂכִּיל לְחָכָׁם יִקַּח־דֵּעַת: 12 מַשְׂבִּיל אֲדִיק לְבֵית רָשֶׁע מְסַלֵּף רְשָׁעֵים לְרֵע: 13 אַטֵם אָזְנוֹ מזַיְעַקַת־דֶּל גַּם־הָוּא 'יִקְרָא וְלָא יֵעָנֶה: 14 מַתָּן בַּסֵּעֶר 'יְכַפֵּר'י-אָף וְשָׁחַד בַּׁחֵק חֵמֶה עַזֶה: 15 שִׂמְחָה

phrase he understands and translates literally in 25:24—with a metaphor representing comfortable, dry plastered rooms. κεκονιαμένοις plays on κοινῷ at the end of the verse. G thus intensifies the good thing (living in plastered rooms with company) that is outweighed by the bad thing (injustice), making the bad thing all the worse by comparison. In G, the proverb complains about injustice rather than unpleasantness. G 21:9 is a deliberately manipulated rendering of a proverb that was understood and translated literally in 25:24. The manipulation reduces the attention given to irascible women, as happens also in 19:13.

The manuscripts differ in their handling of the *ketiv-qere* here, but both forms of the pair are attested in various ways. On this pair see the comment on 6:14. For the sake of consistency I use מדונים in the critical text; though it is well-attested as a *ketiv*, it must be regarded as an emendation vis-à-vis the copy text.

21:10 אותה רע M ] > G S (homoi ע∕ע)

יחן M G (לאנחאלקסבדמו) ] איחז S (הראנעראס) (graph  $1 \rightarrow t)$ 

M has "his neighbor will not be forgiven in his eyes [בעיניו]," but the relationship between the wicked man's appetite for evil and his refusal to forgive or absolve others is tenuous. Reading "in the eyes of" connects the two lines better, with the second stating the consequence of the evil desires mentioned in the first.

G: ψυχὴ ἀσεβοῦς οὐκ ἐλεηθήσεται ὑπ' οὐδενὸς τῶν ἀνθρώπων. "The soul of the impious [sg] shall not receive mercy from any men." G omits אותה apparently a homoioteleuton  $\mathcal{V}$ . ὑπ' οὐδενὸς τῶν ἀνθρώπων does not represent the possessive suffix of בעיניו and indirectly testifies to = S.

V: (a) *anima impii desiderat malum*, (b) *non miserebitur proximo suo*. "(a) The soul of an impious man desires evil; he will not pity his neighbor." V has active meaning despite the deponent verb to mimic the passive voice of M's H-stem יָחָן V includes no equivalent for בעיניו (CB).

## 21:11 בַּעָנָשׁ־ M<sup>A</sup>Y ] בַּעָנָשׁ־ M<sup>L</sup> (vocal)

לחכם [ M לחכם ( $\phi\phi\phi s$ ) אחכם (haplog ל $\phi \to \delta$ ) (haplog ל

## נות [ M לבית 1:12 אלבות [ M לבית 1:12 מלבות [ M לבית 1:12 אלבות [ א לבית 1:12 מור י → 1]

G: (a) סטענו δί×αιος ×αρδίας ἀσεβῶν (b) ×αὶ φαυλίζει ἀσεβεῖς ἐν ×α×οῖς. "(a) A righteous man observes the hearts of the impious [pl] (b) and despises the impious [pl] in evils." That is, he holds them in contempt when they fall into misfortune; compare Lady Wisdom's behavior in 1:26–27. ×αρδίας = לבות (there would be no reason for G to translate "house" as "hearts") thereby producing an awkward locution ("hearts of the wicked man") which then required construing עיונייי ("wicked") as a collective and putting it in the plural (ἀσεβῶν). The reading man's heart, as in Prov 15:11; 17:3; 24:12; and Ps 7:10b. The proto-G scribe probably intended ψιτις ("hearts of wickedness"), on the analogy of אוצרות (Prov 10:2) and similar locutions elsewhere (e.g., Mic 6:10, 11; Job 34:8), but the G translator assumed the vocalization  $\psi$ , as happened in Prov 10:2.

S: אמאבל גרמס, גבמאה הרמס, גבמאה ארמס, "(a) The righteous man observes the hearts of the evildoers, (b) and he repels the evildoers to evil." Since S shows independence of G in 21:12b, its use of "hearts" in 21:12a is also likely to be independent.

21:14 ייְכְפֶה־ [ אי מעמדף א מים א ייַ מון א מעמדף (graph  $\to \to H \approx G$  (מימדף (graph  $\to \to H \approx G$ 

M's כפה' is not attested elsewhere in a sense that would fit this context. BH כפ"ך and RH כפ"ה mean "bend," "force," which is not relevant here, and it is something of a stretch to give it an otherwise unattested sense of "subdue" (HALOT). We should read יכפר ("assuages"); cf. ABP. Prov 6:35; 16:14; and Gen 32:21 use forms of כפ"ר to refer to a payment that (potentially) assuages wrath.

G: (a) δόσις λάθριος ἀνατρέπει ὀργάς, (b) δώρων δὲ ὁ φειδόμενος θυμὸν

ַלַצַּדִּיק עֲשָׂוֹת מִשְׁפֵּט וֹמְחִתָּה לְפַּעֲלֵי אֶוָז: 16 אָדָם הַוֹעָה מִדֶּרֶד הַשְׂפֵּל בִּקְהַל רְפָאִים יָנְוּחַ: 17 אַישׁ מֲחְסוֹר אֹהֵב שִׂמְחֶה אֹהֵב יַיִז־ וְשָׁמֶז לְא יַעֲשִׁיר: 18 כְּפֶר לַצַדִּיק רָשֶׁע וְתַחַת יְשָׁרִים בּוֹגֵד: 19 טוֹב שֶׁבֶת בְּאֶרֶץ־מִדְבֶּר מֵאֵשֶׁת <sup>ר</sup>מְדוֹנִים<sup>-</sup> וְבֶעַס: 20 אוֹצֵר | נֶחְמָד

έγείρει ἰσχυρόν. "(a) A secret gift turns away wrath [pl], (b) and he who refrains from gifts stirs up strong anger." G takes the meaning of  $\neg centre cen$ 

S has הפוסה ("redeem" or "remove"), apparently a guess, for יכפה does not render כפר or כפה elsewhere.)

Some versions have "extinguish":  $\sigma' \sigma\beta \acute{e}\sigma \epsilon$ , T  $\alpha \tau \forall c x tinguet$ . The late versions are unlikely to share variants independently of the major witnesses. They or their source texts were probably influenced by 26:20 and assumed that the *hapax*  $\tau c \epsilon$  has the same meaning as  $\tau c \epsilon$ . The medieval interpreters, as well as Gemser, HALOT 492b, and some other modern scholars share this assumption.

## 21:15

G: (a) εὐφροσύνη δικαίων ποιεῖν κρίμα, (b) ὅσιος δὲ ἀκάθαρτος παρὰ κακούργοις. "(a) The joy of the just is to do justice, (b) but the pious man is (reckoned) unclean by evildoers." G 21:15b is distant from M, though the Hebrew sentence is clear and is translated accurately in 10:29b. There is no religious or literary motivation for the shift. G 21:15 appears to be a variant proverb, created by substituting a new second stich. Whether it was created in Hebrew or Greek cannot be determined. The proverb's assertion resembles 29:10a, claiming that the evil hate the righteous.

#### 21:16

G's δικαιοσύνης ("righteousness") for השכל ("insight") is in line with G's tendency to make moral lessons more explicit.

21:17

G: (a) ἀνὴρ ἐνδεὴς ἀγαπῷ εὐφροσύνην (b) φιλῶν οἶνον καὶ ἔλαιον εἰς πλοῦτον. "(a) A needy man delights in merriment, (b) loving wine and oil in abundance [lit. 'to/as wealth']." G misses the point of the Hebrew proverb, which is that a man who loves pleasures will end up poor, and instead creates a statement that the poor man loves merriment. On the face of it, G's εἰς πλοῦτον looks like a representation of <code>, לעשִר לֹא לֹא לֹישִיר</code>, but it is hard to account for the loss of the <code>א</code> and <code>'</code>. Some other G MSS read ὀυ πλουτήσει, which Baumgartner and de Lagarde prefer, but SyrH ascribes that reading (in Syriac Loop'.

V translates שמחה as *epulas* ("feasts"), a correct interpretation, which is in line with the rabbinic understanding of שמחה as feasting; see C. Gordon 1930, 393.

```
21:18b M S ] > G (unc)
```

There is no evident reason for the absence of 21:18b in G. It is present in V S T  $\alpha\lambda'$ .

## 21:19 מִדְיָנֵים [ M<sup>K</sup> ] מִדְיָנֵים (dial? norm)

G: (a) κρεῖσσον οἰκεῖν ἐν γῆ ἐρήμῷ (b) ἢ μετὰ γυναικὸς μαχίμου καὶ γλωσσώδους καὶ ὀργίλου. "(a) Better to dwell in a desert land (b) than with a quarrelsome and garrulous and irascible woman." μαχίμου καὶ γλωσσώδους is a double rendering of מדינים (αττίτα), giving a clearer picture of the woman's failings. In 21:19b, CVO is treated as an adjectival genitive so as to make it clear that her anger is one of her personality traits and not the anger she *causes*.

On this ketiv-qere pair, see the comment on 6:14.

# 21:20 (מעמדמט (graph ' $\rightarrow$ ו, lig (מעמדמט (קרבון מעמדמ)) (graph ' $\rightarrow$ ו, lig (מ $\leftarrow \subset I$ , lig (מ $\leftarrow \subset I$ , lig (מ $\leftarrow \subset I$ ) א הבנוה M S (כגני) (לבענוה (לבנוה (לבנוה))) א בנוה

M 21:20a is awkward, reading, "A delightful treasure and oil is in the house of the wise," as if "treasure" and "oil" were distinct and commensurate goods. The correct text can be restored by combining G and M: אוֹצָר נֶחְמָד יִשְׁכוֹן בִּנְוֶה חְכָם וּרְסִיל אָדָם יְבַלְעָנוּ dwells in the habitation of the wise, but a foolish man devours it"). That is to say, he devours his family's wealth, quickly and improvidently.

 יִּשְׁכּוֹן בְּנְוָה חָבֶם וּכְסָיל אָדָם יְבַלְּעֵנּוּ: 12 וְבַדָּ צְדָקָה וָחָסֶד יִמְצָא חַיִּיִם יּ וְכָבְוֹד: 22 עֵיר גָּבּּרִים עָלֵה חָבֶם וַיֹּדֶר עָז מִבְטָחָה: 23 שׁמֵר פִּיו וּלְשׁוֹגִוֹ שׁמֵר מִצְרָוֹת נַפְשָׁוֹ: 24 זַד יְהִיר לֵץ שָׁמֵוֹ עוֹשֶׁה בְּעָבְרַת זְדוֹן: 25 תַּאֲוַת עָצֵל הְמִיתֶנּוּ בְּי־מֵאָוּוּ יָדָיו לַעֲשְׁוֹת: 26 כָּל־ הַיוֹם הִתְאַוָּה תַאֲוָה וְצַדִּיק יִׁתֵּן וְלָא יַחְשְׂדָ: 27 זֶבַח רֲשָׁעִים תּוֹעֵבָה אַף כִּי־בָזָאָני זִדוֹן: 28 עַד־בְּזָבִים יאַבֵּר וְאָישׁ שׁוֹמַע לָגָצַח יְדַבָּר:

of the left vertical (Baumgartner), and the reverse fusion is equally possible. This variant is correct. G 21:20a says that the wise man has a "treasure" of wise things to say. This makes sense, but G 21:20b (with "on the mouth of") is problematic, because it seems to assert that the fool will swallow and incorporate the wise man's words, which is exactly what a fool will *not* do.  $\dot{\epsilon}\pi\dot{\epsilon}$   $\sigma\tau\dot{\delta}\mu\alpha\tau\delta\varsigma = 2$  or an ungrammatical אולה שבר. M's בנוה sin the habitation" is correct. A ligature can resemble a b in several early square scripts. Though בפה is the proper form of the construct, and is preferred here as being closer graphically to בנוה is an error.

G's אָרָרָאָ arose before the use of final letters, by metathesis plus  $\mathfrak{D} \rightarrow \mathfrak{I}$ , which could occur before or after the development of final letters. G lacks M's second אַרְרָאָ, which looks like an accidental repetition from the first stich (Baumgartner). Some V<sup>MSS</sup>, presumably earlier in this regard, lack *et iustitiam*. "Righteousness" properly belongs only in the first stich, among the virtues, not in the second, among the rewards; see ABP.

#### 21:22

G: (a) πόλεις ὀχυρὰς ἐπέβη σοφὸς (b) καὶ καθεῖλεν τὸ ὀχύρωμα, ἐφ' ῷ ἐπεποίθεισαν οἱ ἀσεβεῖς. "(a) A wise man went up against strong cities (b) and destroyed the fortress upon which the impious had relied." G adds "the impious" to characterize the defeated city, to show that its defeat was just. Otherwise (the translator may reason) why would a wise man attack it? The plural in 21:22a generalizes the application of the proverbs, so that the singular "fortress" in 21:22b is to be understood as a generality, too.

## 21:23

G represents מצרות as a singular. But there is considerable flexibility with grammatical number in the versions (see the previous verse) and, moreover, moreover, is intended as a collective that can easily be represented by the singular. Four MSS KR have מצרה, perhaps for the same reason as the versions. But just possibly they preserve an old variant.

## 21:24

G: (a) θρασύς καὶ αὐθάδης καὶ ἀλαζών λοιμὸς καλεῖται. (b) ὅς δὲ μνησικακεῖ, παράνομος. "(a) The brash and arrogant and insolent man is called a pest, (b) and he who bears grudges, a criminal." In 21:24a, G compounds the offenses of the pest by adding ἀλαζών. In 21:24b, the notion of grudges is derived from  $\eta$  wrath."

יתן איחן א א א א החב <br/> M S (בתר 'G (לאבמָ אמ' סיאדוֹ<br/> איר א יחן (graph  $\to \to , \, dbl)$ 

G: (a) ἀσεβὴς ἐπιθυμεῖ ὅλην τὴν ἡμέραν ἐπιθυμίας κακάς, (b) ὁ δὲ δίκαιος ἐλεῷ καὶ οἰκτίρει ἀφειδῶς. "(a) The impious man desires bad things [lit. 'desires bad desires'] the entire day, (b) but the righteous man has mercy and shows compassion unsparingly." G changes the subject from the sluggard (which in M is to be supplied from 21:25) to the wicked man (see the comment on 10:5) and moreover adds κακάς "bad [sc. desires]." By moralizing the first stich, G creates a clear antithesis to the second. On the difficulties in M, see ABP. G 21:26b has a double translation of μη, where M has μη. (For ΠΠ interchanges, see LSF §106de.) Both έλεεῖν and οἰκτίρειν are frequent renderings of J<sup>T</sup>.

21:27 תועבה M S (גרא יהוה [ (גרא יהוה א G (βδέλυγμα κυρίω) (explic)

G βδέλυγμα κυρίω ("abomination to the Lord") = תועבת יהוה, as in 15:8. Note the inner-Hebrew variants תועבה מושה in 15:8 (see comment there). In the present verse G adds "the Lord" for specificity.

S: An inner-Syriac variant with מות מו at the end of the first stich is a later addition adjusting to G.

## 21:28 לנצח M ] לנצר G ( $\phi$ υλασσόμενος) (graph $\Pi \rightarrow \eta$ )

G: (a) μάρτυς ψευδής ἀπολεῖται, (b) ἀνήρ δὲ ὑπήκοος φυλασσόμενος λαλήσει. "(a) A lying witness will perish, (b) but an obedient man will speak cautiously." φυλασσόμενος (lit. "guarding himself") = לְנָצֹר (de Lagarde ref Cappelli 1650).

V (21:28b): vir oboediens loquitur victoriam ("an obedient man shall speak of victory"); cf. α', σ', θ': ἐις νῖχος πορεύσεται ("will go to victory").

29 הַעֵּׁז אָישׁ רָשֶׁע בְּפָנְיו וְיָשָׁר הָוּא ו רְיָבִׁין דַרְכְּוֹז: 30 אֵין חֲכְמָה וְאֵין תְּבוּנֶה וְאֵין עֵׁצָּה לְנֶגֶד יְהוֶה: מּ זוֹנ סוּס מוּכָן לְיוֹם מִלְחָמֶה וְלֵיהוֹה הַהְּשׁוּעֵה: 1:22 נִבְחָר שֵׁם מֵעַשׁר רֶב מִכֶּסֶף וֹמִין מְלַחָמֶה וְלֵיהוֹה הַהְּשׁוּעֵה: 1:22 נִבְחָר שֵׁם מֵעַשׁר רֶב מִכֶּסֶף וֹמִין מְלַחָמֶה וְלֵיהוֹה הַהְּשׁוּעֵה: 1:22 נִבְחָר שֵׁם מֵעַשׁר רֶב מִכֶּסֶף וֹמִין מְרָהוֹה הַהְּשׁוּעֵה: 1:22 נִבְחָר שֵׁם מֵעַשׁר רֶב מִכֶּסֶף וֹמוּקוֹה הַהְּשׁוּעֵה: 1:22 נִבְחָר שֵׁם מֵעַשׁר רֶב מִכֶּסֶף וֹמִזְהָב חֵן טְוֹב: 2 עָשִׁיר וְרָשׁ נִפְגָשׁוּ עֹשֵׁה כֵּלֶם יְהוֶה: 3 עָרָוּם וּמִזּהָהָב חֵזְ מוֹב: 1 מָזָה הַהְשׁוּעָה: 1:22 נִבְחָר שֵׁם מֵעַשׁר הַב מְנָשָׁר הָב מְכָסֶף וֹמוּז הַיִזּה הַתְּשָׁר וְבָב מַבֶּסֶף וֹזְהוֹה: 3 עָשָׁה כָלָם יְהוֶה: 3 עָקוּה וּ הַתְּהוֹה הַתְּמָה וְמָיוּ שְׁרִים וּזְרָהוּה: 1 נְתַשָּה כָלָם יְהוֹה: 3 עָבוּה יִהוֹה הַתָּהוּ הְשָׁר וּמָזּהָב חֵן טְשָׁר וּבְבוּה וּבָיוּ הַיָּהוּה בְעָה וּזָהוּה: 3 עָשָׁר וְבָבוּה יְהָאַת יְהוֶה לָאָה רָעָה וּיַהוּה: 3 שְׁכָנוּים שָּבְרָוּ וְנָשֶנִּשׁוּי 1 גַשָּה כָלָם יְהוֹחֵה: 3 עִקּה הַעָּרוּהוּה שָּבוּה יִרְאַת יְהוָה הַתָּהוּה הַתָּהוּה הַהַיָּרוּים שָּבְרָוּ ווֶטְשָׁר הַבְרָהוּ הַיָּהוּה הַיְהוּה הַי הַעָּשָׁר וְכָבוּה וּחַיִים: 5 אַנִים שָּחִים בְּדֶרָהוֹ גַם כִייזין לָא־יָסוּוּר מַמָּרוּ הַבּהוּה.

S's אויבאיה ("truthfully") = לנצח שואים. Perhaps the translator thought that לנצח meant "forever," and finding it strange that a "listening" man would speak *forever*, he substituted an appropriate moral quality.

21:29 יבין  $M^Q$  G (סטענו) ] יבין  $M^K$  S (כאש) מ' ס' (<br/>אנסטן לדטעמסט) V (corrigit) T (מתקן) (graph  $\rightarrow \rightarrow$  )

דרכו  $M^Q$  ] דרכיו  $M^K$  (num)

On בי"ז/כו" איז variants see the comment on 18:15. The *qere* בי"ז/כו" is earlier and correct. In Proverbs (but not in Exodus or Psalms), when דרך is the object of words for understanding, the singular is used; cf. 14:8; 16:9; 20:24. Though the *qere* points it as singular, the *ketiv* writes the word as a plural. Usually it is the *qere* that supplies the *plene* writing of the plural, ". G and S have the plural, but that could be their interpretation of either orthography. For a similar case, see 16:27.

#### 21:30

For לנגד יהוה, G has πρὸς τὸν ἀσεβῆ ("to the impious"). πρός + accusative can mean "against"; hence BAP translates, "... il n'y a pas de conseil à opposer à l'impie!" This would mean that no counsel can (successfully) oppose the ungodly, which is a peculiar sentiment. More likely, πρός here means "to," "toward." G means that no counsel comes *to* the wicked when they most need it (see 1:28; ABP).

S has איס ("like the Lord's"), correctly understanding לנגד as "vis-à-vis," hence, "in comparison with."

21:31 התשועה  $M^{Oc} M^{OrQ}$  התשיעה M<sup>OrK</sup> (graph  $i \rightarrow i$ )

The former is the form of the word attested elsewhere.

22:1

G: ὄνομα καλόν, adding "good" to "name" for moral clarity. This is what "name" implies in the original. Similarly T שמא טבה.

22:3 אויסתר MK  $\approx$  G (דוושסטעוביטי) MK (קתר MK  $\approx$  G (דוושסטע איז MQG (דווסתר (gram, equal)

The *ketiv* is to be vocalized as an imperfect with *waw*-conversive. I take the verse to be in the form of a "mini-narrative," on which see APB 2.531.

G: (a) πανοῦργος ἰδὼν πονηρὸν τιμωρούμενον κραταιῶς αὐτὸς παιδεύεται, (b) οἱ δὲ ἄφρονες παρελθόντες ἐζημιώθησαν. "(a) When a clever man sees a bad man being punished severely, he takes instruction (b) but when the fools pass by they are penalized." The same proverb is handled quite differently in 27:12. There, ¬ע¬ is understood as evil deeds from whose danger the clever man *does* hide. Here it is understood as bad punishment (administered to another), from which hiding would be unnecessary. G associated רעה (*ketiv*) with 'o' ("to discipline," "instruct") and translated it twice, as τιμωρούμενον and παιδεύεται. κραταιῶς is a second translation of ¬V. Renderings elsewhere show that the translator understood the vocabulary involved and has chosen to provide a midrashic-type expansion that explores the verse's semantic potential.

## 22:4

G: (a) γενεὰ σοφίας φόβος χυρίου, (b) = M. "(a) The produce of wisdom is the fear of the Lord." עקב is understood as "consequence" or "reward," hence γενεά. By changing ענוה ("humility") to σοφίας, the translator or a Hebrew scribe associated this verse with Wisdom's promises in 8:18 and 35.

#### 22:5

צנים פחים: The lack of a conjunction between the two nouns of the subject is possible (e.g., Gen 37:27) but awkward, so the fact that the versions supply it (as in G's τρίβολοι καλ παγίδες) does not prove that they had it in their source text.

## **22:6** M S ] > G (unc)

V: (a) proverbium est adulescens iuxta viam suam, (b) etiam cum senuerit non recedet ab ea. "(a) It is a proverb: A young man according to his way, (b) and even when he is old, he will not depart from it." V means that a young man does as he is inclined and persists in this for the rest of his life. Proverbium est for Hebrew  $\exists rtr}$  is puzzling. Perhaps Jerome

ַעֲשִׁיר בְּרָשֵׁים יִמְשֵׁוֹל וְעֶבֶד עֶׁוֹה לְאַישׁ מַלְוֶה: 8 זוֹרַעַ עַוְלָה 7 אֲשִׁיר בְּרָשֵׁים יִמְשֶׁוֹל וְעֶבֶד עָׁוֹה לָאַישׁ מַלְוֶה: 7 יִיְקָצָר<sup>ז</sup>-אָוָן וְשֵׁבֶט עָבְרָתוֹ יִכְלֶה: 9 טְוֹב־עַיִן הוּא יְבֹרֶדְ בְּי־נָתָן

identifies the verse as a "proverb," in this case meaning a folk proverb, somewhat dismissively, because he is not entirely happy with its sentiment. The verse might be read to imply that people act according to their nature, not their education. Jerome greatly valued education, as seen in his *Epistle* 107 on the education of women (CB).

S העד ("he who rebukes") implicitly vocalizes הנך and interprets it as the instructive action commonly recommended in Proverbs.

22:7

G: (a) πλούσιοι πτωχῶν ἄρξουσιν, (b) καὶ οἰκέται ἰδίοις δεσπόταις δανιοῦσιν. "(a) The rich will rule the poor, (b) and servants will lend to their own masters." Mistaking עבד as the subject of לוה, and recognizing that the verse speaks of reversal of social roles, G treats different as "lend" and identifies איש מלוה as the master, on the assumption that the two types are opposites, like the pair in 22:7a. In this way, G shapes the verse into a "world turned upside down" saying, on which see ABP on 30:21–23.

S: ممل مرمص عمدیک مراه لحج تحمد محمد المحدی محمد المحید. "(a) As for the rich man, the poor man will rule over him, (b) and the slave will lend to him who used to lend to him." S, influenced by G, turns the Hebrew into a promise of the eventual domination of the poor over the rich. This is an eschatological expectation. Compare the praise of the poor in S 28:6, and also S's understanding of Ps 37:11: "the *poor* (حتفدیک) will inherit the earth"; similarly Luke 6:20.

22:8 יִקְצָור [<יקצר M (vocal)

עברתו M S (אבאס M S (אבאס) ] א עברתו (קאנאס M S (אבאס) (graph  $\rightarrow \rightarrow \gamma$ 

+ 2 stichoi G (elab) G 22:8a = M 22:8a; G 28:8b ≈ M 22:8b; G 22:8aa ≈ M 22:9a; G 22:8ab ≈ M 22:8b; G 22:9a ≈ M 22:9a; G 22:9b = M 22:9b.

 $Mp^{L}$  confirms the consonants of M's יתיר by designating it 'יתיר ("having a superfluous *waw*"). Contrary to BHS and some other editions, there is no *qere* here, though the Mp note serves as a virtual *qere*. (Mp<sup>Y</sup> has no note.) I use the *defectiva* writing, in accord with Tiberian practices. It seems that a scribe who preferred plene writings was active in this verse and in 22:11, 14.

G: (a) ὁ σπείρων φαῦλα θερίσει κακά, (b) πληγὴν δὲ ἔργων αὐτοῦ

סטעדבאניסבו. "(a) He who sows worthlessness will harvest evils, (b) and the rod will end his works." πληγή ("blow") can correspond to "rod" (Prov 29:15; Isa 30:31; thus Carmignac 1980, 34, who translates here: "celui qui a semé l'injustice moissonnera l'inanité, et le bâton terminera ses oeuvres"). ἔργων αὐτοῦ = עבדתו , understood as plural (עֵבֹלְהָ ). συντελέσει = יָכֵלֶה

#### 22:8a

G: (a) ἄνδρα ίλαρὸν καὶ δότην εὐλογεῖ ὁ θεός, (b) ματαιότητα δὲ ἔργων αὐτοῦ συντελέσει. "(a) God will bless a cheerful and gift-giving man, (b) and he will make an end to the vanities of his works." Carmignac (1980, 36) says that the Hebrew corresponding to G 22:8*a*b was lost by homoioteleuton due to the identical endings of 22:8b (יבלה) and 22:8ab (presumably יבלה). Carmignac restores the source text of G 22:8aa to does not have the required איש רוצה ונותן יברך אלהים. meaning. Moreover, G 22:8ab is an ill-fitting match for G 22:8aa. (G 22:8*a*b cannot mean "et il suppléera à la faiblesse de ses oeuvres," as Carmignac translates it [1980, 40].) According to de Lagarde and Baumgartner, 22:8*a*b is a second translation of a variant of M 22:8b, and G 22:9a is an interpolation. More likely, G 22:8aa corresponds to M 22:9a, while G 22:8*a*b is a deliberate reformulation of M 22:8b. OG was probably G 22:8a (M 22:8a), G 22:8ab (M 22:8b), G 22:9a (M 22:9a), and G 22:9b (M 22:9b). Recensional additions are G 22:8b (≈ M 22:8b) and G 22:8aa ( $\approx$  M 22:9a), though G 22:8b is based on a Hebrew text not identical to M. (ABP offers a different explanation of G 22:8a, but in fact no variants are indicated.)

fin ] + 2 stichoi G (9a) (elab)

G: (a) ό ἐλεῶν πτωχὸν αὐτὸς διατραφήσεται, (b) τῶν γὰρ ἑαυτοῦ ἄρτων ἔδωκεν τῷ πτωχῷ. "(a) He who has mercy on a poor (man), he (too) will be supported, (b) for he gave of his bread to the poor." G 22:9 ≈ M 22:9. πτωχόν = 𝔅μ for μ 𝔅 (Giese 1993a, 114). 𝔅 is understood (ungrammatically) as "good to the poor."

#### 22:9a

G: (a) νίκην καὶ τιμὴν περιποιεῖται ὁ δῶρα δούς, (b) τὴν μέντοι ψυχὴν ἀφαιρεῖται τῶν κεκτημένων. "(a) He who gives gifts secures victory and honor, (b) but he takes away the life of (their) possessors [or 'but they (sc. the gifts) take away the life of (their) possessors']." G 22:9*a*a is associ-

## מִלַּחְמָוֹ לַדֵּל: 10 גְּרֵשׁ לֵץ וְיֵצֵּא מָדְוֹן וְׁיִשְׁבּّת הַין וְקָלְוֹן: 11 אֹהָב רְּטְהָר<sup>ז</sup>-לָב תֵן שְׁפָתָׁיו רֵעֵהוּ מֶלֶדְ: 12 עֵינֵי יֵהוָה נָצְרוּ דֻעַת וֹיְסַלֵּף הְבְרֵי בֹגֵד: 13 אָמַר אָצֵל אֲרֵי בַתָּוּץ בְּתִוֹדְ וְׁרוֹבוֹת אֵרָצֵחַ:

ated with the gift-giving man of 22:8*a*a and is an elaboration of that line (as is 22:9). The two lines of G 22:9*a* are not just fragments (contrary to Baumgartner and de Lagarde) but constitute a new couplet, one that takes the theme of gift giving in a new direction. It says that although gifts (bribes) bring victory (in court, perhaps) and also honor (i.e., prestige), they ultimately destroy their recipients.

G 22:9*a*b recalls G 1:19b.The unhebraic syntax of G 22:9*a*b, with the verb inside the genitive phrase, suggests that the couplet was composed in Greek, not Hebrew. Still, G 22:9*a*b is in the spirit of M (not G) 15:27b, indicating that the couplet is the work of the original translator. V has this addition.

#### 22:10

G: (a) ἔχβαλε ἐχ συνεδρίου λοιμόν, καὶ συνεξελεύσεται αὐτῷ νεῖκος, (b) ὅταν γὰρ καθίσῃ ἐν συνεδρίω, πάντας ἀτιμάζει. "(a) Expel a pest from the assembly and conflict will leave with him, (b) for when he sits in the assembly, he dishonors everybody." G atomizes the Hebrew of 22:10b. It extracts the notion of sitting from "שרת", which it parses as a form of "". Next it extracts the notion of the assembly from ", which the translator construes as the setting of judgment. Then it treats jn as a verb. The translator is familiar with these words; hence his introduction of the new idea is deliberate. Still, the idea of litigation is hinted at in M by the use of the word ", which consistently belongs to the legal sphere and means "litigation" or "judgment"; see further ABP. The assembly is of special importance to the translator; see the comment on 31:23.

S: אחסב לביא אין גלא אי הבנה באלא לבלבה ביב (a) Destroy the evil man and remove the contentious one, (b) and make litigation and insult cease, (c) lest, if he sit in the assembly, he insult you all." S first translates M 22:10b then extends it using G, but loosely. The second-person plural address is unusual. 22:11 טהור  $M^Q$  ] טהור  $M^K$  (orth)

רעהו מלך [ M רעהו המלך [ היערהאלץ ארעה המלך ] ארעה המלך ארעהו מלך (הדערהאת, הכלבאה, גכבאה, dittog ה $\to$ הה הה

S: העראה גדבא בארא גדבא בארא גדבא בורת ססליער בל משטאלא גדערסת, אומא אינא גדבא בורת ססליער בל משטאלא גדעראה. "(a) God loves him who is pure in his heart, (b) and he shows mercy to the lips of the friends of the king." S 22:11a adds "God" ( $\approx$  G;  $\approx$  T). For M's רעהו מלך, S reflects רעה המלך, with אין treated as a plural. The was repeated. This may go back to a *defectiva* orthography: רעה בעה המלך. Though S has picked up "God" from G (unless it also had it in its source), S 22:11b is not a permutation of G but derives from a similar Hebrew variant.

22:12 הברי M S (ג (גערים הארים הארים) אדברים (near dittog ב  $\rightarrow = 0$ 

G: (a) = M; (b) φαυλίζει δὲ λόγους παράνομος. "(b) but the transgressor despises words." The forced construction in 22:12b could come about only if the translator read דברים דברי.

22:13 ארצח M ] רצח G (לסטינטדמל) S (הארצה) (near haplog ארצה  $\bigstar \to \bigstar$  paleo  $\bigstar \star \to \bigstar$ 

G: (a) προφασίζεται και λέγει ὀκνηρός, (b) Λέων ἐν ταῖς ὁδοῖς, (c) ἐν δὲ ταῖς πλατείαις φονευταί. "(a) The sluggard makes excuses and says, (b) 'A lion in the streets (c) and murderers in the plazas!" G's additional "makes excuses" explains the sluggard's (rather obvious) intent. G reads רֹצֵח for רֹצֵח, the <code>%</code> having been lost through haplography with the preceding  $\Pi$ , which is very similar in the archaic script (<code><, <'</code>). Since the translator was alert to the fact that the sluggard is making excuses and inventing pretexts, he would have had no reason to avoid the first-person of M. This is evidence that the predecessor of G's source text diverged from

14 שׁוּחָה עֲמֻקָּה פִּי זָרִוֹת זְעָוּם יְהוָה <sup>ר</sup>ִיִפָּל־יּשֶׁם: 15 אָוָּלֶת קְשׁוּרָה בְּלֶב־גְעַר שֵׁבֶט מוּסָר יַרְחִימֻנָּה מִמֶּנּוּ: 16 עַּשֵׁק דָּל לְהַרְבִּוֹת לֶוֹ נֹתֵז לְעָשִׁיר אַדְ־לְמַחְסוֹר: 17 <sup>ר</sup>ִדְבְרֵי חֵכָמִים הַט אָזְנְדָ וּשְׁמַע דְּבָרִיי

proto-M quite early, when the ancestral text was in the archaic script, and underwent further permutations in the square script.

**22:14 יפול** [ M<sup>K</sup> (orth)

fin ] + 3 stichoi G (14a) (elab)

22:14a

G: (a) εἰσἰν ὁδοὶ κακαὶ ἐνώπιον ἀνδρός, (b) καὶ οὐκ ἀγαπặ τοῦ ἀποστρέψαι ἀπ' αὐτῶν. (c) ἀποστρέφειν δὲ δεῖ ἀπὸ ὁδοῦ σκολιᾶς καὶ κακῆς. "(a) There are evil ways before a man, (b) and (yet) he does not like to turn away from them. (c) But it is necessary to turn away from a crooked and evil way." This is a prosaic, moralizing elaboration on the Hebrew of 22:14. Trint is understood as strange things, hence ἑδοὶ κακαί. The added verse also emphasizes that it is a man's own obduracy, and not an arbitrary curse, that brings a man to ruin.

#### 22:15

G: (a) ਕੈਂvoia ἐξῆπται καρδίας νέου, (b) ῥάβδος δὲ καὶ παιδεία μακρὰν ἀπ' αὐτοῦ. "(a) Ignorance lays hold of the heart of a youth, (b) and the

rod and instruction are far from him." G treats ירחיקנה as an adjective, μάκραν. Apparently the second stich gives the reason for the first.

### 22:16

G: (a) δ συκοφαντῶν πένητα πολλὰ ποιεῖ τὰ ἑαυτοῦ [G<sup>BS\*</sup>; var: + κακά G<sup>ScAMSS</sup>], (b) δίδωσιν δὲ πλουσίω ἐπ' ἐλάσσονι. "(a) He who oppresses the wretched increases his own [var: + 'troubles'] (b) and gives to the rich unto lack [SD: 'er gibt einem Reichen auf Kosten eines Geringeren']." The verse is difficult in both M and G. Some G manuscripts try to clarify it by adding κακά, making the verse say that he who tries to defraud the poor will end up himself destitute. Without the moralizing κακά the verse expresses indignation against the man who brings false charges against the poor and, consequently, cheats them in favor of the wealthy. S + σρkmap = G<sup>A</sup>.

22:17 דברי דברי אונד ושמע דברי G (λόγοις σοφῶν παράβαλλε σὸν οὖς καὶ ἀκουε ἐμὸν λόγον] איב חַכָּמֵים MS (הארא אונד אין גָדָן וְשְׁמַע דִבְרֵי חַכָּמֵים MS (הארא אין גָדָן וִשְׁמַע דִבָרַי חַכָּמֵים) (transp of words, om of ידברי)

Though the mechanisms of the change are unclear, several considerations support the emendation: (1) The restored text provides a title for the unit, whose distinctiveness has been proved by the parallels to Amenemope. (The parallels are extensively explored in ABP on this unit.) (2) The title in 24:23, "These too are of the wise," indicates awareness of the ascription of the foregoing proverbs to "the wise," though a title to that effect is missing in M. (3) Other exordia<sup>5</sup> call for attention to the *speaker's* teachings, not the words of other sages, as M's formulation does. (4) The

<sup>5.</sup> Exordia introduce the discourses of Prov 1–9, which I call "Lectures." An exordium comprises (1) an address to the son or sons; (2) an exhortation to hear and remember the father's teachings; and (3) a motivation that supports the exhortation by extolling the teachings' excellence and value. There are similarly structured instructions elsewhere in Proverbs. See ABP 1.45–46.

ּזְלִבְּדְׁ תְּשִׁית לְדַעְתִּי: 18 בִּי־גָעִים בִּי־תִשְׁמְרֵם בְּבִטְגֶדְ יִבְּנוּ יַחְדָּו עַל־ שְׁפָתֶידְ: 19 לִהְיוֹת בֵּיהוָה מִבְטַחֶדְ הוֹדַעְתִּידְ הַיָּוֹם אַר־אֶתָּה: 20 הַלְא כָתַבְתִי לֲדְ 'שְׁלֹשֵׁים' בְּמִוֹעֵצְוֹת וְדֶעַת: 21 לְהוֹדִיעֲדָׁ לְשִׁט

emendation to "my words" provides a better parallel to "my knowledge."

G: (a) Λόγοις σοφῶν παράβαλλε σὸν οὖς καὶ ἄκουε ἐμὸν λόγον, (b) τὴν δὲ σὴν καρδίαν ἐπίστησον, (c) ἵνα γνῷς ὅτι καλοί εἰσιν. "(a) To the words of the wise direct your ear and hear my word (b) and set your heart, (c) so that you may know that they are beautiful."

In G 22:17a, the phrases "words of the wise" and "incline your ear" are transposed with respect to M. It is likely that G had the transposition in its source text. The reconstructed Hebrew source text of 22:17a (see above) must be construed differently from the way G interpreted it. The sense of the original was "The words of the wise. Incline your ear and hear my words, and set your heart to my knowledge." The reconstructed verse provides the heading of the third part of Proverbs, 22:17–24:22. (See the introduction to this unit in ABP 2.704–7.)

The above emendation, first proposed by Gressmann 1924, 274, is widely accepted. Alex Luc (2000, 253), however, argues that, although a different Hebrew lies behind G 22:17a, it is not a section title. The first line reads: דברי חכמים הט אזנך, which is to be translated, "To the words of the wise direct your ear" (the translation Luc's exegesis calls for). However, דברי חכמים הט אזנך However, דברי חכמים הט אזנך, and the words not provide it directly. Although it is true that G does not intend the words to represent a section heading, the underlying Hebrew does. Also, in 22:17c, G had לדעת or construed לדעת (= M) as an infinitive. It also continued 22:17 into כי נעים in 22:18.

### 22:18

G: (a) καὶ ἐἀν ἐμβάλῃς αὐτοὺς εἰς τὴν καρδίαν σου, (b) εὐφρανοῦσίν σε ἅμα ἐπὶ σοῖς χείλεσιν. "(a) And if you place them into your heart, (b) they will together make you rejoice on your lips." G 22:18 starts with " תשמרם. G substitutes "heart," which is the usual organ of knowledge and memory in Hebrew, for M's "belly," which preserves an Egyptian usage. העמרם is not represented and החד"ה is parsed as " כנו D-stem. The two Hebrew words were well-known to the translator and the reason for these differences is unclear. 22:19 היום M S (האריזם [ האריים M S (האריים  $\sigma'$  ( $\zeta \omega \eta \nu$ ) (graph  $\eta \to \eta, \eta \to \gamma$ )

Neither the ketiv nor the qere makes good sense. The qere, שלישים, is commonly thought to mean "noble things" or "three times" (see ABP for references) but in fact it has neither meaning. A  $\psi \psi$  is a type of military officer, but the word bears no connotations of excellence. Nor is "three times" relevant to anything in the unit. The ketiv, שלשום, means "the day before yesterday," but the word appears only in the fixed idiom א)תמול שלשום (א), literally "yesterday (and) the day before yesterday." Toy hesitantly suggests adding תמול, but this is ad hoc and in any case does not make good sense. "Have I not written for you formerly?" would imply a contrast between the attempt to instill trust in Yahweh today (22:19b) and the intention to teach true words and good answers in the past, a meaningless contrast. See further ABP. We should read שָׁלשִׁים ("thirty"), following Erman 1924, 89. The ketiv and the gere are two orthographic realizations of consonantal שלשים. These two spellings arose once the function of "thirty" to refer to the number of maxims was no longer recognized. The consonantal writing is found in many MSS KR, though without the suggested vocalization. This spelling very likely maintains the primitive orthography.

Erman (1924) took his cue from Amenemope's epilogue, which begins, "Look to these thirty chapters: they divert, they instruct" (\$30; 27.8). The book of Amenemope has thirty chapters, all numbered. Many commentators, including ABP, find thirty maxims in Prov 22:17–23:11 (which is the first section of the third collection, 22:17–24:22).

G: (a) καὶ σὐ δὲ ἀπόγραψαι αὐτὰ σεαυτῷ τρισσῶς (b) εἰς βουλὴν καὶ γνῶσιν (c) ἐπὶ τὸ πλάτος τῆς καρδίας σου. "(a) And as for you—write these for yourself thrice, (b) for counsel and knowledge, (c) on the tablet of your heart." G's loose treatment of the syntax of M 22:20a resulted from

אִמְרֵי אֶמֶת לְהָשִׁיב אֲמָרֵים (אֶמֶת) לְשׂלְחֵידּ: תּּגְזָל־דָּל כִּי דַל־הָוּא וְאַל־תְּדַבָּא עָנִי בַשְׁעַר: 23 כִּי־יֵהוָה יָרֵיב רִיבֶם וְקָבָע אֶת־לְבְעִיהֵם נְפָשׁ: 24 אַל־תִּתְרַע אֶת־בַּעַל אֶף וְאֶת־אִישׁ חַמוֹת לְא תָבְוֹא: 25 פֶּן־תָּאֶלַף אִׂרְחֹתָו וְלָקַחְהֶ מוֹקַשׁ לְנַפְשֶׁדּ: 26 אַל־תְּהֵי בְתְקְעֵי־כֶף בַּעִׂרְבִים מַשָּׁאוֹת: 27 אִם־אֵין־לְדָ לְשַׁלֵּם

the assumptions that אָר אָתה (M 22:19) is the subject of the sentence (גמו שלישים מים שלישים means "threefold" or "thrice." The translator then added 22:20c, borrowed from 7:3b, to clarify the command to write. V's *tripliciter* follows G.

S: באבא איי לא שי איי איי איי איי אוא וביע. "And behold I write them for you three times." While independent of G, S too understands שלישים as "three times." The next words, במועצות ודעת, are used for the next verse.

 $T^W$ : והא כתבית אנון לך על תלתא זמנין בעיצתא וידיעתא "And behold I have written these things for you three times in counsel and knowledge."  $T^{L}$  דיעתא וידיעתא וידיעתא וידיעתא וידיעתא וידיעתא זמנין בעיצתא וידיעתא (lit. "And behold these are for you about counsel and knowledge").  $T^L$  lacks התבית, which is needed for this sentence.

# 22:21 (metath $M \in (a\lambda \eta \theta \tilde{\eta})$ ] ( $\mathfrak{g} \neq \mathfrak{S}$ ( $\mathfrak{astronomega}$ ) (metath $\mathfrak{g} \rightarrow \mathfrak{g}$ ) ( $\mathfrak{g} \neq \mathfrak{g} = \mathfrak{g} = \mathfrak{g}$ ) ( $\mathfrak{g} \neq \mathfrak{g} = \mathfrak{g} = \mathfrak{g}$ ) ( $\mathfrak{g} \neq \mathfrak{g} = \mathfrak{g} = \mathfrak{g}$ ) ( $\mathfrak{g} \neq \mathfrak{g} = \mathfrak{g} = \mathfrak{g}$ ) ( $\mathfrak{g} \neq \mathfrak{g} = \mathfrak{g} = \mathfrak{g} = \mathfrak{g}$ ) ( $\mathfrak{g} \neq \mathfrak{g} = \mathfrak{g} = \mathfrak{g} = \mathfrak{g}$ ) ( $\mathfrak{g} \neq \mathfrak{g} = \mathfrak{g} = \mathfrak{g} = \mathfrak{g}$ ) ( $\mathfrak{g} \neq \mathfrak{g} = \mathfrak{g} = \mathfrak{g} = \mathfrak{g}$ ) ( $\mathfrak{g} \neq \mathfrak{g} = \mathfrak{g} = \mathfrak{g} = \mathfrak{g} = \mathfrak{g}$ ) ( $\mathfrak{g} \neq \mathfrak{g} = \mathfrak{g} = \mathfrak{g} = \mathfrak{g} = \mathfrak{g}$ ) ( $\mathfrak{g} \neq \mathfrak{g} = \mathfrak{g}$

M's אמת in 22:21b is clumsy and redundant after "words of truth" in 22:21a and should be omitted. It is either a dittography (possibly vertical) with אמת in 22:22a (Clifford) or, more likely, a moralizing gloss intended to make it absolutely clear that only the giving of *true* replies is being taught. I consider this emendation to be conjectural, though the word is also lacking in V; see below.

G: (a) διδάσκω οὖν σε ἀληθῆ λόγον καὶ γνῶσιν ἀγαθὴν ὑπακούειν (b) τοῦ ἀποκρίνεσθαι λόγους ἀληθείας τοῖς προβαλλομένοις σοι. "(a) Therefore I will teach you a true word and (how) to understand good knowledge, (b) (so as) to answer (with) true words those who confront you." This is an expansive treatment of M. In all of G, προβάλλειν translates τοῦς οnly here. GELS defines the middle of προβάλλειν as "to confront somebody with a problem, to question." For the latter see Judg 14:12. In 26:18, its only other occurrence in G-Proverbs, προβάλλουσιν renders <sup>[\*</sup>he who shoots"], suggesting the notion of confrontation or insult, as GELS says. G 22:21 pictures a confrontation rather than the delivery of a mes-

sage. It was probably unclear to the translator why a messenger would "answer" those who sent him on an errand.

V: (a) *ut ostenderem tibi firmitatem et eloquia veritatis*, (b) *respondere ex his illi qui misit te*. "(a) That I might show you the certainty and the utterances of truth, (b) to respond from these (words) to him that sent you." *Firmitatem* is a reasonable approximation of קשט. For unclear reasons, V does not represent אמת Though I consider it a moralizing gloss (see ABP), I doubt that it was missing in V's source text. Possibly Jerome considered it adequately implied by the first stich. Compare the way he substitutes a pronoun for the repeated *n* 128:4b.

22:23

G: (a) δ γὰρ κύριος κρινεῖ αὐτοῦ τὴν κρίσιν, (b) καὶ ῥύσῃ σὴν ἄσυλον ψυχήν. "(a) For the Lord will judge his (the poor man's) case, (b) and you will preserve [lit. 'rescue'] your soul safe from harm." G guesses at the rare verb קבע (actually, "rob") in 22:23b, then must change the person of the suffix of קבעיהם G must mean that if you obey the admonition in 22:22, you will be safe. Prov 22:23b seems to be influenced by Ezek 3:19, 21, which is likely the source of the awkward second-person here.

S: (a) = M; (b) הארא הופאסס (b) and will repay the injustice (to) their souls." S must guess at the meaning of  $\eta$  but does not rely on G.

22:25 אָרְחֹתֵיוֹ [ (num or orth) See the comment on 2:8.

G: μὴ δίδου σεαυτὸν εἰς ἐγγύην αἰσχυνόμενος πρόσωπον. "Do not give yourself as surety, being diffident toward others [lit. 'feeling awe/shame toward face']." G explains what it means to clasp hands. For משאת, G had משאות (Jäger), which he understood as משאת (lit. "from lifting up,"

לְמָּה יִקַּח מִשְׁבְּרָדְ מִתַּחְתֵּידְ: 28 אַל־חַפּג גְּרַוּל עוֹלָם אֲשֶׁר עָשִׁוּ אֲבוֹתֵידְ: 29 חָזִׁיתָ אֵישׁ | מְׁהֵיר בִּמְלַארְתּוֹ לְפְנֵי־מְלָכִים יִתְיַאֶב בַּל־ יִׁתִיַצֵב לִפְנֵי חֲשֵׁבְּים: פּ 1:22 בְּי־תֵשֵׁב לִלְחוֹם אֶת־מוֹשֵׁל בִּין תְׁבִצַר אֶת־אֲשֶׁר לְפָנֵידְ: 2 וְשַׂמְתָ שַׂבְּין בְּלֹעֶדְ אִם־בַּעַל נֶפָשׁ אֶתָה: 3 אַל־תִּתְאָו לְמַטְעַמּוֹתֵיו וְהוּא לֶחֶם בְּזָבִים: 4 אַל־תִינַע

sc. כנים). The idiom appears without פנים in Gen 4:7, where G understands it differently.

 $S \approx G$ , using איא איא איז פוא אונ. "because you are ashamed of faces," i.e., diffident). To deal with the difficult משאות, T takes על על from 17:18.

### 22:27

G and S omit למה. It is impossible to determine whether this was missing in their source text (due to haplography) or was omitted in translation because of the difficulty of fitting the word into the Greek and Syriac syntax. V explains משכבך as *operimentum de cubili tuo* ("the covering from your bed"). This is correct, since beds in ancient times were coverings one lay on and wrapped oneself in.

### 22:29

G: (a) δρατικόν ἄνδρα καὶ ὀξύν ἐν τοῖς ἔργοις αὐτοῦ (b) βασιλεῦσι δεῖ παρεστάναι (c) καὶ μὴ παρεστάναι ἀνδράσι νωθροῖς. "(a) It is necessary for a man who is observant and keen in his works (b) to attend on kings (c) and not attend on sluggish men." G derives "observant" from πτιτη (actually, "have you seen?"), which it treats as an adjective and translates by the hapax δρατικόν. (Πτιπ is translated correctly in 29:20.) There is no reason to assume the reading πτιπ (Jäger), since that is no closer to δρατικόν than M is, and איש חזות (lit. "dark," "obscure") by making it the opposite of "keen."

### 23:2

G: (2a) καὶ ἐπίβαλλε τὴν χεῖρά σου, (2b) εἰδὼς ὅτι τοιαῦτά σε δεῖ παρασκευάσαι. (3a) εἰ δὲ ἀπληστότερος εἶ.... "(2a) And lay your hand [sc. on what is placed before you], (2b) knowing that it is right for you to pre-

pare such things. (3a) And if you are insatiable...." (G 23:2a  $\approx$  M 23:2a; G 23:2b  $\neq$  M; G 23:3a = M 23:2b; G 23:3b = M 23:3a; G 23:3c  $\approx$  M 23:3b.) (NETS: "(b) since you know that you will have to prepare such things.") The intention of "prepare" here is unclear. Perhaps puzzled as to why one would put a knife to his throat (as M advises, in a vivid metaphor for self-control), the literal-minded translator composed a sentence affirming that it is right to eat what you are given.

V: (a) = M; (b) si tamen habes in potestate animam tuam. "(b) if nevertheless you have your soul under control." V has an unusual interpretation of  $\mathbf{u}$  -one shared, independently, by T.

T: (a) = M; (b) אין מרא דנפשך את (b) if you are the master of your soul/appetite." 23:2b = V. Joseph Qimhi explains the clause similarly: "If you govern your soul, then control your desire [lit. 'spirit']." T<sup>Z</sup> אין מלושא to M.

23:3 לחם M S (לעגבאי) א לחם א לחם א לחם מושיי?) א לחם א לחם מושיי?) א לחם צו לשגבאין א לחם צו לשגבאין א לחם צו לשגבאין א א לחם צו לשגבאין א לחם א לחם צו א לחם צו א לחם א לחם א לחם צו א לחם צו א לחם א לחם צו א לחם א לחם צו א לחם צו א לחם א לחם צו א לווי א לוויו א לוויא א לוויא א לוויא לוויייא א לוויייא לווי

G: (a) (see M 23:2b); (b) μὴ ἐπιθύμει τῶν ἐδεσμάτων αὐτοῦ, (c) ταῦτα γὰρ ἔχεται ζωῆς ψευδοῦς. "(b) Do not desire his delicacies, (c) for these belong to a false life." (G 23:3a = M 23:2b; G 23:3b = M 23:3a; G 23:3c  $\approx$ M 23:3b.) ζωῆς = "לְחֵיָי The ם of ם split into two yods. See LSF §\$132ef for similar phenomena, such as מ or ם becoming ז׳.

אָזין קיו קאָן: In 23:6 and 24:1, the Mp has 'אין ק', an abbreviated writing of תחאין. This orthography is derived from the third masculine plural noun suffix י-. For other examples, see the *qerayin* in Cant 2:11; Num 11:32; 12:3; and Ps 105:40. The Mp at 23:3 lacks a *qere*. (BHS supplies one.)

23:4

G: (a) μὴ παρεκτείνου πένης ὢν πλουσίω, (b) τῆ δὲ σῆ ἐννοία ἀπόσχου. "(a) (If) you are poor, do not measure yourself [lit. 'stretch yourself out'] by a rich man, (b) but be restrained by your understanding." Rather than allowing the impression that one should refrain from hard work, G warns against dissatisfaction with one's given station in life. Comparלְהַעֲשֶׁיר מְבִּינְתְדְ חֲדָל: 5 <sup>ר</sup>ְהַתָּעִיף<sup>י</sup> עֵינֵיִדְ בּוֹ וְאֵׁינֶנּוּ כִּי עָשָׂה יַעֲשָׂה־ לְּוֹ כְנָפֵיִם כְּנָשֶׁיר "יָעָוּף" הַשָּׁמִיִם: פּ 6 אַל־תִּלְחַם אֶת־לֶחֶם רַע עֵיִן וְאַל־תִׁתְאָו לְמַטְעַמֹתְיו: 7 כִּי | כְּמוֹ־רּשׁׁעֵר" בְּנַפְשׁׁוֹ כֶּזְ־הָוּא אֶכָל וּשְׁתֵה יָאׁמַר לֶדְ וְלִבוֹ בַּל־עִמֶּדָ: 8 פְּתְדָ־אָכַלְתָ תְקִיאֶנָּה וְשׁׁחַתָּ

ing oneself to the rich would just breed envy. Elsewhere too G eliminates or tempers warnings against overwork; see the comment on 10:22. Introducing the theme of envy tightens the connection of this verse to its context, 23:2–5. להעשיר is understood as "to the rich man," with an unassimilated article.

S: אלאסיב אלא הערט אלא הערט אלא הערט גערט גערט אלא ארט שארט איני (a) Do not approach a rich man, (b) but, in wisdom, depart from him." S derives M's אוז דרש דרש דרט גע"ע. This led to a contextual rendering of  $\pi$  as "depart." S 23:4 speaks about the rich man (supposedly העשיר), as do G and T, rather than wealth. S shares G's reluctance to caution against overwork; see the comment on 10:22.

בתעיף M<sup>Q</sup> G (<br/>להתטוס א M<sup>Q</sup> G (לאניז א א V (erigas) T<sup>L</sup> (תצר) א התעוף (graph '  $\rightarrow$  <br/>ו)

יעוף  $M^Q$  ] אועיף  $M^K \approx G$  (גמו האססדף<br/>לקפו) S (הפוט) V (et avolabunt)  $\approx T$  (דטאס) (graph '  $\rightarrow$  ', graph '  $\rightarrow$  ')

G: (a) ἐἀν ἐπιστήσῃς τὸ σὸν ὄμμα πρὸς αὐτόν, οὐδαμοῦ φανεῖται, (b) κατεσκεύασται γὰρ αὐτῷ πτέρυγες ὥσπερ ἀετοῦ (c) καὶ ὑποστρέφει εἰς τὸν οἶκον τοῦ προεστηκότος αὐτοῦ. "(a) If you place your eye on him, he will in no way appear, (b) for he has made for himself wings like an eagle's, (c) and returns to the house of his superior." G continues speaking about the rich man from the preceding verse rather than his wealth, producing a strange image. השמים, which in the original refers to the sky, is here explained as "the house of his superior," that is to say, God, and to return to his house is to die.

V clarifies the image by translating בו ואיננו as *ad opes quas habere non potes* ("to wealth which you are not able to have") (CB).

In the first *ketiv-qere* pair, G and S (גבו) agree with the *qere*. Given the plural of עיניך, the *qere* ("if you make [your eyes] fly") is better than the *ketiv* (יוֹן [your eyes] flies"). In the second pair, the *ketiv* is an impossible form. The second stich would be smoother with a *waw*-conjunctive—קיעוך). But it is not syntactically necessary, and the presence of a conjunction in G and S is uncertain evidence for it, since both versions are flexible in their representation of conjunctions.

# 23:6 הִתְאָיו [ M<sup>Q</sup> ] (orth)

See the comment on 23:3.

23:7 שָׁעֵר (εἰκάζων) V (aestimat) ] שָׁעַר M (vocal)

אלבו בל M S (ملحس المحرف [ (ملحس المحرف) אילהבלו [ (ملحس المحرف) אילהבלו בל ( دולהבלי, in G 8) (graph similarities, unc)

G: (7a) δν τρόπον γὰρ εἴ τις καταπίοι τρίχα, (b) οὕτως ἐσθίει καὶ πίνει. (8a) μηδὲ πρὸς σὲ εἰσαγάγης αὐτόν. "(7a) For just as if someone swallowed down a hair, (b) thus he eats and drinks. (8a) Nor should you bring him toward you." G merges M 23:7b (ולבו בל עמך) with 23:8a, ignoring יאמר אמר.

אמדמחוֹסו דרוֹצָם is a double translation of שער exploiting two senses of the graphic form. As BHQ observes, ישְׁעָרָנוּ is translated אמדמחוֹבדמו שׁמָרָ is translated אמדמחוֹבדמו שׁמָרָ in Ps 58[57]:10. In 23:7b, G implicitly vocalizes אֹרֵל וְשׁׁתָה (Jäger).

Several vocalizations are possible. (1) שָׁעַר (M). By this pointing, the line would be translated, "For as if he calculated in his mind, thus he [it?] is." (2) שָׁעַר (G) "hair" (that is, a blockage in the throat). (3) שָׁעַר ("storm"). (4) שָׁעַר ("calculates") is to be preferred. This provides an antecedent for כן הוא ס and improves the syntax, though the gist is the same as in M. The line should be translated, "For like one who calculates in his mind, thus is he." While the stingy host invites you to eat, he is silently calculating costs and benefits. He is brooding on the cost of every bite you take and wondering if you are worth it. Other vocalizations may be relevant as visual wordplays. See ABP.

שאָלה בּלוֹ עַמָּדָ מטֹדטי = אָלהבלוֹ עַמָּדָ (with the negative carried forward from 23:6a). This is a distortion of M's ולבו בל עמך, though the steps in the permutation cannot be traced exactly. The retroverted Hebrew does not show the usual Greek-Hebrew correspondences, but הל"ך in the H-stem is translated by מאָנוי in Ezek 40:24, and יב"ל in the Hp-stem is translated by מאָנוי (passive) in Isa 53:7 and Jer 11:19.

V: (a) quoniam in similitudinem arioli et coniectoris (b) aestimat quod ignorat. (c) comede et bibe dicet tibi et mens eius non est tecum. "(a)

דְּבָרֶידְ הַנְּעִימִים: 9 בְּאָזְנֵי כֵּסִיל אַל־תִּדַבֵּר כִּי־יָבוּז לְשֵׁכָל מָלֶידְ: 10 אַל־הַפֵּג גְּבַוּל עוֹלֵם וּבִשְׁדֵי יְׁתוֹמִים אַל־תִּבְא: 11 כִּי־גֹאָלָם חָזֶק הוּא־יָרִיב אָת־רִיבָם אִתֵּדְ: 12 הָבִיאָה לַמּוּסֵר לִבֶּדְ וְאָזְנֶדְ לְאִמְרֵי־ דְעַת: 13 אַל־תִּמְנַע מִנַּעַר מוּסֵר כִּי־תַכָּנוּ בַשׁׁבָט לָא יָמוּת: 14 אַתּה בַּשֵׁבֶט תַּכָּנוּ וְנַפְשׁוֹ מִשְׁאוֹל תַּצִיל: 15 בֵּנִי אִם־חָכַם לְבֵּדְ ישִׁמַח לִבִּי גַם־אֲנִיזּו וְתַעְלִזְנָה כִלְיוֹתֶי בְּדַבֵּר שְׁפָּתֶידְ מֵישָׁרִים: 17 אַל־יְקַנֵּא לְבָדְ בַּחַטָּאֵים כִּי אִם־בְיִרָאַת־יְהוָה כָּל־הַיּוֹם: 18 בִּי אָם־ <sup>ו</sup>תִּשְׁמֶרְנָה<sup>י</sup> יֵשׁ אַחַרֵית וְתַקַוְתָדָ לָא תִכָּרַת: 19

### 23:8

G: (23:8a = M 23:7b) (8b) καὶ φάγῃς τὸν ψωμόν σου μετ' αὐτοῦ. (c) ἐξεμέσει γὰρ αὐτὸν καὶ λυμανεῖται τοὺς λόγους σου τοὺς καλούς. "(b) nor eat your morsel with him. (c) For he will vomit it up and spoil your lovely words." See the comment on 23:7b. G continues the description of the envious man. λυμανεῖται = שִׁתֵּת S = M.

23:10 עולם M S (גרך באס) G (מוֹשׁוֹם [ (גרק מוילים G (מוֹשׁוֹם ) rab V (*parvulorum*) (vocal, orth)

V parvulorum ("children") = עוּלִים, for M's עוּלָם. Jerome rendered אולים as antiquos in the doublet in 22:28. The consonantal reading עולים

is presumed in m. Pe'ah 5:6, though there it is construed as עוֹלִים ("those who ascend"). עוֹלִים also seems to be presumed in y. Sotah 4:4. Jerome probably had עולים in his Hebrew text of 23:10. M is preferable (see ABP). עולים arose under the influence of the parallel "orphans."

### 23:11

After "their redeemer," G adds  $\varkappa$ úpioç to specify (unnecessarily) just who the redeemer is.

### 23:16

G: (a) καὶ ἐνδιατρίψει λόγοις τὰ σὰ χείλη (b) πρὸς τὰ ἐμὰ χείλη, ἐἀν ὀθὰ ὦσιν. "(a) And your lips will linger in words (b) with my lips, if they are honest." The relation of G to M is hard to explain. Emending ἐνδιατρίψει to ἐνδιαθρύψει (de Lagarde; Baumgartner) does not provide a satisfactory meaning ("play the prude toward"? [LSJ]) and leaves other discrepancies unresolved.

# 23:18 (ה∩ם G (דחףאָסאָק מטֿדמֹ) ] > M S (homoi ה∩ם)

G: (a) ἐἀν γὰρ τηρήσης αὐτά, ἔσται σοι ἔκγονα, (b) = M. "(a) For if you keep them, you will have offspring." Though many commentators try to make sense of M 23:18a, a verb is missing after אם. M 23:18a would have to be translated, "for if there is a future," which is not a meaning-ful protasis. Following G, we should add a verb meaning "keep them" (Gemser 1963; McKane; BHS; ABP; and many others), either השמרנה or השמרנה. (G's αὐτά must refer to the exhortations in 22:15–17, but the proper antecedent is "fear of the Lord" in 23:17, and the verb should be restored with the 3rd fem sg suffix.) Admittedly, G may be supplying the verb as required by context (BHQ), but if so, it is probably restoring a word once present. V's *quia habebis spem* ("for if you have hope") is supplied, but "keep" is more appropriate to a sentence motivating the exhortation to fear the Lord.

S: (a) א שנוא (b) = M. "(a) And you will have an end [= 'future'?]." S ignores the א מם does V (and many modern translators).  $T \approx S$  but explains "end" as  $\psi \in M$ . "a good end."

### 23:19

G: (a) = M; (b) אמו אמדביט פֿטעס פֿעטעמג מאָג אמד (b) and make the reflections of your heart straight." G associates אש"ר ("straight"). The "way of your heart" is explained as your thoughts.

S: (a) = M; (b) או ביא, בורא (b) and my mind will rejoice

וַחְכֵם וְאַשֵּׁר בַּדֶּרֶדְ לִבֶּדְ: 20 אַל־תְּהֵי בְסְבְאֵי־זְיָזְ בְּזֹלְלֵי בָשָׁר לֶמוֹ: 21 כִּי־סֹבֵא וֲזוֹלֵל יִזְרֵשׁ וּקְרָעִׁים תַּלְבִּישׁ נוּמֵה: 22 שְׁמַע לֱאָבִידְ זֶה יְלָדֶדְ וְאַל־תְבוּז כִּי־זָקְנֶה אִמֶּדְ: 23 <sup>י</sup> 24 <sup>1</sup>גַּיל יְגִיל<sup>ז</sup> אֲבִי צַדִּיק <sup>י</sup>וְיוֹלֵד<sup>י</sup> חְׁכָם <sup>י</sup>ִישְׁמַח<sup>י</sup>-בְּוֹ: 25 יְשְׁמַח־אָבִידְ וְאִמֶּדְ וְתָגֵל יְוֹלַדְתֶדְ: 26 תְּגֵה־בְנֵי לִבְּדֶ לֵי וְעֵינֶׁידְ דְרָכֵי יֹתִאַרְנָה<sup>ז</sup>: 27 כָּי־שׁוּתָה עַמֻקָּה

in your heart." S identifies אשר with its homonym "declare happy" and must adjust the rest of the sentence to make sense of it. Hence retroversion to אשר דרכִי בלבך (Baumgartner; Pinkuss) is not justified.

23:20 בשר M S (הרסאי G (אר שבר G (אר מער היאי)) לא בשר שבר G (אר מאסט א מאסט (dittog שב  $\rightarrow \rightarrow \rightarrow$  בשר בשר בשר שר שר שני)

G: (a) = M; (b) שָקלּ בֹּאדבוֹיסט סטשָלָסאמוֹג אַרָבּשָּׁט דב מֹאַסַסְמסשָטוֹג. "(b) and do not spend much time at feasts or with purchases of meat." אָרָבּשׁי דב מֹאַסָסְמסשָטוֹג = מֹאַסָרָססָבָר) בָּשָׂר שֶׁבֶר ("provisions"] is translated מֹאַסָסְמסשָטי in Gen 42:19). Since דב never represents the *waw*-conjunctive in G-Proverbs, its use here (without καί) indicates the absence of -1.

23:21

G translates זולל (usually "glutton") as πορνοκόπος ("whoremonger") only here (but correctly; see ABP). The translator is extending the warning against drunkenness and gluttony to other types of dissoluteness.

### 23:22

G adds the vocative vié, thus assimilating to 23:15 and 19. See the comment on 1:15.

# 23:23 > G ] אָמֶת קֵנָה וְאַל־תִּמְכֵּר חְרָמֶה וּמוּמֶר וּבִינֶה: [ M S אַמֶת הַישּלא מע.) (theol)

This verse is probably an addition to the text subsequent to the divergence of proto-M and proto-G. The verse is absent from G, and there is no motivation, ideological or graphic, for an omission by a copyist or translator. It was intended to supplement the exhortation in 23:22 by a motivation, in accordance with the usual structure of exordia. (See the footnote at 22:17, above, and ABP 1.45). While the present verse is appropriate to the general context, it is extraneous to the theme and

keywords of the otherwise cohesive epigram, 23:22 + 24-25. Verse 24 would follow nicely on 23:22.

23:24 גול יגול  $M^Q$  גול יגול (אגע גול גדל אדל אדל מגדל אדל אנול גול  $M^K$ ; גול גדל  $M^K$ : graph ' $^1 \rightarrow 1$ ; G: graph ' $^1 \rightarrow 1$ ; G: graph ' $^1 \rightarrow 1$ , ' $^3 \rightarrow 1$ ) (ומאן דמוליד)  $M^Q$  S (מנסלג) (ומאן דמוליד) (שנסלג MK (± conj)

 $M^{Q}$  V (*laetabitur*) איז  $M^{K}$  S ( $(-\infty)$ ) (± conj) (± conj)

The *ketivin* and the *qerayin* readings constitute two forms of the sentence. *Ketiv*: גיל יגיל אבי צדיק יולד חכם וישמח בו. *Qere*: גיל יגיל אבי צדיק ויולד חכם ישמח בו. In support of the *qere* is the fact that "rejoice" is consistently גו"ל , not גו"ל (though the *qere* could admittedly be a normalization). The versions do not testify clearly to *ketiv-qere* variants here, but the above annotation aligns them with the readings as if they were translating literally.

S: אונער סגני ארביש, גוגעראי מטעג ארביש ארביש, אונעראי ביש ארביש, אונעראי ביש ארביש, אונעראי ביש ארביש, אונעראי (a) The father of a righteous man will rejoice and be glad, (b) and he will beget a wise (son), (c) and his father will take pleasure in him." The subject of 23:24b seems to be the righteous man (the second generation), who is also the "father" in 23:24c. A wise son makes his father happy and is rewarded with a wise son of his own. This is a possible construal of the Hebrew, if the source text read: גיל יגיל אבי צדיק ויולד חכם וישמח בו 13:24a and the *ketiv* in 23:24b. S's interpretation presumes the gene in the source text.

נערנה 1:26 אערנה M<sup>Q</sup> G (דאף<br/>פוֹדשסמי) S (גערנה) V (custodiant) T (תינטרן) ] M<sup>K</sup> ס' (<br/> של הערמה) (metath רצ $\to \to \to$ ר)

Both the *qere* and the *ketiv* (הִרְצֶינָה =) are meaningful, but the *qere* is preferable, because exhortations to listen in Proverbs speak of *keeping* the commands rather than *desiring* them.

זְּזֶרֶה' וּבְאֵר צְׁרָה נְכְרִיֶּה: 28 אַף־הֵיא כְּחֶתָף תָּאֶרֶב וּבוֹגְדִים בְּאָדֶם תּוֹסְף: 29 לְמִׁי אוי לְמֵי אֲבוי לְמֵי 'מְדוֹנִים' | לְמֵי שִּׁיחַ לֱמִי פְּצָעֵים חִנֵּם לְמִי חַכְלִלוּת עֵינֵיִם: 30 לַמְאַחֲרֵים עַל־הַיֶּיו לַבָּאִים לַחְקֹר מִמְסֵד: 31 אַל־תֵּרָא יַיִז' בֶּי יִתְאַדֶם בְּי־יִתֵּן 'בַּבְּוֹס' עֵיגֵוֹ יִתְהַצֵּך בְּמֵי

G: (a) πίθος γὰρ τετρημένος ἐστὶν ἀλλότριος οἶκος, (b) καὶ φρέαρ στενὸν ἀλλότριον. "(a) For a strange house is a pierced jar, (b) and a narrow well is strange." G 23:27a is scarcely related to M, though the Hebrew is clear and not contrary to the translator's ideas; see G 22:14. Gerleman (1956, 33) says that G refers to a household whose members behave as strangers and take no interest in the common good.

"Pierced jar" is a Greek mythological motif. The Danaides were condemned to draw water with a pierced jar. This image is used broadly to indicate futility and wastefulness (BAP 105). According to Pseudo-Aristotle, "The master of a house who is unable to conserve his property is what is called a pierced jar" (Economics 1.6 1344b25). G 23:27 seems to echo that very statement. A "strange house" wastes its possessions. Compare "and he who consorts with harlots loses money" (M 29:3b). In spite of the paraphrastic character of G 23:27, άλλότριος in 23:27a can be retroverted to זרה ("strange"), for M's . זונה M's reading could have arisen by graphic confusion ¬/J (cf. Kaminka 1931–32, 178, and compare the comment on 5:3). The change would have been occasioned further by the pragmatic synonymity of זנה and זנה in the context of the Strange Woman. זרה is the better reading. While לנה is possible, the prostitute was less of an object of aversion to the sages than was the adulteress; see ABP 1.138. Also, נבריה (present in 23:27b) is always parallel to זרה in Proverbs (2:16; 5:20; 7:5; 20:16; 23:27; 27:13), with the exception of 6:24. There it is parallel to אשת רעך ("another man's wife") (as emended), which is equivalent to זרה/אשה זרה; see ABP 1.139-41. The doublet in 22:14 has זרות ("strange women").

23:28 תארב M ] האבד G (מהס<br/>גנואה) S (כספר האבד) (graph האבד , metath בד $\to =$ 

G: (a) οὖτος γὰρ συντόμως ἀπολεῖται, (b) καὶ πᾶς παράνομος ἀναλωθήσεται. "(a) For this [sc. house] will suddenly be destroyed, (b) and every transgressor will be cut off." ἀπολεῖται = Τἐκ̄Ξ (Jäger). G treats κɨκ. probably because he does not recognize the motif of the woman in the preceding verse. דרחתי is a reasonable construal of כחתי as "suddenly" ("like a snatch"  $\approx$  in an instant). מעמעש לאסנדמו derives תוסף from תוסף.

V: (a) *insidiatur in via quasi latro*, (b) *et quos incautos viderit interficit.* "(a) She lies in wait in the way as a robber, and him whom she sees unwary, she kills." V adds *in via* to make it clear that the image is of a robber, not a hunter. Hence *quasi latro* = בְּחֹתֵך. As a robber, the woman kills not only traitors, as in M, but all who are *unwary* (CB).

# **23:29** מִדְיָנִים [ M<sup>K</sup> ] מִדְיָנִים (dial? norm)

On the *ketiv-qere*, see the comment on 6:14.

It is uncertain whether ארוסיג (and געשה) is based on the *qere* or the *ketiv*, deriving it from דין ("judgment"), since in 28:25 G parses מדון similarly. In 23:29b, מאלמו אמו אליסעמו ("unpleasantness and gossip") is a double translation of שיה which means both "complaints" and "talk."

V's *cuius patri vae* ("whose father has woe") parses the *hapax* אבוי as Aramaic "his father," (= אבוה', rabbinic אבוים). *cui foveae* ("who has [= 'falls into'] pits") identifies שיח as Aramaic/RH שִׁיח ("pit"). Jerome does not recognize the sense "lament" for שיה, but only "speech" and the like. S approximates this word by תיאלה ("evils").

23:31 בכוס  $M^Q \approx G$  (εἰς τὰς φιάλας καὶ τὰ ποτήρια, dbl) S ( $\sim m \to V$  (in vitro) T (בכסא) J  $M^K$  (graph  $1 \to i$ )

G: (a)  $\mu\dot{\eta}$   $\mu\epsilon\theta\dot{\upsilon}\sigma\kappa\epsilon\sigma\theta\epsilon$  oႆν $\omega$ ,  $\dot{\alpha}\lambda\lambda\dot{\alpha}$   $\dot{\delta}\mu\lambda\epsilon\tilde{\imath}\tau\epsilon$   $\dot{\alpha}\nu\theta\rho\dot{\omega}\pi\sigma\iota\varsigma$   $\delta\imath\kappa\alpha\iota\sigma\varsigma$  (b)  $\kappa\alpha\dot{\imath}$  $\dot{\delta}\mu\lambda\epsilon\tilde{\imath}\tau\epsilon$   $\dot{\epsilon}\nu$   $\pi\epsilon\rho\imath\pi\dot{\alpha}\tau\sigma\iota\varsigma$ . (c)  $\dot{\epsilon}\dot{\alpha}\nu$   $\gamma\dot{\alpha}\rho$   $\epsilon\dot{\imath}\varsigma$   $\tau\dot{\alpha}\varsigma$   $\phi\iota\dot{\alpha}\lambda\alpha\varsigma$   $\kappa\alpha\dot{\imath}$   $\tau\dot{\alpha}$   $\pi\sigma\tau\dot{\eta}\rho\iota\alpha$   $\delta\tilde{\omega}\varsigma$   $\tau\sigma\dot{\imath}\varsigma$  $\dot{\delta}\phi\theta\alpha\lambda\mu\sigma\dot{\imath}\varsigma$   $\sigma\sigma\upsilon$ , (d)  $\ddot{\upsilon}\sigma\tau\epsilon\rho\sigma\nu$   $\pi\epsilon\rho\imath\pi\alpha\tau\dot{\eta}\sigma\epsilon\iota\varsigma$   $\gamma\upsilon\mu\nu\dot{\sigma}\tau\epsilon\rho\sigma\varsigma$   $\dot{\upsilon}\pi\dot{\epsilon}\rho\sigma\upsilon$ . "(a) Do not get drunk [pl] on wine, but converse [pl] with righteous men, (b) and converse [pl] in (public) walks. (c) For if you [sg] set your eyes to the bowls and the cups, (d) later you [sg] will walk about more naked than a pestle." To M's warning G adds advice about what one *should* do, namely keep company with the righteous. Some of the words and roots of M are discernible behind the translation, but they do not appear in M's order and are scarcely related to M's syntax. The fluctuation between

שָׁרִים: 32 אֲחֲרִיתוֹ כְּנָחֲשׁ יִשֶׁדְ וְּכְצִפְּעֹנִי יַפְרְשׁ: 33 אֵינֶידְ יִרְאַוּ זָרֵוֹת וְלִבְּדְׁ יְדַבֵּר תַּהְפָּכְוֹת: 34 וֲהָיִיתָ כְּשׁבֵב בְּלָב־יָם וּכְשׁבֵב בְּרָאשׁ חִבֵּל: 35 הִכְּוּנִי בַל־חָלִיתִי הַלָמוּנִי בַּל־יָּדֶעְתִּי מְתַי אָמֵיץ אוֹסִיף אֲבַקְשָׁנּוּ עוד: 24:1 אַל־תֵּקַנֵּא בְּאַנְשֵׁי רָעֶה וְאַל־תִׁהְאָו לִהְיָוֹת אִתֵּם:

second singular and second plural is strange. μεθύσκεσθε associates אתרא with רָוָה ("slake thirst"), which is philologically feasible (*DCH* 7:361). G 23:31b alludes to the command to converse about God's commandments when walking in the way (Deut 6:7; 11:19) and, possibly, to peripatetic philosophical conversations (BAP). Once the translator decided to treat יתהלך as second person, he needed an adverb that implied something undesirable. This is likely the reason he substituted a strange idiom: "more naked than a pestle," which is a Greek commonplace (BAP 104).

S: κλα στο μι κίκ στο στο κίκι τη κίκ στο στο κίκι τη το το μι το μι κά. "(a) Do not fix your eye on wine, whose eye is reddish in the cup, (b) but meditate on righteousness." S 23:31 is an erroneous rendering of M. Syriac content is paying attention to G even in 23:31c, showing that the S translator is paying attention to G even in places where he is not directly translating it. The clause content is taken from S 23:20a and is unlikely to be a later interpolation (Joosten 1995, 60, countering Pinkuss). Like G, S supplements the warning against drunkenness with moral advice. Since content the warning against drunkenness with moral advice. Since content the translator probably thought that the moralization was inherent in the original. On this verse see Joosten 1995, 69–70.

G: (a) τὸ δὲ ἔσχατον ὥσπερ ὑπὸ ὄφεως πεπληγὼς ἐκτείνεται (b) καὶ ὥσπερ ὑπὸ κεράστου διαχεῖται αὐτῷ ὁ ἰός. "(a) In the end, he is stretched out as one bitten by a snake; (b) the venom is dispersed through him as if he were bitten by a horned viper." G must have read שכב 'שכי alongside 'שכישכ  $\rightarrow$  ישכב could have arisen in three steps:  $\rightarrow$   $\rightarrow$   $\rightarrow$  'שכב  $\rightarrow$  'שכי (But on the actuality of such steps, see §4.3.) διαχεῖται = !!!

CB observes that "V translates כצפעוני as *regulus*, a term that typically means not a snake but rather a government official. It does have a nega-

tive connotation—it was the name of a Roman general captured by the Carthaginians, so it might be trying to convey unworthiness. Also, V supplies *venena* (lit. 'venoms') as an object of 'disperse."

### 23:33

G: (a) οἱ ὀφθαλμοί σου ὅταν ἰδωσιν ἀλλοτρίαν, (b) τὸ στόμα σου τότε λαλήσει σκολιά. "(a) When your eyes see a stranger [fem], (b) your mouth will then speak crooked things." Irin is treated as singular, as in 22:14. For M's "your heart" G substitutes "your mouth" as the logical organ of speech. G says that if you are drunk—the situation described in the preceding verses—you will speak sinful things with a strange woman, that is to say, agree to her blandishments. In a similar fashion, in 23:21, G interprets i not as gluttony but as whoremongering.

### 23:34

G: (a) καὶ κατακείσῃ ὥσπερ ἐν καρδία θαλάσσης; (b) καὶ ὥσπερ κυβερνήτης ἐν πολλῷ κλύδωνι. "(a) And you will lie down as in the heart of the sea, (b) and as a pilot in a great storm [lit. 'waves']." G omits the second ברשכב, probably as semantically superfluous, and construes as הבל ("sailor," "pilot"); similarly S حديد. Having misidentified the subject (actually "one who reclines"), the translator forces שבו to fit the context by translating it as "waves." S הובל "in a great tempest") is likewise contextual. It does not help to derive these renderings from ברשער (which would not be translated in these ways) or בסער (= בסער subject) as BHQ, following Jäger, suggests.

### 23:35

Whereas in M the drunk says, "I will seek more," in G he says, ζητήσω μεθ' ών συνελεύσομαι ("I will seek those with whom I will go about"). G refers again to the choice of the right companions, a topic introduced in 23:31 and 33.

# 24:1 הָתִאָּיו [ M<sup>Q</sup> (orth)

See the comment on 23:3. G prefixes vié; see the comment on 1:15.

2 כִּי־שִׁד יֶהְגֶה לִבֶּם וְׁעָמָׁל שִׂפְתֵיהֶם תִּדַבֵּרְנָה: 3 בֵּחָכְמָה יִבְּנָה בֵּיִת וֹּבִתְבוּנָה יִתְכּוֹנֵן: 4 וְּבְדַעַת חַדָרִים יִמָּלְאֵוּ כִּל־הוֹן יָקָר וְנָעִים: ז וּגָבַרי חָכָם וֹמַעֵז' וְאִישׁ־דַּעַת וֹמַאַמִּץ'־כְּחַ: 6 כִּי בֻתַחְבָּלוֹת תַּעֲשֶׂה־לְךָ מִלְחָמֶה וֹּתְשׁוּעָה בְּרֵב יוֹמֵץ: 7 רָאמַוֹת לֶאֶוִיל חָכְמוֹת

24:2

For M's  $\forall w$  ("destruction"), G has  $\psi \varepsilon \upsilon \delta \eta$  ("deceit") and S uses  $\prec \flat \omega \omega \eta$  ("of the evil man"). Both versions are using broad, obvious terms to explicate a metaphor. The literal-minded translators may have wondered how one can *speak* destruction. On G-Proverbs' predilection for words from the stems  $\psi \varepsilon \upsilon \delta$ - and  $\alpha \lambda \eta \theta$ -, see Dick 1990, 23.

24:5 (سحبحه Μ ≈ G (σοφός) ≈ S (سحبحه) (vocal)

מעז G (לסעטףסטט) S ( בעוז [ (ד $\rightarrow$  ביעוז) M (graph  $\rightarrow$  ב, orth)

 $^* \alpha \lambda'$  (שֹׁתּדָּס אָאָמָזָי S ( געבי אורא העלדס;) א מארץ (שֹׁת אָ מָזָי M; א מארץ (ענערא אורץ); א גערץ (שנערט); א א געני ש' ס' (שנער גענע גענער גערט) (M: vocal; G: graph  $^2 \rightarrow ^$ ?;  $\sigma'$  graph  $^2 \rightarrow ^$ )

G: (a) κρείσσων σοφὸς ἰσχυροῦ (b) καὶ ἀνὴρ φρόνησιν ἔχων γεωργίου μεγάλου. "(a) Better a wise man than a strong one (b) and a man who has good sense than a great field." ἰσχυροῦ = খαι (or ແฟລ understood as a synecdoche for "than a strong man"), with the comparative idea derived from the (ζ). It is hard to see what could give rise to G's "than a field" other than a textual difference, namely, מאמץ (lit. "from land") for M's שארץ. Confusion between a and could occur if the left arm of the a were lost. γεώργιον does not correspond to γ elsewhere, but "a great field" would be a reasonable paraphrase of a faulty text. A Hexaplaric reading (αλ') is ὑπὲρ κράτος ἰσχύῖ ("above one strong in strength") = DAP translates γεωργίου μεγάλου as "un grand domaine." S: אבוא דא גובאא דער אבוא אין גערא איי (a) Better a wise man than a strong one, (b) and a man of knowledge than a mighty man." S געאַמָץ בֿת implicitly vocalizes געאַמָץ בֿת, as in the Hexaplaric reading.

# 24:6

G: (a) μετὰ κυβερνήσεως γίνεται πόλεμος, (b) βοήθεια δὲ μετὰ καρδίας βουλευτικῆς. "(a) War is (conducted) with strategy [lit. 'steering'], (b) and help is with the heart of a counselor." The connective כי is naturally omitted from the translation in G and S, since this verse does not really motivate the preceding. G, followed by S (כאברג) and V (*initur*), parses תעשה as passive/impersonal הַעָשָׁה. The ethical dative ל is ignored in G, V, and S.

Though M 24:6b = M 11:14b, G construes the line differently. In 15:22, as in 24:6b,  $\Box \Box \Box$  is represented as  $\dot{\epsilon}\nu \delta \dot{\epsilon} \times \alpha \rho \delta i \alpha \kappa \varsigma$ . The translator reasons that the source of help is not in the multiplicity of royal advisors but in their *hearts*, the locus of wisdom; cf. 11:14 and 15:22.

S treats the doublet in 20:18 similarly to G here; see the comment there.

## 24:7-10

Working with a difficult text and a few variants, G shaped 24:7–10 into a coherent epigram contrasting the wise (24:7a–8a) with the foolish (24:8b–10b):

- 7a Wisdom and good understanding are in the gates of the wise.
- 7b The wise do not turn away from the mouth of the Lord,
- 8a but, instead, they deliberate in assemblies.
- 8b The ignoramus—death comes upon him,
- 9a and the fool dies by sins,
- 9b and impurity will be a blemish for the pestilent,
- 10a in the evil day and in the day of affliction,
- 10b until he comes to an end.

### 24:7

G: (a) σοφία καὶ ἔννοια ἀγαθὴ ἐν πύλαις σοφῶν. (b) σοφοὶ οὐκ ἐκκλίνουσιν ἐκ στόματος κυρίου. "(a) Wisdom and good understanding are in the gates of the wise. (b) The wise do not turn away from the mouth of the Lord" (continues in 24:8a). M 24:7a means that wisdom is too lofty for the fool to attain (see the comment in ABP). But the Hebrew might be

בַּשַּׁעַר לָא יִפְתַּח־פְּיהוּ: 8 מְחַשֵּׁב לְהָרֵע לוֹ בִּעַל־מְזִמְוֹת יִקְרֶאוּ: 9 זִמַּת אָוָּלֶת חַטָּאת וְתוֹעֲבָת לְאָדָם לֵץ: 10 הֻתְרַפִּיתָ בִּיוֹם צָרָה צַר כֹּחֶכָה: 11 הַצַּל לְקָתִים לַמֶּוֶת וּמְטֵים לַלֶּגֶר אָם־תַּחְשׁוֹדָ: 12 כִּי־ תֹּאמַר הֵן לֹא־יָדֵעְנֿוּ זֶה הֵלֹא־תֹּכֵן לָבּׁוֹת | הְוּא־יָבִין וְנֹצֵר וָנַפְּשְׁדְ הַוּא

understood as "wisdom is corals for the fool" and thought to mean that the fool esteems wisdom's value, which would make no sense. Faced with this apparently puzzling statement, the translator substituted a more predictable sentiment. G makes 24:7b and 8a a single sentence. Stich 24:7b is incompatible with M. It is possible that שיהו was understood as פיהו, but שמח יפתח "turn away from" by any stretch. S המחיעם ("throws down") is a clever homoiophonic translation of ("is rebellious, agitated") is also homoiophonic.

### 24:8

G (continuing 24:7b): (a) ἀλλὰ λογίζονται ἐν συνεδρίοις. (b) ἀπαιδεύτοις συναντῷ θάνατος. "(a) but instead, they deliberate in assemblies. (b) The ignoramus—death comes upon him" (continues in 24:9a). In 24:8b, G seems to be reading ארמות יקראו יקראו בער בעל. בער מות יקראו involves a ל/י interchange, probably due to phonic resemblance of the liquids. Kennedy (1928, 91) gives some examples of this, all conjectural. The most persuasive case is Ezek 19:7, where יקראו ("his palaces") should be read for יקראו. אלמנותיו was misunderstood as a singular imperfect + third masculine singular suffix. מומ ה G's source text derives from מומות, with ip lost by parablepsis מית (מ[זמ]ות) מית (sistorted in the next verse as well. But in view of the complexity of these changes, I have not included them in the apparatus.

# 24:9 אימת M S (האו ביאה ) אימת ( מאו או א א א מר א נו א יימר ) ( graph t - יי

G (continuing 24:8a): (a) ἀποθνήσκει δὲ ἄφρων ἐν ἁμαρτίαις. (b) ἀκαθαρσία δὲ ἀνδρὶ λοιμῷ ἐμμολυνθήσεται. "(a) and the fool dies by sins, (b) and impurity will be a blemish for the pestilent man" (continues in 24:10a). This is a loose (and mistaken) reading of a text which differs from M only in <code>, y</code>, trigraphic confusion is very similar to the t/l interchanges noted in LSF §121ab and Kennedy 1928, 65.

S: Khun Kuin mhar and Khuhu Khuhu mha inda. "(a) And the thoughts of the fool are sin, (b) and evil is a man's impurity." The

translator may have misread M 24:9b to mean "and (folly) is an abomination to a cynical man." This would be puzzling, since such a one would *desire* it. To deal with this puzzle, S redid the syntax, making  $\gamma$ <sup>4</sup> the subject and translating it broadly as "evil." S does not always turn to G in difficulties.

# 24:10 אעד [ M ] אעד G (ἕως ἀν) (graph $\mathfrak{V} \to \mathfrak{V}$ , graph $\mathfrak{T} \to \mathfrak{T}$ ) הכחבה [ M ] הכחבה G (ἐκλίηη) (graph $\mathfrak{L}^2 \to \mathfrak{T}$ )

G (continuing 24:9b): (a) ἐν ἡμέρҳ κακῆ καὶ ἐν ἡμέρҳ θλίψεως, (b) ἕως ầν ἐκλίπῃ. "(a) in the evil day and in the day of affliction, (b) until he comes to an end." G has a double translation of ביום ערה in 24:10a and reads ביום ער כחכה ("until he is destroyed") for ער כַחַרָה. The change was occasioned by graphic similarities: עוע (LSF §108) and רער (LSF §131). ἐκλείπειν = כחד in Zech 11:9, 16.

S: ملحة محمد معدة محمد معدة محمد شما وي محمد المحمد المعنام. "And evildoers—evil takes them in the day of tribulation." S just inserts a standard sentiment in place of an obscure verse.

# 24:11 אל G (μή) S (גאם G (μή) rab (norm)

G: (a) ῥῦσαι ἀγομένους εἰς θάνατον (b) καὶ ἐκπρίου κτεινομένους, μὴ φείσῃ. "(a) Deliver those who are being led to death, (b) and redeem those who are being killed. Do not forbear!" ἐκπρίασθαι means "buy off," "buy back." It was added as a parallel to ῥῦσαι. Though G's μή could reflect the negative Dκ, in the light of the rabbinic readings below, it probably represents 'κ. Similarly S κΔ.

Several midrashim (Lev. Rab. 10:4; Pesiq. Rab. 33; S. Eliyahu Zut. 22) cite this verse using אל תחשוך G, S.

 24:12-20

זַדֶע וְהַשִּׁיב לְאָדָם בְּפְעָלוֹ: 13 אֶכָל־בְּנִי דְבַשׁ בִּי־טֵוֹב וְנָשָּ מְׁתוֹק עַל־חִבֶּדְ: 14 בֵּן | דְּעֵה חָכְמָה לְנַפְשֵׁדְ אִם־מֲצָאת וְיֵשׁ אַחַרֵית וְתַלְוּתְדָ לָא תִבְּרַת: פּ 15 אַל־תּבָא'' ( ָרָשָׁע) לִנְוָה צַדֵּיק אַל־תִּשַׁדֵּד רִבְצוֹ: 16 בִּי שָׁבַע | יִפּוֹל צַדֵּיק וְהָם וֹרְשָׁעִים יִבָּשְׁלָוּ בְּרָעֶה: 17 בִּנְפָל <sup>ו</sup>ְאוֹיִבְדָ' אַל־תִּשְׁמֵח וֹבִכָּשְׁלוֹ אַל־יָגָל לָבֵּדְ: 18 בֶּן יִרְאֶה יֵהוָה וְרַע בְּעִיגֵיו וְהַשִּׁיב מַעָלָיו אַפּוֹ: 19 אַל־תִּקָנָי זַב

1:24). V modifies the rationalization: *si dixeris vires non suppetunt* ("If you say, 'There are not strengths in store...").

24:14

By itself, M's אחרית ("there is an end") might sound ominous. G makes it clear that the intention is beneficial, hence בדמו אמאא א דבאבעדא סט ("your [sg] end (will be) good").

24:15 אבא\* ≈ G (προσαγάγης) ] תְאֶרָב (metath אב → בא, near dittog גרב → ב)

(gloss, explic) (באָע א G (ἀσεβῆ) S (באָ

M reads: "Do not lie in wait, O wicked man, for the habitation of the righteous man!" This is awkward, because one does not "lie in wait" for "habitations" but rather for persons or prey. The reading indicated by G is  $\pi$ , which is correct consonantally but is better vocalized  $\pi$  (G-stem). To *enter* another's property, when the context refers to an

illegitimate act, means to encroach upon it and misappropriate it, as in 23:10b: "nor enter the fields of orphans." The vocative, רשע, "O wicked man," looks like an addition inserted by a scribe who was puzzled why it would be illegitimate to "enter" the habitation of the righteous. The puzzlement was unnecessary, however, because the illegitimacy is adequately indicated by the second stich. The addition overweights the first stich and introduces a new addressee. The direct address to the wicked is of doubtful originality, because Proverbs does not try to speak to or influence the behavior of the wicked, which is assumed to be hopeless; see ABP, comment. The verse as emended should be translated, "Do not enter the habitation of the righteous. Don't plunder the field of his pasturing!"

24:17 אויבך cf. Mp G (ὁ ἐχθρός σου) S (בעגרבר) V (inimicus tuus) T (דבעיל דבבך) M<sup>K</sup>

The Mp of L and Y marks אויביך מ' אויביך ("an extra yod"), which tells the scribe to write אויביך though it is to be understood and pronounced as a singular. Still, the form אויבך is contextually correct, matching the singular of אל יגל לבך M's אויב ("let your heart not rejoice") is translated by G as  $\mu\eta$  έπαίρου ("do not exalt yourself"). G's έπαίρου is an attempt to refine the nuance of rejoicing at the fall of one's enemy. Possibly the underlying idea is that it is inevitable to rejoice when one's enemy falls, but that does not give license to boast and be arrogant. ἐπαίρου is unlikely to represent יגר', because (unlike in 23:24) there is no letter in that would produce a T by graphic resemblance.

24:21 שׁוֹנִים ( (μηθετέρφ αὐτῶν ) שׁוֹנִים ( near haplog שׁוֹנִים ( near haplog בֹּעָבא) ( near haplog הם → ס, orth)

G: (a) φοβοῦ τὸν θεόν, υἰέ, καὶ βασιλέα (b) καὶ μηθετέρῳ αὐτῶν ἀπειθήσης. "(a) Fear God, O son, and the king, (b) and do not disobey either of them." (μηθ)ετέρῳ αὐτῶν = שְׁנֵיָהם for M's שׁוֹנִים ("with dissenters"?). Instead of M's אָל תתערר ("do not disobey") = אל תתערר ("transgress"). Jan de Waard considers G's "do not disobey" a case of exegetical metathesis (1993, 25), but there is no reason to prefer this explanation over textual metathesis. V translates שׁוֹנִים a detractoribus ("detractors").

### **24:22** fin ] + 10 stichoi G (22*a*-22*e*) (elab)

G: (a) ἐξαίφνης γὰρ τείσονται τοὺς ἀσεβεῖς, (b) τὰς δὲ τιμωρίας ἀμφοτέρων τίς γνώσεται; "(a) For suddenly they [sc. God and king] will punish the wicked, (b) and who knows the punishments of either?" In M, it is unclear whether this verse describes the punishment suffered by the unclear whether this verse describes the punishment suffered by the ("dissenters"?) or the disaster they cause. G assumes the former and recasts the verse in the active voice. The verse then clearly describes the actions of the two powers and gives them an explicitly moral motivation. This recasting suits the message of the following verses as well.

S: (a) = M; (b) איזעדס, דא גערס (b) and who knows the end of their years?" That is to say, one does not know how fools will end up. S derives שניהם from לעניהם ("year"), in spite of having associated שני in 24:21. The translator certainly knew that שונים

"the two of." Different treatment of the similar words in two adjacent verses looks like deliberate wordplay.

### G 24:21-22e

Together with 24:21 and 22, five additional couplets in G (24:22a-22e) form a prose epigram on the power and responsibilities of the king and the danger of his wrath. The evidence is insufficient to justify a Hebrew reconstruction. BAP notes that in G, the literary unit ends at 24:20. The copyist of G<sup>B</sup> placed a large dash at this point, apparently to mark a perceived caesura. S and V lack the additional verses. The epigram in its entirety reads:

(24:21) φοβοῦ τὸν θεόν, υἱέ, καὶ βασιλέα, καὶ μηθετέρω αὐτῶν ἀπειθήσης. (22) ἐξαίφνης γὰρ τείσονται τοὺς ἀσεβεῖς, τὰς δὲ τιμωρίας ἀμφοτέρων τίς γνώσεται; (22a) λόγον φυλασσόμενος υἱὸς ἀπωλείας ἐκτὸς ἔσται, δεχόμενος δὲ ἐδέξατο αὐτόν. (22b) μηδὲν ψεῦδος ἀπὸ γλώσσης βασιλεῖ λεγέσθω, καὶ οὐδὲν ψεῦδος ἀπὸ γλώσσης αὐτοῦ οὐ μὴ ἐξέλθῃ. (22c) μάχαιρα γλῶσσα βασιλέως καὶ οὐ σαρκίνη, ὅς δ'ἂν παραδοθῆ, συντριβήσεται. (22d) ἐἀν γὰρ ὀξυνθῆ ὁ θυμὸς αὐτοῦ, σὺν νεύροις ἀνθρώπους ἀναλίσκει (22e) καὶ ὀστᾶ ἀνθρώπων κατατρώγει καὶ συγκαίει ὥσπερ φλὸξ ὥστε ἄβρωτα εἶναι νεοσσοῖς ἀετῶν.

(24:21) Fear God, O son, and the king, and do not disobey either of them. (22) For suddenly they will punish the wicked, and who knows what punishments either may bring? (22*a*) A son who keeps the command will be free from destruction, and when he receives (a command), he truly absorbs it. (22*b*) Let no tongue speak falsehood to the king, nor let any falsehood go forth from his own tongue. (22*c*) For the tongue of the king is a sword, not flesh, and whoever is delivered (to its power) will be shattered. (22*d*) For should his anger be provoked, he destroys men with sinews, (22*e*) and the bones of men he gnaws up. Like a flame he burns them up, so that even young eagles cannot eat them.

This passage recalls a long proverb-poem in the Aramaic Ahiqar (100a– 102 [TAD 1.1.84–86]); see ABP at 16:14. Since Ahiqar was popular in the Hellenistic period and known in Greek, there is a good chance that Ahiqar's influence on G here was direct.

After 24:22*e*, G proceeds as follows (using M's numbering): 30:1–14; 24:23–34; 30:15–31:9; 25:1–29:27; 31:10–31. Also, 31:25–26 are reversed.

# גַּם־אֵּלֶה לַחֲכָמֵים

הַבּּר־פָּגִים בְּמִשְׁפָּט בַּל־טְוֹב: 24 אָׁמֵׁר | לְרָשָׁע צַדֶּיק אָתָה יִקְבָהוּ עַמֵּים יִזְעָמְוּהוּ לְאָמִים: 25 וְלַמּוֹכִיחִים יִנְעָם וַעֲלֵיהֶם תְּבִוֹא בִרְבַּת־ טְוֹב: 26 שְׁפָתִים יִשֶׁק מֵׁשִׁׁיב דְּבָרִים נְכֹחִים: 27 הְּבֵן בַּחוּץ | מְלַאַכְשֶׁדְה בַּשָּׁדָה לֵדְ אַׁחַר וּבָנִית בֵיתֶד: פּ 28 אַל־ תְּהֵי עֵד־חִנְּם בְּרֵעֶד וֹהַפִּתִית בִּשְׂפָתֶידָ: 29 אַל־תּאׁמַר כַּאֲשֶׁר

### 24:23

G: (a) Ταῦτα δὲ λέγω ὑμῖν τοῖς σοφοῖς ἐπιγινώσκειν. (b) αἰδεῖσθαι πρόσωπον ἐν κρίσει οὐ καλόν. "(a) These things I say to you, wise men, to make (them) known. (b) To respect a face in judgment is not good." The sentence that in M is properly an ascription of authorship ("These too are by the wise") G turns into an address *to* the wise. Avoiding ascription of authorship to others than Solomon is consistent with the translator's procedure; see the comment on 10:1. הכר Set to double use: to make 24:23a into a statement and to serve as the subject in 24:23b.

What are "these things" that the speaker says to the wise? As Jäger observes,  $\tau \alpha \tilde{\upsilon} \tau \alpha \delta \tilde{\varepsilon}$  refers back to  $\tau \dot{\alpha} \delta \varepsilon$  in 30:1c. In G, 30:1–14 have been dislocated and appear immediately before 24:23. Thus "these things" in G 24:23 are the cautionary words about the limits of human wisdom in the preceding unit. The mutual entailment of 24:23a and 30:1–14 indicates that the dislocation of 30:1–14 was already present in the translator's source text rather than being his innovation. The translator is just trying to discern meaningful organization in the text before him. In this organization, G 24:23a *concludes* the unit 30:1–14 + 24:23a, and G 24:23b begins a new string of sayings.

V is close to M but construes 24:23a similarly to G: *haec quoque sapientibus* ("These also are *to* the wise").

S: (a)  $rac{}{}$   $rac{}$   $rac{}$   $rac{}{}$   $rac{}{}$   $rac{}{}$  r

23

### 24:25

S: محصقته ("the poor") in the Leiden edition is an error (metathesis) for حتصته ("rebukers"), which is found in the Mosul edition.

# 24:26

G: χείλη δὲ φιλήσουσιν ἀποκρινόμενα λόγους ἀγαθούς [G<sup>BS\*</sup>; vars: σοφούς G<sup>ScA</sup>; ὀρθούς G<sup>V</sup>]. "And they will kiss lips that answer (with) good [vars: 'wise'; 'honest'] words." "They" refers to the rebukers mentioned in 24:25. The subject of "kiss" is indefinite. G is a strained construal of a difficult Hebrew text (= M).

24:27 לך אחר MS (הכאו-) ] ללך אחרי G (אמ' הסףבטט אמד<br/>ל $\pi$ וסשי $(near \ dittog \ i \rightarrow i')$ 

G: (a) ἑτοίμαζε εἰς τὴν ἔξοδον τὰ ἔργα σου (b) καὶ παρασκευάζου εἰς τὸν ἀγρὸν (c) καὶ πορεύου κατόπισθέν μου (d) καὶ ἀνοικοδομήσεις τὸν οἶκόν σου. "(a) Make ready your [sg] works for the departure, (b) and make preparations for the field. (c) And come after me, (d) and you shall build up your house." G read 'ֻ <br/> [ \* אַחָר ("go after me")—a near dittography with the to f the next word—for M's '<br/> לך אחר ("to you"; "after"). This change led to translational deviations from M. G reads like an invitation to go on a journey. Gerleman (1956, 29) reads this verse as a reminder of death. To be sure, this idea might make sense for an Egyptian, who would hope to depart for the paradise in the Field of Reeds and live in his Eternal House, but Alexandrian Jews would not have accepted this foreign belief.

## 24:28

G translates ברעך ("against your neighbor") as  $\dot{\epsilon}\pi\dot{\iota}$  סטי הסלוד ("against your fellow citizen"); see 11:12.

# 24:29

G: (a) μὴ εἶπῃς Ὁν τρόπον ἐχρήσατό μοι χρήσομαι αὐτῷ, (b) τείσομαι δὲ αὐτὸν ἅ με ἦδίκησεν. "(a) Do not say, 'As he treated me so I will treat him; (b) I will repay him (by) the harm he has done me." The addition of "the harm" leaves no doubt that "as he treated me" refers to an inimical act.

S = M but connects this verse to the preceding by a waw-isrda da

עֲשָׁה־לֵי בֵּן אֲשֲשָׁה־לֵּוֹ אָשָׁיב לָאַישׁ כְּפְעֲלוֹ: 30 עַל־שְׁדֵה אִישׁ־עָצֵל עָבֵרְתִּי וְעַל־כָּרֶם אָדֶם חֲסַר־לֵב: 31 וְהַנֵּה עָלֶָה כָלוֹ | קַמְשוֹּיִם כָּפְוּ פְּנְיו חֲרֻלִּים וְגֶדֶר אֲבָנְיו גֶהֶרֶסָה: 32 וְאֶחֶזֶה אֲנֹכְי אָשִׁית לִבֵּי רְאִיתִי לְקַחְתִי מוּסֵר: 33 מְעַט שֵׁנוֹת מְעַט תְּנוּמֵוֹת מְעַט | חִבֵּק יִדַיִם לִשְׁבְּב: 34 וּבֵא־מִתְהַלֵּדְ רֵישֶׁדְ וּמַחְסֹרֶידְ כְּאֵישׁ מְגֵן: פ נוּזַר מַשְׁכָר גַּבָי אָשָׁית מְעַט שְׁנוֹת מְעַט הְנוּמֵוֹת מְעַט וּ חַבָּק לִשְׁבְּב: 34 וּבֵא־מִתְהַלֵּדְ רֵישֶׁדְ וּמַחְסֹרֶידְ כְּאֵישׁ מְגֵן: נוּבר גַּשָׁר הַאָּשָׁר הַיָּבָי שָׁלֹמֵה אָשָׁר הֵטְהָיָהוּ אַנְשֵׁי | חָזָקַיָּה מֵלָדְיהוּנְדָה: 2 כְּבָר אֵלְהָים הַסִתָּר

("And do not say, etc.")—thereby making the verse into a warning against using false testimony to get revenge.

### 24:30-34

G composes an epigram that takes imagery and phraseology from M but uses them very differently, so much so that G cannot be explained in detail from the Hebrew. In 24:30–32, the lazy, foolish man is compared to a field. Then in 24:33, the sluggard announces his intention to be lazy, or perhaps the sage speaks these words as a hypothetical thought—something that one might think but should not. Then, in 24:34, the sage follows up with a warning against such behavior.

(24:30) A foolish man is like a field and the man who lacks sense is like a vineyard. (31) If you leave it alone, it will go to waste and run entirely to weeds and be abandoned. And the walls of stones are demolished. (32) Later I reconsidered, looking at (him), so as to take a lesson. (33) *The sluggard speaks*: "I snooze a little, I sleep a while, I clasp my bosom with (my) hands." (34) *The sage says*: "If you do this, your poverty will arrive advancing, and your neediness (arrive) like a good runner."

### 24:33

G: (a)  $\partial\lambda$ ίγον νυστάζω,  $\partial\lambda$ ίγον δὲ καθυπνῶ, (b)  $\partial\lambda$ ίγον δὲ ἐναγκαλίζομαι χερσὶν στήθη. "(a) I snooze a while, I sleep a while, (b) I clasp my bosom with (my) hands a while." In G 24:33, the sluggard, not the sage, is speaking. G uses the second person in 6:10; here it uses the first. In fact, only the third person is appropriate. In word choice as well, G 24:33 differs from G 6:10.

S sets the verse in the second-person singular, as in 6:10–11, where it was influenced by G. T = S.

### 24:34

G: (a) ἐἀν δὲ τοῦτο ποιῆς, ἥξει προπορευομένη ἡ πενία σου, (b) καὶ ἡ ἐνδειά σου ὥσπερ ἀγαθὸς δρομεύς. "(a) If you do this, your poverty will arrive after advancing, (b) and your neediness, like a good runner." In G 24:34 and 6:11, the fast runner is called "good" (rather than "fast") even though he is the analogy for a misfortune. (See the comment on 6:11.) This harks back to Plato's *Hippias Minor* 373c–e, in which "goodness" is defined as the skill or potential for achieving goals relevant to a particular activity, an example being the "good runner."

(G continues with 30:15–31:9, followed by 25:1–29:27 + 31:10–31.)

### 25:1

G: (a) Αὗται αἱ παιδεῖαι [G<sup>B</sup>; παροιμίαι G<sup>ScA</sup>] Σαλωμῶντος αἱ ἀδιάχριτοι, (b) ἁς ἐξεγράψαντο οἱ φίλοι Εζεκιου τοῦ βασιλέως τῆς Ιουδαίας. "(a) These are the miscellaneous teachings [var: 'proverbs'] of Solomon (b) which the friends of Ezekias king of Judah copied out." BAP understands "these" as retrospective. (See BAP's theory of the structure of G-Proverbs; BAP 31-32.) παροιμίαι looks like an adjustment to 1:1. άδιάκριτοι, a hapax in G, has nothing corresponding in M. Some suggested meanings of άδιάκριτοι are (1) "qui ne sont pas triées" (BAP); (2) "mixed" (GELS); (3) "undistinguishable," "mixed," "not discriminated" (LSJ); (4) "assortment" (MGELS); and (5) בדים ("profound") (S) (T עמיקי). (V omits the word.) In Jas 3:17 άδιάχριτος means "impartial," "without prejudice." It is a quality of the wisdom that comes from God. Definitions 2-4 are basically the same and accord with the common meaning of the antonym, διαχρίνειν "to distinguish." In my view, ἀδιάχριτοι here implies a significant distinction between the preceding collections and the present one. The former were composed of sayings written and organized by Solomon, whereas the latter collection comprises a variety of sayings that were created by him but written down only later by Hezekiah's men.

דְּבֶר וּכְבָד מְּלָכִים חֲקָׂר דְּבֵר: 3 שְׁמַיִם לֲרוּם וָאֲרָץ לְעָׂמֶק וְלֵב מְלָכִים אֵיז חֵקָר: 4 הָגִוֹ סִיגִים מִכְּסֶף וַיֵּצֵא לַצִּרָף כֶּלִי: 5 הָגִוֹ רֲשָׁע לִפְנֵי־מֶלֶד וְיִכָּוֹן בַּצֶדֶק כִּסְאִוֹ: 6 אַל־תִּחְהַדָּר לִפְנֵי־מֶלֶד וּבִמְקוֹם גִּדִלִים אֲל־תַּעֲמְד: 7 כִּי טִוֹב אֲמָר־לְדָׁ עֲלֵה הֵנָּה מֵהַשְׁפֵּילְדָ לִפְנֵי נְדֵיב אֲשֶׁר רָאַוּ עֵינֵידָ: 8 אַל־תַּצֵא לָרִב מַׁהַר כָּן מַה־תַּעֲשָׁה בְּאַחַרִיתֵה בְּהַכְלִים אֹתְדָ בַעֶדּ: 9 רִיבְדָ רֵיב אֶת־בַעֶד וְסוֹד אַחֵר אַל־תְּגַל: 10 כֵּן־יְחַסֶּדְדָ שֹׁמֵע וְדִבָּתְדָׁ לָא תָשׁוּב: 11 תַפּוּחֵי זָהָב

25:2 הקר  $M \approx S$  ( $\kappa \rightarrow \pi$ הקר [ ( $\kappa \rightarrow \pi$  חקר  $M \approx S$  ( $\pi \mu \tilde{\mu}$ ) (graph  $\rightarrow \pi$ 

G: (a) = M; (b) δόξα δὲ βασιλέως τιμῷ πράγματα. "(b) And the glory of a king brings honor to deeds." τιμῷ = ¬μῷr (Jäger); similarly in 25:27. The deeds in question are, given the context, affairs of state (cf. Esth 3:13ef).

25:3

G: (a) ≈ M; (b) καρδία δὲ βασιλέως ἀνεξέλεγκτος. "(b) But the heart of a king is irrefutable [or 'beyond criticism']." ἀνεξέλεγκτος can mean "unfathomable" (BAP: "insondable") or "unchastised," as in Prov 10:17. In 18:17, ¬קר is translated ἐλέγχεται ("refuted").

# 25:4 בלי M S (בלי [ (גאיבא G (ắπαν) (graph '→ ۱)

G: (a) κτεΐνε ἀσεβεῖς ἐκ προσώπου βασιλέως, (b) καὶ κατορθώσει ἐν

δικαιοσύνη ὁ θρόνος αὐτοῦ. "(a) Kill the wicked [pl] before the king, (b) and he will establish his throne in righteousness." G renders גם הגד גדבוֹעב. In both verses 4 and 5, G understands גם הגד smite" but varies the translation to make it fit each verse. Again, this is a type of translational flexibility that rather than being merely "free" responds to the needs of the context, as the translator understands it. The present verse calls for the execution of state enemies.

S translates אהריס as הגר ("let [evildoers] be broken")  $\approx$  G, whereas in 25:4 it used ("select").

### 25:7

The versions do not provide variants for the puzzling sentence fragment in M 25:7c (lit. "what your eyes have seen"). G adds an imperative  $\lambda \epsilon \gamma \epsilon$ : "What your eyes have seen, declare." S and V attach the stich to the following verse, but the connection is strained.

# 25:9 ריבך ריב את רעך MSV ] > G (homoi רעך ∩רעך)

וסוד M S (היהיה) ] אוסוד G (מימעט'רנו) (graph ר  $\rightarrow$  ר)

# **25:10** fin ] + 4 stichoi G (10c–10*c*) (elab)

G (continuing 25:9): (a) μή σε ὀνειδίση μεν ὁ φίλος, (b) ἡ δὲ μάχη σου καὶ ἡ ἔχθρα οὐκ ἀπέσται, (c) ἀλλ' ἔσται σοι ἴση θανάτω. "(a) so that your friend not insult you, (b) and your [sg] quarrel and hatred [consequently] not cease, (c) but (rather) will be with you like death." G clarifies M expansively: "friend" for "listener" and "quarrel and hatred" for "or (which is translated in a variety of ways in G, never as "slander"), and "be with you like death" (i.e., ineluctably or strongly, cf. Cant 8:6) for "not return." G has a double translation of "not return": as "not cease" and "be with you like death." G 25:10c is part of an expansion that continues in 10*a*.

### 25:10a

G: (a) χάρις καὶ φιλία ἐλευθεροῖ, (b) ὡς τήρησον σεαυτῷ, ἵνα μὴ ἐπονείδιστος γένῃ, (c) ἀλλὰ φύλαξον τὰς ὁδούς σου εὐσυναλλάκτως. "(a) Graciousness and love make (one) free. (b) Guard them for yourself, so that you do

25:11-19

בְּמַשְׂכִּיּוֹת כְּסָף דְׁבָּר דְּבָר עַל־אָפְנֵיו: 12 גָזֶם זְהָב וַחַלִי־בָּתָם מוֹכִיח חְּכָם עַל־אָׂזֶן שׂמֵעַת: 13 כְּצִנַּת־שָׁׁלֶג | בְּיוֹם קָצִיר צִיר גָאָמָן לְשׁלְחֵיו וְגָשָׁם אֲדֹנֵיו יָשִׁיב: פּ 14 גְשִׁיאַים וְרוּחַ וְגָשָׁם אֲיִן אִישׁ מִתְהַלֵּל בְּמַתַּת־שֶׁקָר: 15 בְּאַׁרָד אֲפַּים יְפֻתָּה קָצֵין וְלָשִׁוֹן רַׁכָּה תִּשְׁבָּר־גָּרֶם: 16 דְבַשׁ מֻצָּאתָ אֶכָל דַיָּדֶ פָּן־תִׁשְׂבָעָנוּ וַהַמֵאתוֹ: 17 הֹקַר וְבְלָד מִבֵּית בעֵד פֶּן־יִשְׂבָעַד וּשְׁנַאֶד: 18 מַפִּיץ וְחָרֶב וְתַץ שָׁגָוּן אִישׁ עֹנֶה בְׁבִעֵה עַדַ שֶׁקָר: 19 שֵׁן רָעָה וְרָגָל מוּעֶדָת מִבְעָח

not become disgraceful. (c) But guard your ways peacefully." This addition summarizes 25:7c-10.

### 25:11

G: (a) = M; (b) οὕτως εἰπεῖν λόγον. "(b) thus it is to say a word." εἰπεῖν λόγον implicitly vocalizes אָפנין. G skips the difficult על אפניו. There is no mechanical reason for the loss. Difficult phrases can be omitted when the translator can reasonably conclude (as here) that the gist of the verse does not require them. Omissions of obscure phrases are probably the translator's doing. A copyist could simply replicate the text without comprehension.

V (qui loquitur verbum) α' θ' (λαλῶν ῥῆμα) σ' (λαλῶν λόγον) S ( $\tau = - \sigma$ 

S omits של אפניו = G. It is unusual for S to follow G in omissions, but S too was probably perplexed by this phrase, as are modern commentators.

### 25:13

G: (a) ὥσπερ ἔξοδος χιόνος ἐν ἀμήτῷ κατὰ καῦμα ἀφελεῖ, (b) οὕτως ἄγγελος πιστὸς τοὺς ἀποστείλαντας αὐτόν, (c) ψυχὰς γὰρ τῶν αὐτῷ χρωμένων ἀφελεῖ. "(a) Just as the going forth of snow in harvest is beneficial against heat, (b) so is a reliable messenger to those who sent him, (c) for he benefits the souls of those who employ him." Possible variants are ὥσπερ ἔξοδος = תצאת ad κατὰ καῦμα = LAND. Still, since G is being expansive in explaining the analogy and since the retroversions are not supported by graphic similarities, we can best account for the differences as expansive explications of M's analogy of the "chill of snow in harvest." zuch souls are the sense of "chill" and perhaps not known

to the translator (though it is translated correctly in Sir 43:20 and צנ"ן means "be cold" in RH).

S געאת ("descends") explicates כצאת from context and does not depend on G.

### 25:14

G: (a) ὥσπερ ἄνεμοι καὶ νέφη καὶ ὑετοὶ ἐπιφανέστατοι, (b) οὕτως οἱ καυχώμενοι ἐπὶ δόσει ψευδεῖ. "(a) As the wind and clouds and rains are very conspicuous (b) so are those who boast over a false gift." M is to be translated, "Clouds with wind and no rain: a man who boasts of a gift that disappoints." The proverb is something of a riddle that leaves it to the reader to figure out how the metaphor (in the first stich) applies to the referent (in the second). G explains the metaphor by construing cactually "clouds") as "lifted up" (from κων "lift up"), hence ἐπιφανέστατοι ("evident," "conspicuous"). G's proverb derides insincere gift givers for their ostentation (BAP). S  $\approx$  M.

### 25:15

G: (a) ἐν μακροθυμία εὐοδία βασιλεῦσιν, (b) γλῶσσα δὲ μαλακὴ συντρίβει ὀστᾶ. "(a) In patience there is success for kings, (b) and a soft tongue breaks bones." εὐοδία comes from associating 'eπ 'with similar-looking words that mean "make wide, spacious," such as 'reπ' in Prov 24:28 and reπ' in Gen 9:27. (In both, they are rendered πλατύναι and connote good fortune.) In Gen 9:27. (In both, they are rendered πλατύναι and connote good fortune.) In Gen 9:27. 'extually from 'eπ) has a sense close to εὐοδία ("May God make broad for Japheth"). βασιλεῦσιν ≈ 'σ̄ε; see the comment on 19:6. G's proverb instructs kings on how to succeed, while M's tells courtiers how to influence officials.

# 25:19 מועדת מבטח M S (הטבאה ... אמבעה ] > G (unc)

G: (a) όδοὺς κακοῦ καὶ ποὺς παρανόμου (b) ὀλεῖται ἐν ἡμέρα κακῆ. "(a) The tooth of an evil man and the foot of a transgressor (b) will perish in an evil day." The absence of מועדת מבטח in G's source text is shown by the way G translates רגל בוגד as a meaningful phrase. It is hard to explain the parablepsis, but there is no reason for a deliberate omission. By translating "the tooth of an evil man" (against the grammar) rather than "a bad tooth," G, as often, replaces a difficult phrase with an expected message, here the certainty of retribution. G naturally understands רעה" chart than the less common  $\gamma$  "shaky."

S (T) too understands רעה as הכידא (= גרעה (כמאמא = ( גרעה).

V putridus ("rotten") is a narrow rendering of "bad."

בּׁוֹגֵד בְּיָוֹם צָרֶה: 20 <sup>ר</sup> חַמָּץ עַל־גָתֶר וְשָׁר בַּשִּׁרִים עַל עָב־ רֶע: פּ 21 אִם־רָעֵב שַׂנַאֲדְ הַאֲכִלֵהוּ לֶחֶם וְאִם־צְׁמֵא הַשְׁמֵהוּ מֵיִם: 22 כְּי גֶחָלִים אֲתָּה חֹתֶה עַל־רֹאֹשֶׁוֹ וֵיהוָה יְשַׁעֶּם־לֶדְ: 23 רַוּחַ צְפוֹן הְּחַוֹלֵל גָּשֶׁם וּפָגִים נִזְעָמִים לְשִׁוֹן סֶתֶר: 24 טוֹב שֶׁבֶת

25:20 init ] + מִעֲדָה בָּגֶד בְּיָוֹם קָרָה M ≈ S (האבא גם גיאָגא) איז איז (cf. > G) (distorted dittog) (cf. > G) (cf. > G

G: (a) ὥσπερ ὄξος ἕλχει ἀσύμφορον, (b) οὕτως προσπεσὸν πάθος ἐν σώματι χαρδίαν λυπεῖ. "(a) As vinegar is unpleasant [or 'useless'] to a wound, (b) so does suffering that befalls the body grieve the heart." G does not represent the first four words of the verse, העדה בגד ביום קרה. They look like a distorted dittography of the end of 25:19 (Hitzig 1858), though the development cannot be traced exactly. The matter is complicated by the fact that the elaboration in G 25:20*a* is aware at least of ¬ג

Nevertheless, M's 25:20a—"One who removes (מעדה) a garment on a cold day"—looks like a scribal accident, and its absence from G supports this. The words do not relate to the foregoing or the following, nor do they provide a suitable analogy to the person who sings songs to a sad heart (25:20b). Moreover, unlike the actions mentioned in 25:19a and 20c, removing one's garment on a cold day is imprudent to the one who does it but does not cause others discomfort. De Waard (2007, 1–4) says that assonance with the preceding favors the originality of the clause, but assonance would arise anyway by an accidental repetition.

G 25:20b explains (wrongly) the meaning of the analogy. בשרים is construed as בשרים "flesh" (pl in 14:30).

 $S \approx M$ , but adding  $m_{\star}$  ("from his fellow") in 25:20a.

25:20*a* 

G: (a) ὥσπερ σὴς ἱματίω καὶ σκώληξ ξύλω, (b) οὕτως λύπη ἀνδρὸς βλάπτει καρδίαν. "(a) As a moth (does) to a garment and a worm to wood, (b) so does a man's suffering harm (his) heart." G creates a new proverb from components of both M and G 25:20.

S translates G 25:20*a* exactly. T incorporates the Syriac ( $T^Z$  בולטיתא). בולטיתא should certainly be emended to גאוי (Pinkuss), though the former was the form that reached T ( $T^U$ ).

### 25:21

G and S omit "bread" and "water" as adequately implicit in "feed" and "give drink." The Greek and Syriac of Rom 12:20 follow this form of the verse.

### 25:22

G: (a) τοῦτο γὰρ ποιῶν ἀνθρακας πυρὸς σωρεύσεις ἐπὶ τὴν κεφαλὴν αὐτοῦ,
(b) ὁ δὲ κύριος ἀνταποδώσει σοι ἀγαθά. "(a) For when you do this, you will heap coals of fire on his head, (b) and the Lord will repay you good."

G starts with a transitional τοῦτο γὰρ ποιῶν ("for when you do this"). Thus in Rom 12:20. G adds ἀγαθά ("[with] good"), to make it clear that the repayment is a reward. The translator clarifies a sentence that was clear to start with.

### 25:23

G: (a) ἄνεμος βορέας ἐξεγείρει νέφη, (b) πρόσωπον δὲ ἀναιδὲς γλῶσσαν ἐρεθίζει. "(a) The north wind stirs up clouds, (b) and a brazen face provokes the tongue." As in 25:14 and 26:1, G differs from M in its treatment of meteorological phenomena. The correspondences ἀναιδής = <code>Dyna</code> and ἐρεθίζειν = <code>Dyna</code> are unique. They seem to arise from the assumption that the metaphor in 25:23a refers to provocation and that the "face" is what is being compared to the wind. G reorganizes the syntax in 25:23b.

S: אישלים אישים אישיישים אישים אישים אישיישים אישים אישיים אישים אישי אישיים אישים אישי

יתחולל is translated "dissipates," "drives away" by  $\sigma'$  ( $\delta\iota\alpha\lambda \dot{\upsilon}\epsilon\iota$ ) and V (*dissipat*). According to BHQ, these readings, "but also those of Sa'adia and Ibn Ezra [the commentary is actually Moshe Qimhi's (MVF)] 'correct' the proverb according to the climatic situation of Palestine." The problem facing commentators is that in Israel it is the west wind, not the north, that brings rain (1 Kgs 18:44), and the author of the proverb

עַל־פִּנַּת־גָג מֵאֵשֶׁת 'מְׁדוֹנִים' וּבֵית חֶבֶר: 25 מַיִם קָרִים עַל־גָפָשׁ עַיַבֵּה וּשְׁמוּעָה טוֹבָה מֵאֶרֶץ מֶרְחֶק: 26 מַעְיָן גִרְפָּשׂ וּמָקוֹר מְשְׁחֶת צַׁדִּׁיק מֵט לִפְנֵי־רָשֵׁע: 27 אֶכָל דְּבַשׁ הַרְבָּוֹת לֹא־טָוֹב 'וְהֹמֵר דְּבָרים

would have been familiar with Palestinian meteorology. BHQ's explanation of these readings is correct. The translators and commentators, like modern commentators, were probably influenced by 1 Kgs 18:44.

To be precise, Sa'adia explains תחולל המולל מגישים as "make cease." Moshe Qimhi explains the word first as "prevent" (an explanation he ascribes to his father Joseph, who uses an impossible etymology), then as "give birth to." But "north" here is, more precisely, northwest (Toy), and geographical precision is not required in proverbial sayings (especially since תחולל is part of a complex wordplay; see ABP). In any case, the proverb does not say that *only* the north wind has this effect. As noted in ABP, תחולל can mean "make whirl" as well as "produce." (A storm is described as "מתחולל" whirling about" in Jer 23:19.) The powerful north wind makes rain swirl about during a storm, and a secretive tongue makes faces "stormy" or angry at the talebearer.

# 25:24 מִדְיָנִים [ M<sup>K</sup> מִדְיָנִים ( dial? norm)

On the *ketiv-qere*, see the comment on 6:14. G = M. Contrast the freer treatment of the near-doublets in 21:9 and 21:19.

S translates הבית חבר as איש איש ובית ("and in a house of factions"), a guess at an unusual phrase, which was ignored in 21:9.

G: (a) ὥσπερ ὕδωρ ψυχρὸν ψυχῆ διψώσῃ προσηνές, (b) οὕτως ἀγγελία ἀγαθὴ ἐκ γῆς μακρόθεν. "(a) As cold water is pleasant to the thirsting soul, (b) thus is good news from a far land." Whereas M makes the comparison by juxtaposing two clauses, G adds adverbs of comparison to phrase the clauses as an explicit ratio (so also S); likewise in 25:26 and 26:3. G thereby creates a cohesive series of six comparisons: 25:25, 26, 28; 26:1, 2, 3 (BAP). S follows the syntax of G in these verses but chooses its own way of dealing with the Hebrew; e.g., 25:25 adds (a rather superfluous)  $\prec h \prec 1$  ("that comes").

### 25:26

G: (a) ὥσπερ εἴ τις πηγήν φράσσοι καὶ ὕδατος ἔξοδον λυμαίνοιτο, (b)

<sup>25:25</sup> 

οὕτως ἄκοσμον δίκαιον πεπτωκέναι ἐνώπιον ἀσεβοῦς. "(a) Just as if someone were to block up a spring and spoil the exit of water, (b) thus is it improper for a righteous person to fall before the impious [pl]." G changes 25:26a from passive to active, thereby making the first stich of the comparison better aligned with the second. By calling the tottering of the righteous man ἄκοσμος (lit. "unlovely"), G defines the situation as a matter of propriety or aesthetics. This is not far from M-Proverbs' own "moral aesthetics"; see ABP 2.967–76. S 25:26a = G; S 25:26b = M.

25:27 אוהקר  $* = G (\tau \iota \mu \tilde{\alpha} \nu) ] א וחָקר M S (ג געריב'); אוהקר ' σ' (ἐξερευνῶν δέ);$  $ה <math>* = \theta' (\kappa \alpha i ἐξερευνῶντι) V (qui scrutator est) (M S σ' θ' V: graph$  $<math>\rightarrow \Pi$ )

דברים (גלאסטג) S (דברים (graph  $\neg \rightarrow \neg$ , graph  $\neg \rightarrow \neg$ , orth) כ  $\neg \rightarrow \neg$ , orth)  $\mathfrak{q} \subset \mathfrak{s}$  ממ (أد محتد א  $\mathfrak{s} \to \mathfrak{s}$  ( (أر محتد مد  $\mathfrak{s} \to \mathfrak{s}$  ) א מ (haplog מכן א  $\mathfrak{s} \to \mathfrak{s}$  , orth, vocal) I propose emending to והקר דְבָרִים מְכָבָד. The differences between this reading and the consonants of M are common graphic mutations. The theoretical stages leading to M are as follows: דברם , וחקר  $\rightarrow$  דברם  $\rightarrow$  כבדם. The component steps of this change were, theoretically: (1) ה  $\rightarrow$   $\Pi$ , (2)  $\rightarrow$   $\rightarrow$  and  $\top$   $\rightarrow$   $\supset$ , (3)  $\alpha$   $\rightarrow$   $\alpha$  (haplography of  $\alpha$  with the last letter of דברם), (4) mater ו. When analyzed into components, the process seems improbably intricate. However, I do not think that the change occurred in a series of unrelated accidents but rather that a scribe looked at a group of consonants and grasped them wrongly (see §4.3.3). Taken as a whole, the change is not unlikely, especially if we picture the text as written continuously and without final letters: והקר = והקרדברממכבד הברים מכבד. As reconstructed, the proverb reads, "Eating honey too much is not good, and being sparing of words is honorable." In other words, don't overdo good things, not even speech. This saying uses the same analogy as Prov 25:16 and, as emended, the same logic as 25:16-

Raymond Van Leeuwen (1986, 106–7) suggests that G (1) confused the common קבוּדִים with the rare קבוּדִים, (2) vocalized the second word as קבוּדִים (supposedly meaning "difficulties"), and (3) added λόγους ("words") for

מְכֵבֶּד<sup>ז</sup>: 28 עִיר פְּרוּצָה אֵיז חוֹמֶה אִׁישׁ אֲשָׁר אֵיז מַעְצָר לְרוּחִז: 26:1 פַשָּׁלֶג | בַּלַּיִז וְכַמְטָר בַּקָּצִיר בֵּז לֹא־נָאוֶה לִכְסֵיל כָּבְוֹד: 2 פַּצִפְּזר לְנוּד כַּדְרַזר לְעֵוּף בֵז קִלְלַת חִנָּם לָא תָבְא: 3 שׁׁזט לֵסּוּס מֶתֶג לַחֲמֶזר וְשֵׁבֶט לְגֵו בְּסִילִים: 4 אַל־תַּעַן בְּסִיל בְּאוּלְתֵז פֶּז־תִשְׁוֶה־ לִז גַם־אַתַּה: 5 עֵנֵה כָסִיל בָּאוּלִתִז פָּזִ־יִהָיֵה חַכֵם בְּעֵינֵיו: 6 מִקַצֵּה

explication. More likely, G's דוµav resulted from the extremely common  $\pi/\pi$  confusion, not a misunderstanding of a common word. Moreover, בבודים nowhere has the sense assigned to it.

V: (a) sicut qui mel multum comedit non est ei bonum (b) sic qui scrutator est maiestatis opprimitur gloria. "(a) Just as it is not good for one to eat much honey, (b) so shall the examiner of majesty be crushed by glory." qui scrutator est = כבדם וחקר is understood as "majesty," while is associated with בָּבַר ("heavy"). The result recalls the sentiments of Sir 3:21-23: do not probe matters too exalted for you.

S: אבאהבל גרבא שבע, לא שבעו, אם לא למרבא הילא השרעא. (a) It is not good to eat much honey, (b) nor to examine glorious [or 'complimentary'] words." The translator understood the negative in the first line as gapped in the second and so made it explicit. The independence of S from G with regard to ארקר symports S's "words" and "glorious" as evidence for דברים and בבד

#### 25:28 מעצר M S ( $\kappa$ יבעצה [ ( $\mu$ יאי $\kappa$ ) מעצר M S ( $\kappa \rightarrow \pi$

#### 26:1

G: (a) ὥσπερ δρόσος ἐν ἀμήτῷ καὶ ὥσπερ ὑετὸς ἐν θέρει, (b) οὕτως οὐκ ἔστιν ἄφρονι τιμή. "(a) Like dew in harvest and like rain in summer, (b) so is there no honor for a fool." G substitutes "dew" for "snow." G's proverb compares a fool's honor to things that do not exist, whereas M compares it to things that are inappropriate and useless (thus too S). M could not

use dew in this way, because dew does appear in summer in Palestine. G adjusts meteorological phenomena to the Egyptian climate; cf. 25:14, 23.

26:2 לא M<sup>K</sup> G (סטֿסבעוֹ) ≈ S (בחבא) T (לא) ] לא M<sup>Q</sup> ≈ V (in quempiam) (aur) G סטֿג ... סטֿסבעוֹ is a double translation of the ketiv. V in quempiam ("on someone") = M<sup>Q</sup>. S's בחבא ("go astray") ≈ לא תבא = M<sup>K</sup>. The ketiv is correct, since the analogies (a wandering bird, a flying sparrow) exemplify *not* coming (to a particular goal) rather than coming to someone. Also, thas no relevant antecedent. However, a scribe may have thought that the noun "curse" implies an actant (a curser), and it is to him that the curse would return (as in Ps 109:17).

26:3 לגו M  $\theta'(\sigma \omega \mu \alpha \tau \iota) \approx S(\tau \tau) V(dorso) T (לגושמיהון) א לגו G <math>\sigma' \varepsilon'$  (έθνει) (near dittog  $\iota \to \iota$ )

G: (a) ὥσπερ μάστιξ ἵππφ καὶ κέντρον ὄνφ, (b) οὕτως ῥάβδος ἔθνει παρανόμφ. "(a) A whip for a horse, a goad for an ass: (b) so too a rod for a lawless nation." By using "goad" for xπα ("bridle"), G tightens the analogy among the three figures: horse, ass, fool (BAP). ἔθνει = ζκίν for M's τζ ίν ("for the back"); this is supported by σ' ε' ἔθνει. G consequently substitutes "lawless" for "fools," because the former is a key feature of wicked nations. Compare the treatment of the near-doublet in 19:29b.

V dorso ("for the back") agrees with M, as do T לגושמיהון and  $\theta'$   $\sigma \dot{\omega} \mu \alpha \tau \iota$ , which understand  $\iota$ , via metonymy, as "body," cf. S.

S's אבויה ("for a man") = לגו ("for the back"), *pars pro toto*. In 19:29, S has לגוי ("for a people") ברא ("for a people").

26:5

G: (a) ἀλλὰ ἀποκρίνου ἀφρονι κατὰ τὴν ἀφροσύνην αὐτοῦ, (b) ἵνα μὴ φαίνηται σοφὸς παρ' ἑαυτῷ. "(a) Rather, answer the fool against his folly (b) so that he not appear wise to himself." The distinction between πρός in 26:4 and κατά in 26:5 is an attempt to resolve the contradiction: (4) Do not answer a fool πρός (in a way that corresponds to) his folly; but rather (5) respond κατά (against) his folly, that is, in a way that rebukes his nonsense. It is hard to know just which of the many functions of πρός/κατά + accusative express the distinction intended here, but κατά commonly connotes hostility or contrarity.

S: אלא מעני ברא איי עבוצא איי געאר בראי גערי בער איי איי איי איי אייי אייי אייי אייייי ס. "(a) But speak with a fool according to your wisdom, (b) lest he think in his soul that he is wise." S boldly reverses the meaning of M in order to avoid the contradiction with 26:4 (thus also T). ָרַגְלַיִם חָמֲס שֹׁתֶה שֹׁלֵח דְּבָרִים בְּיַד־כְּסִיל: 7 דַּלְיָוּ שִׁקַיִם מִפְּסֵח וּמָשָׁל בְּפִי כְסִילִים: 8 בִּצְרִוֹר אֶבֶן בְּמַרְגַּמֶה בֵּן־נוֹתֵן לִכְסֵיל בָּבְוֹד: 9 חוֹח עָלֵה בְיַד־שִׁבְוֹר וּמְשָׁל בְּפֵי כְסִילִים: 10 רֵב <sup>ר</sup>ְמְחוֹלָל בָּל־

#### 26:6

G: (a) ἐκ τῶν ἑαυτοῦ ποδῶν [G<sup>106</sup>, σ' ε' SyrH; var: ὁδῶν G<sup>BSA</sup>] ὄνειδος πίεται [var: ποιεῖται G<sup>A</sup>] (b) ὁ ἀποστείλας δι' ἀγγέλου ἄφρονος λόγον. "(a) He drinks [var: 'makes for himself'] shame from his own feet [var: 'ways']—(b) (namely,) he who sends a word by a foolish messenger." This text is given in Rahlfs; it conveys an image suggestive of drinking one's own urine. However, ποδῶν is probably Hexaplaric and πίεται looks like an adjustment to M. We should probably read with the variants and translate, "(a) He makes (for himself) shame from his own ways—(b) (namely,) he who sends a word by a foolish messenger." ὄνειδος πίεται for ασμεία is hard to explain. G may be associating the word with μ ("be disgusted"), though this correspondence is not found elsewhere.

S: גם אלא כאר אונא א באר איג מכלא. "(a) From under his feet he drinks iniquity—(b) he who sends a word by means of a fool." This image suggests drinking water from a mud puddle.

#### 26:7

G: (a) ἀφελοῦ πορείαν σκελῶν (b) καὶ παρανομίαν [G<sup>BSA</sup>; var: παροιμίαν G<sup>Rahlfs</sup> SyrH] ἐκ στόματος ἀφρόνων. "(a) Remove walking from legs (b) and transgression [var: 'a parable'] from the mouth of fools." ἀφελοῦ construes rơở as a D-stem imperative from T "draw up" (BHQ). G attempts to explain the metaphor of legs hanging down from a cripple. With παρανομίαν (= OG), the proverb means that crime is as natural to the mouth of fools as walking is to the legs (BAP). παροιμίαν is Hexaplaric. The translator probably used παρανομίαν out of hesitation to credit the fool with the ability to use a proverb at all.

26:8

G's ὅς ἀποδεσμεύει ("who binds") = <br/>קצֹרֵר .

V: (a) *sicut qui mittit lapidem in acervum Mercurii*, (b) *ita qui tribuit insipienti honorem*. "(a) Like him who throws a stone into the heap of Mercury, (b) so is he who gives honor to a fool." This corresponds to Yalqut Shimoni ad loc.; cf. b. Hullin 133a: "Hama bar Hanina said that one who benefits a person whom he does not know is like one who throws a stone to Mercury [i.e., before his idol]." (C. Gordon 1930, 388). This alludes to the Greco-Roman custom of placing votive stones along-side roads in honor of Hermes/Mercury, the patron of roads and travel (Delitzsch, ad loc.).

#### 26:9

G: (a) ἄκανθαι φύονται ἐν χειρὶ τοῦ μεθύσου, (b) δουλεία δὲ ἐν χειρὶ τῶν ἀφρόνων. "(a) Thorns grow in ['by'?] the hand of the drunkard, (b) and bondage in ['by'?] the hand of the fools." The point of comparison is either that thorns and bondage are cultivated and produced by these types or that these misfortunes grow in them and afflict them. The notion of "bondage" comes from parsing Hebrew משל ("proverb" in M) as a passive verb, "being ruled." Since משל in the sense of "proverb" was obviously well-known to the translator, this must be a homiletical word-play; cf. 26:7.

#### 26:10 מְחְוֹלֵל (עניא גע א קוֹוֹל G (עניאמניזע) א מחולל M (vocal)

<->< f (gram) כָּל [ <כָּל

יבשר א G ( $\sigma \alpha \rho \xi$ ) S (ג השוית [ רבשוית M (metath שב השבא, graph ב השוית, t conj)

עובר ים א (div, orth) עלבר ים א ער א א (div, orth)

M (רָב מְחוֹלָל־פֹל וְשֹׁבֵר פְסִיל וְשֹׁבֵר עָבְרִים) is incomprehensible. Literally it means something like "A great man causes [or 'wounds'] everything, and he who hires a fool and hires passersby." A more meaningful text can be reconstructed from components of S and G. I propose reading ירב מְחוֹלָל בְּל־בְּשֵׁר בְּסִיל וְשָׁבּר עוֹבֵר יִם. The emended text presumes only minor variants, but its conjectural nature must be acknowledged. It is to be translated, "All the flesh of the fool is greatly afflicted, and the ַבְּשַׂר' בְּאָׁי רְּשָׁבּׁר עוֹבֵר יֶם': 11 בְּכֶלֶב שָׁב עַל־קֵאָו בְּאָיל שׁוֹנֶה בְאַוַּלְתְּוֹ: 12 רָאִיתָ אֵישׁ חָבָם בְּעֵינֵיו תִּקְוָה לִכְסֵיל מִמֶּנּוּ: 13 אָמַר

drunkard crosses the sea." This describes the condition of the drunkard—who is certainly a fool—during his drinking binge and the next morning. Prov 23:29–35 depicts this condition vividly. Note in particular 23:34–35: "And you will be like one who lies (in bed) in the heart of the sea, or like one who lies in the rigging. 'They beat me, (you will say,) but I felt no hurt, battered me but I was unaware. When will I wake up? I'll go and look for more."

G: (a) πολλὰ χειμάζεται πᾶσα σάρξ ἀφρόνων. (b) συντρίβεται γὰρ ἡ έχστασις αὐτῶν. "(a) All the flesh of fools is greatly battered, (b) because became שבר by metathesis and graphic error. The extra wawconjunctive in M is probably an erroneous scribal addition consequent on this mistake. This led the scribe (and the Masoretes) to coordinate with the following ושכר. Omissions and additions of this conjunction (and the corresponding Greek and Syriac conjunctions) are very frequent in both translation and single-language transmission, often with no identifiable motives other than scribal preference; see Goshen-Gottstein 1960, 59. אָפוּאָלנדמו = (or מחולל, cf. GKC \$90), parsed as passive. (χειμάζειν is a G-hapax, and we do not know what it would ordinarily translate.) G perhaps understands מחולל to mean "driven by a storm," hence "be afflicted" (NETS: "exposed to winter cold"; BAP: "souffre grands dommages"), but מחולל can itself mean, approximately, "afflicted," as in M-Isa 53:5. סטעדףואבר for ושבר 1. The meaning of ἔχστασις in this context is unclear. Elsewhere it means "astonishment," "terror," or "trance"; NETS: "trance"; BAP: "esprit égaré"; SD: "Aussersichsein" or "Verrückstein." Perhaps G associated עברים with עברה ("wrath").

Hexaplaric renderings of אשר 1 – גושכר 3' גע ל φράσσων ἐμφράσσει; θ' καὶ φιμῶν – φιμοῖ. These are based on association with סבר "block up." σ' όργας, θ' χόλους, by association with ײַבְרָה ("wrath").

V: (a) *iudicium determinat causas* (b) *et qui inponit stulto silentium iras mitigat.* "(a) Judgment decides cases, (b) and he who puts a fool to silence soothes anger." This looks like a guess with help from  $\sigma'$  and  $\theta'$ . V derives שכר from דיב and parses שכר both times as  $\sigma = \sigma'$ , cf. Rashi; C. Gordon 1930, 404.

S: אסבה בסויס המבלא. הוס א בבו הבא (a) The flesh of the fool suffers greatly, (b) and the drunkard crosses the sea." (T<sup>L</sup> רויא "רויא") הוס הים.) "Suffers" interprets מחולל in a way similar to G, but without reflecting  $\chi$ ειμάζεται. (S-Isa 53:5, however, translates the verb as  $\chi$ בר ים אבר ים, and "crosses the sea" =  $\chi$ , a different word division.

#### **26:11** fin ] + 2 stichoi G (11*a*) (elab)

G: (a) ὥσπερ κύων ὅταν ἐπέλθῃ ἐπὶ τὸν ἑαυτοῦ ἔμετον καὶ μισητὸς γένηται, (b) οὕτως ἄφρων τῇ ἑαυτοῦ κακίᾳ ἀναστρέψας (c) ἐπὶ τὴν ἑαυτοῦ ἁμαρτίαν. "(a) As a dog, when it comes upon its own vomit and becomes disgusting, (b) thus the fool, when, by his own evil, he returns (c) to his own sin." G explicates the comparison by adding "and becomes disgusting" in 26:11a and giving a second translation of באולתו in 26:11c. "His own sin" emphasizes the moral corruption of the fool's behavior.

S translates אונה as אונה ("who is stupid"), associating the Hebrew word with Syriac אונה ("madman"); see the comment on 24:21.

#### 26:11*a*

G: (a) ἔστιν αἰσχύνη ἐπάγουσα ἁμαρτίαν, (b) καὶ ἔστιν αἰσχύνη δόξα καὶ χάρις. "(a) There is a shame that brings on sin, and (b) there is a shame that is honor and grace." This is a later insertion in G quoting Sir 4:21, which distinguishes worthy and unworthy shame; cf. Sir 41:14–22. This insertion comments on the preceding verse by teaching that one should be ashamed of his own folly so as to avoid repeating it, and on the following verse by implying that a proper "shame" (that is to say, humility, the opposite of being wise in one's own eyes) can bring honor.

26:12 ראית א' M S (איתי אי → אית אי G (בּוֹסּט) (distant dittog אית אי → אית אי ?) G has בוֹסֿט ("I saw") = ראיתי where M has ראיתי. There is no reason for the translator to have avoided the second person, and in fact the first person does not fit well. (G maintains the second person in the identically structured 29:20, though he departs from it in 22:29.) Apparently the additional *yod* arose by distant dittography with שיאיתי appears, independently, in a single MS KR.

26:13 שחל M S ((אובה [ (הובה G (ἀποστελλόμενος) S (אובה ) (G: metath הביאה; S: dbl)

אָצָל שַׁחַל בַּדֶּרֶדְ אֲּרִי בֵּיז הָרְחֹבְוּת: 14 הֲדֶּלֶת תִּפּוֹב עַל־צִירֶה וְׁעָצֵׁל עַל־מִשְׁתִוּ: 15 סְׁמֵן עָצֵל יֲדוֹ בַּצַּלֶחַת וְלְאָה לַהֲשִׁיבָה אֶל־פִּיו: 16 חְכָם עָצֵל בְּעֵינְיו מִׁשִׁבְעָה מְשֵׁיבֵי טֵעַם: 17 מַחַזִיק בְּאָזְנֵי־כֵּלֶב עֹבֵר מִׁתְעַבֵּר עַל־רִיב לְאִ־לְוֹ: 18 בְּמִתְלַהְלַהַ הַיֹּרֶה זִלָּים חִצִים נְמֶוֶת: 19 בֵּן־אִישׁ רִמֵּה אֶת־רֵעֵהוּ וְׁאָמַר הְלָא־מְשַׂחֵק אֶנִי 20 בְּאֶפֶס עֵצִים תִּכְבֶּה־אֵשׁ וּבְאֵיז וֹרְיָגָן יִשְׁתֹּן מִדְזוֹן: 21 כֶּחָם לֵגֶחֶלִים וְעַצִים לְאֵשׁ וְאִישׁ וּמְדוֹנִים ְלַחַרְחַר־רִיב:

streets!<sup>29</sup> ἀποστελλόμενος = <u>μν</u>. Most G manuscripts add 26:13c: ἐν δὲ ταῖς πλατείαις φονευταί ("a murderer in the plazas"). This is taken from G 22:13b in order to explain "lion" metaphorically. Though G 26:13c is secondary, it was known to S.

#### 26:15

The versions did not understand צלחת as "plate" but translated it here and at 19:24 as "bosom" (G בער דָשָׁ אטָא אָטָ מטֿדטָט, S בערם) or "armpit" ( $\alpha', \sigma', \theta' \doteq \nu \pi \mu \alpha \lambda \eta$ ; V sub ascellas suas; T בשחאתיה). In both cases the resulting image signifies inactivity; see G 24:33b; cf. 6:10b.

#### 26:16

G: (a) = M; (b) דסט פֿא האחסעסעה מהסאסעולסידסג מאר (b) Than the one who brings back a report in satisfaction." פֿא האחסעסעה (b) Than the satisfaction (b) משָׁבְעָה = אָשָבְעָה (it. "from satisfaction" = satisfactorily? when he is satisfied [ABP]?). שעם is given the sense of Aramaic שָׁעַם ("report") (Baumgartner).

#### 26:17

G speaks of grabbing the *tail* of a dog, not his ears. BHQ says that the choice was inspired by the resulting alliteration, δ κρατῶν κέρκου κυνός. Perhaps a more salient reason was the fact that the tail is easier to grasp and the image therefore more "logical." δ προεστώς could correspond to either מתעבר (M) or אַמתערב (M) or 14:16.

#### 26:18

G: (a) ὥσπερ οἱ ἰώμενοι προβάλλουσιν λόγους εἰς ἀνθρώπους, (b) ὁ δὲ ἀπαντήσας τῷ λόγῳ πρῶτος ὑποσκελισθήσεται. "(a) As the healers hurl words at men, (b) and he who encounters the word first will stumble" (continues in 26:19). G, which is itself difficult, is unrelated to the Hebrew (except perhaps for προβάλλουσιν = הירה, though this correspondence would be unique). This is a cynical remark about physicians. BAP: "Comme les guérisseurs donnent aux gens de belles paroles—et celui qui s'y prête va rechuter le premier." NETS: "As those who are being treated propound tales to people, and he who encounters the tale will be the first to be tripped up." G 26:18 does not provide a meaningful analogy to the next verse, in spite of ὥσπερ - οὕτως. The translator has simply substituted a verse of his own.

#### 26:19

G (continuing 26:18): (a) οὕτως πάντες οἱ ἐνεδρεύοντες τοὺς ἑαυτῶν φίλους, (b) ὅταν δὲ φωραθῶσιν [var: ὁραθῶσιν G<sup>BSMSS</sup>], λέγουσιν ὅτι Παίζων ἔπραξα. "(a) thus are all who lie in wait for their friends—(b) when they are discovered [var: 'are seen'] they say, 'I acted in jest.'" It is unclear why G uses "ambush" for רמה. Perhaps he could not see how one could even pretend that deceit was in jest, whereas lying in ambush might be playful. G adds "and when they are discovered" to make a logical connection between the misdeed (a) and the excuse (b).

#### 26:20

G: (a) ἐν πολλοῖς ξύλοις θάλλει πῦρ, (b) ὅπου δὲ οὐκ ἔστιν δίθυμος [G<sup>B</sup>; var: ὀξύθυμος G<sup>ScA</sup>] ἡσυχάζει μάχη. "(a) By much wood [pl], fire flourishes, (b) but where there is no dissenter (?) [var: 'quick tempered'], conflict quiets down." G 26:20a is a converse translation that creates an antithesis within the verse but disturbs the analogy.

**26:21** מָדונים M<sup>K</sup> ] מְדְיָנִים M<sup>Q</sup> (dial? norm)

On the *ketiv-qere*, see the comment on 6:14.

22 דִּבְרֵי גְרְגָּן כְּמְתְלַהָמֵים וְׁהֵׁם יְרְדָוּ חַדְרֵי־בֶּטֶן: 23 כֵּסֶף סֵיגִים מְצָפֶּה עַל־חֲרֶשׁ שְׁפָתַיִם יּחַלָקֵים וְעָׁב־רֶע: 24 בֻּשְׂפָתָו יִנְּכֵר שוֹנֵא וּבְקִרְבּוֹ יָשֵׁית מִרְמֵה: 25 כִּי־יְחַנֵּן קוֹלוֹ אַל־תַּאֲמֶן־בְּוֹ כִּי שֶׁבַע תּוֹעֵבְוֹת בְּלִבְּוֹ: 26 יֹמְכַסֶּה' שִׁנְאָה בְּמַשָּׁאֵוֹן תִּגָּלֶה רָעָתוֹ בְקָהֵל:

G, S, and T thought that מרטקל ("coals") means "hearth" or "gridiron": έσχάρα, לה לה לה, ארטקל, ארטקל, לה לה The Hebrew word is translated correctly in Isa 44:12 and 54:16. S's and T's choice is strange, because the exact cognate exists in Syriac and RH. This choice is not due to lexical ignorance (BHQ) but rather to S's reliance on G and T's on S. V, correctly: *carbones*.

#### 26:22

G: (a) λόγοι κερκώπων μαλακοί, (b) οὖτοι δὲ τύπτουσιν εἰς ταμίεια σπλάγχνων. "(a) The words of knaves are soft, (b) but they strike the chambers of the innards." The *hapax* κέρκωψ means "monkey-man," "ape" (LSJ, NETS), hence "knaves" (LSJ) or "tellers of false tales" (GELS), "malins singes" (BAP), or the like. Since he translated μετις differently and fairly literally in 26:20, the present rendering seems to be placing a playful twist on the word. μαλακοί is a contextual guess at the obscure canter. M 26:22 = 18:8, but G is entirely different from M in the latter.

S's במתלהמים ("provocative") is a guess for במתלהמים.

26:23 (graph דלְקֵים ( βεία) א דלְקֵים ( קים ( βεία) ( εία) ( εία

G: (a) ἀργύριον διδόμενον μετὰ δόλου ὥσπερ ὄστρακον ἡγητέον. (b) χείλη λεῖα [var: δόλια  $G^S$ ] καρδίαν καλύπτει λυπηράν. "(a) Silver given with deceit should be considered as a potsherd. (b) Smooth [var: 'deceitful'] lips hide a grieving heart."

G 26:23a explicates the metaphor of drossiness as deceit. In 26:23b,  $\lambda \epsilon \tilde{i} \alpha$ = הלקים ("smooth") (Jäger); see G-Gen 27:11. (Possibly  $\lambda \epsilon \tilde{i} \alpha$  is used of smooth speech in the present verse to allude to Jacob's smoothness. That the translator allows himself the "freedom" of introducing an intertextual factor in the rendering does not argue against the existence of a textual difference.) הלקים alone fits the imagery: just as the sheen of drossy silver disguises worthless clay, so does the gloss of unctuous, deceitful speech mask inner hostility. "Smooth lips" also is appropriate to the theme of the proverb cluster in 26:23–28, and note how הלקים describes an enemy's flattery in Ps 55:22a. (The fire in Prov 16:27 [which Van Leeuwen (1986, 111 n. 4) adduces in support of דלקים here] represents malice and slander, whereas the present verse speaks of a quality that is superficially attractive; see further ABP.) The emendation presupposes only the loss of the left vertical of the  $\Pi$ . The masculine adjective is incongruous with the feminine שׁפּתִים in both the emendation and M.

V *tumentia* ("swelling") is a guess at the unusual דלקים. Swelling may connote anger.

26:24 בשפתו M<sup>K</sup> ] בָּשְׂפָתָיו M<sup>Q</sup> G (χείλεσιν) ≈ S (הב) V (*labiis suis*) ≈ T (מז] שפוותיה) (num or orth)

G: (a) χείλεσιν πάντα ἐπινεύει ἀποκλαιόμενος ἐχθρός, (b) ἐν δὲ τῆ καρδία τεκταίνεται δόλους. "(a) A weeping enemy promises all things by (his) lips, (b) but in the heart he devises deceits." This continues the theme of "lips" from 26:23b. While M vocalizes "ແລר a passive (N-stem), G parses it as active (D-stem, as in Job 21:29 and 34:19) and translates it ἐπινεύει ("to give recognition by nodding," hence "to promise"). (The verb is not used elsewhere in translating a Hebrew text.) "All things" and "weeping" are added to enhance the dramatic quality, a feature that continues in 26:25a ("in a great voice").

S translates M's unusual בכיא (lit. "sets deceits") as בכיא ("an ambush is concealed"), producing a more concrete image. S הא השפתיו (= T) could reflect הא משפתיו was likely chosen as indicating the source of the knowledge.

The *ketiv* can intend a singular or plural; the *qere* indicates the latter.

#### 26:25

G and S treat אחנן קולי (lit. "beseeches [by] his voice") differently. In G the enemy speaks loudly ( $\mu\epsilon\gamma\dot{\alpha}\lambda\eta$  דָןָ שָׁעָאָ ("by a great voice")), in S, quietly ( $\mu\epsilon\gamma\dot{\alpha}\lambda\eta$  דָןָ שָׁעָאָ ("by a great voice")). V submiserit vocem suam  $\approx$  S. Apparently "his voice" by itself seemed superfluous, so the versions elaborated in different ways.

26:26 אמכסה G (καλύπτων) σ' (καλύπτων) S (גרבמסה V (qui operit) T (גרבמסה) V (qui operit) T (דמכסה (דמכסה)

G: (a) ὁ κρύπτων ἔχθραν συνίστησιν δόλον, (b) ἐχκαλύπτει δὲ τὰς ἑαυτοῦ ἑμαρτίας εὔγνωστος ἐν συνεδρίοις. "(a) He who hides hostility contrives deceit, (b) but, being well-known, he exposes his sins in assemblies." G draws together themes from the preceding three verses: enmity (26:24a), deceit (26:23a, 24b), and uncovering/covering (26:23b) (BAP). G adds

27 כְּרָה־שִׁחַת בְּה יִפְּל וְגָלֵל אֶׁבֶן אֵלְיו תָּשִׁוּב: 28 לְשׁוֹן־שֻׁקָר יִשְׂגָא דַכְּיו וּפֶה חָלָק יַשְשָׁה מִדְחֶה: 27:1 אַל־תְּהַלֵּל בְּיוֹם מְחֵר כִּי לֹא תַּדַע מַה־יֵלֶד יוֹם: 2 יְהַלֶּלְדְ זָר וְלֹא־פִידְ גָׁכְרִי וְאַל־שְׂפָתֶידָ: 3 כְּבָד אֶבָּן וְגַטֶל הַחֵוֹל וְכַעַס אָוֹיל כְּבֵד מִשְׁנֵיהֶם: 4 אַכְזְרִיּוּת חֵמָה וְשֵׁטֶף אֶרָן וּמֵי יַעֲמֹד לִפְנֵי קַנְאֵה: 5 טוֹבָה תּוֹכַחַת מְגַלֶה מֵאַהַבֶה מְסָתֵּרת: 6 נֵאֵמְנִים פּּצְעֵי אוֹהֵב וְנַטָתָרוֹת נִשִׁיקּוֹת שוֹנֵא: 7 נֵפָשׁ שְׁבַעָה תָּבוֹס

εὔγνωστος to explain why the secret will be exposed: the hypocrite in question is a public figure maneuvering in the assembly.

תכסה: All the versions treat תכסה as a D-stem participle, but they may have been constrained to do so by context, especially if they did not recognize the Dt form in תכסה. But even apart from this uncertain evidence, we should emend to מכסה (ABP, following BHS and many), because otherwise העתו וה רעתו in the b-stich is left without an antecedent. Moreover, the participle is supported by the similarly structured descriptions of the worthless man in Prov 6:12–15 and 16:27–30, which are built on series of participles.

#### 26:27

In 26:27a, G adds τῷ πλησίον ("for [his] neighbor") to make it clear that the saying does not forbid digging legitimate holes. This is another case of G's tendency to explain the obvious.

#### 26:28

G: (a) γλῶσσα ψευδὴς μισεῖ ἀλήθειαν, (b) στόμα δὲ ἀστεγον ποιεῖ ἀκαταστασίας. "(a) A deceitful tongue hates truth, (b) and an unguarded mouth causes disorders." For M's difficult τ⊂τ ("those it afflicts" [ABP]?, "those crushed by it" [JPSV]?), G has ἀλήθειαν, by association with Aramaic ("pure") (Baumgartner) = S (¬ωνω), V (veritatem).

BHQ says that T's אורחתא דקושטא reads the Hebrew as דרכי and also follows S. But דרכי is dissimilar to דכי, and "ways" could have been added for concreteness.

#### 27:2

G translates Translates Translates  $\pi \epsilon \lambda \alpha \varsigma$  ("[your] neighbor"), used only here in the Sep-

tuagint. This shows a correct understanding of the Hebrew word, which in this context means another person, someone not properly belonging to the relevant group; see ABP 1.139. In this verse, ד is equivalent to "רָּעָ" ("fellow").

#### 27:4

G: (a) ἀνελεήμων θυμὸς καὶ ὀξεῖα ὀργή, (b) ἀλλ' οὐδένα ὑφίσταται ζῆλος. "(a) Anger is merciless and wrath is swift, (b) but jealousy tolerates no one." ὀξεῖα (which can mean "swift" as well as "sharp"; e.g., Prov 22:29 and Amos 2:15) is suggestive of a rush of anger, which, for all its force, is more tolerant than jealousy. The rhetorical question in 27:4b is recast as a negative indicative. (For the phenomenon, see Orlinksy 1958, 244–46.) The Hebrew would call for "but who can stand before jealousy?" Seeing anger and wrath as the active forces in the first stich, G personifies jealousy as the actor in the second.

#### 27:6

G: (a) ἀξιοπιστότερά ἐστιν τραύματα φίλου (b) ἢ ἑχούσια φιλήματα ἐχθροῦ. "(a) More trustworthy are wounds from a friend (b) than willing kisses of an enemy." G introduces a moral lesson by formulating the verse as a comparison and translating moral lesson by formulating the verse as a comparison and translating as ἑχούσια. This is a unique etymologizing translation. It derives the notion of willingness from , twice basic meaning is *entreaty*. (Compare a' ἰχαπιά ["fit for supplicants"]; see LSJ.) G puts the proverb into comparative parallelism in imitation of 27:3–5 (Cook 1999a, 143). In fact, μακικά is an Aramaism meaning "profuse"; see ABP.

σ' θ' τεταραγμένα ("troubled") is unlikely to reflect the dissimilar נערצות (BHQ, comparing G Isa 8:12 and Saʿadia). (Saʿadia actually explains the word to mean "overdone.") The σ' θ' rendering is a guess from context, like the broad translations of T and V; see below.

S: אפאד הכאגרבא אין געשא אין געראר (a) Better are the blows of a friend (b) than the kisses of an enemy." S elides the difficult געתרות. The translator may have considered the omission unproblematic because the kisses of an enemy are *without* qualification worse than the rebukes of a friend. T uses a slot variant, בישן, which is a neat antithesis to שפירן in 27:6a.

#### 27:7

G explains תבוס ("trample") as ἐμπαίζει ("mock") (only here), apparently by homoiophonic association with בו"ו ("mock"). גְּפֶת וְגָפֶשׁ רְׁעֵבָּה כָּל־מִר מָתוֹק: 8 בֻּצִפּוֹר נוֹדָדֶת מִז־קִגָּה בֵּז־אִּישׁ נוֹדֵד מִמְקוֹמְוֹ: 9 שֶׁמֶן וּקְטֹרֶת יְשַׂמֵּח־לֵב וּמֶתֶק <sup>ו</sup>רֵׁשָׁה<sup>י</sup> מֵעְצַת־נֵפָשׁ: 10 רֵעֲדֶ וְוַבֶּשֶה<sup>י</sup> אָבִידְ אַל־תַּעֲזֹב וּבִית אָחִידְ אַל־תְּבוֹא בְּיוֹם אֵידֶדְ סוֹב שְׁכֵן קְׁרוֹב מֵאָח רָחְוֹק: 11 חַכַם בְּנִי וְשַׂמַּח לְבֵּי וְאָשֶׁיבָה חֹרְפִי דְבָר: 12 עְרָוּם רָאֲה רָעֵה נִסְתֵּר פְּתָאיִם עָבְרוּ נֶעֵנֶשׁוּ: 13 קַח־בָּבוֹא כִּי־עָרַב זֶר וּבְעַד וּבְעַד וּבְכָרִים חַבְלֵהוּ: 14

#### 27:8

G: (a) ພຶσπερ ὅταν ὄρνεον καταπετασθῆ ἐκ τῆς ἰδίας νοσσιᾶς, (b) οὕτως ἀνθρωπος δουλοῦται, ὅταν ἀποξενωθῆ ἐκ τῶν ἰδίων τόπων. "(a) As when a bird flies down from its own nest, (b) thus a man is enslaved when he migrates from his own places." Whereas M suggests the loneliness of a wanderer, G makes the analogy signify enslavement and migration. We may hear in this the Alexandrian Jew's feelings of alienation from his surroundings.

#### 27:9 ↔ ביאהו [רעה M ≈ S (סיגבות) (orth)

אַמעצת M הַמָעָאָבָת הַמַעָאָבָת (אמדמף<br/>מאָזיטעדמו או הַמָעָאָבָת הַמַעָאָבָת (div, near dittog בת $\to \mu$  (בת<br/> Mן מעצת (גרעית) א הבעצת (גרעית) ד (בתרעיתא) (graph ברעיתא)

As emended, the verse means, "Oil and incense please the heart, and (likewise) the sweet advice of a friend (pleases the heart) more than the soul's counsel." This requires the deletion of the 1 from רעהו as a false orthographic addition by a scribe who thought that the vocalization intended was  $\alpha \pi \eta$ . (lit. "sweetness") implies advice; "the soul's counsel" is the advice that comes from within; see ABP.

G: (a) שַטָּסוּטָ אמו פֿיעטָ אמו פֿיעוּמָשמיע דַרָּאָדָדָמו אַמאָסָאָ, (b) אַמדמאָסָאָיעטדמו לו טָּדָל סַטָּשָדָדָשָׁמָלישע שָׁטָאָל. "(a) By myrrh and wines and incense the heart is pleased, (b) but by misfortunes the soul is rent." G adds "wines" to fill out the picture of merrymaking. G 27:9b reflects וּמַעַצֶּבֶת (Jäger). The underlying verse should be translated, "Oil and incense and myrrh please the heart, but the soul is torn by grief." It must be said that these Greek-Hebrew correspondences are unique, but G can hardly be explained otherwise. G's saying reflects the sentiment (but not the wording) of Prov 14:13: "Even in merriment a heart may hurt, and the outcome of pleasure is sadness."

In the apparatus I explain G 27:9b as deriving from a text similar to M. M's variant fits the context, which speaks of friendship and its preciousness (27:10, 11). Still, it is possible that the development was from G's source text to M, with the context promoting the change. In any case, even though G (or M) arose by graphic error, interpretive possibilities and mechanical scribal processes reinforced each other to yield valid variant proverbs.

 $S \approx M$ . בעצת – T בתרעיתא, מ'  $\beta$ ουλ $\tilde{\eta}$  = בעצת.

#### 27:10 ורעה M<sup>K</sup> ] וְרֵעָ M<sup>Q</sup> (syn, norm)

The *qere* is a normalization of the rarer synonym רְשָה. The latter occurs also in 2 Sam 15:37; 16:16; 1 Kgs 4:5; as well as Prov 27:9 as reconstructed.

27:11

G: (a) = M; (b) גמו מֹחסֹסדְסָשְׁטּ מֹחסׁ סּסָט פֿהטאנולוסדסטָג אָלאָסטָג. "(b) and remove from yourself very disgraceful words." In M, the father tells his son to become wise, "so that I may reply to him who insults me." G likely found this puzzling, for, after all, who would insult the wise father, and why would he need to reply? In G's rendering it is the callow son who would be subject to insults, and the father logically warns him to remove insults from *himself*. Given this reasoning, G cannot be used to support insults from *himself*. Given this reasoning, G cannot be used to support והשיבה (cf. BHQ). פֿהטאנולוסדסטג אלאסטג construes M's consonants, דבר הקרץ, as "disgraces of words," hence "disgraceful words," though "תַרָּף" "disgrace" is not attested.

S: (a) = M; (b) הרשני האר גרביניסגנ, (b) and make there cease from me the insults of my insulters." S too uses the imperative but not G's  $\dot{\alpha}\pi\dot{\sigma}$  σοῦ. S addresses the same problem as G but solves it differently. In S, instead of the father himself having to answer his abusers, he commands his son to comport himself in such a way that others will not reproach the father for the son's behavior.

#### 27:13 ערב M S (בר $\leftarrow$ רב M S ( $\rightarrow$ יעבר $\subseteq$ ( $\pi \alpha \rho \eta \lambda \theta \varepsilon \nu$ ) (metath בר $\rightarrow$ כב

א א M S (לגהבי א א S (לגהבי (טאר א א א ד )) א דר M S (לגהבי א )

 $( c : (a) \dot{a} \dot{a} \dot{\lambda} \dot{\delta} \dot{\tau} \rho : a) V (alienis) ] גָרָרָיָה M S ( הסביבה) ] (graph קר ה) G: (a) \dot{a} \dot{\phi} c > i \dot{\mu} \dot{a} \dot{\tau} i o v a \dot{v} \tau o \ddot{v}, \pi a \rho \ddot{\eta} \dot{\lambda} \theta c v \gamma \dot{a} \rho \dot{v} \dot{\beta} \rho i \sigma \tau \dot{\eta} s, (b) o \tau i \dot{v} \dot{a} \dot{\lambda} \dot{\delta} \dot{\tau} \rho i a \lambda v \mu a \dot{v} c \tau a. (a) Take away his garment, for an arrogant man has passed by, (b) who damages what does not belong to him." It is unlikely that the translator is availing himself of "remarkable freedom" in rendering M, as Cook (1999a, 145) suggests. Why would G use this freedom to produce such an obscure proverb? In Prov 6:1; 17:18; and 22:26, the$ 

### בַּבְּקֶר הַשְׁבֵּים לְּלָלָה תֵּחָשָׁב לְוֹ: 15 דָּלָף טוֹרֵד בְּיִוֹם סַגְרֵיר וְאֵשָׁת <sup>ר</sup>מְׁדוֹנִים<sup>-</sup> נִשְׁתָּוֶה: 16 <sup>ר</sup>צֹפְנָה<sup>-</sup> צְפַן־רְוּחַ <sup>ר</sup>וּשְׁמוֹ יְמִינִי יִקָּרֵא<sup>-</sup>: 17 בַּרְזֶל בְּבַרְזֶל יֻחֵד וְאִישׁ יַחֵד פְּנֵי־רֵעֵהוּ: 18 נֹצֵר תֵּאַנָה יאכַל

translator understands the warning against going surety correctly. (In 11:15, textual differences and a lexical misunderstanding obscured this theme.) In the present verse too slight textual differences obscured the theme: G reflects ערב לער ערב ערב זר זר זר (cf. 14:10). (Though else-where טָּקָרָסוֹקָ = בְּוֹדָסוֹקָ IProv 6:17] and דר ערב ונקריס (for 14:10). (Though else-where שׁרָה נָרָרָ = בַּוֹדָסוֹקָ אָרָב בַּרָרָם בַּוֹדָסוֹקָ ווּשָׁרָ בַּרָרָ מַרָרָ מָרָב בַּרָרָם בַּרָרָ מָרָב בַּרָרָם בַּרָרָב שָרָב שָרָב שָרָב שָרָב בַרָרָ מָרָב שָרָב שָרָב שָרָב שָרָב שָרָב שָרָב שָרָב שָרָב בַרָרָם בַרָּרָב שָרָב שָרָב שָרָב שָרָב שָרָב שָרָב שָרָב מָרָב מָרָב בַרָרָם בַרָּרָם בַרָּרָב שָרָב שָרָב שָרָב שָרָב שָרָב שָרָב שָרָב בַרָרָב בַרָרָב בַרָרָם בַרָרָב בַרָרָב בַרָרָב שָרָב שָרָב בַרָרָב בַרָרָם בַרָרָב בַרָרָם בַרָּרָב בַרָרָב בַרָרָם בַרָרָב בַרָרָב בַרָרָם בַרָרָב בַרָרָם בַרָרָב בַרַרָרָם בַרָרָב בַרָרָם בַרָרָב בַרָרָב בַרָרָם בַרָרָב בַרָרָם בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרָם בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרַב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרַב בַרָרָב בַרָרָב בַרָרַב בַרָרָב בַרָרָב בַרָרָב בַרָרַב בַרָרָב בַרָרָב בַרָרַב בַרָרָב בַרָרַב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרָרָב בַרַרָים בַרַרָים בַרַרָים בַרַרָרָב בַרַרָים בַרַרָרָב בַרַרָים בַרַרָים בַרַרָים בַרַרָרָב בַרָרַב בַריים בַרַרָרָב בַרַרָים בַרַרָים בַרַרָרָב בַריב בַריב בַריב בַרים בַרַרָרָב בַרָרָב בַרַרָים בַרַרַרָים בַרַרָים בַרַרָרָב בַרָרָב בַרָרָב ברַרָים בַרַרים בַרַרָים בַרַרָרַב בַרָרָב ברַרָרָב ברַרָב ברַרָים בַרברים בַרַרים בַרַרָרָב ברַרַרָב ברַרָים בַרַרָים בַרַרָרָב ברַרָרָב ברַרָרָב ברַרָרָב ברַרים בַרַרָרָב ברַרָרָב ברַרָרָב ברַרָים בַרַרָרָב ברַרָרָב ברַרים בַרַרָים בַרַרָרָב ברַרים בַרַרָרָב ברַרָרָב ברַרים בַרַרָרָב ברַרים בַרַרָרָב ברַרָרָב ברַרָרָב ברַרים בַרַרָרָב ברַרים בַרַרָרָב ברַרָרָב ברַרָרָב ברַרָרַברים ברַרַרָרָב ב

V *alienis* (pl) = 20:16b *ketiv*. But since נכרים there is translated by a different word, *extraneis*, V is not assimilating to that verse. Rather, V = נכרים. The arguments for that reading are given in the comment on 20:16.

27:15 מדונים M<sup>K</sup> ] מַדְיָנִים M<sup>Q</sup> (dial? norm) See the comment on 6:14.

We should emend עפן to the singular, as the singular צפן requires; cf. V. Then, following G's variants, the verse reads אַפְנָה עָפַן־רוּח וּשָׁמוֹ יְמִינִי יָקָרָא . In ABP I translate, "He who hides her [sc. the contentious woman] hides the wind, and he is called, 'Right.'" "Hide" could mean attempting to conceal her from the embarrassing public view (Clifford), but the verse is a riddle, and the solution is not dependent on the translation alone.

G: (a) βορέας σκληρὸς ἀνεμος, (b) ὀνόματι δὲ ἐπιδέξιος καλεῖται. "(a) The north wind is a hard wind, (b) but it is called 'Favorable' by name." G 27:16a is a guess at an obscure text. G 27:16b can be retroverted to שִׁמִי יְקִרָא וֹשְׁמוֹ (lit. "its name is called 'right") (Jäger). (The form 'מִינִי יְקָרָא *ketiv* in 2 Chr 3:17.) The right hand is δεξιός and, in Greek, connotes good fortune. If the retroversion is correct, it shows Greek influence on the Hebrew. As the subject of "is called," G's "its name" is more plausible than M's "oil," and the noun-adjective "right" fits better than M's "his right hand."

It is strange that the wind is called "right" in G and, perhaps, its source text, since the right side is the south in the Palestinian and Greco-Egyptian orientation. According to Plutarch (*Isis* 32), "The Egyptians think the dawn to be the face of the world, northward to be the right, and southward to be the left" (noted by de Lagarde). But for the earlier Egyptians, the west was on the right.

α' and σ' bring out the concept of north here: κεκρυμμένος βορέας ανεμος ("the north wind is hidden"); but this does not represent a different consonantal Hebrew.

V has the singular *qui retinet eam* ("he who restrains her" [sc. the contentious woman in 27:16]).

S: معنیہ معنیہ معنیہ معنیہ محمدیہ (a) The northern wind is hard, (b) and it is called by the name of 'right (hand).'" S (and, indirectly, T) depends on G for the gist of the verse.

#### 27:17

This verse is difficult. M calls for the translation, "Iron in iron together, and a man together with the face of his fellow." This could mean that just as (magnetized) iron is attracted to iron, so is one man drawn to another's face for fellowship and communication. It is, however, doubtful that  $\underline{}$  "together" can govern  $\exists$  in the sense of "with."

The versions take Th' to mean "sharpen," from T''T'', H-stem). G: (a) σίδηρος σίδηρον ὀζύνει, (b) ἀνὴρ δὲ παροξύνει πρόσωπον ἑταίρου. "(a) Iron sharpens iron, (b) but a man provokes the face of (his) fellow." V and S too derive the verb from T''T but seem to construe this as a friendly action. V: ferrum ferro acuitur et homo exacuit faciem amici. "Iron sharpens iron, and a man makes the face of his friend sharp." S: minor, and a man sharpens the face of his friend"; thus also T. The verse 27:18-22

ַפְּרְיֶה וְשֹׁמֵר אֲדֹנֵיו יְכֵבֶּד: 19 כַּמַּיִם הַפָּנִים לַפָּנִים כַּן לֵב־הָאָדָׁם לָאָדֶם: 20 שְׁאוֹל וַאֲבַדּה לָא תִשְׂבֵּעְנָה וְעֵינֵי הָאָדָם לָא תִשְׂבֵּעְנָה: 21 מַצְרֵף לַכֶּסֶף וְכַוּר לַזָּהֶב וְאִישׁ לְפֵי <sup>ר</sup>ְמְהַלְלֵיו<sup>1</sup>: 22 אָם תִּכְתִוֹשׁ־

is understood in this way in b. Ta'anit 7a, which paraphrases, "Just as one piece of iron can sharpen another, so do two wise students sharpen each other in the halakah." Similarly several commentators, including Clifford and McKane. There is, however, no evidence that "sharpen the face" means to challenge and sharpen a person's wits. Moreover, iron would be sharpened by a whetstone.

In ABP I propose vocalizing יחד (pausal יחד), a G imperfect from 'חד'', comparing Gen 49:6 and Isa 14:20. This would be translated, "As iron joins to iron, (so does) a man join with the face of his fellow." The problem is that in the two verses cited the verb governs a preposition meaning "with," which is absent here. No entirely viable emendation presents itself.

במים 19:19 במים M ] כמו ה שה ה הי שה ה הי שי ג (graph  $\bullet \to \mathfrak{l}, \, \mathrm{near} \, \mathrm{haplog}$  ה ה $\to$  ה)

G: (a) ὥσπερ οὐχ ὅμοια πρόσωπα προσώποις, (b) οὕτως οὐδὲ αἰ καρδίαι [G<sup>AVMSS</sup>; vars: διάνοιαι G<sup>BS</sup>; + ὅμοιαι G<sup>SMSS</sup>] τῶν ἀνθρώπων. "(a) Just as faces are not similar to (other) faces, (b) so too are the hearts [var: 'minds'] of men not [var: + 'similar']." (The variant διάνοιαι, more distant from M, is probably original.) G ὥσπερ = ἱςαί for Cara (Jäger), the final of of cara cara by near haplography with the following π.

The Hebrew underlying G (= M except for כמו (כמו לבמו) would be translated, "Like a face to (another) face, so a man's heart to (another) man's." This could mean that just as people see each other externally, so they can look into each others' hearts—though concealment is possible (see ABP). M (with במים) is best translated, "As in water, a face (appears) to a face, so a man's heart (appears) to (another) man." This is open to various interpretations, of which I prefer the idea that when one looks at another's mind to ascertain how he feels he sees a reflection of his own; see ABP. Both forms of the proverb make sense. Instead of the mechanical explanation given above, it might be better to say that a scribe in one of the traditions was inspired by graphic similarity to shape a new proverb. Various other retroversions have been suggested to uncover the negative in G's source text (e.g., כאין דומים), but they are distant from M. V: (a) quomodo in aquis resplendent vultus prospicientium (b) sic corda hominum manifesta sunt prudentibus. "(a) Just as the faces of them that look in the water shine [i.e., reflect back] in it, (b) so are the hearts of men revealed to the wise." V's unusual interpretation of the analogy accords with Mid. Prov. ad loc. (see C. Gordon 1930, 395).

```
27:20 אָבָדוֹן M<sup>Q</sup>; אָבָדוֹן M<sup>Q</sup>; אָבָדוֹן M<sup>MSS</sup> (M<sup>Q</sup>: orth; M<sup>MSS</sup>: norm) fin ] + 2 stichoi G (20a) (elab)
```

The *ketiv* and the *qere* are variants of the more familiar וַאָּבַדּוֹן, which is found in some MSS as the *qere*. The *ketiv*, with  $\pi$  as the marker of  $\bar{o}$ , on the model of the possessive, is the earlier.

#### 27:20*a*

G: (a) βδέλυγμα κυρίω στηρίζων ὀφθαλμόν, (b) καὶ οἱ ἀπαίδευτοι ἀκρατεῖς γλώσσῃ. "(a) An abomination to the Lord is he who stares [lit. 'fixes the eye'], (b) and the uninstructed (are) intemperate of tongue." The addition is based on 16:30 and inserted here because of the mention of the eye in 27:20. The addition is not in S or V.

27:21 מַהַלְלָיו G (פֿאָגשµומגָאיזע מטֿדטֿע) S (ג<del>משׁבעערס,</del> M (vocal, num)

fin ] + 2 stichoi (G 21a) (elab)

G: (a) δοχίμιον ἀργύρῷ καὶ χρυσῷ πύρωσις, (b) ἀνὴρ δὲ δοχιμάζεται διὰ στόματος ἐγκωμιαζόντων αὐτόν. "(a) Burning is a test for silver and gold, (b) but a man is tested by the mouth of those who praise him." This is close to the consonantal M but combines כור with its synonym מצרף.

S approximates M but adds a verb, صحه ("tests"), in 27:21a.

S and G (followed by V<sup>MSS</sup> and T) have the plural "those who esteem" or "praise." Since the disparate witnesses agree, they are probably reading מהלליו (= one MS KR). With M's מָהַלָּלוֹ ("his praise") the analogy is lost, for a man is not assayed according to his praise but by the words of those who praise him; see ABP.

#### 27:21*a*

G: (a) καρδία ἀνόμου ἐκζητεῖ κακά, (b) καρδία δὲ εὐθὴς ἐκζητεῖ γνῶσιν. "(a) The heart of a wicked man seeks evils, (b) but the upright heart seeks אָת־הָאָוֹיל | בְּמַרְתֵּשׁ בְּתַוֹדְ הֲרִיפּוֹת בְּעָלֵי לֹא־תָסְוּר מֵעָלָיו אַוּלְּתוֹ: פּ 23 יְדַעַ הֵּדַע פְּגַי צֹאנֶדְ שִׁית לִבְּדָ לַעָדָרִים: 24 כְּי לָא לְעוֹלָם חֶסֶן וְאִם־נֵזֶר לְדַוֹר <sup>ר</sup>ְדוֹר<sup>י</sup>: 25 גָּלָה חֲצִיר וְנִרְאָה־ דֶשָׁא וְנֶאֶסְפוּ עִשְׂבְוֹת הָרִים: 26 בְּבָשֵׁים לִלְבוּשֶׁדְ וּמְחֵיר שְׁדָה עַתּוּדִים: 27 וְדֵי | חֲלֵב עִזִים לְלַחְמְדָ לְלָחֶם בֵּיתֶדְ וְחַיִים לְנַעֲרוֹתֶידְ:

knowledge." G represents, approximately, אַב יְשָׁרִים מְבַקֵּשׁ רָע וְלֵב יְשָׁרִים מְבַקֵּשׁ רָע וְלֵב יְשָׁרִים מָבַקֵּשׁ רָע וְלֵב יְשָׁרִים מָבַקֵּשׁ רַע וּלֵב הַשָּׁרִים מָבַקֵּשׁ רַעַת G 27:21 $ab \approx M$  18:15a. S has this addition = G; V  $\approx$  G.

נרבבו אתסיר [ M תסיר G (περιέλης) S (כבבו ארטיר (graph 1 ↔ ', equal)

G: (a) ἐἀν μαστιγοῖς ἄφρονα ἐν μέσῷ συνεδρίου ἀτιμάζων, (b) οὐ μὴ περιέλης τὴν ἀφροσύνην αὐτοῦ. "(a) If you whip a fool, disgracing (him) in the midst of the assembly, (b) you will surely not remove his folly." G explicates the metaphor of crushing with a mortar as public punishment by introducing one of its favorite themes, the assembly; see the comment on 15:22.

G 27:22b  $\approx$  M, but où µỳ περιέλης = רְּסִיך. Both proverb variants (M: "his folly will not depart from him"; G: "you will not remove his folly") make good sense. M's point is that the fool is resistant to the most severe of chastisements. G's proverb makes the more interesting observation that publicly disgracing a fool is an ineffective corrective.

27:23

G: (a) γνωστῶς ἐπιγνώσῃ ψυχὰς ποιμνίου σου (b) καὶ ἐπιστήσεις καρδίαν σου σαῖς ἀγέλαις. "(a) You [sg] should know well the souls of your flock, (b) and set your heart to your herds." G "souls" (= appetites) is borrowed from 12:10, for the less clear "faces" (= appearance). G 27:23–24 and 25–26a are formulated as two sentences. In 27:25–26a, G diverges from the Hebrew sharply to convey a new message, though the text is fairly clear.

S gives context to the following epigram by prefixing a stich: אשר אין ("When you shepherd"). Otherwise S stays close to M in this epigram, except in 27:27.

#### 27:24 אָדור M<sup>K</sup> ] (± conj) נָדָוֹר [

G: (a) ὅτι οὐ τὸν αἰῶνα ἀνδρὶ κράτος καὶ ἰσχύς, (b) οὐδὲ παραδίδωσιν ἐκ γενεᾶς εἰς γενεάν. "(a) For power and strength do not belong to a man forever, (b) nor does he transmit (them) from generation to generation." G translates [DT twice: κράτος and ἰσχύς. Then it removes the problematic , making the implied object the earlier-stated "power and strength." This maneuver is an explication of the metaphor of "crown." For the phenomenon, see the comment on 25:11.

S: איזא איז איזא איזא איזא איזער איזא איזער איזא איזער איז איזער איזע גער אסעגנא. (a) For power is not forever, (b) nor does (one) generation transmit (it) to (another) generation." S works from M in the first stich but, like G, elides the puzzling נזר in the second.

#### 27:25

G: (a) ἐπιμελοῦ τῶν ἐν τῷ πεδίῳ χλωρῶν καὶ κερεῖς πόαν (b) καὶ σύναγε χόρτον ὀρεινόν. "(a) Cultivate vegetation in the field and cut grass, (b) and gather mountain-herbage" (continues in 27:26a). G changes the verse from a temporal clause ("When grass disappears and verdure appears, and the grasses of the hills are gathered") into (self-evident) advice.

#### 27:26

G (continuing 25): (a) ίνα ἔχης πρόβατα εἰς ἱματισμόν. (b) τίμα πεδίον, ίνα ὦσίν σοι ἄρνες. "(a) so that you may have sheep for clothing. (b) Honor the field so that you may have lambs." G 27:26b fills out (and misconstrues) the elliptical Hebrew.

#### 27:27

G: (a) vié,  $\pi \alpha \rho'$  ἐμοῦ ἐχεις ῥήσεις ἰσχυρὰς εἰς τὴν ζωήν σου (b) καὶ εἰς τὴν ζωὴν σῶν θεραπόντων. "(a) My son, from me you have powerful words for your life (b) and for the life of your servants [masc]." Though M is clear, G takes the verse in a new direction, making it into the father's praise of his teachings. It first prefixes "my son," which often appears in statements of this sort. (See the comment on 1:15.) By introducing the father's praise of the teachings, G shifts the import of the epigram in 27:23–27 from the material to the spiritual realm, thereby suggesting that in retrospect 27:23–26 can be read as an allegory of studying the

28:1 נָסוּ וְאֵין־רֹדֵף רָשֶׁע וְצַדִּילִים כִּכְפִיר יִבְטֵח: 2 בְּפָשַׁע אֶֶרָץ רַבִּים שָׂרֶיה וּבְאָדֶם מֵבִין <sup>ר</sup> ז בֵּן יַאֲרִידִ: 3 גֶּבֶר וְרָשׁ וְעֹשֵׁק דַּלִים מָטָר סׁמֵף וְאֵין לֶחֶם: 4 עֹזְבֵי וְתוֹרָה יְהַלְלָוּ רָשֶׁע וְשׁמְרֵי תוֹרָה יִתְגָרוּ

teachings of wisdom. G simply omits ביתך ללחם ביתך because that is not what the new proverb is about.

S: אנרא הכאלא האלא ארא ארא ארא ארא ארא ארא איש אוויא הפאלא איש אוויא (continuing 27:26) "and milk for your goats and your food and the food of your house." The omission of "and the life of your maidservants," together with the introductory "When you shepherd" in 27:23, suggests that the translator wants to make the scene pertain to a shepherd rather than the prosperous estate owners whom M implicitly addresses.

28:1

The ancient versions (like modern translations) naturally adjust the number of either the verbs or the nouns in both clauses for the sake of agreement.

V translates בטח with a doublet: *confidens absque terrore erit* ("shall be confident without terror").

28:2 ידעכן (א בן M S (ארג בי איזעכן; גענע בן G (אמדמסβέσει αὐτάς) (M: gloss; G: div, diath)

Neither M nor G makes good sense in 28:2b; M, moreover, is overloaded. I would conjecturally omit ידע as a gloss on מבין or a synonym variant that has entered the text. We can then translate, "but through an understanding man honesty endures." That is to say, a perceptive man in this context, a wise ruler—secures honesty and therefore stability for his land; see ABP.

G: (a) δι' ἁμαρτίας ἀσεβῶν κρίσεις ἐγείρονται, (b) ἀνὴρ δὲ πανοῦργος κατασβέσει αὐτάς. "(a) By the transgression of the impious [pl], contentions are stirred up, (b) but a clever man will extinguish them." Certainly the translator knew that רבים שריה means "its princes are many," but he was probably puzzled by the notion that iniquity multiplies princes and so chose a different vocalization of the words in question. Iniquity is naturally associated with quarreling and suggested that רבים "רבים" (quarrels"; the otherwise unattested masculine form of the plural is implied by the construct רִים ' in 2 Sam 22:44 = Ps 18:44; Lam

3:58). שריה שריה is ignored or possibly translated באריה שריה שריה '' שר"ר ("strive"), though that word is not translated thus elsewhere. The "land" in question is assumed to be wicked, since it has transgressions; hence it is rephrased as "impious." G גמדמסβέσει αὐτάς = אָריך, i.e., ידעכן ("be extinguished," translated by diathesis; Gemser 1963). M's אריך was not translated, perhaps because it could not be fit into the sentence. Though a translation like this can be called "free," it is at every step governed by the translator's assumptions about both the meaning of the verse and his ideas about Hebrew.

V: (a) propter peccata terrae multi principes (b) et propter hominis sapientiam et horum scientiam quae dicuntur vita ducis longior erit. "(a) For the sin of a land (its) princes are many, (b) and for a man's wisdom and the knowledge of what is said, the life of the prince shall be prolonged." V translates 28:2b expansively and on the basis of the antithesis adds *ducis* to make it clear whose life is prolonged. The idea of the last clause is taken from 28:16.

S: جمادت متحب تعلیلین محتر محتر محمد محب علی دمید عب حصید (a) Because of the iniquity of a land its rulers are many, (b) but righteous men who know honesty will live long." This is a reasonable attempt to deal with the difficulties of M.

28:3 רשע M S (דשע [ (גיאביא M S ( $e^{2}$  ארשע [  $e^{2}$  ארשע (distant dittog עוע  $\rightarrow$  ארשע)

G: (a) ἀνδρεῖος ἐν ἀσεβείαις συκοφαντεῖ πτωχούς. (b) ὥσπερ ὑετὸς λάβρος καὶ ἀνωφελής. "(a) A vigorous man by impieties oppresses the lowly [pl]. (b) As a violent and unprofitable rain" (continues in 28:4). G 28:3a is an isolated monostich, while 28:3b introduces a correlative sentence with the apodosis in 23:4a. ἐν ἀσεβείαις = 𝒴𝔅. A scribe apparently skipped to the 𝔅 of the next word and copied it twice. (Compare the reverse change, 𝒴𝔅 → 𝔅, in 28:15.) Some commentators find the notion of a poor man tyrannizing over the poor puzzling and eliminate it by emendation; see ABP. But the poor, too, can be oppressive and arrogant (as is recognized by Sir 25:2). G speaks of a poor but oppressive tyrant in 28:15–16.

It is strange to see the cardinal virtue of מיאסרנ connected with wrongdoing. M's אין לחם is explained as "unprofitable" ( $\approx$  S), that is to say, a rain that does not produce grain.

28:4 יתגדרו M ] איתגדרו \*  $\approx$  G (περιβάλλουσιν ἑαυτοῖς τεῖχος)  $\approx$  S (כבינה) (near dittog רך  $\rightarrow$  ר

G (continuing 28:3b): (a) οὕτως οἱ ἐγκαταλείποντες τὸν νόμον ἐγκωμιάζουσιν ἀσέβειαν, (b) οἱ δὲ ἀγαπῶντες τὸν νόμον περιβάλλουσιν

בְּם: 
בְּם: 
בַּם: 
בַּם: 
בַּם: 
בַּם: 
בַּלִים 
בְּלַי: 
סִוֹבַדְלָשׁ הוֹלֵדְ בְּתֻמֵּוֹ מֵעִמֵּשׁ דְּרָכַּיִם וְהַוּא עָשִׁיר: 
דְּלָ: 
סִוֹבַדְלָשׁ הוֹלֵדְ בְּתַמֵּוֹ מֵעִמֵּשׁ דְּרָכַיִם וְהַוּא עָשִׁיר: 
דְּלָ: 
הַוֹנוֹ בְּנָשֶׁדְ
דְּלָים יִקְבְּצֵנוּ: 
נוֹצֵר הַיֹם יִקְבָעָשׁרַ הַיַלִים 
דְּלֶבוּ הַשְׁמֵׁע הּוֹנֵן הַוּא יִשְׁמֵי הָוֹנוֹ בְּנָשֶׁדְ

έαυτοῖς τεῖχος. "(a) so do those who abandon the law praise wickedness. (b) But those who love the law surround themselves with a wall." "Surround themselves with a wall" = בה) יתגדרו בה (Jäger) or יתגדרו בה "thereby" and is implicit in the Greek. τεῖχος = τος in Isa 5:5.)

G 28:4 alludes to the law of Moses, as indicated by the article (twice). Other likely allusions are in 28:7, 9, 10. Cook (2002, 282) compares the image in G 28:4b to Letter of Aristeas 139: "When therefore our law-giver, equipped by God for insight into all things, had surveyed each particular, he fenced us about with impregnable palisades and with walls of iron, to the end that we should mingle in no way with any of the other nations, remaining pure in body and in spirit, emancipated from vain opinions, revering the one and mighty God above the whole of creation" (quoted in Cook 1999a, 282, from Hadas 1973, 157). Israel is surrounded by the wall of Torah to keep them from mingling with foreign nations. The concept of Torah as a fence or wall is not biblical and would not have been in the Hebrew archetype.

28:5

verse. By M's sentence division, the Hebrew says that the pious "understand everything," which may have seemed too extreme a claim.

#### 28:6

G: κρείσσων πτωχὸς πορευόμενος ἐν ἀληθεία πλουσίου ψευδοῦς. "Better a lowly man who walks in truth than a deceitful rich man." G condenses the second stich in dependence on 19:22b and turns the proverb into a monostich.

#### 28:8 ובתרבית [ M<sup>Q</sup> ] ובתרבית M<sup>K</sup> (gram)

The repeated preposition in the *ketiv* is optional but perhaps added for extra clarity. This is the type of change that is more typical of the *qere* than the *ketiv*.

#### 28:10

G: (a) δς πλανᾶ εὐθεῖς ἐν ὁδῷ κακῆ, (b) εἰς διαφθορὰν αὐτὸς ἐμπεσεῖται. (c) οἱ δὲ ἀνομοι διελεύσονται ἀγαθὰ [G<sup>BSAMSSRahlfs</sup>; οἱ δὲ ἄμωμοι διελοῦνται ἀγαθὰ G<sup>V106174254261≈MSS</sup>] (d) καὶ οὐκ εἰσελεύσονται εἰς αὐτά. "(a) He who misleads the honest into a bad way—(b) he will fall into corruption. (c) The lawless will pass through good things [or 'goodness'; var: 'The innocent will divide up (= inherit) good things'] (d) but not enter into them [or 'it']." G construes בשחות (M inis corruption) stich c in the first variant (per Rahlfs) relates to M only in ἀγαθά = טוב. G 28:10d was added to make sense of the strange statement in 28:10c. However, the second variant in 28:10c, which is supported by de Lagarde, BHQ, and others, is correct.

וּתְמִימִים יִנְחֲלוּ־טְוֹב: 11 חָבֶם בְּעֵינָיו אֵישׁ עָשָׁיר וְדָל מֵבִין יַחְקָרֶנּוּ: 12 בַּעֲלָץ צַדִּיקִים רַבְּה תִפְאָרֶת וּבְקוּם רְשָׁעִים יְחֻפְּשׁ אָדֲם: 13 מְכַפֶּה פֲשָׁעָיו לָא יַצְלֵיח וּמוֹדֶה וְעֹזֵב יְרָחֶם: 14 אַשְׁרֵי אָדָם מְפַחֵד תְּמֵיד וּמַקְשֶׁה לְבוֹ יִפְּוֹל בְּרָעֵה: 15 אָרִי־וָּהֵם וְדַב שׁוֹתֵק משֵׁל רְשָׁע עַל עַם־דֵּל: 16 נָגִיד חַסֵר תֶבוּנוֹת וְרֵב מַעַשַׁקּוֹת ישוֹנֵא

#### 28:11

S: משני usually means "despise," but according to CAL it can also mean "rebuke" or "refute." "Refute" may be an explication of יחקרנו, since what one does to an adversary in litigation is refutation rather than interrogation. Pinkuss mentions the possibility of emending משני ("investigates"). This is graphically feasible, though that correspondence never occurs elsewhere. Perhaps the notion of a poor man "investigating" the rich one made no sense to the translator, so he substituted "despises," since this is the feeling that the rich man's smugness would naturally arouse.

#### יתפש איתפש [ א יחפש אינע מאנגיא איתפש (graph $\to \pi$ ) (graph $\to \pi$

G: (a) διὰ βοήθειαν δικαίων πολλὴ γίνεται δόξα, (b) ἐν δὲ τόποις ἀσεβῶν ἁλίσκονται ἀνθρωποι. "(a) By the help [or 'salvation'] of (the) righteous [pl], honor becomes great, (b) but in the places of the wicked, men are captured." διὰ βοήθειαν for בעלץ is puzzling. Retroversions to בעזר (Baumgartner) or, less likely, בהעל (de Waard 1993, 255–56; cf. Ps 70[69]:2) are graphically distant. In 28:12b, ἐν δὲ τόποις does not necessarily represent במקום (de Lagarde and many) for M's הניר, since the latter is translated the same way in 28:28. בקום may have been interpreted as "places" because it seemed unclear how the wicked could "arise." G's ἁλίσκονται = (Jäger) is a miswriting of Jager).

S: אבהביא אוגים איבה איבה איבה איבה איבה איבה. (a) By the strength of the righteous [pl] honor is great, (b) but by the ascent of the wicked [pl], it is diminished [var: 'is sought']." הראבים is based on G. האבים should be emended to אום ("it [sc. honor] is sought") (Pinkuss); ≈ T מבצי T.

#### 28:13

G: (a) ὁ ἐπικαλύπτων ἀσέβειαν ἑαυτοῦ οὐκ εὐοδωθήσεται, (b) ὁ δὲ

έξηγούμενος ἐλέγχους ἀγαπηθήσεται. "(a) He who covers up his own iniquity will not succeed, (b) but he who pronounces rebukes will be loved." G correctly interprets M's ambiguous first line as a warning against hiding one's own sins. G 28:13b praises rebukers, not penitents, as M does. (ἐλέγχος elsewhere corresponds to תוכחת, so BAP's "ses torts" is not justified.) The root of provide the form) is treated as an Aramaism meaning "loved."

#### 28:14

G: (a) μακάριος ἀνήρ, ὅς καταπτήσσει πάντα δι' εὐλάβειαν, (b) = M. "(a) Happy is the man who fears always out of respectfulness." G adds δι' εὐλάβειαν for clearer definition of what kind of fear is being praised. εὐλάβεια is the virtue of discretion and respect, which the Stoics considered a worthy emotion, whereas simple trepidation (φόβος) was unworthy; see Dick 1990, 25; cf. BAP. καταπτήσσειν, though literally meaning "cower," signifies an admirable type of fear in 29:9b.

#### רשע א S (ארשע א S (גע בארש S (גע בארש S (גע בארא )) א רשע א G ( $\pi \tau \omega \chi \delta \varsigma$ ) (haplog $\gamma \to \gamma$ )

G: (a) λέων πεινῶν καὶ λύκος διψῶν (b) ὅς τυραννεῖ πτωχὸς ἂν ἔθνους πενιχροῦ. "(a) A hungry lion and a thirsty wolf (b) is he who, though poor, tyrannizes over an impoverished people." "Hungry" and "thirsty" are contextual guesses at the hapax קוקם and נהם (The translator may not have known the meaning of נהם; see the comment on 19:12.) G chooses to associate ד with Aramaic ("wolf"), though ד means "bear" in Aramaic, too; see the comment on 17:12. BAP notes the assonance in λέων πεινῶν καὶ λύκος διψῶν ... πτωχὸς ἂν, but it is doubtful that this aesthetic quality was enough to motivate the mistranslation of a well-known word, especially since this is not a factor in 17:12. πτωχός =  $\psi$  for M's  $\psi$  lost by haplography with the next word. The opposite happens in G 28:3,  $\psi$  To  $\psi$ . The possibility of an impoverished ruler oppressing his people is raised in 28:16.

## 28:16 הבונות M S (הונא) V (prudentia) [ הנוא G (הונא) T<sup>L</sup> (הונא) (transp בונ $\rightarrow \rightarrow$ נוב $\rightarrow \rightarrow$

שנא  $M^Q G (\mu \iota \sigma \tilde{\omega} \nu) S (\star \omega \iota num)$  שנא  $M^K (num)$ 

The singular (*qere*) is required grammatically. It is unclear how the plural *ketiv* arose.

G: (a) βασιλεὺς ἐνδεὴς προσόδων μέγας συκοφάντης, (b) ὁ δὲ μισῶν ἀδικίαν μακρὸν χρόνον ζήσεται. "(a) A king lacking revenues is a great oppressor, (b) but he who hates injustice will live a long time." G ele-

# בָצַע יַאֲרֵידְ יָמִים: בּצַע יַאֲרֵידְ יָמִים: דור אָדָם עָשִׁק בְּדַם־גָפָשׁ עַד־בּוֹר זְנוּס אַל־יִתְמְכוּ־בְוֹ: 18 הוֹלֵדְ הֻּמִים יוּשֵׁע וְנֶעְקָשׁ דְּרָכַיִם יִפְּוֹל גַּדַרַאַדְמָתוֹ יִשְׂבַּע־לֶחֶם וּמְרַדֵּף רֵׁלִים יִשְׂבַּע־רִישׁ:

vates the status of גגיד by translating it as βασιλεύς; see the comment on 19:6. In this case, the change turns the proverb's criticism against the king himself; see the comment on 16:12. προσόδων ("public revenues") apparently represents קנובות, though this is a unique correspondence. The variant arose by chiastic transposition בוב  $\rightarrow$ בוג, a change perhaps encouraged by mention of an oppressive ruler in the preceding verse. But מעשקות is translated συχοφάντος, which properly means "false accuser." But συχοφάντο- translates forms of  $\chi$ wg in Proverbs (14:31; 22:16; 28:3, 16), and usually elsewhere, and seems to refer to oppressive practices more generally.

T: It is striking that for M's תנובות T<sup>L</sup> has אונא ("wealth")  $\approx$  תנובות, a reading that is close to G but does not come by way of S (T<sup>Z</sup> corrects to to Correct to R). T does not elsewhere translate תנובה this way, and one wonders if an exceptical tradition lay behind the translations of הבונה as "revenues" or "wealth." The idea that an impoverished ruler may be an oppressor is not self-evident and is unlikely to have arisen from independent interpretation of M alone. This seems to be an independent T variant.

S: אסבאי געשאי געשאי געשאי געשאי איז איז געשאי האלא (a) A ruler who lacks sense—many are his injuries, (b) and the one who hates deceit will extend his days." S, perhaps puzzled by the fact that M seems to say nothing about the foolish prince, interprets מעשקות as the misfortunes that *he* will suffer. (In ABP I argue that 28:16a forms a tristich with 28:15 while 28:16b is an independent monostich.)

28:17 עד בור ינוס M S (באדא גבוס ) ] איר בור ינס G (שטאא מאז א כור שלים) (graph ד היסט), orth, metath רב אר רר haplog א ר רר א div) א סיא, div) א בור M<sup>Or</sup> (orth)

fin ] + 2 stichoi G (17*a*) (elab)

G: (a) ἄνδρα τὸν ἐν αἰτία φόνου ὁ ἐγγυώμενος (b) φυγὰς ἔσται καὶ οὐκ ἐν ἀσφαλεία. "(a) He who provides bail for a man charged with murder (b) will be a fugitive and not (live) in security." According to Jäger, ὁ ἐγγυώμενος = עורב, and indeed that is how G understood his text, but there are other factors to be accounted for as well. G is a paraphrase induced by reading, approximately, ס(ו), ערבו יָנָט This variant could have arisen by the following hypothetical changes, starting from a *defectiva* text (partly supported by M<sup>Or</sup>) with slight or no word separations: ערבינס (עַדברינס) (עַדברינס) ערבינס ערבינס ערבינס, ערבינס), The reconstructed Hebrew verse is ערבינס אל יתמכו יעם, which would best be translated, "A man who oppresses by murder [lit. 'by life's blood']—his guarantor will flee. Let them not support him." This awkward sentiment arose by mechanical errors, not interpretive adjustments. G construes the last phrase as an indicative: they will not support him; hence (effect for cause) will not live in security.

S: The meaningless ("to the pit, lion's] den") = T לגובא (Pinkuss).

#### 28:17*a* + יסר בנך ויניחך ויתן מעדנים לנפשך (transf from 29:17)

G: (a) παίδευε υίόν, καὶ ἀγαπήσει σε (b) καὶ δώσει κόσμον τῆ σῆ ψυχῆ. (c) οὐ μὴ ὑπακούσης [G<sup>B</sup>; var: ὑπακούση G<sup>A</sup>] ἔθνει παρανόμω. "(a) Educate (your) son and he will love you (b) and bestow an ornament on your soul. (c) You should not obey [var: 'he will not obey'] a lawless nation." G 28:17*a*ab is similar to G 29:17 (= M 29:17). However, G 28:17*a*ab differs from G 29:17 in two ways: אינר פון פֿאַמָאָקָרָנוּ פֿעָרָדָן but ἀναπαύσει σε in 29:17, and דְעָרָדָן = ἀγαπήσει σε in 28:17*a* but ἀναπαύσει σε in 29:17, and τῆ ψυχῆ σου in 29:17. These differences indicate that 28:17*a*ab is not a copy of G 29:17 but is based on a Hebrew text. G 28:17*a*ab was then expanded by 28:17*a*c. The latter seems unrelated to the context, except, somewhat, to 28:16. The reasons for these developments are obscure.

#### 28:18

As in 28:6, the versions do not represent the dual of דְרָכֵיִם.

S has לישרי ("pit"), which is either a contextual guess (independent of G) or represents בשחת. The latter makes sense and perhaps is the original reading; thus ABP (cf. the role of the pit in 26:27; 28:10; Pss 7:16; 9:16). It is, however, hard to explain how בשחת became האחת Most of those who maintain באחת understand it to mean "at once," "immediately," like Aramaic כחדא context and RH באחר to the context of the second seco

20 אֵישׁ אֱמוּנוֹת רַבּ־בְּרָכֵוֹת וְאָץ לְהַעֲשִׁיר לָא יִנְּקֶה: 21 הַבּּר־פָּגִים לֹא־טְוֹב וְעַל־פַּת־לֶׁחֶם יִפְשַׁע־גֲבֶר: 22 נְבֶהֶל לַהוֹז אֲישׁ רַע עֻיִז וְלְא־ יֵׁדַע בִּי־חֶסֶר יְבֹאֶנּוּ: 23 מֿוֹכִיַח אָדָם <sup>ר</sup>אַחַר<sup>-1</sup> תַז יִמְצֵא מְמַחַלִיק לְשוֹז: 24 גוֹזֵל | אָביו וְאַמֹּו וְאַמֵר אֵין־פָּשַׁע חָבֵר הוּא לְאֵישׁ מַשְׁחִית: 25 רְחַב־גָפָשׁ יְגָרֶה מְדֵוֹז וּבוֹטֵח עַל־יְהוֶה יְדָשֵׁן:

#### 28:20

G: (a) = M; (b) ό δὲ κακὸς οὐκ ἀτιμώρητος ἔσται. "(b) but the evil man will not go unpunished." G moralizes M 28:20b (which reads, "but he who hastens to get rich will not be held guiltless"). The translator resists threatening punishment for the man who strives for wealth, because this is not prohibited behavior. For this reason, we should not retrovert G κακός to χακός to βHQ). See the comment on 10:22.

#### 28:21

G: (a) δς οὐκ αἰσχύνεται πρόσωπα δικαίων, οὐκ ἀγαθός. (b) ὁ τοιοῦτος ψωμοῦ ἄρτου ἀποδώσεται ἀνδρα. "(a) He who does not respect [lit. 'feel awe/shame toward the faces of'] the righteous [pl] is not good. (b) Such a one would deliver [or 'sell'] a man for a bit of bread." By translating 𝒴𝒷 as "would deliver," G specifies the sin in question and derogates the man who would sell the righteous into debt slavery. Amos 2:6 calls the selling of the poor for a pair of sandals a 𝒴𝒷. In one interpretation of the text, Amos's idiom refers to selling the poor into debt slavery for owing a trivial sum. Quite likely the phrase in Prov 28:21b, "would sin for a bit of bread," recalled that verse to the G translator. G has ἀποδιδόναι here and in Amos 2:6 (for מכר), as well as in the sale of Joseph in Gen 37:27.

S ריפשע ("deliver') for אילה ("deliver") היילה G.

ער אדער 19:22 MS (אר מאנאי) א דער 10 G (לאנאנאי) (graph  $T \to T$ 

G: (a) σπεύδει πλουτεῖν ἀνὴρ βάσκανος (b) καὶ οὐκ οἶδεν ὅτι ἐλεήμων κρατήσει αὐτοῦ. "(a) An envious man hurries to get rich, (b) and he does not know that a merciful man will rule over him." "Merciful man" = ¬, interpreted as an abstract-for-concrete trope. G means that because the mean-spirited man is too fixated on pursuing wealth to give alms (cf. BAP), the merciful one will ultimately prevail over him. This is the message of 28:27 as well. The present verse, to be precise, characterizes the envious man rather than evaluating overwork. G tends to avoid the idea that hurrying after wealth is bad; see the comment on 10:22.

28:23 ↔ ארח ; אַחָרַי (δδούς) (M: unc; G: metath רח → רח א ארח (לסטיג) (M: unc; G: metath רח → רח)

M's אָחָרָי is awkward (lit. "he who reproves after me [?]"). For a survey and critique of explanations, see ABP. אחר fits well and is a reasonable conjecture. Possibly M arose by near dittography  $\rightarrow \rightarrow \neg$ . The verse is to be translated, "He who reproves another will find more favor than a flatterer"; cf. 15:12.

G: (a) ὁ ἐλέγχων ἀνθρώπου ὁδοὺς (b) χάριτας ἕξει μᾶλλον τοῦ γλωσσοχαριτοῦντος. "(a) He who rebukes the ways of a man (b) will have favor more than the flatterer." ὁδοὺς = Π, with change of number, as often in G, or possibly ארחות Neither produces a meaningful Hebrew sentence.

S: אדרים לבוידא העביא האבעי לב כא איז גרפוע בלצוא. "(a) He who reproves a man finds friends (b) more than he who divides by his tongue." S seems to ignore אחרי perhaps considering it adequately implied by "man." He then interprets אחרי ("favor") as an abstract-forconcrete metonymy for "friends." אחרי is translated "etymologically"; cf. S-Prov 29:5. T's קדמוי ("before him") is an attempt to make sense of the difficult אחרי, which he does not see represented in S.

#### 28:24

G translates גוזל ("robs") as ἀποβάλλεται ("expel"). G defines the robbing of one's parents more narrowly as *expelling* them from the ancestral estate. This interpretation, probably correct (see ABP), is based on 19:26a (משדד אב יבריח אם), though these two verses differ in their wording in Hebrew and G.

#### 28:25

G: (a) ἄπληστος [G<sup>A</sup>; var: ἄπιστος G<sup>BS</sup>] ἀνὴρ κρίνει εἰκῆ, (b) ὅς δὲ πέποιθεν ἐπὶ κύριον, ἐν ἐπιμελεία ἔσται. "(a) An insatiable [var: 'unfaithful'] man judges rashly, (b) but he who trusts in the Lord will be attentive." "Judges

26 בּוֹטַח בּּלִבּוֹ הַוּא כְסֵיל וְהוֹלֵדְ בְּחָכְמָה הַוּא יִמְּלֵט: 27 נוֹתַן לְרָשׁ אֵין מַחְסֵוֹר וּמַעְלִים עֵׁינִיו רַב־מְאֵרְוֹת: 28 בְּקוּם רֲשָׁעִים יִפָּתַר אָדֶם וּבְאָרָדָם יִרְבָּוּ צַדִּיקִים: 1:92 אֵישׁ וּזֹכָחוֹת מַקְשָׁה־עֶׁרֶף פֶּתַע יִשְׁבֵׁר וְאֵין מַרְפֵּא: 2 בִּרְבַוֹת צַדִּיקִים יִשְׂמַח הָעֶם וּבִמְשָׁל רְשָׁע יֵאָנַח עֶם:

3 אִישׁ־אֹהֵב חֲכְמָה יְשַׂמַּח אָבֵיו וְרֹעֵה זוֹנוֹת יְאַבָּד־הְוֹן: 4 מָׂלֶד הַמִשְׁפָּט יַעֲמִיד אֶֶרֶץ וְאֵישׁ תְּרוּמַוֹת יֶהֶרְסֶנָּה: 5 גֶּבֶר מַחֲלֵיק עַל־

rashly" derives מדון from דין ("judgment"); cf. 23:29. I understand לע להוועבאבלק, lit. "in attention, care," to mean "being cared for," as in 13:4, too. This has a more edifying tone than the literal "made fat."

#### 28:26

G has θρασεία καρδία ("in an arrogant [or 'bold'] heart") for M's ("in his heart"). Since "heart" is often equivalent to wisdom (e.g., 15:32), the translator adds the adjective to make it clear that the warning concerns a bad type of heart.

#### 28:28

G: (a) ἐν τόποις ἀσεβῶν στένουσι δίκαιοι, (b) ἐν δὲ τῇ ἐκείνων ἀπωλεία πληθυνθήσονται δίκαιοι. "(a) In the places of the wicked [pl], the righteous groan, (b) but in their [sc. the wicked men's] perishing, the righteous will be increased." On ἐν τόποις = בקום, see the comment on 28:12. The groaning of the righteous in unhappiness at the social dissolution (derived from 29:2; and cf. Ezek 9:4) replaces the unbefitting image of them hiding. When the clever man does hide (27:12), it is to avoid a quarrel.

29:1 ישבר M S (שא<br/>לבי) ] ישרף G (קאנאט מטֿדסטֿ) (graph ב $\to$ ב, metath ר<br/>  $\to$ ס)

S: (a) אבא המבע הארא המבא המבא (b) = M. "(a) A man who does not accept reproof and whose neck is stiff." S identifies the problematic איש תוכחות (lit. "a man of reproofs") as one who does not accept them. S מגא מגא מגא M.

## 29:2 ברבות M S ( $\sim$ $\sim$ ) ] בברכות M S ( $\sim$ $\sim$ ) (dittog בבי $* \approx$ G (לאשנענענענענענט) (dittog בב , graph $2 \rightarrow$ )

G's ἐγκωμιαζομένων =  $\Box$  (Jäger), which it treats, loosely, as a participle; cf. the correspondence in G 10:7. 29:2a but ἄνδρες in 29:2b, showing G's tendency (albeit inconsistent) to enhance variation of synonyms, apparently for aesthetic reasons.

For unclear reasons, S has שבי for ישמה. This is not "assimilation" to 29:16 (per BHQ); that verse is very different.

29:3

29:4 יהרסנה M G (אמדמסא<br/>מ $\pi \pi \epsilon \iota)$  ] יהרסנה (graph ה^1  $\to$  <br/>ה, metath סר $\to$ רס)

G's  $\pi$ מףמעסעסג ("transgressor") = איש תרמית (Jäger) or construes M's lin that sense, perhaps correctly.

29:5

G: (a) δς παρασχευάζεται ἐπὶ πρόσωπον τοῦ ἑαυτοῦ φίλου δίκτυον, (b) περιβάλλει αὐτὸ τοῖς ἑαυτοῦ ποσίν. "(a) He who prepares a net before his friend (b) throws it about his own feet." The translator assumed that רשת belongs to the first clause and so found it necessary to give a vague translation ("prepares") that could take "net" as its direct object. He might have had in mind someone smoothing out a net on the ground. M actually means "flatters": a man who *flatters* his fellow is in effect spreading a net to trap him.

ַרַעֵּהוּ דָּשָׁת פּוֹרֵשׁ עַל־פְּעָמְיו: 6 בְּפָשָׁע אַישׁ רֲע מוֹקָשׁ וְצַדִּיק יָרָוּן וְשָׂמֵחַ: 7 יֹדֵעַ צַדִּיק דֵּין דַּלֵּים דְׁשָׁע לֹא־יָבִין דֱעַת: 8 אַנְשֵׁי לָצוֹן יְפִיחוּ קַרְיֶה וְחַכָמִים יָשִׁיבוּ אָף: 9 אִישׁ־חָכָם גַשְׁפָּט אֶת־אַישׁ אָוֵיל וְרָגַז וְשָׁחַק וְאֵין גֶחַת: 10 אַנְשֵׁי דָמִים יִשְׂנְאוּ־תֶם וִישָׁרִים יְבַקְשָׁוּ נַפְשָׁוֹ: 11 כָּל־רָוּחוֹ יוֹצִיא כְסֵיל וְחָכָם בְּאָחוֹר <sup>ו</sup>יַחְשָׂכֶנָה<sup>ז</sup>:

29:6 בישא M S (בישא G ( $\mu \epsilon \gamma \alpha \lambda \eta$ ) (graph  $\mathcal{V} \to \exists paleo \mathcal{P} \to \mathcal{P}$ ?)

G: (a) ἁμαρτάνοντι ἀνδρὶ μεγάλη παγίς, (b) = M. "(a) For a sinning man there is a great trap." It is hard to see a motivation for μεγάλη other than G's having read  $\neg \neg$  for  $\lor \neg$  (Jäger). The letters  $\lor$  and  $\neg$  have a certain resemblance in the paleo script ( $\neg, \neg$ ). G's reading is less suitable than M's, because it is not the size of a trap that makes it dangerous but its "evil."  $\lor$  evil. Letters  $\lor$  and  $\lor$  have a certain source of a trap that makes it dangerous but its "evil."  $\lor$  evil.

S's כאאביג ("is trapped") = מוּקָשׁ, an Hp-stem participle; thus too in 12:13a.

29:7

G: (a) ἐπίσταται δίκαιος κρίνειν πενιχροῖς, (b) ὁ δὲ ἀσεβὴς οὐ συνήσει γνῶσιν, (c) καὶ πτωχῷ οὐχ ὑπάρχει νοῦς ἐπιγνώμων. "(a) The righteous man knows how to judge [on behalf of] the poor [pl], (b) but the impious one does not understand knowledge; (c) and a poor man does not have an understanding mind." BAP translates ἐπιγνώμων more narrowly, as "lesprit d'arbitrage." G 29:7b and 7c are forms of M 29:7b, with G 29:7b adjusting to M. G 29:7c is apparently based on reading  $\mathfrak{W}$  instead of  $\mathcal{W}$ , under the influence of the preceding word,  $\Box \mathcal{V}$ . It cannot be determined if the slip was a scribe's or the translator's. G 29:7c recalls the attitude expressed in Pirqe Avot 2:5: "A common man cannot be pious." **29:9 את** M ] את M ] את M ] את M (śen) (syn)

G: (a) מֹעחֹף ססּשְׁטֹּלָ אָרְוֹעִנּ בְּשׁעָא, (b) מֹעחֹף לּבֹּ שְׁמֹטֹלְסָ סְּרְוֹלָטְנָצְעסָ אָמִדמיָרָאָמִדמוּ אָמוֹ סט אָמדמידחֹיָסְסְבוּ. "(a) A wise man judges nations, (b) but a worthless man, being angry, mocks and does not fear." Baumgartner retroverts אָרְוֹעוּבּי בֿפּעיאָר די רישׁבּט אָם ("will judge a people"), but the graphic difference is great. Also, שׁבּט אָם יוֹר ("will judge a people"), but the graphic difference is great. Also, שׁבּט אָם יוֹר ("will judge a people"), but the graphic difference is great. Also, שׁבּט אָם is not attested, though, to be sure, it could have been mistakenly understood by back-formation from אָמִים אַמי. De Waard (1993, 257) suggests that בווו אוין represents אַמִים אוין האויל, not as a text but as an exegetical metathesis of אויל, not as a text but as an exegetical metathesis of אויל אויל. The latter word, however, lacks a שׁ and is already represented by שְׁמּטֹלָסָ. More likely, the source text had שָׁ, intended as שָ מוּסוֹלָסָ, a synonym of אַר (here, "with"), but the translator misconstrued it as a ("people"). G associates שׁרָל אַרוֹם

#### 29:10

For M's אנשי דמים G has an explanatory ἄνδρες αἰμάτων μέτοχοι ("Men who take part [in crimes of] blood"), with the added μέτοχοι based on μετέχοντες in 1:18.

S: (a) = M; (b) הארביא היאריט (b) but the righteous love him." S creates antithetical parallelism with "hate" to make sense of M's puzzling וישרים יבקשו נפשו (lit. "and the upright seek his [the innocent man's] life").

29:11 יחשבנה (דמµונינידמו)  $\approx V(differt\ et\ reservat)$  ] יחשבנה M; יחשבנה S (כאניד (M: transp שכח  $\rightarrow \square$  שכח קר קר (C scale) (M: transp שכח  $\rightarrow \square$ 

G: (a) = M; (b) σοφός δὲ ταμιεύεται κατὰ μέρος. "(b) but the wise man reserves (it) in part." BAP translates the last phrase as "la dispense à mesure," in accordance with the classical usage, ταμιεύεσθαι = "to be treasurer." But the word means "hold in store" in 4 Macc 12:12. G is reading יחשׂכנה ("hold it back") (de Lagarde), and κατὰ μέρος corresponds loosely to באחור

V: (a) = M; (b) sapiens differt et reservat in posterum. "(b) but the wise man defers and reserves (his thought) until later." Since V shows independence in the handling of באחור, it supports the existence of יחשכנה. The verb אחנה is used of restraining one's speech in Job 7:11; Prov 10:19; and 17:27 (but, to be sure, is not translated by *differo* or *reservo*).

S: (a) = M; (b) האנאיבר הואנאים אונעים. "(b) but the wise man thinks it in his mind." "Thinks it" = יחשבנה.

12 אַשׁל מַקְשִׁיב עַל־דְּבַר־שָׁאֶר בּּל־מְשָׁרְתָיו רְשָׁעִים: 13 רָשׁ וְאִישׁ הְּכָבִים נִפְגָשׁו מֵאִיר־מֵיגֵי שְׁנֵיהֶם יְהוֶה: 14 מֶעָּד שׁוֹפֵט בָּאָמֶת דַּלֵּים הְּכָבִים נִפְגָשׁו מֵאִיר־מֵיגֵי שְׁנֵיהֶם יְהוֶה: 14 מֶעָד שׁוֹפֵט בָּאָמֶת דַּלֵּים בְּסָאוֹ לְעַד יִכְּוֹן: 15 שֵׁבֶט וְתוֹכַחַת יִתּן חְכְמֵה וְנַעַר מְשָׁלְח מֵבִישׁ אִמְוֹ: 16 בִּרְבָוֹת רֲשָׁעִים יִרְבֶּה־פֶּשׁׁע וְצַדִּילִים בְּמַפּּלְתָם יִרְאוּ: 17 יַפֵּר בְּנִדְ וִינִיחֵדְ וִיהֵן מַעֵדַנֵּים לְנַפִּשֵׁדָ: פּ

#### 29:12

G translates משל (M משל), uniquely, as  $\beta \alpha \sigma i \lambda \epsilon \omega \varsigma$  ("king"); see the comment on 19:6.

#### 29:13

G: (a) δανιστοῦ καὶ χρεοφειλέτου ἀλλήλοις συνελθόντων (b) ἐπισκοπὴν ποιεῖται ἀμφοτέρων ὁ κύριος. "(a) When creditor and debtor have a meeting with each other, (b) the Lord has them both under observation" (trans. McKane). G's "creditor" and "debtor" are a narrower interpretation of שר ("poor") and שיש תככים ("man of oppressions"), in reverse order. G thereby creates a more precise antithesis and a closer association between the two types. They interact in a particular way, lending and borrowing. (De Lagarde suggests that G associates ["usury"], which would be a case of interlingual homoiophony.)

The phrase מאיר עיני ("gives light to the eyes of") is understood as God making his own eyes "shine on"—that is to say, look upon—the two men. This formulation provides a warning pertinent to the situation envisioned by G. If, in such an encounter, the lender wishes to pressure the debtor, he should remember that God has both under view.

V translates the first three words as *pauper et creditor* ("a poor man and a creditor"), combining M and G.

S's איש תככים ("a hurtful man") is a good approximation of איש תככים and is reinforced by association with Syriac איש ("harm," "injury"). T's אסבא ("an intermediary") associates תככים with אניא.

#### 29:14

G's μαρτύριον ("for a witness") vocalizes לְעָד for M's לְעָד; similarly in 12:19. However, "forever" is the more appropriate sense here (unlike in 12:19). G says that the royal throne is a *witness*. Possible background is Ps 89:38, which (apparently) calls David's throne a "faithful witness" (though it is not clear that G-Psalms itself construes the Hebrew this way). S: כתבאה געשה בגנאה בסויסבת הגואה שלח. "(a) A king who judges in truth (b) his throne will be established in justice." The omission of דלים extends the dictum to all royal judgments. It is not clear that is based on G (per BHQ) rather than taken from the idea of 16:12; 25:5, though that word is not used there. (It is used of royal justice in S 29:4 and 31:9.)

### 29:15

For M's "his mother," G has γονεῖς αὐτοῦ ("his parents"), since, the translator may presume, both must feel the shame.

### 29:16

G's הסאלא איז ("become fearful") confuses איראו "see" with the homograph "fear."

S's נעגנס ("will rejoice") makes the consequence of *seeing* explicit. It also uses one component of the pair ראה ושמח (see especially Pss 107:42; 119:74) to suggest both.

### 29:17

In G, this verse appears both here and in G 28:17*a*ab; see the comment there.

### 29:18

G: (a) οὐ μὴ ὑπάρξῃ ἐξηγητὴς ἔθνει παρανόμω, (b) ὁ δὲ φυλάσσων τὸν νόμον μακαριστός. "(a) For a lawless nation there is no interpreter, (b) but he who keeps the law is most fortunate." In the translator's time, prophecy was no longer sanctioned or was at least viewed with suspicion. Hence G (in line with the later rabbinic attitude) asserts that *interpreters* are indispensable to it. GELS glosses ἐξηγητής as "expounder," "interpreter" but also suggests that ἐξηγητής (one of whose meanings is "leader," "guide" [LSJ]) can mean "superintendent," "guide," like Aramaic μ. BAP notes that the dream interpreters in Gen 41:8 and 24 are called ἐξήγται (cf. G-Judg B 7:13), but in the present context, the ἐξηγητής is an interpreter of the law. G 29:18a ≈ G 28:17*ac*. V maintains *prophetia* "prophecy."

S: , mand kows it in kows side karding kommender is he multiplicity of evildoers a people is broken, (b) but fortunate is he who keeps the law!" S apparently shares G's scruples about asserting the necessity of prophetic vision, but it solves the problem independently by substituting "evildoers" as an antithesis to 29:18b (recalling in part 29:16).

יִפְּרֵע עֲם וְשֹׁמֵר תּוֹרֶה אַשְׁרֵהוּ: 19 בִּדְבָרִים לֹאִ־יָוָּסֶר עֲבֶד כִּי־יָּבִין וְאֵין מַעָגָה: 20 חָזִיתָ אֵישׁ אֲץ בִּדְבָרֵיו תִּקְוָה לִכְסֵיל מִמֶּנוּ: 21 מְפַנַק מִנִּעַר עַבְדֶוֹ וְאַחַרִיתׁוֹ יִהְיֶה מְנִוֹן: 22 אִישׁ־אַף יְגָרֶה מְדֵוֹן וּבַעַל חֵמֶה רַב־פְּשַׁע: 23 גַּאַוַת אֶדָם תַּשְׁפִּילֶנוּ וּשְׁפַל־רוּחַ יִתְמְדָ כָּבְוֹד: 24 חוֹלֵק עִם־גַּנְּב שוֹנֵא נַפְּשֶׁוֹ אָלֶה יִשְׁמַע וְלָא יַגִּיד: 25 תֶרְדַּת אָדָם יִתֵּן מוֹקַשׁ וּבוֹטֵח בִּיהוֶה יִשָּׁגֵב: 26 רֵבִּים מִבַקְשִׁים פּּנִי־מוֹשֵׁל

### 29:19

G adds σκληρός to οἰκέτης (hence "a stubborn slave"), to avoid the implication that *no* slaves will accept verbal rebuke.

S: (a) = M; (b) בא הלא כלב (b) for he knows that he is not beaten." S vocalizes מְעָנָה (lit. "tortured") (Pinkuss).

### 29:21

G: (a) δς κατασπαταλ $\tilde{q}$  ἐκ παιδός οἰκέτης ἔσται, (b) ἔσχατον δὲ όδυνηθήσεται ἐφ' ἑαυτ $\tilde{\omega}$ . "(a) He who lives luxuriously from childhood will be a servant, (b) and will in the end be miserable about himself." Because G assumes that מפנק is passive (מְכַנָּק), he divides the sentence after מנער and treats עבדו ("his servant") as a predicate clause. όδυνηθήσεται ἐφ' ἑαυτ $\tilde{\omega}$  is a contextual guess for the obscure June.

V interprets מנון from context as contumacem "obstinate."

# 29:22 יגרה M S (באו ג' G (ὀρύσσει) (graph גרה 29:22)

G: (a) ἀνὴρ θυμώδης ὀρύσσει νεῖκος, (b) ἀνὴρ δὲ ὀργίλος ἐξώρυξεν ἑμαρτίας. "(a) A hot-tempered man digs up conflict, (b) and an angry man digs out sins." ἐξώρυξεν is used for the sake of paronomasia with ὀρύσσει. The wordplay would not have come about, however, unless the source text had יִבְרָה) in 29:22a.

#### 29:23

G: (a) = M; (b) τούς δέ ταπεινόφρονας έρείδει δόξη κύριος. "(b) but the

Lord supports the humble-minded [pl] with honor." G assumes that תמך means "support" (see the comment on 4:4) and supplies an appropriate subject, the Lord.

### 29:24

G: (a) δς μερίζεται κλέπτη, μισεῖ τὴν ἑαυτοῦ ψυχήν. (b) ἐἀν δὲ ὅρκου προτεθέντος ἀκούσαντες μὴ ἀναγγείλωσιν. "(a) He who shares with a thief hates his own soul. (b) If, hearing a vow being put forth, they not make a report" (continues in 29:25). Verses 24–25 form a quatrain on the dangers of becoming involved with a thief. The plural ἀναγγείλωσιν (where M has sg) must refer to unmentioned witnesses.

S: (a) = M; (b) אמאס אלא האד, האסת, "(b) And they decree oaths for him but he does not confess." S has a different legal procedure in mind, one in which the suspect is forced to swear to his innocence. In this case, he chooses to perjure himself.

### 29:25

G: (a) φοβηθέντες καὶ αἰσχυνθέντες ἀνθρώπους ὑποσκελισθήσονται [var: ὑπεσκελίσθησαν G<sup>BSRahlfs</sup>], (b) ὁ δὲ πεποιθὼς ἐπὶ κύριον εὐφρανθήσεται. (c) ἀσέβεια ἀνδρὶ δίδωσιν σφάλμα. (d) ὃς δὲ πέποιθεν ἐπὶ τῷ δεσπότῃ, σωθήσεται. "(a) because they fear and reverence [or 'feel shame toward'] men, they will be overthrown [var: 'have been overthrown'], (b) but he who has trust in the Lord shall be happy. (c) Impiety will make a man stumble, (d) but he who trusts in the Master will be saved." There is some confusion in the Hexaplaric markings (see Fritsch 1953, 179), but 29:25cd is clearly a corrective doublet of the freer 29:25ab. Stich 29:25a has a double translation of חרדת אדם.

S translates אדם מחרדת אדם as באליס געבוי ("the iniquity of a man"). באליס לוויד does not occur elsewhere in Proverbs, but in the Pentateuch S understood the meaning of  $\pi$  correctly. Perhaps S-Proverbs assumed, too "logically," that sin rather than fear is the trap, for fear should help a person avoid sin.

### 29:26

G translates מבקשים as θεραπεύουσιν ("serve," "pay court to"), borrowing a term from the near-doublet in G 19:6a, where the verb translates יחלו. ּוֹמֵיְהוָה מִשְׁפַּט־אִישׁ: 27 תּוֹעֲבָת אֲדִיקִים אֵישׁ עָוֶל וְתוֹעֲבָת רָשָׁע יְשַׁר־דֶֶרֶדִ: פּ

דִּבְרֵי | אָגוּר בִּן־יָאֶה 30:1

יהַמַּשָׂא נְאָם הַגֶּבֶר לְאִיתִי אֵל לְאִיתִי אֵל וְאֵכֶל<sup>ז</sup>: 2 בִּי בַעַר אָנֹבִי

G 30:1-14; 31:10-31

G-Prov 30:1–14 follow G 24:22*e*. G 31:10–31 follow G 29:27. G verses are discussed along with M. See the note at 24:22*e*.

30:1 יקח M S ( (שם) G (אמו אנאם G (אם) איקח (שם) (graph הש) המשא נאם M (accents) הַמַּשָּׁא נְאָם M (syntax, accent) הַגָּבְר [ אַגָּבָר [ אַזָגָבָר M (syntax, accent) לאַיתִיאַל לָאָיתִיאַל לָאָיתִיאַל (איתי אל לאיתי אל איתי אל איתי אל איתי אל M; לאיתי אל לאיתי אל M; איתיאל לאיתי אל איתי אל M (bright distribution for the second seco

M 30:1 is enigmatic. Most translations are along the lines of KJV: "The words of Agur the son of Jakeh, even the prophecy: the man spake unto Ithiel, even unto Ithiel and Ucal." In this understanding, Agur is speaking to two sons, one of them mentioned twice. The name Ithiel is known from Neh 11:7; a name Ukhal is otherwise unattested. There are problems with this interpretation. If Ithiel is a name, why is it repeated? If these are personal names, we would expect the addition of "his sons." In other titles that state the audience of a wisdom instruction, the audience is always identified as "his son" or "his sons" and not by name alone. Also, the next verse starts with "<code>></code> (probably meaning "because"), which presumes an earlier sentence, not only a title compounded of personal names.

I translate: "(a) The words of Agur the son of Yaqeh, (b) the pronouncement, the oracle of the man: (c) I am weary, God, (d) I am weary, God, and have wasted away" ( $\neq$  ABP). The proposed translation is uncertain but requires only minor changes and makes sense. (The major disjunctive is now at  $\neg$  $\pi$ L) As emended, this sentence reverberates in the clause "before I die" in 30:7b, which is suggestive of approaching death, as in Gen 27:4 and 45:28. Agur's oracle consists of his last words. They are called a  $\Box$ , as are David's last words in 2 Sam 23:1. The psalmist of Ps 73, confessing ignorance (see 73:22), says that his flesh "is wasting away" (73:26, using  $\neg$  $\Box$ " $\neg$ , as in the emended text).

G: (a) Τοὺς ἐμοὺς λόγους, υἱέ, φοβήθητι (b) καὶ δεξάμενος αὐτοὺς μετανόει. (c) τάδε λέγει ὁ ἀνὴρ τοῖς πιστεύουσιν θεῷ, καὶ παύομαι. "(a) Fear

my words, son, (b) and having received them, repent! (c) These things says the man to those who have faith in God; and I cease." This radical paraphrase eliminates an attribution of authorship to someone other than Solomon; see the comment on 10:1.

G probably thought that the idea of repentance was implicit in המשא, which he associated with נשא. This verb sometimes means "forgive" or, in the passive, "be forgiven" (e.g., Isa 33:24; Ps 32:1). The precondition of forgiveness is repentance; hence G changes "be forgiven" to "repent," to make the human role in the process explicit.

G 30:1b recalls G 24:22*a*b:  $\delta \epsilon \chi \delta \mu \epsilon \nu o \varsigma \delta \epsilon \delta \delta \delta \epsilon \lambda \sigma \tau$  (lit. "and receiving it, he received it"). Indeed, G 30:1 recalls the entirety of G 24:22*a*. That verse is now followed by G 24:22*b*-*e*, but the latter is an independent unit and very likely a later insertion. Without it, 30:1 would have directly followed 24:22*a*:

24:22*a* a A son who keeps the word is free from destruction,

b and receiving it, he has (really) received it.

- 30:1 a Fear my words, son,
  - b and having received them, repent

V: (a) verba Congregantis filii Vomentis (b) visio quam locutus est vir cum quo est Deus (c) et qui Deo secum morante confortatus ait. "(a) The words of the Assembler, son of Vomiter. (b) the vision, which a man, with whom is God, said, (c) and who [sc. the man], having been strengthened by God himself, while delaying, said." Using midrashic techniques of creative rereading, V (tortuously) construes M 30:1ab as if it read, approximately, (a) קי"א is associated with איתי ("vomit"), and גאם is treated like RH יקה. אָל ("speak") and translated as a passive. Congregantis recalls 1 Kgs 8:1, מֵאֶישׁ וְלְאִ־בִינֵת אָדָם לְי: 3 וְלְאִ־לָמַדְתִּי חָכָמֲה וְדַעַת קְדֹשֵׁים אֵדֵע: 4 מִי עָלֶה־שָׁמַׁיִם | וַיֵּבִّד מִי אֱסַף־רֹּוּחַ | בְּחָפְנָיו מִי צֶרַר־מַׁיִם | בַּשִּׁמְלָה מֵי הֵקֵים כָּל־אַפְסֵי־אָָרָץ מַה־שְׁמִו וּמַה־שָׁם־בְּנוֹ כִּי תֵדֵע: 5 כָּל־אִמְרַת אֱלְוֹהַ צְרוּפֶה מְגֵן הוּא לַחֹסִים בְּוֹ: 6 אַל־תִּוֹסְפְּ עַל־ דְּבָרֵיו כֶּן־יוֹכִיחַ בְּדָ וְגִכִזְבְתָּ: פּ 7 שְׁתַּים שָׁאַלְתִי מֵאתָדָ

in which Solomon "assembled" the people (M יְקָהֵל, V *congregavit*). V 30:1c, however, can hardly be correlated with M's לאיתיאל ואכל. Num. Rab. 10:4 (88a) (similarly Tanḥuma Wa'era' [Buber] 8 [73a]) explains that Solomon was called יקה because "he disgorged" (הקיא) the words of Torah as if into a bowl. That same midrash also says that Solomon used to say: אָיִתִי אל ("God is with me").

30:2

G: (a) ἀφρονέστατος γάρ εἰμι πάντων ἀνθρώπων, (b) καὶ φρόνησις ἀνθρώπων οὐκ ἔστιν ἐν ἐμοί. "(a) For I am the most foolish of all men, (b) and intelligence of men is not in me." G rightly carries the force of the negative into the second stich.

### 30:3

 the translator's part, an exegetical metathesis. (On the concept, see comment on 11:27.) אל in the Hebrew would have made למדת' impossible, whereas the translator could take liberties with the grammar. The problem presented by M is that it has Solomon, the prototypical wise man, declaring that he has not learned wisdom. With God as the subject, the verse now has Solomon ascribing his wisdom to God himself, as in Wis Sol 8:17–22; 9:9–11.

S: ملعم جمع ملع ملع ملع ملع ملع معد (a) And I did not know wisdom (b) and I did not learn the knowledge of the holy beings [or 'saints']." S continues the force of the negative into the second stich, denying that the speaker has any knowledge, human or angelic. This is the usual modern interpretation and is distinct from G's.

### 30:4

(a) τίς ἀνέβη εἰς τὸν οὐρανὸν καὶ κατέβη; (b) τίς συνήγαγεν ἀνέμους ἐν κόλπω; (c) τίς συνέστρεψεν ὕδωρ ἐν ἱματίω; (d) τίς ἐκράτησεν πάντων τῶν άχρων τῆς γῆς; (e) τί ὄνομα αὐτῶ, ἢ τί ὄνομα τοῖς τέχνοις αὐτοῦ, ἵνα γνῶς [var: >  $i\nu\alpha$   $\gamma\nu\omega\zeta$  G<sup>BS</sup>]; "(a) Who has gone up to heaven and come down? (b) Who has gathered the winds in (his) bosom? (c) Who has gathered water in a garment? (d) Who has mastered all the heights of the earth? (e) What is his name, or what is the name of his children?--that you may know [var: > 'that you may know']?" G's "his bosom"—that is, the opening of his robe—is influenced by "in his garment" in 30:4c and does not show a different Hebrew. G 30:4d has ἐκράτησεν ("mastered") for הקים ("established"). Whereas M requires the answer "God," G's "who mastered...?" allows for the answer, "No one," which is the answer called for by the other questions in the series. In 30:4e, τοῖς τέχνοις αὐτοῦ =  $\Box$ or μΞ. ἕνα γνῶς is lacking in  $G^{BS}$ . It is possible that ἕνα γνῶς (>  $G^{BS}$ ) is secondary in G and its original absence reflects an absence of כי תדע in proto-M as well (thus, e.g., Clifford), but see ABP.

S prefixes  $\lambda$  in ambiguous form. If the verb is a perfect, the clause means, "He [God] said to me." If it is an imperative, the clause means "Tell me!" The latter would be a challenge, like God's challenge to Job (38:5a). After that, S = M.

30:7

G: (a) = M; (b) μὴ ἀφέλῃς μου χάριν πρὸ τοῦ ἀποθανεῖν με. "(b) Do not remove favor from me before I die." Since M appears to lack a direct object for **πατι** in the second stich (the object is actually present in the first), G supplies one by adding χάριν, referring to God's grace.

אַל־תִּמְנַע מִמָּנִי בְּטָרֶם אָמְוּת: 8 שָׁוְא | וְדְבַר־כָּזְׁב הַרְחֵׁק מִמָּנִי רֵאשׁ וָעֹשֶׁר אַל־תִּתֶזּ־לֵי הַטְרִימֵּנִי לֵחֶם חֻמְי: 9 כָּן אָשְׂבַּע | וְכִחַשְׁתִי וְאָמַרְתִי מֵי יְהֹוֶה וּפֵּן־אִוְרֵשׁ וְגָנֵבְתִי וְׁתָפַּשְׁתִי שֵׁם אֶלֹהֵי: פּ 10 אַל־תַּלְשֵׁן עֶבֶד אֶל־אַדֹנֵו פֶּן־יְקַלֶּלְדָ וְאָשְׁמְתָ: 11 דְּוֹר אָבִיו יְקַלֵּל וְאֶת־אָמוֹ לָא יְבָרֵדָ: 12 דְּוֹר טְהוֹר בְּעֵינֵיו וֹמִצֹאָתוֹ לָא רָחֵץ: 13 דְּוֹר מָה־רָמָוּ עֵינֵיו וְעַפְעַפִיו יִנָּשֵׂאוּ: 14 דְּוֹר ו חֲרָבִוֹת שִׁנִיו וְמַאֲכָלוֹת מְתַלְעָׁתִיו לֶאֶכָל עֲנִיִים מֵאֶרָץ וְאָבֵיוֹנִים מֵאָבֶים: פּ

### 30:8

G (a) = M; (b) ≈ M, with word metathesis; (c) σύνταξον δέ μοι τὰ δέοντα καὶ τὰ αὐτάρκη. "(c) but prescribe for me what is necessary and sufficient." G 30:8c is an accurate paraphrase of M.

 $S \approx M$ . לחם חקי is translated כמשל, ("a living sufficient for me"), an idea probably taken from G.

# יחוה ( איחוה M S (כיישי G (לא G (graph $i \rightarrow n, i \rightarrow t$ ) איחוה ( $i \rightarrow n, i \rightarrow t$ )

G's דוֹג με אָסְמָ ("who sees me?") = מי יָחֲזָה (with "me" inferred), for M's מי יהוה (Similarly in Isa 2:2 בעקמאבל for יחזה where M has מי יהוה וt is possible that M's מי יהוה is the secondary reading, influenced by Exod 5:2. The claim that God does not see the sinner is ascribed to the wicked in Isa 29:15; Ps 64:6; and Job 22:14.

# 30:10 אדנו M<sup>K</sup> ] אַדֹנָיו M<sup>Q</sup> (orth)

G: (a) μή παραδώς οἰκέτην εἰς χεῖρας δεσπότου. (b) μήποτε καταράσηταί σε καὶ ἀφανισθῆς. "(a) Do not deliver a slave to the hands of (his) master, (b) lest he curse you and you be annihilated." De Waard suggests that μή παραδώς is an exegetical metathesis that treats תלשן as if it were תשלם ( תַּשָׁלִים ). This, he believes, belongs to G's "tendency toward intensification and radicalization" (1993, 258). In fact, G is influenced by the command in G-Deut 23:16, which uses παραδώσεις for "עֹם" ("turn over to"). In the present verse, ἀφανισθῆς = אשמת by association with שמ"ם (similarly Hos 5:15; 10:2; 14:1 [G-stem]; and Joel 1:18 [N-stem]). This choice is prompted both by the phonetic similarity of the two words

and the reasoning that the consequence of a malediction is not guilt but punishment.

## 30:11

G: (a) ἔκγονον κακὸν πατέρα καταρᾶται. (b) = M. "(a) Bad offspring curse the father." By translating TIT as ἔκγονον κακόν in verses 11, 12, 13, and 14 (rather than γενεά as in 27:24 and usually), G condemns individuals, as is usual in Proverbs, and not an entire generation. G adds "bad" in these verses to clarify what kind of offspring does these things.

# 30:12

30:14 אביונים M S (כמשביא) ] ארטינים (דאביונים) אביונים (ידע מטידש) (orth or gram)

In G 30:14d, גמן דסטֹק הציאדמק מטֹדῶν ("and their wretched [pl]") represents ואביונים, which is an orthographic variant of M's ואביונים. Barr (1989, 45–47) gives examples of the masculine plural written without the ', a phenomenon he describes as extremely rare but not a sign of antiquity. G construes the word as וָאֶרְיוֹנָם, though the suffixed form does not exist elsewhere.

G continues with 24:23–34.

# 30:15

G: (a) Τῆ βδέλλη τρεῖς θυγατέρες ἦσαν ἀγαπήσει ἀγαπώμεναι, (b) καὶ αἱ τρεῖς αὖται οὐκ ἐνεπίμπλασαν αὐτήν, (c) καὶ ἡ τετάρτη οὐκ ἠρκέσθη εἰπεῖν Ἱκανόν. "(a) To the leech there were three daughters, truly beloved. (b) And these three did not sate her, (c) and the fourth was not satisfied, so as to say 'Enough!" G reads 30:15 as a single saying (though 30:15a is actually an independent monostich) and rewrites it as a prose parable. ἀγαπήσει ἀγαπώμεναι represents a Hebrew paronomastic infinitive-absolute construction, which was suggested to the translator by the repetition of Π = Π. Possibly G had Π = Π (as if from Π = "Ω") (de Lagarde), but given the paraphrastic treatment (or misunderstanding)

הַב שָׁלִוֹשׁ הֲנָּה לָא תִשְׂבֵּעְנָה אַׁרְבַּע לאֹ־אָמְקְרוּ הְוֹזָ: 16 שְׁאוֹל וְעָּצֶׁר רְחַם אֶֶרֶץ לאֹ־שָׂבְעָה מֵּיִם וְאֵשׁ לאֹ־אָמְרָה הְוֹזָ: 17 עַיָז | הַּלְעַג לְאָב וְחָבֶוּז <sup>ו</sup>לְיִקְהַתִ-זאֵם יִקְרוּהָ עִׂרְבֵי־נַחַל וְיאַכְלְוּהָ בְנֵי־ גַשָּׁר: פּ 18 שִׁלֹשֶׁה הֵמָּה נִפְלְאוּ מִמֶּנּוּ וּוְאַרְבַּע' לָא יְדַעְתִּים: 19 דֶרֶך הַנָּשֶׁר | בַּשָׁמַים דֶּרֶך נָחָשׁ עַלֵּי צְוּר דֶרֶך-אֶנְיָה בְלָב־יָם וְדֵרֶך גֶּבֶר בְּעַלְמֵה: 20 בֵּז | דֶרֶך אָשָׁה מְנָאָפָת אֵכְלָה

of 30:15–16, G could as well be associating אה" with אה". M's "two" was changed to "three" to fuse 30:15aa to the number series in 30:15bc. G translates ארבע as an ordinal ( $\dot{\eta}$  דנדמֹסָדָק), as in verses 18, 21, and 29 (similarly V), setting the last item apart from the first three items and making it the climax of the listing.

S has محتج محتج ("three beloved daughters") = G.

G: (a) ἄδης καὶ ἔρως γυναικὸς (b) καὶ τάρταρος καὶ γῆ οὐκ ἐμπιπλαμένη ὕδατος (c) καὶ ὕδωρ καὶ πῦρ οὐ μὴ εἴπωσιν Ἀρκεῖ. "(a) Hades and the love of a woman (b) and Tartarus, and the earth, which is not filled with water, (c) and water and fire do not say, 'It is enough.'" G makes the four insatiables into six. ἔρως appears elsewhere in G only in Prov 7:18, where it also means sexual passion. G's καὶ ἔρως γυναικός departs considerably from M's משניר רחם ("the closing of the womb") to introduce a moral warning against lustful women. It associates שיאור Aramaic "(love"), as in G 28:13 and G<sup>A</sup>-Judg 5:30. The second "water" in G has no exegetical function and is in a different syntactic position than the first. Hence ὕδατος καὶ ὕδωρ is not a double translation but a dittography in the source text. "Sheol," however, is given a double translation, Hades and Tartarus.

 $S \approx M$ , but איזא דערבעה ("and the shutting up of her womb") must refer to the "womb" (plural in singular sense, as in, e.g., Gen 49:25; Exod 13:12; 34:19) of Sheol (fem). The image recalls the metaphor of the earth as the mother of all; see Ps 139:13 with 15; Sir 40:4; Ezra 5:48.

# 30:17 לִיִקְהָת־ M<sup>Erfurt 1</sup> לִיקָהָת־ M<sup>ALY</sup> (vocal)

The major M MSS have the anomalous vocalization לְיָקָהַת. This is preferred by Ben Naphtali, though it seems morphologically impossible.

The minority vocalization is לְיָקְהָת (Cod. Erfurt 1); see Baer 1880, 52. It is very unlikely that this maintains a connection to the original Masoretic form; rather, it looks like a normalization based on Gen 49:10. But even if it is secondary, it is correct. The versions (like modern translators) guess at the meaning of the crux קהת. G renders it as  $\gamma \tilde{\eta} \rho \alpha \varsigma$  ("old age"), taking a cue from 23:22, where also one is warned not to despise (דוו) a parent. S (סערסאליי) follows G, and T (שרסאליי) follows S in understanding this word as meaning "old age." V shows originality in rendering it *partum* ("childbearing").

# **30:18 אַרְבָּעָ**ה [ M<sup>K</sup> ] אַרְבָּעָה (norm, assim)

The *qere* adjusts וארבע to the masculine form, as used in the parallel שלשה.

G: (a) τρία δέ ἐστιν ἀδύνατά μοι νοῆσαι, (b) καὶ τὸ τέταρτον οὐκ ἐπιγινώσκω. "(a) There are three things that are too difficult for me to understand, (b) and the fourth I do not observe." G uses the cardinal τρία and the ordinal τὸ τέταρτον (as in 30:15c [fem]). This sets the fourth item in the list—"the ways of a man in his youth"—in a class by itself. The fourth behavior is what the epigram was leading up to (see the next comment). It is the most puzzling "way" of them all.

S uses the cardinal "fourth" = G.

### 30:19

G: (a) ἴχνη ἀετοῦ πετομένου (b) καὶ ὁδοὺς ὄφεως ἐπὶ πέτρας (c) καὶ τρίβους νηὸς ποντοπορούσης (d) καὶ ὁδοὺς ἀνδρὸς ἐν νεότητι. "(a) The tracks of a flying eagle, (b) and the ways of a snake on a rock, (c) and the paths of a ship travelling the sea, (d) and the ways of a man in youth." G modifies the syntax of the lines of this verse and varies the words for "ways" in 30:19 and 20a. (M repeats דרך; on the meaning of the repeated "way," see ABP.) G translates שלמה ("with a maid") as ἐν νεότητι ("in youth"), probably for propriety's sake; similarly V (*in adulescentula*), S (σωματικής), and T (Ξυάσκης). BAP understands "a flying eagle" (for M's "an eagle in the sky") to reflect 9:12a and "the ways of a man in youth" to echo 9:12b, which condemns walking in wayward paths.

30:20

G: (a) τοιαύτη όδὸς γυναικὸς μοιχαλίδος, (b) η, ὅταν πράξη, ἀπονιψαμένη οὐδέν φησιν πεπραχέναι ἄτοπον. "(a) Such is the way of an adulterous woman, (b) who, when she has done (the deed), after washing says that she has done nothing improper." G explicates the metaphor of eating

וּמְחַתָּה פֵּיהָ וְאָמְרָה לְאֹ־פְּעֵלְתִי אֲזָן: פּ 12 תַּחַת שֵׁלוֹשׁ הְנְאָזָה אֶֶרֶץ וְתַחַת אַרְבַּע לֹאִ־תוּכֵל שְׂאֵת: 22 תַּחַת־עֵּבָד פֵּי יִמְלֵוֹד וְנְבָל פֵּי יִשְׂבַע־לֶחֶם: 23 תַּחַת שֵׁנוּאָה פֵּי תִבְּעֵל וְשִׁפְחָה פְּי תִירַשׁ גְּבִרְתֵּה: פּ 24 אַרְבָּעָה הֵם קְטַנֵּי־אֶרֶץ וְהֵמָה חַכִּמִים מְחֻבָּמִים: 25 הַנְּמָלִים עֵם לֹא־עָז וַיָּכָינוּ בַקַיּץ לַחְמֶם: 10 שֵׁפַנִּים מְחֻבָּמִים: 25 הַנְּמָלִים עֵם לֹא־עָז וַיָּכָינוּ בַקַיץ לַחְמֶם: 26 שֵׁפַנִים מְחַבָּמִים: 25 הַנְּמָלִים עַם לֹא־עָז וַיָּכָינוּ בַקַיץ לַחְמֶם: 26 שֵׁפַנִים מְחַבָּמִים: 27 הַנְּמָלִים עַם לֹא־עָצוּם וַיָּשִׁימוּ בַפָּלַע בֵּיתָם: 27 מֵלָד אֵין לָאַרְבֶּה וַיֵּצֵא חֹצֵץ כָּלוֹ: 28 שֵׁמָמִית בְּיָדַיִם תְּתַפֵּשׁ וְהִיא בְּהַיְכְנִי לָאַרְבֶּה וַיֵּצֵא חַצֵּץ כַּלוֹ: 28 שֵׁמָמִית בְּיָדַיִם תְּתַפָּשׁ וְהִיא בְּהַיְכָלִי מֶלֶדָ: פּ פַּנ שְׁלֹשֶׁה הֵמָה מֵיטִיבִי צָעַד וְאַרְבָּעָה מֵיטָבִי

as committing adultery and the accompanying gesture of wiping the mouth as actual postcoital washing.

### 30:21

G uses ordinals, as in 30:15.

#### 30:23

G: (a) καὶ οἰκέτις ἐἀν ἐκβάλῃ τὴν ἑαυτῆς κυρίαν, (b) καὶ μισητὴ γυνὴ έαν τύχη ανδρός αγαθοῦ. "(a) And (the earth shakes) if a maidservant casts out her own mistress, (b) and if a hateful woman happens to get a good man." G reverses the order of the stichoi, perhaps to tighten the parallelism between oixétnç in 30:22 and oixétiç in 30:23 (BAP). For G, a male slave coming to rule (30:22a) means deposing (ἐκβάλλειν) his master, like a slave woman displacing her mistress. G parses תירש as תֹרִישׁ (cf. Exod 34:24), hence ἐκβάλη, or understands the G-stem to have this sense. This may be an actual case of exegetical metathesis (see index). Alternatively, the translator may simply not be parsing the forms precisely. Whereas M says just that the hateful woman "gets married," G says "happens to get a good man." The latter event, it seems, is just an accident ( $\tau \dot{\nu} \chi \eta$ ), not the expected result of choosing a wife, and, being unpleasant, it disturbs the world only if it befalls a good man, who does not deserve it. G sometimes tries to eliminate implications of absence of retribution that might appear to be allowed for by M; see \$3.1.5.2 and the comments on 10:6 and 13:23.

## 30:24

G: σοφώτερα τῶν σοφῶν ("wisest of the wise") vocalizes אַכְמִים מֵחֲכָמִים מַחֲכָמִים. Similarly S איביד אי איבייד.

# 30:28

G: (a) καλ καλαβώτης χερσίν έρειδόμενος καὶ εὐάλωτος ὢν (b) κατοικεῖ ἐν όχυρώμασιν βασιλέως. "(a) The gecko, though holding himself up by (his) hands and easily caught, (b) dwells in fortresses of the king." ἐρειδόμενος καὶ εὐάλωτος ὢν is a double translation attempting to secure the relevant senses of ৺ν is a double translation attempting to secure the relevant senses of ψοματιν, both of which assume a passive-reflexive N-stem. (ἐρειδόμενος apparently means that the gecko is impeded by the difficulty of holding on to vertical walls.) G has the gecko dwelling "in fortresses" (ὀχυρώμασιν), continuing the martial imagery from 30:27 (BAP).

# 30:29

 $G \approx M$ , but using the ordinal "fourth"; cf. 30:15.

# 30:30-31

G structures the epigram in 30:30–31 as a series of nouns + relative clauses by using relative pronouns or participles.

# 30:31 אל־קום M<sup>Or MSS</sup> V (qui resistat ei) ] אַלקום M<sup>ALOc</sup> (div)

G: (a) καὶ ἀλέκτωρ ἐμπεριπατῶν θηλείαις εὖψυχος (b) καὶ τράγος ήγούμενος αἰπολίου (c) καὶ βασιλεὺς δημηγορῶν ἐν ἔθνει. "(a) And a cock, who walks about boldly among hens, (b) and a he-goat, who leads the herd, (c) and a king, who delivers a harangue among nations." The translator pictures creatures who proudly lead their kind. He construes the obscure index is if it were אלקום עמו ("and a king standing before [lit. 'to'] his people"), though it is unlikely that this was actually in his source. The Greeks viewed the cock as a "proud and pugnacious bird" (Gerleman 1956, 31).

V: (a) (b) = M; (c) *nec est rex qui resistat ei*. "(c) and a king who cannot be resisted." V construes the Hebrew loosely as וּמֶלְדָ אַל קוֹם עָמוֹ. V supports the division אל קום.

S: (a) (b) = M; (c) محلکہ محطل صلا منظ مختکہ. "(c) and a king who speaks among the peoples." S 30:31c is based on G.

The Oriental אל־קום (followed by several MSS KR) makes most sense of the crux. It means "let one not stand with him," i.e., antagonistically; see ABP. This is the sense conveyed by V. B. Sanh. 82b identifies "a king with whom one cannot stand" as God.

#### 30:32

This verse is best translated, "(a) If you have acted vilely [or 'foolishly'] by exalting yourself, (b) or if you have schemed, put your hand on your mouth!" The versions had difficulty with this verse and strain to make sense of it in various ways. They do not evidence variants.

G: (a) ἐἀν πρόῃ σεαυτὸν εἰς εὐφροσύνην (b) καὶ ἐκτείνῃς τὴν χεῖρά σου μετὰ μάχης, ἀτιμασθήσῃ. "(a) If you give yourself to pleasure, (b) and stretch out your hand in conflict, you will be disgraced." This is close to a guess at the meaning of M. In G, the proverb introduces the theme of conflict, which continues in 30:33.

V: (a) *et qui stultus apparuit postquam elatus est in sublime* (b) *si enim intellexisset ori inposuisset manum*. "(a) And one who appears a fool after he was exalted, (b) had he indeed understood, would place a hand on (his) mouth." In other words, one who becomes a fool after reaching a high station should know enough to keep quiet.

S:  $r \to \infty$   $s \to \infty$  s

T: לא תיתרו<מ>ם דלא תיטפש ולא תושיט אידך לפומך ( $T^L$ , cf.  $T^Z$ ). "(a) Do not exalt yourself that you not be made stupid, (b) and do not extend your hand to your mouth." T uses the words of M in a rather random fashion.

### 31:1

G: (a) Oi ἐμοὶ λόγοι εἴρηνται ὑπὸ θεοῦ, (b) βασιλέως χρηματισμός, ὃν ἐπαίδευσεν ἡ μήτηρ αὐτοῦ. "(a) My words have been spoken by God. (b) An oracle of the king which his mother taught him." G reads דברי דברי לְמוֹ אֵל ("words of Lemuel") as לְמוֹאל This unique rendering is an exegetical maneuver to avoid ascribing this poem to a sage other than Solomon. (α' has Λαμμουν, σ' has Ιαμουηλ, and θ' has Ρεβουηλ. These are probably inner-Greek distortions and phonetic changes.) The word order of מלך משא is reversed. (For other examples of G overriding M's word order, see the comment on 11:21.) As in 30:1, G has Solomon claiming a divine source for his oracle, but here the teaching is conveyed by his mother.

 $V \approx M$ , with *visio* ("prophecy") for משא.

S: אלאסת, גרבאאר שלבא גריאן גוגאס ארכה ארבא מארט (a) The words of Muel the king (and) prophet, (b) which his mother taught him, and she said to him." S parses the first למואל as the authorial למואל. For אשא, which can mean "prophecy," S has, by way of metonymy, גרבא ("prophet").

T expresses two meanings of משא by a double translation: נביותא נביותא ("prophecy and instruction").

31:2

G: (a) τί, τέχνον, τηρήσεις; τί; ῥήσεις θεοῦ. (b) πρωτογενές, σοὶ λέγω, υἰέ. (c) τί, τέχνον ἐμῆς χοιλίας; (d) τί, τέχνον ἐμῶν εὐχῶν; "(a) What, child, will you keep? What? The words of God. (b) O (my) firstborn, I speak to you, (my) son. (c) What, child of my womb? (d) What, child of my vows?" Alternatively: "(a) What, child, will you keep? What? (b) The words of God, O (my) firstborn, I speak to you, (my) son; etc." (cf. BAP). (In either case, the syntax is rather choppy.) In 31:2a, the translator completes and answers the repeated rhetorical question, thereby introducing an idea entirely absent from M, namely that the son must keep *God*'s words, which, G implies, the mother is teaching, in essence if not verbally. We have here the bold notion of the mother as Torah mediator and teacher. τηρήσεις τί ῥήσεις is a pun of the translator (BHQ).

 $V \approx M$  but rendering Aramaic ' $\Box$  as *dilecte* ("beloved"), associating it with  $\Box$ " ("to purify," "choose"). The same etymology is used in Lev. Rab. 12:8 (19a) (C. Gordon 1930, 411).

# 31:3 (לְמָחוֹת ( <br/>אַ לַמְחוֹת ( אַ מָחוֹת ) א V (ad delendos) (vocal)

We should vocalize למחות ("to those [fem] who destroy kings"), parallel to נשים (GKC \$53*q*, Delitzsch, Toy, BHS, ABP).

G: (a) μη δῶς γυναιξί σὸν πλοῦτον (b) καὶ τὸν σὸν νοῦν καὶ βίον εἰς ὑστεροβουλίαν. "(a) Do not give your wealth to women, (b) nor your mind and life to regret." G 31:3b is a guess at the obscure Hebrew. GELS glosses the *hapax* ὑστεροβουλίαν as "deliberation after the facts, remorse, wisdom after the events, hindsight." θ' gives the Hebrew a similar sense: וּדְרָכֶּיִדְּ וּלְמֹחִוֹת מְלָכֵיז: 4 אַל לַמְלָכִים | לְמוֹאֵל אַל לַמְלָכִים שְׁתוֹ־יָיִז וּלְרוֹזְנִים וּאֵי שֵׁכָר: 5 פֶּן־יַשְׁתָּה וְיִשְׁפַּח מְחָקֶק וְישַׁנָּה דִּיז כְּל־בְּנֵי־עְׂנִי: 6 תְּנוּ־שֵׁבְר לְאוֹבֵד וְיַיִז לְמָרֵי נְפָשׁ: 7 יִשְׁתָּה וְישִׁנָּח רִישֶׁוֹ וַׁעַמְלוֹ לָא יִזְכָּר־עְוֹד: 8 פְּתַח־פִּידְ לְאָלֵם אֶל־דִיז כָּל־בְּנֵי חֲלְוֹף:

ἐις μεταμέλειαν ("to repentance," "to remorse"). M's מלכין (which has an Aramaic ending) was, not unnaturally, associated with Aramaic ("counsel," hence "thought," "afterthoughts," and the like), and allows ύστεροβουλίαν to stand in for the obscure למחות as well.

S translates גמרות מלכין as למחות מלכין ("as food of kings"). This is a guess whose function is to match the warning against excessive drinking (in 31:5) with one against (excessive) eating.

T translates this difficult phrase as לבנת מלכין ("to daughters of kings"). The Targumist probably has Solomon's wives, including the daughter of Pharaoh, in mind. In any case, S and T are not deriving their renderings from לחם, or לְמֵחוֹת, or לְמֵחוֹת, or לְמֵחוֹת, or לְמֵחוֹת, or אַמְחוֹת, or phissing in Delitzsch).

V's ad delendos reges ("ought to obliterate kings") parses למחות as an infinitive of מח"ה "wipe away." This, I think, is M's intention, and the patah marks the H-stem (for which see Jer 18:23 and Neh 13:14).

31:4 אָי ה ( (µµ́) ) או M<sup>Q</sup> V (ubi); א M<sup>K</sup> S (<br/>o) (M<sup>Q</sup> V: vocal; M<sup>K</sup> S: graph י $\rightarrow$  <br/>)

The *ketiv* או ("or") is impossible in the middle of the clause. The *qere* means "where?" which does not make good sense either. (Various proposals are surveyed in ABP.) We should use the *qere* consonants but vocalize  $\aleph$ , a negative particle = G. This particle negates nouns in 1 Sam 4:21 (in a personal name); Job 22:30; Lachish Ost. 2.6 ( $\aleph$  = "no thing"); and also in Phoenician. Hence: (c) "nor for governors beer."

G: (a) μετὰ βουλῆς πάντα ποίει, (b) μετὰ βουλῆς οἰνοπότει. (c) οἱ δυνάσται θυμώδεις εἰσίν, (d) οἶνον δὲ μὴ πινέτωσαν. "(a) Do all things with counsel. (b) Drink wine with counsel. (c) The princes are hot-tempered—(d) let them not drink wine." G 31:4a replaces M 31:4a with a precept relevant to context, probably in order to avoid mentioning Lemuel (see the comment on 10:1). In G 31:4ab, Hebrew מלכים is twice derived from Aramaic מלכים ("counsel"). G 31:4cd gets the gist of M 31:4bc. In 31:4d, μή = 'ặ, representing the consonants of the *qere*. V: (a) noli regibus o Lamuhel noli regibus dare vinum (b) quia nullum secretum est ubi regnat ebrietas. "(a) Not to kings, O Lamuel, give not wine to kings, (b) because there is no secret where drunkenness reigns." V supplies a different reason for kings to abstain from wine. V associates אושר רוזנים ("secrets"), as does b. Sanh. 70b (C. Gordon 1930, 391).

### 31:5

G substitutes דאָָק ססּלומָק for מחקק. In M's vocalization, מְחָקָק means "that which is inscribed," i.e., the statute. The translator apparently associated the consonantal form with the more frequent מְחֹקֵק, which means "lawgiver." Since this would be awkward as the direct object of "forget," G replaced it with the abstract and contextually appropriate "wisdom."

V's iudiciorum ("judgments") is close to M.

S's גְּחְהֵק ("the lawgiver") = מְחְהֵק ("your senses" or "your conduct") provides a broad term to fit the context.

### 31:6

G's τοῖς ἐν λύπαις ("to those who are in misery") arises from the consideration that there is no value in giving strong drink to those who are perishing. It may, however, help dull the pain of the miserable. The other versions too use words for misery: V *maerentibus*; S . G is the source for them all. Pinkuss proposes that they read לאביל, but // interchange is unlikely.

S: האנתכי (a) Beer is given to mourners, (b) and wine to those bitter of soul." The passive "is given" (instead of the imperative) makes the verse into a description of a common practice—one apparently associated with mourning—rather than a recommendation to give alcohol to the poor, which the translator might have found questionable.

#### 31:8

G: (a) άνοιγε σὸν στόμα λόγω θεοῦ (b) καὶ κρῖνε πάντας ὑγιῶς. "(a) Open your mouth with the word of God (b) and judge everyone soundly." The

9 פְּתַח־פִּיךּ שְׁפָט־צֶעֶדֶק וְדִיז עָנִי וְאָבְיְוֹז: 
10 אֵשֶׁת־חֵיל
מִי יִמְצֵא וְרָחָק מִפְּנִינִים מִכְרֵה: 11 בָּטַח בָּה לֵב בַּעְלֶה וְשָׁלָל לֵא מִי יִמְצֵא וְרָחָק מִפְּנִינִים מִכְרֵה: 11 בָּטַח בָּה לֵב בַּעְלֶה וְשָׁלָל לָא יַחִיחָסָר: 12 גְּמַלַתְהוּ טִוֹב וְלֹא־רֶע בֹּל יְמֵי 'חַ״ֶיָה'': 13 גְּרְשָׁה צֵמֶר

king's duty is to pronounce God's teachings and judgments. גע אלם אלים associates אלאלים with אָלים "God." The translator actively guides the verse's focus to God's Torah. This move recalls G 31:1–2, where the royal teachings are identified with God's. These are what the king should speak in judgment. שני הלוף is puzzling.

V, correctly: aperi os tuum muto ("open your mouth for the dumb").

S: כואר בואר באר הסבעל הואר שליש שליש (a) Open your mouth in a word of truth (b) and judge all wicked sons." The logic behind the rendering of the *hapax* הלוף המביעה מינגעל as "wicked" is that the wicked are what one naturally judges. BAP points to the expression by ("to judge soundly") in Plato's *Republic* 409a, but this seems like a rather abstruse reference and in any case does not explain the idiom's use here.

T: (a) דיז מסטו דיז (a) Open your mouth to/for those who do not pervert justice." It is hard to see this rendering as even "inspired" by S (BHQ). It is just an attempt to give moralizing context to by S (BHQ). It is just an attempt to give moralizing context to the versions struggle to make sense of לאלם. The versions struggle to make sense of the king to speak "to the dumb," since the dumb would have no greater need to be spoken to than anyone else. It is unlikely that G, S, and T have Aramaic למלתא" (BHQ). The expected form is למלא

After 31:9, G continues with 25:1–29:27.

### 31:10-31

Prov 31:10–31 is an encomium on the Woman of Strength. G's understanding of this poem will be described in the following notes. The translator emphasizes the benefits the woman brings her husband, and her real achievement is not her own peace of mind but his (see especially 31:21).

Starting with 31:13, G uses mostly aorist forms mixed with presenttense ones in a way that does not allow for temporal distinctions. With respect to indicative forms, G-Proverbs is fairly consistent in using aorist forms for the Hebrew perfect and *wayyiqtol* and present and

future forms for the Hebrew imperfect. The aorists in G 31:10–31 give the poem the sound of a eulogy, though this is not carried through consistently. The translator will also use a participial construction for a finite verb.

### 31:10

G: (a) Γυναϊκα ἀνδρείαν τίς εὑρήσει; (b) τιμιωτέρα δέ ἐστιν λίθων πολυτελῶν ἡ τοιαύτη. "(a) A capable woman who can find? (b) And her sort is more valuable than precious stones."

G describes the woman as  $\dot{\alpha}\nu\delta\rho\epsilon(\alpha\nu)$ , a term used frequently to denote vigor and efficacy. G cautiously avoids speaking of the woman's "price," mentioned figuratively in M, lest this be misunderstood in an indelicate fashion.

### 31:11

G: (a) θαρσεῖ ἐπ' αὐτῇ ἡ καρδία τοῦ ἀνδρὸς αὐτῆς, (b) ἡ τοιαὑτη καλῶν σκύλων οὐκ ἀπορήσει. "(a) The heart of her husband finds courage in her. (b) Such a one [fem] does not lack good booty." G's use of θαρσεῖ in 31:11a implies that the woman gives her husband courage and confidence. In conjunction with 31:11a, G 31:11b implies that she is rewarded for doing so. G maintains the military connotations of M's לל "booty") while adding a cautionary "good." By making the woman the subject in 31:11b (in spite of the masculine gender of the Hebrew verb), G imbues her with an aura of power even greater than she has in M and enhances the connotations of valor evoked by ἀνδρείαν in 31:10.

S clarifies the metaphorical אשלל as שלל ("stores").

# 31:12 חַיָּיָה M<sup>AY</sup> ] חַיָּיה M<sup>L</sup> (vocal err)

G: (a) ἐνεργεῖ γὰρ τῷ ἀνδρὶ ἀγαθὰ (b) πάντα τὸν βίον. "(a) For she produces good for (her) husband (b) all (her) life." G omits the possessive pronouns (unnecessary in Greek) and eliminates "and not evil" as superfluous.

31:13 בחפץ M V (consilio ... suarum) ] איר א בחפץ S (איר א נכחפץ (graph ב → ב)

G μηρυομένη ("wind off thread") is, as BHQ says, a more specific term than דרשה for the task of weaving. בחפץ is translated εὔχρηστον ("use-

וּפִּשְׁתֵּים וֹהַעַשׁ בְּחַפָּץ פַפֶּיהָ: 14 הֲיְתָה כְּאֲנִיּוֹת סוֹחֵר מִׁמֶּרְחָׁק תְּבִיא לַחְמְה: 15 וַתָּקָם | בְּעוֹד לַיְלָה וַתִּתַּן טֶֶרֶף לְבֵיתָה וְׁחֹק לְנַעֲרֹתֶיהָ: 16 זְמְמָה שֵׁדֶה וַתִּקּחֵהוּ מִפְּרִי כַׁפֶּיה וּנָטְעָה בָּכָיוֹ 17 חֲגְרֵה בְעוֹז מְתְנֵיה וֹתְּאַמֵּץ זְרֹעוֹתֵיהָ: 18 טֵעַמָה בִּי־טָוֹב סַחְרֵה לְאִ־יִרְבֶּה וּבַלַיִל נֵרָה: 19 יָדָיהָ שִׁלְחַה בַבִּישׁוֹר וְכַפִּיה תְּמְכוּ בְּלָי מַשְׁלָג בִּי כָל־בֵּיתָה לְבֵעֵי וְיָדָיהָ שִׁלְחָה לֵאָבְיוֹז: 12 לֹא־תִירָא לְבֵיתָה מִשֶׁלֶג בִּי כָל־בֵּיתָה לְבֵשׁ שְׁנִים: 22 מַרְבַדָּים עֲשְׁתָה־לֵה שֵׁשׁ

fully"), less for alliteration with  $\tau \alpha \tilde{i} \zeta \chi \epsilon \rho \sigma i \nu$  (contrary to BHQ) than to make sense of a word that usually means "with pleasure" or "willingly." Though the woman's zest in her labors is indeed one of the virtues for which she is praised (her attitudes are antithetical to the sluggard's), the practical-minded translator finds her usefulness more significant.

V *consilio manuum suarum* ("by the counsel of her hands") understands שמי as what she wishes to do, hence her counsel or planning.

S: (a) = M; (b) איר אינגעה אינגעה (b) and the work of her hands is according to her wish." S witnesses to  $\Box \Pi \Box$  and reshapes the sentence accordingly. T = M.

### 31:15

G understands חק as "command," "order" and gives it a related sense:  $\xi \rho \gamma \alpha$  ("work assignments"). S  $\prec = G$ .

### נטע M<sup>Q</sup> ] נטע M<sup>K</sup> (logic)

It is doubtful that the third feminine singular perfect would have intentionally been written גטע. (The writing without the *heh* occurs occasionally with verbs III-h, but these represent the archaic –*āt* ending.) In any case, גטעה is correct. BHS proposes גָטע, but the infinitive absolute is not used elsewhere in the poem. I suggest that a scribe changed נטעה to גטעה with the passive גָטַע in mind, intending to say, "a vineyard was planted," for it may have seemed illogical that the lady of the manor would have literally done the planting herself. This is a form of hypercorrection.

#### 31:17

G adds εἰς ἔργον ("for work") at the end to make it clear that the meta-

phors of girding loins and strengthening hands refer to preparations for work, not war.

# **31:18 בַלֵּילָה** [ M<sup>K</sup> בליל M<sup>Q</sup> (norm)

The gere is a normalization of the archaic and rare בליל.

## 31:19

The versions all derive כשור ("spindle") from כשור "be useful" or "successful": G לאוֹ דע סטעסליסטדע ("things that are useful");  $\alpha'$ ,  $\sigma'$ ,  $\theta'$ מיאסקבוֹם ("capable"); V fortia ("strong things"); S כושרא ("skill"); T כושרא ("skill").

# 31:20

G: (a)  $\chi \epsilon \tilde{\rho} \rho \alpha \delta \epsilon$  autig dinvolzev πένητι, (b)  $\kappa \rho \rho \pi \delta v$  de έξέτεινεν πτω $\chi \tilde{\phi}$ . "(a) Her hands she opened to the poor, (b) and she extended fruit to the poor." G has the woman extending *fruit* to the poor, rather than "her hands," as in M. G makes it clear just what was in her hands.

# 31:21

G: (a) οὐ φροντίζει τῶν ἐν οἴκῷ ὁ ἀνὴρ αὐτῆς, ὅταν που χρονίζῃ, (b) πάντες γὰρ οἱ παρ' αὐτῆς ἐνδιδύσκονται. "(a) Her husband does not worry about those who are in his house, whenever he tarries (away from home), (b) because those who are with her are well clothed." G makes the husband the one who need not worry (similarly V). G further eliminates snow as a cause of concern, since this would be irrelevant to Alexandrian Jews.

# 31:22

G: (a) δισσὰς χλαίνας ἐποίησεν τῷ ἀνδρὶ αὐτῆς, (b) ἐκ δὲ βύσσου καὶ πορφύρας ἑαυτῆ ἐνδύματα. "(a) She made for her husband a two-ply mantle, (b) and from linen and purple (she made) clothing for herself." G understands שנים (from the end of M 31:21) as שָׁנִים ("two," hence "double"); similarly V *duplicibus*, but reading it as part of 31:21. G parses as "her husband." The translator is again emphasizing that the woman serves her husband's needs.

וְאַרְגָּמָן לְבוּשֵׁה: 23 נוֹדֵע בַּשְׁעָרֵים בַּעָלֵה בְּשָׁבְתוֹ עִם־זִקְנִי־אֶרָץ: 24 סָדֵין עֲשְׁתָה וַתִּמְכֵּר וֹחֲגוֹר נָתְנָה לַפְנַעֲנִי: <sup>2</sup>61 פֵּיה פְּתְחָה בְחָכְמֵה וְתוֹרַת־חֶסֶד עַל־לְשׁוֹנֶה: <sup>2</sup>51 עוֹ־וְהָדֶר לְבוּשֶׁה וֹתִשְׂחַק לְיוֹם אַחֲרְוֹן: 27 צֵוֹפִיָה הַלִיכִוֹת בֵּיתֶה וְלֶחֶם עַצְלוּת לָא תֹאבֵל: 28 קַמוּ בֵּנֶיה וַיְאַשֶׁרָוּה בַּעָלָה וַיְהַלְלֵה: 29 רַבְּוֹת בְּנוֹת עֲשׁוּ חֵיָל

#### 31:23

G: (a) περίβλεπτος δὲ γίνεται ἐν πύλαις ὁ ἀνὴρ αὐτῆς, (b) ἡνίκα ἂν καθίσῃ έν συνεδρίω μετά τῶν γερόντων κατοίκων τῆς γῆς. "(a) Her husband is notable in the gates (b) whenever he sits in council with the elders who dwell in the land." G enhances the husband's status—he is  $\pi\epsilon\rho(\beta\lambda\epsilon\pi\tau\circ\varsigma)$ — "notable," lit. "looked at from all sides." He sits in the gates, G explains, not for leisure but for service in the city assembly. συνέδριον is added or introduced here and in 11:13; 15:22; 22:10 (2x); 24:8; 27:22. The assembly was clearly an important part of the communal life of the translator (see 15:22). This is not the Sanhedrin in Jerusalem (contrary to Cook 1999a, 151), for the statement is pertinent to all male readers, or at least "the elders who dwell in the land" (31:23b), not only the (traditionally) seventy-one men in the Sanhedrin. Possibly these are formally recognized local assemblies (also called συνέδρια, see Mark 13:9 = Matt 10:17), which had judicial authority in Palestine. But the translator may equally well be thinking of informal gatherings of elders to discuss communal affairs, conduct business, and adjudicate conflicts. In 31:23 the assembly is said to meet in the city gate, though this may be a carryover from the Hebrew and not a reflection of contemporary Hellenistic practice.

V translates זקני as *senatoribus* ("senators"), reflecting the same idea as G about the setting of the husband's activities.

S says that her husband is known حيتلا حبيتلا ("in the cities"), giving "gates" a meaning it has in Deuteronomy and elsewhere (though S does not translate it that way in the Pentateuch).

## 31:26 verse order 26-25 G ] 25-26 M S (assim)

In G, the  $\mathcal{Y}$ -verse (G 31:26 = M 31:25) follows the  $\mathfrak{D}$ -verse (G 31:25 = M 31:26). G uses the alternative, and earlier, alphabetic order found in Lam 2; 3; 4; and Ps 10 and in some epigraphic abecedaries. It must have been present in G's source text, since there was no motive for switching the verses in translation. The change probably took place in the M-trans-

mission, since M's order is the usual one and is standard in postexilic usage. Another motive for the switch could have been that M's order keeps the lines about clothing together.

G (25): (a) στόμα αὐτῆς διήνοιξεν προσεχόντως καὶ ἐννόμως (b) καὶ τάξιν ἐστείλατο τῆ γλώσσῃ αὐτῆς. "(a) She opened her mouth attentively and lawfully, (b) and she controlled [lit. 'set order to'] her tongue." In M, the woman speaks as a teacher—"the teaching of kindness is on her tongue"—whereas in G she speaks obediently and cautiously.

### 31:25

G: (26a) = M; (b) καὶ εὐφράνθη ἐν ἡμέραις ἐσχάταις. "(b) and she rejoiced [aor] in the last days." The last days are the end of her life. M's "she *laughs* at the latter day" must have seemed overconfident.

### 31:27

G: (a) στεγναὶ διατριβαὶ οἴκων αὐτῆς, (b) σῖτα δὲ ὀκνηρὰ οὐκ ἔφαγεν. "(a) The ways of her household are covered, (b) and she has not eaten lazy food."  $\exists$  "22 can mean "cover" (in the D-stem) or "watch" (in the G-stem). The translator uses the first meaning (actually, a distinct root). If διατριβαί means "way of life, practices," then στεγναί ("covered") must be a metaphor for "careful" (GELS). If it means "place of habitation, haunts" (also GELS), as in Lev 13:46, then "covered" can be understood literally: "Ils sont bien couverts, les séjours de sa maison" (BAP). On the plural of οἴκων, see the comment on 7:8.

#### 31:28

G: (a) [≈ M 31:26a] τὸ στόμα δὲ ἀνοίγει σοφοῖς [G<sup>BS</sup>; var: σοφῶς G<sup>A</sup>] καὶ νομοθέσμως, (b) [≈ M 31:26b] ἡ δὲ ἐλεημοσύνη αὐτῆς [≈ M 31:28a] ἀνέστησεν τὰ τέκνα αὐτῆς, καὶ ἐπλούτησαν, (c) [≈ M 31:28b] καὶ ὁ ἀνὴρ αὐτῆς ἦνεσεν αὐτήν. "(a) [≈ M 31:26a] She opens her mouth to the wise [var: 'wisely'] and in accordance with the law. (b) [≈ M 31:26b] And her kindness [≈ M 31:28a] raised her children, and they grew rich. (c) [≈

וְאַׁתְּ עָלִית עַל־כָּלֶנָה: 30 שָׁקָר הַחֵז וְהָבֶל הַיֶּפִי אִשְׁה יִרְאַת־יְהוָה הֵיא תִתְהַלֶּל: 31 תְּנוּ־לָה מִפְּרֵי יָדֶיָה וְיהַלְלוּהָ בַשְׁעָרֵים מַעַשֶׂיהָ:

M 31:28b] And her husband praised her." G 31:28 is a blend of M 31:26 and 28 that probably arose by misreading of a Hebrew text. The changes were caused by (1) duplication and displacement of M 31:26 (present in G 31:26 in a different form); (2) a construal of שם as a causal; and (3) an association of אשרוה ויאשרוה ייאשרוה עש"ר The third confusion was due to the weakening of the gutturals in the first century BCE. Consequently, אשרוה lost its then-meaningless suffix, whether in transmission or translation. The adverb νομοθέσμως (a hapax) represents ותורת (= M). Similarly in G 31:26, the translator adds an adverb meaning *lawfully*, namely, έννόμως.

Of the variants  $\sigma \sigma \phi \tilde{\alpha} \varsigma$  and  $\sigma \sigma \phi \tilde{\omega} \varsigma$ , the latter (though preferred by Rahlfs) looks like an adjustment to M. When read with  $\sigma \sigma \phi \tilde{\alpha} \varsigma$ , the verse presents the interesting picture of the capable woman instructing wise men in God's word. She is a teacher of Torah also in G 31:2, though that role is elided in G 31:26.

### 31:29

G: (a) Πολλαὶ θυγατέρες ἐκτήσαντο πλοῦτον, (b) πολλαὶ ἐποίησαν δυνατά, (c) σὺ δὲ ὑπέρκεισαι καὶ ὑπερῆρας πάσας. "(a) Many daughters have acquired wealth; (b) many have done mighty deeds, (c) but you have surpassed and exceeded all (of them)." G 31:29a and 29b bring out the two meanings of היל "wealth" and "power." G 31:29a (like 31:31a) assumes that women can amass wealth of their own, as was possible in the Hellenistic world. In G 31:29c, ὑπέρκεισαι and ὑπερῆρας are an emphatic double translation of y.

### 31:30

G: (a) ψευδεῖς ἀρέσκειαι καὶ μάταιον κάλλος γυναικός. (b) γυνὴ γὰρ συνετὴ εὐλογεῖται, (c) φόβον δὲ κυρίου αὕτη αἰνείτω. "(a) Desires for favor are deceitful and the beauty of woman is vain. (b) For it is the intelligent woman that is praised. (c) Let her praise the fear of the Lord."

In G 31:30a, η is given a unique translation, the *hapax* ἀρέσχεια. χάρις, its usual translation, is a reward for wisdom and virtue (G 3:3, 22; 13:15; etc.), not something that could be called deceitful. The translator added "of woman" to indicate that not all beauty, but only a woman's, is to be denigrated, probably because female beauty arouses male sexual desire.

G 31:30*bc* is a double translation of M 31:30*b*, each component offering a different interpretation of that line. G first provides an epithet that sums up the woman's qualities as described in the poem, then praises her most important virtue, the fear of God. The woman is said to do the praising of this virtue to indicate that she embraces it. In G 31:30*c*, the passive of התהלל is replaced by the active, because what is important to the ideal is not the receiving of praises but the teaching of virtues.

G's אשת בינה does not represent the anomalous אשת היעטע סער אשת מער אשת בינה, as Toy suggests. Rofé (2002) proposes that G's source text read גבונה, G האישה משכלי often represent forms of איכ"ל, which means both "to be intelligent" and "to succeed." Rofé argues that משכלת is the original and was replaced by הוה יראת יהוה order to imbue the woman with the quality of piety, otherwise unmentioned in the poem. But in 19:14 G treats אשה משכלת quite differently, making it less likely that משכלת was in the source text.

## 31:31

G: (a) δότε αὐτῆ ἀπὸ καρπῶν χειλέων [G<sup>BSAMSS</sup>; var: χειρῶν G<sup>Complut Rahlfs</sup> SyrH] αὐτῆς, (b) καὶ αἰνείσθω ἐν πύλαις ὁ ἀνὴρ αὐτῆς. "(a) Give her from the fruits of her lips [var: 'hands'], (b) and let her husband be praised in the gates." Rahlfs accepts the hexaplaric χειρῶν, but, as BAP observes, "fruit of the lips" is equivalent to פרי פר ("fruit of the mouth"), an established usage (12:14; 13:2; 18:20). G's notion that it is her husband who is praised in the gates was inspired by 31:23, which says that he is "known" or "notable." "Gates" in both verses may have suggested to the translator that they describe the same event.

# Bibliography

- Aejmelaeus, Anneli. 1993. On the Trail of Septuagint Translators: Collected *Essays.* Kampen: Kok Pharos.
- Aitken, James K. 2009. "The Jewish Use of Greek Proverbs." Pages 53–77 in Jewish Reception of Greek Bible Versions: Studies in Their Use in Late Antiquity and the Middle Ages. Edited by Nicholas de Lang, Julia G. Krivoruchko, and Cameron Boyd-Taylor. Texts and Studies in Medieval and Early Modern Judaism 24. Tübingen: Mohr Siebeck.
- Albrektson, Bertil. 1978. Reflections on the Emergence of a Standard Text of the Hebrew Bible. Pages 49–65 in *Congress Volume: Göttingen 1977*. Edited by J. A. Emerton. Supplements to Vetus Testamentum 29. Leiden: Brill.
  - ------. 1981. *Difficilior Lectio Probabilior*: A Rule of Textual Criticism and Its Use in Old Testament Studies. *Oudtestamentische Studiën* 21:5–18.
- Austen, Jane. 1923. *The Novels of Jane Austen: The Text Based on Collation of the Early Editions*. Edited by Robert W. Chapman. 5 vols. Oxford: Clarendon.
- Baer, Seligmann. 1880. Liber proverbiorum: Textum Masoreticum accuratissime expressit e fontibus Masorae codicumque varie illustravit expositionem de legibus dagessationis adjecit. Leipzig: Tauchnitz.
- Barr, James. 1968. Vocalization and the Analysis of Hebrew among the Ancient Translators. Pages 1–11 in *Hebräische Wortforschung*. Edited by Benedict Hartmann et al. Supplements to Vetus Testamentum 16. Leiden: Brill.
  - —. 1975. Ba'areș—molis: Prov. XI.31, I Pet. IV.18. Journal of Semitic Studies 20:149–64.
  - —. 1979. *The Typology of Literalism in Ancient Biblical Translations*. Mitteilungen des Septuaginta-Unternehmens 15. Göttingen: Vandenhoeck & Ruprecht.
  - —. 1981. A New Look at Kethibh-Qere. *Oudtestamentische Studiën* 21:19–37.

- ——. 1989. *The Variable Spellings of the Hebrew Bible*. Oxford: Oxford University Press.
- Barthélemy, Dominique. 1963. *Les Devanciers d'Aquila*. Supplements to Vetus Testamentum 10. Leiden: Brill.
  - ——. 1982–2005. *Critique textuelle de l'Ancien Testament.* 4 vols. Orbis Biblicus et Orientalis 50. Göttingen: Vandenhoeck & Ruprecht.
- ------. 1985. Doubts about Homoeophony in the Septuagint. *Textus* 12:1–77.
- Baumgartner, Antoine J. 1890. Étude critique sur l'état du texte du livre des Proverbes d'après les principales traductions anciennes. Leipzig: Drugulin.
- Beentjes, Pancratius C. 1997. The Book of Ben Sira in Hebrew: A Text Edition of All Extant Hebrew Manuscripts and a Synopsis of All Parallel Hebrew Ben Sira Texts. Supplements to Vetus Testamentum 68. Leiden: Brill.
- Boström, Gustav. 1935. Proverbiastudien: Die Weisheit und das fremde Weib in Sprüche 1–9. Lund: Gleerup.
- Bowers, Fredson. 1956. Hamlet's "Sullied" or "Solid" Flesh: A Bibliographical Case-History. *Shakespeare Survey* 9:44–48.
- Breuer, Mordecai. 2003. נוסח המקרא בכתר ירושלים [The Masoretic Text in the Jerusalem Crown]. Jerusalem: Keren Ha-Masorah.
  - —, ed. 2004. *Keter-Crown Bible*. Jerusalem: Hebrew University Press.
- Brooke, George J. 2005. The Qumran Scrolls and the Demise of the Distinction between Higher and Lower Criticism. Pages 26–42 in New Directions in Qumran Studies: Proceedings of the Bristol Colloquium on the Dead Sea Scrolls, 8–10 September 2003. Edited by Jonathan G. Campbell, William J. Lyons, and Lloyd Pietersen. London: T&T Clark.
- Cappelli, Ludovici. 1650. Critica sacra. Paris: Cramoisy.
- Carmignac, Jean. 1980. Critique textuelle de Proverbes 22,8–9. *Folia Orientalia* 21:33–41.
- Carr, David McLain. 2005. Writing on the Tablet of the Heart: Origins of Scripture and Literature. Oxford: Oxford University Press.
- ——. 2011. *The Formation of the Hebrew Bible: A New Reconstruction*. New York: Oxford University Press.
- Cassuto, Philippe. 1989. *Qeré-Ketib et listes massorétiques dans le manuscrit B 19a*. Judentum und Umwelt 26. New York: Lang.
- Clifford, Richard J. 1999. *Proverbs*. Old Testament Library. Louisville: Westminster John Knox.

- Clines, David J. A., ed. 1993–2011. *Dictionary of Classical Hebrew*. 8 vols. Sheffield: Sheffield Phoenix Press.
- Cohen, Maimon. 2007. הכתיב והקרי שבמקרא [The Ketiv and the Qeré in the Biblical Text]. Jerusalem: Magnes.
- Cohen, Philip, and David H. Jackson. 1991. Notes on Emerging Paradigms in Editorial Theory. Pages 103–23 in *Devils and Angels: Textual Editing and Literary Theory*. Edited by Philip Cohen. Charlottesville: University Press of Virginia.
- Cook, Johann. 1993. The Dating of Septuagint Proverbs. *Ephemerides theologicae lovanienses* 69:383–99.
  - —. 1997. The Septuagint of Proverbs: Jewish and/or Hellenistic Proverbs? Concerning the Hellenistic Colouring of LXX Proverbs. Supplements to Vetus Testamentum 69. Leiden: Brill.
  - ——. 1999a. Contextual Exegetical Interpretations in the Septuagint of Proverbs. *Journal of Northwest Semitic Languages* 25:139–52.
  - ——. 1999b. The Law of Moses in Septuagint Proverbs. *Vetus Testamentum* 49:448–61.
  - ——. 2000. Textual Problems in the Septuagint Version of Proverbs. *Journal of Northwest Semitic Languages* 26:171–79.

 2001. Ideology and Translation Technique: Two Sides of the Same Coin? Pages 195–210 in *Helsinki Perspectives on the Translation Technique of the Septuagint*. Edited by Raija Sollamo and Seppo Sipilä. Publications of the Finnish Exegetical Society 82. Göttingen: Vandenhoeck & Ruprecht.

—. 2002. The Law of Moses as a Fence and a Fountain. Pages 280–88 in *Sense and Sensitivity: Essays on Reading the Bible in Memory of Robert Carroll.* Edited by A. G. Hunter and Philip R. Davies. Journal for the Study of the Old Testament Supplement Series 348. Sheffield: Sheffield Academic Press.

—. 2003. The Greek of Proverbs: Evidence of a Recensionally Deviating Hebrew Text? Pages 605–18 in *Emanuel: Studies in Hebrew Bible, Septuagint and Dead Sea Scrolls in Honor of Emanuel Tov.* Edited by Shalom Paul, Robert Kraft, and Lawrence Schiffman. Leiden: Brill.

——, trans. 2007. Proverbs. Pages 621–47 in Pietersma and Wright 2007.

Crawford, Sidnie W. 2005. Textual Criticism of the Book of Deuteronomy and the Oxford Hebrew Bible Project. Pages 315–26 in Seeking Out the Wisdom of the Ancients: Essays Offered to Honor Michael V. Fox on the Occasion of His Sixty-Fifth Birthday. Edited by Ronald L. Troxel, Kelvin G. Friebel, and Dennis R. Magary. Winona Lake, IN: Eisenbrauns.

- Dahood, Mitchell J. 1963. *Proverbs and Northwest Semitic Philology*. Scripta Pontificii Instituti Biblici 113. Rome: Pontifical Biblical Institute.
- Dathe, Johann A. 1764. De ratione consensus versionis Chaldaicae et Syriacae Proverbiorum Salomonis. Leipzig: Langenheim.
- Davis, Tom. 1998. The Monsters and the Textual Critics. Pages 95–111 in *Textual Formations and Reformations*. Edited by Laurie E. Maguire and Thomas L. Berger. Newark: University of Delaware Press.
- Delitzsch, Franz. 1873. *Biblical Commentary on the Proverbs of Solomon*. Translated by M. G. Easton. Grand Rapids: Eerdmans. Repr., 1983.
- Delitzsch, Friedrich. 1920. *Die Lese- und Schreibfehler im Alten Testament*. Berlin: de Gruyter.
- Dick, Michael B. 1990. The Ethics of the Old Greek Book of Proverbs. *Studia Philonica Annual* 2:20–50.
- Díez Merino, Luis. 1984. *Targum de Proverbios: Edición príncipe del Ms. Villa-Amil no. 5 de Alfonso de Zamora*. Bibliotheca Hispana Bíblica 11. Madrid: Consejo Superior de Investigaciones Científicas.
- Dotan, Aron. 1971. Masorah. Pages 1401–82 in vol. 16 of *Encyclopedia Judaica*. Edited by Cecil Roth and Geoffrey Wigoder. 16 vols. New York: Macmillan, 1971.
  - —. 2001. *Biblia Hebraica Leningradensia*. Peabody, MA: Hendrickson.
- Ehrlich, Arnold B. 1913. *Randglossen zur hebräischen Bibel: Textkritisches, Sprachliches und Sachliches*. Vol. 6. Hildesheim: Olms.
- Erman, Adolf. 1924. Eine ägyptische Quelle der "Spruch Salomos." Sitzungsberichte der Preussischen Akademie der Wissenschaften, philosophisch-historische Klasse 15:86–93.
- Erne, Lukas. 2008. *Shakespeare's Modern Collaborators*. London: Continuum.
- Eshel, Hanan. 2003. 6Q30, a Cursive *šîn*, and Proverbs 11. *Journal of Biblical Literature* 122:544–46.
- Field, Frederick. 1875. Origenis Hexaplorum quae supersunt sive Veterum interpretum graecorum in totum Vetus Testamentum fragmenta. 2 vols. Oxford: Clarendon.
- Fischer, Bonifatius, and Robert Weber, eds. 1983. *Biblia Sacra: Iuxta Vulgatam Versionem*. 3rd edition. Stuttgart: Deutsche Bibelgesellschaft.
- Flint, Peter W. 1997. *The Dead Sea Psalms Scrolls and the Book of Psalms*. Studies on the Texts of the Desert of Judah 17. Leiden: Brill.

Foakes, R. A., ed. 1997. King Lear. London: Thomas Nelson and Sons.

Fox, Michael V. 1991. *The Redaction of the Books of Esther: On Reading Composite Texts*. Society of Biblical Literature Monograph Series 40. Atlanta: Scholars Press.

. 2000. Proverbs 1–9. Anchor Bible 18A. Garden City, NY: Doubleday.

- —. 2002. Translation and Mimesis. Pages 207–21 in *Bible Translation in Context*. Edited by Frederick Knobloch. Studies in Jewish History and Culture 10. Bethesda: University of Maryland Press.
- ——. 2004. Review of Gerhard Tauberschmidt, Secondary Parallelism: A Study of Translation Technique in the LXX. Review of Biblical Literature 11. http://www.bookreviews.org/pdf/4192\_4111.pdf.

------. 2005. LXX-Proverbs as a Text-Critical Resource. *Textus* 22:95–128.

——. 2006. Producing a Critical Text of Proverbs: The Challenge of the Oxford Hebrew Bible. *Journal of Northwest Semitic Languages* 32:1–22.

- ——. 2009. *Proverbs 10–31*. Anchor Bible 18B. New Haven, CT: Yale University Press.
- -----. 2013. How the Peshitta of Proverbs Uses the Septuagint. *Journal of Northwest Semitic Languages* 39:37–56.
- Freedman, David Noel. 1972. The Broken Construct Chain. *Biblica* 53:534–36.

Friedman, Matti. 2012. *The Aleppo Codex: A True Story of Obsession, Faith, and the Pursuit of an Ancient Bible.* Chapel Hill, NC: Algonquin Books.

- Fritsch, Charles T. 1953. The Treatment of the Hexaplaric Signs in the Syro-Hexaplar of Proverbs. *Journal of Biblical Literature* 72:169–81.
- Furness, H. H, ed. 1880. *Shakespeare's King Lear*. Vol. 5 in *Variorum: A New Variorum Shakespeare*. Philadelphia: Lippincott.
- Geiger, Abraham. 1857. Urschrift und Übersetzung der Bibel. Breslau: Hainauer. Repr., Frankfurt am Main: Madda, 1928.
- Gemser, Berend. 1963. *Sprüche Salomos*. 2nd ed. Handbuch zum Alten Testament 1/16. Tübingen: Mohr.
- Gerleman, Gillis. 1950. The Septuagint Proverbs as a Hellenistic Document. *Oudtestamentische Studiën* 8:15–27.
  - —. 1956. *Studies in the Septuagint III: Proverbs*. Lund: Gleerup.
- Gesenius, Wilhelm. 1910. *Gesenius' Hebrew Grammar*. Edited by E. Kautzsch. Translated by A. E. Cowley. 2nd ed. Oxford: Clarendon.
- Giese, Ronald L., Jr. 1990. Wisdom and Wealth in the Septuagint of Proverbs. PhD diss., University of Wisconsin-Madison.

<sup>—. 1992</sup>a. Qualifying Wealth in the Septuagint of Proverbs. *Journal of Biblical Literature* 111:409–25.

- ——. 1992b. Strength through Wisdom and the Bee in LXX-Proverbs. *Biblica* 73:404–11.
- ------. 1993a. Compassion for the Lowly in Septuagint Proverbs. *Journal for the Study of the Pseudepigrapha* 11:109–17.
  - ——. 1993b. Dualism in the LXX of Prov 2: 17: A Case Study in the LXX as Revisionary Translation. *Journal of the Evangelical Theological Society* 36:289–95.
- Gordis, Robert. 1971. *The Biblical Text in the Making: A Study of the Kethib-Qere.* 2nd ed. New York: Ktav. First published 1937.
- Gordon, Cyrus H. 1930. Rabbinic Exegesis in the Vulgate of Proverbs. *Journal of Biblical Literature* 49:384–416.
- Gordon, Robert P. 1999. "Converse Translation" in the Targums and Beyond. *Journal for the Study of the Pseudepigrapha* 19:3–21.
- Goshen-Gottstein, Moshe. 1954. Die Jesaiah-Rolle und das Problem der hebräischen Bibelhandschriften. *Biblica* 35:429–42.
- ——. 1957. The History of the Bible Text and Comparative Semitics: A Methodological Problem. *Vetus Testamentum* 7:195–201.
- ——. 1960. *Text and Language in Bible and Qumran*. Jerusalem: Orient Publishing.
- Grabe, Joannes Ernestus. 1707–1720. Septuaginta interpretum ... ex antiquissimo codice Alexindro. Oxford: Sheldon.
- Graves, Michael. 2003. The Origins of the *Ketiv-Qere* Readings. *TC: A Journal of Biblical Textual Criticism* 8. http://rosetta.reltech.org/TC/ vol08/Graves2003.html.
- Greetham, David C. 1994. *Textual Scholarship: An Introduction*. New York: Garland.
- Gressmann, Hugo. 1924. Die neugefundene Lehre des Amen-em-ope und die vorexilische Spruchdichtung Israels. Zeitschrift für die alttestamentliche Wissenschaft 42:272–96.
- Haar Romeny, Robert Bas ter. 2005. Hypotheses on the Development of Judaism and Christianity in Syria in the Period after 70 C.E. Pages 13–33 in *Matthew and the Didache*. Edited by Huub van de Sandt. Assen: Van Gorcum.
- Hadas, Moses. 1973. *Aristeas to Philocrates (Letter of Aristeas)*. Philadelphia: Harper for Dropsie College, 1951. Repr., New York: Ktav.
- Haefeli, Leo. 1927. Die Peschitta des Alten Testamentes, mit Rücksicht auf ihre textkritische Bearbeitung und Herausgabe. Alttestamentliche Abhandlungen 11/1. Münster: Aschendorff.

- Halio, Jay L., ed. 1992. *The Tragedy of King Lear.* Cambridge: Cambridge University Press.
- ——. 1994. The First Quarto of King Lear. Cambridge: Cambridge University Press.
- Hamonville, David-Marc d'. 2000. La Bible d'Alexandrie: Les Proverbes. Paris: Cerf.
- Hayut, Anat. 2013. The Differences between the Western and Eastern Recensions (Maarvaey and Medinhaey) [Hebrew]. M.A. thesis, Bar Ilan University, Ramat Gan, Israel.
- Healey, John F. 1991. *The Targum of Proverbs*. The Aramaic Bible 15. Collegeville, MN: Liturgical Press.
- ——. 2012. Targum Proverbs and the Peshitta: Reflections on the Linguistic Environment. Pages 325–35 in *Studies on the Text and Versions of the Hebrew Bible in Honour of Robert Gordon*. Edited by Geoffrey Khan and Diana Lipton. Leiden: Brill.
- Heater, Homer. 1982. A Septuagint Translation Technique in the Book of Job. Catholic Biblical Quarterly Monograph Series 11. Washington, DC: Catholic Biblical Association.
- Heidenheim, M. 1865–67. Zur Textkritik der Proverbien. *Deutsche Vierteljarschrift* 2:395–414 [Prov 1–3]; 3:51–60 [Prov 4–7]; 3:327–46 [Prov 8–9]; 3:445–60 [Prov 10–14]; 3:468–87 [Prov 15–31].
- Hendel, Ronald. Forthcoming. What Is a Biblical Book? In From Author to Copyist: Composition, Redaction and Transmission of the Hebrew Bible; Essays in Honor of Zipora Talshir. Edited by Cana Werman. Winona Lake, IN: Eisenbrauns.
- Hengel, Martin. 1974. *Judaism and Hellenism: Studies in Their Encounter in Palestine during the Early Hellenistic Period*. Translated by John Bowden. 2 vols. London: SCM.
- Hitzig, Ferdinand. 1858. Die Sprüche Salomo's. Zürich: Orell, Füssli.
- Holderness, Graham, ed. 1995. *M. William Shak-speare, His True Chronicle Historie of the Life and Death of King Lear and His Three Daughters.* Hemel Hempstead, Hertfordshire: Prentice Hall.
- Holmes, James S. 1988. *Translated! Papers on Literary Translation and Translation Studies*. Amsterdam: Rodopi.
- Holmes, Robert, and James Parson, eds. 1798–1827. *Vetus Testamentum Graecum cum variis lectionibus*. Oxford: Clarendon.
- Hyun, Changhak. 2000. A Study of the Translation Technique of Peshitta Proverbs. PhD diss., University of Wisconsin-Madison.

- Jäger, Johannes G. 1788. *Observationes in Proverbium Salomonis versionem alexandrinam*. Meldorf: Boie.
- Joosten, Jan. 1995. Doublet Translations in Peshitta Proverbs. Pages 63–72 in *The Peshitta as a Translation: Papers Read at the II Peshitta Symposium Held at Leiden, 19–21 August 1993.* Edited by P. B. Dirksen and Arie van der Kooij. Monographs of the Peshitta Institute 8. Leiden: Brill.
  - —. Forthcoming. The Tiberian Vocalization and the Edition of the Hebrew Bible Text. In *Text History of the Hebrew Bible (Nebi'im)* Between the 4th and 1st Centuries BCE: Multiplicity of Text Types and/ or Hierarchy. Edited by Innocent Himbaza. Fribourg.
- Joüon, Paul. 1993. *A Grammar of Biblical Hebrew*. Translated by T. Muraoka. Subsidia Biblica 14/1–2. Rome: Pontifical Biblical Institute.
- Junker, Hubert. 1936. Konsonantenumstellung als Fehlerquelle und textkritisches Hilfsmittel im MT. Pages 162–74 in *Werden und Wesen des Alten Testaments*. Edited by Paul Volz et al. Beihefte zur Zeitschrift für die alttestamentliche Wissenschaft 66. Berlin: Töpelmann.
- Kaminka, Armand. 1931–32. Septuaginta und Targum zu Proverbia. *Hebrew Union College Annual* 8–9:169–91.
- Kedar, Benjamin. 1968. The Vulgate as Translation. PhD diss., Hebrew University of Jerusalem.
- Kennedy, James. 1928. An Aid to the Textual Amendment of the Old Testament. Edited by N. Levison. Edinburgh: T&T Clark.
- Kenney, E. J. 2005. Textual Criticism. *Encyclopaedia Britannica Online*. http://www.search.eb.com/eb/article-58834.
- Kennicott, Benjamin. 1776–1780. Vetus Testamentum Hebraicum cum variis lectionibus. 2 vols. Oxford: Clarendon.
- Klein, Michael L. 1976. Converse Translation: A Targumic Technique. *Biblica* 57:515–37.
- Köhler, Ludwig, Walter Baumgartner, and Johann Jakob Stamm. 1994–2000. *The Hebrew and Aramaic Lexicon of the Old Testament*. Translated by M. E. J. Richardson. 5 vols. Leiden: Brill.
- Krauss, Wolfgang, and Martin Karrer, eds. 2009. *Septuaginta Deutsch*. Stuttgart: Deutsche Bibelgesellschaft.
- Küchler, Max. 1992. Gott und seine Weisheit in der Septuaginta (Ijob 28; Spr 8). Pages 118–43 in Monotheismus und Christologie: Zur Gottesfrage im hellenistischen Judentum und im Urchristentum. Edited by H.-J. Klauck. Freiburg: Herder.

- Kuhn, Gottfried. 1931. *Beiträge zur Erklärung des salomonischen Spruchbuches*. Beiträge zur Wissenschaft vom Alten (und Neuen) Testament 57. Stuttgart: Kohlhammer.
- Lagarde, Paul de. 1863. Anmerkungen zur griechischen Übersetzung der Proverbien. Leipzig: Brockhaus.
  - —. 1873. *Hagiographa Chaldaice*. Leipzig: Teubneri.
- Landes, George M. 1956. The Fountain at Jazer. *Bulletin of the American Schools of Oriental Research* 144:30–37.
- Levinson, Bernard M. 2011. *The Right Chorale: Studies in Biblical Law and Interpretation*. Winona Lake, IN: Eisenbrauns.
- Lichtheim, Miriam. 1973–1980. *Ancient Egyptian Literature*. 3 vols. Berkeley: University of California Press.
- Liddell, Henry George, Robert Scott, and Henry Stuart Jones. 1996. *A Greek-English Lexicon*. 9th ed. Oxford: Clarendon.
- Louw, Theo A. W. van der. 2007. *Transformations in the Septuagint: Towards an Interaction of Septuagint Studies and Translation Studies.* Contributions to Biblical Exegesis and Theology 47. Leuven: Peeters.
- Luc, Alex. 2000. The Title and Structure of Proverbs. *Zeitschrift für die alttestamentliche Wissenschaft* 112:252–55.
- Lust, Johan, Erik Eynikel, and Katrin Hauspie. 1992–1996. A Greek-English Lexicon of the Septuagint. 2 vols. Stuttgart: Deutsche Bibelgesellschaft.
- Maas, Paul. 1956. *Textual Criticism*. Translated by Barbara Flower. Oxford: Clarendon.
- Margolis, Max L. 1910. Complete Induction for the Identification of the Vocabulary in the Greek Versions of the Old Testament with Its Semitic Equivalents: Its Necessity and the Means of Obtaining It. *Journal of the American Oriental Society* 30:301–12.
- Martin, Gary D. 2010. *Multiple Originals: New Approaches to Hebrew Bible Textual Criticism.* Society of Biblical Literature Text-Critical Studies 7. Atlanta: Society of Biblical Literature.
- Maybaum, S. 1871. Über die Sprache des Targum zu den Sprüchen und dessen Verhältnis zum Syrer. *Archiv für wissenschaftliche Erforschung des Alten Testamentes* 2:66–93.
- McKane, William. 1970. *Proverbs*. Old Testament Library. Philadelphia: Westminster.
- McKerrow, Ronald B. 1939. Prolegomena for the Oxford Shakespeare: A Study in Editorial Method. Oxford: Clarendon.

- Meinhold, Arndt. 1991. *Die Sprüche*. Zürcher Bibelkommentare 16. 2 vols. Zürich: Theologischer Verlag.
- Melammed, Ezra Z. 1972. תרגום משלי [Targum Proverbs]. Bar Ilan 9:18– 91.
- Mezzacasa, Giacomo. 1913. Il Libro dei Proverbi di Salomone: Studie Critico sulle Aggiunte Greco-Alessandrine. Rome: Pontifical Biblical Institute.
- Minhat Shai. See Yedidiah Shlomo of Norzi.
- Muraoka, Takamitsu. 2009. A Greek-English Lexicon of the Septuagint. Leuven: Peeters.
- Murphy, Roland E. 1998. *Proverbs*. Word Biblical Commentary 22. Nashville: Nelson.
- Nahmias (Yosef ben Yosef ibn Nahmias). [first half of 14th century]. Commentary on Proverbs [Hebrew]. Edited by Arieh L. Bamberger. Berlin: Itschkowsky, 1911. Repr., Jerusalem, 1969 [?].
- Orlinsky, Harry M. 1958. Studies in the Septuagint of the Book of Job, Chapter II: The Character of the Septuagint Translation of the Book of Job. *Hebrew Union College Annual* 29:229–71.
  - —. 1960. The Origin of the Kethib-Qere System: A New Approach. Pages 184–92 in *Congress Volume: Oxford, 1959*. Supplements to Vetus Testamentum 7. Leiden: Brill.
- Owens, Robert J. 1998. The Relationship between the Targum and Peshitta Texts of the Book of Proverbs: Status Quaestionis. Pages 195–207 in *Targum and Peshitta*. Edited by Paul Flesher. Atlanta: Scholars Press.
- Perles, Felix. 1895. *Analekten zur Textkritik des Alten Testaments*. Munich: Ackermann.
  - —. 1906. The Fourteenth Edition of Gesenius-Buhl's Dictionary. Review of Wilhelm Gesenius, *Hebräisches und Aramäisches Handwörterbuch über das Alte Testament. Jewish Quarterly Review* 18:383–90.
  - —. 1922. *Analekten zur Textkritik des Alten Testaments*. 2nd ed. Leipzig: Engel.
- Peursen, Wido van. 2004. The Peshitta of Ben Sira: Jewish and/or Christian? *Aramaic Studies* 2:243–62.
- Pietersma, Albert, and Benjamin G. Wright, eds. 2007. A New English Translation of the Septuagint. Oxford: Oxford University Press.
- Pinkuss, Hermann. 1894. Die syrische Übersetzung der Proverbien. Zeitschrift für die alttestamentliche Wissenschaft 14:65–141, 161–222.
- Plöger, Otto. 1984. *Sprüche Salomos (Proverbia)*. Biblischer Kommentar, Altes Testament 17. Neukirchen: Neukirchener Verlag.

- Porten, Bezalel, and Ada Yardeni. 1993. *Textbook of Aramaic Documents*. Vol. 3. Jerusalem: Akademon.
- Rahlfs, Alfred. 1979. *Septuaginta*. Stuttgart: Deutsche Bibelgesellschaft. First published 1935.
- Rajak, Tessa. 2009. *Translation and Survival: The Greek Bible of the Ancient Jewish Diaspora*. Oxford: Oxford University Press.
- Rashi (Solomon ben Isaac). [11th century]. Proverbs Commentary in *Rabbinic Bible* II, all editions.
- Roberts, Colin H. 1950. *The Antinoopolis Papyri*. Vol. 1. Graeco-Roman Memoirs 28. London: Egypt Exploration Society.
- Rofé, Alexander. 2002. The Valiant Woman, gunē synetē, and the Redaction of the Book of Proverbs. Pages 145–55 in Vergegenwärtigung des Alten Testaments: Beiträge zur biblischen Hermeneutik; Festschrift für Rudolf Smend zum 70. Edited by C. Bultmann, W. Dietrich, and C. Levin. Göttingen: Vandenhoeck & Ruprecht.
- Rooy, Herrie F. van. 2004. A New Critical Edition of the Hebrew Bible. Journal of Northwest Semitic Languages 30:139–50.
- Rossi, Giovanni Bernardo de. 1784–1788. Variae lectiones Veteris Testamenti. 4 vols. Parma: Regio. Repr., Amsterdam: Philo, 1969.
- Rüger, Hans Peter. 1959. Vier Aquila-Glossen in einem hebräischen Proverbien-Fragment aus der Kairo-Geniza. Zeitschrift für die neutestamentliche Wissenschaft und die Kunde der älteren Kirche 50:275–77.
- Saʿadia Gaon. 1975–76. משלי [Commentary on Proverbs]. [9th–10th century]. Edited and translated from Judaeo-Arabic by Yosef Qafiḥ. Jerusalem: n.p.
- Sandelin, Karl-Gustav. 1986. Wisdom as Nourisher: A Study of an Old Testament Theme, Its Development within Early Judaism, and Its Impact on Early Christianity. Åbo: Åbo Akademi.
- Scoralick, Ruth. 2002. Salomos griechische Gewänder: Beobachtungen zur Septuagintafassung des Sprichwörterbuches. Pages 43–75 in Rettendes Wissen: Studien zum Fortgang weisheitlichen Denkens im Frühjudentum und im frühen Christentum. Edited by K. Löning. Münster: Ugarit-Verlag.
- Scott, Melville. 1927. *Textual Discoveries in Proverbs, Psalms, and Isaiah*. New York: Macmillan.
- Scriba, Albrecht. 1995. Die Geschichte des Motivkomplexes Theophanie. Forschungen zur Religion und Literatur des Alten und Neuen Testaments 167. Göttingen: Vandenhoeck & Ruprecht.

- Seeligmann, I. L. 1953. Voraussetzungen der Midraschexegese. Pages 150–81 in Congress Volume: Copenhagen, 1953. Edited by G. W. Anderson. Vetus Testamentum Supplement 1. Leiden: Brill.
- Shakespeare, William: editions listed by editors' names; see Foakes, Furness, Halio, Weis, and Wells and Taylor.
- Skehan, Patrick W. 1946. Proverbs 5:15–19 and 6:20–24. *Catholic Biblical Quarterly* 8:290–97.
- Skehan, Patrick W., and Alexander A. Di Lella. 1987. *The Wisdom of Ben Sira*. Anchor Bible 39. New York: Doubleday.
- Snell, Daniel C. 1993. *Twice-Told Proverbs and the Composition of the Book of Proverbs*. Winona Lake, IN: Eisenbrauns.
- Sokoloff, Michael. 2009. A Syriac Lexicon: A Translation from the Latin, Correction, Expansion, and Update of C. Brockelmann's Lexicon Syriacum. Winona Lake, IN: Eisenbrauns; Piscataway, NJ: Gorgias.
- Sperber, Alexander. 1966. A Historical Grammar of Biblical Hebrew: A Presentation of Problems with Suggestions to Their Solution. Leiden: Brill.
- Spevack, Marvin. 1996–97. The End of Editing Shakespeare. *Connotations* 6:78–85.
- Szpek Heidi M. 1998. On the Influence of the Septuagint on the Peshitta. *Catholic Biblical Quarterly* 60:251–66.
- Talmon, Shemaryahu. 1960. Double Readings in the Masoretic Text. *Textus* 1:144–84.
  - —. 1975. The Textual Study of the Bible: A New Outlook. Pages 321–400 in *Qumran and the History of the Biblical Text*. Edited by S. Talmon and F. M. Cross. Cambridge: Harvard University Press.

—. 1981. The Ancient Hebrew Alphabet and Biblical Text Criticism. Pages 497–530 in *Mélanges Dominique Barthélemy: Études bibliques offertes à l'occasion de son 60e anniversaire*. Edited by P. Casetti, O. Keel, and A. Schenker. Orbis Biblicus et Orientalis 38. Göttingen: Vandenhoeck & Ruprecht.

- Tanselle, G. Thomas. 1976. The Editorial Problem of Final Authorial Intention. *Studies in Bibliography* 29:167–211.
  - —. 1989. *A Rationale of Textual Criticism.* Philadelphia: University of Pennsylvania Press.
- ------. 1996a. Reflections on Scholarly Editing. Raritan 16:52-64.
- ——. 1996b. Textual Instability and Editorial Idealism. *Studies in Bibliography* 49:1–60.
  - 2001. Textual Criticism at the Millennium. *Studies in Bibliography* 54:1–80.

- Tauberschmidt, Gerhard. 2004. Secondary Parallelism: A Study of Translation Technique in LXX Proverbs. Academia Biblica 15. Atlanta: Society of Biblical Literature.
- Thackeray, Henry St. John. 1912. The Poetry of the Greek Book of Proverbs. *Journal of Theological Studies* 13:46–66.
- Tov, Emanuel. 1979. Loan-Words, Homophony, and Transliteration in the Septuagint. *Biblica* 60:216–36.

—. 1985. The Literary History of the Book of Jeremiah in Light of Its Textual History. Pages 211–37 in *Empirical Models for Biblical Criticism*. Edited by Jeffrey H. Tigay. Philadelphia: University of Pennsylvania Press.

—. 1990a. The Greek Minor Prophets Scroll from Nahal Hever (HevXIIgr). Discoveries in the Judean Desert VIII. Oxford: Clarendon.
—. 1990b. Recensional Differences between the Masoretic Text and the Septuagint of Proverbs. Pages 43–56 in Of Scribes and Scrolls: Studies on the Hebrew Bible, Intertestamental Judaism, and Christian Origins, Presented to John Strugnell on the Occasion of His Sixtieth Birthday. Edited by Harold W. Attridge, John J. Collins, and Thomas H. Tobin. Lanham, MD: University Press of America.

. 1997. The Text-Critical Use of the Septuagint in Biblical Research.
 2nd ed. Jerusalem Biblical Studies 8. Jerusalem: Simor.

—. 2004. Scribal Practices and Approaches Reflected in the Texts Found in the Judean Desert. Studies on the Texts of the Desert of Judah 54. Leiden: Brill.

—. 2008a. The Ketiv/Qere Variations in Light of the Manuscripts from the Judean Desert. Pages 199–205 in *Hebrew Bible, Greek Bible and Qumran: Collected Essays*. Texts and Studies in Ancient Judaism 121. Tübingen: Mohr Siebeck.

— 2008b. The Textual Basis of Modern Translations of the Hebrew Bible: The Argument Against Eclecticism. Pages 92–106 in *Hebrew Bible, Greek Bible and Qumran: Collected Essays.* Texts and Studies in Ancient Judaism 121. Tübingen: Mohr Siebeck.

—. 2012. *Textual Criticism of the Hebrew Bible*. 3rd ed. Minneapolis: Fortress.

- Toy, Crawford H. 1899. A Critical and Exegetical Commentary on the Book of Proverbs. International Critical Commentary. Edinburgh: T&T Clark. Repr., 1959.
- Tur-Sinai, Naphtali H. [=Harry Torcszyner]. 1947. משלי שלמה [Proverbs of Solomon]. Tel Aviv: Yavneh.

- Ulrich, Eugene C., Jr., et al., eds. 2000. *Qumran Cave 4. XI: Psalms to Chronicles*. Discoveries in the Judean Desert XVI. Oxford: Clarendon.
- Van Leeuwen, Raymond C. 1986. Proverbs XXV 27 Once Again. Vetus Testamentum 36:105–14.
- Vawter, Bruce. 1972. Intimations of Immortality and the Old Testament. *Journal of Biblical Literature* 91:158–71.
- Verwijs, Petra. 2005. The Septuagint in the Peshitta and Syro-Hexapla Translations of Amos 1:32–2:16. *Bulletin of the International Organization for Septuagint and Cognate Studies* 38:25–40.
- Vogel, G. V. L. 1769. In A. Schultens, *Versio integra Proverbiorum Salomonis et in eadem commentaries*. Halle: Curt.
- Waard, Jan de. 1981. "Homophony" in the Septuagint. Biblica 62:551–61.
- ——. 1990. Old Greek Translation Techniques and the Modern Translator. *Bible Translator* 41:311–19.
- ——. 1991. Traduction et altération: L'interprète en quête de la fidélité. Revue d'histoire et de philosophie religieuses 71:151–68.
- ——. 1993. Metathesis as a Translation Technique? Pages 249–60 in *Tra-ducere Navem: Festschrift für Katharina Reiss*. Edited by J. Holz-Mänttäri and C. Nord. Tampere: Tampereen.
- -----. 1998. 4QProv and Textual Criticism. Textus 19:87-96.
- ——. 2004. The Exit of the Nagging Wife in Proverbs 21.9 LXX. *Bible Translator* 55:360–64.
  - —. 2005. Indices phonétiques hébreux dans et derrière le grec de la septante de Proverbes. Pages 105–17 in L'apport de la Septante aux études sur l'antiquité: Actes du colloque de Strasbourg, 8–9 novembre 2002. Edited by J. Joosten and P. Le Moigne. Paris: Cerf.
    - —. 2006. Lexical Ignorance and the Ancient Versions of Proverbs. Pages 261–68 in Sôfer mahîr: Essays in Honour of Adrian Schenker Offered by the Editors of Biblia Hebraica Quinta. Edited by Y. Goldman, A. Van der Kooij, and R. D. Weis. Supplements to Vetus Testamentum 110. Leiden: Brill.
    - —. 2007. Difference in Vorlage or Lexical Ignorance? A Dilemma in the Old Greek of Proverbs. *Journal for the Study of Judaism in the Persian, Hellenistic, and Roman Periods* 38:1–8.
- Weis, René, ed. 1993. King Lear: A Parallel Text Edition. London: Longman.
- Weitzman, Michael. 1994. Peshitta, Septuagint and Targum. Pages 51–84 in *VI Symposium Syriacum*, 1992. Edited by René Lavenant. Rome: Pontificio Istituto Orientale.

- ——. 1999. The Syriac Version of the Old Testament: An Introduction. Cambridge: Cambridge University Press.
- Wells, Stanley, Gary Taylor, John Jowett, and William Montgomery, eds. 1986. *The Oxford Shakespeare: Complete Works*. 2nd ed. Oxford: Clarendon.
- Wettlaufer, Ryan. 2012. Unseen Variants: Conjectural Emendation and the New Testament. Pages 171–94 in *Editing the Bible: Assessing the Task Past and Present*. Edited by John S. Kloppenborg and Judith H. Newman. Society of Biblical Literature Resources for Biblical Study 69. Atlanta: Society of Biblical Literature.
- Whybray, R. N. 1994. *The Composition of the Book of Proverbs*. Journal for the Study of the Old Testament Supplement Series 168. Sheffield: JSOT Press.
- Williamson, H. G. M. 2009. Do We Need a New Bible? Reflections on the Proposed Oxford Hebrew Bible. *Biblica* 90:153–75.
- Yardeni, Ada. 1997. The Book of Hebrew Script: History, Palaeography, Script Styles, Calligraphy and Design. Jerusalem: Carta.
- Yedidiah Shlomo of Norzi. 1626. *Minḥat Shai* (Masoretic notes). In *Rabbinic Bible* II, all editions.
- Yeivin, Israel, ed. 1973. *Geniza Bible Fragments with Babylonian Massorah and Vocalization.* Jerusalem: Makhor.
- Žaba, Z. 1956. *Les Maximes de Ptahhotep*. Prague: Editions de l'Académie Tchécoslovaque des Sciences.
- Zakovitch, Yair. 1975. "Assimilation in Biblical Narratives." Pages 175–96 in *Empirical Models for Biblical Criticism*. Edited by Jeffrey Tigay. Philadelphia: University of Pennsylvania Press.
- Zalcman, Lawrence. 2003. Prov 5,19c, שגיאות מי יבין. Zeitschrift für die alttestamentliche Wissenschaft 115:433–34.
- Zuntz, G. 1956. Der Antinoe Papyrus der Proverbia und das Prophetologion. Zeitschrift für die alttestamentliche Wissenschaft 68:124-84.

## 7. INDEX

Verse references show where a phenomenon is discussed in the Commentary (though the phenomenon itself may not occur in that verse). Sections in the introduction are marked by "§." Boldface indicates places where a phenomenon is given special attention.

Abrabanel §2.2.3.3.2 accents §2.2.2 additions \$1.3 agreement ≠ support §4.3.1 accents §2.2.2 afterlife 11:7, 14:23, 14:32, 24:27 allegory 5:5 Aleppo Codex §2.1.1.2 Amenemope 3:28, 22:17, 22:20 antisigma-sigma §1.3 Aristotle 6:8*a*-*c*, 23:27 atomizing changes §4.3.3 Austen, Jane §1.6.2 author \$1.4 authorship, translators' concepts of 10:1, 24:23, 30:1, 31:1 bee 6:8*a*-c "before the Lord" 3:32 "better than" proverb 19:1 Bible preface chiasm 4:7 copyist/scribal errors \$1.3 Delphic oracle 13:10 document §1.3 Edessa §3.2.1 editor §1.4

7.1. GENERAL

Egyptian wisdom 3:28, §3.1.5.0. See Amenemope Ephrem §3.2.1 equal or alternative proverbs \$3.1.8.3 evil counsel 2:16-20 exegesis \$3.143 exordia 22:17 (note) formation of the book of Proverbs \$1.3 goals §1.2 Hellenistic wisdom 2:16-20 hendiadys 1:31 hyparchetypes \$1.2 Holmes-Parsons §3.1.1 immortality 12:28 intentionalist paradigm §2.2.1, §1.4 Interludes (in Prov 1–9) §1.3 (note) Jerome §3.3 Jerusalem Crown §1.4 Kennicott and de Rossi §2.1.1.3 Lady Wisdom 5:19 lectio difficilior §2.2.3.3.5 Lectures (in Prov 1–9) §1.3 (note) Leningrad Codex B19 §2.1.1.1 lexical assumptions §4.3.2 literal translation §3.1.4 logic 14:33 mapping \$3.1.4

Masorah Parva (Mp) §2.2.3.1, \$2.2.3.3.5, \$2.2.3.3.4 meteorology of Palestine 25:23 "midrashic" variants §4.3.4.3. mimesis, mimetic translation §3.1.5.1 oral reading §2.2.3.3.4 parallelism, created and enhanced \$3.1.5.9 parentheses \$1.3 path 8:13, 8:35, 14:12 perceptual variants §4.3.4.1. Peshitta. See S-Proverbs Plato 13:10, 19:15, 24:34 Pseudo-Phocylides 6:8a-c Qumran §2.1.2 revenge 3:30 right/left 4:27a scribal additions \$1.3 Shakespeare \$1.6.1, \$2.2.2 smoothness of speech 7:21 Solomon, ascription to 10:1 stich, stichography §2.1.2, §4.2 Stoicism 8:30, 13:1, 14:30, 28:14 strange woman 2:16-20, 7:6-17, 9:12a-12c, 19:13, 20:16, 22:14, 23:33, 30:16, 31:30 "support" §4.3.1 text, definition \$1.3 textual criticism, theory and practice §1 translation, translation technique §3.1.4 unit titles 10:1 Urtext §1.1 variants: written and perceived; written and remembered \$4.3.4 versions §3 vocalization §2.2.1 wife 18:22 quarrelsome 19:13, 21:9, 19; 25:24; 27:15 wind 11:29 winking 6:13, 10:10, 16:30

wisdom, person of 5:19 Wisdom Literature §3.1.5.0 work §1.3 7.2. HEBREW (see §2) §7.2.1. Letter Interchange and Other Graphic Errors (without regard to priority) 19:28 ד/א **צ/א** 1:18, 31:28 **19:8;** 22:13 ת/א ב/⊐ 3:24, 8:24, 10:23, 11:19, 14:7, 18:15, 20:20, 20:24, 21:29, 23:32, 26:10(2x), 29:2, 29:11, 31:13 ב/ב 5:18, 11:6, 14:35, 17:23, 21:6, 22:12, 24:5 2:2, 16:7, 19:28 נ/ב 29:6 ע/ב ב/ב 4:3, 16:15, 19:29, 29:1 ר/ב 15:6, 16:27, 24:15, 27:13, 28:17 27:9 ת/ב ג 29:22 ⊂/ג 10:32, 23:28 ב/ד 16:26 ה/ד ד/ז 23:24 ד/ד 26:23 י/ד 23:24(2x) ר/⊐ 8:23, 8:24, 24:10, 25:27 ל/ד 31:6? ר/ד 10:21, 10:25, 11:8, 12:9, 12:12, 13:19, 13:20, 14:10, 14:32, 14:34, 15:14, 17:14, 19:19, 19:23, 19:28, 21:21, 22:8, 23:28, 24:10, 24:19, 25:9, 25:27, 27:13, 28:4, 28:17, 28:22 13:15, 14:22 ת/ד ח/ה 1:21, 5:11, 9:1, 10:3, 11:7, 12:18, 20:21, 21:6, 22:19, 25:2, 25:27, 29:4, 30:1, 30:9 20:16, 24:21, 27:13, 27:19 מ/ה 25:28 ר/ה

1/1 8:28, 8:29, 14:28, 21:8, **30:9** 

1/7 2:1, 3:30, 4:21, 5:16, 5:19, 6:15, 7:22, 8:10, 8:19, 8:31, 10:3, 11:3, 12:14, 12:26, 14:22(2x), 14:23, 14:27, 15:6, 17:13, 17:27, 18:20, 19:29, 20:30, 21:8, 21:10, 21:12, 21:20, 21:31, 22:19, 22:20, 23:5 (3x), 23:24(2x), 23:31, 24:12, 24:27, 25:4, 26:3, 27:16, 27:19, 27:22, 31:4. See also עניים/ענוים 1/1 18:1, 27:16 1/ר 13:1 ז/ז 24:9 1/1 21:10 15:10 ר/ז ר/⊓ 21:28, 28:23 ת/ח 2:18, 21:26, 28:12 י/ר 28:23 יו 10:19? ת/י ⊃/ב 3:15, 3:26, 6:24, 11:28, 11:30 ב/⊂ 11:15 16:21 נ/ל 24:6?, 24:8? 11:30 ס/מ 11:30 פ/מ 24:5? 15:14 פ/נ גע 5:3, 6:7, 21:6, 23:27 ע/צ 24:10 10:7? ע/ק ל/ע 3:20, **6:16**, 8:3?, 15:4 29:1 ב/פ 7.2.2. Phenomena and Topics in Hebrew<sup>1</sup> assimilation 1:15, 2:19, 4:10, 5:1, 13:2, 18:8, 19:20, 23:23, 30:18 aural 1:18, 18:9 19:7, 26:2 לא/לו

dittography 8:10, 19:28(2x), 20:6, 22:11, 23:20, 23:32, 29:2, 30:1, 30:16 near dittography 2:2, 3:6, 3:24, 6:10, 7:22, 8:10, 8:24, 11:30, 12:26, 15:6, 17:14, 19:28, 19:29, 21:8, 21:10, 22:12, 23:32, 24:15, 24:27, 26:3, 27:9, 28:4 distant dittography 4:16, 15:6, 15:15, 26:12, 28:3 distorted 25:20 division 2:7, 12:4, 13:14, 14:13, 16:30, 18:6, 22:11, 26:10, 27:9, 28:2, 28:17, 30:31 elaboration 3:16a, 9:12ab, 15:8, 16:30, 25:20a, 27:21a epexegetical addition 3:28 explication 5:22, 21:27 gloss 5:22, 24:15, 28:2 haplography 3:35, 14:14(2x), 15:6, 20:6, 21:11, 24:19, 25:27, 28:17 near haplography 1:21, 2:1, 2:18, 4:4, 5:16, 6:5, 6:24, 11:6, 17:14, 21:8(2x), 22:13, 27:19 homoi (homoiarkton and homoioteleuton) 1:16, 1:27, 3:3, 3:15, 3:28?, 8:26?, 8:29, 8:32-34?, 11:8, 11:10, 11:19?, 18:6, 21:10, 22:8 ?, 23:18, 25:9 imitation 8:35 inner-M variants 8:16, 8:26 ketiv/gere §2.2.3, 1:27, 2:7, 2:8, 3:15, 3:27, 3:28, 3:30, 3:34, 4:16, 6:13, 6:14, 6:16, 8:17, 8:35, 11:3, 12:14, 13:20, 14:21, 15:14, 16:19, 16:27, 17:13, 17:27, 18:17, 18:19, 19:7, 19:16, 19:19, 20:4, 20:16, 20:20, 20:21, 21:9, 20:30, 21:19, 21:29, 22:3, 22:8, 22:11, 22:14, 22:20, 22:25, 23:3, 23:5, 23:6, 23:24, 23:26, 23:29, 23:31, 24:1, 24:17, 25:24,

<sup>1.</sup> These are phenomena that occurred in Hebrew, not necessarily in M. The ones that happened in Hebrew but not in M are attested in the other versions (G S V T etc.).

## 7. INDEX

ketiv/gere (cont.) 26:2, 26:21, 26:24, 27:10, 27:15, 27:20, 27:24, 28:8, 28:16, 30:10, 30:18, 31:4, 31:16, 31:18 conjunction 23:24 (2x) [though usually not mentioned] grammar, yiqtol/weqatal 2:7, 13:20, 18:17, 20:4, 22:3 grammatical assimilation 30:18 grammatical variation 4:16, 19:16, 28:8 number variants 2:8, 3:27, 6:13, 6:16, 16:27, 21:29, 22:25, 24:17, 26:24, 28:16 *gere yod waw* =  $/\bar{a}w/$  (not plural) 23:3, 23:6, 24:1 ligature 13:15, 21:20(2x) metathesis 1:27, 1:32, 2:7, 2:17, 3:20, 3:21, 4:16, 8:24, 8:28, 11:26, 12:21, 12:23, 13:1, 13:11, 13:20, 14:7, 14:9, 14:17, 14:32, 14:35, 15:1, 15:10, 15:30, 16:19, 16:27, 17:20, 21:6, 21:21, 22:9, 22:21, 23:20, 23:26, 23:28, 24:15, 26:10, 26:13, 27:13, 28:17, 28:23, 29:1, 29:4 negative added 14:33 normalization 1:10, 3:28, 31:18 number variation, grammatical 1:15. *See also ketiv/gere* Occidental/Oriental readings §2.2.4; 5:19, 8:19, 8:24, 12:18, 18:19, 18:20, 21:9, 21:19, 21:31, 25:24, 28:17, 30:31 parablepsis 4:4 paleo-Hebrew 1:21, 6:5, 6:16, 10:7?, 15:4, 19:28?, 22:13, 29:6 reformulation 15:8 seghol/qames 7:22 surety 27:13 synonym variants 2:6, 5:1, 5:3, 6:2, 7:2, 10:3, 13:2, 15:14, 15:32, 16:16, 16:19, 17:3, 19:27, 23:27, 29:9, 31:18, 31:30

transfer 16:30 transposition of letters (ABC  $\rightarrow$  CBA) 11:9, 11:27, 15:4, 16:28, 28:16, 29:11 of words 22:17 vertical dittography 10:10 7.2.3. Hebrew Words Given Special Attention 14:12 ארחות /אחרית 24:11 אל/אם **2:19**, 4:10 שנות/ארחות עם/את 29:9 assayed" 8:10, 10:20 בחר כון/בין 18:15, 20:24, 21:29 19:29, 26:3 גוי/גו 28:15 דב/דאבא 22:10, 28:25 דין/מדון 23:21, 23:33 זלל 18:4 חיים 27:27 חלב 14:32 חסה 15:27a-16:9 יהוה " for /āw/ 23:6, 24:1 6:14, 18:19, 21:19, 6:14, 6:14, 18:19, 21:19, 23:29, 25:24, 26:21, 27:15 and געא, association 8:35, 10:13, מצא 18:22a as "thought" 14:14, 15:24 מעלה 2:6 מפני/מפיו 15:30 מאור/מראה **14:16,** 20:2, 26:17 מתערב/מתעבר 19:12 נהם 10:9 נודע 11:13 סוד עניים/ענוים 3:34, 14:21, 16:19 15:14 פני/פי 26:15 צלחת 19:29 שבטים 11:31 שלם תמך 3:18, 4:4-5a, 5:5, 9:12*a*-12*c*, 11:16, 29:23

7.3. Septuagint (see \$3.1)<sup>2</sup>

§7.3.1. Phenomena and Topics in G-Prov

- abstraction, person type treated as 5:5, 14:10, 15:6, 18:5, 23:28
- abstraction treated as a person type 5:19, 10:31, 11:30, 13:6, 14:1, 14:23, 28:22
- additions vis-à-vis M of one stich or more (excluding Hexaplaric insertions)<sup>3</sup> §3.1.5.7, <u>1:7ab</u>, 3:6c, 3:15bc, <u>3:16a</u>, <u>3:22a</u>, <u>3:28c</u>, <u>4:27a</u>, <u>4:27b</u>, <u>6:8a-c</u>, <u>6:11a</u>, 7:1a, <u>8:21a</u>, <u>9:10a</u>, <u>9:12a-c</u>, <u>9:18a-d</u>, <u>10:4a</u>, <u>12:11a</u>, <u>12:13a</u>, <u>13:9a</u>, <u>13:13a</u>, <u>15:1a</u>, <u>15:18a</u>, <u>17:6a</u>, <u>17:21c</u>, <u>18:22a</u>, <u>19:7ef</u>, <u>22:9a</u>, <u>22:14a</u>, 24:22a-e, <u>25:10a</u>, <u>25:20a</u>, <u>26:11a</u>, 27:20a, <u>27:21a</u>, 28:17aabc, <u>31:30c</u>.
- additions to OG in the Greek transmission 3:6d, 10:15, 10:24, 11:16, 12:26bc, 13:15b, 18:22*a*b, 26:11*a*, 26:13c
- adjective, addition 1:4, 1:9, 21:24, 29:19. *See also* moral and/or religious emphasis, addition of adjectives or nouns
- advisors and administration, royal 11:14, 15:22, 24:6
- afterlife 10:28, 11:7, 14:32
- allegory 5:19, 27:27
- alternative proverbs \$3.1.8.3, 11:25,
- 16:6, 16:8, 21:15, 27:9, 27:19, 27:22 ambiguity (in Hebrew source) \$3.1.7.3 anaphora **1**:7, 1:12, 4:12, 27:20*a*

anthropopathism, reduction **3:32**, 3:34, 8:7, 11:1, 16:5, 17:15, 20:10

- antithesis \$3.1.8.8, 1:4 1:22, 5:16, 6:1, 6:11, 6:17, 8:7, 8:28, 10:10, 11:7, 11:10, 11:20, 11:30, 13:4, 13:14, 14:17, 15:15, 15:16, 16:26, 17:4, 17:5, 17:7, 17:21, 18:22*a*, 19:6,19:22, 21:14, 21:26, 24:5, 26:20, 27:9, 29:13
- Aramaic, association with 6:14, 8:10, 9:12*a*-12*c*, 10:20, 11:26, 14:9, 26:16, 26:28?, 28:13, 30:1, 30:16, 31:3, 31:4(2x), 31:22
- Aramaic and Aramaisms: 1:3, 1:21, 32, 3:12, 6:14, 8:2, 8:7, 8:10, 9:12a–12c, 10:20, 11:26, 14:9, 17:12, 23:29, 26:16, 26:28, 28:15, 28, 29:18, 30:1, 30:16, 31:3, 31:4 (2x), 31:22
- artistic and poetic qualities 29:3
- assembly 11:13, **15:22**, 15:23, 22:10, 24:8, 26:26, 27:22, **31:23**
- assimilation 7:24, 13:4, 15:5, 15:32, 18:8, 18:20, 19:13, 30:11
- awkwardness §3.1.7.2
- balance/imbalance **3:16**, 3:23, 5:22, 10:22, 12:4
- bribery and gift giving **17:8**, 19:24, 21:14, 22:9*a*
- broad translation 16:28, 24:2, 29:4
- concealment 10:12, 10:18
- cohesiveness 10:29

compassion 17:5

- combination of indicators supporting translation \$3.1.7.5
- components that do not serve translator's purposes \$3.1.7.1
- conflation 1:27, 6:12, 6:32, 8:2, 8:24, 8:31, 10:13, 27:21, 28:6

<sup>2.</sup> Citations of Greek words in this index refer to cognate forms as well.

<sup>3.</sup> For this lemma, I use single underlining to mark components (verses or stichoi) that I consider to have a Hebrew source (as argued in the commentary) and double underlining to mark components that arose in translation or in the Greek transmission. Components in which there is no evidence either way are left unmarked.

*conflation* (*cont.*) of variants 2:2, 2:19 control §3.1.9 converse translation 1:24, 9:6, 11:7, 11:29, 13:5, 15:32, 17:4, 17:21, 20:2, 26:20 corrupt Hebrew, translation of 12:12 date \$3.1.2 diathesis 6:25, 8:23, 8:25, 9:18, 12:2, 14:2, 14:17, 14:20, 20:8, 25:26, 26:24, 28:2 disambiguation, paraphrase \$3.1.5.6, 5:4, 6:25, 7:6–17, 10:4, 10:5, 10:24, 12:25, 16:7, 17:16a, 23:27, 24:15, 25:13, 28:17, 30:1, 30:8, 31:5 division 5:13, 5:15, 11:15 double translation 1:27, 3:23, 3:31, 4:10, 5:15, 5:19, **6:3**, 6:14, 8:26, 9:6, 9:10*a*, 11:3, 11:29, 12:11*a*, 12:12, 13:11, 14:23, **15:17**, 17:1, 17:9, 17:12, 17:15, 17:18, 17:23, 17:28, 18:22*a*, 21:19, 21:24, 21:26, 22:3, 22:13, 23:7, 23:21, 23:29, 23:31, 24:7, 24:10, 24:23, 25:10, 26:11, 27:24, 28:17*a*, 29:25, 30:16, 30:28, 30:30, 30:33, 31:29 edition used §3.1.1 Egyptian Diaspora 9:18a-d, 27:8, 27:16 climate, adjustment for 10:5, 25:14, 25:23, 26:1, 31:21 elaboration \$3.1.8.6, 3:28, 4:27a-27b, 6:8*a*-*c*, 6:11*a*, 8:21*a*, 9:2, 9:10*a*, 9:12*c*, 9:18*a*-*d*, 10:4*a*, 13:9*a*, 13:13*a*, 15:18*a*, 21:3, 22:9*a*, 22:14*a*, 22:21, 24:22a-22e, 25:10c-10a, 25:20a, 26:11*a*, 26:25, 27:20*a*, 27:21*a*, 28:10 elegance §3.1.8.1 ellipsis, elimination 8:19, 8:20, 8:24, 9:9

elevation to royal status 19:6, 25:15, 28:16, 29:12, 29:12 eschatology 12:24 etymological association 9:12, 12:24, 27:6euphemism/propriety 9:3, 14:31, 30:12, 30:19 exegetical metathesis 11:27, 24:21, 29:9, 30:3, 30:10, 30:23 expansion 1:18, 9:6, 9:16, 17:1 explication, explanation 3:10, 3:26, 5:5, 5:16, 5:22, 6:1, 6:3, 6:6, 6:10, 6:12, 6:22, 6:31, 8:21, 9:12, 9:13, 10:8, 10:10, 10:25, 11:16, 11:21, 11:30, 12:5, 12:11*a*, 12:14, 13:2, 13:19, 14:19, 14:20, 15:30, 16:12, 16:22, 17:16a, 17:26, 17:27, 17:28, 18:1, 18:22, 19:7, 19:10, 19:22, 20:8, 20:10, 20:29, 21:4, 21:19, 22:13, 22:20, 23:11, 24:2, 24:14, 24:22, 25:5, 25:13, 25:14, 26:6, 26:7, 26:13, 26:23, 26:26, 27:22, 27:24, 28:3, 28:10, 28:13, 28:14, 28:26, 29:10, 29:13, 29:19, 30:7, 30:11, 30:15, 30:20, 31:13, 31:17, 31:20, 31:22, 31:23 overexplanation<sup>4</sup> §3.1.8.7, 8:21*a*, 14:20, 17:28, 19:22, 22:13, 22:14a, 25:22, 26:11, 26:19, 26:27, 27:25 external support for variants \$3.1.7.4 flexibility \$3.1.4, \$3.1.8, 11:3, 16:2, 16:22, 21:23, 25:6 freedom, apparent \$3.1.4, \$3.1.8, 5:2, 5, 6:13, 6:15, 6:22, 9:6, 11:25, 12:11*a*, 12:4, 12:11*a*, 19:27, 22:19, 25:5, 27:13, 28:2, 30:1 gapping, suppletion of 1:12, 1:33, 2:9, 5:10, **6:4** gloss 4:21, 6:3, 10:8. See also explana-

tory gloss

<sup>4.</sup> Some items in "explication" may belong in this category as well.

<sup>5.</sup> Only a few of the many examples of this phenomenon are listed here.

grammatical forms, flexible treatment of<sup>5</sup> 6:5, 6:10, 8:16, 8:31, 8:36, 9:17, 10:32, 14:9, 19:7, 22:10, 22:14, 22:15, 22:29, 23:33 number variation 1:15, 5:7, 5:15, 5:17 guessing 6:3, 7:10, 7:20, 12:26, 12:27, 12:28, 14:29, 17:10, 17:14, 18:19, 20:30, 22:23, 24:15, 26:22, 27:16, 29:21, 31:3 heart 22:18, 28:26 Hexaplaric phenomena 1:14, 1:21, 2:2, 2:3, 2:21, 3:3, 3:6, 3:15, 4:4-5a, 5:19, 6:11, 6:25, 7:25, 8:29, 8:32-34, 9:10*a*, 11:20?, 11:23, 12:14, 12:26?, 14:22, 15:18*a*, 15:33, 16:1, 17:23, 18:18, 24:5, 24:12, 26:6, 26:7, 29:25, 31:31 homiletical twist 26:9 homoiophony 11:22, 25:4, 27:7, 29:13, 30:10 imperfect 1:24 inner biblical allusions 28:21? intellectual quality, emphasis of 15:22, 24:6interpretation in translation §3.1.4, \$3.1.8.2, 6:25, 7:12, 13:12 king 14:35, 15:22, 16:12, 16:15, 20:8, 24:6, 28:15-16, 29:14, 31:8 less enthusiasm for kingship 16:12, 20:8, 24:6, 28:16 law of Moses/Torah/νόμος 4:21, 6:20, 6:23, 9:10*a*, 10:21, 13:15, 28:4, 28:7, 28:9, 28:10, 29:18 laziness 10:5, 10:26, 19:15, 19:24, 21:26, 22:16, 24:30-34 literal translation §3.1.4 literary qualities, elegance \$3.1.6.1, 5:4, 14:8, 24:7–10, 25:25; 29:3 litigation 22:10 loan guarantees 17:18 logic (apparent logic), adjustment for \$3.1.5.4, 1:24, 4:4, 4:19, 4:25,

4:27*a*, 5:4, 5:6, 5:9, 5:16, 6:6, 6:12, 6:13, 6:30, 6:31, 7:10, 8:27, 9:3, 9:9, 9:12, 9:13, **10:1**, 10:12, 11:21, 11:30, 11:31, 12:3, 12:6, 12:9,12:23,13:8, 14:3, 14:6, 14:16, 14:33, 15:21, 15:28, 15:30, 17:7, 17:17, 18:1, 18:20, 19:12, 19:21, 19:22, 22:21, 23:2, 23:33, 24:2, 24:7, 26:7, 26:9, 26:19, 27:19, 27:23, 28:2, 28:3, 29:15, 29:19, 30:3, 31:6, 31:30, 31:31 lot-casting 16:33, 18:18 meaningful proverb from errors 12:23, 13:20, 14:7, 17:14 meteorology. See climate metaphor resolving and explicating \$3.1.5.5, 1:17, 3:26, 4:27*a*, 5:6, 5:15, 6:12, 10:5, 12:6, 12:18, 14:11, 17:28; 18:3, 21:4; 21:9, 22:2, 23:2, 24:2, 25:13, 25:14, 25:20, 25:23, 26:7, 26:11, 26:23, 27:22, 27:24, 30:20, 31:10, 31:17 creating 21:9 metonymy 11:30, 12:4, 14:23 minuses §1.3 monotheism 7:1a moralism and/or religious emphasis \$3.1.5.2 άγαθός 5:2, 13:12 άληθής 1:3 άσεβής, άσέβεια 1:19 δίκαιος 3:9, 10:17, 10:22, 16:9 [G 15:29b], 16:11, 17:4, 19:22, 20:8 κακός 1:3, **1:18**, 1:28, 2:11, 3:9, 3:31, 5:2, 6:11, 10:17, 13:10, 14:25, 15:23, 16:7 (=G 15:28*a*), 16:28, 17:12, 17:26, 18:3, 19:22, 20:8, 21:22, 21:26, 22:1, 22:16, 27:21*a*, 28:20, 30:11–14 καλός 2:10, 2:11, 22:1, 22:17, 24:14, 31:11 δσίος 2:11

moralism and/or religious emphasis (cont.) other 1:4, 1:17, 1:19, 3:9, 3:18, 3:21-35, 3:30, 6:30, 9:12, 9:17, 10:4, 10:6, 10:18, 10:22, 10:15, 11:1, 11:16, 12:5, 12:24, 13:2, 13:11, 13:19, 13:23, 14:24, 15:14, 15:15, 15:16, 15:28, 16:26, 18:22a, 19:6, 19:7, 19:22, 19:24, 21:22, 21:30, 22:3, 22:14a, 23:31, 24:22, 26:11, 21:16, 27:6, 28:20, 30:16, 30:19. See also laziness. negative added 3:32, 5:5, 5:16?, 10:12, 14.13, 18:9, 27:19; 30:23. See also rhetorical question, negative added. new proverbs, formation 11:10, 11:16, 14:27, 15:18a, 16:17, 16:33, 18:4, 19:13, 20:11, 26:18, 30:32 omissions vis-à-vis M of one stich or more \$1.3, 1:16, 3:3, 4:4–5, 4:7, 7:25, 8:29, 8:33, 11:4, 11:10, 13:6, 15:31, 16:1, 16:3, 18:23-19:2, 20:14-19, 21:5, 21:18, 22:6, 23:23, 25:9, 27:27 order of G-Prov §3.1.4, 15:27; 24:22, 31:9, 31:26 ordinals 30:15, 30:21, 30:29 overwork/greed 10:22, 13:11, 23:4, 28:20, 28:22 paleo-Hebrew script 1:21, 6:5, 6:16, 10:7?, 15:4, 19:28?, 22:13, 29:6 parallelism (including enhanced parallelism) \$3.1.8.8, 1:23, 2:15, 7:8, 7:17, 8:4, 8:14, 8:16, 8:20, 14:33, 15:13, 15:20, 15:22, 15:26, 17:4, 21:14, 24:11, 27:6, 27:9, 28:15, 30:23 paronomasia 15:18*a*, 29:22 paraphrase. See disambiguation pierced jar 23:27 polis 11:10, 11:12, 24:28

possessive adjective/pronoun addition 1:13, 5:3?, 5:19, 24:14 omission 1:8, 5:11, 9:1, 14:13, 20:21, 25:8, 25:22, 29:21 poverty/poor 3:27, 6:11, 10:4, 10:15, 13:8, 13:11, 14:20, 14:23, 19:7, 19:22: 22:7, 28:3, 28:15, 29:7 preconceptions, following them astray 5:2, 5:5, 6:22, problematic phrases, adjustment or removal of 3:34, 4:16, 21:9, 25:9, **25:11**, 27:24 prophecy 29:18 Rahlfs, Septuaginta §3.1.1 refinement, enhancing dignity and propriety \$3.1.5.3, 3:34, 6:3, 9:1, 18:18, 22:3, 28:25 reformulation 14:27, 21:9, 24:33, 27:27, 28:2 relocations and transfers 8:32-34, 15:27a-16:9, 17:19, 20:9a-c, 25:1-29:27, 28:17*a*, 30:1–14 repetition of a word in parallelism 13:9; 14:20; 29:22. retribution and recompense 10:6, 13:23 rhetorical question, negative added 1:17, 27:4 revenge 3:30, 24:29 reversal of sense 10:12, 10:18, 27:19 sandwiching<sup>6</sup> 3:15, 10:24, 11:16; 12:26 sluggishness 22:29 smoothing 5:9, 8:6, 9:11 son vocative ὑιέ, G addition of 2:17, 9:12, 23:22, 24:1, 27:27 M שבני G טונ 1:8, 1:10, 2:1, 3:1, 3:11, 3:21, 4:10, 4:20, 5:1, 6:1, 6:3, 6:20, 7:1, 23:15, 23:19, 23:26, 24:13, 24:21, 27:11

6. Sandwiching is the creation of a new proverb by inserting a stich or two within an existing couplet.

M בני > G 1:15, 5:20 M plural changed to singular 5:1, 7:24, 8:32 Strange Woman 2:16-20, 5:3, 7:6-17 (intro. and passim), 9:12a-c, 19:13 subsequent transmission in G 6:8a-c, 9:18*a*-*d*, 10:15, 10:17, **10:24**, 11:16, 12:26, 15:18a, 15:27a-16:9?, 16:17?, 19:23 spiritual quality 13:1 striking/clasping the hands 6:1, 17:18, 22:26 syntactic reformulations 2:16-20, 5:15, 5:5, 5:15, 6:16, 10:13, 12:5, 13:2, 14:6, 14:16, 17:23, 17:26, 20:20, 22:20, 25:23, 30:19. See also word order, overriding tautology/banality/redundancy, and elimination of 14:8, 14:24, 16:22, 19:22, 23:34, 25:21, 31:12 tenses 31:10 theology 3:30, 3:34, 8:23, 12:24, 13:23, 22:14a, 31:8 transference/copying of lines from elsewhere in G 3:6 transition, addition of 8:21a, 9:3 translation, translation technique \$3.1.4 transposition 9:17, 17:4, 31:26 trope. See abstract, person type treated as; abstract treated as a person type; explication; metonymy variant treatment of same Hebrew 6:10, 6:11, 7:5, 10:10, 11:14, 13:5, 15:20, 18:8, 19:12, 20:2, 22:29, 24:33, 25:5, 25:24, 29:1, 30:19 variants, indicators of §3.1.5 variety, enhancement 4:16, 8:27, 20:28, 29:2, 30:19 variety, reduction 6:2 verb, addition (suppletion) 1:12, 1:33, 2:9, 5:10, 6:4, 6:31, 7:12, 8:4, 8:19, 8:20, 8:24, 25:7

verb tense 8:25
wealth 10:22, 14:24
wisdom
 as a reward 2:2, 6:22
 causing pain 14:10, 14:30
 personified 10:23
 vulnerability of 17:10, 17:12
women, reduction of attention given
 to irascible 19:13, 21:9
woman as teacher 1:8, 4:4–5, 6:20,
 31:1, 31:2, 31:28, cf. 31:26
word order, overriding 11:21, 15:26,
 16:7, 16:15, 16:24, 17:4, 17:12,
 18:19, 19:7, 20:20, 29:8, 31:1
wordplay 17:12, 26:9, 29:22

7.3.2. Greek Words Given Special Attention

άδικος 19:24 άκάθαρτος 3:32 άλλότρια 7:5 άνδρεῖος 10:4, 11:16, 15:19, 28:3, 31:11 άνεξέλεγκτος 10:17 βασίλεος 29:12 βρόχος 6:5 έμβάλλειν 11:21 έμφανῶς 8:3 έκ = 12:3 ⊐ ζηλοῦν 3:31 θεός 15:27*a*-16:9 κακός. See moral and/or religious emphasis, moralism above καλός. See moral and/or religious emphasis, moralism. above κύριος 15:27*a*-16:9 μόλις 11:31 οἴκος (pl.) 7:8, 31:27 όλιγόψυχος 14:29  $\pi \tilde{\alpha}$ , addition 3:12, 3:26, 11:20, 24:12 συνέδριον 11:13; 15:22; 22:10(2x); 24:8; 27:22 υίέ **1:8**, 1:15, 2:17

7.4. PESHITTA (see §3.2)

- אחרית/ארחות interchange 14:12, 16:25, 19:20
- abstract, person type treated as 18:11
- additions, sepuagintal \$3.2.2.7
- afterlife 14:23
- allegory 5:19, 22:15
- antithesis 28:12, 29:10; 29:18
- Aramaic, lexical association with 3:12, 6:6, 10:21, 26:11, 30:1
- assimilation 13:2, 23:31, 27:13, 28:6 bribery 17:8
- broad translation 12:23, 24:2, 24:9, 24:10, 29:14
- clarification 5:5, 5:9, 6:3, 6:10
- conflation (of Hebrew words) 1:17,
- 4:5, 5:16, 6:3, 6:6, 8:19, 9:14, 13:13*a*? converse translation 15:22
- date \$3.2.1
- dependence on G-Proverbs \$3.2.2
- diathesis (change of voice) 5:22, 8:23,
- 20:18, 24:6, 27:13

difficulty

- guess 12:12, 14:28, 19:7, 22:23, 25:24, 26:22, 29:11?, 29:21, 31:3 ignoring 3:34, **25:11**, 27:6, 27:24
- displacement 14:35
- double rendering \$3.2.2.6, 6:25, 12:4, 14:23, 17:12
- elaboration 9:12*a*-12*c*, 10:25, 26:25, \* eschatology 22:7
- etymological associations 5:3, 25:23, 28:23
- exegetical association 3:21
- exegetical metathesis 3:21, 14:12, 18:1
- explication 5:9, 8:2, 9:13, 10:14, 11:16, 13:2, 18:1, 18:22*a*, 22:13, 25:13, 25:20, 27:22, 29:1, 29:10, 29:16. *See* 
  - also metaphor, resolving
- euphemism/propriety 5:19, 31:6
- generalizing 13:13a

- grammatical forms, freedom in treatment of 6:10, 6:14, 15:2 number variation, grammatical
- 1:15; 5:9. homoiophony 11:22, 15:14?, 24:7, 24:26
- hope 2:7, 8:21, 10:24, 11:3, 13:12
- laziness 21:5, 28:20

logic (apparent), adjustment for 1:24, 4:4, 4:19, 7:12, 9:13, 10:12, 14:12, 19:25, 20:8, 21:28, 23:2, 23:7, 24:2, 24:9, 26:7, 26:9, 28:11, 28:16, 29:25, 31:8, 31:27

- materialism, muting 8:21
- metaphor, resolving and explicating 1:9, 5:3, 5:6, 5:19, 11:10, 14:11, 15:14, 23:2, 24:2, 28:4, 31:10, 31:11, 31:27. *See also* explication
- metonymy 28:23
- moral and/or religious emphasis, moralism 3:30, 6:30, 11:16, 11:24, 12:5, 16:9, 16:24, 18:11; 21:28, 22:16, 23:31, 31:6
- moralism, reduction of 6:3
- 18:8 נרגן
- negative, addition 23:2, 25:27
- noun treated as verb 6:10, 6:14
- omission, deliberate 27:27
- omission vis-à-vis G of one stich or more 1:7, 3:6, 3:15, 3:16a, 3:22a, 3:28, 4:27*a*-*b*, 5:3, 6:8*a*-*c*, 6:11*a*, 7:1*a*, 8:21*a*, 9:10*a*, 10:4*a*, 12:11*a*, 12:13*a*, 13:9*a*, 15:18*a*, 17:6*a*, 17:16*a*, 19:7, 24:22*a*-*e*, 25:10*a*, 26:11*a*, 27:20*a*, 28:17*a*
- omission vis-à-vis M of one stich or more 3:30
- origins of \$3.2.1
- overwork/greed 16:26, 23:4, 28:20
- parallelism 8:14, 16:21?, 29:10
- paraphrase 3:26, 14:17, 26:7
- pluses vis-à-vis M of one stich or more 9:12*a-c*, 9:18*a-d*, 13:13*a*,

14:35, 18:22*a*, 25:20*a*, 27:21*a*, 27:22, 27:23 possessive pronouns 1:13 poverty/the poor 14:23, 22:7, 28:6 profile of \$3.2.3 prophecy 29:18, 30:1, 31:1 reversal of sense 11:24, 26:5. See also negative, addition smoothing 7:27, 9:3, 9:11 Strange Woman 2:16, 6:14, 7:6-17, 8:1 sources, use of 4:15, 5:17, 6:3, 6:6, 6:25, 7:10, 7:20, 8:9, 9:14, 10:13, 10:14, 10:19, 11:16, 11:29, 12:17, 12:25, 12:26, 13:1, 13:2, 14:17, 14:22, 15:10, 15:22, 15:27*a*-16:9, 17:10, 18:3, 19:25, 20:5, 21:27, 22:26, 23:5, 23:7, 23:31, 24:9, 24:34, 30:31, 30:32, 31:29 synonym variants 16:27 syntactic reformulation 16:21, 19:6, 20:18, 23:28 tautology/banality/redundancy, elimination 14:8, 14:24, 16:22, 25:21 tense 7:6-17 theological modification 3:30, 14:23 transfer 4:4-5, 15:4 underworld 17:24, 18:3 variant treatment of same Hebrew 18:8, 25:5, 27:13 variation, creation or enhancement 3:24, 27:13 verb, addition 5:17, 8:4, 8:23, 27:21 waw-conjunctive 1:3 wordplay 24:22 wealth 18:11 14:12, 19:20 11:31 سک 7.5. VULGATE (see §3.3)

Aramaic, association with 31:2 bribery 17:8

difficulty, ignoring 3:34 guess 26:10 moral emphasis 11:24 negative, addition of 19:27 omission vis-à-vis M 1:14 parallelism 8:14, 8:16 Torah 30:1

7.6. TARGUM (see §3.4)

combining M and S 17:14, 27:19, 28:20 date of §3.4.2 differing from M and S 18:6, 28:16 difficulty, ignoring 3:34 double translation 31:1 edition used §3.4.1 homoiophony 24:7 independent attestation to non-M reading 9:1 materialism, muting 8:21(T<sup>L</sup>) metathesis of S 1:21 syntax, restructuring 9:16 (T<sup>L</sup>) versions, use of 7:6–17, 11:15

7.7. Hebrew Bible and Ben Sira

Genesis 1:31 8:31 2:1-2 8:31 32:21 21:14 37:27 28:21 41:6-7 10:5 Exodus 20:17 6:25 Isaiah 29:15 30:9 40:31 4:12 45:23 3:16a 48:10 10:20

53:5 26:10

Isaiah (cont.) 59:7 1:16 66:2 3:34 Jeremiah 3:3 9:13 Ezekiel 3:19 22:23 19:7 24:8 20:32 15:24 Amos 2:6 28:21 Psalms 22:8 17:14 34:17 1:12 37:11 22:7 47:2 17:18 55:22 26:23 64:6 30:9 89:38 29:14 109:15 1:12 111:10 1:7 139:13 8:23, 30:16 Iob 7:11 29:11 10:11 8:23 22:14 30:9 28:11 8:24 33:6 16:30 33:15 19:15 Qoheleth 3:21 14:32 6:9 15:30 7:4 14:10 7:26 6:5 10:16-19 12:11*a*-12 11:9 15:30

Ezra 5:48 30:16 Ben Sira \$0.3 3:18 16:1 3:21-23 25:27 4:21 26:11*a* 4:24a 3:15 6:5-17 6:1 6:18-22 3:15 11:3 6:8*a*-*c* 27:20b 6:5 27:22-23 10:10 41:14-22 26:11*a* 7.8. OTHER EARLY SOURCES, Alphabetically Acts 2:4 15:4 4:8 15:4 Bavli Berakhot 8a 18:22 17a 15:1 53b 1:21 57a 2:3 Bavli Hullin 133a 26:8 Bavli Nedarim 20b 5:15 Bavli Sanhhedrin 70b 31:4 82b 30:31 Bavli Sota 22b 24:21

Bavli Ta'anit 7a 27:17 Bavli Yebamot 63b 18:22 Epistle 140, ad Cyprian 8:22 Genesis Rabbah 49:15 13:23 Iames 3:17 25:1 Letter of Aristeas 139 28:4 Leviticus Rabbbah 10:4 24:11 12:8 31:2 Luke 1:15, 41 15:4 Mekilta Exod 20:17 6:25 Midrash Psalms 94:2 3:12 Mishnah Peah 5:6 23:10 Midrash Proverbs 5:3, 11:25, 27:19 Midrash Shoher Tov 5:18 151ab 18:22 Numbers Rabbah 10:4 30:1 13:6 19:29 15:14 24:21

Pesiqta Rabbati 33 24:11 1 Peter 4:18 11:31 Pirqe Avot 2:5 29:7 Qumran 4QProv<sup>a</sup> (102) 1:32 4QProv<sup>b</sup> (103) 13:6, 14:5, 14:32, 14:34, 15:7, 15:19, 15:28 Covenant of Damascus CD (A 11.21) 15:8 Romans 12:20 25:21 Seder Eliyahu Zutar 22 24:11 Sibylline Oracles. 4.171 11:22 Testament of Abraham A 14:11 11:22 Tanhuma Wa'era' 8 [73a] 30:1 1 Thessalonians 4:4 5:15 Wisdom 6:10 9:10a 6:21 6:1 Yerushalmi Sota 4:4 23:10 Yalqut Shimoni 26:8 7.9. AUTHORS Aejmelaeus, A. §3.1.4

Albrektson, B. §2.2.3.3.3 Barr, J. 1975 11:31 1981 \$2.2.3.3.4 1985 11:22 Baumgartner, A. 2:19 Beentjes, P. §0.3 Bonesho, C. §3.3 Boström, G. 7:6-17 Brooke, G. §1.6.1 Carmignac, J. 22:8 Chapman, R. §1.6.2 Cohen, M. §2.2.3.3.4 Cook, J. 1993 §3.2.1 1997 1:3, 1:7, 2:16-20, 6:1, 31:8 (and frequent references to CSP) 1999a 27:6, 27:13, 28:4, 31:23 2000 15:18 2002 13:16, 28:4 Chapman, R. §1.6.2 Dathe, J. A. §3.4.3 Davis, T. §1.7 Delitzsch, Franz 26:8 Dick, M. §3.1.2, 11:7, 13:15, 14:32, 28:14 Dotan, A. §2.2.3.1 Erne, L. §1.5 Eshel, H. 11:8 Fox, M. V. 2000, 2009 Anchor Bible Proverbs (ABP) §4.1 and passim 2005 \$3.1, \$3.1.8.0 Fritsch, C. §3.1.1, 1:7, 6:11 Geiger, A. 14:32 Gerleman, G. §3.1.4, 8:30, 13:10, 23:27, 24:27 Giese, R. 1990 1:4 1992a 10:22, 13:11 1992b 6:8*a*-*c*, 17:2 1993a 22:9

Gordis, R. §§2.2.3.3.1 #, §2.2.3.3.5, \$2.2.3.3, \$2.2.3.2, \$2.2.3.3.2, \$2.2.3.3.3, \$2.2.3.3.4, 2:8 Gordon, C. §3.3, 10:21, 13:23, 21:17, 26:8, 31:2 Gordon, R. 1:24 Goshen-Gottstein, M. 1960 26:10 Hamonville, D.-M. d' §3.1.4 Healey, J. 1991 §3.4.2 Heater, H. 1:7 Hendel, R. §1.3 Holmes, J. S. §3.1.4 Holmes, R., and J. Parsons §3.1.1 Hyun, C. 11:22 Joosten, J. §3.2.2, §3.2.2.6, 11:29, 14:23, 14:32, 23:31 Kaminka, A. §3.4.3, 23:27 Kennedy, J. 2:2 Klein, M. 1:24 Kuhn, G. 1:3 Lagarde, P. 1863 1:3, 1:16, 1:19, 2:19, 3:15, 3:18, 3:26, 3:32, 4:3, 5:3, 5:5, 6:10, 6:25 1873 §3.4.1 Luc, A. 22:17 Martin, G. §2.2.1 Maybaum, S. §3.4.3 McKerrow, R. §1, §2.2.2 Melammed, E. §3.4.2 Orlinsky, H. §2.2.3.3.3, 1:17, 1:24 Pinkuss, H. §3.2.2, §3.4.3 Qimhi, David 10:21 Qimhi, Joseph 11:25, 20:21, 23:2 Qimhi, Moshe 11:25, 25:23 Rofé, A. 31:30 Sa'adia 17:20 Scoralick, R. 15:27*a*-16:9, 31:8 Skehan, P. 5:19 Skehan, P. and Di Lella, A. §0.3

Snell, D. §1.4, §3.1.6 Talmon, S. 1975 2:6 1981 6:5, 19:8 Tanselle, T 1976 \$1.4 1989 \$1.3, \$1.4 1993 §1.4 Tov, E. 1979 11:22 1990 3:16*a*, 3:22*a*, 6:11*a*, 11:16, 15:27*a*-16:9 2004 \$1.3 Toy, C. 2:6 Tur-Sinai, N. §2.2.1 Ulrich, E. §2.1.2 Van der Louw, T. 6:1, 6:10, 29:3 Van Leeuwen, R. 25:27 Vawter, B. 14:32 Waard, J. de 1993 3:21, 11:27, 14:7, 24:21, 28:12, 29:9, 30:10 1998 14:32, 14:35 2006 8:3, 11:10, 17:14 2007 25:20 Weitzman, M. 10:24, 11:31 1994 \$3.4.2 1999 §3.2.2, Williamson, H. §1.6

ּןְאַׁתְּ עָלִית עַל־כָּלְנָה: 30 שֶׁקֶר הַחֵז וְהָבֶל הַיְּפִי אִשְׁה יִרְאַת־יְהוָה הַיא תִתְהַלֵּל: 31 תְּנוּ־לָה מִפְּרֵי יָדֶיָה וְיהַלְלָוּהָ בַשְׁעָרֵים מַעַשֶׂיהָ:

ם מַה־בֵּרִי וּמֲה־בַּר־בִּטְגֵי וֹמֶה בַּר־נְדָרֵי: 3 אַל־תִּתַּן לַנְּשִׁים חֵילֶדְ 2 וּדְרָכֶּידִ יֹלְמֹחוֹת מֶלְבֵין: 4 אֵל לֵמְלָבִים | לְמוֹאֵל אֵל לֵמְלָבִים שִׁתוֹ־יָיָן וֹלְרוֹזְנִים ראֵי' שֵׁכֵר: 5 פַּן־יָשִׁתֵּה וִיִשִׁבֵּח מִחָקֵק וֵישַׁנֵּה דִּין כּּל־בִּגִי־עְׂנִי: 6 תִּנוּ־שֵׁכָר לִאוֹבֵד וְיַיָן לְמֶרֵי נֶפֵשׁ: 7 וֲשָׁתֵּה וְיִשִׁכַּח ּרִישֵׁוֹ וַעֵמָלוֹ לְא יִזְבָר־עוֹד: 8 פִּתַח־פֵּידְ לָאָלֵם אֶל־דִּין בָּל־בְּגֵי חֵלְוֹף: 9 פְּתַח־פִּידָ שְׁפָט־צֶדֶק וְדִיז עָנִי וְאֶבְיִוֹז: פ 10 אֵשֶׁת־חֵיָל מִי יִמְצֶא וְרָחִׁק מִפְּגִינִים מִכְרֶה: 11 בְּטַח בָּה לֵב בַּעְלֶה וְׁשָׁלָׁל לִא יֶחְסֵר: 12 גְּמָלַתְהוּ טְוֹב וְלֹאֹ־רֶע כֹּל יְמֵי יּחַ״ֶיהִ": 13 דְּרְשָׁה צֵמֶר וּפִּשְׁתֵּים וַׁהַעַשׁ בְּחֵפֶץ כַּפֶּיהָ: 14 הֵיְתָה כְּאֲנִיּוֹת סוֹחֵר מִמֶּרְחָק תְּבִיא לַחְמֶה: 15 וַתְּקָם | בְּעוֹד לַיְלָה וַתִּתֵּן טֶָרָף לְבֵיתֶה וְׁחֹק ּלְנַעֲרֹתֶיהָ: 16 זְמְמֶה שֶׂדֶה וַתִּקָּחֵהוּ מִפְּרֵי כֵׁפֶּׁיהָ 'נְטָעָה' בֶּרֶם: ַד הֶגְרָה בְעִוֹז מְתְנֶיָה וַׁהְאַמֵּׁץ זְרֹעוֹתֶיהָ: 18 אֲעֲמָה כִּי־טָוֹב סַחְרֶה 17 ּלְאֹ־יִרְבֶּה <sup>ו</sup>בַלַּיִל' גֵרֵה: 19 יֵדֶיהָ שִׁלְחֵה בַבִּישִׁוֹר וְכַפֶּיהָ תָּמְכוּ פֵּלֶד: 20 כַּפְּה פְּרְשָׂה לֶעָנֵי וְיָדֶׁיהָ שִׁלְחָה לֱאֶבְיוֹן: 21 לֹא־תִירָא לְבֵיתָה מִשֶׁלֵג כֵּי כָל־בֵּיתָה לָבֵשׁ שָׁנֵים: 22 מַרִבַדִּים עֲשִׁתָה־לָה שֵׁשׁ וְאַרְגָמֶן לְבוּשֶׁה: 23 נוֹדֵע בַּשְׁעָרֵים בַּעְלָה בְּשִׁבְתוֹ עִם־זִקְנֵי־אֶרֶץ: 24 סָדִין עֲשְׂתָה וַתִּמְכָּר וֹחֲגוֹר נְתְנֵה לַפְנַעֲנֵי: 26' פֵּיהָ פְּתְחֵה בְחָכְמֶה וְתְוֹרַת־הֶׁסֶד עַל־לְשׁוֹנֵה: <sup>2</sup>21 עֹז־וְהָדֶר לְבוּשָׁה וֵׁתִשְׁחַק לִיוֹם אַחַרוֹן: 27 צוֹפִיָה הַלִיכִוֹת בֵּיתָה וְלֵחֵם עַצִלוּת לְא תאֹכֵל: 28 קמוּ בֶנֶיהָ וַיְאַשְׁרֵוּהָ בַּעְלָה וַיְהַלְלֶה: 29 רַבְּוֹת בֶּנוֹת עֲשׁוּ חֻיִל

31:2-29

14 דְּוֹר | חֲרָבְוֹת שִׁנְּיוֹ וְמַאֲכָלֶוֹת מֱתַלְּעָׁתִיו לֶאֱכְל עֲנִייֵם מֵאֶֶרָץ ן לַעַלוּלָה | שְׁתֵּי בָנוֹת הָב | 15 מַ וְאֶבְיוֹנִים מֵאָדֶם: ַהָב שָׁלוֹש הֵנָּה לְא תִשְׂבֵּעְנָה אַרְבַּע לא־אָמְרוּ הוֹן: 16 שָׁאוֹל וְעָׂצֶׁר רֶחַם אֱרֶץ לֹא־שָׂבְעָה מֵּיִם וְאֵׁשׁ לֹא־אֶמְרָה הְוֹן: 17 עַיָן ו ּתִּלְעַג לְאָב וְתָבֶוּז <sup>ר</sup>ּלְיִקְהַת־<sup>ז</sup>אָם יִקְרָוּהָ עֹרְבֵי־נָחַל וְיאכְלָוּהָ בְנֵי־ פּ 18 שְׁלֹשֶׁה הֵמָּה נִפְּלְאָו מִמֶּנָו <sup>ו</sup>וְאַרְבַּע<sup>ּ</sup> לְא נֵשֵׁר: יְדַעְתִּים: 19 דֶּרֶךְ הַלֶּשֶׁר | בַּשָּׁמַיִם דֶּרֶךְ נְחָשׁ עֲלֵי צְוּר דֶרֶדְ־אֶנֵיֶה ַבְּלֶב־יֶם וְדֶרֶדְ גֶּבֶר בְּעַלְמֶה: 20 בֵּז | דֶּרֶדְ אִשְׁה מְנָאָפֶת אֲכְלָה ּוּמֶחֲתָה פֵּיהָ וְאָמְרָה לְאֹ־פְעַלְתִּי אֶוֶן: 21 הַחַת שֶׁלוֹשׁ פ ַרֶגְזֶה אֱרֵץ וִתַחַת אַרִבָּע לא־תוּכֵל שָׂאֵת: 22 הַחַת־עֵבֵד כִּי יִמְלָוֹד וְּנָבָּל בֵּי יִשְׂבֵּע־לֶחֶם: 23 תַּחַת שֲׂנוּאָה בֵּי תִבְּעֵל וְשִׁפְחָה בִּי־ 24 אַרְבֶּעָה הֵם קְטַנֵּי־אֶָרֶץ וְהֵמָה תִירַשׁ גְּבִרְתֵּה: פ ְחַכָמִים מְחֻבָּמִים: 25 הַגְּמָלִים עֵם לֹאֹ־עָז וַיָּכִינוּ בַקַּיִץ לַחְמֶם: 26 שְׁפַּגִּים עַם לא־עָצָוּם וַיָּשָׂימוּ בַפֶּלַע בֵּיתֶם: 27 מֱלֶך אֵין לְאַרְבֶּה וַיֵּצֵא חֹצֵץ כָּלְוֹ: 28 שֵׂמָמִית בְּיָדַיִם תְּתַפֵּשׁ וְהִיא בְּהֵיכְלֵי 29 שְׁלֹשֶׁה הֵמָּה מֵיטֵיבֵי צֶעַד וְאַרְבָּעָה מֵיטֵבֵי מֵלֵדְ: פ לֶכֶת: 30 לַיִשׁ גִּבְּוֹר בַּבְּהֵמֶה וְלֹא־יְשׁוּב מִפְּגֵי־כְׂל: 31 זַרְזִיר מְתְנַיִם אוֹ־תֶיִשׁ וֹּמֶׁלֶך ואַל־קוּם' עִמְוֹ: 32 אִם־נְבַלְתָּ בְהִתְנַשֵּׂא וְאָם־זַמוֹתָ ַיֶד לְפֶה: 33 כֵּי מֵיץ חָלָב יוֹצֵיא חֶמְאָה וְמִיץ־אֲף יוֹצִיא דֶם וּמֵיץ אַפַּיִם יוֹצִיא רִיב: פ

דִּבְרֵי לְמוּאֵל מֶלֶך מֵׁשָׂא אֲשֶׁר־יִסְרַתּוּ אִמְוֹ: 31:1 🛛 אַבְרֵי לְמוּאֵל מֶלֶד

30:14 אביונים M S (השביונם [ (כשביא) אביונים M S (השביונים [ (השביונים מים מים אביונים) אביונים (משביא) (orth or gram) || 30:16 מים מים [ (הביא M מים מים (מים מים M מים מים M מים מים  $M^{\rm Erfurt 1}$  מים מים מים  $M^{\rm ALY}$  (vocal) || 30:17 לִיְקָהָֿת־ M<sup>Erfurt 1</sup> אַל־קום M^{\rm ALY} (vocal) || מים מים M<sup>K</sup> אַל־קום M<sup>K</sup> אַל־קום [ M<sup>ALOc</sup> (div)

מַעֲגֶה: 20 חָזִׁיתָ אֲישׁ אֲץ בִּדְבָרֵיו תִּקְוָה לִכְמֵיל מִמֶּנוּ: 21 מְפַנֵּק מִנִּעַר עַבְדֵּוֹ וְאַחַרִיתׁוֹ יִהְיֶה מְנִוֹן: 22 אִישׁ־אֲף יְגֶרֶה מְדֵוֹן וּבָעַל חַמֶּה רַב־פֶּשַׁע: 23 גַּאַוּת אֲדָם תַּשְׁפִּילֵנוּ וּשְׁפַל־רוּחַ יִתְמִך כְּבִוֹד: 24 חוֹלֵק עִם־גַּנְּב שוֹנֵא נַפְשָׁוֹ אָלֶה יִשְׁמַע וְלָא יַגְּיד: 25 חֶרְדַּת אָדָם יִתֵּן מוֹלֵש וּבוֹטֵח בַּיהוָה יְשָׁגֵב: 26 רַבִּים מְבַקְשִׁים פְּנֵי־מוֹשֵׁל וֹמֵיְהוָה מִשְׁפַּט־אִישׁ: 27 תּוֹעַבַת צַדִּיקִים אֵישׁ עֶוֶל וְתוֹעַבַת רְשָׁע יְשַׁר־דֵּרֶדָ:

זָסָמַשָּׁא נְאָם הַגְּבֶר לָאִיתִי אֵל לָאִיתִי אֵל וָאָכָל<sup>ז</sup>: 2 כִּי בְעַר אָנֹכִי הַמַּשָּׁא נְאָם הַגְּבֶר לָאִיתִי אֵל לָאִיתִי אֵל וָאָכָל<sup>ז</sup>: 2 כִּי בְעַר אָנֹכִי מַאִישׁ וְלְא־בִינַת אָדָם לִי: 3 וְלָא־לָמַדְתִּי חָכְמֵה וְדַעַת קִדשׁׁים אַדְע: 4 מִי עָלֶה־שָׁמִים וַיַּבִּד מֵי אֶסַף־רֹּוּח ו בְּחָפְנָיו מֵי צֵרַר־ מַיִם ו בַּשִּׁמְלָה מֵי הַקִים כָּל־אַפְסֵי־אָרָץ מַה־שְׁמִו וּמַה־שָׁסִו וּמַר מַיִם ו בַּשִּׁמְלָה מֵי הַקִים כָּל־אַפְסֵי־אָרָץ מַה־שְׁמוּ וּמַה־שָׁסִי כִּי תַדְע: 5 כְּל־אִמְרַת אֶלְוֹהַ צְרוּפֶה מְגֵן הוּא לַחֹסִים בְּוֹ: 6 אַל־ כְּים בְּעִרדְבְרָרִו בָּוֹבָים כָּל־אַפְסֵי־אָרָים מָגֵן הוּא לַחֹסִים בְּוֹ: 6 אַל־ מָּים ו בַּשִּׁמְלָהִי מַאִתְדָ אַל־תִמְנַע מְנָזוּה יְבְטָרֶם אָמוּת: 8 שְׁוּתוּ הַרְחֵק מָלִהִי מַאִתְדָ אַל־תִמְנַע מְנָזוּה יְבְטָרֶם אָמוּת: 8 שְׁוּא וּדְבַר־כָּזָ שָּאַלְתִי מַאִתְדָ אַל־תִמְנַע מְנָזּי בְּטָרֶם אָמוּת: 9 שָׁמָים הַרְחֵק מָשָּרָתי מַאִתְדָ אַל־תִמְנַע מְנָזּי בְּטָרֶם אָמוּת: 9 שָׁמָים הַרְחֵק מָשָּׁרָי הַאָּתַרָת אָלָוּתִי מַי יְהוּתָה וּפָּרָרָי הַרְחֵק מָלִים מָאָנָר אַל־תַמְיָנָי רָבָין הַעָּשְׁמָנִי 11 הָםחַשְׁתִי 10 אַבָּרָתִי מִי יְהָנָי הָיבָרָי וֹמָרָים גָּוּרָי שְׁם אָלְהִי ביי הַאָּלָרָי מָאינָת וּאָמָן אָיר אַליתַיָּמָין מָי הַרְחֵי מָי יְהוֹתוּי מָאינְתָי מַאָיתִי זיישָּים בּרָין וּדָים מָיָרָים בָּרָרָיָים הָנָין הַיָּבָריי הַיָּם אָעָהיי הַיּשְׁמָי הַקּיין הָרָשָּטִיין זיין מָאָימָין וּמָבּיין וּמָמָיין וּמָין יָשָּמָין אָמָרייָיָרָין בָּרָייָין מָירָיים מָיוּין גָּין בּיין יָהוּעָיין מָאַיָּכָין וּמָנִין וּזיין הָייָרָם מָנָגָין אָירָיזיים מָיוּין אַיריין אָיין הַיָּעָן מָייָרָין אָירייָרָין אָנָין מָעָין מָעָיין מַיָּרָין בָּרָין בּיין הַיָּשָּיין הַיָּין מָירָין מָעוּיין מָין יוּמָרָיין גָעריין אָנָין גָיין מָעוּיןין בּיים בּינוּין וּינִין מָיןייין זיין דָּייןין מָייןיין מָייןיןין מָעָיייין בּייין בָּין ייייוּקָין מַיּרָין מָייןין מָיין וּיין מָעָןין מָעָיין מָיין וּיין בּיין בָּיין ייָיוּקָין מָייןיין מָייןיין מָייןיין מָייןיןין מָייןיין בּיין מָיין מָיין וּייןייןין מָיין וּייין בָיין הָייָיןיןיייןין מָייןיין מָיין בּיין בָיין וּין ג

38\*

לְשׁוֹן: 24 גּוֹזֵל | אָבִיו וְאִמֹו וְאֹמֵר אֵין־פֶּשַׁע חָבֵר הוּא לְאֵישׁ מַשְׁחִית: 25 רְחַב־גֶפֶשׁ יְגֶרֶה מֶדֶוֹן וּבוֹטֵח עַל־יְהוֶה יְדֻשֶׁן: 26 בּוֹטֵח בּּלִבּוֹ הוּא כְסֵיל וְהוֹלֵך בְּחָכְמָה הוּא יִמְלֵט: 27 נוֹתֵן לָרָשׁ אֵין מַחְסֶוֹר וּמַעְלֵים עֵׁינְיו רַב־מְאֵרְוֹת: 28 בְּקָוּם רֲשָׁעִים יִפֶּתַר אָדֶם וֹבִאָבְדָם יִרְבְּוּ צַדִּיקִים: 29:1 אֵישׁ תּוֹכָחוֹת מַקשׁה־עָׂרֶף פֵּתַע יִשָּׁבִר ּוְאֵין מַרְפֵּא: 2 בִּרְבַוֹת צֵדִיקִים יִשְׂמַח הָעֶם וּבִמְשִׁל רְׁשָׁע יֵאָנַח עֶם: זוֹנוֹת יְאַבֶּד-הְוֹן: 4 מָׂלֶד אָישׁ־אֹהֵב חֲכְמָה יְשַׂמַּח אָבֶיו וְרֹעֶה זוֹנוֹת יְאַבָּד-הְוֹן: 4 מָׁלֶּד רַמִשְׁפָּט יַעֲמִיד אָָרֶץ וְאָישׁ תְּרוּמִוֹת יֶהֶרְסֶנָּה: 5 גָּבֶר מַחַלִיק עַל־ רַעֶהוּ רֵשֶׁת פּוֹרֵשׁ עַל־פָּעָמֵיו: 6 בִּפֵשַׁע איש רֶע מוֹקֵשׁ וְצַדִּיק יָרְוּן וִשָּׁמֵחַ: 7 יֹדֵעַ צֵּדִיק דֵּין דַלָּים רָשָׁע לא־יָבֵין דֱעַת: 8 אַנִשֵׁי לֱצוֹן יַפִּיחוּ קִרְיָה וַׁחֲכָמִים יָשִׁיבוּ אֶף: 9 אִישׁ־חָכָּם גַשְׁפָּט אֶת־אֵישׁ אֶוֵיל וְרָגֵז וְשָׁחַק וְאֵין נֶחַת: 10 אַנְשֵׁי דָמִים יִשְׁנְאוּ־תָם וִישָׁרִים יְבַקְשָׁוּ נַפְשִׁוֹ: 11 כָּל־רְוּחוֹ יוֹצֵיא כְסֵיל וְחָכָם בְּאָחוֹר <sup>ר</sup>ַחְשְׂכֶנָה<sup>ז</sup>: 12 מְשֵׁל מַקְשֵׁיב עַל־דְּבַר־שֶׁאֶֶר בֶּל־מְשָׁרְתָיו רְשָׁעֵים: 13 רָשׁ וִאֵישׁ תִּכָבֵים נִפְגֲשׁוּ מֵאִיר־עֵינֵי שְׁנֵיהֵם יְהוֶה: 14 מֱלֶך שׁוֹפֵט בֶּאֱמֶת דַּלֵּים כִּסְאוֹ ּלָעִד יִבְּוֹן: 15 שֵׁבֶט וֲתוֹכַחַת יִתֵּן חְכְמֶה וְנַעַר מְשֵׁלָּח מֵבִישׁ אִמְוֹ 16 בִּרְבְוֹת רֲשָׁעִים יִרְבֶּה־פָּשַׁע וְצַדִּילִים בְּמַפַּלְתָּם יִרְאוּ: 17 יֵמֵּר 18 בְּאֵין חֲזוֹן יִפְּרַע בִּנְד וִינִיחֶד וְיִהֵּן מַעֲדַגִּים לְנַפְשֶׁד: פ עֶם וְשֹׁמֵר תּוֹרָה אַשְׁרֵהוּ: 19 בֵּדְבָרים לֹאֹ־וַוָּסֶר עָבֶד בִּי־יָבִין וְאֵין

משלי

מַטָר סׁמֵף וְאֵין לֶחֶם: 4 עֹזְבֵי תְוֹרָה יְהַלְלָוּ רָשֶׁע וְשֹׁמְרֵי תוֹרָה יִתְגְרוּ

ַ אַנְשֵׁי־רָע לֹא־יָבֵינוּ מִשְׁפֵּט וּמְבַקְשֵׁי יְהוָה יָבִינוּ

metath  $\neg \neg \rightarrow \neg$ )

כְּל: 6 טְוֹב־רָשׁ הוֹלֵךְ בְּתֻמֵּוֹ מֵעִקֵּשׁ דְּרָכַּיִם וְהוּא עָשִׁיר: 7 נוֹצֵר

ּתִוֹרָה בֵּן מֵבֶין וִרֹעֵה זְוֹלִלִים יַכִלִים אָבִיו: 8 מַרְבֵּה הְוֹנוֹ בְּגֵשֶׁדְ

ןּתַרְבֶּית ּ לְחוֹגֵן דַּלִּים יִקְבְּצֶנּוּ: 9 מֵסֵיר אָזְנוֹ מִשְׁמִע תּוֹרֶה גַּם־

12 בַּעֲלָץ אֲדִיקִים רַבֶּה תִפְאָרֶת וּבְקוּם רְשָׁעִים יְחֻפֵּשׂ אָדֱם:

13 מְכַסֶּה פֵּשְׁעָיו לָא יַצְלֵיחַ וּמוֹדֶה וְעֹזֵב יְרָחֶם: 14 אַשְׁרֵי אָדָם מִפַּחֵד תָּמֵיד וּמַקִּשֵׁה לְבֵּוֹ יִפְּוֹל בִּרָעֵה: 15 אַרִי־נְהֵם וִדְב שׁוֹקֵק

ּתְפִלְתוֹ תּוֹעֵבֶה: 10 מַשְׁגֶה יְשָׁרִים | בְּדֶרֶךְ רָע בִּשְׁחוּתוֹ הְוּא־יִפָּוֹל

וּתְמִימִים יִנְחֲלוּ־טְוֹב: 11 חֶבֶם בְּעֵינְיו אֵישׁ עָשֵׁיר וְדָל מֵבִין יַחְקָרֶנּוּ:

ּמֹשֵׁל רְׁשָׁע עַל עַם־דֶּל: 16 נְגִיד חֲסַר הֻּבוּנוֹת וְרָב מַעֲשֵׁקֶוֹת ישְׁנֵא

בָּצַע יַאֲרִידְ יָמֵים: פּ 17 אֲדָם עָשֵׁק בְּדַם־גָפֶשׁ עַד־בּוֹר 🖕

ּיְנוּס אַל־יִתְמְכוּ־בְוֹ: 18 הוֹלֵךְ הֵמִים יוָּשֵׁעַ וְגָעְקָשׁ דְׁרָכַּיִם יִפְּוֹל

ּבְּאֶחֶת: 19 עֹבֵד אֲדְמָתוֹ יִשְׂבֵּע־לֶחֶם וּמְרַדֵּף בֹּקִים יִשְׂבֵּע־רִישׁ:

יַדַע כִּי־חֵסֶר יִבֹאֵנּוּ: 23 מֿוֹכִיַח אָדֵם <sup>ו</sup>אַחֵר<sup>י</sup> חֵז יִמְצָא מְמַחֵלֵיק

20 אֵישׁ אֱמוּנוֹת רַב־בְּרָכִוֹת וְאָץ לְהַעֲשִׂיר לָא יִנְקָה: 21 הַכֵּר־פָּנִים

לאֹ־טָוֹב וְעַל־פַּת־לֶּחֶם יִפְשַׁע־גְּבֶר: 22 נְבֶהָל לַהוֹן אֲישׁ רַע עָיָן וְלְא־

28:4 ייתגרו [ איתגרו ] איתגרו [ איתגרו ] ייתגרו [ איתגרו ] יעגרו [ איתגרו ] יעגרו [ יעגרו ] יעגרו [ יעגרו ] יעגרו ]

בם:

פ

לִבֶּי וְאָשִׁיבָה חֹרְפִי דָבֵר: 12 עָרָוּם רָאָה רָעֵה נִסְתָּר פִּתָאיִם עָבְרָו נֶעֲנֵשׁוּ: 13 קַח־בֶּגְדוֹ כִּי־עָרֵב זֶר וּבְעַד 'נְכְרֵים' חַבְלֵהוּ: 14 מְבְּתֵד ַרַעֵּׁהוּ | בְּקוֹל גָּדוֹל בַּבָּּקֶר הַשְׁבֵּים קְלָלָה תֵּחָשֶׁב לְוֹ: 15 דֶלֶף טוֹרֵד בִּיוֹם סַגְרֵיר וָאֵשֶׁת 'מְׁדוֹנִים' נִשִׁתָּוֶה: 16 'צֹפְנֵה' צֵפַן־רָוּח 'וּשָׁמוֹ יְמִינִי יִקָּרֵא<sup>י</sup>: 17 בַּרְזֶל בְּבַרְזֶל יֶחֵד וְאִישׁ יַחַד פְּנֵי־רֵעֵהוּ: 18 נֹצֵר הֵאָנָה יאכַל פִּרְזֶה וְשֹׁמֵר אֲדֹנֵיו יְכָבֶּד: 19 כַּמַּיִם הַפְּנִים לַפְּנִים כֵּן לֵב־הָאָדָם לָאָדֶם: 20 שְׁאוֹל וַאֲבַדֹּה לָא תִשְׂבֵּעְנָה וְעֵינֵי הָאָדָם לָא תִשְׂבֵּעְנָה: 21 מַצְרֵף לֵכֶּסֶף וְכִוּר לַזְהָב וְאִישׁ לְפִי 'מְהַלְלֵיוּ': 22 אֵם תִּכְתִּוֹשׁ־אֶת־הָאֱוִיל | בַּמַּכְתֵּשׁ בְּתַוֹדְ הָרִיפוֹת בֵּעֱלֶי לֹא־תָסוּר מֵעָלָיו :ביִדִע הֻדַע פְּגֵי צֹאנֶדְ שִׁית לִבְּדָ לַעֲדָרִים אַוַּלְתוֹ: פ בּי לְא לְעוֹלֵם חֶסֶן וְאִם־נֵוֶר לְדִוֹר רְדִוֹר<sup>י</sup>: 25 גְּלֵה חֲצִיר וְנִרְאָה־ 24 דֶשָּׁא וְנָאֶסְפֿוּ עִשְׂבִוֹת הָרֵים: 26 כְּבָשִׂים לִלְבוּשֶׁדְ וּמְחֵיר שְׁדָה ַעַתּוּדִים: 27 וְדֵי | חֲלֵב עִזִּים לֱלַחְמְדָ לְלֶחֶם בֵּיתֶדְ וְחַיִּים לְנַעֲרוֹתֶידָ: נָסוּ וִאֵין־רֹדֵף רָשֶׁע וְצַדִּילִים כִּכְפִיר יִבְטֶח: 2 בִּפֵשָׁע אֱרָץ 28:1 רַבְּים שָׂרֵיהָ וּבִאָדֵם מֵבְין יֹן כֵּן יַאֵרִידְ: 3 גֵּבֶר רֲשׁ וִעֹשֵׁק דַלָּים

אֶל־פִּיו: 16 חָבָם עָצֵל בְּעֵינֵיו מִשִּׁבְעָה מְשִׁיבֵי טֱעַם: 17 מַחַזִיק בְּאָזְנִי־כֶלֶב עֹבֶר מִׁתְעַבֵּר עַל־רֵיב לְאֹ־לְוֹ: 18 בְּמִתְלַהְלֵהַ הַיֶּרֶה זִלָּים ּחִצִּים וְמֶוֶת: 19 בֵּן־אֶישׁ רִמְּה אֶת־רֵעֵהוּ וְאָמַר הֲלָא־מְשַׂחֵק אָנִי 20 בִּאֲפָס עֵצִים תִּכְבֵּה־אֲשׁ וּבָאֵין נְׁרָגָּן יִשִׁתִּק מָדְוֹן: 21 פֶּחָם יָלָגָחָלִים וְעֵצֵים לְאֵשׁ וְאֵישׁ 'מְדוֹנִים' לְחַרְחַר־רֵיב: 22 דִּבְרֵי נֵרְגָן כִּמֶתלַהַמֵים וְהֵם יָרְדָוּ חַדְרֵי־בֶטֵן: 23 בֵּסֵף סֵיגִים מְצֻפֶּה עַל־חֶרֶשׁ שְׂפָתַיִם 'חֲלָקֵים' וְלֶב־רֶע: 24 בֵּשְׂפָתָו יִנְּכֵר שוֹגֵא וּבְקִרְבּוֹ יָשֶׁית מִרְמֶה: 25 בִּי־יְחַנֵּן קוֹלוֹ אַל־תַּאָמֶז־בָּוֹ כִּי שֶׁבַע ּתּוֹעֵבְוֹת בְּלִבְוֹ: 26 <sup>ר</sup>מְכַפֶּה<sup>ּ</sup> שִׂנְאָה בְּמַשְּׁאֵוֹן תִּגְּלֶה רָעָתוֹ בְקָהֵל: 27 בְּרָה־שֲׁחַת בְּה יִפְּל וְגְלֵל אֶׁבֶן אֵלְיו תָּשְׁוּב: 28 לֲשׁוֹן־שֶׁקֶר יִשְׂנָא דַבֶּיו וּפֶה חְלָק יַעֲשֶׂה מִדְחֶה: 27:1 אֵל־הֵּתְהַלֵּל בְּיוֹם מָחֶר בִּי לאֹ־ :גַּדַע מַה־יֵּלֶד יְוֹם: 2 יְהַלֶּלְדָּ זָר וְלא־פֵידָ גְׁכְרִי וְאַל־שְׂפָתֶיד 3 כְּבֶד־אֶבֶן וְגֵטֶל הַחֵוֹל וְכַעַס אֶׁוֹיל כְּבֵד מִשְׁגֵיהֶם: 4 אַכְזְריּוּת חֵמָה וְשֶׁטֶף אֱף וּמֵי יַׁשְמֹד לִפְנֵי קִנְאֶה: 5 טוֹבָה תּוֹכַחַת מְגַלֶה מֵאַהֲבָה מְסֶתֶּרֶת: 6 גֵאֱמְנִים פִּצְעֵי אוֹהֶב וְנַעְתָּרוֹת נִשִׁיקוֹת שוֹנֵא: 7 גֵפָשׁ ַשְׂבַעָה תִּבְוּס גָֹפֶת וְגֶפֶשׁ רְׁעֵבָּה כָּל־מֵר מָתִוֹק: 8 כֵּצִפּוֹר נוֹדֶדֶת מִז־ קנֶה בֵּן־אִישׁ נוֹדֵד מִמְקוֹמְוֹ: 9 שֶׁמֶן וּקְטֹרֶת יְשַׂמַּח־לֵב וּמֶתֶק יֹרֵשָׁה' ַמֵעֲצַת־נֱפֶשׁ: 10 רֵעֲדֶ וֹוְרֵעֶה' אָבִידְ אַל־תַּעֲזֹב וּבֵית אָחִידְ אַל־ תֵּבוֹא בִּיוֹם אֵידֶדְ טְוֹב שָׁבֵן קָרוֹב מֵאֶח רָחִוֹק: 11 חַכֵם בֵּנִי וִשַּׁמֵח

26:21 מדונים  $M^{K}$  מדונים (dial? norm) || 26:23 הלקים  $M^{Q}$  מדונים ( $M^{K}$  מדונים  $M^{K}$  מדונים ( $M^{K}$  מדונים  $M^{K}$ ) א דלקים (graph  $\to \to M^{K}$ ) ( $\pi \to M^{K}$ 

עַל־פָּנַת־גָג מַאָּשֶׁת <sup>ר</sup>ְמְדוֹנִים<sup>7</sup> וּבֵית חֶבֶר: 25 מַיִם קֵרִים עַל־גָפָשׁ שְׁיַפֶּה וּשְׁמוּעֵה טוֹבְה מַאָּרֶץ מָרְחֵק: 26 מַעְיֵן גַרְפָּשׁ וּמָקוֹר מִשְׁחֶת שְׁיַפֶּה וּשְׁמוּעֵה טוֹבְה מֵאָרֶץ מָרְחֵק: 26 מַעְיֵן גַרְפָּשׁ וּמָקוֹר מְשִׁחָת מְכַבֶּד<sup>7</sup>: 28 עַיר פֵּרוּצָה אֵין חוֹמֶה אִישׁ אֲשֶׁר אֵין מַעְצֵר לְרוּחְוֹ: 1:30 פַּשֶׁלָג | בַּלַּיִץ וְכַמְּטֶר בַּקָצֵיר כֵּן לֹא־נָאוֶה לִכְסֵיל כָּבוֹד: 2 בַּצִּפּוֹר לְנוּד כַּדְרָוֹר לְעֵוּף כֵן קִלְלָת חִוֹמֶה אִישׁ אֲשֶׁר אֵין מַעְצֵר לְרוּחְוֹ: 2 בַּצִּפּוֹר לְנוּד בַדְרָוֹר לְעֵוּף כֵן קַלְלָת חִוֹבֶם לָא תָבְאוּ הּלְסֵיל כָּבוֹד: 3 בַּצִפּוֹר לְנוּד בַדְרָוֹר לְעֵוּף כֵן קַלְלָת חִוֹבָם לָא תָבְאוּ הַלָּחוֹ פָּן־ מָתֶג לַחַמְוֹר וְשָׁבָט לְגוּ פְסִילִים: 4 אַל־תַעַן בֵּסיל פָאוּלַתוּ פָן־ מְשָׁעָה לַחַמְוֹר וְשָׁבָט לְגוּ פְסִילִים: 4 אַל־תַעַן בָּסיל פָאוּלַתוּ פָןד מִשְּעָה לַחַמְוֹר וְשָׁבָט לְגוּ פְסִילִים: 4 אַליתַעַן בָּסיל פָאוּלַתוּ פָן־ מְסָבֶבָּד: 9 חְזֹם עָלֶה בְיַדישִׁכָּוֹים: 8 בִּצְרָזר אֶבָן בְּכַיל מִפְּפָּח וֹמְשָׁל בְפֵי כְסילִים: 8 בִּצְרָים בְּרָים בְּיָרָים וּזוָן לְכְסֵיל בָּבִיר: פָּחוֹם עָלֶה בִיד-שָׁכָּוֹים: 8 בִּצְרָזר אֶבָּירִים בְּיחוֹלָל מִמָּשָּר וּמָשָּר וּמָשָׁל בְפִי כְסילִים: 11 בְּכָיי מָמָשָּר הָאָנָין הוּמָט שׁתָה בַיִד-שָׁכּוּב יין בּבָירוּים בְּרוּים הָאָיָין הָמָיָיי בּאַיָּר אָיַין אַבָּר עָיבוּמָין וּמָשָׁל בְמָי לַשָּמָין בַיַין גָין אָבָיןין הַבָּירָי אָאָשָר אָמָין בּבָרָים בְיּצָין אַנָין בּיין בּיָין אַבָּין הַישָּין הַיָּתוֹין בּיָין בָּיָין בַיִין בָּעָיָר אָמָין בּייָין בּאַבָּין הָעַצָּי שָּרָין בַּיָרָי אוּיָר בְיָרָי בָיָין בָיָשָּין בּיָיוּים גַין הַיָין בּירָין גַייים בַּרָיןים בּירָים בּייָרָין בּעָבין בּיין בּאָין בְישָרָין בִירָים בָיים בְיוּן בָּשָּים בְיוּין בּייים בּאַיין הַעָין בּייָין בּייןין בַיןיין בּישָרָין בּייין בּייןין בּייןיין בּייןין בּיין בּייָין בּיין בּייִין בּייִין בִיין בָיין בִיין בָייין בָיין בָיין בּיין בָיין בּיין בַיָּיןין בּייָיןין בָייָין בּיין בָּיין בָּין בָרָין בָיין בָּיין בּיין בָיןין בּייןן בָייָין בָין בָיוּין בָיין בָיין בָיין בָיין בָיין בּיין בָיין בָיין

25:24 מדונים  $M^{K}$  (לא מדונים  $M^{K}$  (לא מדונים  $M^{K}$  (לא מדונים  $M^{K}$  (מו לא מדונים)  $M^{K}$  (מו לא מו לא מדונים)  $M^{K}$  (מו לא מדיע) V (qui scrutator est) (M S σ' θ' V: graph  $\rightarrow \rightarrow n$  (מו לא מנ לא מטעט)  $M^{K}$  (מו לא מו לא מו

25:2 || (ה → ח M ≈ S (خ , الت ( ( ל , הקר ( ( ל , الت M ≈ S ( ( א , الت M ≈ S ( ) ) ) ) ) ( א בלי M S ( (מאני (graph י → י) || 25:9 ריבד ריב את רעד MSV] (מאני MSV] > G (homoi רעך רעך) || רעד M S (היהוה) אוסור G (ἀναχώρει) (graph ד → ר || 25:10 fin ] + 4 stichoi G (10c-10c) (elab) || 25:19 מועדת מבטח  $M \approx S$  (הא הבה השהיה השהיה) (cf. > G) (distorted dittog) || fin ] + 2 stichoi G (20*a*) (elab)

שְׁנִוּן אֵישׁ עֹנֶה בְּרֵעֵׁהוּ עֵד שְׁקֶר: 19 שֵׁן רִשָה וְרָגָל מוּעָדֶת מִבְטָח בוֹגֵד בְּיִוֹם צָרֶה: 20 <sup>ר</sup>ֹחְמֶץ עַל־גָתֶר וְשָׁר בַּשִּׁרִים עַל לֶב־ 21 אִם־רָעֵב שַׂנַאָד הַאֲכִלֵהוּ לֶחֶם וְאָם־צְׁמֵא :רַע פ הַשְּׁמֵהוּ מֵיִם: 22 כֵּי גֵחָלִים אֲתָּה חֹתֵה עַל־ראשוֹ וַיהוָה יִשַׁלֵּם־לֶדְ: 23 רוּחַ צָפוֹן תּחוֹלֵל גָשֶׁם וּפָגִים רָזְעָמִים לְשִׁוֹן מֶתֶר: 24 טוֹב שֶׁבֶת

ד הֹקַר רַגִלִד מִבֵּית רַעֵּד פָּן־יִשִּׁבָעַד וּשִׂנַאָד: 18 מֵפִיץ וְחֵרֶב וְחֵץ 17

לִפְנֵי־מֶלֶדְ וִיְכְּוֹן בַּאֲדֶק כִּסְאָוֹ: 6 אַל־תִּתִהַדְּר לִפְנֵי־מֵלֶדְ וּבִמְקוֹם ּגְדלִים אַל־תַּעַמִד: 7 כִּי טִוֹב אֲמָר־לְדָּ עֵלֵה הֵנָּה מֵהַשְׁפִּילְדָ לִפְגַי נְדֵיב אֲשֶׁר רָאָוּ עֵינֵידָ: 8 אַל־תֵּצֵא לְרָב מַתֵּר פֶּן מַה־תַּעֲשָׂה בְּאַחֲרִיתֶה בְּהַכְלֵים אֹתְדֶ רֵעֶד: 9 רֵיבְדָ רֵיב אֶת־רֵעֶד וְסְוֹד אַחֵר אַל־תְגֵל: 10 פֶּן־יְחַפֶּדְדְ שֹׁמֵע וְדִבְּתְדָ לָא תָשׁוּב: 11 תַפּוּחֵי זָהָב בִּמַשִׂכִּיּוֹת כָּסֶף דֶׁבָּר דָבֶר עַל־אָפְנֵיו: 12 גֵזֵם זָהָב וַחַלִּי־כָּתֵם מוֹכֵיחַ ּחְכָּם עַל־אָׂזֶן שֹׁמֱעַת: 13 בְּצִנַּת־שֶׁׁלֶג | בְּיוֹם קָצִיר צִיר גָאָמָן לְשֹׁלְחֵיו וְגֶפֶשׁ אֲדֹנֵיו יָשִׁיב: פּ 14 נְשִׂיאִים וֲרוּחַ וְגֵשֶׁם אָיָן אֵישׁ מִתְהַלֵּל בְּמַתַּת־שֶׁקֶר: 15 בְּאֵׂרֶד אֲפַּיִם יְפֻתֶּה קָצֵין וְלָשִׁוֹן וֹרָבָּה ּתִּשְׁבָּר־גֵּרֵם: 16 דְּבַשׁ מֱצָאת אֵכְל דַיֶּדָ פֶּן־תִּשְׁבָעָׁנּוּ וַהַקֵאתוֹ:

אֲשֶׁע הֶׁעְהִיקוּ אַנְשֵׁי | חִזְקוֹּה מֱלֶדְ־יְהוּדֵה: 2 כִּבְד אֱלֹהִים הַסְתֵּר

מָלָבִים אֵין חֵקֵר: 4 הָגוֹ סִיגִים מִבֶּסֶף וַיֵּצֵא לַצֹּרֵף בֵּלִי: 5 הָגוֹ רֲשָׁע

דְּבֶר וּכְבִד מְלָבִים חֲלֵר דְּבֶר: 3 שָׁמַיִם לֶרוּם וָאָָרֶץ לָעָׂמֶק וּלֵב

משלי גַּם־אֵלֵה מִשָּׁלֵי שָׁלֹמָה 25:1

מְלַאּכְשָׁדְ וְעַתְּדָה בַּשָׂדָה לֵךְ אַׁחַׁר וּבְנִיתָ בֵיתָדָ: פּ 28 אַל־ תְּהֵי עֵד־חִנְם בְּרֵעֶדְ וֹהַפִּתִּיתָ בִּשְׂפָתֶידָ: 29 אַל־תּאׁמַר פַּאֲשֶׁר עֵשָּׁה־לֵי בֵּן אֶעֲשָׁה־לְוֹ אָשָׁיב לָאֵישׁ כְּפְעֲלוֹ: 30 עַל־שִׁדֵה אִישׁ־עָצֵל עָבְרְתִי וְעַל־בָּרֶם אָדָם חְסַר־לֵב: 31 וְהַנֵּה עָרֶהָ כָלוֹ | קִמְשוֹּנִים כָּסְוּ פָנְיו חַרֻלֵים וְגֶדֶר אֲבָנְיו נֶהֶרְסָה: 32 וֵאֶחֶזֶה אֲנֹכִי אָשִׁית לִבֵּי לָאִיתי לְקַחְתִי מוּסֵר: 33 מְעַט שֵׁנוֹת מְעַט תְּנוּמֵזת מְעַט ו חִבֶּק יִדַיִם לִשְׁבָּב: 34 וּבֵא־מִתְהַלֵּדְ רֵישָׁדְ וּמַחְסֹרֶידָ פָאַישׁ מָגַן: פ

ּטְוֹב: 26 שְׂפְתַיִם יִשֶׁק מֵשִׁיב דְּבָרֵים נְכֹחִים: 27 הְּבֵן בַּחוּץ | מְלַאַכְהֶדְ וְעַתְּדָה בַּשָׂדָה לֶךְ אֵׁחַר וּבָנִית בַיֶתֶד: פּ 28 אַל־

23 גַּם־אֵלֶּה לַחֲכָמֵים הַכֵּר־פָּגָים בְּמִשְׁפֵּט בַּל־טִוֹב: 24 אָׂמֵׁר | לְרָשָׁע ֿצַדֶּיק אָתָּה יִקְבָהוּ עַמֵּים יִזְעָמִוּהוּ לְאָמֵים: 25 וְלַמּוֹכִיתִים יִנְעֶם וַעֲלֵיהֶם תָּבְוֹא בִרְכַּת־

וְּלָשֶׁת מְׁתוֹק עַל־חִבֶּדְּ: 14 בֵּן | דְּעֵה חְרָמָה לְנַפְשָׁדָ אִם־מֲצָאת וְיֵשׁ אַחֲרֵית וְׁתִקְוָתְדְׁ לֵא תִבְּרֵת: פּ 15 אַל־רּתָבַא'' (רֻשָׁע) לִנְוָה צַדֵּיק אָל־תִּשַׁדֵּד רִבְצְוֹ: 16 בִּי שָׁׁבַע | יִפּוֹל צַדַּיק וָהֶם וּרְשָׁעִים יִבָּשְׁלָוּ בְרָעֵה: 17 בִּנְפַל <sup>(</sup>אוֹיִבְדָ<sup>-</sup>) אַל־תִּשְׂמֵח וּבִבָּשְׁלוֹ אַל־יָגֵל לְבֵּדָ: 18 פָּז־יִרְאָה יֵהוָה וְרַע בְּעֵינֵיו וְהַשִׁיב מֵעָלָיו אַפּוֹ: 19 אַל־תִּחָעָר בַּמְרַעִים אַל־תְקַנֵּא בְּרְשָׁעִים: 20 בִּי | לְא־תִהְיֶה אַחֲרֵית לְרָע גַר רְשָׁעַים יִדְעֵדְ: 12 יְרָא־אֶת־יְהוָה בְּנֵי וְמֵעָרָ עִם־רֹשְׁנֵיהָם אַל־ תִּתְעָרֵב: 22 בִּי־פָּתָאם יָקוּם אֵידֶם וּפִיד שְׁנֵיהָם מִי יוֹדַעַ: ס

תּוֹסֵף: 29 לִמִי אֲוֹי לִמֵי אֲבוֹי לִמֵי וֹמְדוֹנִים | לִמֵי שִׁיחַ לֵמִי פִּצָעֵים חַנֶּם לְמִי חַכְלִלוּת עֵינֵיִם: 30 לֵמְאַחֲרִים עַל־הַיֶּיָז לַבָּאִים לַחְקֹר מִמְסֶד: 31 אַל־תֵּרָא יַיִן בֶּי יִתְאַדֶּם בְּי־יִתֵּן 'בַּבּוֹס' עֵיגָוֹ יִתְהַלֵּךָ בְּמֵי שָׁרִים: 32 אֲחַרִיתוֹ בְּנָחָשׁ יִשָּׁך וְכָצִפְעָנְי יַפְרָשׁ: 33 אֲחַרִיתוֹ בָּנָחָשׁ יִשָּׁך זֶרָוֹת וְלִבְּדְׁ יְדַבֵּר תַּהְפֻּכְוֹת: 34 וֶהָיִתָ בְּשֹׁכֵב בְּלֶב־יָס וֹכְשׁבֵב בְּרָאשׁ חִבֵּל: 35 הִכְּוּנִי בַל־חָלִיתִי הֲלָמוּנִי בַּל־יָָדֶעְתִּי מְתַי אָקֵיץ אוֹסִיף אֲבַקְשֶׁנּוּ עוֹד: 24:1 אַל־הֵקַנּא בְּאַנְשֵׁי רָעֶה וְאַל־הִתְאָו לִהְיָוֹת אַתְּם: 2 בִּי־שִׁד יֶהְגֶה לִבֶּם וְׁעָמָׁל שִׂפְתֵיתֶם הְּדַבֵּרְנָה: 3 בֵּחְכְמָה יִבְּנֶה בֶּיִת וֹּבִתְבוּנָה יִתְכּוֹנֵן: 4 וִּבְדַעַת חֲדָרֵים יִמָּלְאֵוּ כָּל־הָוֹן יָקֶר וְנָעֵים: 5 וֹגָּבַרי חָכָם וֹמֵעָזי וִאָישׁ־דִּעַת וֹמֵאַמָּץ'־כְּחַ: 6 בְּי בְּתַחְבֻּלוֹת תַּעֲשֶׂה־לְךָ מִלְחָמֶה וֹּתְשׁוּעָׁה בְּרִב יוֹעֵץ: 7 רָאמְוֹת לֱאֱוִיל ְחְכְמֵוֹת בַּשַּׁעַר לָא יִפְתַּח־פִּיהוּ: 8 מְחַשֵּׁב לְהָרֵע לוֹ בַּעַל־מְזִמָּוֹת יִקְרֶאוּ: 9 זִמֵּת אָוֶּלֶת חַטָּאת וְתוֹעֲבָת לְאָדֶם לֵץ: 10 הֶתְרַפִּיתָ בְּיָוֹם צְרָה צַר כֹּחֶכָה: 11 הַצֵּל לְקָחֵים לַמֲוֶת וּמְטֵים לַהֶּג אִם־תַּחְשִׂוּדִ: 12 בְּי־תאֹמַר הֵן לא־יָדָעְנֿוּ זֶה הֲלא־תֹּבֵן לִבוֹת | הְוּא־יָבִין וְנֹצֵר ַנַפְשְׁדָ הַוּא יֵדֶע וְהֵשִׁיב לְאָדָם כְּפָעֵלוֹ: 13 אֱכָל־בְּנֵי דְבַשׁ כִּי־עָוֹב

לֶּדְ וְלִבוֹ בַּל־עִמֶּדְ: 8 פְּתְּדְּאָכִלְתָ תְקִיאָנָה וְשׁׁחַׁתְ דְבָרֵידְ הַנְּעִימִים: 9 פִּאָזְנֵי בֵסִיל אַל־תִּדַבֵּר פִּי־יָבוּז לְשֵׁכָל מִלֵּידְ: 10 אַל־תַּפַג גְּבוּל עוֹלֶם וּבִשְׁדֵי יְתוֹמִים אַל־תִּבָּא: 11 בִּי־גֹאַלָם חָזֶק הְוּא־יָרֶיב אֶת רִיבְם אִתֵּדְ: 12 הָבִיאָה לַמּוּסֵר לְבֵּדְ וְאָזְנֶדְ לְאָמְרֵי־דֵעַת: 13 אַל־תַמְנַע מִנַּעַר מוּסֵר כִּי־תַבֶּנוּ בַׁשָׁבָט לָא יָמוּת: 14 אַתָּ בַּשַׁבֶט תַּכָּנוּ וְנַפְשׁוֹ מִשְׁאוֹל תַּצְּיל: 15 בִּנִי אִם־חָכֵם לְבֵּדְ יִשְׁמָח לְבַּי גַם־אֲנִי:16 וְתַעְלְזְנָה כִלְיוֹתֵי בְּדַבֵּר שְׁפָּגָיד מֵישְׁרִים: 17 אַל־יִקנּגָ גַם־אַנִי:16 וְתַעְלִזְנָה כִלְיוֹתֵי בְּדַבֵּר שְׁפָגָיד מֵישְׁרִים: 17 אַל־יִקנָגַ גַם־אַנִי:17 וְתַעְלִזְנָה כִלְיוֹתֶי בְּדַבֵּר שְׁפָגָיד מֵישְׁרִים: 18 בִי אָם וְמַשְׁמָר בְּדָרָאַר'יְהָוֹתָ בְּדַבָר שְׁפָגָיד בַּרָים: 19 גַּבָּדְ בַּחַשָּאֵים בִּי אִם־בְּירָאַת'יְהָוֹת לָמְתָי בְזַנָע וּמִשְׁמְרָנָה יַנִש אַחַרִית וְתַקְנָתְדְ לָא תִבְּרֵת: 19 גַּבָּדְ וְמַליתִים אַלִיתָבוּ לְבָישׁ נוּמָר בְנָי וּחָשָׁמְרָנָים וּזּשִים בְּי אַם־בְיָאָת וְרָיאָנָר בָיָנִים וּתַכָּר וּמָשָּאוֹם בִיים אָבָּיָי בְעָרָים אַלָרִים אַרְרָיםי בּנִי וְמָשָׁרָים: 19 גַּים מִישָּרָים וּאַבָּר לְמָנָם וּבִשְׁרָים וּחַכָּם וְאַשָּר בַבְין לְבָישׁ אַתְרָים וּתָרְים הַיָּבָרָים בּנִיים וּחַכָּב וּבִיסוּבָאוּים בְעָים וּחַכָּבוּין בִיסַבָּר וּמָעַין בִיים וּתַבֶר וּמָנָר הָבָיי וּזוֹלַל יוּוֹתַרָים הַעָּבָים וּמָרָי בְסַרְבָאיי וּמָשָּר בְנִיים בָּבָיין בִישׁים וּתַהָר בְנִירָים בָּנָין בְּבָרָיָהוּ בָייִים וּתַרָּים וּמָבָין בָּים מָעָל אָנָה בְנָיוֹתָים בָּבָר וּיָם מָעָר בְיָים בַיָּין בָּרִיקוּנָעָר בַיּאַרָרָים הַיָּין בָּרִים בָּיָר בְנָיר מָיָרָים בְעָין בּיישָּרָין אַרין בְנָיין בְירִים בָּייים וּיוֹנַרָרין מוּידָין בִירים בָּעָר בְירָין בּיים מָבָין וּייןין בִיים בָּיים בְעָיין וּבָריים בָּבָין וּזָרָרָים בָּייָרָה בָרָים בּיים בְעָין וּבָייָרָים מִירָר בָיין בָריים גַרָיין בַייָרָים בָּעָרָין הוּבָרָין הַיָּנָרין וּנָרָין בָירָרָרָרָים בָּיוּרָין בָיין וּיין בַייין בָּבָין בָּיןינָר בָייָין בָריים בָּיןים בּיין בָּיין בִיין בָּיים גַייָין בָייי

שִׁפְתֶידִּ: 19 לִהְיוֹת בֵּיהוָה מִבְטַחֶדְ הוֹדַעְתָּידְ הַיָּוֹם אַף־אָתָּה: 20 הַלָּא כְתַבְתִּי לֵדְ <sup>ר</sup>ּשָׁלֹשֵׁים בְּמִוֹטַצִּוֹת וָדֵעַת: 21 לְהוֹדִיעַדְ לָשְׁטְ 20 אִמְרֵי אֶטֶת לְהָשִׁיב אַמְרֵים (אָטֶת) לְשׁלְחֵידָ: 3 אַמְרֵי אָטֶת לְהָשִׁיב אַמְרֵים (אָטֶת) לְשׁלְחֵידָ: 3 הִגְזָל־דָּל כֵּי דַל־הָוּא וְאַל־תְּדַבֵּא עָנִי בַשְׁעַר: 23 כִּי־יֶהוָה יְרֵיב רִיבֶם זְקָבָע אֶת־לְבְעֵיהֶם נְפָשׁ: 24 אַל־תִּתְרַע אֶת־בַּעַל אֶף וְאֶת־אִישׁ זְמָבָע אֶת־לְבְעַיהֶם נְפָשׁ: 24 אַל־תִּתְרַע אָת־בַּעַל אָף וְאֶת־אָישׁ זְמָבָע אֶת־לְבְעִיהֶם נְפָשׁ: 24 אַל־תִּתְרַע אָת־בַּעַל אָף וְאֶת־אָישׁ זְמָבָע אֶת־לְבְעִיהֶם נְפָשׁ: 24 אַל־תִּתְרַע אָת־בַּעַל אָף וְאֶת־אָישׁ זְמָבָע אֶת־לְבְעִיהָם נְפָשׁ: 24 אַל־תִּתְרַע אָת־בַּעַל אָף וְאָת־אָישׁ זְמָהַת לָא תְבִוֹא: 25 כָּזֶ־תִּאָלָף אִרְחֹתָו וְלָקַחְתָּ מוֹקַשׁ לְנַפְשָׁדָּי זֹמוֹת לָא תִבוֹא: 27 בָּלִים גָרָבִים מַשָּׁאוֹת: 27 אִם־אֵין־לְדָּ לְשַׁגַּם לְמָה יִקַּח מִשְׁכָבָדְ מִתַּחְתֶידָ: 28 אַל־תִּמֵג גְבוּל עוֹלֶם אֲשָׁר עָשִׁי אַבוֹתֶידָ: 29 חְזִית אָישׁ | מְהָיִה בִּמְלַאַרְהוֹז לַפְנֵי־מְלָכִים יִתִיאָב בַּל־ זְתִיצַב לִפְנֵי חֲשָׁכִים: פּ

23:1 בִּי־ְתַשֵׁב לִלְחַוֹם אֶת־מוֹשֵׁל בִּין הָּבִין אֶת־אֲשֶׁר לְפָנֵידָ: 2 וְשַׁמְתֵּ שַׁבִּין בְּלֹעֶדְ אִם־בָּעַל נֶבָּשׁ אֲתָּה: 3 אַל־תִּתְאָו לְמַטְעַמּוֹתֶיו זְהוּא לֶחֶם בְּזָבִים: 4 אַל־תִּיגַע לְהַעַשִׁיר מְבִּינְתְדָ חֲדֵל: 5 רְהַתָּעִיף עֵינֶידְ בּוֹ וְאֵׁינֶנּוּ בִּי עָשָׂה יַעַשָּׁה־לָוֹ כְנָפָים בְּנָשִׁיר וְיָעָוּף 5 הַשְׁמֵים: פּ 6 אַל־תִּלְחַם אֶת־לֶחֶם בַע עֶיון וְאַל־תִׁתְאָו לַמַטְעַמִּתְיו: 7 בִּי ן בְּמוֹ־וֹשׁעֵר בְּנַפְשׁוֹ בָּוָהָוּא אֶכָל וּשְׁתֵה יָאמַר

דַּרְכִּוֹ": 30 אֵיז חֶכְמָה וְאֵיז תְּבוּנֵה וְאֵיז עֵׁצָּה לְנֵ גֶד יְהוֶה: פ 31 סוּס מוּכָן לְיוֹם מִלְחָמֶה וְלֵיהוָה הַתְּשׁוּעֵה: 22:1 נִבְחָר שֵׁם ַמַעְשֶׁר רֶב מִפֶּסֶף וֹמִזְּהָב חֵז טִוֹב: 2 עָשִׁיר וְרָשׁ נִפְגָשׁוּ עֹשֵׂה כָּלָם יִהוֶה: 3 עָרִוּם | רָאֶה רָעֶה <sup>ו</sup>וִיּסֶתֵר<sup>ז</sup> וֹפִתִיִים עָבִרְוּ וְנֵעֵנֶשׁוּ: 4 עֵקֵב עַנְוָה יִרְאַת יְהוֶה עִשֶׁר וְכָבִוֹד וְחַיִּים: 5 צִנִּים פַּחִים בְּדֶרֶד עַקֵּשׁ שׁ וּמֶר יַנְפְשׁוֹ יִרְחֵק מֵהֶם: 6 חֲנִּךְ לַנַּעַר עַל־פֵּי דַרְבֶּוֹ גַּם בִּי־יַזְאָיז לְאֹ־ יָסָוּר מִמֶּנְה: 7 עֲשִׁיר בְּרָשֵׁים יִמְשֶׁוֹל וְעֶבֶד לֶוֹה לְאֵישׁ מַלְוֶה: 8 זוֹרֵע ַעַוּלָה ריִקְצָר־אָוָן וְשֵׁבֶט עֶבְרָתִוֹ יִכְלֶה: 9 טִוֹב־עַיִן הוּא יְבֹרֶךְ בְּי־נָתַן מִלַּחְמָוֹ לַדֶּל: 10 גְּרֵשׁ לֵץ וְיֵצֵא מְדָוֹן וְיִשְׁבּׁת דְּין וְקָלְוֹן: 11 אֹהֵב ּרְטְהָר<sup>ז</sup>-לָב חֵן שְׁפָתָׁיו רֵעָהוּ מֶלֶדְ: 12 עֵינֵי יֶהוָה נְצְרוּ דֶעַת וֹיְסַצֵּׁר דִּבְרֵי בֹגֵד: 13 אָמֵר עֲצֵל אֲרֵי בַחָוּץ בְּתוֹדְ רְׁחֹבוֹת אֵרָצֵחַ: 14 שׁוּחָה אֲמֻקָּה פִּי זָרִוֹת זְעָוּם יְהוָה יּיִפָּל־יֹשֶׁם: 15 אֵוֶּלֶת קְשׁוּרָה בְּלֶב־נְעַר שֵׁבֶט מוּסָׁר יַרְחִיהֶנָּה מִמֶּנּוּ: 16 עִׂשֵׁק דָּל לְהַרְבִּוֹת לֵוֹ נֹתֵן ֹלְעָשִׁיר אַדְ־לְמַחִסְוֹר: 1<sub>7</sub> וּדִבְרֵי חֵכָמִים הַט אָזִנְדָ וּשִׁמַע דְבָרֶי וֹלִבְדָּ תַּשֵׁית לְדַעָתֵי: 18 בֵּי־נַעִים בֵּי־תָשָׁמְרֵם בִּבְטָגֵדְ יִבְּנוּ יַחִדֵּו עַל־

 $\begin{aligned} M^{K} &= \left[ \begin{array}{c} (\operatorname{corr} G (\operatorname{corr} \operatorname{cor} (\operatorname{corr} \operatorname{cor} (\operatorname{corr} \operatorname{corr} G (\operatorname{corr} G (\operatorname{c$ 

עָז מִבְּטֶחָה: 23 שֹׁמֵר אֵּיו וּלְשׁוֹגֵוֹ שֹׁמֵר מִצְרוֹת נַפְשׁוֹ: 24 זַד יְהִיר לֵלֵץ שְׁמֵוֹ עׁוֹשֶׁה בְּעָבְרָת זְדְוֹן: 25 תַאָוָה תָצֵרָיק יְּוֹמֵן וְלָא יַחְשָׂדָ: 27 זָבַח לַעֲשָׁוֹת: 26 בְּל־הֵיּוֹם הִתְאַוָּה תַאֲוָה וְצַדָּיק יְוֹמֵן וְלָא יַחְשָׁדָ: 27 זָבַח הַשְׁעִים תּוֹעֵבֶה אַׁף בְּיבְזְמָה יְבִיאֶנּוּ: 28 עַד־בְּזָבִים יֹאבֵד וְאָישׁ הַשָּׁעִים תּוֹעֵבֶה אַף בְּיבִזְמָה יְבִיאֶנּוּ: 28 עַד־בְּזָבִים יֹאבֵד וְאָישׁ הַשָּׁעִים תּוֹעֵבֶה אַף בְּיבִזְמָה יְבִיאֶנּוּ: 28 עַד־בְּזָבִים יֹאבֵד וְאָישׁ הַשְׁעִים תּוֹעֵבֶה אַף בְּיבִזְמָה יְבִיאֶנּוּ: 28 עַד־בְּזָבִים יֹאבֵד וְאָישׁ איחז [ (עריע בי 23 בעִדיבין וּאותה רע 1:10 שׁוֹמַע לְגַצַח יְדַבְּר: 29 הַעֵּז אָישׁ רְשָׁע בְּפְנָיו וְיִשָּׁר הָוּא ו<sup>ַ</sup>יְיָ</sup>בִיץ ניבעיני [ זיהן מוֹתה רע 1:10 אירז [ (בּבעּדָ) אותה רע 1:10 איקריביים אותה רע 1:10 איקריביים אותה רע 1:11 אותה רע 1:12 אובעיני [ גרביא] אותה רע אותה מוֹנים (מאַמדעיַרָבָשָׁר : 10 בַעוּבָים אַרָרָשָּינִי אישבון 1:21 איקריבָשָּה : גענינין 1:13 איקריבָשָּה : גענינין 1:13 איקרינים 1:21 און גַיָשָׁמָן (מוֹאַמִאַנָין 1:21 אישבון 21:21 און גענים 1:21 אישבון 1:21 אישבון 1:21 אישבון 1:21 איקרינים 1:21 איקריניים 1:21 איקרינים 1:21 איקריניים 1:21 א

חֲבֶר: 10 גָפָשׁ רָשָׁע אוּתָה־רֵע לא־יָחָן 'בְּעַינֵי' בעֵהוּ: 11 'בַּעַנָשׁ' לֵץ יָחִכַּם־פֶּתִי וּרְהַשִׁכֵּיל לְחָכָם יַקַח־דֵּעַת: 12 מַשִׂכֵּיל צֵּדִיק לְבֵית רָשֶׁע מְסַלֵּף רְשָׁעֵים לְרֵע: 13 אֹטֵם אוּזְנוֹ מִזַּעַקַת־דֶּל גֵם־הוּא יִקָרָא וְלָא יֵעָגָה: 14 מַתָּן בַּסַּעֶר 'יְכַפָּר'־אָף וְשָׁחַד בַּחֵק חַמָה עַזֵּה: 14 שִׁמְחָה לַצַדִיק עַשִּוֹת מִשִׁפֶּט וּמְחִתָּה לְפַּעְלֵי אוָן: 16 אָדָם זוּנעה מִדֶּרֶד הַשִּׁפֶל בְּקָהָל רְפָאִים יְגוּחַ: 17 אַיש מַחְסוֹר אֹהֵב קוֹעָה מִדֶּרֶד הַשִּׁפֶל בְּקָהָל רְפָאִים יְגוּחַי: 17 אַיָש מַחְסוֹר אֹהֵב שִׁמְחָה אֹהֵב יִיִדּיָשָׁטוֹת מִשְׁפֶט וּמְחִתָּה לְפַעְלֵי אָוָן: 16 אָדָם אוֹעָה מִדֶּרֶד הַשִּׁפֶל בְּקָהָל רְפָאִים יְנוּחַ: 17 אַישׁ מַחְסוֹר אֹהֵב שִּמְחָה אֹהֵב יִיוּדיָשֶׁטוֹ לָא יַעֲשָׁיר: 18 בְּפָר לַצַּדְיק רָשָׁע וְתָחַת שִׁמְחָה אֹהֵב יִיוּדיָשָׁמון לָא יַעֲשָׁיר: 18 בְפָר מַאָּדָיק יְבָשָׁע וְתָחַת ישִׁמְחָה אֹהֵב יִיוּדיָשָׁמון לָא יַבָּשָׁיר: 19 בְּבָּר מַאָשָׁת וֹמְחָנוּ עַלָּחַמָר וּמָדוּנִים וּנְתַס גער גַּדְרָם בּוֹגָד: 19 אוֹשָׁ מוּתָר אָעָיירי גַּרָים אָבָרָים עָלָה חָכָם וּבְעַסי מַלָּין שִׁמְתָה אַהָּב יַיִרוּשָׁין וּקָטָר יַמְצָע וּתַיחַנִים גער גַּדְרָים בוּגָרי וּמָען וּתָסָר ישָׁרָים בּוֹגָר: 19 גַים וּבְעָטין מַי מִבְּרָשָׁין וּמָסָר יִמְצָא חַיִים וּין בִּעָם וּין בַשְּעָר אָבָרים יעָלָה חָכָם וֹדָינִים עַיַז מִבְנָה וּחַסָר יִמְצָא חַיִים יוּיוֹים וּשָׁרָי אַנָין אָבוּיים גַירָי אָנָין אָבָין רָייָרָ עַרָּעָשָרוּ אַבָּר מַאָנוּן וּיָבָין אוּיים לַעַשְׁים הּמִיתָנוּ בָּירִים אָנָרוּים הַעָּצָר וּיָרָים בּירָין אָרָים גַירָין גַיןיי לַרָשָּעִים הּזעַרָן גַין שָּיָר וּיָשָעוּיים וּין בָּעָיים גָינוּין גַיןיין גַיןין לַעָּשָּים יִיקוּקרים גָיןין בָּין אָרָין וּין בָּין וּיןין לַשָּעוּ הַישְּעָןים בּירָין אָרָין בָּעָיים גָין וּשָעוּים וּין בָיין גָין וּין בָּיין אָעָעָיין גָיןין גָיןין בּעָין שָּעָיים גַין וּזשְרָר וּשָּרָר בָיין בָייןין גָין אָישָרין גָּעָן אָישָרָין אָיין בּעָיין שָּבָין שָּעָרון וּין בָּעָיין אָיין בָין אָין אָיין בּעָין שָּעָר וּיוּין בָרָרָר וּיין בָרָין גָין ייָין בָיין אָין גַין בָין אָין גַין בּעָעָר וּין בָ 26\*

מִלְחָמֶה: 19 גִּוֹלֶה־סִּוֹד הוֹלֵדְ רְכֵיל וּלְפֹתֶה שְׁפָתָׁיו לְא תִתְעָרֵב: 20 מֱקַלֵּל אָבְיו וְאָמֵוֹ יִדְעָדְ וֹרוֹ רְבָאִישִׁוֹן חְשֶׁדְ: 21 וַחַלָה ּמְבֹהֶלֶת וּ בָּרָאשׁנֶה וְאַחֵרִיתָה לְא תְבֹרֶדְ: 22 אַל־תּאמֵר אֲשַׁלְמָה־ רֶע קַוּה לֵיהוָה וִיְשָׁע לֶדְ: 23 תּוֹעַבָת יֶהוָה אֶבֶן וָאָבֵן וּמֹאוֹנֵי מִרְמֶה לאֹ־טְוֹב: 24 מֵיְהוֶה מִצְעֲדֵי־גֶבֶר וְאָדָׁם מַה־יָבֵין דַּרְכָּוֹ: 25 מוֹקֵשׁ אָדָם יָלַע קֹדָשׁ וְאַחָר נְדָרֵים לְבַקֵּר: 26 מְזָרֵה רֵשָׁעִים מֱלֶד חָבֶם וַיֶּשֶׁב עֲלֵיהֶם אוֹפֵן: 27 גַר יֶהוָה נִשְׁמֵת אָדֶם חֹפֵׁשׂ כְּל־חַדְבִי־בֶטָן: 28 חֱסֶד וֲאֱמֶת יִצְרוּ־מֱלֶך וְסָעֵד בַּחֱסֶד כִּסְאוֹ: 29 תִּפְאֱרֶת בַּחוּרֵים כּׁחֶם וַהַדַר זְהֵנְים שֵׁיבֶה: 30 חַבָּרוֹת פֵּצַע 'תַּמְרֵיק' בְּרָע וּמַכּוֹת חַדְבִי־בֶּטֶן: 1:12 פַּלְגִי־מַיִם לֶב־מֶלֶדְ בְּיַד־יְהוֶה עֵל־כָּל־אֲשֶׁר יַחְפִּץ יַטֶּנּוּ: 2 בְּל־דֶּרֶדְ־אָישׁ יָשֶׁר בְּעֵינֵיו וְתֹכֵן לִבַּוֹת יְהוֶה: 3 אֲשָׁה צְדָקָה וּמִשְׁפֶּט נִבְחֶר לַיהוֶה מִזְּבַח: 4 רוּם־עֵינַיִם וּוְרֹחַב־ֹלֶב וּגֵר ּ רְשָׁעֵים ַחַטֶּאת: 5 מַחְשְׁבִוֹת הֶרוּץ אַדְ־לְמוֹתֶר וְכָל־אָּץ אַדְ־לְמַחְסוֹר: 6 <sup>ר</sup>פּּעֵלי אוֹצְרוֹת בִּלְשִׁוֹן שֶׁקֶר הֶבָל יֹרֹדֵּף בְּמֹקְשֵׁייֹ-מֶוֶת: 7 שׁד-רִשָּׁעֵים יִגוֹרֶם כֵּי מֵאַנוּ לַעֵשְׂוֹת מִשִּׁפֶּט: 8 הַפַּכִפַּד הֵרֶד אֵישׁ יוָר וְזַדְ יָשֶׁר פָּעֵלְוֹ: 9 טוֹב לָשֵׁבֵת עַל־פִּנַת־גָג מֵאֵשֵׁת ימִדוֹנִים' וּבֵית

20:20 (איך אראישון  $M^{K}$  G (ai δż אלסמו) איך אראישון  $M^{Q}$  באישון ( $M^{Q}$  M<sup>K</sup> G: orth, equal; S T: graph באישון ( $M^{Q}$  M<sup>K</sup> G: orth, equal; S T: graph מבהלת 20:21) ( $M^{Q}$  M<sup>K</sup> G: orth, equal; S T: graph ( $m \in M^{Q}$  G (at 20:9b) (żπισπουδαζομένη) S (הסטיסה) V (festinatur) איך אתונא) M<sup>Q</sup> G (at 20:9b) (żπισπουδαζομένη) S (הסטיסה) V (festinatur) (graph  $\rightarrow \rightarrow M^{K}$  (graph  $\rightarrow \rightarrow M^{K}$  (graph  $\rightarrow \rightarrow M^{K}$  (graph  $\rightarrow \rightarrow M^{K}$  (graph  $\rightarrow m \in M^{K}$  (second) ( $m \in M^{K}$  (graph  $\rightarrow m \in M^{K}$ )) (vocal) ( $m \in M^{K}$  (graph  $\rightarrow m \in M^{K}$ )) (vocal) ( $m \in M^{K}$  ( $m \in M^{K}$ )) (graph  $\rightarrow m \in M^{K}$  ( $m \in M^{K}$ )) ( $m \in M^{K}$  ( $m \in M^{K}$ )) ( $m \in M^{K}$ ) ( $m \in M^{K}$ )) ( $m \in M^{K}$ ) ( $m \in M^{K}$ )) ( $m \in M^{K}$ ) ( $m \in M^{K}$ )) ( $m \in M^{K}$ ) ( $m \in$ 

19:27 עד בליעל 19:28 M<sup>AY</sup> [ חָדָל M<sup>L</sup> (vocal err) || 19:28 מסתג M<sup>AY</sup> M S (א מסג א M S (א מסג א שסג א ש  $(\Delta \sim)$  ] אערב בן בליעל G (א לאין ל הער ד המוֹט מאָרב בן בליעל G (א באין האין ה ה), dittog לוttog ב  $\rightarrow$  בב, near dittog בל הבנ לוttog לוttog און (בב  $\rightarrow$  בב M און M און G (געד (געד S ( הער sc. געד) (graph paleo?  $\prec \rightarrow \sigma$ ) || 19:29 שבטים G (μάστιγες) S ? (באיבה ( (גוי [ (באיבא M (graph  $t \to t)$  אלגוי M G ( $ec{\omega}$ μοις) ) אלגוי M G ( $ec{\omega}$ μοις) אלגוי ( (באיבא M G ( $ec{\omega}$ μοις) ) אלגוי ( S ( $\bigtriangleup M^{AY}$  [ ניין  $M^{AY}$  [ ניין  $M^{AY}$  ] (vocal err) (20:4ישאל MK [ אישאל MQ G (אמו א ל אמון א מענלטעבעסג) S (גבאר) (gram, equal) || 20:6 (حאסت א S (حאסי ) V (vocantur) ] יקרא M (misinterp) || א קרא G (ἐλεήμων) S (אדיערביא) |  $\square$  (dittog  $\square \rightarrow \square$ ) || 20:14–19 M S ] > G (unc) || נכרים 20:16  $M^{K} V (extraneis) S (גרים M^{QMSS} (graph <math>\rightarrow \rightarrow )$ 

וס אֶבֶן וָאֶבֶן אֵיפֶה וְאֵיפֶה תּוֹעֲבַת יְהוָה גַם־שְׁנֵיהֶם: 11 גַם רֹאֶה יְהוָה עָשָׂה גַם־שְׁנֵיהֶם: 13 אַל־תֶּאֶהַב שֵׁנָה בֶּן־תּוָרֵשׁ פְּקָח יַעַיגֶידְ שְׂבֵע־לֶחֶם: 14 רֵע רֵע יֹאמַר הַקּוֹגֶה וְאֹזֵל לוֹ אָז יִתְהַלֵּל: ַזָּהָב וְרָב־פִּגִיגֵים וּכִלֵי יִקָּר שִׂפְתֵי־דֶעַת: 16 לְקַח־בְּגִדוֹ כִּי־ עָרֵב זָר וּבִעָד 'נָכְרֵים' חַבְלֵהוּ: 17 עָרֵב לָאִישׁ לֵחֵם שָׁקֵר וְאַחֵׁר יִמֵּלֵא־פֵיהוּ חָצֵץ: 18 מֲחַשָּׁבוֹת בְּעֵצֵה תִכְּוֹן וֹבְתַחִבָּלוֹת עֵשֵׂה

משלי 25\* יִלְנָבוֹן יָבֵין דֶעַת: 26 מֵשַׁדֵּד־אָב יַבְרֵיחַ אָם בֵּן מֵבֵישׁ וּמַחִפֵּיר: 27 וּחֲדַליּ־בֶּנִי לִשְׁמְעַ מוּסֶֶר לִשְׁגוֹת מֵאמְרֵי־דֶעַת: 28 עֵד בְּלִיַּעַל יַלֵיץ מִשְׁפֶּט וּפִי רְשָׁעִים יְבַלַּע־אֶוֶן: 29 נָכִונוּ לַלֵּצֵים וּשְׁבָטֵים וֹמַהַלָמות לְגֵו בִּסִילֵים: 20:1 לֵץ <sup>ר</sup>ְהַיַיִן הֹמֵה שֵׁכֵר וִכָּל־שְׁגֵה בּוֹ לְא יֵחִכֵּם: 2 נֵהַם כֵּכִּפִיר אֵימַת מֱלֶך מְתָעַבְּרוֹ חוֹטֵא נַפִּשוֹ: 3 כַּבִוֹד לאיש שֵׁבֵת מֵרֵיב וְכָל־אֵוִיל יִתְגַּלֵּע: 4 מֵחֹרֵף עָצֵל לא־יַחֵרָשׁ ּיִשְׁאַלי בַּקָּצִיר וָאֶיוָ: 5 מַיִם עֵמָקּים עֵצֶה בְלֶב־אֵישׁ וְאֵישׁ תְּבוּנֶה יִדְלֵנָה: 6 רָב־אָדָם וּיֶקָרֵא' אֵישׁ וּחָסֵד' וָאֵישׁ אֵמוּנִים מֵי יִמְצֵא: ז מִתְהַלֵּך בְּתֵמוֹ צַדֵּיק אַשְׁרֵי בְנְיֵו אַחֲרֶיו: 8 מֶׁלֶד יוֹשֵׁב עַל־בָּסֵא־ ּדִין מִזֶרֵה בִעֵינֵיו כָּל־רֵע: 9 מֵי־יְאׁמַר זִכִּיתִי לִבָּי טָהַרִתִּי מֵחַטָּאתֵי: רַאָטָר אָם־זָשָר פָּעָלו: 12 אָאָן שִׁמַעַת וְעַיַן בַּמַעַלָיו יִתְנַכֶּר־גָעַר אָם־זָד וְאָם־יִשָּׁר פָּעָלו

19:1-25

בּתָמָו מֵעָקֵשׁ שִׁפָּתָיו וְהוּא <sup>ר</sup>ּעָשֵׁיר<sup>י</sup>: 2 גַם בִּלֹא־דַעַת גֵפָשׁ לֹא־טָוֹב וְאָץ בְּרַגְלַיִם חוֹטֵא: 3 אָוֶלֶת אֶדָם תְּסַלֵּף דַרְכָּוֹ וְעַל־יְהוָה יוְעָף לִבְּוֹ: אַקרים לָא יִנָּאֶה יִנָּאָה זיִפָּרָד: 5 אַד שֶׁקָרים לָא יִנָּאָה 4 הוֹז יִסִיף רַעִים רַבָּים יְדָל וִיָּפְיַחַ כִּזְבִים לְא יִמֶּלֵט: 6 רַבִּים יִחַלְּוּ פְגֵי־נָדָיב וְכָל־הֶבִע לִאִישׁ מַתֶּז: 7 בֶּל אֲחֵי־רָשׁ | שְׂנֵאָהוּ אָף בֵּי מֶרֵעֵהוּ רָחֵקוּ מִמֵּנוּ מִרַדֵּף אַמֶרִים לאֹ־הֵמָּה: 8 קֹנֶה־לֵב אֹהֵב נַפְּשָׁוֹ שֹׁמֵר הְבוּנָה לִמְצאֹ־טָוֹב: 10 לא־ 9 עֵד שֶׁקָרִים לָא יִנְּאֶה וְיָפֶיחַ כְּזָבִים יאבִד: פ נָאוֶה לִכְסֵיל תַּעֲגָוּג אַׁף בִּי־לְעָבָד | מְשׁׁל בְּשָׂרִים: 11 שֵׂכָל אָדָם ָהֶאֶרִידְ אַפֶּוֹ וֹןְתִפְאַרְתּוֹ עֲבִר עַל־פֶּשַׁע: 12 נַהַם בַּכְּפִיר זַעַף מֶלֶד וּכְטַל עַל־עֵשֶׂב רְצוֹגִוֹ: 13 הַוֹת לֱאָבִיו בֵּן בְּסֵיל וְדֶלֶף טֹבִיד מִדְיְגֵי אִשֶׁה: 14 בִּיִת וֶהוֹז נַחֲלֵת אָבָוֹת וֹמֵיְהוָה אִשֶׁה מַשְׂבֶּלֶת: 15 עֵצְלָה תַּפִּיל תַּרְדֵּמֶה וְגֶפָשׁ רְמִיָּה תִרְעֶב: 16 שֹׁמֵר מֵצְוָה שֹׁמֵר נַפְשָׁוֹ בּוֹזֵה דְרָבֶיו "יְמְוּת": 17 מַלְוֵה יֶהוָה חִוֹגֵן דֶּל וֹּגְמֵלוֹ יְשַׁלֶם־לְוֹ: 18 יֵמֵּר ְבֵּנְדְ בִּי־יֵשׁ תִּקְוָה וְאֶל־הֲמִיתוֹ אַל־תִּשָׂא נַפְשֶׁדְ: 19 <sup>ר</sup>ְגָּדָל<sup>ז</sup>־חֵמָה נְשֵׁא עֶׁנֶשׁ בִּי אִם־הַאָּיל וְעַוֹד תּוֹסְף: 20 שְׁמַע עֵצָה וְקַבֵּל מוּסֶר ּלְמַעַן תֶּחְכֵּם בְּאַחֲרִיתֶדְ: 21 רַבְּוֹת מַחֲשָׁבְוֹת בְּלֶב־אֶישׁ וַעֲצַת יְהוָה הֵיא תָקוּם: 22 רֹתְבוּאַת אָדָם חַסְדָוֹ וְטְוֹב־רְשׁ מֵאָישׁ כָּזֶב: נ2 יִרְאַת יְהוָה לְחַיָּיֶם וְשָׂבֵע יְלִין בַּל־יִפְּקָד רֶע: 24 טְّמַן עָצֵל יָדוֹ בַּצַּאֶֶחַת גַּם־אֶל־פִּיהוּ לָא יְשִׁיבֶנָה: 25 לֵץ הַכָּה וּפָתִי יַעָרֵם וְהוֹבֵיח

לאֹ־טָוֹב לְהַטָּוֹת צַׁדִּיֹק בַּמִּשְׁבֶּּט: 6 שִׁפְתֵי בֵּסִיל יָבָּאוּ בְרֵיב וּפִיוּ לְמִהַלָּמִוֹת יִקְרֵא: 7 בְּיַרְבָסִיל מְחִתָּה־לֵו וּשִׁפְתָיו מוֹקֵשׁ נַפְשוֹ: 8 דִּבְרֵי גָרְגָן בְּמְתְלַהַמֵּים וְהֵם יְרְדָוּ חַדְרֵי־בֶּטָן: 9 גַּם מִתְרַבֶּה בְמְלַאְכְתֵוֹ אֶח הוּיא לְבַעַל מַשְׁחִית: 10 מִגְדַל־עָז שֵׁם יְהוֶה בְּוֹ־יָרָוּיָ בְמְלַאְכְתֵוֹ אֶח הוּיא לְבַעַל מַשְׁחִית: 10 מִגְדַל־עָז שֵׁם יְהוֶה בְּוֹ־יָרָוּיָ צַדֵּיק וְנִשְׁגֵב: 11 הוֹז עֵשִׁיר קַרַיַת עָזָו וּכְחוֹמָה נִשְׁגָּבָה בְּמַשְׁבִּיתוּי זַצַּדְיק וְנִשְׁגֵב: 11 הוֹז עֲשִׁיר קַרַיַת עָזָו וּכְחוֹמָה נִשְׁגָּבָה בְּמַשְׁבָּיתוּ: 12 לִפְנֵי־שֶׁבֶר יִגְבַה לֵב־אֵישׁ וֹלְפְנֵי כְבָוֹד עַנְוָה: 13 מַשְׁיב דְּבָר בְּטָרֶם יִשְׁמֶע אוֶּלֶת הִיא־לוֹ וּכְלְמֵה: 14 רְיחַ־אֶישׁ יְכַלְבֵּל מַחַלֵּהוּ בְּעָרָם יִשְׁמֶע אוֶלֶת הִיא־לוֹ וּכְלִמֶה: 14 רְיחַדָּאִישׁ יְכַלְבֵּל מַחַלֵּהוּ בְּעָרָם יִשְׁמֶע אוּזָעָת הִיא־לוֹ וּכְלִמֵה: 14 רוּחַדָּאִישׁ יְכַוְבֵּלִים תְּבַקָּשׁ־ בְּעַרָים יִשְׁמָע אוּלָת הִיא־לוֹ וּכְלִמֵה: 14 רוּחַדָּשִים וּאָזן חְכַמִים תְּבַקָּשִׁי בְּעָרָי זם יְרָזּבָאים מִי ישָׁאֶנָה: 17 עַבָּלָמוּי גוּדָרָים יַנְקָנָירוּין בְּתָים הַנְיוֹז וּאָדָם יַרְחַלֵאוּין מְרָים בְּרָרִים גָּים הְמָז אָנָהם יַיחַמָרוּין גָיוּים הְדָרָים כָּיחָלָיון בְּתַרָּבָיה ישָׁגָּנִי הַבָּים יִשְנָעוּי בַיּשְׁתִיו זישְׁבָּעוּר וּשְׁנָין וּזָרָם בְיָרָים בְיחָים בְּבָעוּין בַיּבָרִיד: 19 בְּיִשְׁים הַיּשְׁנִין מִימִים הָנָים יִדָּבָים יַישְנִים הַירָחַים אָנוּין בַיוּז בַּירִיד: 19 אָתְיזוּז וּחָקרוּין גִשָּנִים הַיחַים בּיוּים בּרָיקים הַיָּנָיים בְירָים אַימוּין בַיַרִידי בָּין וּשְׁיוּחוּין בּיים הַיָּשִים מִיּים הָיםים יָּבָים יַעָרוּין בּבָּרִיד: 19 אָנְעָים הַיוּבוּין בָּרָיים בּירָים בְיתָרָים מָעוּין בּישְׁבָין בּייזים בְידִיין שְּבָרין גוּיןים בּיישָּמִים בְרָים מְחָרוּין בָּרָיים בְירָים בָּבָין בַיעָים בַייּבָּים בְיבִייים בְיישִים בְיים בְיעָמוּין בָּרָים בְיבָין בּייוּם בּייוָה בָיוּיין בָיין בָיים בְיתָריים בָּבָין בָיים בְייוּים בּיים בּייוּים בּייוּים בָּיוּים בָיין בּייין בָיייים בְּיייים בָּיוּין וּזים בִייןים בִיין בּשִיין בּיייים בָּי

17:11-18:5

22\*

יְבַקֶּשׁ־רֶע וּמַלְאָָך אַכְזָרִי יְשֶׁלַּח־בְּוֹ: 12 פָּגוֹשׁ דַּב שַׁכּוּל בְּאֵישׁ וְאַל־ ּבְּאַוּלְהְוֹ: 13 מֵשֵׁיב רָעָה הַּחַת טוֹבֶה לא<sup>ַר</sup>תָמוּשׁ<sup>י</sup> רְׁעָה מִבֵּיתִוֹ: 14 פּוֹטֵר מֵיִם רֵאשֵׁית מַדָוֹן וִלִפְנֵי הָתִגַּלֵּע הָרֵיב נְטִוֹשׁ: דַנַמָר אַזיָק רָשָׁע וּמַרְשִׁיעַ צַדָּיק תּוֹעֲבָת יְהוָה גַּם־שָׁנֵיהֵם: 16 לַמָּה־ זֶה מְתַיר בְּיַד־בְּסֵיל לִקְגוֹת חָכְמֵה וְעָב־אֶיָן: 17 בְּכָל־עֵת אֹהֵב הָרֵעַ וְאָח לְצָרָה יִוְלֵד: 18 אָדֵם חַסַר־לֵב תּוֹהֵע כֶּף עֹרֵב עְׁרֵבָּה לִפְגֵי ַרַעֵהוּ: 19 אַהֵב פֶּשַׁע אֹהֵב מַצֶּה מַגְבֵּיהַ פִּתְחוֹ מְבַקֶּשׁ־שֶׁבֶר: 20 עִקֶּשֹׁ־לֵב לְא יִמְצָא־טָוֹב וְנֶהְפָּך בִּלְשׁוֹנוֹ יִפְּוֹל בְּרָעֵה: 21 יֹלֵד ַרְּוּתַה לָו וְלְאֹ־יִשְׂמַח אֲבִי נְבֶל: 22 לֵב שָׂמַח יֵיטֵב גֵּתָה וִרוּחַ ּעְׁכָאָה הְיַבֶּשֹׁ־גֶּרֶם: 23 שִׁׁחַד רְּבֵחֵיק רָשָׁע יִקֶּח לְהַטּׁוֹת אָרְחָוֹת מִשְׁפֵּט: 24 אֶת־פְּנֵי מֵבִין חָכְמֶה וְעֵינֵי כְׁסִיל בִּקְצֵה־אֶרֶץ: 25 בַּעַס רְאָבִיו בֵּן בְּסֵיל וֹמֶׁמֶר לְיוֹלַדְתּוֹ: 26 גַּם עֲנַוֹשׁ לַצַּדִּיק לאֹ־טָוֹב לְהַכָּוֹת ּנְדִיבִים עַל־יְשֶׁר: 27 חוֹשֵׂדְ אֱמָרָיו יוֹדֵעַ דֶעַת וְקַר־רוּחַ אֵישׁ תְּבוּנֶה: 28 גַּם אֱוִיל מֲחַרִישׁ חָבָם יֵחָשֵׁב אֹטֵם שְׂפָתֵיו נְבְוֹן: 18:1 לְתַאֲוָה יְבַקֵּשׁ נִפְרֶד בְּכָל־תוּשִׁיָה יִתְגַּלְּט: 2 לְא־יַחְפִּץ בְּסִיל בִּתְבוּנֶה בִּי אֶם־ בְּהִתְגַּלְּוֹת לִבְּוֹ: 3 בְּבוֹא־ֻרְשָׁע בָּא גַם־בָּוּז וְעִם־קָלְוֹן תֶרְפָּה: 4 מַיִם עַמָקִים דָבְרֵי פִי־אָישׁ נַחַל נְבָע מִקוֹר חָכְמָה: 5 שָׂאֵת פּנֵי־רָשָׁע

20 מַשְׂהֵיל עַל־דֶּבָר יִמְצָא־טָוֹב וּבוֹטֵח בַּיהוֶה אַשְׁרֵיו: 21 לַחֲכַם־לֵב יִקָּרֵא נָבְוֹן וּמֶתֶק שְׁפָתַיִם יֹסֵיף לֱקַח: 22 מְקוֹר חֵיִים שֵׁכֶל בְּעָלֵיו וּמוּסַר אֶוּלִים אָוֶלֶת: 23 לֵב חֶכָם יַשְׂבֵּיל בִּיהוּ וְעַל־שְׁפָתָיו יֹסֵיף לֵנֶפָּשׁ וּמַרְפֵּא לָעֶצָם: 25 צוּף־דְּבַשׁ אָמְרֵי־גְׁעַם מֶתְוֹק לַנֶּפֶשׁ וּמַרְפֵּא לָעֶצָם: 25 יֵשׁ דֶרֶדְ וְשָׁר לִפְנֵי־אֶישׁ וְאַחֵרִיתָה דַּרְבֵי־מֱוֶת: 26 נֶפֶשׁ עָמֵל עֵמְלָה לֶוֹ ַכִּי־אָכַף עָלָיו פִּיהוּ: 27 אֵישׁ בֵּלִיַעַל כֹּרֶה רְעָה וְעַל־רֹשְׁפָתָיו ּ כְּאֵשׁ צְרֶבֶת: 28 אֵישׁ הַהְפָּכוֹת יְשַׁלֵּח מְדָוֹן וְנִרְגָּן מַפְרָיד אַלְוּף: 29 אֵישׁ חַמָס יְפַתֶּה רֵעֵהוּ וְהוֹלִיכֿוֹ בְּדֶרֶךְ לֹא־טְוֹב: 30 עֹצֶה וֵעִינִיו לַחְשִׁב תַּהִפָּכְוֹת קֹרֵץ שִׁפָתָׁיו כִּלֶה רָעֵה: 31 עֵטֶרֵת תִּפְאֵרֶת שֵׁיבֶה בִּדֶרֶדְ צִׁדְלָה תִּמְצֵא: 32 טִוֹב אֶרֶךְ אֲפַּיִם מִגְּבָּוֹר וּמֹשֵׁל בְּרוּחוֹ מִלֹבֵד עִיר: 33 בַּחֵיק יוּטַל אֶת־הַגּוֹרֶל וּמֵיְהוָה בָּל־מִשְׁפָּטִוֹ: 17:1 טָוֹב פַּת חֲבַבְה וְשַׁלְוָה־בֶהּ מִבַּיִת מְלֵא זִבְחֵי־רִיב: 2 עֶבֶד־מַשְׂבִּיל יֵמְשׁׁל בְּבֵן מֵבֶישׁ וּבְתָוֹדְ אֵׁחִים יַחֲלָק נַחֲלֶה: 3 מַצְרֵף לֵכֶּסֶף וְכָוּר לַזְּהֶב וּבֹחֵן ּלִבְּוֹת יְהוֶה: 4 מֵרַע מַקְשִׁיב עַל־שְׂפַת־אָָוָן שֶׁקֶר מֵזִין עַל־לְשִׁוֹן הַוֹת: ז לעֵג לֶרָשׁ חֵרֵף עֹשֵׂהוּ שָּׁמֵח לְאֵיד לְא יִנְקֵה: 6 עֵטֵרֵת זְקַנִים בְּנֵי בְגֵים וְתִפְאֶרֶת בְּגֵים אֲבוֹתֶם: 7 לאֹ־נָאוֶה לְנָבָל שְׂפַת־יֶעֶר אַׁף בִּי־ לְנָדֵיב שְׁפַת־שֶׁקֶר: 8 אֶבֶז־חֵז הַשֹּׁחַד בְּעֵינֵי בְעָלֵיו אֶל־כָּל־אֲשֶׁר יִפְגֶה יַשְׂבֶּיל: 9 מְכַסֶּה־בֶּשֹׁע מְבַקֵּשׁ אַהֲבֶה וְשֹׁגֶה בְׁדָבָר מַפְרֵיד אַדְ־מָרֵי 11 אַדְ־מָרֵי אַלְוּף: 10 תֵּחַת גִּעָרֶה בִמֵבֵין מֵהַכְּוֹת כִּסֵיל מֵאֶה: 11 אַדְ־מְרֵי

30 <sup>ר</sup>מַרְאֵה<sup>ז</sup>־ְעֵינִיִם יְשַׂמֵּח־לֵב שְׁמוּעָה טוֹבָה תְּדַשָּׁן־עֲצָם: 31 אָׂזָן שִׁמַעַת תּוֹכַחַת חַיִּיִם בְּקֶרָב חֲכָמֵים תָּלִין: 32 פּוֹרֵע מוּסָר מוֹאֵס נַפְּשָׁוֹ וְשׁוֹמֵע תוֹנֹכַחַת קוֹנֶה לֵב: 33 יִרְאַת יְהוָה מוּסַר חָכְמֵה וְלִפְגֵי כָּבְוֹד עַנְוֶה:

16:1 לְאָדֵם מֵעַרְכֵי־לֵב וֹמֵיְהוָה מַעֲגָה לָשְׁוֹן: 2 בֵּל־דַרְכֵי־אֵישׁ זַךָּ בְּעֵיגֵיו וְתֹכֵן רוּחִוֹת יְהוֵה: 3 גַּל אֶל־יְהוָה מַעֲשָׁידּ וְיִבּנוּ מַחְשְׁבֹתֵידָ: 4 בָּל בְּעַל יֵהוָה <sup>ר</sup>ּלְמַעַגָהוּ וְגַם־רָׁשָׁע לְיוֹם רָעֲה: 5 תּוֹעַבַת יֵהוָה כָּל־ גְּבַהּ־לֵב יֶד לְיָד לָא יִנְּקָה: 6 בְּחֶסֶד וֶאֲמֶת יְכַבּּר עָוֹן וּבִיִרְאָת יְהוָה סְוּר מֵרְע: 7 בִּרְצַוֹת יֲהוָה דַּרְכֵי־אֵישׁ גַּם־אוֹיְבָיו יַשְׁלִם אִתּוֹ: 8 טוֹב־ מַעַּט בִּצְדְקָה מֵרְב הְבוּאוֹת בְּלָא מִשְׁבֶּט: 9 לֵב אָדָם יְחַשֵּׁב דַּרְכֵּו מֵעַט בִּצְדְקָה מֵרְב הְבוּאוֹת בְּלָא מִשְׁבֶּט: 9 לֵב אָדָם יְחַשֵּׁב דַרְכֵּו וְיְהוָה יְכֵין צַעֲדְוֹ: 10 לָסָסֵם | עַל־שִׁבְתִי־מֶלֶד בְּמִשְׁבָּט לָא יִמְעַל־בָּיו מַעַט בִּצְדְקָה מֵרְב הְבוּאוֹת בְּלָא מִשְׁבֶּט: 9 לֵב אָדָם יְחַשֵּׁב דַרְכָּוֹ מֵעַט בִּצְדְקָה מֵרְב הְבוּאוֹת הָלָשִׁם | עַל־שִׁבְרִיכֵיאָנִים גַּרְכָים אוּתִי מְעַט בִּצְדְקָה מֵרְב וֹשְׁבוּים יָהוֹה מַעֲשָׁהוּ בַּרְבֵי אָדָם יחַשׁב מַעַט בִּצְדְקָה מֵרְב וּזְבוּאוֹת הָלָשִׁת בְּרָזוּ הַזְבָר מָשְׁבָּתִי־מֶלֶד בְּמַשְׁבָּטוּ מַעַט בִּשְׁרָים יָבְין בַּרְבוּים מָעָרִים הַעָּבָה מִים בּרָבוּתוּבָת מְרָבָה מָבוּר הַבָּרוּ הַמָּשְׁרָם מַיְרָבוּ מַקוּבי מַעָדָים וּבָין מַתַלָּד מַלְאַבִי־מָטוּנוּ זַמָעַרים הַשְׁבָר מָלָרִים יָשְׁרָים הָבוּהוּה הַכָּין בַּבְרָבָר מָרָבין הַבָּין בַּקָּרִים הָשָּבוּה הַמָרים הַבָּר הַיָּרִים הָיבָים שִּבָּר מָלָר מַבְעָב מַרְבוּים בַּרָים הַיָּבוּה וּבוּרִים הַיָּבוּחוּ בַּיָר מָעָר בַיָּים מַעָּבָר יְבָרָה מָבְרָבוּה הַבָּרָה הַרָעוֹים בַּרָים הַים הַיָּהוּה הַיּבָין הַיָּרָה מַיָּרָים הַיָּרָים הַיָּבָר מָבָר מָבָין הַמָּעַרים הַאָּבָה הַעָּבוּה הַנָר הַיָּרָרִים הָירִים הַירִים הַיָּקָר הַבּעָר שִׁבָר הַיָּרָה הוּים הַירִים הָינוּה מַיּים הַים מָיקוּין בַרָּרָים הַיהָרָים הָינִים הַיבּקרָה הַיּהָה הַיןהוּתוּבוּר הַיָּרָין בָיקוּין בַיקוּים בּרָיקוּים בּיין הָים הַיהוּבריים הָירָר הַיוּה הַירָים הָיהָה הַיקוּקוּר הַיָר הַיָרָים הָיוּין הוּירָים הַייָה הוּיר בּרָרָים הַירָין הַיוּין הַיים הַיין הַיין בּיים הַירָי הַישָּרָים מָיהוּין הַים הַירָים הַירָרָין הַיוּיה הַירָין הַיוּין הַי הַיר מ

8 זֶבַח רֲשָׁעִים תּוֹעֲבָת יְהוֶה וּתְפִלֵּת יְשָׁרֵים רְצוֹנִוֹ: 9 תּוֹעֲבָת יֶהוָה דֶרֶדְ רָשֶׁע וּמְרַדֵּף צְדָקָה יֶאֱהֶב: 10 מוּסֵר רָע לְעֹזֵב אָׁרַח שווֹגֵא תוֹכַחַת יָמְוּת: 11 שְׁאוֹל וַאֲבַדּוֹן נֶגֶד יְהוֶה אֵׁף כִּי־לִבְּוֹת בְּנֵי־אָדֶם: 12 לְא יֶאֱהַב־לֵץ הוֹבֵחַ לְוֹ אֶל־חֲכָמִים לְא יֵלֵדְ: 13 לֵב שְׂמֵח יֵיטָב ַפְּגֵים וּבְעַצְבַת־לֵב רַוּחַ נְכֵאֶה: 14 לֵב נָבוֹן יְבַקֶּשׁ־דֶּעַת <sup>ר</sup>וּפִי ֹבְסִילִים יִרְעֶה אָוֶּלֶת: 15 בְּלֹ־יְמֵי עָנֵי רָעֵים וְטִוֹב־לֵב מִשְׁתֶּה תָמִיד: וּסוֹב־מֱעַט בְּיִרְאַת יְהוֶה מֵאוֹצֶר רָב וּמְהוּמָה בְוֹ: 17 טָוֹב אֲרָחֵת 16 וֶרָק וְאַהֲבָה־שֶׁם מִשִּׁוֹר אָבוּס וְשִׂנְאָה־בִוֹ: 18 אֵישׁ חֵמָה יְגָרֵה מָדֵוֹן וְאֶרֶך אֵׁפִּיִם יַשְׁקִיט רִיב: 19 דֶּרֶךְ אָצֵל כִּמְשֵׂכַת חֱדֶק וְאָׁרַח יְשָׁרֵים ַרְסֵיל אָדָם בּוֹזֶה אִמְוֹ: 20 בֵּן חֲכָם יְשַׂמַּח־אָב וּכְסֵיל אָדָם בּוֹזֶה אִמְוֹ: 21 אָוָּלֶת יסֵיּלְבֶה שִׁמְחֶה לַחֲסַר־לֶב וְאֵישׁ הְּבוּנָה יְיַשֶּׁר־לֱכֶת: 22 הָפֵּר מֲחֲשָׁבוֹת בְּאֵין סֶוֹד וּבְרָב יוֹעֲצֵים תָּקוּם: 23 שִׂמְתָה לָאִישׁ בְּמַעֲנֵה־אֵיו וְדָבֶר בְּעִתּוֹ מַה־טְוֹב: 24 אִרֵח חֵיִּים לְמַעְלָה לְמַשְׂבֶּיל לְמַעַז סוּר מִשְׁאָוֹל מֶטָּה: 25 בֵּית גֵּאִים יִפָּח | יְהוֶה וְיַצֵּׁב גְּבָוּל אַלְמָנֶה: 26 תּוֹעֲבָת יֶהוֶה מַחְשְׁבְוֹת רֶע וֹּטְהֹרִים אִמְרֵי־נְעַם: 27 עֹבֵר בֵּיתוֹ בּוֹצֵע בֶּצַע וְשוֹגֵא מַתְּנְת יִחְיֶה: 28 לֵב צֵדִיק יֶהְגֶה לַעֲנָוֹת וּפִי רְשָׁעִים יַבֵּיעַ ַרְעִוֹת: 29 רְחִוֹק יֶהוָה מֵרְשָׁאָים וּתְפַלֵּת צַדִּיקִים יִשְׁמֶע:

19\*

חַכָּמִים עָשְׁרֵם אָוֶּלֶת בְּסִילִים אָוֶלֶת: 25 מַצֵּיל גֲפָּשׁוֹת עֵד אֶכֶּת זִיְפָח בְּזְבִים מִרְמֵה: 26 בִּיִרְאַת יֵהוָה מִבְטִח־עֵׁז וּיְבְנָיו יִהְיֶה מַחְסֵה: 27 יִרְאַת יֵהוָה מְקוֹר חַיִּים לְסׁוּר מִמִּקׁשִׁי מֶוֶת: : 28 בְּרָב־עֶם הַדְרַת־מֶלֶד וּבְאֶפָס לְאֵם מְחִתַּת <sup>ר</sup>רוֹזֵן<sup>י</sup>: 29 אֶרֶך אַפַּים רַב־תְבוּגָה וּקְצַר־רּוּח מֵרִים אָוֶלֶת: 30 חַיֵּי בְשִׁרִים לֵב מַרְפֵּא וּרְקָב עַצְמוֹת קּנְאֵה: 31 עַשֵׁק־דֶּל חֵרֵף עֹשֵׁהוּ וֹמְכַבְּרֹו חֹגֵן אֶבְיוֹז: 29 בְּרָב־עָם קּנְאֵה: 31 עַשֵׁק־דֶל חֵרֵף עֹשֵׁהוּ וֹמְכַבְּרֹו חֹגֵן אֶבְיוֹז: 29 בְּרָמִתוּ הַדְעַת־מֶלֶד וּבְאֶפָס לְאֵם מְחִתַּת <sup>ר</sup>רוֹזֵן<sup>י</sup>: 29 אֶרֶף אַפַּים רַב־תְבוּגָה הַקַּצַר־רּוּח מֵרִים אָוֶלֶת: 30 חַיֵּי בְשָׁרִים לֵב מַרְפָּא וּרְקָב עַצְמוֹת הַנְשָׁר רְשָׁע וְחֹסֶה <sup>ר</sup>ְבְתָמוֹז צַדִּיק: 33 בְּבָרָ וֹחֹגֵן אֶבְיוֹז: 23 בְּרָעָתוּ יִדְּתֶה רְשָׁע וְחֹסֶה <sup>ר</sup>ְבְתָמוֹז צַדְיקָה תְרוֹמֵם־גּוֹי וְחָסֶד לְאָמַים חַשָּאת: 35 רְשָׁע וְחֹסֶה וּדְבַר־שָׁצָב מַשְׁבָּיל וְעָבְרָתוֹ תִהְיָה מֵבְישׁו זוֹתָן ישָׁיב חַמְה וּדְבַר-שָׁצָב מַשְׁבָּי הְשָּעָנָה בַיָּים אָוּשְׁרָם תַרְיּמָם הַיָּשָּתוּ זְשָּים חַמָּמוּ וּדְבַר זּאָנָר זיָר מָבָים מְזַים הַיָּנָים הַשָּעָה מַמָּה וּדְבַר-שָׁע הַשְׁים חַמָּתוּים הַנָּים הַיּנְים אָשָּבָר מַשְׁבָים הַבּישָׁת הַשְׁים בַמְתָה וּשְּכָר הַמָּחָם וּתִיים וְזָלָם בַּבָּרָים מִיּנִים הַבּיםים מַיעָים וּחוּבָים: לְשָׁע נָעְבָּרַת אוּשְׁרָם הּוֹבְרָם מִיתְים וּחָכָר הַיוֹם זּרָים בְּבָר בְעָשָּים בְעָמָים מַיַים הַיָּכָרם בְשָׁע נָעְבָר הוֹשְׁמָר הּוֹבָתוּי הַעָרָם: 6 רְשָּביים וּחָכָר בִיין אָרָים וּשָּרָים זיוָרים בַּבָים מְרָים מָים מָרָרם וּשָּלָר בָיּיָים גַישָּים בַים מָעוּין הַיָים בַיּבָים אָעָרָים בּית בְשָּעוּים הַמְהַים הַיבִים אָעָרָים הַים הַים הַים וּים בּרָשָּים אָרָים הַים בּיים אָבָים אָיוּהוּים בַישָּים בְישָּרָים מָיים מָּים בְיבָר בּישָּרָים בַיּשָרָים בּעָרָם בּישָרָים מָרָים הַיתוּין בּישְרָים בְעָרָים בְיַים בּיתוּין בּעָין הַיָרָים הַים מָרָים מָים בּייוּים בַייָים בְייָרָים בַיָּבָים בְיָרָם בָיָין בּיין בָים בּישָּים בְייָים בָים בְיים בְיים בְייוּים בְייוּבָין בּיַרָים בָיוּים בָיו

14:27 אראת M S (געלאס (πρόσταγμα) (reform, equal) || לסור || לסור || M ] לסיר (ποιει δὲ ἐκκλίνειν) (graph 1→ י) || 14:28 לסיר G (ποιει δὲ ἐκκλίνειν) (graph 1) אלסיר (β רזון (metath דבדעתו [ 14:32 ברעתו M G (לא אמאנת מטֿדטט) (זו  $\rightarrow$  14:32 בדעתו S S (העם מת  $\leftarrow \pi$ ם מותו [ (גרא לה מת  $\rightarrow$  theol, orth) א במותו (גרא לה ה || 14:33 (בכון [ M נבון M ] (גבאינא G (מֹצְמטָק) א נכון [ M נבון 14:33] (קודע || 14:33 ) || א נבון 14:33 לא G α' θ' (οὐ) S (בא) (logic) || 14:34 וחסר [ Μ ] אלא G (ἐλασσονοῦσι  $\delta \epsilon$ ) S (העביה) (graph  $\tau \rightarrow \tau$ ) || 14:35 משכיל 14:35bc) (העביה) (graph  $\tau \rightarrow \tau$ ) (displ from 14:32) || ועברתו [ M ועברתו \* G (דאָ לב במטדסט בטסדףסקוֹם) (graph  $\neg \rightarrow$  metath  $\neg \rightarrow \rightarrow$  metath  $\neg \rightarrow \rightarrow$  15:1 init + 1 stichos G (1a) (transition) || 15:2 אולת M ≈ G (אמאמ ( מאמא M ≈ G (אמאמ) (metath אולת ) → ופלס [ M וסלף M : פלס (ל M אופלס (ל M אופלס (ל M אופלס (ל M ) (ל M ) (ל M ) (ל M ) (ל Mשבר M ] שבע G ( $\pi\lambda\eta\sigma\theta\eta\sigma\epsilon\tau\alpha\iota$ ) S (ממרב) (graph  $\neg \rightarrow \lor$  paleo  $\mathbf{q} \rightarrow \mathbf{\sigma}$ ) || א ברבות (לי, גבית 15:6 א ברבות א: G (לי  $\pi\lambda$ εοναζούση) (M: haplog ב $\rightarrow$  ב; G: near dittog בר  $\rightarrow$  בר, graph  $\rightarrow$  ותבואת M<sup>MSS</sup> G (אמא מלא לל) מ $\lambda'$  $(\lambda o \gamma i \sigma \mu o i \delta \epsilon) S ( בוב \leftarrow בו M (distant dittog ובת בואַת [$ 

וּתִּשְׁמְרוּם<sup>7</sup>: 4 בְּאֵיז אֲלָפִים אֵבָוּס אֵבוּס בֵּר וְרָב־אְבוּאוֹת בְּכָח שׁוֹר:
5 עַד אֲמוּנִים לָא יִכַזֵּב וְיָפִיח בְּזָבִים עֵד שֶׁקָר: 6 בִּקָשׁ־לֵץ חָכְמָה
זְאָז וְדָעַת לְנְבוֹן נְבֶל: 7 לֵךְ מְנָגָד לְאֵישׁ בְּסֵיל וּבַל־יָדַשְׁתָ שִׁבְּתִי
דְעַת: 8 חָכְמַת עֻרוּם הְבֵיז דַרְכֵּוֹ וְאָוֶלֶת בְּסִילִים מִרְמֶה: 9 אֱוֹלִים זְיַשְׁתִי
דְעַת: 8 חָכְמַת עֻרוּם הְבֵיז דַרְכֵּוֹ וְאָוֶלֶת בְּסִילִים מִרְמֶה: 9 אֱוֹלִים זְיַשְׁתִי
זְלָיז אִשְׁם וּבִיח יְשָׁרִים רְצָזוֹ: 10 לֵב יוֹדַע מְרַת נִבְשְׁחוֹז וּבְשִׁחְתוֹ
זְשָׁתִי אַשֶׁם וּבִיז יִשְׁרִים רְצָזוֹ: 10 לֵב יוֹדַע מְרַת נִבְשָׁחוֹז וּבְשִׁחְתוֹ
זְשָׁרִים זְשְׁרִים זְבְיּחִיים בְיָשְׁתִים הַצְמָר וּאָהָל יְשָׁרִים יַבְרָחַ: 12 יֵשׁ הָירָשְׁרִים זְשָׁרִים הְבָזוֹ זּבְשִׁתוּים הַבְיז דַרְכֵי־מֵוֶת: 13 גַם־בִּשְׁחוֹז וּבְשָׁחְלִוֹי יַבְאַבילָבון וּאָהָרִים יַפְרָיחַ: 12 יֶשְׁרִים הַבְּרִיחָ אָשָׁרִים רָצוֹזי וּשָׁבָים זוּג זַר וּבָרָים יַבְּרָיחַ וּבָרַים בְעָשְׁחוֹז וּז יִשְׁבִית וְשָׁבִים בְעָשָׁר וּאָרָרִים הַבְּרִיחָבוּ גַין שָׁרִים הַבְיּז וּאָהָרִים הַאָּמִים וּאָהָביי זַר וּאָתוּים הַבָּיז וּשְׁבָרים בְּבִים בְעָבִים בַּמְרִים בְּבְיּרִים בְּבִירִים בְּרָיחַ בַרְים בְּבָרִים בְּעִים הַשְׁמָרוֹם בְּבִירִם בְּשִׁרִים בַּבְיחוֹים בְעָבַרִים בַּבְיוּרִים בְבָיוּ ז אָשָׁרִים בְעָרִים בְּבְיוֹים מְבִיים בְעָרִים בַיּבְירִים בְּבָרוּים בְּבָרִים בְּעָרִים בַיּבְיוּרִים בְּיּשְׁרִים בְּיּשְׁרִים בְיּשְׁרָים בְיּבִים בְיַשְׁבוּים בְעָבִים בְיּשְׁים בּבּין בְיּשְׁרָים בְיּשְׁים בּיוּים בְעָרִים בְּשְׁיוּים בּבִיחוֹ בְעָרוּים בּבּיּשְׁיִים בְּישְׁבוּים בְעָרוּב זּבוּבוּים בְּשְׁתוּים בִיחוּים בִים בְיּשְׁבִים בּיוּים בְיּשְׁים בּים בִיעִים בּעָרוּים בְעָרוּים בּין בְישְבּיים בַין בּרים בְיים בִיין בְיבִים בְיּבוּביים בּין בּעָים בְּעָרוּייים בִיים בְיים בּין בְישָּיים בּיוּבָים בַיּבִיים בְיּבּיים בּים בְיוּים בּבּיוּ בְעָיוּים בְּעָרוּים בִיים בְיוּבוּיוּים בּיוּשָרוּין בְיין בְיים בּיין בְים בּייןנִים בּיוּים בּיוּין בְעָרוּ בְשָּיין בּיין בְים בְיין בּיין בְיים בְיין בְירָים בּיין

יִשְׂמֶח וְגֵר רְשָׁעֵים יִדְעֶדְ: 10 רַק־בָּזָדוֹן יִתֵּן מַצֶּה וְאֶת־נוֹעָצֵים יִשְׁמֵח וְגֵר בְשָׁעֵים יִדְעֶדָ: חְכְמֵה: 11 הוֹז 'מְבֹהֶל' יִמְעֵט וְקֹבֵץ עַל־יָד יַרְבֶּה: 12 הּוֹחֶלֶת מֱמֻשָּׁכָה מַחֲלָה־לֶב וְעֵץ חֵׁיִּים תַּאֲוָה בָאֶה: 13 בֶּז לֶדְבָר יֵחֶבֶל לָוֹ וִירֵא מִצְוָה הַוּא יְשָׁלֶם: 14 תּוֹרַת חֲכָם מְקוֹר חַיֵּים לְסוּר מִמָּקְשׁי מֶוֶת: 15 שֵׂכָל־טוֹב יִתֶּזִ־חֵז וְדֶרֶדְ בַּגְדֵים <sup>ו</sup>אֵידֵם<sup>ז</sup>: 16 בְּל־אֲרוּם ַיַעָשֶׂה בְדֶעַת וֹּכְסִיל יִפְרִשׂ אָוֶלֶת: 17 מַלְאָד רָשָׁע יִפָּל בְּרֶע וְצִיר אֶמוּנִים מַרְפֵּא: 18 רֵישׁ וְקָלוֹז פּוֹרֵעַ מוּסֶר וְשׁוֹמֵר תּוֹכַחַת יְכֵבֶּד: נּהַאָוֶה וֶהְיָה תֶעֱרָב לְגָפָשׁ וְתוֹעֲבָת כְּסִילִים סְוּר מֵרֶע: 20 'הָלָוֹדְ' אֶת־חֲכָמֵים 'וַחֲכֶם' וְרֹעֶה כְסִילִים יֵרְוֹעַ: 21 חֲטָּאִים הְרַדֵּף רָאֶה וְאֶת־צַׁדִיקִים יְשַׁלֶּם־טְוֹב: 22 טוֹב יַנְחֵיל בְּגֵי־בָגִים וְצָפְוּן לַצַּדִּיק חֵיל חוֹטֵא: 23 <sup>ר</sup>רַב אֹכֵל<sup>ז</sup> גִיר רָאשָׁים וְיֵשׁ גִׁסְפֶּה בְּלָא מִשְׁפֵּט: 24 חוֹשֵׂך שֶׁבְטוֹ שוֹנֵא בְנָוֹ וְאֹהֲבוֹ שֶׁחֲרָוֹ מוּסֶר: 25 צַדִּיק 14:1 רְּחָכְמְוֹת אַכַל לְשְׁבַע נַפְּשֶׁו וּבֶטֶן רְשָׁעֵים תֶּחְסֵר: פ (נָשִׁים) בָּנְתָה בֵיתָה וֹאָוֹּלֵת בִּיָדֵיהָ תֶהֵרְסֵנּוּ: 2 הוֹלֵך בֵּיָשִׁרוֹ יָרֵא יִהוֶה וּנִלְוֹז דְּרֶכֵיו בּוֹזֵהוּ: 3 בֵּפִי־אֵוִיל חְטֵר גַּאַוֶה וְשִׂפְתֵי חֵכָמִים

וּגְמָוּל יְדֵי־אָדָם רְּשָׁוּב לו: 15 דֶרֶךְ אֱוִיל יְשֶׁר בְּעֵינְיו וְשֹׁמֵעַ לְעֵצֶה חָבֶם: 16 אֶוֹיל בֵּיּוֹם <sup>ר</sup>יוֹדַע<sup>ּ</sup> פַּעְסָוֹ וְכֹסֶה קָלָוֹן עָרוּם: 17 יָפִּיַח אֶמוּנָה יַגִּיִד עֶדֶק וְעֵד שְׁקָרֵים מִרְמֶה: 18 יֵשׁ בִּוֹטֶה כְּמַדְקְרָוֹת חֶרֶב וּלִשְׁוֹן חֵכָמֵים מַרִפֵּא: 19 שֵׂפַת־אֱמֵת תִּכְּוֹן לָעָד וִעַד־אַרִגִּיעָה לִשְׁוֹן שֵׁקֶר: 20 מֶרִמֶה בָּלֶב־חִרְשֵׁי רֶע וּלִיּעֵצֵי שָׁלוֹם שִׂמְחֶה: 21 לא יְאָנֶה לַצַדְיק כָּל־אָוֶן וּרְשָׁעִים מֶלְאוּ רֶע: 22 תּוֹעֲבָת יֵהוָה שִׂפְתֵי־שֶׁקֶר וְעֹשֵׂי אֱמוּנָה רְצוֹגוֹ: 23 אָדָם עֲרוּם כִּסֶה דֶעַת וְלֵב כְּסִילִים יִקְרָא אָוֶלֶת: 24 יַד־חָרוּצִים תִּמְשָׁוֹל וּרְמִיָּה תִּהְיֶה לְמַס: 25 דִאָגֶה בְלֵב־אֵישׁ יַשִׁחֶנָּה וְדָבֶר טְוֹב יִשִׂמְחֵנָּה: 26 יָתֵר ַמֵרַעֵהוּ צַדֶּיק וְדֶרֵךְ רִשָּׁעֵים הַּתִעֵם: 27 לֹא־יַחַרֹדְ רִמִיָה צֵידָוֹ וְהוֹן־ אָדֵם יָקֵר חָרִוּץ: 28 בְּאִׂרַח־צְדָקָה חַיָּיֵם וְדֶרֶדְ נְתִיבֶה <sup>ר</sup>אֶל<sup>ז</sup>־מֱוֶת: ואַן הַכָם מְוּסַר אָָב וְלֵץ לֹא־שָׁמַע גְּעָרֵה: 2 מִפְּרֵי פִי־אֵישׁ 13:1 יַאכַל טָוֹב וְגֶפָשׁ בֹּגְדַים חָמֶס: 3 נֹצֵר פִּיו שׂמֵר נַפְשָׁו פּשֵׂק שְׁפָתָיו מִחִתּה־לְוֹ: 4 מִתִאַוֶּה וָאַיִן נַפִּשׁוֹ עָצֵל וְגֵפֶשׁ חָרָצִים תִּדֶשֶׁן: 5 דְבַר־ שֶׁקֵר יִשְׁנֵא צַדֵּיק וְרָשָׁע יַבִאֶישׁ וִיַחִפֵּיר: 6 צֵדָקָה תִּצְׂר תָּם־דֶּרֶדְ ּוְרִשְׁעָה הְסַלֵּף חַטֶּאת: 7 יֵשׁ מֶתְעַשֵּׁר וְאֵין כָּל מִתְרוֹשֵׁשׁ וְהָוֹן רֶב: 8 כִּפֵר גֵפֶשׁ־אֵישׁ עָשְׁרָוֹ וְׁרָשׁ לֹא־שָׁמֵע גְּעָרֶה: 9 אוֹר־צַדִּיקִים 8 14\*

אַדְ־טָוֹב תִּקְוָת רְשָׁעֵים עֶבְרֶה: 24 יֵשׁ מֱפַזֵּר וְנוֹסֶף עָוֹד וְחוֹשֵׂך מִיּשֶׁר אַדְ־לְמַחְסוֹר: 25 גֵפֶשׁ־בְּרָבֶה תְדָשָׁז וֹמַרְוֶה גַּם־הָוּא <sup>ו</sup>יִרְוָא<sup>י</sup>: 26 מְגַעַ בֶּר יִקְבָהוּ לְאָוֹם וּבְרָכָה לְרָאשׁ מַשְׁבִּיר: 27 שְׁחֵר טוב יִבַקֵשׁ רָצָוֹן וִדֹרֵשׁ רָעֶה תִבוֹאֵנּוּ: 28 בּוֹטֵח בֶּעָשִׁרוֹ הְוּא יִפְּל וְכֵעָלֶה צַדִּיאָים יִפְרֶחוּ: 29 עוֹבֵר הֵיתוֹ יִנְחַל־רְוּחַ וְעֶבֶד אֶׁוֹיל לַחֲכַם־לֵב: 30 פְּרִי־צֵדִיק עֵץ חַיָּיֶם וְלֹקֵח וֹנְפָשְׁוֹת' חָכֶם: 31 הֵן צֵדִיק בָּאָרֶץ יְשֶׁלֶם אַׁף כִּי־רָשֶׁע וְחוֹטֵא: 12:1 אֹהֵב הֲוּסָר אָהֵב דֶעַת וְשוֵא תוֹכַחַת בֶּעַר: 2 טוֹב יָפִיק רָצוֹן מֵיְהוֶה וְאָישׁ מְזִמְוֹת יַרְשִׁיעַ: 3 לֹא־ יִפּוֹז אָדָם בְּרֶשַׁע וְשִׁרֶשׁ צַׁדִּילִים בַּל־יִמְוֹט: 4 אֵשֶׁת־חֵיל עֲטֶרֶת בַּעְלֶה וּכְרָקֶב בְּעַצְמוֹתֵיו מְבִישֶׁה: 5 מַחְשְׁבִוֹת צַדִּיקֵים מִשְׁפֵּט תַּחְבֻּלְוֹת רְשָׁעֵים מִרְמֶה: 6 דִּבְרֵי רְשָׁעֵים אֶֶרָב־דֶּם וּפִּי יְשָׁרִים יַצִּילֵם: 7 הָפִוֹדְ רְשָׁעֵים וְאֵינְגֵם וּבֵית צַדִּיקִים יַעֲמְד: 8 לְפִי־שֵׂבְלוֹ יְהֻלַּל־אֵישׁ וְנַעֲוֵה־לֵב יִהְיֶה לְבְוּז: 9 טְוֹב גָקְלֶה <sup>ו</sup>וַעֲבֵרי לֵו ּמְמִתַכַּבֵּׁד<sup>י</sup> וַחֵסַר־לֶחֵם: 10 יוֹדֵעַ צֵּדִיק נֵפֶשׁ בְּהֶמְתָּוֹ וֵרַחֵמֵי רִשְׁעִים אַכְזָרִי: 11 עֹבֵד אֲדְמָתוֹ יְשְׂבֵּע־לֶחֶם וּמְרַדֵּף רֵיהֵים חֲסַר־ לֵב: 12 חָמַד רָשָׁע מְצִוֹד רָעֵים וְשְׁרֶשׁ צַדִּיקִים יִתֵּן: 13 בְּפֶשׁע שֵׁפָתִים מוֹקֵשׁ רָע וַיָּצֵא מִצְרָה צַדֵּיק: 14 מִפְּרֵי פִי־אֵישׁ יִשְׂבַע־טָוֹב

10:32 ישדם 11:3 (מדססדάζει) א ידעון ( (מדם א G (מדססדάζει)) איבעון (ג. ב- ב- 10:32) (graph איבעון 11:3  $M^Q$  ]  $M^K$  (graph '→ 1) || 11:4 M S ] > OG (unc, equal) || 11:6 בבאלמי (ד (ד άπωλεία αὐτῶν) S (בבאלמי) V (in insidiis suis) ] M (near haplog  $L \to J$  ) || 11:7 ותוחלת M S (סמבוסים) ] ותהלת G (דט אנאמט (graph  $\neg \rightarrow \neg$ , orth) || 11:8–10a MGS > 6Q30 (homoi אבד∩ אבד 11:8 (אבד אבד M S (ح, אסל, נאר) β (לא G ( θήρας) (graph  $\neg \rightarrow \gamma$ ) || נאבינה א א א א א א א א א א א א יחלצו (נאבינה ביא G (εὔοδος) \* G (εΰοδος) (transp צלח → חלצ 11:10b-11a MS] > OG (homoi (קרת ∩קריה) || 11:14 בן M S (حجه) G (ὥσπερ φύλλα) (antic 11:28) || 11:19 دربار (حجه) Μ S (حجه) M ] (בי $\rightarrow$  G (viós) S ( $\rightarrow$ בן (graph  $\rightarrow \rightarrow$  ב)

ַרְשָׁע תּאבִר תִקוֶה וְתוֹחֶלֶת אוֹנִים אָבְדָה: 8 צַדִּיק קַ רְשָׁע תּאבִר מִקוֶה וְתוֹחֶלֶת אוֹנִים אָבְדָה: מִצְרֵה נֶחֶלֶץ וַיְּבָא רְשֵׁע תַּחְתָּיו: 9 בְּפֶה חֲנֵף יַשְׁחֶת רֵעֵהוּ וֹבְדַעַת צַדִּיקֵים יֵחָלֵצוּ: 10 בְּטָוּב צֵדִיקִים תַּעֲלָץ קִרְיֶה וּבַאֲלִד רְשָׁעֵים רְוֶה: וּז בְּבִרְפַּת יֲשֶׁרִים תְּרַוּם קֶרֶת וּבְפֵי רְשָׁעִים תֵּהֶרֵס: 12 בָּז־לְרֵעֵהוּ חֲסַר־לֵב וְאִישׁ תְּבוּנְוֹת יַחֲרִישׁ: 13 הוֹלֵדְ רָכִיל מְגַלֶּה־סָּוֹד וְנָאֱמַן־ ּרוּחַ מְכַמֶּה דָבֶר: 14 בְּאֵין הַּחְבֶּלוֹת יִפָּל־עָם וּתְשׁוּטָה בְּרַב יוֹעֵץ: ַדע־יֵרוֹעַ כִּי־עָרַב זֶר וְשֹׁגֵא תֹקְעִים בּוֹטֵחַ: 16 אֵשֶׁת־חֵן תִּתְמְד ּפָּבְוֹד וְׁעָרִיצִּים יִתְמְכוּ־עְּשֶׁר: 17 גַּמֵל גַפְשׁוֹ אֵישׁ חֶסֶד וְעָבֵר שְׁאֵרוֹ אַכְזָרִי: 18 רָשָׁע עֹשֶׂה פְּאֵלַת־שֶׁקֶר וְזֹרֵע אֲדָקָה שֵׂכֶר אֲמֶת: 19 בֵּן־ אַדָקָה לְחַיָּיֵם וּמְרַדֵּף רָעָה לְמוֹתוֹ: 20 תּוֹעֲבַת יֱהוָה עִקְשֵׁי־לֵב וּרְצוֹנוֹ תְּמֵימֵי דֶרֶדְ: 12 יָד לֵיָד לֹאֹ־יְנֶקֶה הֶע וְזֵרַע צַדִּיקִים נִמְלֵט: 22 גֶזֶם זֶהָב בְּאָף חֲזֵיר אִשֶׁה יְפָָה וְקָרַת טֱעַם: 23 תַּאֲוַת צַדִּיקִים

31 פִּי־אַדִיק יָנוּב חָכְמָה וּלְשִׁוֹן תַּהְפָּכוֹת תִּכְּרֵת: 32 שִׂפְתֵי אֲדִיק

רּזַבּעַוּן' רָצֵוֹן וּפִי לְשָׁעִים תַּהְפֵּכְוֹת: 11:1 מאוני מֵרְמָה תּוֹעֲבָת יְהוֶה 'יַבּעַוּן'

3 תַּמַת יְשָׁרֵים תַּנְחֵם וְסֶלֶף בּוֹגְדֵים וּיְשָׁדֵּם": 4 לא־יוֹעֵיל הוֹן בְּיָוֹם

עָבְרֶה וּצְדָקָה תַּצִיל מִמְוֶת: 5 צִדְקַת תָּמִים תְּיַשֵּׁר דַּרְכָּוֹ וּבְרִשְׁעָתוֹ

ּוְאֶבֶן שְׁלֵמֶה רְצוֹנְוֹ: 2 בֶּא־זָדוֹן וַיָּבָא קָלָוֹן וֱאֶת־צְנוּעֵים חָכְמֶה:

יִפִּל רְשֵׁע: 6 צִדְקָת וֲשָׁרִים תַּצִּילֵם וּבְהַוְּתָׁם בֹּגְדֵים יִלְּבֵדוּ:

13\*

8 חַכַם־לֵב יַקָּח מִצְוֹת וֶאֱוֵיל שְׁפְתַׁיִם יִלְבֵט: 9 הוֹלֵך בֵּתֹם יֵלֶך בֶּטַח וּמְעַקֵּשׁ דְּרָכָיו יוָדֵעַ: 10 קְׁרֵץ עֵיוָ יִתֵּן עַצֵּבֶת וּוִתוֹכַחַת מִגְלָה שָׁלְוֹם': 11 מָקוֹר חֵיִים פֵּי צַדֵּיק וּפֵי רְשָׁעִים יִכַּמֶה חָמֶס: 12 שֶׂנָאָה תְּעוֹרֵר מְדָנִים וְעַל כָּל־פְּשָׁעִים תְּכַסֶּה אַהֲבֶה: 13 בְּשִׂפְתֵי נָבוֹן תִּמְצֵא חְרְמֶה וְשֵׁבֶט לְגֵו חֲסַר־לֵב: 14 חֲרָמֵים יִצְפְּנוּ־דֶעַת וּפִי־שֶׁוִיל מְחָתֶּה קְרֹבֶה: 15 הְוֹן עֲשִׁיר קִרְיֵת עֻזֶוֹ מְחִתַּת דַּלִים רֵישֵׁם: וּפְעֻלַת צַדְּיק לְחַיֶּיֶם תְּבוּאַת רְשָׁע לְחַטֶּאת: 17 אַרח לֲחַיִים 16 שׁוֹמֵר מוּסֵר וִעוֹזֵב תּוֹכַחַת מַתְעֵה: 18 מְכַסֵּה שֶׁנָאָה שִׁפְתֵי־שָׁקֵר וּמוֹצָא דְׁבָּה הַוּא כְסֵיל: 19 בְּרָב דֲבָרִים לָא יֶחְדַל־אָּשַׁע וְחֹשֵׂך יִשְׁפָתָיו מַשְׂבִּיל: 20 בֶּסֶף וֶבְחָר לְשִׁוֹן צַדֵּיק לֵב רְשָׁעִים בִּמְעֵט: 21 שִׁפְתֵי צֵדִיק יִרְעָוּ רַבֶּים וֱאֵוילִים בַּחֲסַר־לֵב יָמְוּתוּ: 22 בִּרְבַּת יֶהוֶה הֵיא תַעֲשֶׁיר וְלְאֹ־יוֹסֵף עֵצֶב עִמֱה 23: בִּשְׂחִוֹק לֵכְסִיל עֲשָׂוֹת זִמֶּה ןְׁחָכְמָה לְאֵישׁ תְּבוּגֶה: 24 מְגוֹרַת רֲשָׁע הֵיא תְבוֹאֶנוּ וְתַאֲוָת צַדִּיקֵים יִהֵּן: 25 בַּעֲבִוֹר סוּפָה וְאֵין רָשֶׁע וְצַדִּיק יְסְוֹד עוֹלֶם: 26 בַּחָׂמֶץ | לַשִּׁנַּיִם וְבֶעָשָׁן לְעֵינְיִם בֵּן הֶעָצֵׁל לְשֹׁלְחֵיו: 27 יִרְאַת יֵהוָה תּוֹסִיף יָמֵים וּשְׁוֹת רְשָׁעֵים תִּקְצְׂרְנָה: 28 תּוֹחֶלֶת צַדִּיקִים שִׁמְחֶה וִתִקוֵת רִשָּׁעֵים תּאבֵד: 29 מָעוֹז לַתּם דֶרֵך יָהוֶה וֹמִחִתָּה לִפְעֵלֵי אֶוֶן: 30 צַדְּיק לִעוֹלֶם בַּל־יִאָּוֹט וֹרִשָּׁעִים לְא יִשְׁכָּנוּ־אֶרֵץ:

בֵּן מַשִּׂבֵּיל נִרְדֵּם בַּקַּאָיר בֵּן מֵבֵישׁ: 6 בֵּרָכוֹת לִרְאשׁ צַדִּיק וּפֵי

ּרְשָׁעִים יְכַמֶּה חָמֱס: 7 זֵכֶר צֵּדִּיק לִבְרָכֶה וְשֵׁם רְשָׁעֵים יִרְקֶב:

פּ 1:1 חֶכְמוֹת בְּנְתָה בֵיתָה <sup>ו</sup>הִצֶּבְה<sup>ז</sup> עַמּוּדֶיהָ 🛚 אֵהֵבוּ מֵוֵת: שִׁבְעֵה: 2 טָבְחָה טֵבְחָה מָסְבָה יֵינֶה אַ עְרְבָה שֵׁלְחָגֵה: 3 שֵׁלְחָ נַעֲרֹתֵיהָ תִקְרֶא עַל־צַּׁבֵּׁי מְרִמֵי קֶרֶת: 4 מִי־שֶׁתִי יָסֵר הֵנְּה חֲסַר־צֵׁב ּאַמְרָה' לְוֹ: 5 לֲכוּ לַחֲמָוּ בְּלַחֲמֵי וֹּשְׁתוּ בְּיַיָן מֶסֶכְתִּי: 6 עַזְבָוּ פְתָאיָם וַחְיֶוּ וְאשִׁרוּ בְּדֶרֶךְ בִּינֶה: (7 יָֹמֵׁר | לֵּץ לֹמֵחַ לָוֹ קָלֶוֹן וּמוֹכִיחַ לְרָשַָׁע מוּמִוֹ: 8 אַל־תִּוֹכַח לֵץ פֶּן־יִשְׂנָאֶךָ הוֹכַח לְחָכָם וְיֶאֶָהָבֶךָ: 9 תֵּן לֶחָכָם וְיֶחְכַּם־עָוֹד הוֹדַע לְצַדִּיק וְיִוֹסֶף לֶקַח:) (10 תִּחִלֵּת פ חַכְמָה יִרָאַת יְהוֶה וְדַעַת קִדֹשִׁים בִּינֶה:) 11 כִּי־בֵי יִרְבוּ יָמֵיד וְיוֹסֵיפּוּ לְדָׁ שְׁנָוֹת חַיֶּים: (12 אִם־חָכַמְתָּ חָכַמְתָּ לֶדְ וְלַצְתָּ לְבַדְדָ תִשְּׂא:) 13 אֵשֶׁת כֵּסִילוּת הְמִיָּה פְׁתַיּוּת וּבַל־יָדְעָה מֶה: 14 וְיָשְׁבָה לְפֶתַח בִּיתֶה עַל־בִּמָּא מְרַמֵי קֶרֶת: 15 לִקְרָא לְעִׂבְרֵי־דֶרֶד ַהְמְיַשְׁרִים אְרְחוֹתֶם: 16 מִי־פֶּתִי יָסֵר הֵנְּה וַחֲסַר־לֵב וּןְאַמְרָה לִו: 17 מַיִם־גְּנוּבִים יִמְתֶּקוּ וְלֶחֶם סְתָרֵים יִנְעֱם: 18 וְלֹאֹ־יָדַע בִּי־רְפָאֵים יָשֶׁם בְּעִמְהֵי שְׁאוֹל קְרָאֶיהָ: פ

10:1 בֵּן חֲכָם יְשַׁמַּח־אֶב וּבֵן לְּסִיל תּוּגַת אִמְוֹ: 2 לאֹ־יִוֹעִילוּ אוֹצְרַוֹת רֶשַׁע וֹצְדָקָה תַּצִיל מִמְוֶת: 3 לְאֹ־יַרְעֵיב יֶהוָה גֶפָשׁ צַדֵּיק <sup>ו</sup>וְחַיַּת<sup>י</sup> רְשָׁעֵים יֶהְדְּף: 4 וֹבֵאשׁ עֹשָׂה' כַּף־רְמִיֶּה וְיַד חָרוּצִים תַּעֲשָׁיר: 5 אֹגֵר בַּקַיִץ

משלי

10\*

מְלָכֵים יִמְלָכוּ וְׁרוֹזְגִים יְחִקְקוּ עֱדֶק: 16 בֵּי שָׂרֵים יָשֶׂרוּ וֹנְדִיבִּים כָּל־ שִׁפְטֵי וֹאֶרֶץ": 17 אֲנִי וֹאהֲבֵיּן אֵהֶב וֹמְשַׁחֲרֵי יִמְצָאֵנְנִי: 18 עְשֶׁר־וְכָבְוֹד אִתֶּי הְוֹן עֶׁתֵׁק וּצְדָקֶה: 19 (טְוֹב פֵּרְיִי מֵחָרְוּץ ומִפֶּז וּתִבוּאָתִי מִבֵּסֶף נִבְחֶר:) 20 בִּאְׂרַח־צִדְקֵה אֵהַלֵּך בִּתוֹדְ נִתִיבְוֹת מִשְׁפֵּט: 21 לְהַנְחֵיל אֹהַבֵי | יֵשׁ וְאֹצְרְתֵיהֵם אֲמַלֵּא: ַרַסַלָּתִי באשִׁית דַּרְבָּוֹ הֶדָב מִפְעָלָיו מֵאֶז: 23 מֵעוֹלָם <sup>ר</sup>נְסַלָּתִי 22 מֵרֹאשׁ מִקַּדְמֵי־אֶֶרֶץ: 24 בְּאֵין־תְּהֹמָוֹת חוֹלֶלְתִּי בְּאֵין מַׁעְיָנוֹת רּנִבְכֵי־־מֶיִם: 25 בְּטֶרֶם הָרֵים הָטְבֶּעוּ לִפְנֵי גְבָעֵוֹת חוֹלְלְתִי: 26 עַד־ ַלְא אֲשָׂה אֶָרֶץ וְחוּצְוֹת וְׁרֹאשׁ <sup>ו</sup>עַפְרָוֹת<sup>י</sup> תֵּבֵל: 27 בַּהֲכִינָוֹ שֶׁמַיִם שָׁם אָגִי בְּחוּקוֹ חוּג עַל־פְּגֵי תְהוֹם: 28 בְּאַמְצִוֹ שְׁחָקִים מִמֶּעַל 'בְּעַזְּוֹו' עִינְוֹת תְּהוֹם: 29 בְּשׁׁוּמָוֹ לַיְּם | חֻקּוֹ וְמַיִם לְא יֵעַבְרוּ־פֵּיו רּבְּחַזְּקוֹ מִוֹסְדֵי אֱרֶץ: 30 וֱאֶהְיֶה אֶצְלוֹ אָמִוֹן וֱאֶהְיֶה שְׁעֲשִׁעִים יָוֹם | יָוֹם ַמְשַׂחֶקֶת לְפָנְיו בְּכָל־עֵת: 31 מֲשַׂחֶקֶת בְּתֵבֵל אַרְצֶוֹ וְשַׁעֲשֵׁעֵׁי אֶת־ זַעַתָּה בָנִים שִׁמְעוּ־לֵי וְאַשִׁבִי דְרָכֵי 32 פ בני אדם: יִשְׁמְרוּ: 33 שִׁמְעָוּ מוּסָר וַחֲכָּמוּ וְאַל־תִּפְרֶעוּ: 34 אֵשְׁרֵי אָדָם שֹׁמֵעַ לִי לִשְׁקִּד עַל־דַּלְתֹתֵי יָוֹם | יָוֹם לִשְׁמֹר מְזוּזְת פְּתָחֵי: 35 בִּי מִצְאִי ַמַצֵאי חַײֵם וַיָּפֵק רָצוֹן מֵיָהוֶה: 36 וחֹטָאִי חֹמֵס נַפִּשָׁוֹ כָּל־מִשַּׁנִאַי

7:22 וֹבעֹבָס [ <וּבָעַבֹס M<sup>A</sup>] אַבָח M<sup>L</sup> (graph טָ → טָ M (vocal) || א וֹבעֹבָס M<sup>A</sup>) א אַבָח M (vocal) || מוֹסֵר ( έπὶ δεσμούς) S (באימהיא G (ἐπὶ δεσμούς) S (איל [ (בא מהיא G (ἐπὶ δεσμούς) S ( in G 23) S (הער in S 23) M אַויל (near dittog ' $\rightarrow$ '), context) || עד יפַלָּח הָץ כָּבֶדוֹ [ 12:3 kransp to end of 7:23 עד יפַלָּח הַץ כָבָדוֹ M init (displ; accents) || 8:10 מוסר G (דותא) S (הביגהאה) T<sup>L</sup> (מרדותא) ] מוּסָרָי MV (disciplinam meam)  $T^Z$  (מרדותי) (near dittog נבחר || (יו  $\leftarrow$  ו M מ + דבחר G (מילמוס (dittog בבחר  $\rightarrow$  נבחרבחר (נבחרבחר ) (3:13 ) אבחר יראת ... רע 13MGS (add)

וְעָקֵשׁ: 9 כָּלָם גֲכֹחִים לַמֵּבֶין וִישָׁרִים לְמִצְאֵי דֶעַת: 10 קְחְוּ־<sup>ר</sup>מוּסָר<sup>י</sup> וְאַל־הֶסֶף וְדַעַת מֵחָרְוּץ נִבְחֶר: 11 בִּי־טוֹבֶה חֲכְמָה מִפְּנִינֵים וְכָל־ ְּחֵפָּצִים לְא יִשְׁוּ־בֶה: 12 אֲנִי־חֶכְמָה שָׁכַנְתִּי עָרְמֶה וְדַעַת מְזִמָּוֹת אֶמְצֵא: 13 (יְרָאַת יְהוָה שֶׂנֹאת רֶע גֵּאָה וְגָאוֹן | וְדֶרֶךְ רָע) וּפִי תַהִפְּכוֹת שָׂנֵאתִי: 14 לִי־עֵצָה וִתוּשִׁיָה אַנִי בִינָה לִי גִבוּרֵה: 15 בֵּי

נִרְוֶה דְדִים עַד־הַבְּקֶר נִׁתְעַלְּסָׁה בָּאֲהָבִים: 19 כִּי אֵין הָאֵישׁ בְּבִיתִוֹ ֹהְלַךְ בְּדֶרֶהְ מֵרְחְוֹק: 20 צְרוֹר־הַכֶּסֶף לְקַח בְּיָדֵוֹ לְיוֹם הַבֵּסֶא יָבָא בֵיתִוֹ: 21 הֵטַתּוּ בְּרָב לִקְחֵה בְּחֵלֶק שְׁפָתֶׁיהָ תַּדִיחֶנּוּ: 22 הָוֹלֵד אַחֲדֶׁיהָ פִּֿתְאָם בֵּשׁוֹר אֶל־וֹטֶבַח' יְבָוֹא ווּכְעָכֹס' אֶל־וֹמוֹמֵר אַיֶּל'י נז רֹן הוּא יעַד יְפַלַח חֵץ רֹאָריָבָע בְּי־בְנַפְשָׁוֹ הוּא יעַד יְפַלַח חֵץ 23 נין אַמְרֵי־פִי: אַמְעוּ־לֵי וְהַקְשִׁיבוּ לְאָמְרֵי־פִי: כִּבֵדָוֹ־: 25 אַל־יֵשְׂטְ אֶל־דְּרָכֵיהָ לִבֶּדְ אַל־תֵׁתַע בִּנְתִיבוֹתֶיהָ: 26 בִּי־רַבִּים חַלָלִים הִפֵּילָה וַעֲצָמִים כָּל־הַרָגֵיהָ: 27 דַרְבֵי שְׁאוֹל בֵיתָה יֹרְדוֹת 8:1 הַלְאֹ־חְכְמֵה תִקְרֵא וֹתְבוּנְה תִּמֵן אַל־חַדָרֵי־מֶוֶת: קוֹלֶה: 2 בְּראשׁ־מְרוֹמֵים עֲלֵי־דֶרֶךְ בֵּית נְתִיבְוֹת נִצְּבָה: 3 לְיַד־ שְׁעָרֵים לְפִי־הֶֶרֶת מְבְוֹא פְתָחֵים תְּרְנָה: 4 אֲלֵיכֶם אִישִׁים אֶקְרָא ּוְקוֹלִי אֶל־בְּגֵי אָדֶם: 5 הָבִינוּ פְתָאיַם עָרְמֶה וּכְסִילִים הָבִינוּ לֵב: 6 שֵׁמְעוּ בְּי־נְגִידִים אֲדַבֶּר וּמִפְתַּח שְׁפָתֵׁי מֵישָׁרִים: *7* בִּי־אֱמֶת יֶהְגֶה חؚڿۣڹ ٳٮڗڹۑٟڿؚڔ ۛۻؘ۪ۏؚؠۣٙڹ ڕۻۣ۬ۘ۬ۘٮڒ۬: 8 ج۪ۑٚۑڗۄ ڿؚڂ۬؉ؚڡؚۭڍٮ۬ۦڿۣڹ אۣڹٳ ڂ۪ۑٛڡ ڹؚ؋ؚۺۣڬ

פ אִישׁ אֵשׁ בִּחֵיקָוֹ וֹבְגָדָיו לְא תִשָּׂרַפִנָה: 28 אָם־יִהַלֶּך אֱישׁ עַל־ הַגֶּחָלֵים וְרַגְלָיו לְא תִכְּוֵינָה: 29 בֵּן הַבָּא אֶל־אֵשֶׁת רֵעֵהוּ לְא יִנְאָה בִּל־הַנֹגֵע בֶּה: 30 לֹא־יָבְוּזוּ לֵגַנְּב בִּי יִגְנָוֹב לְמַלֵּא נַפְשׁוֹ בִּי יִרְעָב: גוּנִמְצָא יִשַׁלֵם שָׁבְעָתָיִם אֱת־כָּל־הָוֹן בֵּיתִוֹ יָתֵן: 32 נֹאָף אָשָׁה 31 ְחַסַר־לֵב מֵשְׁחֵית יַנְפְשׁוֹ הַוּא יַעֲשֶׂנָה: 33 גֶגַע־וְקָלָוֹן יִמְצֵא וְחֶרְפָּתוֹ לַא תִמֶּחֶה: 34 בִּי־קִנְאָה חֲמַת־גֶּבֶר וְלְא־יַחְמוֹל בְּיָוֹם נְקֶם: 35 לא־ ַיַשָּׂא פְּגֵי כָל־כִּפֶר וְלְא־יֹאבֶה כִּי תַרְבֶּה־שְׂחַד: ד:1 בְּנִי פ שְׁמִר אֲמֶרֶי וֹמִצְוֹתֵׁי תִּצְפְּן אָתֶדְ: 2 שְׁמִר מִצְוֹתֵי וֶחְיֶה וְתוֹרָתִי כָּאִי שוֹן עֵינֶידְ: 3 קָשְׁרֵם עַל־אֶצְבְּעֹתֶידְ בְּחָבֵם עַל־לְוּחַ לְבֶדְ: 4 אֱמְׂר לַחָכִמָה אֲחִתִי אֶתִּ וֹמֹדָע לַבִּינֵה תִקָרֵא: 5 לֵשִׁמָרִדְ מֵאִשֵׁה זָרָה מִנַּכְרִיָּה אֲמֵרֵיה הֶחֱלֵיקָה: 6 בֵּי בִּחַלָּוֹן בֵּיתֵי בִּעֲד אֲשִׁנַבֵּי נִשְׁקֵפְתִּי: ןאָרֶא בַפְּתָאיִם אָבִינָה בַבְּנִים נַעַר חֲסַר־לֵב: 8 עֹבֵר בֲשׁוּק אֵצֶל 7 פּנֶה וְדֶרֶך בֵּיתָה יִצְעֵד: 9 בְּנֵשֶׁף־רּבַּעַרְב' יָוֹם בְּאִישִׁוֹן לַיְלָה וַאָּפֵלֶה: וּהְגֵּה אֲשָׁה לִקְרָאתוֹ שֵׁית זוֹנָה וּנְצֵרַת לֵב: 11 הֹמִיֶה הֵיא 10 וְסֹרֶרֶת בְּבִיתָה לֹא־יִשְׁבְּנִוּ רַגְלֶיהָ: 12 פַּעַם | בַּחוּץ פַּעַם בְּרְחֹבִוֹת וְאֵצֵל כָּל־פִּגָּה תֶאֵרֹב: 13 וְהֵחֵזֶיקָה בִּוֹ וְגָשִׁקָה־לְוֹ הֵעֵזָה פְׁגֵיהָ וַתְּאֹמַר לְוֹ: 14 זִבְחֵי שְׁלָמֵים עָלֶי הֵיוֹם שִׁלַמְתִי נְדָרֵי: 15 עַל־בֵּן יַצֶאתִי לִקָרָאתֵדְ לִשַׁחֵר פָּגִידְ וָאֵמִצָאֵדָ: 16 מֵרִבַדִּים רָבַדְתִּי עַרִשִי וֹחַטָבוֹת אֵטוּן מִצְרֵיִם: 17 נֵפִתִּי מִשִּׁכָּבֵי מִר אֵֵהָלִים וִקּנָמוֹן: 18 לְכֵה

תְשִׁיתֶדּ: 23 בִּי גַר מֲצְוָה וְתוֹרָה אֵוֹר וְדֶרֶדְ חֵיִים תּוֹכְחְוֹת מוּסֵר: 24 לִשְׁמְרְדָ מַאֵשֶׁשֶׁת <sup>ו</sup>רֵעֶד<sup>ּ</sup> מֵחֶלְלֵת לְשָׁוֹן נְכְרִיֶה: 25 אַל־תַּחְמִד זְפְיָה בִּלְבָבֶדְ וְאַל־תִׁקְתַדְ בְּעַפְעַפֶּיהָ: 26 בִּי בְעַד־אִשְׁה זוֹנָה עַד־ כִּבַּר לְחֶם וְאַשָּׁת אֵישׁ גֶפֶשׁ יְקָרֶה תָצְוּד: פּ 27 הַיַחְתֶּה

בִּנִי וֶהִנָּצֵׁל כִּי בָאתָ בְכַף־רֵעֵדְ לֵדְ הִתְרַפֵּס וּרְהֵב רֵעֵידְ: 4 אַל־תִתַן שֵׁנֵה לְעֵינֶיָדְ וֹתְנוּמָה לְעַפְעַפֶּידְ: 5 הֵנְּצֵל כִּצְבִי 'מִמְּצִוֹד' וֹכְצִפֿוֹר 6 לֵדְ־אֵל־נִמָלֵה עַצֵּל רָאֵה דְרָכֵיהָ וַחֵכָם: מִיַּד יַקוּש: פ ד אַשֶׁר אֵין־לֶה קָצִין שֹׁטֵר וּמֹשֵׁל: 8 תָּכֵין בַּקַיָץ לַחָמֶה אָגְרֶה *7* בַקָּצִיר מַאַכְלֶה: 9 עַד־מָתַי עָאֵל | תִּשִׁבֶּב מְׁתִי תָּקוּם מִשִּׁנָתֶדְ: 10 מְעַט שֵׁנוֹת מְעַט תְּנוּמֵוֹת מְעַט | חִבֶּק יִדְיִם לִשְׁבְּב: 11 וּבָא־ 12 אָדֶם בְּלִיַעַל ּבִמְהַלֵּך רֵאשֶׁד וֹמַחְסְרְדָ בְּאֵישׁ מָגַן: ก אֵישׁ אָון הוֹלֵך עִקִשׁוּת פֵּה: 13 קֹרֵץ בֵּעֵינָו מֹלֵל בְּרַגְלָו מֹרֵה בּאֶצְבָּעֹתֵיו: 14 תַּהפָּכוֹת | בִּלְבוֹ חֹרֵשׁ רֶע בְּכָל־עֵת יֹמִדֹנֵים' יִשַׁלֵּח: נעל־בֵּן פֶּתָאֹם יָבְוֹא אֵידָוֹ פֶּתַע יִשָּׁבֵר וִאֵין מַרְפֵּא: וּשָׁש־הַנָּה שָׂנֵא יְהוֶה יְשָׁבַע וּתּוֹעֲבַת' נַפְּשִׁוֹ: 17 עֵינַיִם רָמוֹת 16 לְשׁוֹן שֶׁקֶר וְיָדַיִם שֹׁפְכָוֹת <sup>ר</sup>דַם<sup>-</sup>־נָקֵי: 18 לֵב חְרֵשׁ מַחְשְׁבְוֹת אָוֶן ַרַגְלַיִם מְמַהֲרוֹת לְרָוּץ לְרָעָה: 19 יָפִיַח בְּזָבִים עֵד שֶׁאֶר וּמְשַׁלֵח 20 נִצְׂר בְּנִי מִצְוֵת אָבִידְ וִאַל־תִּטֹש פ מְדָנִים בֵּין אַחִים: ּתּוֹרַת אָמֵּד: 21 קָשָׁרֵם עַל־לִבָּדְ תָמֵיד עָנִדֵׁם עַל־גַּרִגִּרֹתֵד 22 בִּהִתִהַלֶּכְדְּ | תַּנְחֶה אֹתָׁדְ בְּשָׁכְבְּדְ תִּשְׁמִׁר עָלֶיָדְ וֹהְקִיצׁוֹתָ הֵיא

3 יאַל תַּקְשִׁיב לְאֵשֶׁת אָוֶלֶת כִּי נְפֶת תֵּטֹפְנָה שִׂפְתֵי זְרָה וְחָלֶק מִשֶּׁמֶז חִבְּה: 4 וְאַחֲרִיתָה מְרָה כְלַעֲנָה חֵׁדָּה כְּחֶרֶב פִּיּוֹת: 5 רַגְלֶיהָ יִרְדְוֹת מֵוֵת שָׁאוֹל צִעָדֵיהָ יִתְמִכוּ: 6 אָרַח חֵיִים פֵּן־תִּפַלֵּס נָעָו וַעַתֶּה בָּנִים שִׁמְעוּ־לֵי וִאַל־תְּסוּרוּ 🛛 💿 מַעַגָּלֹהֵיהָ לְא הֵדֶע: ַמָאִמְרֵי־פֵי: 8 הַרְחֵק מֵעָלֵיהָ דַרְכֶּדְ וְאַל־תִּקְרֵב אֶל־פֶּתַח בֵּיתֶה: פ פּן־תִּתֵּן לַאֲחֵרֵים הוֹדֶך וֹּשְׁנֹתֶׁידְ לְאַכְזָרֵי: 10 פֶּן־יִשְׂבְּעָוּ זָרֵים כֹּחֶדְ ּיַנְצָבֶידְ בְּבֵית נְכְרֵי: 11 וְנָהַמְתֵּ בְאַחֲרִיתֶדְ בִּכְלִוֹת בְּשָׂרְדָ וּשְׁאֵרֶדְ: וןאָמַרְהָ אֵידְ שָׂגֵאתִי מוּסֵר וְתוֹכַּחַת נָאָץ לִבְּי: 13 וֱלֹא־שֶׁמַעָתִי 12 בְּקוֹל מוֹרֶי וְלְמְלַמְדִי לֹא־הִטִּיתִי אָזְנֵי: 14 בֵּמְעַט הָיֵיתִי בְכָל־רֶע בִּתְוֹדְ קָהֶל וְעֵדֶה: 15 שִׁתֵה־מֵיִם מִבּוֹתֵדְ וְנִזִלִים מִתְּוֹדְ בִּאֵתֵדְ: וּרָקָן יָפִוּצוּ מַעִיְנֹתֶידָ חָוּצָה בָּרְחֹבוֹת פַּלְגֵי־מֵיִם: 17 יְהִיוּ־לְדָ ּלְבַדֶּך וְאֵין לְזָרֵים אִתֶּד: 18 יְהִי־מְקוֹרְדֶ בְרֵוּד וּשְׂמַח מֵאֵשֶׁת נְעוּרֶד: 19 אַיֶּלֶת אֲהָבִים וְיַעֲלַת־תֵן דֵדֶיהָ יְרַוָּדְ בְכָל־עֵת בְאַהֲבָתָה תִּשְׁגֵה תְמֵיד: 20 וְלֵמָה תִשְׁגֵה בְגֵי בִזְרֶה וֹתִחַבֵּק חֵק נְכִרְיֶה: 21 כֵּי נֹכַח | עֵינֵי יֶהוֶה דַּרְבֵי־אָישׁ וֶכָל־מַעָגָלֹתָיו מִפַּלֵּס: 22 עַווֹנוֹתִיו יִלְבָדָנְוֹ י וּבְחַבְלֵי חַׁטָּאתוֹ יִתָּמֵד: 23 הוּא יֻמוּת בְּאֵין מוּסֶר וּבְרָב אִוּלְתָו : אָם־עָרַבְתָּ לְרֵעֶדְ תְּקַעְתָ לַזָּר <sup>ר</sup>בַפֶּדי 1:0 אַם־עָרַבְתָּ לְרֵעֶד ישגה: ַ נוֹקַשְׁתָּ בְאִמְרֵי־פֵּידָ נְלְבַּדְתָ בְּאִמְרֵי־פֵידָ: 3 עֲשֵׂה זָאת אֵפָוֹא |

דְּבָרֵי לְבֶּדְ שְׁמְׁר מִצְוֹתֵי ווֶחִיֵה: 5 קְנֵה חֲכְמָה קְנֵה בִינֶה אַל־תִּשְׁכַּח וְאַל־הֵט מֵאִמְרֵי־פִי: 6 אַל־תַּעַזְבֶהָ וְתִשְׁמְרֶדְ אֱהָבֶהָ וְתִצְּרֶדָ: 7 רֵא שִׁית חֲכְמָה קְגֵה חָכְמֵה וּבְכָל־לִקְנְיָנָדְׁ קְגֵה בִינֵה: 8 סַלְסְלֶהָ וּתְרוֹמְמֶדְ תְּכַבִּדְדְ בֵּי תְחַבְּקֶנָה: 9 תִּתֵּן לֱרֹאשְׁדְ לוְיַת־חֵן עֲשֶׁרֶת ּתִּפְאֶרֶת הְמַגְנֶדָ: 10 שְׁמַע בְּנִי וְקָח אֲמְרֵי וְיִרְבָּוּ לְדָׁ שְׁנִוֹת חַיִּים: 11 בְּדֶרֶך חֻכְמָה הֹבתִיף הִדְרַכְתִּיף בְּמַעְגְלֵי־יְשֶׁר: 12 בְּלֶכְתְּף לֹא־ ַיַצַר צַעֲדֶדְ וְאִם־תְּרוּץ לְא תִכְּשֵׁל: 13 הַחֲזֵק בַּמּוּסָר אַל־תֶּרֶף גִאָרֶה בִּי־הֵיא חַ״ֶידָ: 14 בְּאָרַח רֲשָׁעִים אַל־תָּבָא וְאַל־תְאַשֵּׁר בְּדֶרֶדְ רָעֵים: ַדַּרָעֵהוּ אַל־תַּעֲבָר־בָּוֹ שְׂטֵה מֵעָלָיו וַעֲבִוֹר: 16 כֵּי לָא יִשְׁנוּ אָם־ לַא יְרֵעוּ וְנִגְזְלָה שְׁנָתָם אִם־לְא יֹיַכְשִׁילוּ": 17 בִּי לָחֲמוּ לָחֶם רֶשַׁע וְיֵיָן חֲמָסֵים יִשְׁתְּוּ: 18 וְאַׁרַח אֲדִיקִים כְּאַוֹר גְּגַהּ הוֹלֵדְ וְאוֹר עַד־נְכָוֹן ַהַיִּוֹם: 19 דֶּרֶךְ רֲשָׁעִים בְּאֲפֵלֶה לְאׁ יְדְעוּ בַּמֶּה יִבְּשֵׁלוּ: פ ַמַעֵינָיָד 20 בְּנִי לִדְבָרֵי הַקְשֶׁיבָה לַאֲמָרֵי הַט־אָזְנֶד: 21 אַל־<sup>ר</sup>ִיְלְוּזוּ<sup>ז</sup> מֵעֵינֶיָד 20 שָׁמִרֵם בִּתְוֹדְ לִבְבֵדְ: 22 בְּיּדחַיֵּים הֵם לִמֹצְאֵיהֵם וְּלִכָל־בִּשָׂרְוֹ מַרִפֵּא: דַמַר מָמָד באָר לָבֶּד בִּי־מִמָּנוּ תּוֹצָאָוֹת חַיִים: 24 הָסַר מָמְד ּעִקְשִׁוּת פֶּה וּלְזָוּת שְׁפָתַיִם הַרְחֵק מִמֶּדָ: 25 עֵינֶידְ לְנְכַח יַבָּיטוּ ּוְעַפְעַפֶּידְ יַיְשֵׁרוּ נֶגְדֶדְ: 26 פַלֵּס מַעְגַל רַגְלֶך וְכָל־דְרָכֵידְ יִכְּנוּ: ַבְּנִי לְחָכְמָתִי 27 אֵל־תֵּט־יָמֵין וּשְׂמָאול הָסֵר רַגְלְדֶ מֵרֶע: 5:1 בְּנִי לְחָכְמָתִי ַהַקִשִׁיבַה לְתִבוּנַתִי הַט־אַזְגֵדָ: 2 לִשְׁמִר מִזְמָוֹת וְדֵעַת שִּׁפַתֵידָ יְנִצְרוּ:

<sup>4:4</sup> ארחות [ (א מצותי M ] מצותי G (לאיסאלג) (near haplog וי א וחיה || מצותי M ] א G (לאיסאלג) (near haplog וחיה || א מצותי M ] ארחות [ (חע, עבדאלא הסע, שהבלאי) M S (קנה חכמה קנה בינה 4:5 || G (theol) || 4:7 M S ] ארחות [ (דעבדין תיקלא) M S (שנות 100 M S (לארחות [ (דעבדין תיקלא) M S (לארחות וו 4:16 M) ארחות [ (לארחות וו 4:16 M) מיבשילו 16:16 M יבשילו 16:20 M מצור יבשולו (לארחות וו 4:16 שנות 16:20 M מצור יבשולו (M (graph וו אישנו ארחות וו ארחות וו ארחות וו ארחות וו ארחות וו ארחות ארחות וו ארחות ארחות מארחות (לארחות וו 4:20 מארחות וו 4:20 מארחות וו 4:20 מארחות וו ארחות הבינה מהבלאים ארחות ארחות (לארחות הבינה 16:20 מארחות וו 4:20 מארחות וו 4:20 מארחות ארחות ארחות היקלא) ארחות (לארחות וו 4:20 מארחות ארחות וו 4:20 מארחות היקלא) ארחות ארחות (לארחות הוו 4:20 מארחות וו 4:20 מארחות ארחות וו 4:20 מארחות ארחות ארחות ארחות היקלא ארחות וו 4:20 מארחות היקלא ארחות ארחות היקלא ארחות היקלא (לארחות 16:20 מארחות ארחות ארחות ארחות ארחות היקלא) ארחות ארחות היקלא (לארחות 1:20 מארחות ארחות ארחות ארחות ארחות ארחות ארחות היקלא ארחות ארחות ארחות היקלא ארחות ארחות ארחות ארחות ארחות היקלא ארחות היקלא ארחות אר

וְכָבְוֹד: 17 דְּרָכֵיהָ דַרְבֵי־גְׁעַם וְכָל־נְתָיבוֹתָיהָ שָׁלְוֹם: 18 עֵץ־חַיַּים 19 יֱהוָה בְּחָכְמֶה ָהֵיא לַמַּחֲזִיהֵים בֶּה וֲתֹמְכֵיהָ מְאָשֶׁר: 🔹 יֶסַד־אֶֶרֶץ כּוֹגֵן שָׁמַיִם בִּתְבוּגָה: 20 בְּדַעְתּוֹ תְּהוֹמָוֹת נִבְקֶעוּ וֹשְׁחָלִים יִרְעֵפוּ־טֵל: 21 בֵּנִי אַל־יָלֵזוּ מֵעֵינֵידְ נִאָר הְאָשִׁיָה וּמִזְמֵה: 22 וִיָהִיו חַיֵּים לְנַפְשֶׁדְ וְהֵז לְגַרְגְרֹתֶידְ: 23 אָז תֵּלֵדְ לְבֶטַח דַרְכֶּדְ וְרַגְלְדָ לָא תּגִּוֹף: 24 אָם־רּמֵשֵׁב' לְאֹ־תִפְחֻד וְׁשָׁכַבְהָָ וְעָרְבָה שְׁנָתֶד: 25 אַל־ הֵּיִרָא מִפַּחַד פִּתְאֶׁם וּמִשֹׁאַת רְשָׁעִים כֵּי תָבְא: 26 כִּי־יֶהוָה יִהְיֶה בְכִסְלֶֶדְ וְשָׁמֵר רַגְלְדָ מִלְּכֶד: 27 אַל־תִּמְנַע־טִוֹב מִבְּעָלֶיו בִּהְיוֹת לְאֵל יּיְדְדְּ' לַעֲשִׂוֹת: 28 אַל־תּאמֵר יּלְרֵעֲדָ'' לֵדְ וָשׁוּב וּמְחָר אֶהֵׁן וְיֵשׁ :אָתֶד: 29 אַל־תַחֲרָשׁ עַל־רַעֲדָ רָעֶה וְהוּא־יוֹשֵׁב לְבֶטַח אִתֶּד 30 אַל־רְּתָרִיב<sup>ּ</sup> עִם־אָדֶם חִנָּם אִם־לְא גְמְלְדֶ רְעֵה: 31 אַל־הֶקַנֵּא ַבְּאֵישׁ חָמֱס וְאַל־תִּבְחַר בְּכָל־דְּרָבֶיו: 32 בִּי תוֹעֲבַת יְהוֶה נְלֵוֹז וְאֶת־ יְשָׁרֵים סוֹדִוֹ: 33 מְאֵרַת יֶהוָה בְּבֵית רְשָׁע וּנְוֵה צַדִּיקֵים יְבָרֵדָ אם־לַלַּצִים הוּא־יָלָיץ וֹןלַעֲנִיִים יִשָּן־חֵן: 35 בָּבוֹד חֲכָמֵים יִנְחֻלוּ 34 וּכְסִילִים <sup>ר</sup>מְרִימִים קָלָוֹן: אַ אַמְעָוּ בֶגִים מְוּסַר אָב ּןְהַקְשִּׁיבוּ לְדָעַת בִּיגֶה: 2 בִּי לֶלֵח טוֹב נְתַתִּי לְכֶם תִוֹרָתִי אֵל־תַּעֲזְבוּ: ז כִּי־בֵן הַיֵיתִי לָאָבֵי רֵדְ וְיָחִיד לִפְגֵי אִמֵי: 4 וַיֹּרֵינִי וַיְּאמֵר לִי יֶתִמָדְ־

3:16 fin ] + 2 stichoi G (16a) (elab) || 3:20 איזלי M G (ἐρρύησαν) S ( $(\infty)$ ] ( $(\infty)$ ) M<sup>MSS</sup> (metath  $(-\infty)$ ) || 3:21 איזלי M ] איזלי G (παραρρυῆς) (metath  $(\infty)$ ) || 3:22 fin M ] + 2 stichoi G (22a) (assim to 3:8) || 3:24 השב G ( $(\alpha d \theta \eta)$ ) SyrH ( $(-\infty)$ ) M S ( $(-\infty)$ ) (near dittog -- C ( $(\alpha d \theta \eta)$ ) SyrH ( $(-\infty)$ ) M S ( $(-\infty)$ ) (near dittog -- C ( $(\alpha d \theta \eta)$ ) SyrH ( $(-\infty)$ ) (

2:18 אחה M ] (נתיבתה G (ਫੱθετο)  $\sigma'$  (ωρισεν) (graph  $\rightarrow$  ח ) ( $\mu \rightarrow \pi$ מות נת MG (דא איאא מידאָכ) א האלס) א מות נת (באאה) (near haplog מות  $\rightarrow$  מות  $\rightarrow$  מות אמות לדא, metath  $M \approx S$  (בי $\rightarrow$ יב) || 2:19 ארחות חיים M  $\approx S$  (אונות חיים G) אירחות חיים  $M \approx S$ (פֿעומטדשט (מאָר (assim) || 2:22 יָסָחוֹ CG, Bab MS Ba ] יָסָחוֹ (vocal, gram) || 3:3 רתבם על-לוח לבך || M G S (add) || כתבם על-לוח לבד M S MS דעהו 3:6 || (۲∩ך G (homoi )| ) حد لمتنه الحرم) (הבעהו [ (גבית, G (אָמָטאָץ) (near dittog הדעהו [ הד $\leftarrow$  ד stichos) || + 1 stichos G (later addition) || 3:8 לְשָׁרֵדְ G (τῷ σώματί σου) S (ארסט ) [ (גרסט ) [ לשָׁרֵד  $M \text{ (vocal) } \parallel 3:15}$  מפנינים  $M^{Q}$  מפניים  $M^{K} \text{ (homoi ) } \parallel 2$ stichoi G (15bc) || הרפצים א G (ד(עסנאט)  $\approx$  S (כנות) V (quae desiderantur) T (כ → מ M (graph חֵפַצִיד [ (מדעם)

וּמֵרֵאשִׁית כָּל־תְבוּאָתֶדְ: 10 וְיִמָּלְאָוּ אֲסָמֶידָ שָׂבֶע וְתִירוֹשׁ יְקָבֵידָ 11 מוּסַר יֲהוָה בְּנִי אַל־תִּמְאֶס וְאַל־תְּקׂץ יפרצו: פ בְּתוֹכַחְתִּוֹ: 12 בֵּי אֶת אֲשֶׁר יֶאֶהָב יְהוֶה יוֹכֵיח וֹכִאָב אֶת־בֵּן יִרְצֶה: ז אַשְׁרֵי אָדָם מָצָא חָכְמֶה וְאָדָם יָפִיק הְבוּגָה: 14 בִּי טָוֹב סַחְרָה מִפְּתִיגֵים' וְכָל־ מִפְתִיגֵים' וְכָל־ מִפְתִיגֵים' וְכָל־ 'וֹחֵפָּצִים' לְא יֵשִׁוּ־בֵה: 16 אָרֵך יֵמִים בֵּימִינֵה בְּשִׂמאולָה עָשֵׁר

וְאֶת־בְּרֵית אֱלֹהֵיהָ שְׁבֵחָה: 18 בִּי שְׁחָה אֶל־מָוֶת <sup>ו</sup>נְתִיבָתֶה<sup>ּז</sup> וְאֶל־ רְפָאִים מַעְגְלֹתֶיהָ: 19 כָּל־הָָאֶיהָ לְא יְשׁוּבְוּן וְלְא־יֵשִׂיגוּ אָרְחָוֹת חַיֵּים: 20 לִמַעַן תֵּלֵך בִּדֵרֵך טוֹבִים וִאָרְחוֹת צַדִּיקִים תִּשִׁמִר: 21 כִּי־ יִשָּׁרִים יִשִׁבְּנוּ־אָרֵץ וֹתִמִימִים יוּוֶתִרוּ בֶה: 22 וְרִשָּׁעִים מֵאֶרֵץ יִבְּרֵתוּ ז:1 בֵּנִי תּוֹרָתֵי אַל־תִּשָּׁבָּח 3:1 פ וֹבוֹגִדִׁים יּיָסְחְוּ מִמֵּנָה: וּמִצְוֹתִי יִצְּר לְבֶּדְ: 2 בִּי אֵׁרֶדְ יָמִים וּשְׁנָוֹת חַיֶּיֶם וְשְׁלוֹם יוֹסֵיפּוּ לֶדְ: : (תֶּסֶד וֶאֶֶמֶׁת אֵל־יַעַֿזְבָדָ) קַשְׁרֵם עַל־גַּרְגְרוֹתֶיִדְ בְּתְבֵם עַל־לָוּחַ לִבָּדָ: : וּמְצָא־חֵז וְשֵׂכֶל־טָוֹב בְּעֵינֵי אֱלֹהֵים וְאָדֵם 5 בִּטֵח אֵל־ פ יֵהוָה בִּכָל־לְבֵּדְ וְאֵל־בֵּינָתִדְׁ אַל־תִּשָּׁעֵן: 6 בִּכָל־דִּרְבֵידְ דָעֵהוּ וְהוּא ּיִיַשֵּׁר אִׂרְחֹתֶידָ: 7 אַל־תְּהֵי חָבֶם בְּעֵינֶיָדְ יְרָא אֶת־יְהוָה וְסַוּר מֵרֶע: אַרַפְאוּת הְהֵי 'לְשֵׁרֶדְ' וְשִׁקֿוּי לְעַצְמוֹתֶידְ: 9 בַּבֵּד אֶת־יְהוָה מֵהוֹנֶדְ 8 רַפְאוּת הַמָי

25 וַתִּפְרְעָוּ כָל־עֲצָתֶי וְתוֹכַחְתִּי לְא אֲבִיתֶם: 26 גַּם־אֲנִי בְּאֵידְכֶם אֶשְׂחֶק אֶׁלְעַׁג בְּבְא פַחִדְכֶם: 27 בְּבָא רְכָשוֹאָה' | פַּחִדְכָם וֱאֵידְכֶם ּכְּסוּפָה יֶאֱתֶה בְּבָא עֲלֵיכֶּם צְרָה וְצוּקֶה: 28 אָז יֵקְרָאֻנְנִי וְלָא אֶעֱגֶה יִשַׁחַרָּגִי וְלָא יִמְצָאֶנִי: 29 הַחַת כִּי־שָׂנָאוּ דֶעַת וְיִרָאָת יְהוֹה לְא בְחֶרוּ: 30 לֹא־אָבְוּ לַעֲצָתֵי נְאֲצֹוּ כָּל־תּוֹכַחְתֵי: 31 וְיֹאכְלוּ מִפְּרֵי דַרְבֶּם וְּמִמִעֲצְׁתֵיהֶם יִשְׂבֶּעוּ: 32 בִּי מְשׁוּבַת פְּתָיֵם תֵּהַרְגֵם וְשִׁלְוַת ּכְּסִילִים הְאַבְּדֵם: 33 וְשֹׁמֵעַ לֵי יִשְׁכָּן־בֶּטַח וְשַׁאֲנַן מִפָּחַד :בּגִי אִם־תַּקַּח אֲמָרֶי וֹמִצְוֹתֵׁי תִּצְפָׂן אָתֶד: :רַעָה פ 2 לְהַקְשֵׁיב לֵחְרְמֵה אָזְגֵדְ תַּשֶּׁה לִבְּדְׁ לַתְּבוּגָה: 3 כִּי אֵם לַבִּינָה תִקְרֶא לַתְּבוּנָה תִּתֵּן קוֹלֶד: 4 אִם־תְבַקְשֶׁנָה כַכֵּסֶף וְכַמַּטְמוֹנִים תַּחְפְּשֶׂנָה: 5 אָׁז תָּבִין יִרְאַת יְהוֶה וְדַעַת אֱלֹהִים תִּמְצֵא: 6 בְּי־יֶהוָה יִתֵּן חְכְמֶה מִׁפִּׁיו דַּעַת וּתְבוּגָה: 7 רוְצָפַן' לֵיְשָׁרִים תּוּשִׁיֶה מְגֹן לְהָלְבֵי ּתְּם: 8 לֵנְצֹר אָרְחוֹת מִשְׁפֶּט וְדֶרֶךְ חֲסִידֵו יִשְׁמְר: 9 אָׁז הָּבִין צֶדֶק וּמִשְׁפֵּט וּמֵישָׁרִים כָּל־מַעְגַּל־טִוֹב: 10 בִּי־תָבִוֹא חָכְמָה בְלָבֶּדְ וְדַעַת לְנַפְשְׁךָּ יִנְעֵם: 11 הֱזִמְה תִּשְׁמִר עָלֶידְ תְבוּנֶה תִנְצְרֶכְּה: 12 לֱהַצִּילְדָ מַדֶּרֶדְ רֶע מֵאִישׁ מְדַבֵּר תַּהְפָּכְוֹת: 13 הַעִזְבִים אָרְחוֹת יָשֶׁר לְלָכֶת בְּדַרְכֵי־חְשֶׁדְ: 14 הֲשְׂמֵחִים לַעֲשְׂוֹת רֶע יְגִילוּ בְּתַהְפָּכְוֹת רֱע: ַדַן אָשֶׁר אָרְחֹתֵיהֶם עִקְשֵׁים וֹּנְלוֹזִים בְּמַעְגְלוֹתֶם: 16 לְהַצֵּילְד מֵאִשֵׁה זָרֶה מִנְכִריָה אֵמֶרֵיהָ הֵחֵלְיקָה: 17 הַעֹזֶבֶת אַלְוּף נְעוּרֵיהָ

ַמִשְׁלֵי שְׁלֹמְה בֶּן־דְוָד מֶלֶך יִשְׂרָאֵל: 1:1 2 לְדַעַת חָכְמֵה וּמוּסֵר לְהָבִין אִמְרֵי בִינֶה: 3 לְקַחַת מוּסַר הַשְׂבֵּל ַצֶּדֶק וֹמִשְׁפָּט וּמֵישָׁרֵים: 4 לָתֵת לִפְתָאיֵם עָרְמֶה לְנַעַר דֵּעַת וּמְזִמֶה: 5 יִשְׁמַע חֲכָם וְיִוֹסֶף לֶקַח וְנָבוֹן תַּחְבָּלִוֹת יִקְנֶה: 6 לְהָבִין מֲשָׁל וּמְלִיצֶה דִּבְרֵי חֲכָמִים וְחִידֹתֶם: 7 יִרְאַת יֶהוֶה רֵאשִׁית דֶעַת חְכָמֶה וֹמוּסָר אֶוִילִים בָּזוּ: פּ 8 שְׁמַע בְּנִי מוּסַר אָבֵידְ וְאַל־תִּטֹשׁ תּוֹרֵת אָמֶדְ: 9 כֵּי | לִוְיָת הֵׁן הֵם לְרֹאשֶׁך וַעֲנָקִים לְגַרְגְרֹתֶידְ: 10 בְּנִי אָם־יְפַתוּדְ חַׁטָּאִים אַל־תֹּבֵא: 11 אָם־יאֹמְרוּ לְכֶה אָׁתֶּנוּ נֶאֶרְבָה לְדֶם נִצְּפְגָה לְנָקֵי חִגְּם: 12 גַבְלָעֵם כִּשְׁאַוֹל חַיִּיֶם וּתְמִימִים כְּיָוֹרְדֵי בְוֹר: 13 כְּל־הַוֹן יֵקָר נִמְצֵא נְמַלֵּא בְתֵּינוּ שְׁלֵל: 14 גָוֹרָלְדְ תַפֵּיל בְּתוֹבֵנוּ בִּיס אֶׁחָׁד יִהְיֶה לְכֵלֶנוּ: 15 בְּנִי אַל־תֵּלֵך בְּדֶרֶך אִתָּם מְנַע ַרַגְלְדָׁ מִנְתִיבָתֶם: 16 בִּי רַגְלֵיהֶם לְרַע יָרָוּצוּ וִימַהֲרוּ לִשְׁפָּדְ־דֶם: ן הֵי־חֶנָּם מְזֹרֶה הָרֶשֶׁת בְּעֵינֵי כָל־בִּעַל כְּגֵף: 18 וְהֵם לְדָמֶם יֶאֱרֶבוּ 17 יִצְפְנוּ לְנַפְשֹׁתֵם: 19 בֵּן אֲרְחוֹת כָּל־בִּצֵע בֶּצַע אֶת־גֶפָשׁ בְּעָלָיו 20 הַכָמוֹת בַּחוּץ תָּרְנָה בָּרִחֹבוֹת תִּתֵּן קוֹלָה: יַקָּח: פ בז בְּרָאשׁ הֹמִיּוֹת הִֿקְרָא בְּפִתְחֵי שְׁעָרִים בָּעִיר אֲמְוֶרִיהָ תאֹמֵר: 22 עַד־מְתַי | פְּתָים װְאֵהְבֿוּ פֶתִי וְלֵצִים לָצוֹן חָמְדַוּ לְהֶם וֹרְסִילִים יִשְׂנְאוּ־דֶעַת: 23 תְּשׁוּבוּ לֱתֿוֹכַחְתֵּי הִנֵּה אַבִּיעָה לְכֶם רוּחֵי אוֹדֵיעָה ִדְבָרֵי אֶתְכֵם: 24 יַעַן קָרָאתִי וַתְּמָאֵנוּ נָטְיתִי יָדִי וְאֵין מַקִשִׁיב:

משלי

Michael V. Fox

The Hebrew Bible: A Critical Edition